ACC/AHA PRACTICE GUIDELINES—FULL TEXT

ACC/AHA Guidelines for the Management of Patients With ST-Elevation Myocardial Infarction

A Report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the 1999 Guidelines for the Management of Patients With Acute Myocardial Infarction)

Developed in Collaboration With the Canadian Cardiovascular Society

WRITING COMMITTEE MEMBERS

Elliott M. Antman, MD, FACC, FAHA, Chair

Daniel T. Anbe, MD, FACC, FAHA Paul Wayne Armstrong, MD, FACC, FAHA Eric R. Bates, MD, FACC, FAHA Lee A. Green, MD, MPH Mary Hand, MSPH, RN, FAHA Judith S. Hochman, MD, FACC, FAHA Harlan M. Krumholz, MD, FACC, FAHA Frederick G. Kushner, MD, FACC, FAHA Gervasio A. Lamas, MD, FACC Charles J. Mullany, MB, MS, FACC Joseph P. Ornato, MD, FACC, FAHA David L. Pearle, MD, FACC, FAHA Michael A. Sloan, MD, FACC Sidney C. Smith, Jr., MD, FACC, FAHA

TASK FORCE MEMBERS

Elliott M. Antman, MD, FACC, FAHA, *Chair* Sidney C. Smith, Jr., MD, FACC, FAHA, *Vice Chair*

Joseph S. Alpert, MD, FACC, FAHA* Jeffrey L. Anderson, MD, FACC, FAHA David P. Faxon, MD, FACC, FAHA Valentin Fuster, MD, PhD, FACC, FAHA Raymond J. Gibbons, MD, FACC, FAHA*† Gabriel Gregoratos, MD, FACC, FAHA* Jonathan L. Halperin, MD, FACC, FAHA Loren F. Hiratzka, MD, FACC, FAHA Sharon Ann Hunt, MD, FACC, FAHA Alice K. Jacobs, MD, FACC, FAHA

Joseph P. Ornato, MD, FACC, FAHA

*Former Task Force Member †Immediate Past Chair

This document was approved by the American College of Cardiology Foundation Board of Trustees on May 7, 2004 and by the American Heart Association Science Advisory and Coordinating Committee on May 5, 2004.

The ACC/AHA Task Force on Practice Guidelines makes every effort to avoid any actual or potential conflicts of interest that might arise as a result of an outside relationship or personal interest of a member of the writing panel. Specifically, all members of the writing panel are asked to provide disclosure statements of all such relationships that might be perceived as real or potential conflicts of interest. These statements are reviewed by the parent task force, reported orally to all members of the writing panel at the first meeting, and updated and reviewed by the writing committee as changes occur. See Appendix 1 for relationship with industry information for writing committee members; see Appendix 2 for peer reviewer names and relationships with industry for this document.

When citing this document, the American College of Cardiology Foundation and the American Heart Association would appreciate the following citation format: Antman EM, Anbe DT, Armstrong PW, Bates ER, Green LA, Hand M, Hochman JS, Krumholz HM, Kushner FG, Lamas GA, Mullany CJ, Ornato JP, Pearle DL, Sloan MA, Smith SC Jr. ACC/AHA guidelines for the management of patients with ST-elevation myocardial infarction: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the 1999 Guidelines for the Management of Patients With Acute Myocardial Infarction). *Circulation*. 2004;110;e82-e293.

Copies: This full-text guideline and its executive summary are available on the World Wide Web sites of the American College of Cardiology (www.acc.org) and the American Heart Association (www.americanheart.org). Single copies of the executive summary published in the August 4, 2004, issue of the Journal of the American College of Cardiology or the August 3, 2004, issue of Circulation or the companion full-text guideline are available for \$10.00 each by calling 1-800-253-4636 or writing to the American College of Cardiology Foundation, Resource Center, 9111 Old Georgetown Road, Bethesda, MD 20814-1699. To purchase bulk reprints (specify version and reprint number—71-0294 for the executive summary; 71-0293 for the full-text guideline): up to 999 copies, call 1-800-611-6083 (US only) or fax 413-665-2671; 1000 or more copies, call 214-706-1789, fax 214-691-6342, or e-mail pubauth@heart.org.

Permissions: Multiple copies, modification, alteration, enhancement, and/or distribution of this document are not permitted without the express permission of the American College of Cardiology Foundation. Please direct requests to copyright_permissions@acc.org.

1. Preamble
2. Pathology
 Management Before STEMI
4. Onset of STEMI
5. Prehospital Issues
 6. Initial Recognition and Management in the Emergency Department
6.2.2. Physical Examination
6.2.7. Global Risk Assessment Tools
6.3.1.5. Beta-Blockerse118 6.3.1.6. Reperfusione119 6.3.1.6.1. General Conceptse119 6.3.1.6.2. Selection of Reperfusion Strategye120 6.3.1.6.3. Pharmacological Reperfusion.e126 6.3.1.6.4. Percutaneous Coronary Intervention e136
6.3.1.6.5. Acute Surgical Reperfusione130 6.3.1.6.6. Patients With STEMI Not Receiving Reperfusione148 6.3.1.6.7. Assessment of Reperfusione148 6.3.1.6.8. Ancillary Therapye149 6.3.1.6.9. Other Pharmacological Measurese158
7. Hospital Managemente163 7.1. Locatione163

7.1.1. Coronary Care Unitel6	53
7.1.1.1 Monitoring and Treatment for Adverse	
Figure Fi	54
7.1.2 Standown Unit	55 5
7.2. Early Canaral Massuras	55
7.2. Early, General Measureserc	5
7.2.1. Level of Activityel6	5
7.2.2. Diete16	56
7.2.3. Patient Education in the Hospital Settingel	67
7.2.4. Analgesia/Anxiolyticsel	68
7.3. Risk Stratification During Early Hospital Coursee1	69
7.4. Medication Assessment	70
7 4 1 Beta-Blockers	70
7.1.2 Nitroglycerin	70
7.4.2. Inhibition of the Donin Angiotanoin Aldesterone	/0
7.4.5. IIIIII0III0II 0I IIIE Keliiii-Aligiotelisiii-Aldosteloite	71
Systeme1	/1
7.4.4. Antiplateletsel	12
7.4.5. Antithromboticse17	13
7.4.6. Oxygene1	73
7.5. Estimation of Infarct Sizeel	74
7.5.1. Electrocardiogram Techniquese1	74
7.5.2. Cardiac Biomarker Methods	74
753 Radionuclide Imaging	74
7.5.1 Echocardiography	7/
7.5.4. Echocardiographyer	74 74
7.5.5. Wagnetic Resonance Imaginger	74
7.6. Hemodynamic Disturbancesel	/4
7.6.1. Hemodynamic Assessmentel	/4
7.6.2. Hypotensionel	76
7.6.3. Low-Output Statee1	76
7.6.4. Pulmonary Congestionel	77
7.6.5. Cardiogenic Shocke1	79
7.6.6. Right Ventricular Infarction	31
7.6.7 Mechanical Causes of Heart Failure/Low-	
Output Syndrome	85
7671 Diagnosis	25
7.6.7.2 Mitral Value Deguratorian)) =
7.6.7.2. Wintral valve Regurgitation	55
7.6.7.3. ventricular Septal Rupture After	
STEMIelt	36
7.6.7.4. Left Ventricular Free-Wall Rupturee18	37
7.6.7.5. Left Ventricular Aneurysme18	57
7.6.7.6. Mechanical Support of the Failing	
Hearte18	7
7.6.7.6.1. Intra-Aortic Balloon Counter-	
nulsation e15	37
7677 Cardiac Transplantation After STEML e1	88
7.7 Arrhythmias After STEMI	8
7.7.1 Ventricular Arrhythmics	0 20
7.7.1.1. Ventricular Armyunniaserc	0
7.7.1.2 View in the Technic in the second se	ð
7.7.1.2. Ventricular Tachycardiae19	10
7.7.1.3. Ventricular Premature Beats)1
7.7.1.4. Accelerated Idioventricular Rhythms and	1
Accelerated Junctional Rhythmse19)1
7.7.1.5. Implantable Cardioverter Defibrillator	
Implantation in Patients After STEMI.e19	12
7.7.2. Supraventricular Arrhythmia/AEe19)5
773 Bradvarrhythmias	07
7731 Acute Treatment of Conduction	,
Disturbances and Draduorrhythmics 200	1
77211 Vantei and Drauyarriyuniniase20	/1 \1
7.7.3.1.1. ventricular Asystole	11
1.1.3.2. Use of Permanent Pacemakers	11
7.7.3.2.1. Permanent Pacing for Brady-	
cardia or Conduction Blocks	
Associated With STEMIe20)1
7.7.3.2.2. Sinus Node Dysfunction After	
STEMIe20	3

77323 Pacing Mode Selection in	
Patients With STEMI	e203
7.8 Recurrent Chest Pain After STEMI	e203
7.8.1 Pericarditis	e204
7.8.2. Recurrent Ischemia/Infarction	e205
7.9 Other Complications	e207
7.9.1 Ischemic Stroke	e207
7.9.2 Deep Venous Thrombosis and Pulmonary	.0207
Embolism	e208
7.10 Coronary Artery Bypass Graft Surgery After STEM	.0200 Le209
7.10.1. Timing of Surgery	e209
7 10 2 Arterial Grafting	e210
7 10 3 Coronary Artery Bypass Graft Surgery Afte	r
Fibrinolytic Therapy	e210
7.10.4. Coronary Artery Bypass Graft Surgery for	
Recurrent Ischemia After STEMI	.e210
7.10.5. Case Selection Concerns in CABG After	
STEMI	.e210
7.10.6. Elective CABG After STEMI in Patients W	ith
Angina	.e211
7.10.7. Coronary Artery Bypass Surgery After STE	MI
and Antiplatelet Agents	.e211
7.11. Convalescence, Discharge, and Post-MI Care	.e212
7.11.1. Risk Stratification at Hospital Discharge	.e212
7.11.1.1. Role of Exercise Testing	.e215
7.11.1.2. Role of Echocardiography	.e216
7.11.1.3. Exercise Myocardial Perfusion	
Imaging	.e217
7.11.1.4. Left Ventricular Function	e217
7.11.1.5. Myocardial Viability	.e218
7.11.1.6. Invasive Evaluation	e218
7.11.1.7. Ambulatory ECG Monitoring for	
Ischemia	.e219
Ischemia	.e219
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias	.e219 e219
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention	.e219 e219 .e220
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge	.e219 .e219 .e220 .e221
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management	.e219 e219 .e220 .e221 .e223
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management	.e219 .e219 .e220 .e221 .e223 .e225
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation	.e219 .e220 .e221 .e223 .e225 .e225
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy	e219 e219 .e220 .e221 .e223 e225 .e225 .e226
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin-	e219 e219 .e220 .e221 .e223 e225 .e225 .e226
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System	e219 e219 .e220 .e221 .e223 e225 .e225 .e226 .e227
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers	e219 e219 .e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control	.e219 .e220 .e221 .e223 .e225 .e225 .e226 .e227 .e227 .e229 .e229
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management	.e219 e219 .e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy	.e219 e219 .e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy	.e219 e219 .e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity	.e219 e219 .e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants	.e219 e219 .e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 e234
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management.	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 e234 e235
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8. 1. Psychosocial Impact of STEMI	e219 e219 .e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e230 .e230 .e231 e234 e235 .e235 .e235
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8.1. Psychosocial Impact of STEMI 8.2. Cardiac Rebabilitation	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e230 .e230 .e231 e234 .e235 .e235 .e235
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8.1. Psychosocial Impact of STEMI 8.2. Cardiac Rehabilitation 8.3. Follow-Un Visit With Medical Provider	e219 e219 .e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e230 .e230 .e231 e234 e235 .e235 .e235 .e235 .e236
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8.1. Psychosocial Impact of STEMI 8.2. Cardiac Rehabilitation 8.4. Return to Work and Disability	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e225 .e226 .e227 .e229 .e230 .e230 .e230 .e231 e234 e235 .e235 .e235 .e235 .e236 .e237 .e239
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8.1. Psychosocial Impact of STEMI 8.2. Cardiac Rehabilitation 8.4. Return to Work and Disability 8.5. Other Activities	e219 e219 e220 .e221 .e223 .e225 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias 7.12. Secondary Prevention 7.12.1. Patient Education Before Discharge 7.12.2. Lipid Management 7.12.3. Weight Management 7.12.4. Smoking Cessation 7.12.5. Antiplatelet Therapy 7.12.6. Inhibition of the Renin-Angiotensin- Aldosterone System 7.12.7. Beta-Blockers 7.12.8. Blood Pressure Control 7.12.9. Diabetes Management 7.12.10. Hormone Therapy 7.12.11. Warfarin Therapy 7.12.12. Physical Activity 7.12.13. Antioxidants 8. Long-Term Management 8.1. Psychosocial Impact of STEMI 8.2. Cardiac Rehabilitation 8.3. Follow-Up Visit With Medical Provider 8.4. Return to Work and Disability 8.5. Other Activities	e219 e219 e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239
Ischemia	e219 e219 e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e239
Ischemia	e219 e219 e220 .e221 .e223 .e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e239 .e239
Ischemia 7.11.1.8. Assessment of Ventricular Arrhythmias	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e231 e234 e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e236 .e237 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e237 .e230 .e230 .e230 .e230 .e230 .e230 .e229 .e230 .e230 .e230 .e230 .e229 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e231 .e235 .e235 .e236 .e237 .e239 .e230 .e231 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e230 .e237 .e236 .e237 .e236 .e237 .e236 .e237 .e236 .e237 .e236 .e237 .e237 .e236 .e237 .e236 .e237 .e236 .e237 .e236 .e237 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e239 .e236 .e237 .e239 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239
Ischemia	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e229 .e230 .e230 .e230 .e231 e234 e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e234 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e229 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e231 .e235 .e235 .e236 .e237 .e239 .e230 .e231 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e230 .e231 .e237 .e239 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e236 .e237 .e239 .e239 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239
Ischemia	e219 e219 e220 .e221 .e223 e225 .e225 .e225 .e226 .e227 .e229 .e230 .e230 .e230 .e231 e234 .e235 .e235 .e235 .e235 .e235 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e231 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e230 .e231 .e236 .e237 .e239 .e230 .e230 .e230 .e231 .e236 .e237 .e239 .e230 .e230 .e231 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e230 .e231 .e237 .e239 .e239 .e239 .e236 .e235 .e235 .e235 .e236 .e237 .e239 .e239 .e239 .e236 .e237 .e237 .e239 .e236 .e237 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e236 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e235 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239 .e239

1. PREAMBLE

It is important that the medical profession play a significant role in critically evaluating the use of diagnostic procedures and therapies in the detection, management, or prevention of disease states. Rigorous and expert analysis of the available data documenting relative benefits and risks of those procedures and therapies can produce helpful guidelines that improve the effectiveness of care, optimize patient outcomes, and generally have a favorable impact on the overall cost of care by focusing resources on the most effective strategies.

The American College of Cardiology (ACC) and the American Heart Association (AHA) have jointly engaged in the production of such guidelines in the area of cardiovascular disease since 1980. This effort is directed by the ACC/AHA Task Force on Practice Guidelines, whose charge is to develop and revise practice guidelines for important cardiovascular diseases and procedures. Experts in the subject under consideration are selected from both organizations to examine subject-specific data and write guidelines. The process includes additional representatives from other medical practitioner and specialty groups where appropriate. Writing groups are specifically charged to perform a formal literature review, weigh the strength of evidence for or against a particular treatment or procedure, and include estimates of expected health outcomes where data exist. Patientspecific modifiers, comorbidities and issues of patient preference that might influence the choice of particular tests or therapies are considered as well as frequency of follow-up. When available, information from studies on cost will be considered, however review of data on efficacy and clinical outcomes will be the primary basis for preparing recommendations in these guidelines.

The ACC/AHA Task Force on Practice Guidelines makes every effort to avoid any actual or potential conflicts of interest that might arise as a result of an outside relationship or personal interest of a member of the writing panel. Specifically, all members of the writing panel are asked to provide disclosure statements of all such relationships that might be perceived as real or potential conflicts of interest. These statements are reviewed by the parent task force, reported orally to all members of the writing panel at the first meeting, and updated and reviewed by the writing committee as changes occur.

These practice guidelines are intended to assist healthcare providers in clinical decision making by describing a range of generally acceptable approaches for the diagnosis, management, and prevention of specific diseases or conditions. These guidelines attempt to define practices that meet the needs of most patients in most circumstances. These guideline recommendations reflect a consensus of expert opinion after a thorough review of the available, current scientific evidence and are intended to improve patient care. If these guidelines are used as the basis for regulatory/payer decisions, the ultimate goal is quality of care and serving the patient's best interests. The ultimate judgment regarding care of a particular patient must be made by the healthcare provider and patient in light of all of the circumstances presented by that patient. There are circumstances where deviations from these guidelines are appropriate.

The executive summary and recommendations are published in the August 4, 2004, issue of the Journal of the American College of Cardiology and August 3, 2004, issue of Circulation. The full text is published on the ACC and AHA World Wide Web sites. Copies of the full text and the executive summary are available from both organizations.

Elliott M. Antman, MD, FACC, FAHA Chair, ACC/AHA Task Force on Practice Guidelines

1.1. Introduction

The process of guideline development for management of patients with acute myocardial infarction (AMI) has undergone substantial evolution since the inaugural publication entitled "Guidelines for the Early Management of Patients with Acute Myocardial Infarction" in 1990 under the auspices of the ACC/AHA Task Force on Assessment of Diagnostic and Therapeutic Cardiovascular Procedures (Chairman, Dr. Rolf Gunnar; Figure 1) (1). Subsequently, the ACC/AHA Task Force on Practice Guidelines convened a committee (Chairman, Dr. Thomas J. Ryan) in 1994 to revise the 1990 guidelines. In the 1996 guideline publication, "ACC/AHA Guidelines for the Management of Patients with Acute Myocardial Infarction," the term acute coronary syndrome was used, reflecting the emerging overarching concept that disruption of a vulnerable or high-risk plaque causes an episode of ischemic discomfort (2). Emphasis was placed on the 12-lead electrocardiogram (ECG) that was used to categorize patients into 2 broad cohorts: those presenting with ST elevation and those presenting without ST elevation (ultimately diagnosed as unstable angina or non-Q-wave myocardial infarction (MI) depending on whether a biomarker of necrosis was detected in the patient's blood). The 1996 guidelines discussed the management of both the ST-elevation and non-ST-elevation presentations of the acute coronary syndromes. The same approach was taken in the 1999 update of the guideline (also chaired by Dr. Ryan) that was posted as an electronic update on the ACC and AHA World Wide Web sites (3).

In parallel to the above efforts, in 1994, the Agency for Health Care Policy and Research and the National Heart, Lung, and Blood Institute jointly published guidelines for the diagnosis and management of patients with unstable angina



Figure 1. Evolution of ACC/AHA guidelines for management of patients with acute MI. The first guideline published by the ACC/AHA described the management of patients with acute myocardial infarction (AMI). The subsequent three documents were the Agency for Healthcare and Quality/National Heart, Lung and Blood Institute sponsored guideline on management of unstable angina (UA), the revised/updated ACC/AHA guideline on AMI, and the revised/updated ACC/AHA guideline on unstable angina/non–ST-segment myocardial infarction (UA/NSTEMI). The present guideline is a revision and deals strictly with the management of patients presenting with ST-elevation myocardial infarction (STEMI). The names of the chairs of the writing committees for each of the guidelines are shown at the bottom of each box.

(Chairman, Dr. Eugene Braunwald). In recognition of rapid advances in the understanding and management of patients with acute coronary syndromes, the ACC/AHA Task Force on Practice Guidelines convened a committee (also chaired by Dr. Braunwald) to revise the 1994 unstable angina guideline. That committee focused on patients presenting without ST elevation and introduced the nomenclature of unstable angina/non–ST-elevation MI (UA/NSTEMI). The "ACC/ AHA Guidelines for the Management of Patients with UA/NSTEMI" were published in 2000 and were updated in electronic form in 2002 (4).

Although considerable improvement has occurred in the process of care for patients with ST-elevation MI (STEMI), room for improvement exists (5-7). The purpose of the present guideline is to focus on the numerous advances in the diagnosis and management of patients with STEMI since 1999. This is reflected in the changed name of the guideline: "ACC/AHA Guidelines for the Management of Patients With ST-Elevation Myocardial Infarction" (Figure 1). It is recognized that there are areas of overlap among this guideline on patients with STEMI, the guideline on patients with UA/NSTEMI, and other guidelines. The committee has handled this overlap by reiterating important concepts and recommendations in this guideline and by providing cross-references to other guidelines.

The final recommendations for indications for a diagnostic procedure, a particular therapy, or an intervention in patients with STEMI summarize both clinical evidence and expert opinion. Once recommendations were written, a Classification of Recommendation and Level of Evidence grade was assigned to each recommendation. Classification of Recommendations and Level of Evidence are expressed in the ACC/AHA format as follows:

Classification of Recommendations

- Class I: Conditions for which there is evidence and/or general agreement that a given procedure or treatment is beneficial, useful, and effective.
- Class II: Conditions for which there is conflicting evidence and/or a divergence of opinion about the usefulness/efficacy of a procedure or treatment.

Class IIa: Weight of evidence/opinion is in favor of usefulness/efficacy.

Class IIb: Usefulness/efficacy is less well established by evidence/opinion.

Class III: Conditions for which there is evidence and/or general agreement that a procedure/treatment is not useful/effective and in some cases may be harmful.

Level of Evidence

• Level of Evidence A: Data derived from multiple randomized clinical trials or meta-analyses.

- Level of Evidence B: Data derived from a single randomized trial, or nonrandomized studies.
- Level of Evidence C: Only consensus opinion of experts, case studies, or standard-of-care.

The schema for classification of recommendations and level of evidence is summarized in Table 1, which also illustrates how the grading system provides an estimate of the size of the treatment effect and an estimate of the certainty of the treatment effect.

The committee recognizes the importance of timely reperfusion for patients with STEMI and spent considerable effort reviewing the literature published since 1999 when formulating recommendations. Along with reperfusion by pharmacological and catheter-based means, the committee emphasized the use of established therapies such as aspirin, betaadrenoceptor-blocking vasodilator therapy. agents, angiotensin converting enzyme (ACE) inhibitors, and cholesterol-lowering therapy. To provide clinicians with a set of recommendations that can easily be translated into the practice of caring for patients with STEMI, this guideline is organized around the chronology of the interface between the patient and the clinician (Figure 2) (8-10). Thus, readers will find material on prevention of STEMI, patient education, prehospital issues, initial recognition and management in the emergency department (ED), hospital management, and long-term management after treatment for the index STEMI event. The reorganization of the material in this guideline along the timeline noted above necessitated considerable modification of the sequence of text presented in the 1996 and 1999 guidelines on AMI. Whenever possible, the writing committee used the term STEMI rather than AMI. Given the reorganization of the guideline along the chronology of clinical care of patients with STEMI and the anticipated desire of readers to search the guideline for specific advice on management of patients with STEMI at different phases of their illness, in a few selected instances, recommendations and, to a lesser extent, some portions of the text are repeated.

Although these guidelines on STEMI have been shaped largely within the context of evidence-based medical practice, the committee clearly understands that variations in inclusion and exclusion criteria from one randomized trial to another impose some limitation on the generalizability of their findings. Likewise, in its efforts to reconcile conflicting data, the committee emphasized the importance of properly characterizing the population under study.

Writing committee members were selected with attention to cardiovascular subspecialties, broad geographical representation, and involvement in academic medicine and primary practice, including neurology, emergency medicine, and nursing. The Writing Committee on the Management of Patients with ST-Elevation Myocardial Infarction also included members of the ACCF Board of Governors, the American Academy of Family Physicians (AAFP), and the Canadian Cardiovascular Society (CCS).

of Evidence
Level
ns and
Recommendatior
of
Classification
1. Applying
Table ⁻

"Size of Treatment Effect"

3	Class 1 Benefit >>> Risk	Class IIa Benefit >> Risk Additional studies with focused objectives needed	Class IIb Benefit ≥ Risk Additional studies with broad objectives needed: Additional	Class III Risk ≥ Benefit No additional studies needed
	Procedure/Treatment SHOULD be performed/administered • Recommendation that	IT IS REASONABLE to perform procedure/administer treatment • Recommendation in favor of	registry data would be helpful Procedure/Treatment MAY BE CONSIDERED • Recommendation's	Procedure/Treatment should NOT be performed/administered SINCE IT IS NOT HELPFUL AND MAY BE HARMFUL • Recommendation that
e (3-5) population risk valuated* l consistency of direction ignitude of effect	procedure or treatment is useful/effective • Sufficient evidence from multiple randomized trials or meta-analyses	treatment or procedure being useful/effective • Some conflicting evidence from multiple randomized trials or meta-analyses	usefulness/efficacy less well established • Greater conflicting evidence from multiple randomized trials or meta-analyses	procedure or treatment not useful/effective and may be harmful • Sufficient evidence from multiple randomized trials or meta-analyses
: (2-3) population risk valuated*	 Recommendation that procedure or treatment is useful/effective Limited evidence from single randomized trial or non- randomized studies 	 Recommendation in favor of treatment or procedure being useful/ effective Some conflicting evidence from single randomized trial or non-randomized studies 	 Recommendation's usefulness/efficacy less well established Greater conflicting evidence from single randomized trial or non-randomized studies 	 Recommendation that procedure or treatment not useful/effective and may be harmful Limited evidence from single randomized trial or non- randomized studies
nited (1-2) population risk valuated*	 Recommendation that procedure or treatment is useful/effective Only expert opinion, case studies, or standard-of-care 	 Recommendation in favor of treatment or procedure being useful/ effective Only diverging expert opinion, case studies, or standard-of- care 	 Recommendation's usefulness/efficacy less well established Only diverging expert opinion, case studies, or standard-of-care 	 Recommendation that procedure or treatment not useful/effective and may be harmful Only expert opinion, case studies, or standard-of-care
ted phrases for writing aendations †	should is recommended is indicated is useful/effective/beneficial	is reasonable can be useful/effective/ beneficial is probably recommended or indicated	may/might be considered may/might be reasonable usefulness/effectiveness is unknown /unclear/uncertain or not well established	is not recommended is not indicated should not is not useful/effective/beneficial may be harmful

"Estimate of Certainty (Precision) of Treatment Effect"



Figure 2. Acute coronary syndromes. The top half of the figure illustrates the chronology of the interface between the patient and the clinician through the progression of plaque formation, onset and complications of STEMI along with relevant management considerations at each stage. The longitudinal section of an artery depicts the "timeline" of atherogenesis from a normal artery (1) to (2) lesion initiation and accumulation of extracellular lipid in the intima, to (3) the evolution to the fibrofatty stage, to (4) lesion progression with procoagulant expression and weakening of the fibrous cap. An acute coronary syndrome develops when the vulnerable or high risk plaque undergoes disruption of the fibrous cap (5); disruption of the plaque is the stimulus for thrombogenesis. Thrombus resorption may be followed by collagen accumulation and smooth muscle cell growth (6). Following disruption of a vulnerable or high-risk plaque, patients experience ischemic discomfort resulting from a reduction of flow through the affected epicardial coronary artery. The flow reduction may be caused by a completely occlusive thrombus (bottom half, right side) or subtotally occlusive thrombus (bottom half, left side). Patients with ischemic discomfort may present with or without STsegment elevation on the ECG. Of patients with ST-segment elevation, most (large red arrow in bottom panel) ultimately develop a Q-wave MI (QwMI), while a few (small red arrow) develop a non-Q-wave MI (NQMI). Patients who present without ST-segment elevation are suffering from either unstable angina or a non-STsegment elevation MI (NSTEMI) (large open arrows), a distinction that is ultimately made on the presence or absence of a serum cardiac marker such as CK-MB or a cardiac troponin detected in the blood. Most patients presenting with NSTEMI ultimately develop a NQMI on the ECG; a few may develop a QwMI. The spectrum of clinical presentations ranging from unstable angina through NSTEMI and STEMI are referred to as the acute coronary syndromes. This STEMI guideline is arranged along the chronologic interface of the clinician with the patient, as diagrammed in the upper panel, and includes sections on management prior to STEMI, at the onset of STEMI, and during the hospital phase. Secondary prevention and plans for long-term management begin early during the hospital phase of treatment. Dx = diagnosis; NQMI, non-Q-wave myocardial infarction; QwMI = Q-wave myocardial infarction; CK-MB = MB isoenzyme of creatine kinase. Modified with permission from Libby. Circulation 2001;104:365-72 (8), Hamm, Bertrand, Braunwald. The Lancet 2001;358:1533-8 (9), and Davies. Heart 2000;83:361-6 (10) with permission from the BMJ Publishing Group.

The committee conducted comprehensive searching of the scientific and medical literature on AMI, with special emphasis on STEMI. Literature searching was limited to publications on humans and in English from 1990 to 2004. In addition to broad-based searching on MI, specific targeted searches were performed on MI and the following subtopics: 9-1-1, patient delays, emergency medical services (EMS), prehospital fibrinolysis, prehospital ECG, ED, supplemental oxygen, nitroglycerin, ASA, clopidogrel, arrhythmia, reperfusion, fibrinolysis/fibrinolytic therapy, angioplasty, stent, coronary artery bypass graft surgery (CABG), glycoprotein (GP) IIb/IIIa, pericarditis, beta-blockers, ischemia, intraarterial pressure monitoring, ACE inhibitors, amiodarone, procainamide, lidocaine, electrical cardioversion, atropine, temporary pacing, transvenous pacing, permanent pacing, cardiac repair, heparin, low-molecular-weight heparin (LMWH), unfractionated heparin (UFH), ramipril, calcium channel blockers, verapamil, nifedipine, magnesium, stress ECG, invasive strategy, secondary prevention, statins, and cholesterol. The complete list of keywords is beyond the scope of this section. The committee reviewed all compiled reports from computerized searches and conducted additional searching by hand. Literature citations were generally restricted to published manuscripts appearing in journals listed in Index Medicus. Because of the scope and importance of certain ongoing clinical trials and other emerging information, published abstracts were cited when they were the only published information available.

This document was reviewed by 3 outside reviewers nominated by the ACC and 3 outside reviewers nominated by the AHA, as well as 1 reviewer each from the AAFP and the CCS, and 58 individual content reviewers. (See Appendix 2 for details.)

This document was approved for publication by the governing bodies of the American College of Cardiology Foundation and the American Heart Association and endorsed by the Canadian Cardiovascular Society. These guidelines will be reviewed annually by the ACC/AHA Task Force on Practice Guidelines and will be considered current unless they are revised or withdrawn from distribution.

2. PATHOLOGY

2.1. Role of Acute Plaque Change

Slowly accruing high-grade stenoses of epicardial coronary arteries may progress to complete occlusion but do not usually precipitate STEMI, probably because of the development over time of a rich collateral network. However, during the natural evolution of atherosclerotic plaques, especially those that are lipid laden, an abrupt and catastrophic transition may occur, characterized by plaque disruption by rupture of the fibrous cap or erosion of the surface of the fibrous cap (Figure 2) (8-10). Plaques that are prone to disruption are usually nonobstructive, are characterized by abundant macrophages and other inflammatory cells, and are often located at branch points or bends in the arterial tree (11-15). They are referred to as vulnerable or high-risk plaques. After plaque disruption, there is exposure of substances that promote platelet activation, adhesion, and aggregation, thrombin generation, and ultimately thrombus formation (16,17). The resultant thrombus can completely occlude the epicardial infarct artery. If there is an insufficient collateral supply, a wave front of myocardial necrosis begins within 15 minutes and spreads from the endocardium toward the epicardium (18). This may be modulated by the extent of collateral flow and determinants of myocardial oxygen consumption, affording opportunity for significant myocardial salvage (19). Of note, intravascular ultrasound studies suggest that in addition to the disruptured plaque, several other vulnerable or high-risk plaques may coexist throughout the coronary vasculature.

2.2. Acute Coronary Syndromes

Disruption of vulnerable or high-risk plaques is the common pathophysiological substrate of the acute coronary syndromes that comprise a spectrum of myocardial ischemia. Patients with an acute coronary syndrome include those whose clinical presentations cover the following range of diagnoses: unstable angina, MI without ST elevation (NSTE-MI), and MI with ST elevation (STEMI) (Figure 2) (8-10). Patients with STEMI have a high likelihood of a coronary thrombus occluding the infarct artery (20,21). Angiographic evidence of coronary thrombus formation may be seen in more than 90% of patients with STEMI but in only 1% of patients with stable angina and about 35% to 75% of patients with unstable angina or NSTEMI (20-24). However, not every STEMI evolves into a Q-wave MI; likewise, a patient with NSTEMI may develop Q waves. The acute coronary syndrome spectrum concept is a useful framework for developing therapeutic strategies. Antithrombin therapy and antiplatelet therapy should be administered to all patients with an acute coronary syndrome regardless of the presence or absence of ST-segment elevation. Patients presenting with persistent ST-segment elevation are candidates for reperfusion therapy (either pharmacological or catheter-based) to restore flow promptly in the occluded epicardial infarctrelated artery and are the subject of this guideline (Figure 3) (24-40). Patients presenting without ST-segment elevation are not candidates for immediate pharmacological reperfusion but should receive anti-ischemic therapy and catheterbased therapy where applicable as discussed in the ACC/AHA Guidelines for Management of Patients with UA/NSTEMI (4).

2.3. Pathophysiology

A key concept in the pathophysiology of STEMI is ventricular remodeling, a term that refers to changes in size, shape, and thickness of the left ventricle involving both the infarcted and noninfarcted segments of the ventricle (41,42). Acute dilatation and thinning of the area of infarction that is not due to additional myocardial necrosis is referred to as infarct expansion (43). An extra load is placed on the residual functioning myocardium, which results in compensatory hyperACC - www.acc.org AHA - www.americanheart.org



Figure 3. Relative treatment effect associated with several acute reperfusion modalities in patients presenting with STEMI. Data are odds ratios and 95% confidence intervals. UFH = unfractionated heparin; PCI = percutaneous coronary intervention; Exp = experimental group; Ctrl = control group. *Data for myocardial reinfarction as a single end point were not available for meta-3; in this case the figure presents odds ratios for the composite of death or myocardial reinfarction. †Intracranial hemorrhage was not reported in meta-1-data were derived from the HERO-1, HIT-4, TIMI9b, and GUSTO2b trials that were included in this meta-analysis. Modified with permission from Elsevier (Boersma et al. The Lancet 2003;361:847-58) (24).

trophy. Inhibition of the renin-angiotensin-aldosterone system is a key therapeutic maneuver in patients with STEMI (44). Additional important pathophysiological concepts in patients with STEMI that form the basis for recommendations in this guideline include cardiac arrhythmias such as those that result from electrical instability, pump failure/excessive sympathetic stimulation, and conduction disturbances. Mechanical problems that result from dysfunction or disruption of critical myocardial structures (e.g., mitral regurgitation [MR], rupture of the interventricular septum, ventricular aneurysm formation, and free wall rupture) may require a combination of pharmacological, catheterbased, and surgical treatments.

2.4. Epidemiology

STEMI continues to be a significant public health problem in industrialized countries and is becoming an increasingly sig-

nificant problem in developing countries (45). Although the exact incidence is difficult to ascertain, using first-listed and secondary hospital discharge data, there were 1 680 000 unique discharges for ACS in 2001 (46). Applying the conservative estimate of 30% of the ACS patients who have STEMI from the National Registry of Myocardial Infarction [NRMI-4] (46a), we estimate 500 000 STEMI events per year in the U.S. However, there has been a steady decline in the mortality rate from STEMI over the last several decades. This appears to be due to a combination of a fall in the incidence of MI (replaced in part by an increase in the incidence of unstable angina) and a reduction in the case fatality rate once an MI has occurred (47-49). There has been a progressive increase in the proportion of patients who present with NSTEMI compared with STEMI.

The committee strongly endorses several public health campaigns that are likely to contribute to a reduction in the incidence of and fatality from STEMI in the future. These include the following: 1) recognition of diabetes mellitus and chronic kidney disease as "risk equivalents" to coronary heart disease (CHD) and therefore recommendation for more aggressive attempts at control of other risk factors (50,51); 2) recognition of the importance of dyslipidemia as a major risk factor for CHD and recommendation for aggressive attempts at cholesterol reduction and treatment of the metabolic syndrome (50); 3) aggressive primary prevention efforts at smoking cessation as emphasized by the World Health Organization; 4) patient education campaigns regarding the signs and symptoms of MI and the appropriate courses of action to be taken (52,53); and 5) implementation at a professional level of quality assurance projects such as the ACC's "Guidelines Applied in Practice" (54) and the AHA's "Get with the Guidelines" (55) to improve compliance with established treatment strategies when caring for patients with MI. A proposal that holds great promise for reducing the morbidity and mortality associated with STEMI is the regionalization of care for patients with acute coronary syndromes using centers of excellence (56-58).

3. MANAGEMENT BEFORE STEMI

One third of patients who experience STEMI will die within 24 hours of the onset of ischemia, and many of the survivors will suffer significant morbidity (24). For many patients, the first manifestation of CHD will be sudden death. The major risk factors for development of CHD and STEMI are well established. Clinical trials have demonstrated that modification of those risk factors can prevent the development of CHD (primary prevention) or reduce the risk of experiencing STEMI in patients who have CHD (secondary prevention). All practitioners should emphasize prevention and refer patients to primary care providers for appropriate long-term preventive care. In addition to internists and family physicians, cardiologists have an important leadership role in primary (and secondary) prevention efforts.

3.1. Identification of Patients at Risk of STEMI

Class I

- 1. Primary care providers should evaluate the presence and status of control of major risk factors for CHD for all patients at regular intervals (approximately every 3 to 5 years). (Level of Evidence: C)
- 2. Ten-year risk (National Cholesterol Education Program [NCEP] global risk) of developing symptomatic CHD should be calculated for all patients who have 2 or more major risk factors to assess the need for primary prevention strategies (59). (Level of Evidence: B)
- **3.** Patients with established CHD should be identified for secondary prevention, and patients with a CHD risk equivalent (e.g., diabetes mellitus, chronic kidney disease, or 10-year risk greater than 20% as calculated by Framingham equations) should receive equally intensive risk factor intervention as those with clinically apparent CHD. (*Level of Evidence: A*)

Major risk factors for developing CHD (i.e., smoking, family history, adverse lipid profiles, and elevated blood pressure) have been established from large long-term epidemiological studies (59,60). These risk factors are predictive for most populations in the United States. Primary prevention interventions aimed at these risk factors are effective when used properly. They can also be costly in terms of primary care physician time, diversion of attention from other competing and important healthcare needs, and expense, and they may not be effective unless targeted at higher-risk patients (61). It is therefore important for primary care providers to make identifying patients at risk, who are most likely to benefit from primary prevention, a routine part of everyone's health care. The Third Report of the NCEP provides guidance on identifying such patients (59).

Patients with 2 or more risk factors who are at increased 10-year risk will have the greatest benefit from primary prevention, but any individual with a single elevated risk factor is a candidate for primary prevention. Waiting until the patient develops multiple risk factors and increased 10-year risk contributes to the high prevalence of CHD in the United States (59,62). Such patients should have their risk specifically calculated, by any of the several available valid prognostic tools available in print (59,63), on the internet (64), or for use on a personal computer or PDA (Personal Digital Assistant) (59). Patients' specific risk levels determine the absolute risk reductions they can obtain from preventive interventions and guide selection and prioritization of those interventions. For example, target levels for lipid lowering and for antihypertensive therapy vary by patients' baseline risk. A specific risk number can also serve as a powerful educational intervention to motivate lifestyle changes (65).

3.2. Interventions to Reduce Risk of STEMI

The benefits of prevention of STEMI in patients with CHD are well documented and of large magnitude (62,66-68). Patients with established CHD should be identified for secondary prevention, and patients with a CHD risk equivalent (e.g., diabetes mellitus, chronic kidney disease, or 10-year risk greater than 20% as calculated by Framingham equations) should receive equally intensive risk factor intervention for high-risk primary prevention regardless of sex (69). Patients with diabetes and peripheral vascular disease have baseline risks of STEMI similar to patients with known CHD, as do patients with multiple risk factors predicting calculated risk of greater than 20% over 10 years as estimated by the Framingham equations (59). Such patients should be considered to have the risk equivalents of CHD, and they can be expected to have an absolute benefit similar to those with established CHD.

All patients who smoke should be encouraged to quit and should be provided with help in quitting at every opportunity. Even a single recommendation by a clinician to quit smoking can have a meaningful impact on the rate of cessation of smoking. The most effective strategies for encouraging quitting are those that identify patients' level or stage of readiness and provide information, support, and, if necessary, pharmacotherapy targeted at the individual's readiness and specific needs (66,70). Pharmacotherapy may include nicotine replacement or withdrawal-relieving medication such as bupropion. Most patients require several attempts before succeeding in quitting permanently. Additional discussion in this area can be found in the ACC/AHA 2002 Guideline Update for the Management of Patients With Chronic Stable Angina (71).

All patients should be instructed in and encouraged to maintain appropriate low-saturated-fat and low-cholesterol diets high in soluble (viscous) fiber and rich in vegetables, fruits, and whole grains. The statin drugs have the best outcome evidence supporting their use and should be the mainstay of pharmacological intervention (62). The appropriate levels for lipid management are dependent on baseline risk; the reader is referred to the NCEP report for details (59).

Primary prevention patients with high blood pressure should be treated according to the recommendations of the Seventh Joint National Committee on High Blood Pressure (JNC-7) (72,73). Specific treatment recommendations are based on the level of hypertension and the patient's other risk factors. A diet low in salt and rich in vegetables, fruits, and low-fat dairy products should be encouraged for all hypertensive patients, as should a regular aerobic exercise program. Most patients will require more than 1 medication to achieve blood pressure control, and pharmacotherapy should begin with known outcome-improving medications (primarily thiazide diuretics as first choice, with the addition of betablockers, ACE inhibitors, angiotensin receptor blockers, and long-acting calcium channel blockers) (72,74). Systolic hypertension is a powerful predictor of adverse outcome, particularly among the elderly, and should be treated even if diastolic pressures are normal (75).

Aspirin prophylaxis can uncommonly result in hemorrhagic complications and should only be used in primary prevention when the level of risk justifies it. Patients whose 10-year risk of CHD is 6% or more are most likely to benefit, and aspirin 75 to 162 mg/d as primary prophylaxis should be discussed with such patients (76-79).

3.3. Patient Education for Early Recognition and Response to STEMI

Class I

- 1. Patients with symptoms of STEMI (chest discomfort with or without radiation to the arms[s], back, neck, jaw, or epigastrium; shortness of breath; weakness; diaphoresis; nausea; lightheadedness) should be transported to the hospital by ambulance rather than by friends or relatives. (Level of Evidence: B)
- 2. Healthcare providers should actively address the following issues regarding STEMI with patients and their families:
 - a. The patient's heart attack risk (Level of Evidence: C)
 - b. How to recognize symptoms of STEMI (Level of Evidence: C)
 - c. The advisability of calling 9-1-1 if symptoms are unimproved or worsening after 5 minutes, despite

feelings of uncertainty about the symptoms and fear of potential embarrassment (Level of Evidence: C)

- d. A plan for appropriate recognition and response to a potential acute cardiac event, including the phone number to access EMS, generally 9-1-1 (80) (Level of Evidence: C)
- **3.** Healthcare providers should instruct patients for whom nitroglycerin has been prescribed previously to take ONE nitroglycerin dose sublingually in response to chest discomfort/pain. If chest discomfort/pain is unimproved or worsening 5 minutes after 1 nitroglycerin dose has been taken, it is recommended that the patient or family member/friend call 9-1-1 immediately to access EMS. (Level of Evidence: C)

Morbidity and mortality from STEMI can be reduced significantly if patients and bystanders recognize symptoms early, activate the EMS system, and thereby shorten the time to definitive treatment. Patients with possible symptoms of STEMI should be transported to the hospital by ambulance rather than by friends or relatives, because there is a significant association between arrival at the ED by ambulance and early reperfusion therapy (81-84). In addition, emergency medical technicians (EMTs) and paramedics can provide life-saving interventions (e.g., early cardiopulmonary resuscitation [CPR] and defibrillation) if the patient develops cardiac arrest. Approximately 1 in every 300 patients with chest pain transported to the ED by private vehicle goes into cardiac arrest en route (85).

Several studies have confirmed that patients with STEMI usually do not call 9-1-1 and are not transported to the hospital by ambulance. A follow-up survey of chest pain patients presenting to participating EDs in 20 US communities who were either released or admitted to the hospital with a confirmed coronary event revealed that the average proportion of patients who used EMS was 23%, with significant geographic difference (range 10% to 48%). Most patients were driven by someone else (60%) or drove themselves to the hospital (16%) (86). In the National Registry of Myocardial Infarction 2, just over half (53%) of patients with STEMI were transported to the hospital by ambulance (82).

Even in areas of the country that have undertaken substantial public education on warning signs of STEMI and the need to activate the EMS system rapidly, either there were no increases in EMS use (87-91) or EMS use increased (as a secondary outcome measure) but was still suboptimal, with a 20% increase from a baseline of 33% in all 20 communities in the Rapid Early Action for Coronary Treatment (REACT) study (92) and an increase from 27% to 41% in southern Minnesota after a community campaign (93). Given the importance of patients using EMS for possible acute cardiac symptoms, communities, including medical providers, EMS systems, healthcare insurers, hospitals, and policy makers at the state and local level, need to have agreed-upon emergency protocols to ensure patients with possible heart attack symptoms will be able to access 9-1-1 without barriers, to secure their timely evaluation and treatment (94).

As part of making a plan with the patient for timely recognition and response to an acute event, providers should review instructions for taking nitroglycerin in response to chest discomfort/pain. If a patient has previously been prescribed nitroglycerin, it is recommended that the patient be advised to take ONE nitroglycerin dose sublingually promptly for chest discomfort/pain. If symptoms are unimproved or worsening 5 minutes after ONE nitroglycerin dose has been taken, it is also recommended that the patient be instructed to call 9-1-1 immediately to access EMS. Although the traditional recommendation is for patients to take 1 nitroglycerin dose sublingually, 5 minutes apart, for up to 3 doses before calling for emergency evaluation, this recommendation has been modified by the writing committee to encourage earlier contacting of EMS by patients with symptoms suggestive of STEMI. Self-treatment with prescription medication, including nitrates, and with nonprescription medication (e.g., antacids) has been documented as a frequent cause of delay among patients with STEMI, including those with a history of MI or angina (95,96). Both the rate of use of these medications and the number of doses taken were positively correlated with delay time to hospital arrival (95).

Family members, close friends, or advocates should be included in these discussions and enlisted as reinforcement for rapid action when the patient experiences symptoms of a possible STEMI (3,80,97) (Figure 4). For patients known to their providers to have frequent angina, physicians may consider a selected, more tailored message that takes into account the frequency and character of the patient's angina and their typical time course of response to nitroglycerin. Avoidance of patient delay associated with self-medication and prolonged re-evaluation of symptoms is paramount.

Taking an aspirin in response to acute symptoms by patients has been reported to be associated with a delay in calling EMS (86). Patients should focus on calling 9-1-1, which activates the EMS system, where they may receive instructions from emergency medical dispatchers to chew aspirin (162 to 325 mg) while emergency personnel are en route, or emergency personnel can give an aspirin while transporting the patient to the hospital (98). Alternatively, patients may receive an aspirin as part of their early treatment once they arrive at the hospital if it has not been given in the prehospital setting (3).

Providers should target those patients at increased risk for STEMI, focusing on patients with known CHD, peripheral vascular disease, or cerebral vascular disease, those with diabetes, and patients with 10-year Framingham risk of CHD of greater than 20% (99). They should stress that the chest discomfort will usually not be dramatic, such as is commonly misrepresented on television or in the movies as a "Hollywood heart attack." Providers should also describe anginal equivalents and the commonly associated symptoms of STEMI (e.g., shortness of breath, a cold sweat, nausea, or lightheadedness) in both men and women (83), as well as the

increased frequency of atypical symptoms in elderly patients (100).

In September 2001, the NHAAP and the AHA launched a campaign urging patients and providers to "Act in Time to Heart Attack Signs" (101). The campaign urges both men and women who feel heart attack symptoms or observe the signs in others to wait no more than a few minutes, 5 minutes at most, before calling 9-1-1 (101,102). Campaign materials point out that patients can increase their chance of surviving a STEMI by learning the symptoms and filling out a survival plan. They also are advised to talk with their doctor about heart attack and how to reduce their risk of having one. The patient materials include a free brochure about symptoms and recommended actions for survival, in English (103) and Spanish (104), as well as a free wallet card that can be filled in with emergency medical information (105). Materials geared directly to providers include a Patient Action Plan Tablet (106), which contains the heart attack warning symptoms and steps for developing a survival plan individualized with the patient's name; a quick reference card for addressing common patient questions about seeking early treatment to survive a heart attack (107), including a palm pilot version (108); and a warning signs wall chart (109). These materials and others are available on the "Act in Time" Web page (www.nhlbi.nih.gov/actintime) (51,101) (Figure 5).

4. ONSET OF STEMI

4.1. Recognition of Symptoms by Patient

Early recognition of symptoms of STEMI by the patient or someone with the patient is the first step that must occur before evaluation and life-saving treatment can be obtained. Although many lay persons are generally aware that chest pain is a presenting symptom of STEMI, they are unaware of the common associated symptoms, such as arm pain, lower jaw pain, shortness of breath, and diaphoresis (111) or anginal equivalents. The average patient with STEMI does not seek medical care for approximately 2 hours after symptom onset, and this pattern appears unchanged over the last decade (45,87,112). Average and median delays for patients with STEMI were 4.7 and 2.3 hours, respectively, from the 14-country Global Registry of Acute Coronary Events (GRACE) project. Approximately 41% of patients with STEMI presented to the 94 study hospitals within 2 hours of acute cardiac ischemia symptom onset (113).

A baseline analysis from the REACT research program demonstrated longer delay times among non-Hispanic blacks, older patients, and Medicaid-only recipients and shorter delay times among Medicare recipients (compared with privately insured patients) and among patients who came to the hospital by ambulance (87). In the majority of studies examined to date, women in both univariate and multivariate adjusted analyses (in which age and other potentially confounding variables have been controlled) exhibit more prolonged delay patterns than men (113).

A number of studies have provided insight into why patients delay in seeking early care for heart symptoms



ion occurs with non-enteric-coated formulations.



Figure 5. Act in Time to **Heart Attack Signs**

Use the **T.I.M.E.** Method To Help Your Patients Make a Heart Attack Survival Plan

Why Your Patients Need To Act in Time to Heart Attack Signs

Coronary heart disease is the leading killer of both men and women in the United States. Each year, about 1.1 million Americans suffer a heart attack. About 460,000 of those heart attacks are fatal. Disability and death from heart attack can be reduced with prompt thrombolytic and other artery-opening therapies—ideally given within the first hour after symptom onset. Patient delay is the largest barrier to receiving therapy quickly.

Heart Attack Warning Signs

- Chest discomfort (pressure, squeezing, fullness, or pain in the center of the chest)
- Discomfort in one or both arms, back, neck, jaw, or stomach
- ▲ **Shortness of breath** (often comes with or before chest discomfort)
- Breaking out in a cold sweat, nausea, or light-headedness

Uncertainty Is Normal

Most people think a heart attack is sudden and intense, like a "movie heart attack." The fact is that many heart attacks start slowly as mild pain or discomfort. People who feel such symptoms may not be sure what is wrong.

Delay Can Be Deadly



Most heart attack victims wait 2 or more hours after symptoms begin before they seek medical help. People often take a wait-and-see approach or deny that their symptoms are serious.

Every minute that passes without treatment means that more heart muscle dies.

Calling 9-1-1 Saves Lives

Minutes matter. Anyone with heart attack symptoms *should not wait more than a few minutes*-

5 minutes at most-to call 9-1-1.

From: Act in Time to Heart Attack Signs. Action Plan. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3313, September 2001 http://nhlbi.nih.gov/health/prof/heart/mi/provider.pdf (110)

Use the T.I.M.E. Method:

Talk with your patients about—

- A Risk of a heart attack.
- A Recognition of symptoms.
- Right action steps to take/rationale for rapid action.
- Rx-give instructions for when symptoms occur (based on patient history).
- Remembering to call 9-1-1 quickly– within 5 minutes.

nvestigate—

- Feelings about heart attack.
- Barriers to symptom evaluation and response.
- Personal and family experience with AMI and emergency medical treatment.

Make a plan—

- Help patients and their family members to make a plan for exactly what to do in case of heart attack symptoms.
- Encourage patients and their family members to rehearse the plan.

Evaluate—

- The patient's understanding of risk in delaying.
- The patient's understanding of your recommendations.
- The family's understanding of risk and their plan for action.

Additional Resources

Find information and educational materials at the National Heart, Lung, and Blood Institute Web site: www.nhlbi.nih.gov and the American Heart Association Web site: www.americanheart.org

NATIONAL INSTITUTES OF HEA



Table 2. Reasons Patients Delay Seeking Medical Attention for

 Symptoms of ST-Elevation Myocardial Infarction

Expected a dramatic presentation

Thought symptoms were not serious/would go away

- Took a "wait and see" approach to the initial symptoms that included self-evaluation, self-treatment, and reassessment until "certain" Tended to attribute symptoms to other chronic conditions (e.g.,
- arthritis, muscle strain) or common illnesses (e.g., influenza)
- Lacked awareness of the benefits of rapid action, reperfusion treatment, or of the importance of calling EMS/9-1-1 for acute MI symptoms
- Expressed fear of embarrassment if symptoms turned out to be a false alarm; reluctant to "bother" physicians or EMS unless "really sick"; need "permission" from others such as healthcare providers, spouses, family to take rapid action
- Few ever discussed symptoms, responses, or actions for a heart attack in advance with family or providers

Stereotypes of who is at risk for a heart attack

Not perceived at risk if:

- Young and healthy (especially men)
- A woman
- Under a doctor's care or making lifestyle changes (especially men with risk factors)

Based on findings from Finnegan et al. Preventive Medicine 2000;31:205-13 (114).

(Table 2) (114). Focus groups conducted for the REACT research program (92,115) revealed that patients commonly hold a pre-existing expectation that a heart attack would present dramatically with severe, crushing chest pain, such that there would be no doubt that one was occurring. This was in contrast to their actual reported symptom experience of a gradual onset of discomfort involving midsternal chest pressure or tightness, with other associated symptoms often increasing in intensity. The ambiguity of these symptoms, due to this disconnect between prior expectations and actual experience, resulted in uncertainty about the origin of symptoms and thus a "wait-and-see" posture by patients and those around them (114). Other reported reasons for delay were that patients thought the symptoms were self-limited and would go away or were not serious (95,116,117); that they attributed symptoms to other pre-existing chronic conditions, especially among the elderly with multiple chronic conditions (e.g., arthritis), or sometimes to a common illness such as influenza; that they were afraid of being embarrassed if symptoms turned out to be a "false alarm"; that they were reluctant to trouble others (e.g., providers, EMS) unless they were "really sick" (95,116,117); that they held stereotypes of who is at risk for a heart attack; and that they lacked awareness of the importance of rapid action, knowledge of reperfusion treatment, or knowledge of the benefits of calling EMS/9-1-1 to ensure earlier treatment (Table 2) (Figure 5) (51,114). Notably, women did not perceive themselves to be at risk (53).

4.1.1. Silent and Unrecognized Events

Patients experiencing STEMI do not always present with chest discomfort (118). The Framingham Study was the first to show that as many as half of all MIs may be clinically

silent and unrecognized by the patient (119). Canto et al. (100) found that one third of the 434 877 patients with confirmed MI in the National Registry of Myocardial Infarction (NRMI) (100) presented to the hospital with symptoms other than chest discomfort. Compared with MI patients with chest discomfort, MI patients without chest discomfort were more likely to be older (74.2 versus 66.9 years), women (49.0%) versus 38.0%), diabetic (32.6% versus 25.4%), and/or have prior heart failure (26.4% versus 12.3%). MI patients without chest discomfort delayed longer before they went to the hospital (mean 7.9 versus 5.3 hours) and were less likely to be diagnosed as having an MI when admitted (22.2% versus 50.3%). They also were less likely to receive fibrinolysis or primary percutaneous coronary intervention (PCI) (25.3%) versus 74.0%), aspirin (60.4% versus 84.5%), beta-blockers (28.0% versus 48.0%), or heparin (53.4% versus 83.2%). Silent MI patients were 2.2 times (95% confidence interval [CI] 2.17 to 2.26) more likely to die during the hospitalization (in-hospital mortality rate 23.3% versus 9.3%). Healthcare providers should maintain a high index of suspicion for MI when evaluating women, diabetics, older patients, and those with a history of heart failure, as well as those patients complaining of chest discomfort but who have a permanent pacemaker that may confound recognition of STEMI on their 12-lead ECG (120).

4.2. Out-of-Hospital Cardiac Arrest

Class I

- 1. All communities should create and maintain a strong "Chain of Survival" for out-of-hospital cardiac arrest that includes early access (recognition of the problem and activation of the EMS system by a bystander), early CPR, early defibrillation for patients who need it, and early advanced cardiac life support (ACLS). (Level of Evidence: C)
- 2. Family members of patients experiencing STEMI should be advised to take CPR training and familiarize themselves with the use of an automated external defibrillator (AED). In addition, they should be referred to a CPR training program that has a social support component for family members of post-STEMI patients. (Level of Evidence: B)

The majority of deaths from STEMI occur within the first 1 to 2 hours after symptom onset, usually from ventricular fibrillation (VF). Survival from VF is inversely related to the time interval between its onset and termination. For each minute that a patient remains in VF, the odds of survival decrease by 7% to 10% (121). Survival is optimal when both CPR and ACLS, including defibrillation and drug therapy, are provided early.

The AHA has introduced the "chain of survival" concept to represent a sequence of events that ideally should occur to maximize the odds of successful resuscitation from cardiac arrest (121). The links in the chain include early access (recognition of the problem and activation of the EMS system by a bystander), early CPR, early defibrillation for patients who need it, and early ACLS.

Although estimates of overall survival from out-of-hospital cardiac arrest in the United States are as low as 5%, survival in patients who are in VF initially can be much higher. The percentage of patients who are found in VF and the likelihood of survival are higher if the patient's collapse is witnessed, bystander CPR is performed, and a monitor/defibrillator can be applied quickly. For example, 27% of patients with witnessed out-of-hospital cardiac arrest in Seattle, WA, survived to leave the hospital when bystanders performed CPR (122). Only 13% survived without bystander CPR. Emerging data suggest that treatment of VF with immediate defibrillation, irrespective of "down time," may not be optimal for all patients and that as the duration of cardiac arrest increases, different interventions may take priority over defibrillation, such as a period of chest compressions (with associated tissue oxygen delivery) after 3 minutes of VF before defibrillation (123).

There is often a long delay from the recognition of cardiac arrest to defibrillation in rural areas where travel time is long and in densely populated urban areas. Survival rates are often extremely low in such settings (124-126). In Seattle, WA, the majority of out-of-hospital cardiac arrest victims receive defibrillation within 5 to 7 minutes after the recognition of out-of-hospital cardiac arrest. In Rochester, MN, the addition of a police defibrillation program to conventional EMS services resulted in a median time to first shock of 5.9 minutes for patients in VF and a 49% rate of survival to discharge (127). Total cumulative survival experience at 7 years in this community was 40% (128). Outcomes data on all Rochester, MN, patients who had an out-of-hospital cardiac arrest with VF from 1990 to 2000 who received defibrillation from emergency personnel showed that 72% survived to hospital admission and 40% were neurologically intact at discharge, with a mean follow-up of 4.8 years (129).

The key to improved survival appears to be the availability of early defibrillation. In the Ontario Pre-hospital Advanced Life Support (OPALS) study, which involved 19 suburban and urban communities, improving the proportion of out-ofhospital cardiac arrest patients who were reached by a defibrillation-equipped ambulance within 8 minutes from 77% to 93% increased survival to hospital discharge from 3.9% to 5.2% (130). A 2-year prospective study at 3 Chicago, IL, airports of readily accessible AEDs in well-marked areas of the airport reported successful resuscitation in 11 of 18 patients with VF. Ten of the 18 were alive and neurologically intact at 1 year of follow-up (131).

Family members of patients with STEMI should be referred to a CPR program that combines CPR training with social support (132,133) (see Section 7.12.1). One study of the impact of in-home defibrillators on post-MI patients and their significant others reported that AEDs were valued by the participants and increased their perception of control over their heart disease, notably for those who believed their risk of cardiac arrest to be high (134). Research is under way

to test the safety and effectiveness of home use of AEDs by family members of patients after MI (135).

5. PREHOSPITAL ISSUES

5.1. Emergency Medical Services Systems

Class I

- **1.** All EMS first responders who respond to patients with chest pain and/or suspected cardiac arrest should be trained and equipped to provide early defibrillation. (*Level of Evidence: A*)
- 2. All public safety first responders who respond to patients with chest pain and/or suspected cardiac arrest should be trained and equipped to provide early defibrillation with AEDs. (Provision of early defibrillation with AEDs by non-public safety first responders is a promising new strategy, but further study is needed to determine its safety and efficacy.) (Level of Evidence: B)
- 3. Dispatchers staffing 9-1-1 center emergency medical calls should have medical training, should use nationally developed and maintained protocols, and should have a quality-improvement system in place to ensure compliance with protocols. (Level of Evidence: C)

EMS systems vary considerably among communities in their ability to evaluate and treat suspected patients with STEMI, with some providing little beyond first aid and early defibrillation, whereas others have highly trained paramedics with sophisticated technology and advanced protocols.

EMS systems have 3 traditional components: emergency medical dispatch, first response, and EMS ambulance response.

Emergency Medical Dispatch. Early access to EMS is promoted by a 9-1-1 system currently available to more than 90% of the United States population. Enhanced 9-1-1 systems provide the caller's location to the dispatcher, which permits rapid dispatch of prehospital personnel to locations even if the caller is not capable of verbalizing or the dispatcher cannot understand the location of the emergency. A major challenge is the widespread proliferation and use of cell phones. Current cell phone technology does not provide the location of the caller to an enhanced 9-1-1 center. Instead, such calls are usually answered by the state police, who then attempt to determine the location of the emergency and forward the call to the appropriate 9-1-1 center. Such additional steps often result in substantial delays in the dispatch of emergency units to the scene. Several technological solutions to this problem exist but have not yet been implemented by the cell phone industry. EMS healthcare workers should encourage enhanced cell phone technology to identify caller location on 9-1-1 systems.

In most communities, law enforcement or public safety officials are responsible for operating 9-1-1 centers, because in most locations, 85% of calls are for police assistance, 10% are for EMS, and 5% are for fire-related emergencies.

Dispatchers who staff 9-1-1 centers typically have only minimal medical background and training and usually operate by following written cards and protocols that in many cases are designed and updated locally. High-performance centers employ EMTs and/or paramedics who are specially trained and certified as emergency medical dispatchers. They, too, operate under written protocols, but such protocols are usually developed and upgraded at the national level. Such centers typically have intense quality-assurance programs to ensure that emergency medical dispatchers follow protocols and procedures correctly and consistently. This is particularly true for the prearrival instructions that are given to cardiac arrest bystanders to instruct them on how to perform CPR while awaiting arrival of emergency personnel (phone CPR) (136). Efforts to shorten the time for contact of the STEMI patient with the medical system are likely to require expansion of the number of trained emergency response personnel and consideration of streamlining methods for distinguishing emergency calls for medical assistance from other emergencies using separate phone numbers, as is the practice currently in some European countries.

First Response. To minimize time to treatment, particularly for cardiopulmonary arrest, many communities allow volunteer and/or paid firefighters and other first-aid providers to function as first responders, providing CPR and, increasingly, early defibrillation using AEDs until EMTs and paramedics arrive. AEDs have been shown to be safe and effective when used by trained first responders with a duty to act (137-139). Systems that incorporate AEDs to shorten response times are highly desirable. Ideally, there should be a sufficient number of trained personnel so that a trained first responder can be at the victim's side within 5 minutes of the call.

Another popular community approach to increase the number of out-of-hospital VF patients who receive early defibrillation is public access defibrillation (PAD), so named because the intent is to have laypersons perform early defibrillation. Experience thus far has been favorable in terms of efficacy and safety when trained public safety laypersons (e.g., flight attendants or security officers) have been allowed to use AEDs to treat cardiac arrest victims (131,140,141). Provision of early defibrillation with AEDs by non-public safety first responders is a promising new strategy to prevent sudden cardiac death after the onset of STEMI, but further study is needed to determine its safety and efficacy (142-147) (Ornato JP; oral presentation, American Heart Association 2003 Annual Scientific Sessions, November 2003, Orlando, FL).

EMS Ambulance Response. Most cities and larger suburban areas provide EMS ambulance services with providers from the fire department, a private ambulance company, and/or volunteers. The most common pattern is a tiered system in which some of the ambulances are staffed and equipped at the basic EMT level (which includes first aid and early defibrillation with AEDs) and other units (either transporting or

nontransporting) are staffed by paramedics or other intermediate-level EMTs (who can, in addition to basic care, start intravenous [IV] drips, intubate, and administer medications). In some systems, the advanced providers can also perform 12-lead ECGs, provide external pacing for symptomatic bradycardia, and utilize other techniques. Some highperformance EMS systems have only advanced life support-staffed ambulances (all-ALS systems). Advantages of such systems are that they provide a uniform standard of care and, surprisingly, can actually lower cost by eliminating the need to dispatch 2 units in response to calls in which it is not clear to dispatchers initially that the patient needs advanced life support (149). The potential disadvantage of such models is that they typically have a relatively large number of paramedics, each of whom gets to perform their advanced skills less frequently than the smaller number of paramedics typically found in tiered systems (150).

Rural areas typically provide primarily basic life support ambulance services, usually by volunteers supplemented by a relatively small number of ALS units. In some cases, ALS is provided by paramedics or helicopter personnel who respond to the scene in addition to the basic life support ambulance.

5.2. Prehospital Chest Pain Evaluation and Treatment

Class I

Prehospital EMS providers should administer 162 to 325 mg of aspirin (chewed) to chest pain patients suspected of having STEMI unless contraindicated or already taken by the patient. Although some trials have used enteric-coated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated formulations. (Level of Evidence: C)

Class IIa

- 1. It is reasonable for all 9-1-1 dispatchers to advise patients without a history of aspirin allergy who have symptoms of STEMI to chew aspirin (162 to 325 mg) while awaiting arrival of prehospital EMS providers. Although some trials have used enteric-coated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated formulations. (Level of Evidence: C)
- 2. It is reasonable that all ACLS providers perform and evaluate 12-lead ECGs routinely on chest pain patients suspected of STEMI. (*Level of Evidence: B*)
- **3.** If the ECG shows evidence of STEMI, it is reasonable that prehospital ACLS providers review a reperfusion "checklist" and relay the ECG and checklist findings to a predetermined medical control facility and/or receiving hospital. (*Level of Evidence: C*)

Because the potential benefits of early aspirin use are great and the risks and costs are low, it is reasonable for physicians to encourage the prehospital administration of aspirin via EMS personnel (i.e., EMS dispatchers and providers) to patients with symptoms suggestive of STEMI unless its use is contraindicated (151). The AHA chest pain algorithm can be adapted for use by prehospital emergency personnel. This protocol recommends empirical treatment of patients with suspected STEMI with Morphine, Oxygen, Nitroglycerin, and Aspirin (MONA) (102). Although short-acting nitroglycerin is often administered for temporary symptomatic relief, it can precipitate hypotension (especially if right ventricular [RV] infarction is present), and long-term nitrates have not been shown to decrease mortality in patients with STEMI (152). To facilitate earlier aspirin administration, it is reasonable that 9-1-1 dispatchers advise non-aspirin-allergic patients with symptoms of STEMI to chew 162 to 325 mg of aspirin while awaiting arrival of prehospital EMS providers. In the absence of instructions by emergency medical dispatchers, prehospital EMS providers (under medical direction) should administer an aspirin en route to the hospital as noted above. Although some trials have used enteric-coated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated formulations.

The AHA (102), the 31st Bethesda Conference of the American College of Cardiology (153), and a technology review supported by the NHLBI's NHAAP (154) strongly encourage the use of 12-lead ECGs by paramedics to evaluate all patients with chest discomfort suspected to be of ischemic origin in the prehospital setting (Figure 6) (Table 3) (155). This requires providing training and 12-lead ECG equipment to all ACLS personnel.

For patients who have ECG evidence of STEMI, it is reasonable that paramedics review a reperfusion checklist and relay the ECG and checklist findings to a predetermined medical control facility and/or receiving hospital (Table 3). The checklist should be designed to determine the presence or absence of comorbid conditions and underlying conditions in which fibrinolytic therapy may be hazardous. The checklist should also facilitate detection of patients with suspected STEMI who are at especially high risk (see Table 3), including those with severe heart failure or cardiogenic shock, for whom primary PCI is generally the preferred reperfusion strategy. (See Section 6.3.1.6.4.2.)

Active involvement of local healthcare providers, particularly cardiologists and emergency physicians, is needed to formulate local EMS protocols for patients with suspected STEMI, provide training, and secure equipment. In the future, regional centers of excellence for care of patients with STEMI may facilitate improvement of EMS protocols (56-58).

5.3. Prehospital Fibrinolysis

Class IIa

Establishment of a prehospital fibrinolysis protocol is reasonable in 1) settings in which physicians are present in the ambulance or 2) well-organized EMS systems with full-time paramedics who have 12-lead ECGs in the field with transmission capability, para-

medic initial and ongoing training in ECG interpretation and STEMI treatment, on-line medical command, a medical director with training/experience in STEMI management, and an ongoing continuous quality-improvement program. (Level of Evidence: B)

The selection of reperfusion strategy is discussed in Section 6.3.1.6.2 and involves assessment of the time from onset of symptoms, risk of STEMI, risk of bleeding, and the time required for transport to a skilled PCI lab. This section discusses issues related to prehospital fibrinolysis, which may bear on the timing and selection of reperfusion therapy.

Randomized controlled trials of fibrinolytic therapy have demonstrated the benefit of initiating fibrinolytic therapy as early as possible after onset of ischemic-type chest discomfort (155-157) (Figure 6). It seems reasonable to expect that if fibrinolytic therapy could be started at the time of prehospital evaluation, a greater number of lives could be saved. In Assessment of the Safety and Efficacy of a New Thrombolytic Regimen (ASSENT)-3, 53% of patients received prehospital fibrinolysis within 2 hours after symptom onset (158). The value of reducing delay until treatment depends not only on the amount of time saved but also on when it occurs. Available data suggest that time saved within the first 1 to 2 hours has greater biological importance than time saved during the later stages of STEMI (156,157,159-164). Several randomized trials of prehospital-initiated fibrinolysis have advanced our understanding of the impact of early treatment (Table 4) (159,165-174). Acquisition of 12lead ECGs in the field and use of a reperfusion checklist (Table 3) lead to more rapid prehospital and hospital care (159,175). Although none of the individual trials showed a reduction in mortality with prehospital-initiated fibrinolytic therapy, a meta-analysis of all available trials (before the Comparison of Angioplasty and Prehospital Thrombolysis in Acute Myocardial Infarction trial [CAPTIM], which was performed after the meta-analysis) (173) demonstrated a 17% relative improvement in outcome associated with prehospital fibrinolytic therapy compared with in-hospital fibrinolytic therapy (95% CI 2% to 29%) (171). In the CAPTIM trial, patients randomized less than 2 hours after symptom onset had a strong trend toward lower 30-day mortality with prehospital fibrinolysis than did those randomized to primary PCI (2.2% versus 5.7%, p equals 0.058) (176). Similarly, patients in PRimary Angioplasty in patients transferred from General community hospitals to specialized PTCA Units with or without Emergency thrombolysis (PRAGUE-2) who were randomized within 3 hours of symptom onset (n equals 551) had no difference in mortality whether treated by fibrinolysis (7.4%) or transferred for PCI (7.3%) (177). Systems that have extensive experience with prehospital fibrinolysis with physician attendance in the ambulance and a well-integrated mechanism for obtaining and transmitting a 12-lead ECG continue to show excellent short- and long-term mortality results with prehospital fibrinolysis. Using data from a national registry, investigators in France reported 1-year mortality from STEMI of 6% in patients receiving prehospi-



Figure 6. Options for transportation of patients with STEMI and initial reperfusion treatment. Panel A: Patient transported by EMS after calling 9-1-1: Reperfusion in patients with STEMI can be accomplished by the pharmacologic (fibrinolysis) or catheter-based (primary PCI) approaches. Implementation of these strategies varies based on the mode of transportation of the patient and capabilities at the receiving hospital. Transport time to the hospital is variable from case to case, but the goal is to keep total ischemic time within 120 min. There are 3 possibilities: 1) If EMS has fibrinolytic capability and the patient qualifies for therapy, prehospital fibrinolysis should be started within 30 min of EMS arrival on scene; 2) If EMS is not capable of administering prehospital fibrinolysis and the patient is transported to a non-PCI-capable hospital, the hospital door-needle time should within 30 minutes for patients in whom fibrinolysis is indicated; 3) If EMS is not capable of administering prehospital fibrinolysis and the patient is transported to a PCI-capable hospital, the hospital door-to-balloon time should be within 90 min. Interhospital transfer: It is also appropriate to consider emergency interhospital transfer of the patient to a PCI-capable hospital for mechanical revascularization if: 1) there is a contraindication to fibrinolysis; 2) PCI can be initiated promptly (within 90 minutes after the patient presented to the initial receiving hospital or within 60 minutes compared to when fibrinolysis with a fibrin-specific agent could be initiated at the initial receiving hospital); 3) fibrinolysis is administered and is unsuccessful (i.e., "rescue PCI"). Secondary non-emergency interhospital transfer can be considered for recurrent ischemia. Patient self transport: Patient self-transportation is discouraged. If the patient arrives at a non-PCI capable hospital, the door-to-needle time should within 30 min. If the patient arrives at PCIcapable hospital, the door-to-balloon time should be within 90 min. The treatment options and time recommended after first hospital arrival are the same. Panel B: For patients who receive fibrinolysis, noninvasive risk stratification is recommended to identify the need for rescue PCI (failed fibrinolysis) or ischemia driven PCI. See Sections 6.3.1.6.4.5 and 6.3.1.6.7 in the full-text guidelines. Regardless of the initial method of reperfusion treatment, all patients should receive late hospital care and secondary prevention of STEMI. †The medical system goal is to facilitate rapid recognition and treatment of patients with STEMI such that door-to-needle (or medical contact-to-needle) for initiation of fibrinolytic therapy can be achieved within 30 minutes or that door-to-balloon (or medical contact-to-balloon) for PCI can be achieved within 90 minutes. These goals should not be understood as 'ideal' times, but rather the longest times that should be considered acceptable for a given system. Systems that are able to achieve even more rapid times for treatment of patients with STEMI should be encouraged. Modified with permission from Armstrong et al. Circulation 2003;107:2533-7 (155).



Antman *et al.* 2004 e101 ACC/AHA Practice Guidelines

Table 4. Prehospital	and Inhospital l	Fibrinolysis							
First Author	C								
Reference	Year	Location	Z	Design	Agent	End Points	Prehospital	Inhospital	d
McNeil (165)	1989	United Kingdom	57	R, PC, DB	Alteplase	Time to treatment (mean)	119 min	187 min	0.0005
		D				Confirmed MI	NR	NR	NR
						Cardiac mortality	7.4%	10%	NR
						EF (MUGA)	41%	35%	NR
Castaigne (166)	1989	France	93	R, PC	Anistreplase	Time to treatment	131 min	180 min	NR
						Confirmed MI	88%	NR	NR
						Inhospital mortality	3.5%	2.8%	NR
Barbash (167)	1990		87	R	Alteplase	Time to treatment	96 min	132 min	less than
						(mean)			0.0001
						Vontirmed MI	NR	NR	NR
						Mortanty (60 d) less than 120 min	0%0	0%0	NS
						Mortality (60 d)	202 C	6 901	lace than
						120 min or greater	0% C.7	0.0%0	n 01
						Discharge EF (MUGA)	50%	45%	SN
						CHF at discharge	4.6%	3.2%	0.07
Schofer (168)	1990		78	R, DB, PC	Urokinase	Time to treatment	85 min	137 min	less than
						(mean)			0.0005
							%66	NR	NR
						Innospital mortality	2.5%	5.2%	NS
						Coronary patency prior to discharge	61%	67%	NS
						EF (LV angiography)	20 22	2 C	
						Dlooding (minor)	%1¢	53%0	SN
						bleeding (minor)	2.5%	5.2%	NS
McAleer (169)	1992		145	R	Streptokinase	Time to treatment	138 min	172 min	less than
						Confirmed MI			0.02
						Mortality (14 d)	NR 2.3%	NR 11.5%	NR less than
						EF (MUGA) Bleeding (minor)	57%	52%	0.05 NS

e102

Continued on next page

ACC - www.acc.org w.americanheart.org

Antman et al. 2004 е



Figure 7. Major components of time delay between onset of symptoms from ST-elevation MI and restoration of flow in the infarct artery. Plotted sequentially from left to right are shown the time for patients to recognize symptoms and seek medical attention, transportation to the hospital, in-hospital decision-making, and implementation of reperfusion strategy, in time for restoration of flow once the reperfusion strategy has been initiated. The time to initiate fibrinolytic therapy is the "door-to-needle" (D-N) time; this is followed by the period of time required for pharmacologic restoration of flow. More time is required to move the patient to the catheterization laboratory for a percutaneous coronary interventional (PCI) procedure, referred to as the "door-to-balloon" (D-B) time, but restoration of flow in the epicardial infarct artery occurs promptly after PCI. At the bottom are shown a variety of methods for speeding the time to reperfusion along with the goals for the time intervals for the various components of the time delay. Cath = catheterization; PCI = percutaneous coronary intervention; min = minutes; ECG = electrocardiogram; MI = myocardial infarction; Rx = therapy. *These bar graphs are meant to be semiquantitative and not to scale. Modified with permission from Cannon et al. J Thromb Thrombol 1994;1:27-34 (180).

tal fibrinolysis compared with 11% in patients receiving inhospital fibrinolysis or primary PCI; the survival difference in favor of prehospital fibrinolysis persisted after adjustment for baseline characteristics (Danchin N; oral presentation, American Heart Association 2003 Annual Scientific Sessions, November 2003, Orlando, FL).

The difference between time to fibrinolytic therapy in the prehospital setting versus the hospital setting can be minimized by improved hospital triage with a decrease in doorto-needle time to within 30 minutes (179) (Figure 7) (180). However, only a small percentage (5% to 10%) of patients with chest pain in the prehospital setting have STEMI and are eligible for fibrinolytic therapy (159,181,182). Ensuring proper selection of patients for therapy can be difficult, and administration of therapy when it is contraindicated has important medical, legal, and economic implications. For these reasons, a general national policy of prehospital fibrinolytic therapy cannot currently be advocated. Prehospital fibrinolysis is reasonable in those settings in which physicians are present in the ambulance or prehospital transport times are more than 60 minutes in high-volume (more than 25000 runs per year) EMS systems (102).

Other considerations for implementing a prehospital fibrinolytic service include the ability to transmit ECGs, paramedic initial and ongoing training in ECG interpretation and MI treatment, online medical command, and the presence of a medical director with training/experience in management of STEMI and full-time paramedics (183). An example of the time saved by prehospital fibrinolysis is illustrated in a report from Scotland. The National Health Service in the United Kingdom established the standard that patients thought to be suffering from STEMI should receive fibrinolysis within 60 minutes of calling for medical assistance (http:// www.doh.gov.uk/nsf/coronarych3.htm). Three groups of patients in Scotland were studied: group 1 consisted of patients (n equals 107) within an urban area who received fibrinolytic therapy in the hospital, group 2 consisted of patients (n equals 43) from rural areas who received fibri-



Call-to-needle time (minutes)

Figure 8. Cumulative distribution of call-to-needle time. Group 1 refers to patients from an urban area receiving fibrinolysis in the hospital. Group 2 are patients from rural areas who received fibrinolysis in the hospital. Group 3 are patients from the study area who received prehospital fibrinolysis. Modified from Pedley et al. BMJ 2003;327:22-6 (183) with permission from the BMJ Publishing Group.

nolytic therapy in the hospital, and group 3 consisted of patients (n equals 28) in a rural area who received fibrinolytic therapy (tenecteplase) in the ambulance by trained paramedics who were supervised by a medical control officer (183). Administration of prehospital fibrinolytic therapy resulted in a median time savings of 73 minutes compared with patients from rural areas and 28 minutes compared with patients from urban areas (p less than 0.001). A greater proportion of patients who received prehospital fibrinolytic therapy were in compliance with the National Health Service standard of "call-to-needle" of 60 minutes (Figure 8) (183).

5.4. Prehospital Destination Protocols

Class I

- 1. Patients with STEMI who have cardiogenic shock and are less than 75 years of age should be brought immediately or secondarily transferred to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG) if it can be performed within 18 hours of onset of shock. (Level of Evidence: A)
- 2. Patients with STEMI who have contraindications to fibrinolytic therapy should be brought immediately or

secondarily transferred promptly (i.e., primaryreceiving hospital door-to-departure time less than 30 minutes) to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG). (Level of Evidence: B)

3. Every community should have a written protocol that guides EMS system personnel in determining where to take patients with suspected or confirmed STEMI. (Level of Evidence: C)

Class IIa

- **1.** It is reasonable that patients with STEMI who have cardiogenic shock and are 75 years of age or older be considered for immediate or prompt secondary transfer to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG) if it can be performed within 18 hours of onset of shock. (*Level of Evidence: B*)
- 2. It is reasonable that patients with STEMI who are at especially high risk of dying, including those with severe congestive heart failure (CHF), be considered for immediate or prompt secondary transfer (i.e., primary-receiving hospital door-to-departure time less

than 30 minutes) to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG). (Level of Evidence: B)

Every community should have a written protocol that guides EMS system personnel in determining where to take patients with suspected or confirmed STEMI. Active involvement of local healthcare providers, particularly cardiologists and emergency physicians, is needed to formulate local EMS destination protocols for these patients. In general, patients with suspected STEMI should be taken to the nearest appropriate hospital. However, patients with STEMI and shock are an exception to this general rule (Table 3).

Emergency revascularization improves 1-year survival in patients with STEMI complicated by cardiogenic shock (184). Subgroup analysis suggested a differential treatment effect, with the clearest benefit for those under 75 years of age. Therefore, whenever possible, patients with STEMI less than 75 years of age with shock should be transferred to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG). On the basis of observations in the SHOCK Trial Registry and other registries, it is reasonable to extend such considerations of transfer to invasive centers for elderly patients with shock (see Section 7.6.5). Patients with STEMI who have contraindications to fibrinolytic therapy and an especially high risk of dying, including severe CHF or cardiogenic shock, should be brought immediately or secondarily transferred promptly (i.e., primary-receiving hospital door-to-departure time less than 30 minutes) to facilities capable of cardiac catheterization and rapid revascularization (PCI or CABG). Given the importance of avoiding delays in time to reperfusion (see Section 6.3.1.6.3.1), direct transport to a facility capable of rapid revascularization is strongly preferred to interhospital transfer.

6. INITIAL RECOGNITION AND MANAGEMENT IN THE EMERGENCY DEPARTMENT

A variety of treatment options (Figure 3) (24-40) are available that can reduce mortality and morbidity in patients with STEMI, but the effectiveness of these therapies diminishes rapidly within the first several hours after symptoms onset (162,185). The traditional ED evaluation of patients with chest pain relies heavily on the patient's history, physical examination, and the ECG. This approach not infrequently fails to identify patients who are actually suffering from STEMI, which results in an inappropriate discharge home from the ED (186). Such missed MI patients are at relatively high risk of death or complications for the next 4 to 6 weeks after ED discharge (187-192).

In a large study on this subject, Pope et al. (188) found that 889 of 10689 patients who presented to 10 US hospital EDs with chest pain or other symptoms that suggested acute cardiac ischemia had STEMI; 19 patients (2.1%, 95% CI 1.1% to 3.1%) were discharged from the ED. Patients with STEMI

were more likely not to be hospitalized if they were nonwhite (odds ratio [OR] for discharge 4.5; 95% CI 1.8 to 11.8) or had a normal or nondiagnostic ECG (OR 7.7; 95% CI 2.9 to 20.2). The risk-adjusted mortality ratio for MI patients who were not hospitalized compared with those who were hospitalized was 1.9 (95% CI 0.7 to 5.2).

6.1. Optimal Strategies for Emergency Department Triage

Class I

Hospitals should establish multidisciplinary teams (including primary care physicians, emergency medicine physicians, cardiologists, nurses, and laboratorians) to develop guideline-based, institution-specific written protocols for triaging and managing patients who are seen in the prehospital setting or present to the ED with symptoms suggestive of STEMI. (Level of Evidence: B)

The advent of highly effective, time-dependent treatment for STEMI coupled with the need to reduce healthcare costs adds further incentive for clinicians to get the right answer quickly and to reduce unnecessary admissions and length of hospital stay. Investigators have tried various diagnostic tools such as clinical decision algorithms, cardiac biomarkers, echocardiography, and myocardial perfusion imaging in an attempt to avoid missing patients with MI or unstable angina. The most successful strategies to emerge thus far are designed to identify MI patients and, when clinically appropriate, screen for unstable angina and underlying coronary artery disease. Most strategies use a combination of cardiac biomarkers, short-term observation, diagnostic imaging, and provocative stress testing. An increasing number of highquality centers now use structured protocols, checklists, or critical pathways to screen patients with suspected MI or unstable angina (193-205). It does not appear to matter whether the institution designates itself a chest pain center. Rather, it is the multifaceted, structured approach to the problem that appears to provide clinical, cost-effective benefit (206,207). One randomized trial has confirmed the safety, efficacy, and cost-effectiveness of the structured decisionmaking approach compared with standard, unstructured care (208).

6.2. Initial Patient Evaluation

Class I

- 1. The delay from patient contact with the healthcare system (arrival at the ED or contact with paramedics) to initiation of fibrinolytic therapy should be less than 30 minutes. Alternatively, if PCI is chosen, the delay from patient contact with the healthcare system (typically, arrival at the ED, or contact with paramedics) to balloon inflation should be less than 90 minutes. (Level of Evidence: B)
- 2. The choice of initial STEMI treatment should be made by the emergency medicine physician on duty



Figure 9. Algorithm for evaluation and management of patients suspected of having acute coronary syndrome. STEMI = ST-elevation myocardial infarction. Modified from Braunwald et al. J Am Coll Cardiol 2000;36:970-1062 (4).

based on a predetermined, institution-specific, written protocol that is a collaborative effort of cardiologists (both those involved in coronary care unit management and interventionalists), emergency physicians, primary care physicians, nurses, and other appropriate personnel. For cases in which the initial diagnosis and treatment plan are unclear to the emergency physician or are not covered directly by the agreedupon protocol, immediate cardiology consultation is advisable. (*Level of Evidence: C*)

Regardless of the approach used, all patients presenting to the ED with chest discomfort or other symptoms suggestive of STEMI or unstable angina should be considered high-priority triage cases and should be evaluated and treated on the basis of a predetermined, institution-specific chest pain protocol. The protocol should include several diagnostic possibilities (Figure 9) (4). The patient should be placed on a cardiac monitor immediately, with emergency resuscitation equipment, including a defibrillator, nearby. An ECG should be performed and shown to an experienced emergency medicine physician within 10 minutes of ED arrival. If STEMI is present, the decision as to whether the patient will be treated with fibrinolytic therapy or primary PCI should be made within the next 10 minutes (Figure 7) (180). The goal for patients with STEMI should be to achieve a door-to-needle time of within 30 minutes and a door-to-balloon time of within 90 minutes (Figure 6) (155). If the initial ECG is not diagnostic, the patient remains symptomatic, and there is a high clinical suspicion for STEMI, serial ECGs at 5- to 10minute intervals or continuous ST-segment monitoring should be performed.

Ideally, such decisions should be made by the emergency medicine physician on duty in the ED based on a predetermined, institution-specific, written protocol that has been developed with input from cardiologists (both those involved in coronary care unit management and interventionalists), emergency medicine physicians, primary care physicians, nurses, and other appropriate personnel. For noninterventional hospitals, this will usually require formal, written transfer agreements and protocols that will permit expeditious transfer of patients who require urgent mechanical revascularization to the nearest appropriate interventional facility (Figure 6) (155). The protocol should also include the level of training and certification of personnel required to accompany the patient during transfer, the minimum equipment requirements, and the type(s) of transport vehicles (e.g., standard ground ambulance, mobile intensive care unit, helicopter, or fixed-wing aircraft) that can be used on the basis of patient condition. For cases in which the initial diagnosis and treatment plan are unclear to the emergency medicine physician or are not covered directly by the agreed-upon protocol, immediate cardiology consultation is advisable.

6.2.1. History

Class I

The targeted history of STEMI patients taken in the ED should ascertain whether the patient has had prior episodes of myocardial ischemia, such as stable or unstable angina, MI, coronary bypass surgery, or PCI. Evaluation of the patient's complaints should focus on chest discomfort, associated symptoms, sexand age-related differences in presentation, hypertension, diabetes mellitus, possibility of aortic dissection, risk of bleeding, and clinical cerebrovascular disease (amaurosis fugax, face/limb weakness or clumsiness,

face/limb numbness or sensory loss, ataxia, or vertigo). (Level of Evidence: C)

The history taken in the ED must be concise and detailed enough to establish the probability of STEMI but should be obtained expeditiously so as not to delay implementation of reperfusion therapy.

Chest Discomfort. The severity of discomfort varies and is typically graded on a scale of 1 to 10, with 10 being the most severe pain. It is important to keep in mind that many patients will not admit having chest "pain" but will acknowledge the presence of chest "discomfort," because of their definition of pain. The chest discomfort is often described as a crushing, vice-like constriction, a feeling equivalent to an "elephant sitting on the chest," or heartburn. Usually, the discomfort is substernal but may originate in or radiate to areas such as the neck, jaw, interscapular area, upper extremities, and epigastrium. The duration of the discomfort, which typically lasts longer than 30 minutes, may wax and wane and may be remitting. It may be described as "indigestion in the chest" and occasionally may be relieved with belching. The possibility of precipitation of STEMI by use of illicit drugs such as cocaine should be considered.

The targeted history of patients with STEMI taken in the ED should ascertain whether the patient has had prior episodes of myocardial ischemia such as stable or unstable angina, MI, coronary bypass surgery, or PCI. Evaluation of the patient's complaints should focus on chest discomfort, associated symptoms, sex- and age-related differences in presentation, hypertension, diabetes mellitus, possibility of aortic dissection, risk of bleeding, and clinical cerebrovascular disease (amaurosis fugax, face/limb weakness or clumsiness, face/limb numbness or sensory loss, ataxia, or vertigo).

Associated Symptoms. Other symptoms to be aware of when taking a patient's history include nausea and vomiting. Diaphoresis associated with a pale complexion may also appear, as well as weakness or profound fatigue. Dizziness, lightheadedness, syncope, and paresthesia may occur because of pain and hyperventilation.

Hypertension. Hypertension should be assessed, because chronic, severe, poorly controlled hypertension or severe uncontrolled hypertension on presentation is a relative contraindication to fibrinolytic therapy (see Section 6.3.1.6.3.2).

Sex- and Age-Related Differences in Presentation. It has been noted in studies that women present with STEMI at an older age and later after the onset of symptoms than men (53,210). There must be an elevated index of suspicion during the evaluation of women for STEMI. Although some variation exists, when large databases of MI patients are examined, symptom profiles for STEMI by sex generally appear more similar than different between men and women (211-215). Elderly patients with STEMI are significantly less likely than younger patients to complain of chest discomfort. However, elderly patients with STEMI are more likely to complain of shortness of breath, as well as other atypical symptoms such as syncope or unexplained nausea. (181). *Diabetes Mellitus*. Diabetics may have impaired angina (pain) recognition, especially in the presence of autonomic neuropathy. A diabetic may misinterpret dyspnea, nausea, vomiting, fatigue, and diaphoresis as disturbance of diabetic control. Up to 50% of diabetic individuals with type 2 diabetes for longer than 10 years will have autonomic nervous system dysfunction manifested by impaired heart rate variability. Diabetics with STEMI should be evaluated for renal dysfunction (216).

Possibility of Aortic Dissection. Severe tearing pain radiating directly to the back associated with dyspnea or syncope and without ECG changes indicative of STEMI should raise the suspicion for aortic dissection, and appropriate studies should be undertaken. Clinicians should have a heightened index of suspicion for aortic dissection in elderly hypertensive patients. However, it must be kept in mind that the dissection may extend to the pericardial sac and produce cardiac tamponade or disrupt the origin of a coronary artery.

Risk of Bleeding. Patients should be questioned about previous bleeding problems, e.g., during surgery or dental procedures, history of ulcer disease, cerebral vascular accidents, unexplained anemia, or melena. The use of antiplatelet, antithrombin, and fibrinolytic agents as part of the treatment for STEMI will exacerbate any underlying bleeding risks.

Clinical Cerebrovascular Disease. The patient with STEMI frequently has medical conditions that are risk factors for both MI and stroke. Evidence for prior episodes suggestive of clinical cerebrovascular disease should be sought. For example, the patient should be asked whether he/she has ever had symptoms of transient retinal or cerebral ischemia such as amaurosis fugax, face/limb weakness or clumsiness, face/limb numbness or sensory loss, ataxia, or vertigo. Transient ischemic attacks (TIAs) typically last less than 30 minutes, whereas symptoms that last more than 60 to 90 minutes are more likely to indicate the presence of a stroke (217). In addition, the patient should be asked whether he/she has ever had an ischemic stroke, intracerebral hemorrhage [ICH], or subarachnoid hemorrhage. A brief summary of the details for diagnosis of the different stroke subtypes is available (218). Finally, a history of cognitive decline/dementia may indicate the presence of cerebral amyloid angiopathy and increased risk of ICH, and information regarding head and facial trauma should be obtained.

6.2.2. Physical Examination

Class I

1. A physical examination should be performed to aid in the diagnosis and assessment of the extent, location, and presence of complications of STEMI. (Level of Evidence: C) Antman et al. 2004

Table 5. Brief Physical Examination in the Emergency Department

- 1. Airway, Breathing, Circulation (ABC)
- 2. Vital signs, general observation
- 3. Presence or absence of jugular venous distension
- 4. Pulmonary auscultation for rales
- 5. Cardiac auscultation for murmurs and gallops
- 6. Presence or absence of stroke
- 7. Presence or absence of pulses
- 8. Presence or absence of systemic hypoperfusion (cool, clammy, pale, ashen)
- 2. A brief, focused, and limited neurological examination to look for evidence of prior stroke or cognitive deficits should be performed on STEMI patients before administration of fibrinolytic therapy. (Level of Evidence: C)

A brief physical examination may promote rapid triage (Table 5), whereas a more detailed physical examination aids in the differential diagnosis and is useful for assessing the extent, location, and presence of complications of STEMI (Tables 6 and 7) (219).

Evidence of prior stroke or dementia may be suggested by the finding on examination of focal neurological or cognitive deficits (Table 6). A brief but focused examination can identify focal neurological or cognitive deficits.

6.2.2.1. Differential Diagnosis

The differential diagnosis of STEMI includes conditions that can be exacerbated by fibrinolysis and anticoagulation (Table 8). The pain of aortic dissection is typically described as searing, ripping, or tearing and frequently radiates to the back or lower extremities. The pain is worse at onset and lasts for hours. Major pulses may be absent, and a murmur of aortic regurgitation may be present. A transesophageal echocardiogram, computed tomography (CT) scan, or magnetic resonance imaging scan is useful for establishing the diagnosis of aortic dissection. Active peptic ulcer disease can be present with chest or epigastric pain, sometimes radiating posteriorly, and may be associated with syncope, hematemesis, or melena. Free subdiaphragmatic air may be seen on upright chest X-ray in perforations. Acute pericarditis may show PR-segment depression and ST-segment elevation on the ECG but without reciprocal ST-segment depression (220). Pain from pericarditis is usually pleuritic and can radiate to the shoulder and trapezius ridge and is often relieved by sitting up and leaning forward, characteristics not found in STEMI. A rub is often present. Pulmonary embolus, with or without infarction, presents with dyspnea and knifelike pleuritic pain, sometimes with hemoptysis. Pulmonary embolism can present with chest pain similar to that of STEMI. Costochondral pain is described as sharp or sticking,

Table 6. Physical Findings and Possible Implications in Complicated and Uncomplicated ST-Elevation Myocardial Infarction Patients

Cardiovascular

- General: Restless agitated, anguished facies, clenched fist (Levine's sign)
- Skin: Cool, clammy, pale, ashen
- Low-grade fever: Nonspecific response to myocardial necrosis
- Hypertension, tachycardia: High sympathetic tone (anterior MI)
- Hypotension, bradycardia: High vagal tone (inferior-posterior MI) Small-volume pulses: Low cardiac output
- Fast, slow, or irregular pulse: Atrial or ventricular arrhythmias, heart block
- Paradoxical "ectopic" systolic impulse: LV dyskinesis, ventricular aneurysm (anterior MI)
- Soft S₁: Decreased LV contractility; first-degree AV block (inferior MI)
- S₄ gallop: Decreased LV compliance
- Paradoxically split S₂: Severe LV dysfunction, LBBB
- S_3 gallop, pulmonary rales, pulsus alternans: LV systolic dysfunction (signs of CHF greater than 25% of myocardium)
- Hypotension: Skin cool, clammy, cyanotic; CNS altered mental status; kidneys – oliguria (signs of cardiogenic shock)
- Jugular venous distension: with Kussmaul's sign, hypotension, RV S_4 and S_3 gallops, clear lungs (RV infarction)
- Systolic murmur of VSR: VSR (LSB, palpable thrill common) Differentiate from systolic murmur of MR: papillary muscle rup-
- ture
- Pericardial friction rub: Pericarditis (accompanies transmural MI) – late post-MI (Dressler's) syndrome
- Signs of cardiac tamponade, EM dissociation: Cardiac rupture Absent pulses and murmur of aortic regurgitation: Aortic dissection

Screening Neurological Examination

- Cognitive disorientation: memory loss, dysarthria, aphasia, hemispatial neglect
- Motor: facial asymmetry, pronator drift, reflex symmetry, limb dysmetria

Sensory: loss of sensation of pinprick

MI = myocardial infarction; LV = left ventricular; AV = atrioventricular; LBBB = left bundle-branch block; CHF = congestive heart failure; CNS = central nervous system; RV = right ventricular; VSR = ventricular septal rupture; LSB = left sternal border; mitral regurgitation; EM = electromechanical.

with associated localized tenderness. Pneumothorax may present with acute dyspnea, pleuritic pain, and differential decrease in breath sounds with hyperresonance over 1 lung field. Acute cholecystitis may mimic STEMI, and rightupper-quadrant abdominal tenderness should be sought on physical examination.

6.2.3. Electrocardiogram

Class I

1. A 12-lead ECG should be performed and shown to an experienced emergency physician within 10 minutes of ED arrival on all patients with chest discomfort (or anginal equivalent) or other symptoms suggestive of STEMI. (Level of Evidence: C)

			Fibrinolytic Trials (30 Days)				
Killin	Killin and	GIS	SSI-1 (157)	International Study Group:	ASSENT-2:		
Class	Kimball (Inhospital)	Placebo	Fibrinolytic	Fibrinolytic (354)	Fibrinolytic (28)		
I	6	7	6	5	5		
II	17	20	16	18	13		
III	38	39	33	32	26		
IV	81	70	70	72	56†		

Table 7. Percent Mortality by Killip Class*

Class I = no rales, no S₃; Class II = rales less than 50%; Class III = pulmonary edema; Class IV = cardiogenic shock.

*Values cited are subject to survivor bias.

†Highly selected group of patients.

Modified with permission from Topol. Textbook of Cardiovascular Medicine. 2nd ed. Philadelphia, PA: Lippincott-Williams & Wilkins; 2002:438 (219).

- 2. If the initial ECG is not diagnostic of STEMI but the patient remains symptomatic, and there is a high clinical suspicion for STEMI, serial ECGs at 5- to 10-minute intervals or continuous 12-lead ST-segment monitoring should be performed to detect the potential development of ST elevation. (Level of Evidence: C)
- 3. In patients with inferior STEMI, right-sided ECG leads should be obtained to screen for ST elevation suggestive of RV infarction. (See Section 7.6.6 and the

Table 8. Differential Diagnosis of ST-Elevation Myocardial Infarction

Life-threatening
Aortic dissection
Pulmonary embolus
Perforating ulcer
Tension pneumothorax
Boerhaave syndrome (esophageal rupture with mediastinitis)
Other cardiovascular and nonischemic
Pericarditis
Atypical angina
Early repolarization
Wolff-Parkinson-White syndrome
Deeply inverted T waves suggestive of a central nervous system
lesion or apical hypertrophic cardiomyopathy
LV hypertrophy with strain
Brugada syndrome
Myocarditis
Hyperkalemia
Bundle-branch blocks
Vasospastic angina
Hypertrophic cardiomyopathy
Other noncardiac
Gastroesophageal reflux (GERD) and spasm
Chest-wall pain
Pleurisy
Peptic ulcer disease
Panic attack
Biliary or pancreatic pain
Cervical disc or neuropathic pain
Somatization and psychogenic pain disorder

ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography) (Level of Evidence: B)

The 12-lead ECG in the ED is at the center of the therapeutic decision pathway because of the strong evidence that ST-segment elevation identifies patients who benefit from reperfusion therapy (221). Mortality increases with the number of ECG leads showing ST elevation. Important predictors of mortality on the initial 12-lead ECG include left bundlebranch block (LBBB) and anterior location of infarction (Figure 10) (222,223). Diagnostic criteria of greater than 0.1 mV in leads V₁ through V₄ may have reduced specificity for STEMI in patients with early repolarization. Some evidence exists to support the use of greater than or equal to 0.2 mV anteroseptal elevation as a preferable threshold for diagnosing STEMI, because a higher proportion of patients are correctly classified as having STEMI than with a threshold of greater than 0.1 mV in these leads (221).

In the absence of ST elevation, there is no evidence of benefit of fibrinolytic therapy for patients with normal ECG or nonspecific changes, and there is some suggestion of harm (including increased bleeding risk) for patients with ST-segment depression only (221,224). Notwithstanding this, fibrinolytic therapy may be appropriate when there is marked STsegment depression confined to leads V_1 through V_4 and accompanied by tall R waves in the right precordial leads and upright T waves indicative of a true posterior injury current and circumflex coronary occlusion. In circumstances where there is a suggestive clinical history and suggestive evidence of true posterior infarction, confirmatory data from posterior leads (i.e., V_7 and V_8) as well as 2-dimensional echocardiography may be especially helpful; this latter investigation has a high negative predictive value (225,226). Primary PCI is another reperfusion strategy that may be effective in patients with true posterior MI (see Section 6.3.1.6.4.2).

Initial errors in ECG interpretation can result in up to 12% of patients being categorized inappropriately, demonstrating a potential benefit of accurate computer-interpreted electrocardiography and fax transmission to an expert (227). It is less likely that STEMI is present if the upward-directed ST-segment changes are concave rather than convex (228).



Figure 10. Risk stratification: electrocardiogram (ECG). This 12-lead ECG was obtained from a middle-aged man admitted with an extensive anterior acute myocardial infarction. (Note pathological Q waves in the precordial leads and marked repolarization abnormalities in the anterior and lateral leads.) A 5-beat salvo of non-sustained VT is seen extending over the transition between leads III and aVF. Reprinted with permission from Antman and Rutherford. Coronary Care Medicine. Boston, MA: Martinus Nijhoff Publishing; 1986:81 (223).

Because lethal ventricular arrhythmias may develop abruptly in patients with STEMI, all patients should be monitored electrocardiographically on arrival in the ED. It is important to examine serial tracings approximately 5 to 10 minutes apart, or if symptoms recur, during evaluation in the ED for development of ST elevation if the initial ECG is nondiagnostic. ST elevation may also be detected by intermittent visual inspection of the oscilloscope or auditory alarms in systems with continuous ST-segment monitoring capability.

Although the Fibrinolytic Therapy Trialists' (FTT) Collaborative Group overview indicates that patients with

Table 9. Laboratory Evaluations for Management of ST-Elevation

 Myocardial Infarction

Serum biomarkers for cardiac damage (do not wait for results before implementing reperfusion strategy) CBC with platelet count

INR aPTT Electrolytes and magnesium BUN Creatinine Glucose Serum lipids

CBC = complete blood count; INR = international normalized ratio; aPTT = activated partial thomboplastin time; BUN = blood urea nitrogen.

new or presumably new LBBB are at high risk when presenting with presumed MI, this ECG presentation is a frequent cause of delay or lack of reperfusion therapy because of the concern of the validity of the ECG criteria for MI diagnosis and the risk of therapy. This is also a situation in which direct PCI may be preferable to fibrinolytic therapy (156). It has been suggested that patients with new or presumably new LBBB coupled with a typical ischemic history be approached with a plan to rule in MI using 1 of 3 ECG criteria that provide independent diagnostic value. These consist of ST elevation greater than or equal to 0.1 mV in leads with a positive QRS, ST depression greater than or equal to 0.5 mV in leads with a negative QRS (229,230).

6.2.4. Laboratory Examinations

Class I

Laboratory examinations should be performed as part of the management of STEMI patients but should not delay the implementation of reperfusion therapy. For specific laboratory examinations, see Table 9. (Level of Evidence: C)

In addition to serum cardiac biomarkers for cardiac damage, several routine evaluations have important implications

Biomarker	Molecular Weight, Da	Range of Times to Initial Elevation, h	Mean Time to Peak Elevations (Nonreperfused)	Time to Return to Normal Range
Frequently used in clinical practice				
CK-MB	86 000	3-12 h	24 h	48-72 h
cTnI	23 500	3-12 h	24 h	5-10 d
cTnT	33 000	3-12 h	12 h–2 d	5-14 d
Infrequently used in				
clinical practice				
Myoglobin	17 800	1-4 h	6-7 h	24 h
CK-MB tissue isoform	86 000	2-6 h	18 h	Unknown
CK-MM tissue isoform	86 000	1-6 h	12 h	38 h

 Table 10. Molecular Biomarkers for the Evaluation of Patients With ST-Elevation Myocardial Infarction

Da = Daltons; h = hours; CK-MB = MB isoenzyme of creatine kinase; cTnI = cardiac troponin I; cTnT = cardiac troponin T; CK-MM = MM isoenzyme of creatine

for management of patients with STEMI (Table 9). Although these studies should be undertaken when the patient is first examined, therapeutic decisions should not be delayed until results are obtained because of the crucial role of time to therapy in STEMI.

6.2.5. Biomarkers of Cardiac Damage

Class I

- 1. Cardiac-specific troponins should be used as the optimum biomarkers for the evaluation of patients with STEMI who have coexistent skeletal muscle injury. (Level of Evidence: C)
- 2. For patients with ST elevation on the 12-lead ECG and symptoms of STEMI, reperfusion therapy should be initiated as soon as possible and is not contingent on a biomarker assay. (Level of Evidence: C)

Class IIa

Serial biomarker measurements can be useful to provide supportive noninvasive evidence of reperfusion of the infarct artery after fibrinolytic therapy in patients not undergoing angiography within the first 24 hours after fibrinolytic therapy. (Level of Evidence: B)

Class III

Serial biomarker measurements should not be relied upon to diagnose reinfarction within the first 18 hours after the onset of STEMI. (*Level of Evidence: C*)

The nomenclature of acute coronary syndromes is illustrated in Figure 2 (8-10). The central position of the 12-lead ECG and initial triage of patients are emphasized. Serum cardiac biomarkers (creatine kinase [CK], CK-MB, cardiac-specific troponins, myoglobin) are useful for confirming the diagnosis of MI and estimating infarct size. Serum cardiac biomarkers also provide valuable prognostic information. For patients with ST-segment elevation, the diagnosis of STEMI is secure; initiation of reperfusion therapy should not be delayed while awaiting the results of a cardiac biomarker assay (231,232) (Table 10). Quantitative analysis of cardiac biomarker measurements provides prognostic information as well as a noninvasive assessment of the likelihood that the patient has undergone successful reperfusion when fibrinolytic therapy is administered (Figure 11) (233,234).

Because there are differences in the clinical need for biomarkers in STEMI versus NSTEMI patients and differences in the characteristics of the various cardiac biomarkers, preferential use of a particular biomarker should be based on the clinical syndrome. CK-MB is found in the skeletal muscle and blood of healthy subjects; therefore, the cutoff value for an elevated CK-MB is typically set a few units above the upper end of the reference (normal) range. In contrast, because cardiac troponin I (cTnI) and cardiac troponin T (cTnT) are not normally detected in the blood of healthy people, the definition of an abnormally increased level is a value that exceeds that of 99% of a reference control group. Given the nearly absolute myocardial tissue specificity and high sensitivity for even microscopic zones of myocardial necrosis, the ACC and the European Society of Cardiology subsequently declared cardiac troponins to be the preferred biomarker for diagnosing MI (233). A single cutoff point was recommended such that an MI would be diagnosed if, as a result of myocardial ischemia, cTnI or cTnT were detected at least once within 24 hours of the index clinical event at a level exceeding the 99th percentile of the mean value measured in a normal control population (233). The superior sensitivity makes troponin the preferred marker for patients with UA/NSTEMI. In contrast, patients with STEMI are recognized on the basis of the 12-lead ECG, and in general, subsequent confirmation of MI can be ascertained by measurement of any of the available cardiac biomarkers. Occasionally, a very small infarct will be missed by CK-MB; therefore, troponin should be measured for patients suspected to have STEMI who have negative serial CK-MBs.

It should be recognized that in patients with STEMI, cTnT and cTnI may first begin to rise above the reference limit by



Figure 11. Cardiac biomarkers in ST-elevation myocardial infarction (STEMI). Typical cardiac biomarkers that are used to evaluate patients with STEMI include the MB isoenzyme of CK (CK-MB) and cardiac specific troponins. The horizontal line depicts the upper reference limit (URL) for the cardiac biomarker in the clinical chemistry laboratory. The URL is that value representing the 99th percentile of a reference control group without STEMI. The kinetics of release of CK-MB and cardiac troponin in patients who do not undergo reperfusion are shown in the solid green and red curves as multiples of the URL. Note that when patients with STEMI undergo reperfusion, as depicted in the dashed green and red curves, the cardiac biomarkers are detected sooner, rise to a higher peak value, but decline more rapidly, resulting in a smaller area under the curve and limitation of infarct size. Modified with permission from Alpert et al. J Am Coll Cardiol 2000;36:959 (233) and Wu et al. Clin Chem 1999;45:1104 (234).

3 to 6 hours from the onset of ischemic symptoms. Therefore, a significant number of patients will present to the emergency room with negative biomarkers. Myoglobin, a low-molecular-weight heme protein found in cardiac and skeletal muscle, is not cardiac specific but is released more rapidly from infarcted myocardium than CK-MB and may be detected as early as 2 hours after STEMI.

Antman et al. 2004

In some patients, cardiac-specific troponins may not be detectable for up to 6 hours after onset of chest pain. Thus, when CK-MB, cTnI, or cTnT levels are elevated in less than 6 hours after the onset of discomfort in patients with STEMI, clinicians should suspect that an antecedent episode of unstable angina was in fact MI and the patient is exhibiting a stuttering course of occlusion and release of the infarct artery. Data from the Global Utilization of Streptokinase and TPA for Occluded Arteries (GUSTO) III Study suggest that patients with STEMI who have elevated cTnT levels and who are less than 6 hours from the onset of discomfort have an increased mortality risk (235).

CK-MB is the preferred, widely available cardiac biomarker for most patients with STEMI, for whom the need to diagnose reinfarction and noninvasively assess reperfusion is greater than the need to make the diagnosis. By mapping the time course of the rise and fall of a biomarker (typically CK-MB), clinicians can detect an interruption of the progressive fall of the biomarker level to a point below the upper reference limit (Figure 11) (233,234). Re-elevation of the biomarker level is evidence of myocardial reinfarction (Figure 12). A more rapidly rising and falling biomarker such as CK-MB or myoglobin is superior for diagnosing reinfarction. As a consequence of continuous release from a degenerating contractile apparatus in necrotic myocytes, elevations of cTnI may persist for 7 to 10 days after MI, and elevations of cTnT may persist for up to 10 to 14 days. The more protracted time course of kinetic release of cTnI and cTnT limits the ability of clinicians to make the diagnosis of reinfarction within several days after the index STEMI event. An algorithm illustrating the decision-making process that incorpo-



tion; CABG = coronary artery bypass graft surgery; h = hours; Hem Decomp = hemodynamic decompensation; CK-MB = MB isoenzyme of creatine kinase; ULN = upper limit of normal; re-elev = re-elevated; ECG = electrocardiogram. * Because of a more protracted time course of elevations after the index STEMI event, diagnosis of recurent MI may be problematic if cardiac troponins are used. A more rapidly rising and falling cardiac biomarker such as CK-MB is preferable.

rates biomarker measurements, ECG findings, clinical symptoms, and, if available, autopsy data for making the diagnosis of reinfarction is shown in Figure 12.

In addition to monitoring the patient for resolution of ischemic-type chest discomfort and regression of the magnitude of ST-segment elevation on the ECG, clinicians can obtain serial measurements of serum cardiac markers to buttress the noninvasive diagnosis of reperfusion of the infarct-related artery after fibrinolytic therapy (Figure 11) (233,234,236). An early peak of CK-MB (12 to 18 hours) suggests reperfusion. Because of its rapid-release kinetics, myoglobin is also an attractive marker for the early diagnosis of reperfusion.

CK-MB isoforms are another serum cardiac biomarker less frequently used for evaluating patients with STEMI. CK-MB exists in only 1 form in myocardial tissue but in different isoforms (or subforms) in the plasma. An absolute level of CK-MB2 greater than 1 U/L or a ratio of CK-MB2 to CK-MB1 of 1.5 has improved sensitivity and specificity for diagnosis of MI within the first 6 hours compared with conventional assays for CK-MB (237).

6.2.5.1. Bedside Testing for Serum Cardiac Biomarkers

Class I

- 1. Although handheld bedside (point-of-care) assays may be used for a qualitative assessment of the presence of an elevated level of a serum cardiac biomarker, subsequent measurements of cardiac biomarker levels should be done with a quantitative test. (Level of Evidence: B)
- 2. For patients with ST elevation on the 12-lead ECG and symptoms of STEMI, reperfusion therapy should be initiated as soon as possible and is not contingent on a bedside biomarker assay. (*Level of Evidence: C*)

Handheld rapid bedside assays are clinically available for measuring cTnI, cTnT, myoglobin, and CK-MB, but in general, bedside assays are less sensitive and less precise than quantitative assays. Small desktop rapid analyzers are also available for the same purpose. A rapid, high-voltage electrophoretic system is available for measuring CK-MB isoforms. Monitoring the timing of the appearance of a positive bedside assay result may provide clinicians with a tool for a semiquantitative estimate of a serum cardiac biomarker level at the patient's bedside (238). A positive bedside test should be confirmed by a conventional quantitative test. However, reperfusion therapy should not be delayed while one awaits the results of a quantitative assay.

6.2.6. Imaging

Class I

1. Patients with STEMI should have a portable chest X-ray, but this should not delay implementation of reperfusion therapy (unless a potential contraindi-

cation is suspected, such as aortic dissection). (Level of Evidence: C)

2. Imaging studies such as a high-quality portable chest X-ray, transthoracic and/or transesophageal echocardiography, and a contrast chest CT scan or magnetic resonance imaging scan should be used for differentiating STEMI from aortic dissection in patients for whom this distinction is initially unclear. (Level of Evidence: B)

Class IIa

Portable echocardiography is reasonable to clarify the diagnosis of STEMI and allow risk stratification of patients with chest pain who present to the ED, especially if the diagnosis of STEMI is confounded by LBBB or pacing or if there is suspicion of posterior STEMI with anterior ST depressions. (See Section 7.6.7, Mechanical Causes of Heart Failure/Low-Output Syndrome.) (Level of Evidence: B)

Class III

Single-photon emission CT (SPECT) radionuclide imaging should not be performed to diagnose STEMI in patients for whom the diagnosis of STEMI is evident on the ECG. (Level of Evidence: B)

Various forms of imaging are often used to evaluate patients with symptoms that are suggestive of MI or acute coronary syndrome. Cardiac imaging can be of value in further determining the cause of chest discomfort in patients suspected of having an acute MI or unstable angina but whose initial ECG is normal or nondiagnostic. The 2 most studied techniques thus far have been echocardiography and radionuclide imaging.

Bedside echocardiography is useful for diagnosis and risk stratification of chest pain patients in the ED (226). A highquality portable chest X-ray, transthoracic and/or transesophageal echocardiography, and a contrast chest CT scan can be useful for differentiating acute MI from aortic dissection in patients for whom this distinction is clinically unclear.

SPECT radionuclide imaging at rest is not routinely indicated to establish the diagnosis of MI in patients with STEMI, although it can provide valuable, accurate diagnostic and prognostic information in patients who present to the ED with symptoms suggestive of acute cardiac ischemia and a normal or nondiagnostic ECG (239). During the recuperative phase of hospitalization for STEMI, SPECT imaging can be used to study myocardial perfusion and to look for segmental abnormalities of LV wall motion.

6.2.7. Global Risk Assessment Tools

Global risk assessment provides an opportunity to integrate various patient characteristics into a single score that can convey an overall estimate of a patient's prognosis over a given period of time. Beyond being informative about prognosis, the general value of these risk assessment tools is that they can influence clinical strategies. In general, the risk of the intervention should be commensurate with the underlying risk of the patient without the intervention and the expected benefit of the intervention. That is, a high-risk intervention should usually not be used for a very low-risk patient. The expected increase in risk associated with the intervention would very likely outweigh the expected benefit.

Several risk assessment tools have been proposed for patients with STEMI (240-243). One such tool uses clinical and ECG characteristics to predict risk of mortality for a patient if and if not treated with fibrinolytic therapy, as well as the risk of intracranial hemorrhage and major bleeding. This decision aid suggests that some patients with small infarctions may not have a substantial benefit from fibrinolytic therapy, particularly those who may have a risk factor for bleeding. These estimates are based on trials and registries. The use of this aid in clinical practice did not increase the use of fibrinolytic therapy overall (244). Whether the widespread application of these tools can improve decision making is not clear. Nevertheless, they provide estimates of risk that may be useful in the tailoring of therapy for individual patients. In general, however, patients who present with STEMI require evaluation for rapid reperfusion therapy and treatment with aspirin, beta-blockers, and ACE inhibitors. Nevertheless, any patient with a risk from the intervention that exceeds their STEMI risk reduction will, on average, do better without that treatment. This group will generally include patients with a higher risk from the intervention or a lower absolute risk reduction (generally because of a low absolute STEMI risk). This issue may be particularly important for younger patients, who tend to have a lower absolute risk of mortality (245), and for the elderly, who tend to have a higher risk from interventions, particularly with respect to fibrinolytic therapy (246). Precise estimates of risks and benefits are useful because the low STEMI risk in younger patients is often accompanied by a lower risk of interventions. In contrast, in the elderly, the higher intervention risk is accompanied by a higher STEMI risk (and thus a larger absolute reduction in risk with the intervention) (247).

The use of any risk assessment tool should not contribute to any delay in providing the time-sensitive assessment and treatment strategies that patients with STEMI require. Further research is necessary to determine how these tools may best contribute to optimizing patient outcomes.

6.3. Management

6.3.1. Routine Measures

6.3.1.1. Oxygen

Class I

Supplemental oxygen should be administered to patients with arterial oxygen desaturation (SaO₂ less than 90%). (Level of Evidence: B)

Class IIa

It is reasonable to administer supplemental oxygen to all patients with uncomplicated STEMI during the first 6 hours. (Level of Evidence: C) It has become universal practice to administer oxygen, usually by nasal prongs, to virtually all patients suspected of having acute ischemic-type chest discomfort, although it is not known whether this therapy limits myocardial damage or reduces morbidity or mortality. If oxygen saturation monitoring is used, therapy with supplemental oxygen is indicated if the saturation is less than 90%. Experimental results indicate that breathing oxygen may limit ischemic myocardial injury (248), and there is evidence that oxygen administration reduces ST-segment elevation (249). The rationale for use of oxygen is based on the observation that even with uncomplicated MI, some patients are modestly hypoxemic initially, presumably because of ventilation-perfusion mismatch and excessive lung water (250).

In patients with severe congestive heart failure, pulmonary edema, or a mechanical complication of STEMI, significant hypoxemia may not be corrected with supplemental oxygen alone. Continuous positive-pressure breathing or endotracheal intubation and mechanical ventilation may be required in such cases (251).

For patients without complications, excess administration of oxygen can lead to systemic vasoconstriction, and high flow rates can be harmful to patients with chronic obstructive airway disease. In the absence of compelling evidence for established benefit in uncomplicated cases, and in view of its expense, there appears to be little justification for continuing its routine use beyond 6 hours.

6.3.1.2. Nitroglycerin

Class I

- **1.** Patients with ongoing ischemic discomfort should receive sublingual nitroglycerin (0.4 mg) every 5 minutes for a total of 3 doses, after which an assessment should be made about the need for intravenous nitroglycerin. (*Level of Evidence: C*)
- 2. Intravenous nitroglycerin is indicated for relief of ongoing ischemic discomfort, control of hypertension, or management of pulmonary congestion. (Level of Evidence: C)

Class III

- 1. Nitrates should not be administered to patients with systolic blood pressure less than 90 mm Hg or greater than or equal to 30 mm Hg below baseline, severe bradycardia (less than 50 beats per minute [bpm]), tachycardia (more than 100 bpm), or suspected RV infarction. (Level of Evidence: C)
- 2. Nitrates should not be administered to patients who have received a phosphodiesterase inhibitor for erectile dysfunction within the last 24 hours (48 hours for tadalafil). (Level of Evidence: B)

The physiological effects of nitrates include reducing preload and afterload through peripheral arterial and venous dilation, relaxation of epicardial coronary arteries to improve coronary flow, and dilation of collateral vessels, potentially creating a more favorable subendocardial to epicardial flow ratio (252-254). Vasodilation of the coronary arteries, especially at or adjacent to sites of recent plaque disruption, may be particularly beneficial for the patient with acute infarction. Nitrate-induced vasodilatation may also have particular utility in those rare patients with coronary spasm presenting as STEMI.

Clinical trial results have suggested only a modest benefit from nitroglycerin used acutely in STEMI and continued subsequently. A pooled analysis of more than 80 000 patients treated with nitrate-like preparations intravenously or orally in 22 trials revealed a mortality rate of 7.7% in the control group, which was reduced to 7.4% in the nitrate group. These data are consistent with a possible small treatment effect of nitrates on mortality such that 3 to 4 fewer deaths would occur for every 1000 patients treated (152).

Nitroglycerin may be administered to relieve ischemic pain and is clearly indicated as a vasodilator in patients with STEMI associated with LV failure. Nitrates in all forms should be avoided in patients with initial systolic blood pressures less than 90 mm Hg or greater than or equal to 30 mm Hg below baseline, marked bradycardia or tachycardia (256), or known or suspected RV infarction. Patients with RV infarction are especially dependent on adequate RV preload to maintain cardiac output and may experience profound hypotension during administration of nitrates (257). Phosphodiesterase inhibitors potentiate the hypotensive effects of nitrates because of their mechanism of action in releasing nitric oxide and increasing cyclic guanosine monophosphate (258). Therefore, it is useful clinical practice to ascertain whether such agents have been used, and nitrates should not be administered to patients who have received a phosphodiesterase inhibitor for erectile dysfunction in the prior 24 hours (48 hours for tadalafil).

Nitroglycerin is commonly given sublingually at doses of 0.4 mg when patients present with STEMI. Arterial pressure may decline precipitously because of limited control of the initial dose and rate of absorption. An intravenous infusion of nitroglycerin allows clinicians to titrate the therapy in response to the patient's blood pressure. A useful intravenous nitroglycerin regimen employs an initial infusion rate of 5 to 10 mcg per minute with increases of 5 to 20 mcg per minute until symptoms are relieved or mean arterial blood pressure is reduced by 10% of its baseline level in normotensive patients and by up to 30% for hypertensive patients, but in no case below a systolic pressure of 90 mm Hg or a drop greater than 30 mm Hg below baseline. In view of their marginal treatment benefits, nitrates should not be used if hypotension limits the administration of beta-blockers, which have more powerful salutary effects.

6.3.1.3. Analgesia

Class I

Morphine sulfate (2 to 4 mg IV with increments of 2 to 8 mg IV repeated at 5- to 15-minute intervals) is the analgesic of choice for management of pain associated with STEMI. (Level of Evidence: C)

Pain relief is an important element in the early management of the patient with STEMI. There is a tendency to underdose patients with STEMI because of the desire to assess the response to anti-ischemic or reperfusion therapy. This should be avoided, because patients with STEMI have a hyperadrenergic state particularly early after the onset of coronary occlusion. Conversely, it should not be assumed that resolution of discomfort after administration of analgesics indicates reperfusion has occurred (see Section 6.3.1.6.3.7 for further discussion). Pain, which is commonly severe in the acute phase of the event, contributes to increased sympathetic activity.

Pain management should be directed toward acute relief of symptoms of ongoing myocardial ischemia and necrosis and toward general relief of anxiety and apprehension, the latter of which can heighten pain perception. Surges of catecholamines have been implicated as having a role in plaque fissuring and thrombus propagation and in reducing the threshold for ventricular fibrillation (259). Because the pain of STEMI is related to ongoing ischemia, interventions that affect the oxygen supply-demand relationship (i.e., by either increasing supply or decreasing demand) may lessen the pain of STEMI (260).

Control of cardiac pain is typically accomplished with a combination of nitrates, opiate analgesic agents, oxygen, and beta-adrenergic blockers. Treatment with these agents extends from the ED to the critical care unit. An important consideration when using intravenous nitrates is not to lower blood pressure to a level that would preclude adequate dosage of morphine sulfate for pain control. Morphine sulfate remains the analgesic agent of choice for management of pain associated with STEMI, except in documented cases of morphine sensitivity. The dose required for adequate pain relief varies in relation to age and body size, as well as blood pressure and heart rate. Anxiety reduction secondary to morphine administration reduces the patient's restlessness and the activity of the autonomic nervous system, with a consequent reduction of the heart's metabolic demands. Morphine administration for patients with pulmonary edema is clearly beneficial and may promote peripheral arterial and venous dilation, reducing the work of breathing and slowing the heart rate secondary to combined withdrawal of sympathetic tone and augmentation of vagal tone (259,260).

Side effects of morphine administration such as hypotension can be minimized by keeping the patient supine and elevating the lower extremities if systolic pressure goes below 100 mm Hg systolic, assuming pulmonary edema is not present. The concomitant use of atropine in 0.5- to 1.5-mg doses intravenously may be helpful in reducing the excessive vagomimetic effects of morphine if significant bradycardia or hypotension occurs. Although respiratory depression is relatively uncommon, patients' respirations should be monitored, particularly as their cardiovascular status improves. The narcotic reversing agent naloxone, 0.1 to 0.2 mg intravenously, can be given initially if indicated and repeated after 15 minutes if necessary. Nausea and vomiting as poten-
tial side effects of large doses of morphine may be treated with a phenothiazine (260).

See "Hospital Management" (Section 7.2.4) for additional discussion of analgesia.

6.3.1.4. Aspirin

Class I

Aspirin should be chewed by patients who have not taken aspirin before presentation with STEMI. The initial dose should be: 162 mg (*Level of Evidence: A*) to 325 mg (*Level of Evidence: C*). Although some trials have used enteric-coated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated aspirin formulations.

At a dose of 162 mg or more, aspirin produces a rapid clinical antithrombotic effect caused by immediate and near-total inhibition of thromboxane A₂ production. The Second International Study of Infarct Survival (ISIS-2) has shown conclusively the efficacy of aspirin alone for treatment of evolving acute MI, with an absolute risk difference in 35-day mortality of 2.4% (relative risk reduction [RRR] 23%) (261). When aspirin was combined with streptokinase, the absolute risk difference in mortality was 5.2% (RRR 42%). A metaanalysis demonstrated that aspirin reduced coronary reocclusion and recurrent ischemic events after fibrinolytic therapy with either streptokinase or alteplase (262). Accordingly, aspirin now forms part of the early management of all patients with suspected STEMI and should be given promptly, certainly within the first 24 hours, at a dose between 162 and 325 mg and continued indefinitely at a daily dose of 75 to 162 mg (263). Although some trials have used entericcoated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated formulations (264).

Unlike fibrinolytic agents, there is little evidence for a time-dependent effect of aspirin on early mortality. However, data do support the contention that a chewable aspirin is absorbed more quickly than one swallowed in the early hours after infarction, particularly after opiate therapy. The use of aspirin is contraindicated in those with a hypersensitivity to salicylate. Aspirin suppositories (300 mg) can be used safely and are the recommended route of administration for patients with severe nausea and vomiting or known upper-gastrointestinal disorders. In patients with true aspirin allergy (hives, nasal polyps, bronchospasm, or anaphylaxis), clopidogrel or ticlopidine may be substituted.

6.3.1.5. Beta-Blockers

Class I

Oral beta-blocker therapy should be administered promptly to those patients without a contraindication, irrespective of concomitant fibrinolytic therapy or performance of primary PCI. (Level of Evidence: A)

Class IIa

It is reasonable to administer IV beta-blockers promptly to STEMI patients without contraindica-

tions, especially if a tachyarrhythmia or hypertension is present. (*Level of Evidence: B*)

During the first few hours after the onset of STEMI, betablocking agents may diminish myocardial oxygen demand by reducing heart rate, systemic arterial pressure, and myocardial contractility. In addition, prolongation of diastole caused by a reduction in heart rate may augment perfusion to ischemic myocardium, particularly the subendocardium. As a result, immediate beta-blocker therapy appears to reduce 1) the magnitude of infarction and incidence of associated complications in subjects not receiving concomitant fibrinolytic therapy, 2) the rate of reinfarction in patients receiving fibrinolytic therapy, and 3) the frequency of life-threatening ventricular tachyarrhythmias.

In patients not receiving fibrinolytic therapy, intravenously administered beta-blocking agents exert a modestly favorable influence on infarct size (265). Large early trials suggested a mortality benefit as well. In ISIS-1 (266), more than 16000 patients with suspected acute MI were enrolled within 12 hours of onset of symptoms; immediate atenolol, 5 to 10 mg IV, followed by oral atenolol, 100 mg daily, reduced 7-day mortality from 4.3% to 3.7% (p less than 0.02; 6 lives saved per 1000 treated). The mortality difference between those receiving and not receiving atenolol was evident by the end of day 1 and was sustained subsequently. In the Metoprolol In Acute Myocardial Infarction (MIAMI) trial (267), more than 5700 subjects with evolving MI were randomly assigned to receive placebo or metoprolol, up to 15 mg IV in 3 divided doses followed by 50 mg orally every 6 hours for 48 hours and then 100 mg twice per day thereafter. Fifteen-day mortality was reduced with metoprolol from 4.9% to 4.3%. As in ISIS-1, the mortality difference between those given placebo and those receiving metoprolol was evident by the end of day 1, after which it was sustained.

In subjects receiving concomitant fibrinolytic therapy, intravenously administered beta-blocking drugs diminish the incidence of subsequent nonfatal reinfarction and recurrent ischemia. In addition, they may reduce mortality if given particularly early (i.e., within 2 hours) after onset of symptoms. In the Thrombolysis In Myocardial Infarction Phase II (TIMI-II) trial (268), in which all patients received IV alteplase, those randomly assigned to receive metoprolol, 15 mg IV, followed by oral metoprolol, 50 mg twice per day for 1 day and then 100 mg twice per day thereafter, had a diminished incidence of subsequent nonfatal reinfarction and recurrent ischemia compared with those begun on oral metoprolol 6 days after the acute event. Among those treated especially early (i.e., within 2 hours of symptom onset), the composite end point, death or reinfarction, occurred less often in those given immediate IV metoprolol than in those who did not receive it.

The benefits of routine early IV use of beta-blockers in the fibrinolytic era have been challenged by 2 later randomized trials of IV beta-blockade (269,270) and by a post hoc analysis of the use of atenolol in the GUSTO-I trial (271). A subsequent systematic review of early beta-blocker therapy in

STEMI found no significant reduction in mortality (67). Therefore, data on the early use of intravenous beta-blockade in STEMI are inconclusive, and patterns of use vary.

Beta-blockers should not be administered to patients with STEMI precipitated by cocaine use because of the risk of exacerbating coronary spasm (272). If IV beta-blockade induces an untoward effect, such as atrioventricular (AV) block, excessive bradycardia, or hypotension, the condition is quickly reversed by infusion of a beta-adrenergic agonist (i.e., isoproterenol 1 to 5 mcg/min). The presence of moderate LV failure early in the course of STEMI should preclude the use of early IV beta-blockade until the heart failure has been compensated but is a strong indication for the oral use of beta-blockade before discharge from the hospital.

The following are relative contraindications to beta-blocker therapy: heart rate less than 60 bpm, systolic arterial pressure less than 100 mm Hg, moderate or severe LV failure, signs of peripheral hypoperfusion, shock, PR interval greater than 0.24 second, second- or third-degree AV block, active asthma, or reactive airway disease.

Randomized trials of beta-blocker therapy in patients with STEMI undergoing PCI without fibrinolytic therapy have not been performed. However, it seems reasonable pending further information to extrapolate data from those receiving another form of revascularization, fibrinolytic therapy, to the PCI population. The more contemporary CAPRICORN (Carvedilol Post-infarct Survival Controlled Evaluation) trial (273), which includes patients undergoing either form of revascularization, confirms the benefits of beta-blocker therapy in patients with transient or sustained postinfarction LV dysfunction.

6.3.1.6. Reperfusion

6.3.1.6.1. GENERAL CONCEPTS

Class I

All STEMI patients should undergo rapid evaluation for reperfusion therapy and have a reperfusion strategy implemented promptly after contact with the medical system. (Level of Evidence: A)

Although rapid spontaneous reperfusion of the infarct artery may occur, in the majority of patients there is persistent occlusion of the infarct artery in the first 6 to 12 hours while the affected myocardial zone is undergoing necrosis. Prompt and complete restoration of flow in the infarct artery can be achieved by pharmacological means (fibrinolysis), PCI (balloon angioplasty with or without deployment of an intracoronary stent under the support of pharmacological measures to prevent thrombosis), or surgical measures (Figure 3) (24-40). Despite the extensive improvement in intraoperative preservation with cardioplegia and hypothermia and in numerous surgical techniques, it is not logistically possible to provide surgical reperfusion in a timely fashion, and therefore patients with STEMI who are candidates for reperfusion routinely receive either fibrinolysis or a catheter-based treatment.

Evidence exists that expeditious restoration of flow in the obstructed infarct artery after the onset of symptoms in patients with STEMI is a key determinant of short- and longterm outcomes regardless of whether reperfusion is accomplished by fibrinolysis or PCI (24,274,275). As discussed previously (Section 4.1), efforts should be made to shorten the time from recognition of symptoms by the patient to contact with the medical system. All healthcare providers caring for patients with STEMI from the point of entry into the medical system must recognize the need for rapid triage and implementation of care in a fashion analogous to the handling of trauma patients. When considering recommendations for timely reperfusion of patients with STEMI, the Writing Committee reviewed data from clinical trials, focusing particular attention on enrollment criteria for selection of patients for randomization, actual times reported in the trial report rather than simply the allowable window specified in the trial protocol, treatment effect of the reperfusion strategy on individual components of a composite primary end point (e.g., mortality, recurrent nonfatal infarction), ancillary therapies (e.g., antithrombin and antiplatelet agents), and the interface between fibrinolysis and referral for angiography and revascularization. When available, data from registries were also reviewed to assess the generalizability of observations from clinical trials of reperfusion to routine practice. Despite the wealth of reports on reperfusion for STEMI, it is not possible to produce a simple algorithm given the heterogeneity of patient profiles and availability of resources in various clinical settings at various times of day. This section introduces the recommendations for an aggressive attempt to minimize the time from entry into the medical system to implementation of a reperfusion strategy using the concept of medical system goals. More detailed discussion of these goals and the issues to be considered in selecting the type of reperfusion therapy may be found in Section 6.3.1.6.2, followed by a discussion of available resources in Section 6.3.1.6.2.1.

The medical system goal is to facilitate rapid recognition and treatment of patients with STEMI such that door-to-needle (or medical contact-to-needle) time for initiation of fibrinolytic therapy can be achieved within 30 minutes or that door-to-balloon (or medical contact-to-balloon) time for PCI can be kept under 90 minutes. These goals may not be relevant for patients with an appropriate reason for delay, such as uncertainty about the diagnosis (particularly for the use of fibrinolytic therapy), need for evaluation and treatment of other life-threatening conditions (e.g., respiratory failure), or delays associated with the patient's informed choice to have more time to consider the decision. In the absence of such types of circumstances, the emphasis is on having a system in place such that when a patient with STEMI presents for medical care, reperfusion therapy can be provided as soon as possible within these time periods. Because there is not considered to be a threshold effect for the benefit of shorter times to reperfusion, these goals should not be understood as "ideal" times but rather the longest times that should be considered acceptable. Systems that are able to achieve even

more rapid times for patients should be encouraged. Also, this goal should not be perceived as an average performance standard but as a goal that an early treatment system in every hospital should seek for every appropriate patient.

A critically important goal of reperfusion is to restore flow in the infarct artery as quickly and as completely as possible, but the ultimate goal of reperfusion in STEMI is to improve myocardial perfusion in the infarct zone. Despite adequate restoration of flow in the epicardial infarct artery, perfusion of the infarct zone may still be compromised by a combination of microvascular damage and reperfusion injury (276-278). Microvascular damage occurs as a consequence of downstream embolization of platelet microemboli and thrombi followed by the release of substances from activated platelets that promote occlusion or spasm in the microvasculature. Reperfusion injury results in cellular edema, free radical formation, calcium overload, and acceleration of the apoptotic process. Cytokine activation in the infarct zone leads to neutrophil accumulation and inflammatory mediators that contribute to tissue injury.

Thus, construction of an ideal reperfusion regimen in patients with STEMI not only should focus on the primary means of restoring flow in the epicardial infarct artery (pharmacological or catheter-based) but should also include adjunctive and ancillary treatments that minimize the amount of microvascular damage and protect the jeopardized myocardial infarct zone that contains cells in various stages of ischemia, necrosis, and apoptosis (279,280). The Writing Committee endorses further research to identify the optimum strategies for achieving these goals.

6.3.1.6.2. SELECTION OF REPERFUSION STRATEGY. The literature provides very strong evidence that among patients with suspected STEMI and without contraindications, the prompt use of reperfusion therapy is associated with improved survival (156). Despite such strong evidence, studies continue to indicate that reperfusion therapy is underutilized and often not administered soon after presentation (281-283). Indecision about the choice of reperfusion therapy should not deter physicians from using these strategies or delay them in administering therapy.

There is controversy about which form of reperfusion therapy is superior in various clinical settings. Part of the uncertainty derives from the continual introduction of new agents, devices, and strategies, which quickly make previous studies less relevant to contemporary practice. With pharmacological reperfusion therapies, there are new agents, dosing regimens, adjunctive treatments, and combined strategies with procedures that are in a continual process of refinement and evaluation. Similarly, with catheter-based approaches, there are new devices, adjunctive therapies, technologies, and combined strategies with medications that are being introduced and evaluated. As a result, the evidence base regarding the best approach to reperfusion therapy is quite dynamic. Several issues should be considered in selecting the type of reperfusion therapy, as discussed below.

Time From Onset of Symptoms. Time from onset of symptoms to fibrinolytic therapy is an important predictor of MI size and patient outcome (284). The efficacy of fibrinolytic agents in lysing thrombus diminishes with the passage of time (279). Fibrinolytic therapy administered within the first 2 hours (especially the first hour) can occasionally abort MI and dramatically reduces mortality (Figure 13) (156,159). The National Heart Attack Alert Working Group (179) recommends that EDs strive to achieve a 30-minute door-to-needle time to minimize treatment delays. Prehospital fibrinolysis reduces treatment delays by up to 1 hour and reduces mortality by 17% (285).

The amount of myocardium at risk, presence of collateral blood flow, and duration of coronary occlusion are major determinants of myocardial infarct size (286-289). In animal models (18), occlusions persisting greater than 30 minutes produce myonecrosis. Reperfusion at 90 minutes salvages approximately half of the myocardium at risk. Myocardial salvage is minimal after 4 to 6 hours of ischemia unless ischemic preconditioning and/or collateral flow have modified the wave front of necrosis.

A time-dependent decrease in efficacy of fibrinolytic therapy may also contribute to the higher mortality rate in patients with longer symptom duration (279). In contrast, the ability to produce a patent infarct artery is much less dependent on symptom duration in patients undergoing PCI. Several reports claim no influence of time delay on mortality rates when PCI is performed after 2 to 3 hours of symptom duration (290,291). One study suggests that time to PCI is only important for patients presenting with shock (292). Another showed that time was associated with outcome in higher-risk but not lower-risk patients (293). Conversely, others have reported increasing mortality rates with increasing door-to-balloon times (294,295). Importantly, after adjustment for baseline characteristics, time from symptom onset to balloon inflation is significantly correlated with 1year mortality in patients undergoing primary PCI for STEMI (relative risk [RR] equals 1.08 for each 30-minute delay from symptom onset to balloon inflation, p equals 0.04) (275,275a). Interestingly, although the CAPTIM (173) and PRAGUE-2 (177) studies reached different conclusions about the overall superiority of PCI over fibrinolysis, important observations were made in the subset of patients presenting very early after the onset of symptoms. In the subset of patients presenting within 3 hours of the onset of symptoms in PRAGUE-2, mortality was equivalent in those treated with streptokinase and those transferred with PCI (177). Patients treated within 2 hours of symptom onset in CAP-TIM had improved outcomes with prehospital tissue plasminogen activator (tPA) versus transfer for PCI (176). (See Section 6.3.1.6.2.1.)

It is also possible that time-to-treatment analyses have been confounded by other variables (293,296). First, higher-

Presentation features	Perce Fibrii	ent of patien nolytic	ts dead Cont	d rol	Stratifie O-E	ed statistics Variance	Odds ratio Fibrinolytic better	o & Cls Control	Chi-square test of In different patien Heterogeneity	odds ratios ts categories: Trend
ECG							, sector	, better	notorogeneity	mond
BBB	18 7	%	23.6	%	-24 5	83.3	-			
ST elev anterior	13.2	%	16.9	%	-122	420.6		7		
ST elev inferior	7.5	%	84	%	-27.1	237.4	-			
ST elev other	10.6	%	13.4	%	-42 1	159.6		+-	21.26 on 6 df	
ST depression	15.2	%	13.8	%	12.9	108.7			(P<0.1)	
Other abnormality	5.2	%	5.8	%	-9.6	103.2	÷		(1 -0.1)	
Normal	3.0	%	23	0/6	34	12.9				
Normal	0.0	70	2.0	70	0.4	12.0				
Hours from onset										
0 - 1	9.5	%	13.0	%	-29.3	83.3	:			
2 - 3	8.2	%	10.7	%	-100.2	354.8				
4 - 6	9.7	%	11.5	%	-78.5	387.6			9.69 on 4 df	9.55 on 1 df
7 - 12	11 1	%	12.7	%	-51.5	336.7	_		(P < 0.05)	(2P=0.002)
13 - 24	10.0	%	10.5	%	-11 1	212.6			(1 0.00)	(21 0.002)
	10.0	<i>,</i> ,,	10.0			212.0				
Age (years)										
<55	3.4	%	4.6	%	-45.9	155.6	1			
55-64	7.2	%	8.9	%	-56.3	360.0			8 27 on 3 df	6 58 on 1 df
65-74	13.5	%	16 1	%	-113 7	533.0			(P < 0.05)	(2P=0.01)
75+	24.3	%	25.3	%	-12.6	266.6			(1 0.00)	(21 0.01)
101	24.0	<i>,</i> ,,	20.0	70	12.0	200.0				
Gender										
Male	8.2	%	10.1	%	-208.1	926.0			1.99 on 1 df	
Female	14.1	%	16.0	%	-62.2	436.8			(NS)	
1 officio		,,,	10.0	,,,	02.2	10010		-	(110)	
Systolic BP (mm Ha)							i i	1		
<100	28.9	%	35.1	%	-38.7	132.2				
100-149	9.6	%	11.5	%	-168.9	850.0			1.31 on 3 df	0.68 on 1 df
150-174	7.5	%	8.7	%	-58.2	290.0			(NS)	(NS)
175+	7.2	%	8.2	%	-10.8	74.1		·	()	()
Heart rate										
<80	7.2	%	8.5	%	-83.2	454.0		-		
80-99	9.2	%	11.3	%	-55.8	267.2			0.51 on 2 df	0.31 on 1 df
100+	17.4	%	20.7	%	-51.7	238.6		_	(NS)	(NS)
								-		
Prior MI										
Yes	12.5	%	14.1	%	-43.7	322.4		-1	2.09 on 1 df	
No	8.9	%	10.9	%	-228.5	1001.9	-		(NS)	
Diabetes										
Yes	13.6	%	17.3	%	-41.4	145.7			1.57 on 1 df	
No	8.7	%	10.2	%	-142.6	830.4			(NS)	
							· · · · · · · · · · · · · · · · · · ·			
ALL PATIENTS	2820	29315	3357	/29285		-269.5	1377.4 🛒		18% SD 2 odd	s reduction
	9.6	%	11.5	%					2P < 0.00001	
							, ,			
							0.5	1.0	1.5	

Figure 13. Mortality differences during days 0 through 35 subdivided by presentation features in a collaborative overview of results from nine trials of fibrinolytic therapy. At center absolute mortality rates are shown for fibrinolytic and control groups for each clinical feature at presentation listed at left. The odds ratio of death in fibrinolytic group to that in control group is shown for each subdivision (black square) along with 95% confidence interval (horizontal line). The summary odds ratio at bottom corresponds to an 18% proportional reduction in 35-day mortality and is highly statistically significant. This translates to a reduction of 18 deaths per 1000 patients treated with fibrinolytic agents. O-E indicates observed versus expected ratio; Cls = confidence intervals; ECG = electrocardiogram; BBB = bundle-branch block; ST elev = ST-segment elevation; df = degrees of freedom; BP = blood pressure; MI = myocardial infarction; SD = standard deviation. Reprinted with permission from Elsevier (Fibrinolytic Therapy Trialists' (FTT) Collaborative Group. The Lancet 1994; 343:311-22) (156).

risk patients report later to the hospital and may respond better to PCI than to fibrinolytic agents. Second, shorter doorto-balloon times may be a surrogate for better quality of care and adherence to treatment guidelines. The Task Force on the Management of Acute Myocardial Infarction of the European Society of Cardiology (297) and this Committee both recommend a target medical contact– or door-to-balloon time of less than 90 minutes.

Risk of STEMI. Several models have been developed that assist clinicians in estimating the risk of mortality in patients

with STEMI (240-242,298,299). Although these models vary somewhat in the factors loaded into the risk prediction tool and also vary with respect to statistical measures of their discriminative power (e.g., C statistic), all the models provide clinicians with a means to assess the continuum of risk from STEMI. None of the models have been tested prospectively by randomizing patients to a reperfusion strategy based on estimated mortality at presentation. Retrospective analyses do suggest that the absolute difference in mortality at 30 days between PCI and fibrinolysis increases in favor of PCI as the estimated risk of mortality with fibrinolysis increases (300). Conversely, as the estimated mortality benefit with fibrinolysis decreases, the absolute mortality benefit of PCI decreases, with equipoise appearing (i.e., similar 30-day mortality rates) when the estimated mortality with fibrinolysis is approximately 2% to 3% (300).

When the estimated mortality with fibrinolysis is extremely high, as is the case in patients with cardiogenic shock, compelling evidence exists that favors a PCI strategy. The SHOCK trial (SHould we emergently revascularize Occluded Coronaries for cardiogenic shock?) demonstrated that patients with cardiogenic shock have a better 1-year survival if they have undergone early coronary revascularization (184). At 1 year, patients in the early revascularization (184). At 1 year, patients in the early revascularization the group that had initial medical stabilization followed by no or late revascularization (184,301). Observational data from NRMI suggest superiority of PCI over fibrinolysis for patients with Killip class greater than or equal to II (302).

Risk of Bleeding. Choice of reperfusion therapy is also affected by the patient's risk of bleeding. When both types of reperfusion are available, the higher the patient's risk of bleeding with fibrinolytic therapy, the more strongly the decision should favor PCI. If PCI is unavailable, then the benefit of pharmacological reperfusion therapy should be balanced against the risk. A decision analysis suggested that fibrinolytic therapy should be favored against no reperfusion treatment until the risk of a life-threatening bleed exceeds 4% in older patients who have a risk profile similar to those in the classic randomized trials of fibrinolytic therapy allow for the calculation of this risk (246). Because they are derived from less restricted populations, the scores that are most generalizable are those derived from observational studies (246).

Time Required for Transport to Skilled PCI Laboratory. The availability of interventional cardiology facilities is a key determinant of whether PCI can be provided. For facilities that can offer PCI, the literature suggests that this approach is superior to pharmacological reperfusion (303). The trials comparing pharmacological and PCI strategies, however, were conducted before the advent of more recent pharmacological and PCI strategies. When a composite end point of death, nonfatal recurrent MI, or stroke is analyzed, much of the superiority of a PCI strategy is driven by a reduction in the rate of nonfatal recurrent MI (Figure 14) (40). The rate of nonfatal recurrent MI can be influenced both by the adjunctive therapy used (Figure 3) (24-40) and by the proportion of patients who are referred for PCI when the initial attempt at fibrinolysis fails or myocardial ischemia recurs after initially successful pharmacological reperfusion (Figure 14) (155).

The experience and location of the PCI laboratory also plays a role in the choice of therapy. The trials were performed in centers with highly experienced teams, and their results may not be generalizable to all PCI laboratories throughout the country. Not all laboratories can provide prompt, high-quality primary PCI. Even centers with interventional cardiology facilities may not be able to provide the staffing required for 24-hour coverage of the catheterization laboratory. Despite staffing availability, the volume of cases in the laboratory may be insufficient for the team to acquire and maintain skills required for rapid PCI reperfusion strategies. A study from NRMI investigated the effect of volume on the outcomes of patients treated with PCI versus pharmacological reperfusion strategies (303). They studied 446 acute-care hospitals, with 112 classified as low-volume (fewer than or equal to 16 procedures), 223 as intermediatevolume (17 to 48 procedures), and 111 as high-volume (49 or more procedures) based on their annual primary angioplasty volume. They reported that patients hospitalized at intermediate- and high-volume centers had lower mortality with PCI reperfusion, whereas in the low-volume centers, there was no significant difference between the 2 reperfusion strategies. In another article from the NRMI investigators, the volume of primary PCI procedures, but not pharmacological treatment, was inversely associated with the mortality rate for patients with STEMI (304).

A decision must be made when a STEMI patient presents to a center without interventional cardiology facilities. Fibrinolytic therapy can generally be provided sooner than primary PCI (Figure 7) (180). As the time delay for performing PCI increases, the mortality benefit associated with expeditiously performed primary PCI over fibrinolysis decreases (305). Compared with a fibrin-specific lytic agent, a PCI strategy may not reduce mortality when a delay greater than 60 minutes is anticipated versus immediate administration of a lytic (Figure 15) (305).

The balance of risk/benefit between the transfer of patients for PCI and more immediate treatment with fibrinolytic therapy remains uncertain. The DANAMI-2 trial (DANish trial in Acute Myocardial Infarction), conducted in Denmark, found that patients treated at facilities without interventional cardiology capabilities had better composite outcomes with transfer for PCI within 2 hours of presentation than with pharmacological reperfusion treatment at the local hospital (306). Whether these results could be replicated elsewhere is not known. An alternative to transfer is for hospitals without on-site cardiac surgery to develop the capability to provide primary mechanical reperfusion therapy. A study by Aversano and colleagues with 11 hospitals in Massachusetts and Maryland suggested that this approach may improve outcomes (307). It can be expected, however, that only a limited number of hospitals could develop such a program, and it has yet to be determined whether a certain volume of cases would be necessary to maintain the effectiveness of the service. The economic implications of expansion of the number of PCI-capable centers that are able to maintain an inventory of the necessary catheters and other devices and provide 24hour coverage, 7 days per week, deserve further evaluation from the perspectives of individual institutions and the global healthcare delivery system. See additional discussion in Section 6.3.1.6.2.1.

Given the current literature, it is not possible to say definitively that a particular reperfusion approach is superior for all patients, in all clinical settings, at all times of day Long-Term Clinical Outcomes

Short-Term Clinical Outcomes

PTCA

35 30

PCI vs Fibrinolysis:

P less than 0.0001

Fibrinolysis

PTCA

35 7 30 -

PCI vs Fibrinolysis:



TNN

Figure 14. Percutaneous coronary intervention (PCI) versus fibrinolysis for ST-elevation myocardial infarction (STEMI). The short term (4-6 weeks) (top left) and long term (top right) outcomes for the various endpoints shown are plotted for patients with STEMI randomized to percutaneous coronary intervention (PCI) or fibrinolysis for reperfusion in 23 trials (N=7739). Based on the frequency of events for each endpoint in the two treatment groups the number needed to treat (NNT) or number needed to harm (NNH) are depending on whether PCI is compared with streptokinase or a fibrin-specific lytic. For example, when primary PCI is compared with alteplase (tPA) and the SHOCK trial is shown for the short term (bottom left) and long term (bottom right) outcomes. The magnitude of the treatment differences for death, non-fatal reinfarction, and stroke vary excluded, the mortality rate is 5.5% versus 6.7% (OR 0.81, 95% CI 0.64-1.03, P=0.081) (421a). See references (40,421a) for additional discussion. PTCA = percutaneous transluminal coronary angioplasty; ReMI = recurrent MI; Rec. Isch = recurrent ischemia; Hem. Stroke = hemorrhagic stroke; MI = myocardial infarction; and CVA = cerebrovascular accident. Modified with permission from Elsevier (Keeley et al. The Lancet 2003;361:13-20) (40).



Figure 15. PCI versus lysis with fibrin-specific agents: is timing (almost) everything? RCT = randomized controlled trial; N = Number of patients; PCI = percutaneous coronary intervention. Modified from Nallamothu and Bates. Am J Cardiol 2003;92:824-6 (305). Copyright 2003, with permission from Excerpta Medica, Inc.

Table 11. Assessment of Reperfusion Options for Patients with STEMI

STEP 1: Assess Time and Risk

- Time since onset of symptoms
- Risk of STEMI
- Risk of fibrinolysis
- Time required for transport to a skilled PCI lab

STEP 2: Determine if Fibrinolysis or an Invasive Strategy is Preferred If presentation is less than 3 hours and there is no delay to an invasive strategy, there is no preference for either strategy



An Invasive Strategy is generally preferred if (See Section 6.3.1.6.4.2):

- Skilled PCI lab available with surgical backup †; Medical Contact-to-Balloon or Door-to-Balloon is less than 90 minutes (Door-to-Balloon) – (Door-to-Needle) is less than 1 hour *
- High Risk from STEMI Cardiogenic shock Killip class is greater than or equal to 3
- Contraindications to fibrinolysis including increased risk of bleeding and ICH
- Late Presentation The symptom onset was greater than 3 hours ago
- Diagnosis of STEMI is in doubt

STEMI = ST-elevation myocardial infarction; PCI = percutaneous coronary intervention; ICH = intracranial hemorrhage. *Applies to fibrin-specific agents (See Figure 15).

†Operator experience greater than a total of 75 Primary PCI cases/year.

‡Team experience greater than a total of 36 Primary PCI cases/year.

\$This calculation implies that the estimated delay to the implementation of the invasive strategy is greater than one hour versus initiation of fibrinolytic therapy immediately with a fibrin-specific agent.

(173,176,177) (Danchin N; oral presentation, American Heart Association 2003 Annual Scientific Sessions, Orlando, FL, November 2003). The main point is that some type of reperfusion therapy should be selected for all appropriate patients with suspected STEMI. The appropriate and timely use of some reperfusion therapy is likely more important than the choice of therapy, given the current literature and the expanding array of options. Clinical circumstances in which fibrinolytic therapy is generally preferred or an invasive strategy is generally preferred are shown in Table 11.

6.3.1.6.2.1. Available Resources

Class I

STEMI patients presenting to a facility without the capability for expert, prompt intervention with primary PCI within 90 minutes of first medical contact should undergo fibrinolysis unless contraindicated. (Level of Evidence: A)

The preferred reperfusion therapy for STEMI must take into account the location of the patient, the response time and expertise of the paramedical/ambulance personnel, their relationship to the regional healthcare facility(s), and the availability, capability, and expertise of the medical personnel at the facility. The most recent NRMI data continue to support advantages of primary PCI versus fibrinolysis in high-PCI-volume hospitals but not among those institutions with low volume (fewer than or equal to 16 procedures per year) (303). Approximately 20% of US hospitals have cardiac catheterization laboratories; and less than that number have the capacity for primary PCI. The C-PORT study (Atlantic Cardiovascular Patient Outcomes Research Team), which randomized 451 fibrinolysis-eligible patients with STEMI treated in 11 community hospitals with diagnostic catheterization but not onsite PCI facilities, is of interest. Patients were randomized within 12 hours of symptom onset to an accelerated alteplase regimen (median door-to-lytic time was 46 minutes) versus primary PCI (median door-to-balloon time 101.5 minutes). At 6 months, the incidence of death, re-MI, and stroke was 12.4% for PCI and 19.9% for fibrinolytic therapy (p equals 0.03). Because only 18% of the intended sample size was actually enrolled, this study is significantly underpowered, and its conclusion can only be hypothesis-generating rather than definitive (307). Importantly, most C-PORT patients were randomized between 0800 and 1600 hours, an experience consistent with NRMI data. The NRMI report also demonstrated a substantially longer doorto-balloon time when patients with STEMI undergo direct PCI outside of daylight hours (308). The Zwolle group evaluated 1702 consecutive patients and found that the 47% of patients who presented outside "routine duty hours" (i.e., 1800 to 0800 hours) had a higher rate of both PCI failure and 30-day mortality than those within the 0800 to 1800 period (6.9% and 4.2% versus 3.8% and 1.9%, respectively; p less than 0.01) (309). The reasons for these differences are unclear but could relate to both patient and process-of-care

factors, including variations in both cognitive function and manual dexterity in sleep-deprived healthcare providers (308,310).

Two studies germane to STEMI care and resource utilization are noteworthy. The first, CAPTIM, a comparison of angioplasty and prehospital fibrinolysis (accelerated alteplase) in STEMI, fell short of its planned 1200-patient enrollment, and hence was underpowered (173). Eight hundred forty patients were randomized to prehospital fibrinolysis versus primary PCI. The primary end point was the composite of all-cause mortality, nonfatal recurrent MI, and nonfatal disabling stroke at 30 days, which occurred in 8.2% of patients assigned fibrinolytic therapy and 6.2% of patients assigned to PCI (p not significant [NS]). The components for death, reinfarction, and disabling stroke were 3.8%, 3.7%, and 1.0% for fibrinolytic therapy and 4.8%, 1.7%, and 0% for PCI. Unlike C-PORT, this trial liberally used rescue angioplasty (28%), which probably accounts for the relatively low reinfarction rate in the lytic-treated group. A subsequent analysis from CAPTIM of the 55% of patients treated within 2 hours of symptom onset revealed a mortality trend in favor of prehospital fibrinolysis versus primary PCI (2.2%) versus 5.7%, p equals 0.058), whereas those patients treated beyond 2 hours had a 5.9% versus 3.7% (p equals 0.47) 30day mortality rate, respectively (176). Interestingly, there was a significant reduction in the frequency of cardiogenic shock for patients treated within 2 hours with prehospital fibrinolysis (1.3% versus 5.3%, p equals 0.032), whereas the frequency of this event after 2 hours was similar (i.e., 3.9% versus 4.4%, respectively) (176).

The DANAMI-2 study, which compared primary PCI versus accelerated alteplase, enrolled 1572 patients versus the 2000 patients planned (306). Patients were eligible if they had a sum of greater than 0.4 mV of ST elevation in 2 contiguous leads on their presenting ECG within 12 hours of symptom onset; however, patient enrollment consisted of 37% of those screened, and patients deemed to be at high risk during ambulance transport were excluded (306). Twentynine hospitals, of which 5 conducted primary PCI and were located a mean of 35 miles from referring hospitals (maximum 95 miles), participated. The median door-to-needle time for patients randomized to fibrinolysis was approximately 50 minutes for patients presenting either to a community (referral) hospital or an invasive center. For patients who presented to a community hospital (where 1129 patients were enrolled), the time from initial presentation to balloon inflation at an invasive center was 108 minutes; the door-toballoon time was 93 minutes for patients presenting to an invasive center and randomized to PCI. The primary composite end point of death, reinfarction, and stroke through 30 days occurred in 14.2% of the fibrinolysis-treated patients and 8.5% of the PCI-treated patients (p less than 0.001). The individual end-point components of death, reinfarction, and stroke occurred in 7.8%, 6.3%, and 2.0% of the fibrinolysistreated patients and 6.6%, 1.6%, and 1.1% of the PCI-treated patients, respectively. In addition to the exclusion of highrisk patients for transport, some caveats in DANAMI-2 are



Figure 16. Effect of fibrinolytic therapy on mortality according to admission electrocardiogram. Patients with bundle-branch block (BBB) and anterior ST-segment elevation (ANT ST Elevation) derive the most benefit from fibrinolytic therapy. Effects in patients with inferior ST-segment elevation (INF ST Elevation) are much less, while patients with ST-segment depression (ST DEP) do not benefit. Reprinted with permission from Elsevier (Fibrinolytic Therapy Trialists' Collaborative Group. The Lancet 1994;343:311-22) (156).

noteworthy: 1) The antithrombotic dosing regimen was in excess of ACC/AHA guidelines. 2) The protocol specified that repeat fibrinolysis was to be used for failed reperfusion, reinfarction, and recurrent ST-elevation ischemia; this strategy was used in 26 patients within 12 hours after randomization, and only 1.9% underwent rescue PCI. 3) Patients with prior stroke were included, and an imbalance in this baseline characteristic was present, (i.e., 4.0% for fibrinolysis versus 2.7% for PCI ([1-sided p equals 0.06]). 4) The difference in reinfarction rates between the 2 groups was likely exaggerated by the exclusion of those patients associated with invasive procedures (311).

Hence, on the basis of the data, patients with STEMI presenting to a facility without the capability for expert, prompt intervention with primary PCI within 90 minutes of first medical contact should undergo fibrinolysis unless contraindicated. (See Sections 6.3.1.6.4.2, Primary PCI, and 6.3.1.6.4.2.4, Interhospital Transfer for Primary PCI.)

6.3.1.6.3. PHARMACOLOGICAL REPERFUSION. *RATION-ALE FOR FIBRINOLYTIC THERAPY*. Although the clinical features of coronary obstruction were described nearly a century ago (312,313), thrombotic obstruction of the infarct artery as a cause of STEMI was not proven until 1980 (20). The benefits of fibrinolytic therapy are maximal when there is prompt, adequate restoration of flow in the epicardial infarct artery and perfusion of the myocardium in the infarct zone. Controlled clinical trials have demonstrated the potential for functional, clinical, and mortality benefits only if fibrinolytic therapy is given within 12 hours. (See additional discussion on the use of antithrombins and antiplatelet agents as ancillary therapy in Sections 6.3.1.6.8.1 and 6.3.1.6.8.2.)

The reduction in mortality with fibrinolytic therapy is present regardless of sex, presence of diabetes, blood pressure (if less than 180 mm Hg systolic) (246,314), heart rate, or history of previous MI (156). The mortality benefit is greater in the setting of anterior STEMI, diabetes, low blood pressure, (less than 100 mm Hg systolic) or high heart rate (greater than 100 bpm) (Figure 13) (156). The earlier therapy begins, the better the outcome, with the greatest benefit decidedly occurring when therapy is given within the first 3 hours. Benefit occurs, however, up to at least 12 hours from the onset of symptoms. The absolute benefit is less with inferior STEMI, except for the subgroup with associated RV infarction or anterior ST-segment depression indicative of a greater territory at risk (Figure 16) (156).

6.3.1.6.3.1. Indications for Fibrinolytic Therapy

Class I

1. In the absence of contraindications, fibrinolytic therapy should be administered to STEMI patients with symptom onset within the prior 12 hours and ST elevation greater than 0.1 mV in at least 2 contiguous
 Table 12. Contraindications and Cautions for Fibrinolysis in ST

 Elevation Myocardial Infarction*

Absolute contraindications

Any prior ICH

Known structural cerebral vascular lesion (e.g., arteriovenous malformation)

Known malignant intracranial neoplasm (primary or metastatic) Ischemic stroke within 3 months EXCEPT acute ischemic stroke within 3 hours

Suspected aortic dissection

Active bleeding or bleeding diathesis (excluding menses) Significant closed-head or facial trauma within 3 months

Relative contraindications

History of chronic, severe, poorly controlled hypertension Severe uncontrolled hypertension on presentation (SBP greater than 180 mmHg or DBP greater than 110 mmHg)⁺

- History of prior ischemic stroke greater than 3 months, dementia, or known intracranial pathology not covered in contraindications
- Traumatic or prolonged (greater than 10 minutes) CPR or major surgery (less than 3 weeks)

Recent (within 2-4 weeks) internal bleeding

Noncompressible vascular punctures

For streptokinase/anistreplase: prior exposure (more than 5 days ago) or prior allergic reaction to these agents

Pregnancy

Active peptic ulcer

Current use of anticoagulants: the higher the INR, the higher the risk of bleeding

ICH = intracranial hemorrhage; SBP = systolic blood pressure; DBP = diastolic blood pressure; CPR = cardiopulmonary resuscitation; INR = international normalized ratio; MI = myocardial infarction.

*Viewed as advisory for clinical decision making and may not be all-inclusive or definitive.

 \dagger Could be an absolute contraindication in low-risk patients with MI (see Section 6.3.1.6.3.2).

precordial leads or at least 2 adjacent limb leads. (Level of Evidence: A)

2. In the absence of contraindications, fibrinolytic therapy should be administered to STEMI patients with symptom onset within the prior 12 hours and new or presumably new LBBB. (Level of Evidence: A)

Class IIa

- 1. In the absence of contraindications, it is reasonable to administer fibrinolytic therapy to STEMI patients with symptom onset within the prior 12 hours and 12lead ECG findings consistent with a true posterior MI. (Level of Evidence: C)
- 2. In the absence of contraindications, it is reasonable to administer fibrinolytic therapy to patients with symptoms of STEMI beginning within the prior 12 to 24 hours who have continuing ischemic symptoms and ST elevation greater than 0.1 mV in at least 2 contiguous precordial leads or at least 2 adjacent limb leads. (Level of Evidence: B)

Class III

- 1. Fibrinolytic therapy should not be administered to asymptomatic patients whose initial symptoms of STEMI began more than 24 hours earlier. (Level of Evidence: C)
- 2. Fibrinolytic therapy should not be administered to patients whose 12-lead ECG shows only ST-segment depression except if a true posterior MI is suspected. (Level of Evidence: A)

Because the benefit of fibrinolytic therapy is directly related to the time from symptom onset, treatment benefit is maximized by the earliest possible application of therapy. The constellation of clinical features that must be present (although not necessarily at the same time) to serve as an indication for fibrinolysis includes symptoms of myocardial ischemia and ST elevation greater than 0.1 mV, in 2 contiguous leads, or new or presumably new LBBB on the presenting ECG (156,315). In the very early phase of STEMI, giant hyperacute **T** waves may precede ST elevation (316). True posterior MI may be manifested by tall R waves in the right precordial leads and ST-segment depression in leads V_1 through V_4 , especially when T waves are upright (317). Repeat ECGs and incorporation of additional leads such as V_7 through V_9 are more specific for the detection of posterior infarction (225). Patients with LBBB or anterior ST elevation are at greater inherent risk from MI and achieve greater benefit with fibrinolytic therapy. Additional valuable information may be garnered from concurrent echocardiography to identify the location and extent of regional wallmotion abnormalities. Patients with inferior MI and ST elevation in V₁, V₄R, or both are more likely to have concomitant RV infarction (318). Attainment of additional ECG leads (right sided and/or posterior) or an echocardiogram may help clarify the location and extent of infarction and anticipated risk of complications, but it is important that acquisition of such ancillary information not interfere with the strategy of providing timely reperfusion in patients with STEMI (319).

6.3.1.6.3.2. Contraindications/Cautions

Class I

- 1. Healthcare providers should ascertain whether the patient has neurological contraindications to fibrinolytic therapy, including: any history of intracranial hemorrhage or significant closed head or facial trauma within the past 3 months, uncontrolled hypertension, or ischemic stroke within the past 3 months. (See Table 12 for a comprehensive list.) (Level of Evidence: A)
- 2. STEMI patients at substantial (greater than or equal to 4%) risk of ICH should be treated with PCI rather than with fibrinolytic therapy. (See Table 11 for further management considerations.) (Level of Evidence: A)

A detailed list of contraindications and cautions for the use of fibrinolytic therapy is shown in Table 12. Specific neurological considerations are addressed below.

Hemorrhage represents the most important risk of fibrinolytic therapy, especially ICH, which may be fatal in half to two thirds of patients. There is both legitimate concern and confusion surrounding the issue of whether fibrinolytic therapy should be contraindicated in patients with a history of prior cerebrovascular disease (2,320-322). The 1996 ACC/AHA Guidelines for the Management of Acute Myocardial Infarction (2) stated that "Previous hemorrhagic stroke at any time; other strokes or cerebrovascular events within 1 year" was a contraindication to use of thrombolytic therapy and that "...history of prior cerebrovascular accident or known intracerebral pathology not covered in contraindications" was a caution/relative contraindication. After the first 627 patients were enrolled in TIMI-II (320), a number of protocol changes were made: the dose of tPA was reduced from 150 to 100 mg, use of 80 mg of aspirin was postponed for 24 hours, patients who had a history of stroke or intermittent cerebral ischemic attacks were excluded, and patients with blood pressures greater than or equal to 180 mm Hg systolic or greater than or equal to 110 mm Hg diastolic were excluded. The reduction of ICH frequency by the exclusion of patients with any history of cerebrovascular disease was likely confounded by the influence of the other 3 protocol changes. The basis for the 1996 recommendation for a time frame of 1 year for ischemic stroke as a contraindication to coronary fibrinolysis was a consensus opinion without specific supporting data.

In subsequent trials (26-31,33,323,324), the use of prior TIA or stroke as an exclusion criterion has varied: stroke within 2 years (30), stroke within 6 months (323), TIA within 6 months/any history of stroke (33), any stroke (26,27,324), and any history of prior TIA or stroke (28,29,31) have each been used as exclusion criteria. In some studies, the frequency of ICH in patients older than 75 years ranged from 0.5% with streptokinase and heparin (33) to 2.5% with reteplase (26). Giugliano et al. (325) showed that the higher ICH frequencies with lanoteplase or tenecteplase may be explained in part by the effect of the UFH (InTIME-2 [Intravenous NPA for the Treatment of Infarcting Myocardium Early] equals 1.12%, InTIME-IIb equals 0.50%) and dose of the UFH infusion (ASSENT-1 and TIMI-10B: higher heparin dose equals 1.83%, lower heparin dose equals 0.74%). (See discussion on the use of LMWH in the elderly in Section 6.3.1.6.8.1.2.) In the Maximal Individual Therapy in Acute Myocardial Infarction (MITRA) registry (326), previous stroke within 3 months was the strongest predictor of stroke (OR equals 9.3, 95% CI 6.0 to 14.2) after STEMI. On the basis of these data, it appears that the effect of prior stroke/TIA per se on the frequency of ICH after fibrinolysis may be influenced by a number of factors. However, in patients with STEMI with prior ischemic stroke and other ICH risk factors who have a substantial risk for ICH, another reperfusion strategy should be pursued. Additional contraindications to fibrinolytic therapy include a recent history of significant closed head or facial trauma (327).

Estimation of risk of ICH. Several models have been developed for estimating the risk of ICH after fibrinolysis (246,328-330). These models incorporate baseline demographic features of the patient and also illustrate the impact of certain therapeutic decisions (e.g., selection of streptokinase versus tPA; dose of tPA used) (Table 13) (29,246,329,330). Streptokinase without heparin is the regimen associated with the lowest ICH rates (Figure 17) (29,246,329,330).

6.3.1.6.3.3. Effect on Mortality

Efficacy of intravenous fibrinolytic therapy in STEMI. It has now been well established that fibrinolytic therapy provides a survival benefit for patients with STEMI, based on large, well-controlled clinical trials (157,261,331,332). The mechanisms of benefit, which may have different time dependencies, include salvage of myocardium with reduced infarct size, favorable effect on infarct healing and myocardial remodeling, and reduced electrical heterogeneity and potential for life-threatening ventricular arrhythmia (333). An overview from 9 trials of fibrinolytic therapy (versus control) for STEMI has shown a highly significant 18% relative reduction in 35-day mortality (9.6% fibrinolysis versus 11.5% control), which corresponds to a reduction of 18 deaths per 1000 patients treated when data from all patient groups are pooled (156). In patients with ST elevation, a relative mortality reduction of 21% occurred. This survival benefit is maintained over the long term (up to 10 years) (334,335). Mortality reduction from fibrinolytic therapy is greatest within the first hour after symptom onset; thereafter, a decline in benefit of approximately 1.6 lives per 1000 patients treated is seen per 1-hour delay. Additionally, patients with presumed new LBBB, anterior infarction, and the greatest area of risk, as exemplified by the number of ECG leads affected and the extent of ST deviation, derived maximal benefit from fibrinolytic therapy (Figure 16) (156, 336).

Elderly patients. Although the elderly constitute a minority of the general population, they are the fastest-growing segment of the population and account for the majority of patients presenting with MI and a disproportionately high component of death from MI (46,213,338). In persons older than 75 years, the overall risk of mortality from MI is high with and without therapy. Although the proportionate reduction in mortality for patients older than 75 years treated within 12 hours with ST elevation or LBBB is somewhat less for patients less than or equal to 75 years, the absolute number of lives saved per 1000 patients treated is actually greater (i.e., 34 lives saved per 1000 patients treated versus 28 for those less than 75 years) (339).

Registry observations from the Cooperative Cardiovascular Project (CCP) database by Thiemann and colleagues of 2673 patients between the age of 75 and 86 years suggested that the 1607 patients receiving fibrinolytics had a lower 30-day survival than those not treated with this thera-

Table 13. Models for Estimating Risk of ICH

Risk F	'actor*
--------	---------

Risk Factor*		Model	
	Simoons et al (329)	CCP (246)	InTIME-2 (29)
Age, y	Greater than $65 = 1$ point	75 or greater = 1 point	75 or greater $= 1$ point
Weight	70 kg or less = 1 point	65 kg or less (women) = 1 point	
		80 kg or less (men) = 1 point	67 kg or less = 1 point
Hypertension	SBP:	SBP:	SBP:
on admission	170 mm Hg or greater = 1 point	160 mm Hg or greater = 1 point	160 mm Hg or greater = 1 point
	DBP:		
	95 mm Hg or greater = 1 point		
	Both SBP and DBP		
	above limits = 1 point		
Treatment			
assignment			
rtPA	1 point	1 point	_
nPA	_	_	1 point
Female	_	1 point	_
Black race	_	1 point	1 point
Prior stroke	_	1 point	1 point
Prior nifedipine use	_		1 point
Excessive anticoagulation	_	1 point	_

ICH = intracranial hemorrhage; CCP = Cooperative Cardiovascular Project; InTIME-2 = Intravenous nPA for treatment of Infracting Myocardium Early-2; SBP = systolic blood pressure; DBP = diastolic blood pressure; rtPA = recombinant alteplase; nPA = lanoteplase.

*The sum of the points for each of the models is used for interpretation of the graphs shown in Figure 17.



Figure 17. Estimation of risk of ICH with fibrinolysis. ICH = intracranial hemorrhage; CCP = Cooperative Cardiovascular Project; InTIME-2 = Intravenous nPA for treatment of Infracting Myocardium Early-2; *The number of risk factors is the sum of the points based on the criteria shown in Table 13. †If the overall incidence of ICH is assumed to be 0.75%, patients without risk factors who receive streptokinase have a 0.26% probability of ICH. The risk is 0.96%, 1.32%, and 2.17% in patients with 1, 2, or 3 risk factors, respectively. See Simoons et al. (329) for further discussion.



Figure 18. Adjusted probability of death or cerebral bleeding in relation to fibrinolytic therapy in patients with ST-elevation myocardial infarction (STEMI) who were 75 years or older (dotted line) versus that among patients with STEMI not receiving fibrinolysis (solid line). At 30 days and 1 year this was 23% and 32% versus 26% and 36%, respectively. Modified with permission from Stenestrand and Wallentin. Arch Intern Med 2003;163:965-71. Copyrighted © 2003, American Medical Association. All rights reserved (342).

py (340). Some caution in the interpretation of these observational data is appropriate because 1) a reversal of this effect toward benefit was seen in women between the ages of 65 and 75 years derived from a larger sample, and 2) a substantial proportion of the patients possessed conventional clinical trial fibrinolytic exclusion criteria (i.e., 11.8% had systolic blood pressure greater than 180 mm Hg, 18% had a history of recent trauma or a remote history of peptic ulcer or internal bleeding, and 6.9% had a history of stroke). Twice as many patients receiving fibrinolysis (i.e., 2.2% versus 0.9%) had CPR before arrival as opposed to those without fibrinolysis (340). In contrast to the CCP study by Thiemann et al., another CCP analysis in the elderly by Berger et al. indicated that both fibrinolytic therapy and primary angioplasty were associated with a survival benefit at 1 year compared with patients receiving neither (341). Data from the Swedish National Register on the use of fibrinolysis in 6891 patients 75 years and older with first registry-recorded STEMI also confirm a 13% adjusted relative risk reduction (95% CI 0.80 to 0.94; p equals 0.001; absolute risk reduction 4%) in the composite of mortality and cerebral bleeding after 1 year (Figure 18) (342).

6.3.1.6.3.4. Effect on LV Function

Early reperfusion of ischemic myocardium within the risk region of an occluded infarct-related artery interrupts the wave front of necrosis (18), reduces ultimate infarct size, preserves regional and global ventricular function, and improves survival. Clinical evidence for this paradigm was inconclusive until the GUSTO-I angiographic trial (343-348). Global LV ejection fraction (LVEF), a load-dependent measurement, is an imperfect surrogate for infarct size. Compensatory remote-segment hyperkinesis, the important prognostic effect of ventricular dilation, and the potential effects of a patent infarct artery independent of myocardial salvage confound the relationship between early reperfusion, global LV function, and survival. Poor perfusion at the myocardial cellular level due to microvascular obstruction further confounds the relationship between early reflow in the epicardial coronary artery, wall-motion improvement, and survival. In patients with TIMI 3 flow, those with abnormal myocardial tissue perfusion have worse LV function and survival than those with normal perfusion. Myocardial blush on angiography, contrast perfusion on echocardiography, and prompt complete resolution of ST elevation are measures of tissue perfusion. Poor perfusion at the myocardial cellular level is associated with increased morbidity and mortality. However, the mechanism by which poor tissue perfusion confers an adverse prognosis is not clear. Evidence of poor tissue perfusion may be the result of extensive transmural infarction with tissue edema and increased microvascular resistance (349). Alternatively, poor microvascular flow may result from distal embolization of atherothrombotic debris and hence be a target for therapeutic interventions. Flow in the infarct artery before PCI is associated with smaller infarct size and better outcome. Infarct size can be measured with SPECT sestamibi imaging (350), and this has been done in more than a dozen randomized trials.



Figure 19. Algorithm for evaluation of intracranial hemorrhage complicating fibrinolytic therapy for ST-elevation myocardial infarction. PT = prothrombin time; aPTT = activated partial thromboplastin time; LOE = level of evidence; H = hours; sOsm = serum osmolality; mOsm = milliosmoles; UFH = unfractionated heparin; LMWH = low-molecular weight heparin. *As dictated by clinical circumstances (see Section 6.3.1.6.3.5). Modified with permission from NINDS rt-PA Stroke Study Group. Stroke 1997;28:1530-40 (370).

6.3.1.6.3.5. Complications of Fibrinolytic Therapy: Neurological and Other

Class I

- 1. The occurrence of a change in neurological status during or after reperfusion therapy, particularly within the first 24 hours after initiation of treatment, is considered to be due to ICH until proven otherwise. Fibrinolytic, antiplatelet, and anticoagulant therapies should be discontinued until brain imaging scan shows no evidence of ICH. (Level of Evidence: A)
- 2. Neurology and/or neurosurgery or hematology consultations should be obtained for STEMI patients who have ICH, as dictated by clinical circumstances. (Level of Evidence: C)
- **3.** In patients with ICH, infusions of cryoprecipitate, fresh frozen plasma, protamine, and platelets should be given, as dictated by clinical circumstances. (*Level of Evidence: C*)

Class IIa

In patients with ICH it is reasonable to:

a. Optimize blood pressure and blood glucose levels (Level of Evidence: C)

- b. Reduce intracranial pressure with an infusion of mannitol, endotracheal intubation, and hyperventilation (*Level of Evidence: C*)
- c. Consider neurosurgical evacuation of ICH (Level of Evidence: C)

Hemorrhagic complications of fibrinolytic therapy primarily include ICH and other moderate or severe bleeding that may or may not require transfusion. The slight but definite excess risk of ICH occurs predominantly within the first day of therapy (156). Summaries of the incidence of ICH with various pharmacological reperfusion regimens are shown in Table 14 (26-31,33,158,261,320,322,324,331,351-369).

ICH may refer to parenchymal hemorrhage (PH), intraventricular hemorrhage, subarachnoid hemorrhage, subdural hematoma, and epidural hematoma. Between 65% and 77% of ICHs occur within 24 hours of initiation of treatment, up to 77% occur at lobar/subcortical lobar sites, 15% to 33% are multiple PHs, and up to 15% are combined PH and subdural hematoma. Typical presenting features include an acute change in level of consciousness, unifocal or multifocal neurological signs, coma, headache, nausea, vomiting, and

ials
Tr
ysis
lou
brii
E
snoi
ver
ıtra
n Ir
ctio
farc
l In
dial
car
Iyo
n N
ution
eva
Ξ
- F 1
-TS
Acute ST-
ve Acute ST-
ative Acute ST-
parative Acute ST-
Comparative Acute ST-
te Comparative Acute ST-
arge Comparative Acute ST-
n Large Comparative Acute ST-
ns in Large Comparative Acute ST-
ations in Large Comparative Acute ST-
lications in Large Comparative Acute ST-
omplications in Large Comparative Acute ST-
e Complications in Large Comparative Acute ST-
troke Complications in Large Comparative Acute ST-
4. Stroke Complications in Large Comparative Acute ST-
e 14. Stroke Complications in Large Comparative Acute ST-

Table 14. Stroke Complications in Larg	ce Comparative Acute S	ST-Elevation Myocardial Ir	farction Intravenou	s Fibrinolysis Trials		
	No. of Strokes/ No. of Patients			Strake Subt	vnes n (%)	
Study (References)	Enrolled	% Stroke*	ICH	CI	NS	HUS
Fibrinolytics with/without SC/IV UFH Strentokinase(33261,322,324,381-389)						
Subtotal	785/75221	0.90	258 (33)	316 (40)	205 (26)	16(1)
Percent of all patients			0.34	0.42	0.27	0.02
rtPA(26-29,320,322,331,353-355,357,359-366) ct	C22 V L/L V	1 60	E15 1461	150 (11)		14 (1)
Subtotal Derroart of all motionto	111/1/4 003	06.1	(04) CI C	(14) 4C4 17 0	13/ (12) 0.18	14 (1) 0 01
Fercent of all patients APSAC(357)			60.0	10.0	01.0	10.0
Subtotal	172/13 599	1.26	54 (31)	67 (39)	12 (7)	16 (9)
Percent of all patients			0.40	0.65	0.12	0.15
Accelerated rtPA plus						
reduced sueptoxinase	170/10374	1.64	84 (49)	(2) (39)	12 (7)	16 (9)
Percent of all patients			0.81	0.65	0.12	0.15
r-scuPA ⁽³⁶⁷⁾						
Subtotal	15/1698	0.88	8 (53)	7 (47		-
Percent of all patients			0.47	0.41		
Reteplase (r-PA)(26,30,324)						
Subtotal	276/21482	1.28	164 (59)	92 (33)	20 (8)	
Percent of all patients			0.76	0.43	0.09	-
Tenecteplase ^(28,31,158)						
Subtotal	193/11 320	1.70	106 (55)	72 (40)†	12 (6)†	
Percent of all patients		0.93	0.69	0.11		
Lanoteplase (nPA) ⁽²⁹⁾						
Subtotal	188/10038	1.86	112 (60)	76 (40)		
Percent of all patients			1.12	0.75		-
Grand total	2916/218395	1.34	1301 (45)	1134 (39)	459 (16)	46 (2)
Percent of all patients			0.59	0.52	0.21	0.02
Fibrinolytics with direct						
thrombin inhibitors ^(33,359,368)						
Total	124/9308	1.3	65 (52)	44 (35)	15 (13)	
Percent of all patients			0.7	0.47	0.16	
Reduced-dose fibrinolytics with						
GP IIb/IIIa inhibitors ^(30,31,369)						
Total	112/10628	1.08	72 (64)	29 (26)	11 (10)	I
Percent of all patients			0.68	0.27	0.13	
Fibrinolytics with LMWH(31,158)						
Total	57/2858	2.0	36 (63)	12(36)†	$(9)^{\ddagger}$	

seizures, at times with acute hypertension. In many cases, onset is catastrophic and rapidly fatal.

Management of suspected ICH. An algorithm for the management of ICH in the STEMI setting is provided in Figure 19 (370,371). Any change in neurological function, particularly in the first 24 hours after treatment, should be regarded as strongly indicative of PH/intraventricular hemorrhage/ subarachnoid hemorrhage/subdural hematoma/epidural hematoma until proven otherwise. Fibrinolytic, anticoagulant, antiplatelet, and combined therapies should be discontinued as soon as symptoms and signs are recognized. An emergency CT scan should be performed as soon as possible to identify the specific type of hemorrhagic complication and to measure the volume of hematoma (372,373). It is useful to document the severity of the coagulopathy, although emergency patient management should not await the results of laboratory testing. Immediate measures to reduce intracranial pressure are reasonable and include mannitol infusion, elevation of the head of the bed to 30 degrees, endotracheal intubation, and hyperventilation to achieve a pCO₂ of 25 to 30 mm Hg. Early involvement of neurologists, neurosurgeons, and hematologists will optimize treatment decisions.

Once PH, intraventricular hemorrhage, subarachnoid hemorrhage, subdural hematoma, or epidural hematoma is documented, the patient should be given 10 U of cryoprecipitate, which will increase the fibrinogen level by approximately 0.70 grams per liter and the factor VIII level by approximately 30% in a 70-kg adult. Fresh frozen plasma can be used as a source of factors V and VIII and as a volume expander. In patients who are receiving UFH, 1 mg of protamine for every 100 U of UFH given in the preceding 4 hours may be administered. If the bleeding time is abnormal, infusion of 6 to 8 U of platelets is indicated. In rare cases, antifibrinolytic agents, such as epsilon-aminocaproic acid, may be necessary (374). These replacement/reversal therapies may theoretically be accompanied by reocclusion of the infarctrelated artery. Control of blood pressure and blood glucose levels may require a compromise between competing cardiologic and neurological concerns. The decision to use various measures to reduce increased intracranial pressure, such as elevation of the head of the bed to 30 degrees, mannitol, hyperventilation, and ventriculostomy, may be based on the consensus of the management team. The use of mannitol and hyperventilation are reserved for incipient brain herniation syndromes. After the patient is stabilized, catheter-based angiography may be necessary if a ruptured berry aneurysm or arteriovenous malformation is suspected.

In GUSTO-I (375), 46 (17.2%) of 268 PH/subdural hematoma patients underwent neurosurgical evacuation by open craniotomy or burr-hole craniectomy. Patients who underwent neurosurgical evacuation had significantly higher 30-day survival rates than patients who did not (65.2% versus 35.1%, p less than 0.001), particularly in patients with

		Clasgow	Come Scale	Time from	Fibringh	reis to	Total Ua	morrhogie
Age	e, y	Glasgow	Coma Scale	Fine from	Creat ha	515 10	Volum	morriagie
A ~~	Deinte	- GCS	Deinte	Time	Unset, no	Deinte	Volum	Deinte
Age	Points	Score	Points	1 ime		Points	Volume	Points
		3	20	0		100	to 50	0
40	2	4	18	20		93	60	2
50	4	5	16	40		87	80	5
55	5	6	15	60		80	100	9
60	7	7	13	80		73	120	13
65	9	8	12	100		67	140	16
70	11	9	10	120		60	160	20
75	13	10	8	140		53	180	24
80	14	11	7	160		47	200	2.7
85	16	12	5	180		40	220	31
90	18	13	3	200		33	240	35
		14	2	220		27	260	38
		15	0	240		20	280	42
				260		13	300	46
				280		7		
				greater than or equal				
				to 300		0		
			2. Sum P	oints for All Risk F	actors			
Age	plus	GCS score	plus	Time to stroke	plus	Volume	equals	Point Tota
		3	3. Look Up Ris	sk Corresponding to	Point To	tal		
Points	Risk		Points	Risk		Points	Risk	
87	1%		109	30%		128	90%	
97	5%		115	50%		132	95%	
101	10%		120	70%		142	009/	

Figure 20. Nomogram for prediction of 30-day mortality following intracranial hemorrhage in patients receiving fibrinolysis. Reproduced with permission from Sloan et al. Circulation 1998;98:1376-82 (377).

PH. Patients with both PH and subdural hematoma had a very poor prognosis regardless of surgical treatment. There was a trend for improved functional status in patients who underwent neurosurgical evacuation compared with those who did not (nondisabling stroke 20% versus 12%, p equals 0.15). Although not definitive, these data suggest that physicians actively consider neurosurgical interventions in selected patients. For patients with spinal epidural hematoma and significant neurological deficits occurring after fibrinolysis, early multilevel decompressive laminectomy should be performed to minimize long-term disability (376). Survivors of PH/subdural hematoma/epidural hematoma should receive supportive care measures, including physical therapy, occupational therapy, speech therapy, swallowing evaluation, aspiration precautions, deep vein thrombosis (DVT) prophylaxis (pneumatic compression device), antibiotic therapy, nutritional support, and rehabilitation, where appropriate.

The risk of 30-day mortality from ICH after coronary fibrinolysis may be predicted on the basis of prior trial experience. In GUSTO-I (322), the ICH mortality rate was 59.7%. Glasgow Coma Scale score, age, time from fibrinolysis to ICH onset, hydrocephalus, herniation, mass effect, intraventricular hemorrhage, and volume and location of ICH were significant univariable predictors, with a strong trend (p equals 0.0546) for neurosurgical evacuation. The multivariable model showed that Glasgow Coma Scale score, time from thrombolysis to ICH onset, ICH volume, and age were significant predictors of mortality (377). A nomogram has been developed to calculate the risk of dying of ICH (Figure 20) (377). Prospective studies are needed to confirm the utility of this nomogram.

6.3.1.6.3.6. Comparison of Fibrinolytic Agents

All of the fibrinolytic agents currently available and under investigation are plasminogen activators (378). They work enzymatically, directly or indirectly, to expose the active enzymatic center of plasmin. Some comparative features of the approved fibrinolytic agents for intravenous therapy are presented in Table 15 (379-381).

Data from GUSTO-I (25) and GUSTO-III (26) suggest that accelerated alteplase and reteplase (administered as a double bolus) with intravenous heparin are effective therapies for achieving early coronary reperfusion and may provide an advantage over streptokinase; however, both are substantially more expensive and confer a slightly greater risk of ICH. Thus, the cost-benefit ratio is most favorable for alteplase or reteplase in patients who present early after onset of chest pain or symptoms and in those with a large area of injury (e.g., anterior infarction) and at low risk of ICH. In ASSENT-2, weight-adjusted TNK-tPA (tenecteplase) and alteplase were compared in 16949 patients. Covariate-adjusted 30-day mortality was virtually identical (i.e., 6.18% for tenecteplase and 6.15% for alteplase), which met the predefined criteria for equivalence. The rates of ICH were also similar (i.e., 0.93% for tenecteplase and 0.94% for alteplase), but in patients receiving tenecteplase, there were fewer systemic mild-to-moderate bleeding complications (26.3% versus 28.95%, p equals 0.0003) and less requirement for blood transfusion (4.25% versus 5.49%, p equals 0.0002) (28).

There is considerable ongoing investigation into the effectiveness of fibrinolytic therapy with various ancillary therapies (see Section 6.3.1.6.3.8). In 2 studies that evaluated the combination of hirudin (desirudin) with alteplase and streptokinase, there was no improvement in mortality rate, and the therapeutic-to-severe bleeding window appeared to be very narrow (382,383). See Section 6.3.1.6.8.1.3 for further discussion.

A number of proposals for selection of fibrinolytic regimens after GUSTO-I have been suggested (384-387). Additional considerations include avoiding the reuse of streptokinase, preferably indefinitely because of a high prevalence of potentially neutralizing antibody titers. Alternatively, Simoons and Arnold (386) proposed consider-

	Streptokinase	Alteplase	Reteplase	Tenecteplase-tPA
Dose	1.5 MU over 30-60 min	Up to 100 mg in 90 min (based on weight)*	$10 \text{ U} \times 2 \text{ each over } 2 \text{ min}$	30-50 mg based on weight (379)†
Bolus administration	No	No	Yes	Yes
Antigenic	Yes	No	No	No
Allergic reactions (hypotension most common)	Yes	No	No	No
Systemic fibrinogen depletion	Marked	Mild	Moderate	Minimal
90-min patency rates, approximate %	50	75	7	75 (380)
TIMI grade 3 flow, %	32	54	60	63
Cost per dose (US \$)(381)	\$613	\$2974	\$2750	\$2833 for 50 mg

Table 15. Comparison of Approved Fibrinolytic Agents

MU = mega units.

*Bolus 15 mg, infusion 0.75 mg/kg times 30 minutes (maximum 50 mg), then 0.5 mg/kg not to exceed 35 mg over the next 60 minutes to an overall maximum of 100 mg.

†Thirty milligrams for weight less than 60 kg; 35 mg for 60-69 kg; 40 mg for 70-79 kg; 45 mg for 80-89 kg; 50 mg for 90 kg or more.

ing primary PCI for those at highest risk (approximately 10% of patients), alteplase for those at moderate to high risk (40%), streptokinase for those at low to moderate risk (40%), and no lytic therapy for those at lowest risk (10%). All of these recommendations await prospective testing, and no data are available to determine the best modes for routine clinical practice.

Current use rates for fibrinolytic therapy. Because many patients have contraindications or other exclusions for fibrinolytic agents, it has been difficult to ascertain the proportion of patients with ST elevation who fail to receive fibrinolytic therapy who actually should have received such therapy (388). Critical to any such assessment of appropriateness of care, however, is whether the diagnosis of STEMI was suspected on entry into the healthcare system or whether a diagnosis made after 12 to 24 hours in the hospital or at some later point before hospital discharge. Some increase in use rates probably can be achieved, but contraindications prohibit a vast increase in the rate of use of fibrinolysis.

6.3.1.6.3.7. Net Clinical Benefit

The decision to use reperfusion therapy is based on an estimate of the patient's underlying risk without treatment, the expected benefit of the treatment, and the risk of the therapy. In general, the higher the underlying risk, the more benefit that can be gained and the fewer the number of patients who need to be treated to save 1 life.

Because decisions about reperfusion must be made rapidly in order for the intervention to be maximally effective, it is not possible to take the time to confirm the diagnosis of STEMI before administration of therapy. Data from the Multicenter Chest Pain Study suggest that approximately 80% of patients with chest pain and ST-segment elevation who present to the ED are having a STEMI (227).

Several prediction models for short-term (30 days) (240,242,298,389-393) and long-term (1 to 6 years) (392,393) mortality after reperfusion therapy for STEMI have been developed. These models have been derived from clinical trials (240,242,389,394), administrative data sets (298,390,392), and registries (391), and some have been validated in other clinical trials (242) or registries (391,394). Performance for some of these risk-prediction models, as defined by the area under the receiver operating characteristic curve, c-index, or standardized mortality ratio, is similar, with c-indices ranging from 0.74 to 0.80 and correlation coefficients for the standardized mortality ratios ranging from 0.89 to 0.92 (298). The TIMI risk score developed in the InTIME-II trial (242) performs well compared with data from the TIMI-9 trial and NRMI-3 (394). In NRMI-3, prognostic discriminatory capacity was similar between patients who received fibrinolytics and primary PCI (c-index equals 0.80). Substantial differences in predicted mortality rates (greater than 5%) between those patients who did or did not receive reperfusion therapy were observed if the TIMI risk score was greater than or equal to 7(394).

The expected benefit of fibrinolytic therapy can be estimated from the clinical trials (156,282). The fibrinolytic trials showed "absolute mortality reductions of about 30 per 1000 for those presenting within 0-6 h and of about 20 per 1000 for those presenting 7 to 12 h from onset, and a statistically uncertain benefit of about 10 per 1000 for those presenting at 13 to 18 h (with more randomized evidence needed in this latter group to assess reliably the net effects of treatment)" (156). The relative benefit, however, appeared to vary by age, with a smaller relative reduction in risk for the oldest patients.

The major risk of pharmacological reperfusion therapy is life-threatening hemorrhage (105). (See Section 6.3.1.6.3.2 for further information.) A study among Medicare patients identified older age, female sex, black race, prior stroke, systolic blood pressure greater than or equal to 160 mm Hg, lower weight (less than or equal to 65 kg for women, less than or equal to 80 kg for men), excessive anticoagulation (international normalized ratio [INR] greater than or equal to 4, prothrombin time greater than or equal to 24 sec), and choice of fibrinolytic therapy (tPA associated with greater risk than streptokinase) as risk factors for intracranial bleeding (246). Patients with none or 1 of these factors had a risk of 0.69% for intracranial bleeding, whereas those with 5 or more factors had a risk of 4.1%. Simoons et al., using data primarily from trials, identified older age, lower body weight (less than 70 kg), systolic blood pressure greater than or equal to 170 mm Hg or diastolic blood pressure greater than or equal to 95 mm Hg, and the use of tPA (versus streptokinase) as risk factors (329). Using these factors, a risk score was developed that classified patients with a risk of hemorrhage from 0.26% to 2.2% (for zero to 4 risk factors). Although this score uses dichotomous cutpoints for the risk factors, it is likely that the risk is a continuous function of the factor (e.g., the higher the blood pressure or the older the patient, the higher the risk) (322). Also, the studies of risk were based on older regimens of fibrinolytic therapy and higher dosages of UFH than are used currently. There is some evidence that newer agents, with increased fibrin specificity and bolus administration, may not increase the risk of ICH, but this issue deserves continued evaluation (156). In addition, the use of ancillary therapies may influence the risk of bleeding with fibrinolytic therapy.

6.3.1.6.3.8. Combination Therapy With GP IIb/IIIa Inhibitors

Class IIb

1. Combination pharmacological reperfusion with abciximab and half-dose reteplase or tenecteplase may be considered for prevention of reinfarction *(Level of Evidence: A)* and other complications of STEMI in selected patients: anterior location of MI, age less than 75 years, and no risk factors for bleeding. In two clinical trials of combination reperfusion, the prevention of reinfarction did not translate into a survival benefit at either 30 days or 1 year (394a) (Level of Evidence: B).

2. Combination pharmacological reperfusion with abciximab and half-dose reteplase or tenecteplase may be considered for prevention of reinfarction and other complications of STEMI in selected patients: anterior location of MI, age less than 75 years, and no risk factors for bleeding in whom an early referral for angiography and PCI (i.e., facilitated PCI) is planned. (Level of Evidence: C)

Class III

Combination pharmacological reperfusion with abciximab and half-dose reteplase or tenecteplase should not be given to patients aged greater than 75 years because of an increased risk of ICH. (Level of Evidence: B)

Studies evaluating the use of glycoprotein IIb/IIIa inhibitors as the sole means of reperfusion (i.e., without a fibrinolytic or in conjunction with PCI) do not suggest that the isolated use of a GP IIb/IIIa inhibitor restores TIMI 3 flow in a sufficient proportion of patients to make it a viable pharmacologic strategy (395a). To improve rates of achieving TIMI 3 flow by pharmacological reperfusion therapy, GP IIb/IIIa antagonists have been combined with fibrinolytic agents to achieve both platelet disaggregation and fibrinolysis (395). The phase 2 angiographic studies, TIMI-14, SPEED (Strategies for Patency Enhancement in the Emergency Department), and Intro AMI (Integrilin and low-Dose Thrombolysis in Acute Myocardial Infarction) (369,396,397), have demonstrated higher TIMI 3 flow rates at 60 to 90 minutes. When combination therapy is used, the dose of fibrinolytic agent is reduced by 50%. A large-scale mortality study, GUSTO-V (30), tested half-dose reteplase (5 U and 5 U) and full-dose abciximab (abciximab 0.25 mg/kg bolus and 0.125 mcg/kg/min [maximum of 10 mcg/min] for 12 hours) compared with full-dose reteplase (10 U and 10 U) in 16588 patients in the first 6 hours of STEMI. Thirty-day mortality rates were similar in the 2 groups (5.9% versus 5.6%). However, nonfatal reinfarction rates were reduced in the combination therapy group (2.3% versus 3.5%, p less than 0.0001), as were other complications of MI, including VF and tachycardia, high-grade AV block, and septal or freewall rupture. ICH rates were the same (0.6%), but moderate to severe bleeding was significantly increased from 2.3% to 4.6% (p less than 0.001). Excess bleeding risks appear to be limited to those over the age of 75 years, and the greatest mortality benefit was seen for those with anterior MI. ICH rates for those older than 75 years were 2.1% versus 1.1% (p equals 0.069) for combination versus full-dose reteplase. In contrast, the rates were similar for those younger than 75 years (0.5% versus 0.4%). However, there was an interaction between age and risk of ICH with therapy. Younger patients (age less than 70 years) appeared to have significantly lower ICH rates with combination therapy (398). Despite the reduction in reinfarction by combination therapy, the 1-year

mortality rates were the same (8.38%) in both groups (399). Although early reinfarction was associated with a marked increase in 1-year mortality, (22.6% versus 8.0% without reinfarction), this did not result in an overall mortality difference owing to the low reinfarction rates. For those younger than 75 years with anterior MI, 30-day mortality was 4.4% for combination therapy versus 5.8% (p equals 0.029) for full-dose rPA, and 1-year mortality was 7.1% versus 8.0% (p equals 0.260), respectively.

ASSENT-3 (31) randomized 6095 patients with STEMI to full-dose tenecteplase with UFH versus full-dose tenecteplase with enoxaparin or half-dose tenecteplase plus abciximab plus weight-adjusted, reduced-dose UFH. Similar to the GUSTO-V trial, combination of abciximab (abciximab 0.25 mg/kg bolus and 0.125 mcg/kg/min [maximum of 10 mcg/min] for 12 hours) and half-dose tenecteplase did not reduce mortality compared with full-dose tenecteplase but did result in significantly reduced in-hospital infarction and refractory ischemia. Notably, the major bleeding rate other than ICH, which was the same in the 2 groups, was increased from 2.2% to 4.3% (p less than 0.0005). Those over the age of 75 years were at greatest risk for excess bleeding, with a 3-fold increase in major bleeding complications. The tenecteplase plus enoxaparin arm showed superiority compared with UFH (see Section 6.3.1.7.8.2.1). The need for urgent PCI was reduced in the GP IIb/IIIa antagonist and fibrinolytic combination therapy arms in both trials. The heparin regimen, when combination therapy is used, is a weight-adjusted bolus of 40 U/kg (ASSENT-3) or 60 U/kg (GUSTO-V) followed by a reduced infusion dose of 7 U/kg/h. The lower bolus dose is preferable for patients who have an increased risk for bleeding.

Given the observation that patients with TIMI 3 flow before primary PCI have the best outcomes (346) and given the role of GP IIb/IIIa antagonists in PCI, some have hypothesized that administration of combination GP IIb/IIIa antagonists and half-dose fibrinolytics will facilitate primary PCI, particularly when it cannot be accomplished very rapidly. This remains to be tested prospectively in appropriately sized trials. Combination pharmacological reperfusion regimens may be associated with a slightly higher frequency of ICH and a slightly lower frequency of cerebral infarction and stroke of unknown cause than other reperfusion regimens (Table 14) (26-30,31,33,158,261,320,322,324,331,351-369,400).

6.3.1.6.4. PERCUTANEOUS CORONARY INTERVENTION. Percutaneous coronary intervention is a very effective method for re-establishing coronary perfusion and is suitable for at least 90% of patients. Considerable data (40,282,401) support the use of PCI for patients with STEMI. Reported rates of achieving TIMI 3 flow range from 70% to 90%. There is a 15% reocclusion rate after PTCA and a 5% reocclusion rate after stenting (402). Although most evaluations of PCI have been in patients who are eligible to receive fibrinolytic therapy, considerable experience supports the value of PCI for patients who may not be suitable for fibrinolytic therapy because of an increased risk of bleeding (403).

6.3.1.6.4.1. Coronary Angiography

Class I

Diagnostic coronary angiography should be performed:

- a. In candidates for primary or rescue PCI. (Level of Evidence: A)
- **b.** In patients with cardiogenic shock who are candidates for revascularization. (*Level of Evidence: A*)
- c. In candidates for surgical repair of ventricular septal rupture (VSR) or severe MR. (Level of Evidence: B)
- d. In patients with persistent hemodynamic and/or electrical instability. (Level of Evidence: C)

Class III

Coronary angiography should not be performed in patients with extensive comorbidities in whom the risks of revascularization are likely to outweigh the benefits. (Level of Evidence: C)

Acute cardiac catheterization has been proposed as an anatomic risk stratification strategy. A subset of patients will have severe 3-vessel or left main disease or anatomic features unfavorable for PCI and may be candidates for urgent or emergency CABG. Another subset of patients will have spontaneously reperfused and will have minimal evidence of atherosclerotic obstruction. They can be treated medically, which avoids the risks of fibrinolytic therapy or PCI. Additionally, identification of high-risk patients may facilitate additional strategies that will improve outcome, whereas low-risk patients may be eligible for early hospital discharge. Coronary angiography should not be performed in patients with extensive comorbidities or who will not consent to coronary revascularization regardless of the findings.

6.3.1.6.4.2. Primary PCI

See Table 11 for additional consideration for selecting reperfusion therapy.

Class I

- 1. General considerations: If immediately available, primary PCI should be performed in patients with STEMI (including true posterior MI) or MI with new or presumably new LBBB who can undergo PCI of the infarct artery within 12 hours of symptom onset, if performed in a timely fashion (balloon inflation within 90 minutes of presentation) by persons skilled in the procedure (individuals who perform more than 75 PCI procedures per year). The procedure should be supported by experienced personnel in an appropriate laboratory environment (a laboratory that performs more than 200 PCI procedures per year, of which at least 36 are primary PCI for STEMI, and has cardiac surgery capability). (Level of Evidence: A)
- 2. Specific considerations:
 - a. Primary PCI should be performed as quickly as possible with a goal of a medical contact-to-balloon or door-to-balloon interval of within 90 minutes.

(Level of Evidence: B)

b. If the symptom duration is within 3 hours and the expected door-to-balloon time minus the expected door-to-needle time is:

i) within 1 hour, primary PCI is generally preferred. (Level of Evidence: B)

ii) greater than 1 hour, fibrinolytic therapy (fibrinspecific agents) is generally preferred. (Level of Evidence: B)

- c. If symptom duration is greater than 3 hours, primary PCI is generally preferred and should be performed with a medical contact-to-balloon or doorto-balloon interval as short as possible and a goal of within 90 minutes. (Level of Evidence: B)
- d. Primary PCI should be performed for patients less than 75 years old with ST elevation or LBBB who develop shock within 36 hours of MI and are suitable for revascularization that can be performed within 18 hours of shock unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (Level of Evidence: A)
- e. Primary PCI should be performed in patients with severe CHF and/or pulmonary edema (Killip class 3) and onset of symptoms within 12 hours. The medical contact-to-balloon or door-to-balloon time should be as short as possible (i.e., goal within 90 minutes). (Level of Evidence: B)

Class IIa

- 1. Primary PCI is reasonable for selected patients 75 years or older with ST elevation or LBBB or who develop shock within 36 hours of MI and are suitable for revascularization that can be performed within 18 hours of shock. Patients with good prior functional status who are suitable for revascularization and agree to invasive care may be selected for such an invasive strategy. (Level of Evidence: B)
- 2. It is reasonable to perform primary PCI for patients with onset of symptoms within the prior 12 to 24 hours and 1 or more of the following:
 - a. Severe CHF (Level of Evidence C)
 - b. Hemodynamic or electrical instability (Level of Evidence: C)
 - c. Persistent ischemic symptoms. (Level of Evidence: C)

Class IIb

The benefit of primary PCI for STEMI patients eligible for fibrinolysis is not well established when performed by an operator who performs fewer than 75 PCI procedures per year. (*Level of Evidence: C*)

Class III

- **1.** PCI should not be performed in a noninfarct artery at the time of primary PCI in patients without hemodynamic compromise. (*Level of Evidence: C*)
- 2. Primary PCI should not be performed in asymptomatic patients more than 12 hours after onset of

STEMI if they are hemodynamically and electrically stable. (*Level of Evidence: C*)

Primary PCI has been compared with fibrinolytic therapy in 22 randomized clinical trials (173,177,306,307,404-421). An additional trial, SHOCK (301), which compared medical stabilization with immediate revascularization for cardiogenic shock, was included along with the above 22 trials in an overview of primary PCI versus fibrinolysis (40). These investigations demonstrate that PCI-treated patients experience lower short-term mortality rates (5.0% versus 7.0%, RR 0.70, 95% CI 0.58 to 0.85, p equals 0.0002), less nonfatal reinfarction (3.0% versus 7.0%, RR 0.35, 95% CI 0.27 to 0.45, p equals 0.0003), and less hemorrhagic stroke (0.05% versus 1.0%, RR 0.05, 95% CI 0.006 to 0.35, p equals 0.0001) than those treated by fibrinolysis but with an increased risk for major bleeding (7.0% versus 5.0%, RR 1.3, CI 1.02 to 1.65, p equals 0.032) (40). These results were achieved in medical centers with experienced providers and under circumstances in which PCI could be performed promptly after patient presentation (Figure 14) (40).

Additional considerations that affect the magnitude of the difference between PCI- and fibrinolysis-treated patients include the fact that UFH was used as the antithrombin with fibrinolytics as opposed to other antithrombins such as enoxaparin (see Section 6.3.1.6.8.1.1) or bivalirudin (see Section 6.3.1.6.8.1.2) that are associated with a reduction in the rate of recurrent MI after fibrinolysis; a smaller but statistically significant advantage for PCI compared with a fibrin-specific fibrinolytic versus streptokinase; and variation among the PCI arms as to whether a stent was implanted or GP IIb/IIIa antagonists were administered. Figure 14 shows the short-term and long-term outcomes of patients with STEMI treated by fibrinolysis versus PCI and the number of patients who need to be treated to prevent 1 event or cause 1 harmful complication when selecting PCI instead of fibrinolysis as the reperfusion strategy (Figure 14) (40). When primary PCI is compared with tPA and the SHOCK trial is excluded, the mortality rate is 5.5% versus 6.7% (OR 0.81%, 95% CI 0.64 to 1.03, p equals 0.081) (421a).

There is serious and legitimate concern that a routine policy of primary PCI for patients with STEMI will result in unacceptable delays in achieving reperfusion in a substantial number of cases and produce less than optimal outcomes if performed by less-experienced operators. The mean time delay for PCI instead of fibrinolysis in the randomized studies was approximately 40 minutes (40). Strict performance criteria must be mandated for primary PCI programs so that long door-to-balloon times and performance by low-volume or poor-outcome operators/laboratories do not occur. Interventional cardiologists and centers should strive for outcomes to include 1) medical contact-to-balloon or door-toballoon times less than 90 minutes, 2) TIMI 2/3 flow rates obtained in more than 90% of patients, 3) emergency CABG rate less than 2% among all patients undergoing the procedure, 4) actual performance of PCI in a high percentage of patients (85%) brought to the laboratory, and 5) risk-adjusted in-hospital mortality rate less than 7% in patients without cardiogenic shock. This would result in a risk-adjusted mortality rate with PCI comparable to that reported for fibrinolytic therapy in fibrinolytic-eligible patients (40) and would be consistent with previously reported registry experience (422-425). Otherwise, the focus of treatment should be the early use of fibrinolytic therapy (Figure 14) (40).

PCI appears to have its greatest mortality benefit in highrisk patients. In patients with cardiogenic shock, an absolute 9% reduction in 30-day mortality with coronary revascularization instead of immediate medical stabilization was reported in the SHOCK trial (301). In NRMI-II, patients with CHF had a 33% relative risk reduction with primary PCI compared with a 9% relative risk reduction with fibrinolytic therapy (302). Primary PCI in patients with anterior STEMI reduces mortality compared with fibrinolytic therapy, but there is no difference in patients with nonanterior STEMI (426,427).

Time from symptom onset to reperfusion is an important predictor of patient outcome. Two studies (294,295) have reported increasing mortality rates with increasing door-toballoon times. Other studies have shown smaller infarct size, better LV function, and fewer complications when reperfusion occurs before PCI (345,346,428). An analysis of the randomized controlled trials that compared fibrinolysis with a fibrin-specific agent versus primary PCI suggests that the mortality benefit with PCI exists when treatment is delayed by no more than 60 minutes (Figure 15) (305). Mortality increases significantly with each 15-minute delay in the time between arrival and restoration of TIMI 3 flow (doorto-TIMI 3 flow time), further underscoring the importance of timely reperfusion in patients who undergo primary PCI (429). Importantly, after adjustment for baseline characteristics, time from symptom onset to balloon inflation is significantly correlated with 1-year mortality in patients undergoing primary PCI for STEMI (RR equals 1.08 for each 30minute delay from symptom onset to balloon inflation, p equals 0.04) (Figure 21) (275). Given that the door-to-needle time goal is 30 minutes, this Writing Committee joins the Task Force on the Management of Acute Myocardial Infarction of the European Society of Cardiology in lowering the recommended medical contact-to-balloon or door-to-balloon time goal from 120 to 90 minutes in an attempt to maximize the benefits for reperfusion by PCI (297) (Figure 22) (294).

If the expected door-to-balloon time exceeds the expected door-to-needle time by more than 60 minutes, fibrinolytic treatment with a fibrin-specific agent should be considered unless it is contraindicated. This is particularly important when symptom duration is less than 3 hours but is less important with longer symptom duration, when less ischemic myocardium can be salvaged. In both the CAPTIM trial (173), which showed lower mortality with prehospital fibrinolysis than with primary PCI, and the PRAGUE-2 trial (177), which showed lower mortality with primary PCI after interhospital transfer than with on-site fibrinolysis, PCI was superior to fibrinolysis when symptom duration was greater



Figure 21. Symptom onset-balloon time and mortality in primary PCI for ST-elevation myocardial infarction. RCT = randomized controlled trial; PCI = percutaneous coronary intervention. The relationship between time-to-treatment and 1-year mortality, as continuous functions, was assessed using a quadratic regression model. The dotted lines represent 95% confidence intervals of the predicted mortality. Modified with permission from De Luca. Circulation 2004;109:1223-25 (275).

than 2 to 3 hours but not when symptom duration was shorter (see Section 6.3.1.6.2.1). In the early hours of STEMI, prompt fibrinolytic therapy can decrease infarct size and the risk of developing cardiogenic shock (176).

6.3.1.6.4.2.1. Complications of primary PCI.

Potential complications of an invasive strategy for treating STEMI include problems with the arterial access site;

adverse reactions to volume loading, contrast medium, and antithrombotic medications; technical complications; and reperfusion events. Reocclusion occurs in 10% to 15% of patients after PTCA but in fewer than 5% after stent implantation. Likewise, angiographic restenosis occurs in 30% to 40% of patients after PTCA but in 15% to 20% after stent implantation. The management of these complications is beyond the scope of this guideline (430-432).



Time to PTCA (minutes)

Figure 22. Relationship between 30-day mortality and time from study enrollment to first balloon inflation. Patients assigned to angioplasty in whom angioplasty was not performed are also shown. PTCA = percutaneous transluminal coronary angioplasty. Reprinted with permission from Berger et al. Circulation 1999;100:14-20 (294).

6.3.1.6.4.2.2. Primary PCI in fibrinolytic-ineligible patients.

Class I

Primary PCI should be performed in fibrinolyticineligible patients who present with STEMI within 12 hours of symptom onset. (Level of Evidence: C)

Class IIa

It is reasonable to perform primary PCI for fibrinolytic-ineligible patients with onset of symptoms within the prior 12 to 24 hours and 1 or more of the following:

- a. Severe CHF (Level of Evidence: C)
- b. Hemodynamic or electrical instability (Level of Evidence: C)
- c. Persistent ischemic symptoms. (Level of Evidence: C)

Randomized controlled trials evaluating the outcome of PCI for patients who present with STEMI but who are ineligible for fibrinolytic therapy have not been performed. Few data are available to characterize the value of primary PCI for this subset of patients with STEMI; however, the recommendations in Section 4.2 are applicable to these patients. Nevertheless, these patients are at increased risk for mortality (433), and there is a general consensus that PCI is an appropriate means for achieving reperfusion in those who cannot receive fibrinolytics because of increased risk of bleeding (403,434-436).

6.3.1.6.4.2.3. Primary PCI without on-site cardiac surgery.

Class IIb

Primary PCI might be considered in hospitals without on-site cardiac surgery, provided that a proven plan for rapid transport to a cardiac surgery operating room exists in a nearby hospital with appropriate hemodynamic support capability for transfer. The procedure should be limited to patients with STEMI or MI with new, or presumably new, LBBB on ECG, and should be done in a timely fashion (balloon inflation within 90 minutes of presentation) by persons skilled in the procedure (at least 75 PCIs per year) and at hospitals performing a minimum of 36 primary PCI procedures per year. (*Level of Evidence: B*)

Class III

Primary PCI should not be performed in hospitals without on-site cardiac surgery capabilities and without a proven plan for rapid transport to a cardiac surgery operating room in a nearby hospital or without appropriate hemodynamic support capability for transfer. (Level of Evidence: C)

Reports on emergency primary PCI from hospitals without established open heart surgery or elective PCI programs have demonstrated generally favorable results (307,437-450). PCI **Table 16.** Criteria for Performance of Primary PCI at Hospitals

 Without Onsite Cardiac Surgery

- The operators must be experienced interventionalists who regularly perform elective PCI at a surgical center (at least 75 cases/year). The catheterization laboratory must perform a minimum of 36 primary PCI procedures per year.
- The nursing and technical catheterization laboratory staff must be experienced in handling acutely ill patients and must be comfortable with interventional equipment. They must have acquired experience in dedicated interventional laboratories at a surgical center. They participate in a 24-hour, 365-day call schedule.
- The catheterization laboratory itself must be well equipped, with optimal imaging systems, resuscitative equipment, and IABP support, and must be well stocked with a broad array of interventional equipment.
- The cardiac care unit nurses must be adept in hemodynamic monitoring and IABP management.
- The hospital administration must fully support the program and enable the fulfillment of the above institutional requirements.
- There must be formalized written protocols in place for immediate and efficient transfer of patients to the nearest cardiac surgical facility that are reviewed/tested on a regular (quarterly) basis.
- Primary PCI must be performed routinely as the treatment of choice around the clock for a large proportion of patients with STEMI, to ensure streamlined care paths and increased case volumes.
- Case selection for the performance of primary PCI must be rigorous. Criteria for the types of lesions appropriate for primary PCI and for the selection for transfer for emergent aortocoronary bypass surgery are shown in Table 17.
- There must be an ongoing program of outcomes analysis and formalized periodic case review.
- Institutions should participate in a 3- to 6-month-period of implementation, during which time development of a formalized primary PCI program is instituted that includes establishment of standards, training of staff, detailed logistic development, and creation of a quality assessment and error management system.

PCI = percutaneous coronary intervention; IABP = intra-aortic balloon pump; STEMI = ST-elevation myocardial infarction.

Modified from Wharton et al. J Am Coll Cardiol 1999;33:1257-65 (445).

in the early phase of an acute STEMI can be difficult and requires even more skill and experience than routine PCI in the stable patient. The need for an experienced operator and experienced laboratory technical support with availability of the broad range of catheters, guidewires, stents, and other devices (e.g., intra-aortic balloon pump [IABP]) required for optimum results in an acutely ill patient is of major importance. Careful patient selection and continuous quality

Antman *et al.* 2004 e141 ACC/AHA Practice Guidelines

Table 17. Patient Selection for Primary PCI and Emergency Aorto-Coronary Bypass at Hospitals Without Onsite Cardiac Surgery (445)

Avoid intervention in hemodynamically stable patients with:

- Significant (greater than or equal to 60%) stenosis of an unprotected left main coronary artery upstream from an acute occlusion in the left coronary system that might be disrupted by the angioplasty catheter
- Extremely long or angulated infarct-related lesions with TIMI grade 3 flow
- Infarct-related lesions with TIMI grade 3 flow in stable patients with 3-vessel disease (432)

Infarct-related lesions of small or secondary vessels

Hemodynamically significant lesions in other than the infarct artery

Transfer for emergency aortocoronary bypass surgery patients:

After primary PCI of occluded vessels if high-grade residual left main or multivessel coronary disease with clinical or hemodynamic instability present Preferably with IABP support

PCI = percutaneous coronary intervention; TIMI = Thrombolysis in Myocardial Infarction; IABP = intra-aortic balloon pump.

improvement are critical components of a successful program. If these complex patients are treated by interventionalists with limited experience at hospitals with low volume, then the gains of early intervention may be lost because of increased complications. In such circumstances, transfer to a center that routinely performs complex PCI will often be a more effective and efficient course of action. Fibrinolysis is an acceptable form of therapy and is preferable to primary PCI by an inexperienced team.

Criteria have been suggested for the performance of primary PCI at hospitals without on-site cardiac surgery (432,445) (Tables 16 and 17). Large-scale registries have shown an inverse relationship between the number of primary PCI procedures performed and in-hospital mortality (295,303,304). The data suggest that both door-to-balloon time and in-hospital mortality are significantly lower in institutions that perform a minimum of 36 primary PCI procedures per year (295). Suboptimal results may relate to operator/staff inexperience and capabilities and to delays in performing PCI for logistical reasons. From clinical data and expert consensus, the Committee recommends that primary PCI for acute STEMI performed at hospitals without established elective PCI programs should be restricted to those institutions capable of performing a requisite minimum number of primary PCI procedures (36 per year) with a proven plan for rapid and effective PCI and rapid access to cardiac surgery in a nearby hospital. The benefit of primary PCI is not well established for operators who perform fewer than 75



Figure 23. Comparison of elapsed time to fibrinolysis versus primary PCI. Time is presented as a continuous variable in minutes on the horizontal axis. For DANAMI-2, times reflect components of delay from symptom onset to randomization (vertical bar) and are further separated according to whether patients presented at community referral hospitals versus those equipped for primary PCI. For those patients randomized to PCI at a referral hospital, the 3 components of delay after randomization are related to duration of stay at referral hospital, time for transport to PCI hospital and delay from arrival at PCI hospital to balloon inflation. PCI = percutaneous coronary intervention; Rand = Randomization; SK = streptokinase; Transp = Transportation. PRAGUE-2 data modified from Widimsky et al. Eur Heart J 2003; 24:94-104.



Total Door 1 to Balloon Time: 185 minutes (137-276 minutes) Percent of Patients with Door-to-Balloon Time Less Than 90 Minutes: 3.0%

Sample Size: 1346; Time Period: January 2002 – December 2002

Figure 24. Door-to-balloon times: patients transferred in NRMI 4. Data are expressed in minutes as median time (25th percentile-75th percentile) Cath Lab = catheterization laboratory. Modified from NRMI-4 Investigators: The National Registry of Myocardial Infarction-4 Quarterly Report. Genentech, South San Francisco, CA; March, 2003:2 (458).

PCIs per year or in a hospital that performs fewer than 36 primary PCI procedures per year. In addition, the benefit of timely reperfusion of the infarct artery by primary PCI at sites without on-site surgery must be weighed against the small but finite risk of harm to the patient related to the time required to transfer the patient to a site with CABG surgery capabilities (452,453).

6.3.1.6.4.2.4. Interhospital transfer for primary PCI.

The enthusiasm for primary PCI has led to the concept of emergency interhospital transfer for catheter-based reperfusion rather than fibrinolytic therapy in the initial hospital (454-456). Complication rates are low during transport, but time to reperfusion is delayed, which results in larger infarct size and lower LVEF (457). However, as noted in Section 6.3.1.6.2.1, selection bias of patients enrolled in randomized trials likely resulted in an underestimation of the risk of interhospital transfer expected in routine practice. Five randomized trials enrolled 2466 patients, with favorable results for PCI versus fibrinolytic therapy (177,306,408,419,421). Mortality was reduced with PCI (6.8% versus 9.6%, RR 0.69, 95% CI 0.51 to 0.0.92, p equals 0.01), as was the combined end point of death, nonfatal reinfarction, and stroke (8.5% versus 15.5%, RR 0.51, 95% CI 0.39 to 0.65, p less than 0.0001). Importantly, mean time to treatment was delayed only 44 minutes in these studies (Figure 23) (177,306). In contrast, the time from presentation at the door of the first hospital to balloon inflation in the second hospital, as recorded in 1346 patients in NRMI-4, was 185 minutes in the United States in 2002 (Figure 24) (458). Emergency transport in Europe is centrally organized and more efficient United than in the States (Table 18)(177,306,408,419,421,459) (Van de Werf; oral presentation, American College of Cardiology 52nd Annual Scientific Session, Chicago, IL, March 2003). Delays in door-to-balloon time versus door-to-needle time of more than 60 minutes because of interhospital transfer might negate the potential mortality benefit of transfer for primary PCI over immediate intravenous fibrinolysis with a fibrin-specific agent as shown in these trials (305). To achieve optimal results, time from the first hospital door to the balloon inflation in the sec-

Table 18. Transport of Patients With STEMI for Primary PCI

			Time Between Randomization
Study (Reference)	No. Transported	Distance, km	and First Balloon Inflation, min
Vermeer et al. (419)	75	25-50	85*
PRAGUE-1 (408)	101	5-74	80*
AIR-PAMI (421)	71	51 plus or minus 58*	155†
PRAGUE-2 (177)	429	5-120	97*
DANAMI-2 (306)	559	3-150	90†
Total	1235	3-150	

*Mean. †Median



Figure 25. Primary stenting versus primary angioplasty. Odds Ratios and their 95% confidence intervals for primary stenting versus primary angioplasty for the risk of death, reinfarction, target vessel revascularization (TVR), and major adverse cardiac events (MACE) over a 6- to 12-month follow-up after ST-elevation myocardial infarction (STEMI). The meta-analysis was performed without (subtotal) and with (total) the abciximab arms of the CADILLAC trial. Modified from Zhu M et al. Am J Cardiol 2001;88:299. Copyright 2001, with permission from Excerpta Medica, Inc. (37).

ond hospital should be as short as possible, with a goal of within 90 minutes. Significant reductions in door-to-balloon times might be achieved by directly transporting patients to PCI centers rather than transporting them to the nearest hospital, if interhospital transfer will subsequently be required to obtain primary PCI.

6.3.1.6.4.3. Primary Stenting

Of the 22 randomized trials that compared primary PCI with fibrinolytic therapy, 12 involved a comparison of primary PCI with stenting and fibrinolytic therapy (40,173,177,306,307,408,409,414,417-421). These investigations demonstrate that PCI-treated patients experience lower mortality rates (5.9% versus 7.7%, OR 0.75, 95% CI 0.60 to 0.94, p equals 0.013), less reinfarction (1.6% versus 5.1%, OR 0.31, 95% CI 0.21 to 0.44, p equals 0.0001), and less hemorrhagic stroke than those treated by fibrinolysis (40). Compared with PTCA, intracoronary stents achieve a better immediate angiographic result with a larger arterial lumen, less reocclusion and restenosis of the infarct-related artery, and fewer subsequent ischemic events.

Primary stenting has been compared with primary angioplasty in 9 studies (38,37,460-467). There were no differences in mortality (3.0% versus 2.8%) or reinfarction (1.8% versus 2.1%) rates. However, major adverse cardiac events were reduced, driven by the reduction in subsequent targetvessel revascularization with stenting (Figure 25) (37). Preliminary reports suggest that compared with conventional bare metal stents, drug-eluting stents are not associated with increased risk when used for primary PCI in patients with STEMI (468). Postprocedure vessel patency, biomarker release, and the incidence of short-term adverse events were similar in patients receiving sirolimus (n equals 186) or bare metal (n equals 183) stents. Thirty-day event rates of death, reinfarction, or revascularization were 7.5% versus 10.4%, respectively (p equals 0.4) (468).

6.3.1.6.4.4. Facilitated PCI

Class IIb

Facilitated PCI might be performed as a reperfusion strategy in higher-risk patients when PCI is not immediately available and bleeding risk is low. (*Level of Evidence: B*)

Facilitated PCI refers to a strategy of planned immediate PCI after an initial pharmacological regimen such as fulldose fibrinolysis, half-dose fibrinolysis, a GP IIb/IIIa inhibitor, or a combination of reduced-dose fibrinolytic therapy and a platelet GP IIb/IIIa inhibitor. Facilitated PCI should be differentiated from primary PCI without fibrinolysis or GP IIb/IIIa inhibitor therapy, from primary PCI with a GP IIb/IIIa inhibitor started at the time of PCI, and from rescue PCI after unsuccessful fibrinolysis. Potential advantages include earlier time to reperfusion, improved patient stability, greater procedural success rates, higher TIMI flow rates, and improved survival rates (36,346,428,469,470). However, preliminary studies have not demonstrated any benefit in reducing infarct size or improving outcomes (471-473). It is unlikely that this strategy would be beneficial in low-risk patients.

A strategy of facilitated PCI holds promise in higher-risk patients when PCI is not immediately available. Potential risks include increased bleeding complications, especially in those 75 years of age or older (see Section 6.3.1.6.3.8), and potential limitations include added cost. Several randomized trials of facilitated PCI with a variety of pharmacological regimens are in progress (473a).

6.3.1.6.4.5. Rescue PCI

Class I

- 1. Rescue PCI should be performed in patients less than 75 years old with ST elevation or LBBB who develop shock within 36 hours of MI and are suitable for revascularization that can be performed within 18 hours of shock unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (Level of Evidence: B)
- 2. Rescue PCI should be performed in patients with severe CHF and/or pulmonary edema (Killip class 3) and onset of symptoms within 12 hours. (Level of Evidence: B)

Class IIa

- 1. Rescue PCI is reasonable for selected patients 75 years or older with ST elevation or LBBB or who develop shock within 36 hours of MI and who are suitable for revascularization that can be performed within 18 hours of shock. Patients with good prior functional status who are suitable for revascularization and who agree to invasive care may be selected for such an invasive strategy. (Level of Evidence: B)
- 2. It is reasonable to perform rescue PCI for patients with 1 or more of the following:
 - a. Hemodynamic or electrical instability (Level of Evidence: C)
 - b. Persistent ischemic symptoms. (Level of Evidence: C)

Immediately after failed fibrinolysis. Intravenous fibrinolytic therapy successfully restores coronary TIMI 2/3 flow at 90 minutes in 50% to 85% of patients with STEMI (474). In those in whom fibrinolysis is unsuccessful, antegrade coronary flow can usually be restored with PCI. Several studies have demonstrated the marked beneficial effect of infarct-related artery patency (obtained via endogenous, pharmacological, or mechanical recanalization) on survival in patients with STEMI (475,476). Survivors of STEMI with a patent infarct-related artery demonstrated at 90 minutes after treatment have an improved long-term outcome compared with those with an occluded infarct-related artery, even when LV systolic function is similar (476). Rescue (also known as salvage) PCI is defined as PCI within 12 hours after failed fibrinolysis for patients with continuing or recurrent myocardial

ischemia. Rescue PCI has resulted in higher rates of early infarct artery patency, improved regional infarct-zone wall motion, and greater freedom from adverse in-hospital events than with a deferred PCI strategy or medical therapy. The Randomized Evaluation of Rescue PCI with Combined Utilization End Points (RESCUE) trial demonstrated a reduction in rates of in-hospital death and a combined end point of death and CHF that was maintained up to 1 year after study entry for patients presenting with anterior STEMI who failed fibrinolytic therapy, when PCI was performed within 8 hours after the onset of symptoms (477). Improvement in TIMI grade flow from less than or equal to 2 to 3 may offer additional clinical benefit. Similar data are not available for patients with nonanterior STEMI.

A major problem in adopting a strategy of rescue PCI lies in the limitation of accurate identification of patients for whom fibrinolytic therapy has not restored antegrade coronary flow. Unless unsuccessful fibrinolysis is recognized and corrected quickly (within 3 to 6 hours of onset of symptoms), salvage of ischemic myocardium is unlikely. Unfortunately, clinical markers of reperfusion, such as relief of ischemictype chest discomfort, partial resolution of ST-segment elevation, and reperfusion arrhythmias, have limited predictive value in identifying failure of fibrinolysis (478). In a prior era in which the practice of PCI was less mature, immediate catheterization of all patients after fibrinolytic therapy to identify those with an occluded infarct artery was found to be impractical, costly, and often associated with bleeding complications (479,480). This strategy is being re-evaluated in clinical trials testing facilitated PCI in the contemporary PCI setting.

Even in the patient with documented failure of fibrinolysis, rescue PCI has limitations. Because extensive myocardial necrosis occurs when coronary occlusion has been present for more than 3 hours (18), PCI may not salvage a substantial amount of myocardium, considering the time delay associated with presentation of the patient to the hospital after onset of symptoms, infusion of the fibrinolytic agent, recognition of failed fibrinolysis, and subsequent initiation of PCI. Rescue PCI fails to reestablish antegrade coronary flow in approximately 10% of patients, and reocclusion of the infarct artery occurs in as many as 20% of the remainder (481), although use of GP IIb/IIIa inhibitors and stent implantation may improve these results. Unsuccessful rescue PCI is associated with a high mortality rate (482,483). Finally, coronary reperfusion occurs over the subsequent hours after fibrinolytic therapy in many patients. Although infarct artery patency is achieved in only 50% to 85% of patients 90 minutes after fibrinolytic therapy, it rises to 90% by 24 hours (474). Such late reperfusion may improve survival without the risk of invasive procedures coupled with fibrinolytic therapy. Confounding the issue, both fibrinolytic therapy and PCI may successfully restore flow in the epicardial artery but fail to improve microvascular perfusion.

Hours to days after failed fibrinolysis. Patency of the infarctrelated artery is an important predictor of mortality in survivors of STEMI (475,476). Compared with those with a patent infarct artery, survivors of STEMI with a persistently occluded artery after fibrinolysis, PCI, or no reperfusion therapy have 1) increased LV dilatation (484), 2) a greater incidence of spontaneous and inducible ventricular arrhythmias (485), and 3) a poorer prognosis (486). On the basis of observational and experimental data, it has been hypothesized that infarct artery patency may favorably influence LV remodeling and electrical stability, even if accomplished at a time when salvage of ischemic myocardium is unlikely (i.e., more than 12 hours to days after coronary artery occlusion). Five small randomized trials, which enrolled a total of 562 patients, have directly tested the hypothesis that mechanical opening of persistent total occlusions late after MI will improve long-term LV remodeling and clinical outcomes (the late open-artery hypothesis). Most studies enrolled a combination of patients that included those who had failed fibrinolysis and those who had not received reperfusion therapy (487-489), with a range from almost no fibrinolytic therapy (490) to fibrinolytic therapy in nearly all patients (491). There was wide variation in the effect of routine PCI compared with only medical therapy on LV size and function. Most studies showed no significant differences between the treatment groups (487,488). One single-center study of 83 patients with occlusions of the left anterior descending coronary artery (LAD) reported improved LV volumes and clinical outcomes (composite of CHF, MI, and death) at 6 months in the PCI group (490). In contrast, a multicenter study of 66 patients with LAD occlusions reported significantly worse LV remodeling, with progressive LV dilation at 1 year and more clinical events in the PCI group than in those assigned to optimal medical therapy alone (491). The latter included very high rates of beta-blocker and ACE inhibitor use. The largest multicenter study, DECOPI, enrolled 212 patients and reported no difference in the primary end point, the composite of death, VT, and MI at 6 months (Steg PG; oral presentation, European Society of Cardiology Congress 2003, Vienna, Austria, September 2003). Stents were used in 80% of patients in the PCI group, and GP IIb/IIIa antagonists were used in 9%. The study reached fewer than one third of the target sample size and was severely underpowered, as were all the other studies, to assess clinical events.

There are no convincing data to support the routine use of adjuvant PCI days after failed fibrinolysis or for patients who do not receive reperfusion therapy. Nevertheless, this is being done in some patients with STEMI as an extension of the invasive strategy for patients with NSTEMI. The Occluded Artery Trial (OAT) is currently randomizing patients to test whether routine PCI days to weeks after MI improves long-term clinical outcomes in asymptomatic highrisk patients with an occluded infarct-related artery (493).

6.3.1.6.4.6. PCI for Cardiogenic Shock

Class I

Primary PCI is recommended for patients less than 75 years old with ST elevation or LBBB who develop

shock within 36 hours of MI and are suitable for revascularization that can be performed within 18 hours of shock unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (*Level of Evidence: A*)

Class IIa

Primary PCI is reasonable for selected patients 75 years or older with ST elevation or LBBB who develop shock within 36 hours of MI and are suitable for revascularization that can be performed within 18 hours of shock. Patients with good prior functional status who are suitable for revascularization and agree to invasive care may be selected for such an invasive strategy. (Level of Evidence: B)

Observational studies support the value of PCI for patients who develop cardiogenic shock in the early hours of STEMI. For patients who do not have mechanical causes of shock, such as acute MR or septal or free wall rupture, mortality among those having PCI is lower than for those treated medically. However, undergoing cardiac catheterization alone, with or without PCI, is associated with a lower mortality owing to patient selection bias (494).

Two small randomized clinical trials (301,495) have further clarified the role of emergency revascularization in STEMI complicated by cardiogenic shock. Both showed a statistically insignificant but clinically important absolute 9% reduction in 30-day mortality. In the SHOCK trial (301), the survival curves continued to progressively diverge such that at 6 months and 1 year, there was a significant mortality reduction with emergency revascularization (53% versus 66%, p less than 0.03) (184). The prespecified subgroup analysis of patients less than 75 years old showed an absolute 15% reduction in 30-day mortality (p less than 0.02), whereas there was no apparent benefit for the small cohort (n equals 56) of patients more than 75 years old. These data strongly support the approach that patients younger than 75 years with STEMI complicated by cardiogenic shock should undergo emergency revascularization and support measures. Three registries (496-498) have demonstrated a marked survival benefit for elderly patients who are clinically selected for revascularization (approximately 1 of 5 patients), so age alone should not disqualify a patient for early revascularization. (See Section 7.6.5.)

Several additional discussions elsewhere in this guideline are important to consider in these patients. Intra-aortic balloon pump support or ventricular assist devices can stabilize hemodynamics so that revascularization procedures can be performed (see Section 7.6.7.6). Post hoc analyses (499-501) have suggested that GP IIb/IIIa inhibitors reduce mortality, but the studies are limited by lower than expected mortality rates, larger than expected mortality reduction, and small sample sizes (see Section 6.3.1.6.8.2.3). Although PCI in a noninfarct artery is not recommended in stable patients, it can be beneficial in hemodynamically compromised patients if the stenotic artery perfuses a large area of myocardium and



Figure 26. Recommendations for initial reperfusion therapy when cardiogenic shock complicates STEMI. Early mechanical revascularization with PCI/CABG is a class I recommendation for candidates less than 75 years of age with ST elevation or LBBB who develop shock less than 36 hours from STEMI and in whom revascularization can be performed within 18 hours of shock, and a class IIa recommendation for patients 75 years of age or older with the same criteria. Eighty-five percent of shock cases are diagnosed after initial therapy for STEMI, but most patients develop shock within 24 hours. An IABP is recommended when shock is not quickly reversed with pharmacological therapy, as a stabilizing measure for patients who are candidates for further invasive care. Dashed lines indicate that the procedure should be performed in patients with specific indications only. Recommendations for staged CABG and PCI are discussed in the text, as are definitions of moderate and severe 3-vessel CAD. PCI = percutaneous coronary intervention; STEMI = ST-elevation myocardial infarction; IABP = intra-aortic balloon pump; CAD, coronary artery disease; IRA = infarct-related artery; CABG = coronary artery bypass graft surgery; LBBB = left bundle-branch block. Modified with permission from Hochman. Circulation 2003;107:2998-3002 (502).

the procedure can be done efficiently. In patients with significant left main disease or severe 3-vessel disease and without RV infarction or major comorbidities such as renal insufficiency or severe pulmonary disease, CABG can be considered as the revascularization strategy (see Section 6.3.1.6.5) (Figure 26) (502).

6.3.1.6.4.7. Percutaneous Coronary Intervention After Fibrinolysis

Class I

- 1. In patients whose anatomy is suitable, PCI should be performed when there is objective evidence of recurrent MI. (Level of Evidence: C)
- 2. In patients whose anatomy is suitable, PCI should be performed for moderate or severe spontaneous or provocable myocardial ischemia during recovery from STEMI. (*Level of Evidence: B*)
- 3. In patients whose anatomy is suitable, PCI should be performed for cardiogenic shock or hemodynamic instability. (See Section 6.3.1.6.4.6.) (Level of Evidence: B)

Class IIa

- **1.** It is reasonable to perform routine PCI in patients with LVEF less than or equal to 0.40, CHF, or serious ventricular arrhythmias. (*Level of Evidence: C*)
- 2. It is reasonable to perform PCI when there is documented clinical heart failure during the acute episode, even though subsequent evaluation shows preserved LV function (LVEF greater than 0.40). (Level of Evidence: C)

Class IIb

Routine PCI might be considered as part of an invasive strategy after fibrinolytic therapy. (Level of Evidence: B)

Immediately after successful fibrinolysis. Randomized prospective trials examined the efficacy and safety of immediate PCI after fibrinolysis (479,480,503). These trials showed no benefit of routine PCI of the stenotic infarct-related artery immediately after fibrinolytic therapy. The strategy did not appear to salvage myocardium, improve LVEF, or prevent reinfarction or death. Those subjected to this approach appeared to have an increased incidence of adverse

events, including bleeding, recurrent ischemia, emergency CABG, and death. These studies have not been repeated in the modern interventional era with improved equipment, improved antiplatelet and anticoagulant strategies, and coronary stents, thus leaving the question of routine PCI early after successful fibrinolysis unresolved in contemporary practice. Studies of facilitated PCI are presently enrolling patients (36,396,471,504).

Hours to days after successful fibrinolysis. It was initially suggested that elective PCI of the stenotic infarct-related artery hours to days after fibrinolysis might allow sufficient time for development of a more stable hemostatic milieu at the site of previous thrombotic occlusion. In this setting, PCI would be safer and more effective in reducing the incidence of reocclusion and improving survival. Two large randomized, prospective trials from an earlier PCI era tested this hypothesis, with both concluding that 1) there are fewer complications if PCI is delayed for several days after fibrinolytic therapy and 2) routine PCI in the absence of spontaneous or provocable ischemia does not improve LV function or survival (268,505-507). Thus, in unselected patients receiving fibrinolytic therapy, PCI of the stenotic infarctrelated artery in the absence of evidence of recurrent ischemia within 48 hours did not appear to be beneficial.

Great improvements in equipment, operator experience, and adjunctive pharmacotherapy have increased PCI success rates and decreased complications. More recently, the invasive strategy for patients with NSTEMI has been given a Class I recommendation by the ACC/AHA 2002 Guideline Update for the Management of Patients With Unstable Angina/NSTEMI (4). Patients with STEMI are increasingly being treated similarly as an extension of this approach. Although 6 published reports (472,508-512) and 1 preliminary report (Lablanche JM; oral presentation, American Heart Association 2002 Annual Scientific Sessions, November 2002, Chicago, IL) support this strategy, randomized studies similar to those in NSTEMI need to be performed.

One study supports the policy of performing catheterization and subsequent revascularization for patients who do have spontaneous or inducible angina after STEMI. The DANAMI trial (515) randomly assigned 1008 survivors of a first acute MI treated with fibrinolytic therapy within 12 hours of onset of symptoms to catheterization and subsequent revascularization or standard medical therapy if they showed evidence of spontaneous or inducible angina. Those who underwent revascularization had less unstable angina and fewer nonfatal MIs during a 2.5-year period of follow-up compared with those patients randomly assigned to medical treatment only (18% and 5.6% versus 30% and 10.5%, respectively).

Days to weeks after successful fibrinolysis. Continued thrombus lysis and remodeling of the infarct artery stenosis occur over the days to weeks after successful fibrinolysis, which makes the underlying residual coronary stenosis more

stable and less prone to rethrombosis and reocclusion. Thus, delaying PCI for days to weeks after fibrinolysis might improve survival, even though earlier routine PCI does not. To date, there have not been adequately sized trials to evaluate this treatment strategy. Two older, small, randomized trials (516,517) demonstrated similar LV function, rates of reinfarction, and mortality in patients randomized to PCI or conservative therapy.

6.3.1.6.5. Acute Surgical Reperfusion.

Class I

Emergency or urgent CABG in patients with STEMI should be undertaken in the following circumstances:

- a. Failed PCI with persistent pain or hemodynamic instability in patients with coronary anatomy suitable for surgery. (Level of Evidence: B)
- b. Persistent or recurrent ischemia refractory to medical therapy in patients who have coronary anatomy suitable for surgery, have a significant area of myocardium at risk, and are not candidates for PCI or fibrinolytic therapy. (Level of Evidence: B)
- c. At the time of surgical repair of postinfarction VSR or mitral valve insufficiency. (*Level of Evidence: B*)
- **d.** Cardiogenic shock in patients less than 75 years old with ST elevation or LBBB or posterior MI who develop shock within 36 hours of STEMI, have severe multivessel or left main disease, and are suitable for revascularization that can be performed within 18 hours of shock, unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care (Level of Evidence: A)
- e. Life-threatening ventricular arrhythmias in the presence of greater than or equal to 50% left main stenosis and/or triple-vessel disease. (Level of Evidence: B)

Class IIa

- 1. Emergency CABG can be useful as the primary reperfusion strategy in patients who have suitable anatomy and who are not candidates for fibrinolysis or PCI and who are in the early hours (6 to 12 hours) of an evolving STEMI, especially if severe multivessel or left main disease is present. (Level of Evidence: B)
- 2. Emergency CABG can be effective in selected patients 75 years or older with ST elevation, LBBB, or posterior MI who develop shock within 36 hours of STEMI, have severe triple-vessel or left main disease, and are suitable for revascularization that can be performed within 18 hours of shock. Patients with good prior functional status who are suitable for revascularization and agree to invasive care may be selected for such an invasive strategy. (Level of Evidence: B)

Class III

1. Emergency CABG should not be performed in patients with persistent angina and a small area of

risk who are hemodynamically stable. (Level of Evidence: C)

2. Emergency CABG should not be performed in patients with successful epicardial reperfusion but unsuccessful microvascular reperfusion. (Level of Evidence: C)

These recommendations are supplementary to those published recently in a more complete set of general guidelines and indications for CABG (518) and are restricted to patients with STEMI and associated complications. The basis for recommending surgery in emergency circumstances is the documented benefit of CABG for severe multivessel disease or left main coronary artery stenosis, particularly with reduced LV function (518-521), with the recognition that risk of emergency CABG is greater than that for elective operation.

The widespread use of fibrinolysis and primary PCI has largely superseded CABG for acute reperfusion of patients with STEMI. However, CABG still plays an integral role in the early reperfusion strategy for some patients. In the PAMI (Primary Angioplasty in Myocardial Infarction)-2 trial (522), of 1100 patients with MI and without cardiogenic shock, 5% underwent CABG as the primary reperfusion strategy with STEMI. Mortality was 6.4% if surgery was undertaken on an urgent or emergency basis versus 2.0% if elective. Major risk factors for death included poor LV function and advanced age. In the setting of cardiogenic shock complicating STEMI, emergency CABG has been used where other interventions have failed or have not been indicated. In the SHOCK registry (523), of 136 patients undergoing emergency CABG for cardiogenic shock due to LV failure, mortality was 27.9% compared to 45.5% in 268 patients undergoing PTCA. For patients undergoing CABG within 18 hours of the onset of shock, mortality was 39.6%. In a review of 25 papers reporting the outcome of CABG in 391 patients with cardiogenic shock, mortality was 35% (524). In GUSTO-I, mortality in a similar group of patients was 29% (98 of 340) after CABG and 29% (165 of 567) (422,525,526) after PTCA. On the basis of these studies, emergency CABG should only be considered for patients with STEMI with severe coronary artery disease. In the SHOCK trial, emergency CABG was performed at a median of 14 hours after the onset of STEMI in 40% of those who underwent early revascularization; most of the patients undergoing CABG had significant left main or 3-vessel coronary artery disease. The 30-day mortality rate was similar to those with less severe coronary artery disease who underwent PTCA (42% versus 45%).

6.3.1.6.6. PATIENTS WITH STEMI NOT RECEIVING REPERFUSION. Many patients with suspected STEMI do not receive reperfusion therapy. For some of these patients, the lack of treatment represents a missed opportunity. For others, patient preference led to a decision that the clinical benefit was not worth the risk of the therapy. Other patients may have contraindications to treatment owing to comorbid disease. Few studies have examined the care and outcomes of

patients with suspected STEMI who do not receive reperfusion therapy. Many of the studies (e.g., beta-blocker trials) that established therapies for MI patients preceded the reperfusion era, and so their efficacy in patients with STEMI who did not receive reperfusion is clear. The acute use of aspirin was shown to be effective in patients who did and did not receive fibrinolytic therapy. Guideline-based recommendations for nonreperfusion treatments should not vary whether or not patients received reperfusion therapy. The major difference is that patients not receiving reperfusion therapy are considered to have a higher risk for future adverse events (261). (See Section 6.3.1.6.8.1.2 for discussion of the TETA-MI trial.)

6.3.1.6.7. Assessment of Reperfusion.

Class IIa

It is reasonable to monitor the pattern of ST elevation, cardiac rhythm, and clinical symptoms over the 60 to 180 minutes after initiation of fibrinolytic therapy. Noninvasive findings suggestive of reperfusion include relief of symptoms, maintenance or restoration of hemodynamic and or electrical stability, and a reduction of at least 50% of the initial ST-segment elevation injury pattern on a follow-up ECG 60 to 90 minutes after initiation of therapy. (Level of Evidence: B)

A high priority exists for the development of simple, accurate, readily available noninvasive techniques to assess the success of pharmacological reperfusion early, i.e., 60 to 90 minutes after administration of therapy. Prior studies evaluating clinical and ECG outcome measures of reperfusion used angiographic TIMI 2 or 3 flow as the "gold standard"; angiographic assessment of epicardial flow is now considered inadequate to completely assess myocardial perfusion. Indeed, it is now clear that microvascular perfusion may be impaired despite achievement of TIMI 3 flow and less than 50% coronary narrowing; moreover, abnormal microperfusion has negative prognostic implications (395,527,528).

Myocardial contrast echocardiography, myocardial angiographic perfusion with assessment of angiographic blush in the myocardium, and ECG assessment of ST resolution are recognized as useful techniques for assessing myocardial perfusion. The relatively simple and readily available evaluation of the ECG ST-segment resolution that exceeds 50% at 60 to 90 minutes after reperfusion is a good indicator of enhanced myocardial perfusion (527). This finding is also associated with enhanced recovery of LV function, reduced infarct size, and improved prognosis (277,349,395,529-531). In the TIMI-14 study of 888 patients, those with TIMI 3 perfusion and greater than 70% ST-segment resolution had substantial enhancement of survival compared with those without ST-segment resolution and angiographically patent infarct arteries (531).

Santoro and colleagues (532) evaluated 158 consecutive patients with STEMI referred for direct angioplasty within 6

hours of symptom onset. In their observational study of patients with TIMI grade 3 flow and less than 30% residual stenosis, 42 patients had less than 50% reduction in maximal ST elevation in a single lead versus 75 patients with at least 50% reduction in ST elevation. Those with ST-segment resolution had enhanced infarct-zone functional recovery and improved ejection fraction. The reduction of ST-segment elevation was the only independent predictor of functional recovery.

Persistence of unrelenting ischemic chest pain, absence of resolution of the qualifying ST-segment elevation, and hemodynamic or electrical instability are generally indicators of failed pharmacological reperfusion and the need to consider rescue PCI. Aggressive medical support may be necessary in the interim. (See Section 6.3.1.6.4.5.)

6.3.1.6.8. ANCILLARY THERAPY. Ancillary therapy plays a key role in the overall management of patients with STEMI and can be usefully categorized as conjunctive, in which case it facilitates and maintains coronary reperfusion, or adjunctive, which aims to limit the consequences of myocardial ischemia, enhance myocardial healing, and reduce the likelihood of recurrent events.

6.3.1.6.8.1. Antithrombins as Ancillary Therapy to Reperfusion Therapy

After rupture of a vulnerable or high-risk plaque, its contents are exposed to the passing bloodstream. Vulnerable plaques are laden with both lipid and collagen and are rich in tissue factor, thereby resulting in activation of the coagulation cascade, which ultimately results in the deposition of fibrin strands. In addition, platelets are activated and aggregate. Thrombin that is generated as a consequence of activation of the coagulation cascade is a pivotal molecule not only for the formation of fibrin strands but also for activation of platelets. Therefore, there is considerable rationale for ancillary therapy to inhibit the coagulation cascade in patients with STEMI, including both those who do and do not receive reperfusion therapy. The general term used to include agents that alter the function of 1 or more proteins in the coagulation cascade is antithrombins (533). However, such a broad term does not do justice to the biochemical complexities of agents that may inhibit the coagulation cascade at multiple positions (e.g., UFH and LMWH) or in a single position (e.g., direct antithrombins). In addition to establishing and maintaining patency of the infarct-related artery, the rationale for prescribing antithrombins in selected patients with STEMI includes prevention of DVT, pulmonary embolism, LV mural thrombus formation, and cerebral embolization.

6.3.1.6.8.1.1. Unfractionated heparin as ancillary therapy to reperfusion therapy.

Class I

1. Patients undergoing percutaneous or surgical revascularization should receive UFH. (Level of Evidence: C)

- 2. Unfractionated heparin should be given intravenously to patients undergoing reperfusion therapy with alteplase, reteplase, or tenecteplase with dosing as follows: bolus of 60 U/kg (maximum 4000 U) followed by an infusion of 12 U/kg/hr (maximum 1000 U) initially adjusted to maintain activated partial thromboplastin time (aPTT) at 1.5 to 2.0 times control (approximately 50 to 70 seconds). (Level of Evidence: C)
- 3. Unfractionated heparin should be given intravenously to patients treated with nonselective fibrinolytic agents (streptokinase, anistreplase, urokinase) who are at high risk for systemic emboli (large or anterior MI, atrial fibrillation (AF), previous embolus, or known LV thrombus). (Level of Evidence: B)
- 4. Platelet counts should be monitored daily in patients taking UFH. (Level of Evidence: C)

Class IIb

It may be reasonable to administer UFH intravenously to patients undergoing reperfusion therapy with streptokinase. (*Level of Evidence: B*)

Despite the use of UFH (533) in STEMI for over 40 years, there is continued controversy regarding its role. In patients who are treated with fibrinolytic therapy, recommendations for UFH therapy depend on the fibrinolytic agent chosen. The nonspecific fibrinolytic agents (streptokinase, anistreplase, and urokinase) that produce a systemic coagulopathy, including depletion of factors V and VIII and massive production of fibrin(ogen) degradation products, are themselves anticoagulants. From this perspective, the need for conjunctive systemic anticoagulation with these agents conceptually is less compelling. However, the procoagulant potential of streptokinase, which induces extensive plasmin-mediated thrombin activity, has been noted as the rationale for antithrombotics (534). The rationale for UFH is clear for the more fibrin-specific agents, such as alteplase, reteplase, and tenecteplase. They induce less effect on the systemic coagulation system, and in many patients, very little breakdown of fibrinogen or depletion of coagulation factors is evident (535,536). Furthermore, the same procoagulant increase in thrombin activity is seen (534).

Over 60 000 patients were enrolled in the randomized ISIS-3 (357) and GISSI-2 (Gruppo Italiano per lo Studio della Streptochinasi nell'Infarto Miocardico)/International (353,354) trials comparing subcutaneous UFH with no routine heparin in conjunction with streptokinase, anistreplase, and alteplase. During the period in which UFH was given, a small reduction in mortality (4 to 5 lives per 1000 treated) was observed in ISIS-3; however, by 30 days, the 2 to 3 lives saved per 1000 treated was no longer statistically significant. A small excess rate of hemorrhagic stroke (1 to 2 per 1000 treated patients) was observed together with a larger excess in systemic bleeding (3 to 5 per 1000 patients), although total stroke rate was not significantly increased. A meta-analysis of these and several smaller studies enrolling a total of 68 000 patients showed that 5 lives were saved per 1000 patients treated with UFH in addition to streptokinase (537). In the GUSTO-I trial (25), more than 20 000 patients treated with streptokinase were randomly assigned to routine intravenous versus routine subcutaneous UFH. No significant differences were observed in death, reinfarction, or nonhemorrhagic stroke rates, whereas excess rates of systemic bleeding and hemorrhagic strokes (trend) were observed in the intravenous UFH group. There was a 36% crossover rate from subcutaneous to intravenous UFH in this trial.

Several angiographic studies have evaluated coronary perfusion as a function of UFH therapy (538-540). More rapid resolution of ST-segment elevation has been reported in patients treated with intravenous UFH immediately at the time of streptokinase infusion than in those treated with intravenous heparin started at a later time, but the OSIRIS study (Optimization Study of Infarct Reperfusion Investigated by ST Monitoring) showed no difference in perfusion at 24 hours (539). In the GUSTO-I angiographic substudy, patients treated with intravenous UFH had an 88% patency rate at 5 to 7 days compared with a 72% rate in patients treated with subcutaneous UFH (p less than 0.05), although less reinfarction occurred in the subcutaneous UFH group (3.4% versus 4.0%, p less than 0.05) (538). When these angiographic studies are viewed as a whole, intravenous UFH appears to have no clinical advantage over subcutaneous administration when used with a nonspecific fibrinolytic agent, and the evidence for use of subcutaneous UFH is equivocal (541). There are few data comparing intravenous UFH to placebo.

The clinical importance of the procoagulant increase in thrombin activity after streptokinase administration is supported by the beneficial effect of newer antithrombins used in conjunction with streptokinase (see Section 6.3.1.6.8.1.3). The HERO (Hirulog and Early Reperfusion or Occlusion)-2 trial demonstrated reduced reinfarction with intravenous bivalirudin compared with intravenous UFH (33). The AMI-SK study demonstrated in patients treated with streptokinase improved ST-segment resolution at 180 minutes and higher rates of infarct-related artery patency at 8 days for enoxaparin compared with placebo. The composite of death, MI, and recurrent angina was reduced, but severe bleeding was increased (1.6% versus 0.8%), with no difference in ICH (0% to 0.4%) (542). Additionally, a preliminary report of 5year GUSTO-I follow-up data demonstrated similar survival rates for streptokinase with UFH versus alteplase-assigned patients. In the context of these new data and the event reduction (5 fewer deaths per 1000 patients) demonstrated in the meta-analysis (537), the recommendation for intravenous UFH administration with non-fibrin-specific fibrinolytic agents was changed from Class III to Class IIb.

When alteplase is the fibrinolytic agent, the empirical information to confirm the pathophysiological reasoning discussed above is primarily inferential. In a series of angiographic trials (543-545), intravenous UFH led to higher rates of infarct-related artery perfusion in conjunction with alteplase. A direct relation between duration of aPTT and the likelihood of infarct-related artery perfusion was observed (544,545). An overview (546) points out, however, that the

effects of intravenous UFH on clinical outcomes from these studies are not so convincing; a significant increase in the rate of bleeding and nonsignificant increases in rates of reinfarction and hemorrhagic and nonhemorrhagic stroke are evident (546). These negative findings are tempered by a point estimate of an 18% reduction in mortality with broad confidence limits. Until the uncertainty is resolved, it appears judicious to use UFH for at least 48 hours with alteplase and to target the aPTT to 1.5 to 2.0 times control (approximately 50 to 70 seconds).

When primary PCI is chosen as the route of reperfusion, weight-adjusted boluses of heparin of 70 to 100 U/kg are recommended. This recommendation does not come specifically from empirical data in the setting of STEMI but from general observations in the setting of angioplasty that an activated clotting time of at least 250 to 350 seconds with the HemoTec device and 300 to 350 seconds with the Hemochron device is associated with a lower rate of complications than lower activated clotting times (432,547,548).

When GP IIb/IIIa antagonists are used (see Section 6.3.1.6.8.2.3), the UFH bolus should be reduced to 50 to 70 U/kg to achieve a target activated clotting time of 200 seconds with either the HemoTec or Hemochron device (432). UFH doses used during PCI for failed fibrinolysis should be similarly reduced and further lowered if used with GP IIb/IIIa antagonists as well.

The dose of UFH in the fibrinolytic-treated patient remains somewhat controversial. Given the infarct-related artery perfusion results just described, it would be reasonable to recommend an aPTT value more than 3-fold higher than the median control value. However, observational data strongly support a lower aPTT, because death, stroke, reinfarction, and bleeding were found to be lowest in the aPTT range of 50 to 70 seconds, or approximately 1.5 to 2.0 times the control value (549). Because of the evidence that the measured effect of UFH on the aPTT is important for patient outcome and that the predominant variable mediating the effect of a given dose of heparin is weight (549), it is important to administer the initial doses of UFH as a weight-adjusted bolus (541). For fibrin-specific (alteplase, reteplase, and tenecteplase) fibrinolytic-treated patients, a 60 U/kg bolus followed by a maintenance infusion of 12 U/kg/h (with a maximum of 4000 U bolus and 1000 U/h initial infusion for patients weighing more than 70 kg) is recommended. The recommended weight-adjusted dose of UFH, when it is administered without fibrinolytics, is a 60 to 70 U/kg IV bolus and a 12 to 15 U/kg/h infusion (4). Higher UFH doses are required for DVT and pulmonary embolism (80 U/kg and 18 U/kg/h) (550,551). Other factors that prolong aPTT include age, sex, and creatinine level. Elderly women may require lower bolus doses. Diabetics, smokers, and very heavy patients (weight more than 100 kg) may require higher UFH doses (550,552). When used with fibrinolytic therapy, an aPTT goal of 60 to 90 seconds is associated with an unacceptably high rate of ICH (359,553). The recommendation of an aPTT of 50 to 70 seconds was based on the GUSTO trials and supported by an overview of several fibrinolytic trials (325,380). ASSENT-3 was the first large-scale trial that used the recommended reduced-dose weight-adjusted UFH regimen (31). This regimen resulted in similar ICH rates but less bleeding than the higher dose used in ASSENT-2, without an increase in ischemic events. An aPTT measurement and dose adjustment are required beginning at 3 hours for those who receive UFH with fibrinolytics (28). The aPTT should be remeasured 6 hours after each dose adjustment until it is in the target range and daily thereafter. There is wide variability in aPTT measurement between laboratories, and it is not known what UFH level, as measured by anti-Xa activity, corresponds with an aPTT of 50 to 70 seconds. However, for most thromboplastin reagents, this corresponds to 0.2 to 0.5 U/mL heparin by anti-Xa activity.

Once UFH has been started, the appropriate duration of therapy is uncertain. The only randomized trial to address this issue found that discontinuation of UFH after 24 hours after fibrinolytic therapy with alteplase resulted in no measurable increase in ischemic events (554), although this study did not have adequate power to detect modest differences. A reasonable approach is to use intravenous UFH for 48 hours and then to use UFH according to the clinical characteristics of the patient. UFH may be discontinued in low-risk patients, given subcutaneously in patients at high risk of systemic embolization, and given intravenously in patients at high risk for coronary reocclusion.

There is concern that when UFH is discontinued abruptly, the patient undergoes a high-risk period for recurrent thrombosis (heparin rebound) because of increased thrombin activity (555,556). Despite this concern, no specific policy has been tested to attempt to reduce this clinical rebound effect. Several ongoing studies, however, are reducing UFH infusions in a gradual fashion (e.g., by half within 6 hours, then discontinuing over the subsequent 12 hours).

Platelet counts should be monitored daily in patients being treated with UFH. Evidence suggests the incidence of heparin-induced thrombocytopenia is 3% and that this is associated with a substantial risk of prothrombotic events (557). If the platelet count drops below 100 000, a test for heparin-induced thrombocytopenia should be obtained, and the clinician should be vigilant for thrombotic complications, because the prognosis in patients with thrombocytopenia is substantially worse (558).

6.3.1.6.8.1.2. Low-molecular-weight heparin as ancillary therapy to reperfusion therapy.

Class IIb

Low-molecular-weight heparin might be considered an acceptable alternative to UFH as ancillary therapy for patients aged less than 75 years who are receiving fibrinolytic therapy, provided that significant renal dysfunction (serum creatinine greater than 2.5 mg/dL in men or 2.0 mg/dL in women) is not present. Enoxaparin (30-mg IV bolus followed by 1.0 mg/kg SC every 12 hours until hospital discharge) used in combination with full-dose tenecteplase is the most comprehensively studied regimen in patients aged less than 75 years of age. (Level of Evidence: B)

Class III

- 1. Low-molecular-weight heparin should not be used as an alternative to UFH as ancillary therapy in patients aged more than 75 years who are receiving fibrinolytic therapy. (Level of Evidence: B)
- 2. Low-molecular-weight heparin should not be used as an alternative to UFH as ancillary therapy in patients less than 75 years who are receiving fibrinolytic therapy but have significant renal dysfunction (serum creatinine greater than 2.5 mg/dL in men or 2.0 mg/dL in women). (Level of Evidence: B)

There have been no definitive phase III randomized trials of LMWH in patients with STEMI to provide a firm basis for recommendations. However, a number of phase II clinical trials provide encouraging information that suggests that LMWH may be an attractive alternative to UFH. These clinical trials include those with LMWH as ancillary therapy to fibrinolysis and those in patients not receiving fibrinolysis (31,158,542,559-567). These 2 broad categories of trials with LMWH involve either enoxaparin or dalteparin, the 2 LMWHs studied most extensively in patients with STEMI. Clinical evaluation of LMWH as ancillary therapy to most of the commonly prescribed fibrinolytics has been reported with the exception of reteplase (Table 19) (31,158,542,559-567).

The available data suggest that the rate of early (60 to 90 minutes) reperfusion of the infarct artery either assessed angiographically or by noninvasive means is not enhanced by administration of a LMWH. However, a generally consistent theme of a lower rate of reocclusion of the infarct artery, reinfarction, or recurrent ischemic events emerges in patients receiving LMWH regardless of whether the control group was given placebo or UFH.

The most comprehensive data available are from the ASSENT-3 trial, in which patients received tenecteplase and either UFH (bolus 60 U/kg; initial infusion 12 U/kg/h; duration of treatment equals 48 hours) or enoxaparin (bolus 30 mg; subcutaneous injections 1.0 mg/kg every 12 hours; duration of treatment equals duration of hospital stay) (31). Each of the elements of the composite end point of 30-day mortality, in-hospital reinfarction, or in-hospital recurrent ischemia were reduced with enoxaparin treatment. This was associated with a slight, nonsignificant increase in noncerebral bleeding complications. In patients aged more than 75 years, the rate of noncerebral major bleeds was 4.1% with UFH and 7.2% with enoxaparin. Patients with significant renal dysfunction (serum creatinine greater than 2.5 mg/dL for men and greater than 2 mg/dL for women) were excluded from ASSENT-3, and therefore enoxaparin cannot be recommended for use in combination with tenecteplase in patients with severe renal dysfunction until more data are available. At 1 year, no difference was noted in the composite end point noted above between the UFH and enoxaparin groups in ASSENT-3 (31).

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	9. Irials of I	This (The formation)	N.			T MANTE Dans	$\mathbf{D}_{i+1} = \mathbf{D}_{i+1}$	Control %	I.MWH 62
MP 16 HART MLS50 40 PA UFH 5000 Ublus: 30 mg IV holis: 90 min TMI 3 flow grade 48 53 HART MLS50 40 PA 15 Mag at least 72.h 12.h for at least 72.h Parent IK-AP 1 rK. 30 50 43 AMI SK (542) 49 SK Dang IV holis: 12.h for at least 72.h 12.h for at least 72.h 20.mg IV holis: 20.mg IV holis: 20.mg IV holis: 20.mg IV holis: 21.h for at least 72.h 21.h for 3.mg holis: 21.h for 3.mg holis: <th></th> <th>ITIAI (REIEFEIICE)</th> <th>.0N</th> <th>r IDTHIOIYUC</th> <th>COLLEG</th> <th>TIM M I DOSE</th> <th>EAR FORM</th> <th></th> <th></th>		ITIAI (REIEFEIICE)	.0N	r IDTHIOIYUC	COLLEG	TIM M I DOSE	EAR FORM		
HART ILG550 40 PA Discrete Total Production of Find Production of Find Program Production of Find Program Production of Find Production Productin Production Production Productin Production Productin	rapy tı	0							
AMI SK (542) 49 5K Placebo 12 h for at laast 27 h (for at laast 27 h) 12 h for at laast 27 h (may benochage 50 45 AMI SK (542) 49 5K Placebo 30 mg IV bolts: Complete ST segment 25 30 45 35 30 45 35 30 45 35 36 45 36 30 45 35 36 45 36 36 45 36 45 36 45 36 45 36	_	HART II (559)	400	tPA	UFH 5000 U bolus;	30 mg IV bolus;	90-min TIMI 3 flow grade	48 9.8	53 5.9
AMI SK (542) 40 5K Pacebo 30 ang IVbolts; Comptementing 30 <td></td> <td></td> <td></td> <td></td> <td>15 U/kg/h infusion</td> <td>1 mg/kg SC every</td> <td>Reocclusion of initially</td> <td></td> <td></td>					15 U/kg/h infusion	1 mg/kg SC every	Reocclusion of initially		
All IK (542) 49 9K Pacebo 30 mg V bolus; Complete ST segment 25 36 All IK (542) 49 9K Pacebo 30 mg V bolus; Complete ST segment 25 36 BARED (560) 300 SK, UFH (5000 U) 30 mg V bolus; Complete ST segment 27 88 BARED (560) 300 SK, UFH (5000 U) 40 mg V bolus; Complete ST segment 26 36 APSAC; bolus, then 30000 Mitcement and Bin min 210 134 MICCentert argin Mitcement and Bin min 264 255 APSAC; bolus, then 3000 40 mg V bolus; Death, reinfarction, or 364 255 ASENOX (561) 312 SK (15 MU U per 241) Bing and 90 days 100 100 ASENOX (561) 312 SK (15 MU U min 120 (sec) 364 255 364 255 ASENOX (561) 312 SK (15 MU U min 120 (sec) 264 256 30 ASENOX (561) 312 SK (15 MU U min 210 (sec) 264 271 264 <td< td=""><td></td><td></td><td></td><td></td><td>TOT at least 12 h</td><td>12 n for at least 12 n</td><td>patent IKA by 1 WK</td><td>3.0</td><td>3.6</td></td<>					TOT at least 12 h	12 n for at least 12 n	patent IKA by 1 WK	3.0	3.6
AMI SK (542) 49 K Placebo 30 ng V bolus: common and R0 min 25 36 RAMI SK (542) 49 K Placebo 30 ng V bolus: complete ST-segment 25 36 RAMI SK (542) 300 SK UFH (5000 U W) 40 mg V bolus: Placef RA anglography 72 8 RAMISD (560) 300 SK UFH (5000 U W) 40 mg V bolus: Placef RA anglography 71 214 RAMISD (560) 300 SK UFH (5000 U W) 40 mg V bolus: Placef Adv 216 235 RAMISD (510) 300 SK UFH (5000 U W) 40 mg V bolus: 216 214 RAMISD (510) 300 SK UFH (5000 U M) 40 mg V bolus: 216 214 RAMISD (510) 312 SK (15 MU UFH (5000 U M) 40 mg V bolus: 210, 00 mg V 214 214 ABSNOX (561) 312 SK (15 MU UFH (4000 U M) 40 mg V bolus: 10, 00 mg V 214 214 ABSNOX (561) 31							111911 major nemormage 30-dav mortalitv	5.0	4.5
AMI SK (542) 40 50 0 mg V bolus: I 2h for 5-8 days Complete Steepment at 8 d (p = 0.01) 72 88 BAIRD (560) 300 S.K. UFH f500 U IV PRAC 0 mg V bolus: Rate of dam/sterurent and a grant and APSAC 210 234 235 BAIRD (560) 300 S.K. UFH f500 U IV PRAC 0 mg V bolus: PRAC Death, reinfraction, or and a dip = 0.01) 210 234 APSAC Niles, then 30000 0 mg V bolus: PRAC Death, reinfraction, or and a r 90 days (p=0.03) 7.4 2.4 APSAC U per 241) 0 mg V bolus: PRAC Death, reinfraction, or angin a r 90 days 364 2.55 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg V bolus: Practinision with unstable 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg V bolus: Practinision with unstable 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg V bolus: Practinision with unstable 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg V bolus: Practinision with unstable 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg V bolus: Practinision with unstable 4.0 3.0 ASENOX (5			0		-			25	36
Ingress C every and G = 0001) resource and own and G = 0001) 21 8 BAIRD (560) 300 SK, TAX UFH (5000 U M) 40 mg IV bolus; Rate of darperory 210 134 BAIRD (560) 300 SK, TAX UFH (5000 U M) 40 mg IV bolus; Rate of darperory 210 134 BAIRD (560) 300 SK, TAX UFH (5000 U M) 40 mg IV bolus; Recurrent MI 210 245 ASENOX (561) 312 SK, U.J. MU UFH (3000 U M) 40 mg IV bolus; Recurrent MI 204 255 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Readmission with unstable 40 30 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Readmission with unstable 40 30 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Readmission with unstable 40 30 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Readmission with unstable 40 30 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Read of noninvasive signs of 0.03 50 40 ASENOX (561) 312 SK (1.5 MU UFH (1000 U M) 40 mg IV bolus; Read of noninvasive signs of 0.03 50		AMI SK (542)	49	SK	Placebo	30 mg IV bolus;	Complete S1-segment		
BAIRD (560) 300 SK, IPA UHI (5000 UN PASAC, IPA 40 mg IV bolus: APSAC, IPA UHI (5000 UN PASAC, IPA 40 mg IV bolus: APSAC, IPA 210 134 BAIRD (560) 300 SK, IPA UHI (5000 UN PASAC, IPA 40 mg IV bolus: APSAC, IPA 40 mg IV bolus: APSAC, IPA 24 24 APSAC, IPA UPE 24 h) Description of angina 190 days 364 255 APSAC, IPA UPE 1000 40 mg IV bolus: APSAC, IPA 24 24 APSAC, IPA UHI (5000 UN B) for 4 days 40 mg IV bolus: angina 490 days 100 26 ASENOX (561) 312 SK (15 MU Description death 100 100 23 ASENOX (561) 312 SK (15 MU Description death 100 23 26 ASENOX (561) 312 SK (15 MU Description death 24 23 ASENOX (561) 312 SK (15 MU Description death 24 23 ASENOX (561) 312 SK (15 MU Description death 23 26 ASENOX (561) 312 SK (15 MU Description death 24 23 ASENOX (561) 312 SK (15 MU Description death 24 23 ASENOX (561) 312 SK (15 MU Description death 26 26 ASENTRE-TIM 123 (562) 483 <						1 mg/kg oC every 12 h for 5-8 davs	Patent IRA at angiography	72	88
BAIRD (560) 300 S.K. UFH (5000 U M) 40 mg IV bolus: Nifrecurrent angina 100 mog S0 (5000 O) 74 24 BAIRD (560) 300 S.K. UFH (5000 U M) 40 mg IV bolus: Death, reinfraction, or 36.4 25.5 RASAC, bolus, then 30000 8 h for 4 days Core admission with unstable 36.4 25.5 Reurear MI 2 mg N at 30 days Core admission with unstable 30.4 25.5 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 10.6 6.0 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 10.6 6.0 2.4 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 10.5 0.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 2.6 2.7 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 2.0 2.0 ASENOX (561) 312 SK (1.5 MU UFH 1000 U/h 40 mg IV bolus: 2.6 2.7 ASENOX (561)<							at 8 d (p = 0.001)	21.0	13.4
BAIRD (560) 300 SK, UFH (5000 U) 40 mg V bolus; Recurren MI 7.4 2.4 PSAC, Uper 241) ApSAC, Uper 241) 8 h for 4 days 7.4 2.4 APSAC, Uper 241) 8 h for 4 days 7.4 2.4 2.5 APSAC, Uper 241) 8 h for 4 days 7.4 2.5 ASENOX (561) 312 SK (1.5 MU UFH 1.000 Uh 40 mg V bolus; 7.6 2.0.4) 10.6 6.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 Uh 40 mg V bolus; Reinfarction, or 3.0 4.7 Reinfarction over 20 min) for 48-72 h Reinfarction 10.6 6.0 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1.000 Uh 40 mg V bolus; Reinfarction 6.0 4.7 Reinfarction over 20 min) for 48-72 h Reinfarction 6.0 4.7 ASENOX (561) 312 SK (1.5 MU UFH (bolus 60 U/k; Reinfarction 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH (bolus 60 U/k; 8							Rate of death/recurrent	0.17	
BAIRD (560) 300 SK, UFH (5000 U/V) 40 mg V bolus; Death, reinfraction, or 364 255 BAIRD (560) 300 SK, UFH (5000 U/V) 40 mg V bolus; Death, reinfraction, or 364 255 Recurrent MI Uper 241h) 8 h for 4 days readinasion with unstable 106 60 RENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; 10.5 0.6 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; 10.6 6.0 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; Readmission with unstable 6.0 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; Readmission with unstable 6.0 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; 10.6 6.0 4.0 3.0 ASENOX (561) 112							MI/recurrent angina through 30 days (n=0.03)		
BAIRD (560) 300 SK, TPA, TPA UFH (5000 U IV) bolus, then 30000 40 mg IV bolus; 40 mg SC every angina at 90 days 36.4 25.5 APSAC, TPA Uper 24 h) 40 mg SC every angina at 90 days 10.6 6.0 APSAC, TPA Uper 24 h) 40 mg SC every angina at 90 days 10.6 6.0 APSAC, TPA Uper 24 h) 8 h for 4 days 0.04) 10.6 6.0 APSAC, TPA Uper 24 h) 8 h for 4 days 0.04) 10.6 6.0 4.7 APSAC, TPA Uper 24 h) APSAC 40 mg SC every 10.8 10.6 6.0 4.7 APSAC 312 SK (1.5 MU UTH 1.000 U/h 40 mg IV bolus: Ratificartion 6.0 4.7 APSAC 312 SK (1.5 MU UTH 1.000 U/h 40 mg IV bolus: Angina 6.0 4.7 APSAC 312 SK (1.5 MU UTH 1.000 U/h 40 mg IV bolus: 6.0 4.0 5.6 APSAC APSAC APSAC APSAC 6.0 4.0 5.6 APS							Recurrent MI	7.4	2.4
APAC Dots, then 30000 4 mag SC events readmission with unstable (PA Uper 24 h) Uper 24 h) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) (0 = 0.0) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA Uper 24 h) Uper 24 h (0 = 0.0) (0 = 0.0) (0 = 0.0) (0 = 0.0) (0 = 0.0) (PA (PA PA PA PA PA		RAIRD (560)	300	SK	11FH (\$000 11 IV	40 mg IV holus:	Death reinfarction or	36.4	25.5
(PA) U per 24 (h) 8 h for 4 days (n = 004) (n = 004) (n = 004) ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 60 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h 40 mg IV bolus: Readmission with unstable 6.0 4.0 3.0 ASENOX (561) 112 Readmission with unstable 6.0 4.0 5.0 6.0 5.0 ASENOX (561) 12 Med SeNOX Med No 1 0.0 6.0 7.14			000	APSAC.	bolus. then 30000	40 mg SC everv	readmission with unstable		
ASENOX (561) 312 SK (1.5 MU UFH 1.000 U/h $\frac{1}{100}$ Readmission with unstable $\frac{1}{60}$ $\frac{1}{60}$ $\frac{1}{47}$ Readmission with unstable $\frac{1}{60}$ $\frac{1}{60}$ $\frac{1}{47}$ argina $\frac{1}{100}$ Barl reaction theorethage $\frac{1}{60}$ \frac				tPA	U per 24 h)	8 h for 4 days	angina at 90 days		
ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus; Reinfraction 60 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus; Reinfraction 60 4.7 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus; Reinfraction 60 4.7 angina Clinically significant hemorrhage 40 3.0 Early noninvasive signs of 76 80 76 80 76 80 76 80 76 80 76 80 76 80 714 714 714 714 30 00 JV 80 714 45-72 h 1 mg/kg SC every 12 h for 48-72 h 2.6 over 20 min) 7 UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina conclusion through 30 days 8.2 7.14 dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IN bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IN bolus dose TNK; UFH (holus 60 U/kg; 1 mg/kg SC every 12 h 100 minvasive signs of 0.3 angina 30 mg IV bolus dose 7 h 100 minvasive 100 minvasiv							(p = 0.04)	10.6	6.0
ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus: Reinfarction with unstable 60 4.7 angina angina angina clinically significant hemorrhage 4.0 3.0 Clinically significant hemorrhage 4.0 3.0 cver 20 min) for 48-72 h 1 mg/kg SC every 2 mg rate of noninvasive signs of 6.3 2.6 g and 1.2 h for 48-72 h 1 mg/kg SC every 12 h mortality and 12 h for 48-72 h 1 mg/kg SC every 12 h mortality and 12 U/kg/h) for either with our without 12 U/kg/h) for either with or witho							Cardiac death	19.9	14.8
ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h dong IV bolus; angina continuent in angina cover 20 min) for 48-72 h 1 mg/kg SC every 3 angina continvasive signs of 5.3 2.6 and 5.1 mg/kg SC every 12 h for 48-72 h 1 mg/kg SC every 12 h for 48-72 h 2 h for 48-72 h 2 h 1 mg/kg SC every 12 h for 48-72 h 2 h 1 mg/kg SC every 12 h for 48-72 h 2 h 1 mg/kg SC every 12 h for 48-72 h 2 h 1 mg/kg SC every 12 h for 48-72 h 2 h 1 mg/kg SC every 12 h 1 mg/							Reintarction Readmission with unstable	6.0	4.7
ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus; Cinically significant hemorrhage 4.0 3.0 ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg IV bolus; Early nomivasive signs 76 80 over 20 min) for 48-72 h 1 mg/kg SC every Rate of noninvasive signs of 6.3 2.6 BNTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 67 51 ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 50 51 ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 50 51 Gos TNK; UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 50 51 Gos TNK; UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 50 51 Gos TNK; UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortaliy 50 51 Gos TNK; UFH (bolus 60 U/kg; 1 mg/kg SC every 1 mg/kg SC every 65 5.5 U/kg/h) for patients 0 U/kg; 1 mg/kg SC every <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>incautilission with unstauto</td><td></td><td></td></t<>							incautilission with unstauto		
ASENOX (561)312SK (1.5 MU over 20 min)UFH 1,000 U/h for 48-72 h40 mg IV bolus; 1 mg/kg SC every 12 h for 48-72 hEarly noninvasive signs of reperfusion7680ASENOX (561)312SK (1.5 MU over 20 min)UFH 1,000 U/h40 mg IV bolus; I mg/kg SC everyEarly noninvasive signs of reperfusion6.32.6ASENTRE-TIMI 23 (562) 483TNKUFH (bolus 60 U/kg; infusion 12 U/gdh) for either with or without patients receiving full- inful-dose TNK group.Montality Death or recurrent MI through 30 days (p = 0.005)7.14 5.0ADU/kg; infusion 7Img/kg SC every 12 h through 30 days (p = 0.005)Montality 5.05.94.4 1.9ADU/kg; infusion 7Dose ranging from 0.3 with full-dose TNK plus abcixinab2.68.2 5.57.14 5.2ADU/kg; infusion 7Dose ranging from 0.3 with full-dose TNKMontality through 30 days (p = 0.005)15.94.4 1.9ADU/kg; infusion 7Dose ranging from 0.3 with full-dose TNKWith full-dose TNK with full-dose TNK5.28.5ADU/kg; initial IV bolus of 30 mg in mitial IV bolus of 30 mg in mitial IV bolus of 30 mg in with full-dose TNK2.41.9ADU/kg; initial IV bolus of 30 mg in mitial IV bolus of 30 mg in mitial IV bolusTIMI major heorating with full-dose TNK2.41.9ADU/kg; mitial IV bolus of 30 mg in mitial IV bolus of 30 mg in mitial IV bolus of 30 mg in mitial IV bolusTIMI major heorating mitial IV bolus2.41.9							auguta Clinically significant hemorrhage	4.0	3.0
ASENOX (561) 312 SK (1.5 MU UFH 1,000 U/h 40 mg V bolus; Early noninvasive signs 6.3 2.6 over 20 min) for 48-72 h 1 mg/kg SC every Rate of noninvasive signs of 6.3 2.6 over 20 min) for 48-72 h 1 mg/kg SC every Rate of noninvasive signs of 6.3 2.6 over 20 min) for 48-72 h 1 mg/kg SC every Rate of noninvasive signs of 6.3 2.6 over 20 min) for 48-72 h 1 mg/kg SC every Rate of noninvasive signs of 6.3 2.6 FNTRE-TIM 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIM 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIM 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIM 23 (562) 483 TNK UFH (bolus 60 U/kg; 1 mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIM 23 (562) 483 TNK TIM 13 0 mg IV bolus TIM 13 0 mg V (p = 0.005) 5.9 5.9 4.4							-0	76	80
over 20 mm) for 48-72 h I mg/kg SC every kate of nomivasive signs of of reperfusion ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; infusion 12 U/kg/h) for patients receiving full- I mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; infusion 12 U/kg/h) for I mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; infusion 12 U/kg/h) for I mg/kg SC every 12 h Mortality 8.2 7.14 ENTIRE-TIMI 23 (562) 483 TNK U/kg/h) for either with or without TIMI 3 flow rate at 60 min 8.2 7.14 ENTIRE-TIMI 23 (562) 483 TO/kg/h) for either with or without TIMI 3 flow rate at 60 min 50 51 Alower TNK U/kg/h) for patients to 0.75 mg/kg SC every with full-dose TNK 6.5 5.5 U/kg/h) for patients to 0.75 mg/kg SC every with full-dose TNK 5.2 5.5 Plus abciximab initial IV bolus of 30 mg in with full-dose TNK 5.2 8.5 Plus abciximab patients receiving half-dose with full-dose TNK 5.2 8.5		ASENOX (561)	312	SK (1.5 MU	UFH 1,000 U/h	40 mg IV bolus;	Early noninvasive signs	6.3	2.6
ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; i) mg/kg SC every 12 h Mortality reocclusion through 30 days 8.2 7.14 ENTIRE-TIMI 23 (562) 483 TNK UFH (bolus 60 U/kg; i) mg/kg SC every 12 h Mortality for cocclusion through 30 days 8.2 7.14 infusion 12 U/kg/h) for either with or without TIMI 3 flow rate at 60 min 50 51 dose TNK; UFH (bolus in full-dose TNK group. through 30 days (p = 0.005) 15.9 4.4 40 U/kg; infusion 7 Dose ranging from 0.3 with full-dose TNK 5.5 5.5 U/kg/h) for patients to 0.75 mg/kg SC every through 30 days (p = 0.005) 15.9 4.4 plus abciximab tintial IV bolus of 30 mg in with full-dose TNK 5.2 8.5 plus abciximab patients receiving half-dose to 10.75 mg/kg SC every with full-dose TNK 5.2 8.5				over 20 min)	tor 48-72 h	1 mg/kg SC every	Rate of noninvasive signs of		
ENTIRE-TIMI 23 (562) 483TNKUFH (bolus 60 U/kg; infusion 12 U/kg/h) for patients receiving full- dose TNK; UFH (bolus dose TNK group. dose ranging from 0.3 U/kg/h) for patients treceiving half-dose TNK plus abciximabImg/kg SC every 12 h through 30 days (p = 0.005) through 30 days (p = 0.005) through 30 days (p = 0.005) through 30 days (p = 0.005)8.2 5.0 4.4 4.4 6.5O/kg/h) for patients plus abciximab10.0.75 mg/kg SC every titial IV bolus of 30 mg in with full-dose TNK with combination therapy with combination therapy8.2 5.27.14 8.5							reocclusion through 30 days		
ENTIRE-TIMI 23 (562) 483TNKUFH (bolus 60 U/kg;)1 mg/kg SC every 12 hMortality5051infusion 12 U/kg/h) foreither with or withoutTIMI 3 flow rate at 60 min5051patients receiving full-initial 30 mg IV bolusDeath or recurrent MI5051dose TNK; UFH (bolusin full-dose TNK group.through 30 days (p = 0.005)15.94.4U/kg/h) for patientsto 0.75 mg/kg SC everywith full-dose TNK6.55.5U/kg/h) for patientsto 0.75 mg/kg SC everywith full-dose TNK5.28.5plus abciximabinitial IV bolus of 30 mg inwith full-dose TNK5.28.5							reconnection antonion of any of	8.2	7.14
infusion 12 U/kg/h) foreither with or withoutTIMI 3 flow rate at 60 minpatients receiving full-initial 30 mg IV bolusDeath or recurrent MIdose TNK; UFH (bolusin full-dose TNK group.through 30 days (p = 0.005)15.94.4dose TNK; infusion 7Dose ranging from 0.3with full-dose TNK6.55.5U/kg/h) for patientsto 0.75 mg/kg SC everywith full-dose TNK6.55.28.5plus abciximabinitial IV bolus of 30 mg inwith full-dose TNK5.28.5		ENTIRE-TIMI 23 (562	?) 483	TNK	UFH (bolus 60 U/kg;	1 mg/kg SC every 12 h	Mortality	50	51
pattents receiving full-initial 30 mg IV bolusDeath or recurrent MIdose TNK; UFH (bolusin full-dose TNK group.through 30 days (p = 0.005)15.94.4dose TNK; infusion 7Dose ranging from 0.3with full-dose TNK6.55.5U/kg/h) for patientsto 0.75 mg/kg SC everywith combination therapy6.55.5plus abciximabinitial IV bolus of 30 mg inwith full-dose TNK5.28.5					infusion 12 U/kg/h) for	either with or without	TIMI 3 flow rate at 60 min		
dose TNK; UFH (bolusin full-dose TNK group.through 30 days (p = 0.005)15.94.440 U/kg/h) for patientsto 0.75 mg/kg SC everywith full-dose TNK6.55.5U/kg/h) for patientsto 0.75 mg/kg SC everywith combination therapy6.55.5plus abciximabinitial IV bolus of 30 mg inwith full-dose TNK5.28.5					patients receiving full-	initial 30 mg IV bolus	Death or recurrent MI		
40 UKg; intusion / Dose ranging from 03 with rull-dose UNK 6.5 5.5 U/kg/h) for patients to 0.75 mg/kg SC every with combination therapy 6.5 5.5 U/kg/h) for patients to 0.75 mg/kg SC every with combination therapy 6.5 5.2 receiving half-dose TNK 12 h either with or without TIMI major hemorrhage 2.4 1.9 plus abciximab initial IV bolus of 30 mg in with full-dose TNK 5.2 8.5					dose TNK; UFH (bolus	in full-dose TNK group.	through 30 days ($p = 0.005$)	15.9	4.4
receiving half-dose TNK 12 h either with or without TIMI major hemorrhage plus abciximab initial IV bolus of 30 mg in with full-dose TNK 5.2 8.5 patients receiving half-dose with combination therapy 5.2 8.5					40 U/Kg; intusion / II/ko/h) for natients	to 0.75 mo/ko SC everv	with run-dose LNK with combination therapy	6.5	5.5
plus abciximab initial IV bolus of 30 mg in with full-dose TNK 5.2 8.5 patients receiving half-dose with combination therapy 5.2 8.5					receiving half-dose TNK	12 h either with or without	TIMI maior hemorrhage		
patients receiving half-dose with combination therapy 0.2					plus abciximab	initial IV bolus of 30 mg in	with full-dose TNK	2.4	1.9
						patients receiving half-dose	with combination therapy	7.6	C.8

dial Infa M -7 With ST-Fle tio. P. .; Woicht He ÷ Mole fΤ Trials

ACC - www.acc.org

e152

Continued on next page

	Trial (Reference)	No.	Fibrinolytic	Control	LMWH Dose	End Point	Control, %	LMWH, %
Ancillary therapy to fibrinolvsis (continue	;(b;							
	ASSENT3 (31)	6095*	TNK	UFH IV bolus of 60 U/kg (maximum 4000 U), initial infusion of 12 U/kg/h (maximum 1000 U/h) for a	30 mg IV bolus, 1 mg/kg SC q 12 h until hospital discharge or successful revascu-	30-day mortality, inhospital reinfarction, or inhospital refractory ischemia	15.4	11.4
				minimum of 48 h	larization (maximum 7 days)	30-day mortality Inhospital reinfarction Inhospital refractory ischemia	6.0 6.5 6.5	5.4 2.7 4.6
						Major bleeding other than ICH	2.2	3.0
	ASSENT: 3 PLUS (158)	1639	INK	UFH IV bolus of 60 U/kg (maximum 4000 U), initial infusion of 12 U/kg/h (maximum 1000 U/h) for a minimum of 48 h	30 mg IV bolus, 1 mg/kg SC q 12 h until hospital discharge or successful revascu- larization (maximum 7 days)	30-day mortality, inhospital reinfarction, or inhospital refractory ischemia (p = 0.08) Major bleeding ICH (p = 0.05) ICH in patients greater than	17.4 2.8 1.0 0.78	14.2 4.0 6.71
						or equal to 75 years old		
Dalteparin	ASSENT-PLUS (563)	439	tPA	UFH	120 IU/kg q 12 h for 4-7 days	TIMI 3 flow at median of 6 days 30 days mortality	62.5	69.35
						In-hospital reinfarction $(p = 0.01)$	5.4 5.4	4.1 1.4
	FRAMI (564)	517	SK	Placebo		LV thrombus or arterial embolism (p=0.03)	22	14
						Major hemorrhage $(p = 0.006)$	0.3	2.9
	BIOMACS II (565)	101	SK	Placebo		TIMI 3 flow at 20-28 hours Ischemic episodes $6-24$ h after start of Rx (p = 0.04)	51 38	68 16
LMWH in patients r receiving fibrinolvsis	not ::							
Enoxaparin	TETAMI (566)	1224		Enoxaparin 30 mg IV bolus;	30 mg IV bolus; 1 mg/kg SC BID	Death, MI, or recurrent angina at 30 days†	15.4	16.1
				1 mg/kg SC BID for 2-8 days	+ tirofiban 10 mcg/kg IV and 0.1 mcg/kg/min infusion for 2.8 days	Death, recurrent MI† Major hemorrhage (TIMI)‡ Hemorrhagie errobet	8.7 1.0	8.5 2.0
				UFH 70 U/kg bolus; 15 U/kg/h IV for 2-8	UFH 70 U/kg bolus; 15 U/kg/h IV +	Death, MI, or recurrent angina at 30 days ⁺	0.0 17.3	$0.3 \\ 17.2$
				days	tirofiban 10 mcg/kg IV and 0.1 mcg/kg/min infusion for 2-8 days	Death, recurrent MI† Major hemorrhage (TIMI)‡ Hemorrhagic stroke‡	9.2 1.0 0.0	8.0 1.6 0.3
Dalteparin	FAMI (567)	1178		Placebo	Open label dalteparin 120 IU/kg twice daily for 1-3 days followed by double-blind dalteparin 7500 IU (or placebo) for 30 days	Death, MI, or recurrent angina, emergency revascularization	6.7	7.0
LMWH = low-molecular streptokinase; min = *UFH = 2038; enoxaparin †Randomized patients. ‡Treated patients.	weight heparin; tPA = altepla minute(s); $d = day(s)$; $MI =$ = 2040; half-dose TNK plus abc	se; IV = in myocardial ciximab plus	I infarction; TIMI = I infarction; APSAC reduced-dose UFH	Thrombolysis in Myocardial Infarction; 1 C = anistreplase; MU = mega units; TN = 2017.	UFH = unfractionated heparin; h = 1 dK = tenecteplase; No. = number;	hour; IRA = infarct-related artery; wk = we ICH = intracranial hemorrhage; LV = left	eek(s); SC = sub ventricular; BID	cutaneous; SK = = twice a day.

Table 19. Continued

e153
However, results from the third Assessment of the Safety and Efficacy of a New Thrombolytic PLUS (ASSENT-3 PLUS) trial underscore the need for continued evaluation of the safety of LMWH as an adjunct to fibrinolysis (158). Among 1639 patients with STEMI receiving tenecteplase and either enoxaparin or UFH in a prehospital setting, higher rates of both major bleeding (4.0% versus 2.8%; p equals)0.18) and ICH (2.2% versus 1.0%; p equals 0.05) were seen in the enoxaparin group than in the UFH group. There was a significant interaction between patient age and risk of bleeding because almost all cases of excess ICH were confined to patients older than 75 years (158,568). The Enoxaparin and Thrombolysis Reperfusion for Acute Myocardial Infarction Treatment-Thrombolysis In Myocardial Infarction - Study 25 (EXTRACT-TIMI-25) trial is evaluating enoxaparin versus UFH in patients receiving fibrinolytic therapy and will provide information on the efficacy and safety of reduced doses of enoxaparin in the elderly.

The combination of tirofiban and enoxaparin was studied in 1224 patients presenting with STEMI in the Treatment of Enoxaparin and Tirofiban in Acute Myocardial Infarction (TETAMI) trial (566). Patients ineligible for fibrinolysis were randomized in a 2×2 fashion to receive either enoxaparin (intravenous 30 mg bolus and subcutaneous injection of 1 mg/kg twice daily) or UFH (intravenous 70 U/kg bolus and 15 U/kg/h infusion) with or without tirofiban (intravenous 10 mcg/kg bolus and 0.1 mcg/kg/min infusion) for 2 to 8 days. There were no differences noted in the primary efficacy end point (30-day combined incidence of death, reinfarction, or recurrent angina) between either enoxaparin and UFH monotherapy groups (15.4% versus 17.3%) or between enoxaparin and UFH combination groups (16.1% versus 17.2%). Major bleeding was rare and not statistically different among all 4 groups.

6.3.1.6.8.1.3. Direct antithrombins as ancillary therapy to reperfusion therapy.

Class IIa

In patients with known heparin-induced thrombocytopenia, it is reasonable to consider bivalirudin as a useful alternative to heparin to be used in conjunction with streptokinase. Dosing according to the HERO 2 regimen (a bolus of 0.25 mg/kg followed by an intravenous infusion of 0.5 mg/kg/h for the first 12 hours and 0.25 mg/kg/h for the subsequent 36 hours) (33) is recommended, but with a reduction in the infusion rate if the partial thromboplastin time is above 75 seconds within the first 12 hours. (Level of Evidence: B)

A number of direct thrombin inhibitors are now available for use in heparin-induced thrombocytopenia and DVT but have not yet been approved for the treatment of acute coronary syndrome (Table 20) (39,359,368,382,383,553,569-573). One, (bivalirudin) has been approved for use in patients with unstable angina undergoing PCI. A meta-analysis evaluated 11 trials that collectively enrolled more than 35 000 patients, comparing direct thrombin inhibitors with UFH (39). There was an approximately 25% reduction in the incidence of MI in patients with STEMI treated with either hirudin or bivalirudin, but there was less evident efficacy for univalent thrombin inhibitors (argatroban, efegatran, and inogatran). Major bleeding was reduced with bivalirudin compared with heparin (4.2% versus 9.0%; OR 0.44 [0.34 to 0.56]). There was an excess of bleeding after the use of hirudin and no difference with univalent inhibitors; statistical heterogeneity among these 3 groups of trials existed (39).

Subsequent to the meta-analysis, a large phase 3 study (HERO-2) of 17 073 patients with STEMI who presented within 6 hours of onset of chest pain was performed to evaluate the efficacy of bivalirudin versus UFH administered in conjunction with streptokinase (33). In the HERO-2 trial, bivalirudin did not reduce mortality compared with UFH (10.8% versus 10.9%) but was associated with a lower rate of adjudicated myocardial reinfarction within 96 hours (1.6% versus 2.3%, p equals 0.005). Although it was anticipated there would be fewer hemorrhagic complications with bivalirudin, severe bleeding occurred in 0.7% of the bivalirudin group versus 0.5% for heparin (p equals 0.07), and intracerebral bleeding occurred in 0.6% versus 0.4% (p equals 0.09), respectively, possibly related to higher aPTT levels in the bivalirudin group. The frequency of moderate and mild bleeding was also greater with bivalirudin (33,323).

Bivalirudin is currently indicated only for anticoagulation in patients with unstable angina who are undergoing percutaneous coronary angioplasty (574). On the basis of the data in the HERO-2 trial, the Writing Committee believes that bivalirudin could be considered an acceptable alternative to UFH in those patients with STEMI who receive fibrinolysis with streptokinase, have heparin-induced thrombocytopenia, and who, in the opinion of the treating physician, would benefit from anticoagulation.

6.3.1.6.8.1.4. Other.

A phase 2 angiographic trial Pentasaccharide as an Adjunct in ST-Segment Myocardial Infarction (PENTALYSE) evaluated fondaparinux, a synthetic pentasaccharide that is a highly selective inhibitor of factor Xa. Fondaparinux selectively binds antithrombin III, inducing a conformational change that increases the anti-Xa activity of antithrombin III more than 300 times, which results in dose-dependent inhibition of factor Xa (575). A total of 333 patients with evolving STEMI were treated with aspirin and alteplase and randomized to UFH given intravenously for 48 to 72 hours or to a low, medium, or high dose of fondaparinux. The percentage of patients achieving TIMI grade 3 flow at 90 minutes was 68% in the UFH control group and ranged between 60% and 69% with fondaparinux. Thus, selective factor Xa inhibition appears to be an attractive therapeutic concept in patients presenting with STEMI; however, further study is required before it can be recommended for routine administration.

Table 20. Trials	of Direct Thrombin In	hibitors Used as	s Ancillary Therapy i	in Patients With ST-Elevation My	ocardial Infarction			
Direct Thrombin Inhibitor	Trial (Reference)	а	Fibrinolytic	Control/Ref. Arm(UFH)	Direct Thrombin Inhibitor	End Point(s)	UFH	Direct Thrombin Inhibitor
Hirudin	TIMI 5 (569)	246	цРА	UFH 5000 U bolus; 1000 U/h IV to PTT 65-90 sec for 5 days	4 Ascending doses; bolus 0.15-0.6 mg/kg followed by fixed infusion 0.05-0.2 mg/kg for 5 days	TIMI 3 flow at 90 min & 18-36 h (without death or re-MI) (p=0.07) Reocclusion (p=0.07) Death/reinfarction (p=0.02) Major hemorrhage (p=0.09)	49.4% 6.7% 16.7% 4.7%	61.8% 1.6% 6.8% 1.2%
	GUSTO IIa (359)	1264	rtPA or SK	5000 U bolus & 1000 -1300 U/h to PTT 60-90 sec	0.6 mg/kg bolus & 0.2 mg/kg/h infusion without PTT adjustment	Death or MI within 30 days Major hemorrhage ICH* [95% CI] *Early termination of trial due to excess ICH	1.5% (0.5-2.4)	2.2% (1.0-3.3)
	TIMI 9a (553)	757	rtPA or SK	5000 U bolus & 1000- 1300 U/h to PTT 60-90 sec	0.6 mg/kg bolus followed by 96-h infusion of 0.2 mg/kg/h	Death/re-MI CHF Shock LVEF less than 0.40 ICH or major hemorrhage ICH or major hemorrhage (p=0.02) [95% CI] *DSMB requested early termination	$\begin{array}{c} 1.9\%\\ (0.8-3.9)\\ 3.0\%\\ (1.5-5.3) \end{array}$	$\begin{array}{c} 1.7\% \\ (0.5-3.7) \\ 7.0\% \\ 4.5-10.2) \end{array}$
	НІТ III (368)	302	rtPA	70 U/kg bolus & 15 U/kg/h	0.4 mg/kg bolus & 0.15 mg/kg/h (48-72 h duration)	Death/reinfarction at 30 days *Early termination due to excess ICH in hirudin group, i.e., 5 of 148 patients (3.4%)	%0	3.4%
	TIMI 9B (382)	3002	rtPA or SK	5000 U followed by 1000/h; PTT 55-85 sec for 96 h	0.5 mg/kg bolus; infusion 0.1 mg/kg/h for 96 h	Unsatisfactory outcome, i.e., death, re-MI, CHF, shock, and major bleeding OR 1.02 [95% CI 0.80-1.31] ICH (p=NS) Major hemorrhage (p=NS)	9.5% 0.9% 1.1%	9.7% 0.4% 1.8%
	GUSTO IIb (383)	4131	rtPA or SK	5000 U followed by 1000 U/h and 75 + 29 hours (SD)	0.1 mg/kg bolus followed by infusion of 0.1 mg/kg/h 75 + 29 hours (SD)	Death/re-MI at 30 days OR 0.86 [95% CI 0.70-1.05] ICH Severe bleeding	11.3% 0.4% 1.5%	9.9% 0.5% 1.1%

Continued on next page

Table 20. Contin	ned									
Direct Thromhin	Trial			Control/Ref	Direct Thrombin			Direct Thromhin		
Inhibitor	(Reference)	u	Fibrinolytic	Arm(UFH)	Inhibitor	End Point(s)	UFH	Inhibitor		
	HIT 4 (572)	1208	SK	12 500 U SC every 12 h for 5-7 davs	0.2 mg/kg IV bolus & 0.5 mo/kg twice a	TIMI 3 patency at 90 min $(n = 0.16)$	33.5%	40.7%		
					day SC for 5-7 days	ST resolution at 90 min $(n - 0.05)$	22%	28%		
						Composite clinical end point	24.3%	22.7%		
						(p = NS) ICH $(p = NS)$	0.3%	0.2%		
						Major bleeding Death. nonfatal stroke	3.5%	3.3%		
						Re-MI, PCI, or refractory				
						auguna ar ou days	47%	77%		
Hirulog	Pilot - hirulog strentokinase (570)	45	SK	1000 U/h to PTT 2.0-2.5 × control	0.5 mg/kg/h for 12 hours & then 0.1	TIMI 2 or 3 flow at 90 min (n less than				
				for 4.7 days	mg/kg/h for 4.7 days	0.05)	27%	13%		
						Serious bleeding (p = NS) Absence reocclusion	8%	0%0		
						(p = NS)		0.5 mg/kg/h	1.0 mg/kg/h	
	Hirulog vs heparin in STEMI (571)	70	SK	5000 U bolus & 1000 U/h	0.5 mg/kg bolus - 1.0 mg/kg for 12 h -		46% (n=13)	96% (27)	79% (28)	
					at 12 h, dose reduction to 0.1 mg/kg/h or placebo	TIMI 2/3 flow 90 min post-SK	0/13	1/27	4/28	
					for 4-6 days (4:1 random- ization)	Reocclusion Serious bleeding	4/13	6/27	5/28	
Efegatran	ESCALAT (573)	245	SK & rtPA	5000 U bolus	4 Ascending doses:		Plus rtPA 53%	Plus SK 40%		
)				followed by 1000 U/h with rtPA	0.05 mg/kg - 0.2 mø/kø holins	TIMI 3 flow at 90 min $(n = 0.10)$	7%	15%		
					infusion 0.3-1.0 mg/kg/h;	Composite clinical end	11.07			
					NG UM C.I MIM N 06-7/ JOI	point ($p = 0.14$) Major bleeding ($p = 0.07$)	0%0	0% C7		
						ICH	6.9%	6.3%		
Meta-analysis: hirudin	Direct thrombin inhibitor trialists'	9947	SK & rtPA	UFH		Death or MI at 30 -days OR = 0.91 [95% CI	(
bivalirudin argatroban	(39) (5 trials with ST elevation)					0.77-1.06]	3.4%	2.5%		
efegatran						OR = 0.75 [95% CI 0.59-0.94]	3.3%	2.9%		
0						Major bleeding OR = $0.89 [95\% \text{ CI}]$				
						0.71 - 1.11]				

UFH = unfractionated heparin; rtPA = recombinant alteplase; U = unit; min = minutes; TIMI = Thrombolysis in Myocardial Infarction; h = hours; IV = intravenous; PTT = prothrombin time; re-MI = recurrent myocardial infarction; sec = seconds; SK = streptokinase; ICH = intravenous; PTT = prothrombin time; re-MI = recurrent myocardial infarction; sec = seconds; SK = streptokinase; ICH = intravenous; PTT = prothrombin time; re-MI = recurrent myocardial infarction; sec = seconds; SK = streptokinase; ICH = intravenous; PTT = prothrombin time; re-MI = recurrent myocardial infarction; SK = streptokinase; ICH = intravenous; OR = odds ratio; SD = standard deviation; SC = subcutaneous; NS = not significant; PCI = intravenial hemorrhage; CI = confidence interval; CHF = congestive heart failure; LVEF = left-ventricular ejection fraction; DSMB = Data Safety Monitoring Board; OR = odds ratio; SD = standard deviation; SC = subcutaneous; NS = not significant; PCI = intraventage; CI = confidence interval; CHF = congestive heart failure; LVEF = left-ventricular ejection fraction; DSMB = Data Safety Monitoring Board; OR = odds ratio; SD = standard deviation; NS = not significant; PCI = intraventage; CI = confidence interval; CHF = congestive heart failure; LVEF = left-ventricular ejection; DSMB = Data Safety Monitoring Board; OR = odds ratio; SD = standard deviation; NS = not significant; PCI = interval; CHF = interval; CHF = congestive heart failure; LVEF = left-ventricular ejection; DSMB = Data Safety Monitoring Board; OR = odds ratio; SD = standard deviation; NS = not significant; PCI = interval; CHF = confidence interval; CHF = con

e156

6.3.1.6.8.2. Antiplatelets

6.3.1.6.8.2.1. Aspirin.

Class I

A daily dose of aspirin (initial dose of 162 to 325 mg orally; maintenance dose of 75 to 162 mg) should be given indefinitely after STEMI to all patients without a true aspirin allergy. (Level of Evidence: A)

As discussed in Section 7.4.4 and Section 6.3.1.4, aspirin should be given to the patient with suspected STEMI as early as possible and continued indefinitely, regardless of the strategy for reperfusion and regardless of whether additional antiplatelet agents are administered. True aspirin allergy is the only exception to this recommendation.

6.3.1.6.8.2.2. Thienopyridines.

Class I

- 1. In patients who have undergone diagnostic cardiac catheterization and for whom PCI is planned, clopidogrel should be started and continued for at least 1 month after bare metal stent implantation, for several months after drug-eluting stent implantation (3 months for sirolimus, 6 months for paclitaxel), and up to 12 months in patients who are not at high risk for bleeding. (Level of Evidence: B)
- 2. In patients taking clopidogrel in whom CABG is planned, the drug should be withheld for at least 5 days, and preferably for 7 days, unless the urgency for revascularization outweighs the risks of excess bleeding. (Level of Evidence: B)

Class IIa

Clopidogrel is probably indicated in patients receiving fibrinolytic therapy who are unable to take aspirin because of hypersensitivity or major gastrointestinal intolerance. (*Level of Evidence: C*)

Ticlopidine and clopidogrel are ADP-receptor antagonists and are quite similar chemically. Ticlopidine can cause neutropenia and thrombotic thrombocytopenia. Clopidogrel is preferred because of fewer side effects, lack of need for laboratory monitoring, and once-daily dosing. Clopidogrel combined with aspirin is recommended for patients with STEMI who undergo coronary stent implantation (576-580). There are no safety data available regarding the combination of fibrinolytic agents and clopidogrel, but ongoing trials will provide this information in the future. However, in patients in whom aspirin is contraindicated because of aspirin sensitivity, clopidogrel is probably useful as a substitute for aspirin to reduce the risk of occlusion (581). There are no safety data comparing 300 and 600 mg as loading doses for clopidogrel. We do not recommend routine administration of clopidogrel as pretreatment in patients who have not yet undergone diagnostic cardiac catheterization and in whom CABG surgery would be performed within 5 to 7 days if warranted (431).

6.3.1.6.8.2.3. Glycoprotein IIb/IIIa inhibitors.

Class IIa

It is reasonable to start treatment with abciximab as early as possible before primary PCI (with or without stenting) in patients with STEMI. (*Level of Evidence: B*)

Class IIb

Treatment with tirofiban or eptifibatide may be considered before primary PCI (with or without stenting) in patients with STEMI. (Level of Evidence: C)

The use of intravenous GP IIb/IIIa receptor inhibitors in combination with fibrinolytic agents is discussed in Section 6.3.1.6.3.8. Intravenous GP IIb/IIIa receptor inhibitors have also been studied as supportive antiplatelet therapy in patients undergoing PCI. Five randomized trials compared abciximab to placebo control in a collective total of 3666 patients undergoing primary PCI for STEMI (34-36,38,582). A total of 1843 patients received abciximab, a relatively small data set on which to base recommendations for treatment. In addition, in the setting of primary PCI, periprocedural recurrent MI is not easily measured, so the benefit of antiplatelet therapy with GP IIb/IIIa inhibitors is harder to determine. Finally, only 1 of the trials, CADILLAC (Controlled Abciximab and Device Investigation to Lower Late Angioplasty Complications), provided data on the effect of abciximab on patients who underwent PTCA without stenting and on patients who had a stent implanted at the time of PCI (38).

The ADMIRAL study (Abciximab Before Direct Angioplasty and Stenting in Myocardial Infarction Regarding Acute and Long-Term Follow-Up) (36) enrolled 300 patients with STEMI undergoing primary stenting; half received placebo and half received abciximab in the mobile intensive care unit or the ED before arrival at the catheterization laboratory. Abciximab-treated patients had higher infarct-artery patency (TIMI 2/3 flow) rates (25.9% versus 10.8%) before revascularization and a better LVEF (0.61 versus 0.57) 6 months after revascularization. Abciximabtreated patients had a lower rate of death, reinfarction, or need for subsequent target-vessel revascularization at 30 days (6.0% versus 14.6%, p equals 0.01) and at 6 months (7.4% versus 15.9%, p equals 0.02); the majority of the benefit of abciximab on the composite primary end point in ADMIRAL was driven by a reduction in urgent target-vessel revascularization. The CADILLAC study (38) enrolled 2082 patients (88% with STEMI) undergoing primary PTCA or stenting; half received placebo, and half were treated with abciximab in the catheterization laboratory. At 30 days, the incidence of the primary composite end point of death, reinfarction, revascularization, or disabling stroke was highest in the group assigned to receive PTCA alone (8.3%), and the lower rates in the other 3 groups were not significantly different from one another (4.8% PTCA plus abciximab, 5.7% stenting alone, 4.4% stenting plus abciximab). The Anticoagulation for Cardioversion using Enoxaparin (ACE)

study (582) randomized 400 patients to stenting alone or stenting plus abciximab (administered immediately before the procedure). At 30 days, the incidence of the primary composite end point of death, reinfarction, target-vessel revascularization, or stroke was reduced in the stent-plus-abciximab group (4.5%) versus the stent-alone group (10.5%); p equals 0.023); the majority of the benefit of abciximab on the primary end point in the ACE study was driven by a reduction in the rate of reinfarction. It is unclear whether the different 30-day results in the studies described above are related to patient selection and risk, timing of abciximab administration, or patency rates before revascularization (583). Assessment of the benefit of abciximab at 6 months varies depending on the composite end point, with evidence in favor of its use derived from composite end points of death/reinfarction or death/reinfarction/urgent target-vessel revascularization, whereas evidence of long-term benefit of abciximab is lost if elective revascularization is added to the end point (34,36,583).

The Writing Committee believes that it is reasonable to start treatment with abciximab as early as possible in patients undergoing primary PCI (with or without stenting), but given the size and limitations of the available data set, assigned a Class IIa recommendation. The data on tirofiban and eptifibatide in primary PCI are far more limited than for abciximab. However, given the common mode of action of the agents, a modest amount of angiographic data (584), and general clinical experience to date, tirofiban or eptifibatide may be useful as antiplatelet therapy to support primary PCI for STEMI, with or without stenting (Class IIb recommendation).

6.3.1.6.9. OTHER PHARMACOLOGICAL MEASURES.

6.3.1.6.9.1. Inhibition of Renin-Angiotensin-Aldosterone System

Class I

- 1. An ACE inhibitor should be administered orally within the first 24 hours of STEMI to patients with anterior infarction, pulmonary congestion, or LVEF less than 0.40, in the absence of hypotension (systolic blood pressure less than 100 mm Hg or less than 30 mm Hg below baseline) or known contraindications to that class of medications. (Level of Evidence: A)
- 2. An angiotensin receptor blocker (ARB) should be administered to STEMI patients who are intolerant of ACE inhibitors and who have either clinical or radiological signs of heart failure or LVEF less than 0.40. Valsartan and candesartan have established efficacy for this recommendation. (Level of Evidence: C)

Class IIa

An ACE inhibitor administered orally within the first 24 hours of STEMI can be useful in patients without anterior infarction, pulmonary congestion, or LVEF less than 0.40 in the absence of hypotension (systolic blood pressure less than 100 mm Hg or less than 30 mm Hg below baseline) or known contraindications to that class of medications. The expected treatment benefit in such patients is less (5 lives saved per 1000 patients treated) than for patients with LV dysfunction. (Level of Evidence: B)

Class III

An intravenous ACE inhibitor should not be given to patients within the first 24 hours of STEMI because of the risk of hypotension. (A possible exception may be patients with refractory hypertension.) (Level of Evidence: B)

A number of large, randomized clinical trials have assessed the role of ACE inhibitors early in the course of acute MI. All trials in which ACE inhibitors were administered orally demonstrated a benefit in mortality. In the ISIS-4 trial, 58000 patients with suspected acute MI were randomly assigned within the first 24 hours (median 8 hours) to receive either oral captopril or placebo; a significant 7% relative reduction was observed in 5-week mortality among those randomly assigned to captopril (absolute difference of 4.9 fewer deaths per 1000 patients treated for 1 month) (152). The largest benefit was among those with an anterior infarction. Among the 143 fewer deaths in the group allocated captopril, 44 occurred in days 0 through 1 and 37 in days 2 through 7 (585), which demonstrates that early therapy is important. In the GISSI-3 trial, more than 19000 patients with either STsegment elevation or depression were randomly assigned to lisinopril or open control (586). There was a significant reduction in 6-week mortality (OR 0.88; 95% CI 0.79 to 0.99); 60% of the lives were saved during the first 5 days of treatment. The SMILE (Survival of Myocardial Infarction: Long-Term Evaluation) study involved 1556 patients randomly assigned within 24 hours to receive either placebo or zofenopril (587). The patient population was restricted to those with anterior MI who had not received fibrinolytic therapy. Use of an early ACE inhibitor in this trial suggested a trend of more lives saved in the first 6 weeks (RRR 25%, p equals 0.19). A Chinese captopril study involving more than 13 600 patients with suspected acute MI also revealed an approximate 0.5% absolute mortality benefit among those who were randomly assigned to the ACE inhibitor compared with the control population (588). A meta-analysis of these major trials along with 11 smaller trials that collectively enrolled more than 100000 patients revealed a 6.5% overall odds reduction (p equals 0.006) with an absolute benefit of 4.6 fewer deaths per 1000 patients treated among those who received the ACE inhibitor (585). These data conclusively support a role for ACE inhibitors in the early and convalescent phases of STEMI.

All trials with oral ACE inhibitors have shown benefit from its early use, including those in which early entry criteria included clinical suspicion of acute infarctions. Data from these trials indicate that ACE inhibitors should generally be started within the first 24 hours, ideally after fibrinolytic therapy has been completed and blood pressure has stabilized. ACE inhibitors should not be used if systolic blood pressure is less than 100 mm Hg or less than 30 mm Hg below baseline, if clinically relevant renal failure is present, if there is a history of bilateral stenosis of the renal arteries, or if there is known allergy to ACE inhibitors.

The meta-analyses of the large ACE inhibitor trials have been useful in defining those patient subgroups most likely to demonstrate the greatest benefit from early post-MI ACE inhibitor therapy. According to a meta-analysis of nearly 100000 randomized patients, the benefits of early oral ACE inhibitors are greatest among those aged 55 to 74 years, with an anterior infarct, and with a heart rate of 80 bpm or higher (589).

ACE inhibitor therapy after STEMI should start with lowdose oral administration and increase steadily to achieve a full dose within 24 to 48 hours. For example, in ISIS-4, an initial 6.25-mg dose of captopril was given and, if tolerated, was followed by 12.5 mg 2 hours later, 25 mg 10 to 12 hours later, and then 50 mg twice per day. GISSI-3 patients received 5 mg or oral lisinopril at the time of randomization, 5 mg after 24 hours, 10 mg after 48 hours, and then 10 mg daily for 6 weeks or open control. Similar graded-dose schedules should be used with other ACE inhibitors such as ramipril, zofenopril, enalapril, and quinapril. Regarding the potential for aspirin to blunt the effect of ACE inhibitors, the Writing Committee thought that any adverse drug interaction between aspirin and ACE inhibitors was of a small magnitude and was far outweighed by the benefit of the combined administration of both drugs to patients recovering from STEMI (430,590,591). Lower doses of aspirin are likely to minimize any potential interaction.

Finally, the only trial that did not show a benefit with ACE inhibitors was the Cooperative New Scandinavian Enalapril Survival Study (CONSENSUS) II, in which patients were randomly assigned within the first day to receive either intravenous enalaprilat or placebo followed by increasing oral dosages of either enalapril or placebo (592). This trial was terminated early by its Safety Committee because of the high probability that a significant benefit of enalapril over placebo was unlikely to be demonstrated with continuation of the trial, as well as concern over an adverse effect among elderly patients who experienced an early hypotensive reaction. The 95% confidence limits ranged from showing a 7% benefit to 29% harm. Thus, intravenous enalaprilat should be avoided.

The use of ARBs has not been explored as thoroughly as ACE inhibitors in patients with STEMI. However, clinical experience in the management of patients with heart failure and data from clinical trials in patients with STEMI (see Sections 7.4.3 and 7.6.4) suggest that ARBs may be useful in patients with depressed LV function or clinical heart failure who are intolerant of an ACE inhibitor. Use of aldosterone antagonists in patients with STEMI is discussed in Sections 7.4.3 and 7.6.4.

6.3.1.6.9.2. Metabolic Modulation of the Glucose-Insulin Axis

Metabolic modulation of patients with STEMI was originally proposed by Sodi-Pallares et al. (593) in 1962. A meta-

analysis of 1932 patients in trials conducted between 1965 and 1987 demonstrated a 28% relative mortality reduction, with an absolute benefit of 49 lives saved per 1000 patients treated (Table 21) (594-597). Subsequent trials in the reperfusion era show promising but variable results. High-dose infusions of glucose-insulin-potassium (GIK) (25% glucose, 50 IU/L soluble insulin, and 80 mmol/L KCl at a rate of 1.5 mL/kg/h for 24 hours) or a low-dose infusion (10% glucose, 20 IU/L soluble insulin, and 40 mmol/L KCl at a rate of 1 mL/kg/h for 24 hours) were compared to usual care. The ECLA (Estudios Cardiológicos Latinoamérica) pilot suggested a relationship between the time from symptom onset and impact of GIK infusion; a significant reduction in mortality rate was observed in patients treated 12 hours or less after symptom onset (595). The high-dose GIK regimen is being tested in large, ongoing international trials. The potential beneficial effect of GIK in high-risk patients with acute ischemic syndromes who have been revascularized is supported by a study of 322 post-cardiac surgery patients with postoperative cardiogenic shock (598). Those assigned to GIK had a 34% (p less than 0.02) reduction in in-hospital mortality. GIK was not superior to placebo in a study of 940 patients who underwent primary PCI (597). There appeared to be an interaction between treatment and Killip class, with possible mortality reduction at 30 days in Killip class I patients and excess mortality for those in Killip class II or higher. No definitive recommendations regarding GIK can be formulated until ongoing trials are completed.

6.3.1.6.9.2.1. Strict glucose control during STEMI.

Class I

An insulin infusion to normalize blood glucose is recommended for patients with STEMI and complicated courses. (*Level of Evidence: B*)

Class IIa

- **1.** During the acute phase (first 24 to 48 hours) of the management of STEMI in patients with hyper-glycemia, it is reasonable to administer an insulin infusion to normalize blood glucose even in patients with an uncomplicated course. (*Level of Evidence: B*)
- 2. After the acute phase of STEMI, it is reasonable to individualize treatment of diabetics, selecting from a combination of insulin, insulin analogs, and oral hypoglycemic agents that achieve the best glycemic control and are well tolerated. (Level of Evidence: C)

The acute phase of STEMI is associated with a dramatic increase in catecholamine levels in the blood and ischemic myocardium. The insulin level remains low while cortisol and glucagon levels increase, which leads to decreased insulin sensitivity that contributes to impaired glucose utilization. Free fatty acid levels and the concentration of their metabolites increase, potentiating ischemic injury through several mechanisms: direct myocardial toxicity, increased oxygen demand, and direct inhibition of glucose oxidation. It has been suggested that agents that support glucose oxidation could reduce postischemic contractile dysfunction. Insulin

Author (Beference)	Voar	Samula Siza	Hours From Onset of Symptoms to Starting CIK	Reperfusion	Mortality: CIK Dose	GIK vs Placebo	OB (05% CI)*	Rollow-Un
Fath-Ordoubadi (meta-analysis) (594)	1997	1932	12-48	0.6% (100% Fibrinolytics)	4 Trials with high-dose GIK; 5 trials with low-dose GIK	16% vs 21%	0.72 0.57-0.90) 0.58 (0.30-1.10)	Inhospital (3-4 wk)
Diaz (ECLA pilot) (595)	1998	407	Less than 24	62% (95% Fibrinolytics)	135 Patients with high-dose GIK; 133 patients with low-dose GIK	6.7% vs 11.5%	$\begin{array}{c} 0.34 \\ (0.78-10.1); \\ 0.43 \\ (0.2-0.9) \end{array}$	Inhospital (average 10.1 days)
Ceremuzynski (596)	1999	954	Less than 24	59% (100% Fibrinolytics)	Low-dose GIK	8.9% vs 4.5%	1.45 (0.79-2.68)	35 days
Zijlstra (597)	2003	940	N/A	100% (100% Primary angioplasty)	High-dose GIK	4.8% vs 5.8%	0.82	30 days
GIK = glucose-insulin-potas: *GIK compared with control †Fibrinolytic-treated patients ‡GIK less than 12 hours.	ium; OR = odds	s ratio; CI = confidence in	terval; N/A = data not availabl	ن ن		S .		

Table 21. Trials of Glucose-Insulin-Potassium (GIK) for ST-Elevation Myocardial Infarction

promotes glucose oxidation, increases adenosine triphosphate levels, and may improve the fibrinolytic profile of patients with STEMI (599,600). Insulin reduces free fatty acids by reducing lipolysis and enhances glycolysis. Insulin specifically enhances glucose, lactate, and pyruvate uptake and switches the reliance of the myocardium from fat to carbohydrate without a change in oxygen consumption. The oxygen requirement of the heart is stimulated by free fatty acids without an improvement in mechanical activity (601).

Intensive insulin management of endogenous elevation of glucose in diabetics, supplemented by potassium as needed, has potential metabolic benefits similar to GIK for nondiabetics. The DIGAMI study randomized 620 diabetic patients to intensive insulin therapy with an insulin-glucose infusion for 24 hours followed by 3 months of subcutaneous injections of insulin 4 times daily or usual care (602). With continuous insulin infusion, blood glucose decreased in the first 24 hours from 15.4 to 9.6 mmol/L in the infusion group versus 15.7 to 11.7 mmol/L in the control group (p less than 0.0001). There was a trend toward lower 30-day mortality and significantly lower 1-year mortality (18.6% versus 26.1%, p equals 0.027).

Compelling evidence for tight glucose control in intensive care unit patients (a large proportion of whom were there after cardiac surgery) supports the importance of intensive insulin therapy to achieve a normal blood glucose (80 to 110 mg/dL) in critically ill patients (603,603a). Van den Berghe et al. reported that 12-month mortality rates were reduced from 8.0% to 4.6% (p less than 0.04; n equals 1548) for critically ill patients assigned to intensive insulin therapy (604). Goldberg et al. reported successful implementation of a nursing protocol with an insulin infusion to achieve a target blood glucose of 100 to 139 mg/dL in an intensive care setting (605). The studies by Van den Berghe et al. and Goldberg et al. underscore the importance and feasibility of intensive infusion therapy in the intensive care setting. The precise target blood glucose range requires further study.

Management of diabetic patients with STEMI should also involve consideration of long-term hypoglycemic therapy. A review of the oral hypoglycemic therapy of type 2 diabetes mellitus indicated that with few exceptions, the available oral antidiabetic agents are equally effective in lowering glucose levels. Their mechanisms of action are different. As a result, they appear to have distinct metabolic effects that may influence their profile and affect cardiovascular risk (606). As suggested by Inuzuchi, in terms of hypoglycemic effect alone, there is no compelling reason to favor one of the major classes of oral antidiabetic agents (606). The overarching principle is that diabetic patients with STEMI should ultimately receive a regimen that achieves the best glycemic control, is well tolerated, and is likely to be maintained by the patient over the long term.

Although it is well appreciated that type I diabetic patients require insulin, most type 2 diabetics will also eventually need insulin to achieve the target of a HbA1C level less than 7%, a value that has been shown to be associated with reduced cardiovascular complications. It is reasonable that

the prescription for care of diabetics with STEMI be individualized, selected from an armamentarium of insulin, insulin analogs, and oral hypoglycemic agents alone or in combination (607). A popular combination is metformin with insulin because it results in similar metabolic control, less weight gain, lower insulin doses, and fewer hyperglycemic episodes than insulin alone or insulin plus sulfonylurea therapy (607). The use of metformin must be tempered with the knowledge that metformin is contraindicated in the presence of CHF and renal failure. It should be withheld for 48 hours after intravenous contrast injection (608).

6.3.1.6.9.3. Magnesium

Class IIa

- **1.** It is reasonable that documented magnesium deficits be corrected, especially in patients receiving diuretics before the onset of STEMI. (*Level of Evidence: C*)
- 2. It is reasonable that episodes of torsade de pointestype VT associated with a prolonged QT interval be treated with 1 to 2 grams of magnesium administered as an IV bolus over 5 minutes. (*Level of Evidence: C*)

Class III

In the absence of documented electrolyte deficits or torsade de pointes-type VT, routine intravenous magnesium should not be administered to STEMI patients at any level of risk. (*Level of Evidence: A*)

Meta-analyses of 7 randomized trials published between 1984 and 1991 suggested a significant mortality benefit of magnesium (4.4% absolute risk difference [ARD]; OR 0.44, CI 0.27 to 0.71) (609,610). The Second Leicester Intravenous Magnesium Intervention Trial (LIMIT-2) subsequently reported a significant reduction in mortality with magnesium treatment (2.5% ARD; 24% RRR; p equals 0.03) (611).

The ISIS-4 investigators enrolled 58 050 patients, of whom 29 011 were allocated to magnesium and 29 039 to control (152). There were 2216 deaths (7.64%) by 35 days in the magnesium group and 2103 deaths (7.24%) in the control group (OR 1.06; CI 0.99 to 1.13), which suggests no mortality benefit of magnesium administration and even the possibility of slight harm. Critiques of ISIS-4 raised the possibility that the null effect of magnesium resulted from late administration of treatment to patients who were predominantly at low risk (612,613).

The MAGIC (Magnesium in Coronaries) trial investigated the benefits of early administration of intravenous magnesium to high-risk patients with STEMI (stratum I: age 65 years or older and eligible for reperfusion therapy; stratum II: patients of any age who were not eligible for reperfusion therapy) (614). At 30 days, 475 (15.3%) patients in the magnesium group and 472 (15.2%) in the placebo group had died (OR 1.0, 95% CI 0.9 to 1.2, p equals 0.96). Potential explanations for the null effect of magnesium in MAGIC include the possibility that publication bias and an inadequate sample size in several earlier trials could have led to an overesti-

First Author.	Start of		Duration of		Early Mortality*		
Year (Reference)	Trial Treatment	Total Mg ²⁺ Dose	Treatment, h	Control	${ m Mg^{2+}}$	P Value	
Morton, 1984 (615)	Less than 8 h after pain	90 mmol/80 kg	36	2/36	1/40	NS	
Rasmussen, 1986 (616)	Less than 3 h after admission	62 mmol	48	23/135	9/135	Less than 0.01	
Smith, 1986					15/180	NIC	
(LIMIT-1) (617,618)	Less than 1 h after admission	65 mmol	24	23/185	1/48	SN	
Abraham, 1987 (619)	Soon after admission	30 mmol	48	1/46	1/25	SN	
Ceremuzynski, 1989 (620)	Less than 12 h after pain	32 mmol	24	3/23	1/59	Less than 0.01	
Shechter, 1990 (621)	Soon after admission	88 mmol	48	9/56	10/150	NS	
Feldstedt, 1991 (622)	Less than 8 h after pain	80 mmol	24	8/148	2011/00	2000 IV I	
Woods, 1992					0511/06	CU.U nan U.U.	
(LIMIT-2) (623)	Median time less than 3 h	72 mmol	24	119/1129	4/96	Less than 0.01	
Thogersen, 1993 (624)	Less than 1 h after admission	50 mmol	20	20/122	2216/29011	NS	
Shechter, 1995 (625)	Mean time 7 h after pain	92 mmol	48	17/98			
ISIS-4, 1995 (255)	Median time less than or equal to 3	80 mmol	24	2103/29039	6/169 2750	Less than 0.01	
	hours after fibrinolytic†				00/7	co.u nant con	
Raghu, 1999 (626)	Less than 7 h after pain	76 mmol	24	18/181	475/3110	SN	
Gyamlani, 2000 (627)	Less than 6 h after pain; less	62 mmol	48	10/50		2	
	than 2 h after admission						
MAGIC, 2002 (614)	Median 3.8 h	77 mmol	24	472/3098			
h = hours; NS = not significant. *Variously defined less than or equal to ? †Median 12 hours from onset in those nc Modified with permission from Woods a	35 days. ot given fibrinolytic (30%). ınd Abrams. Progress in Cardiovascular Diseases 200	2;44:267-74 (628).		2			

fai dial In f Ma2+ in My P 5

mation of the benefit of magnesium through a large type I error and that the combination of mechanisms proposed for the benefits of magnesium overlapped with and were superseded by aspirin, beta-blockers, and ACE inhibitors (prescribed infrequently in earlier trials but commonly in ISIS-4 and MAGIC).

Between 1980 and 2002, a total of 68 684 patients were studied in a series of 15 randomized trials (Table 22) (614-628). On the basis of the totality of available evidence, in current coronary care practice, there is no indication for the routine administration of intravenous magnesium to patients with STEMI at any level of risk. Magnesium can continue to be administered for repletion of documented electrolyte deficits and life-threatening ventricular arrhythmias such as torsade de pointes (629).

6.3.1.6.9.4. Calcium Channel Blockers

Class IIa

It is reasonable to give verapamil or diltiazem to patients in whom beta-blockers are ineffective or contraindicated (e.g., bronchospastic disease) for relief of ongoing ischemia or control of a rapid ventricular response with AF or atrial flutter after STEMI in the absence of CHF, LV dysfunction, or AV block. (Level of Evidence: C)

Class III

- 1. Diltiazem and verapamil are contraindicated in patients with STEMI and associated systolic LV dys-function and CHF. (Level of Evidence: A)
- 2. Nifedipine (immediate-release form) is contraindicated in the treatment of STEMI because of the reflex sympathetic activation, tachycardia, and hypotension associated with its use. (*Level of Evidence: B*)

Nifedipine. In patients with STEMI, immediate-release nifedipine does not reduce the incidence of reinfarction or mortality when given early (less than 24 hours) or late. Immediate-release nifedipine may be particularly detrimental in patients with hypotension or tachycardia; in these patients, it may induce a reduction in coronary perfusion pressure, disproportionate dilatation of the coronary arteries adjacent to the ischemic area (so-called "steal"), and/or reflex activation of the sympathetic nervous system, with an increase in myocardial oxygen demands (630-637).

Verapamil. Although the overall results of trials with verapamil showed no mortality benefits, subgroup analysis showed that immediate-release verapamil initiated several days after STEMI in patients who were not candidates for a beta-blocking agent may have been useful in reducing the incidence of the composite end point of reinfarction and death, provided LV function was well preserved with no clinical evidence of heart failure. Verapamil is detrimental to patients with heart failure or bradyarrhythmias during the first 24 to 48 hours after STEMI (638-641). One randomized study of 1700 patients less than 75 years of age using verapamil within 2 weeks of STEMI showed a significant reduction in major

events (death or reinfarction) over 18 months (3.6% ARD; 17% RRR; p equals 0.03) (642).

Diltiazem. Data from the Multicenter Diltiazem Postinfarction Trial (MDPIT; Q-wave and non-Q-wave infarction) (643) and the Diltiazem Reinfarction Study (DRS; non-Q-wave infarction) (639,640,644,645) suggest that patients with non-O-wave MI or those with O-wave infarction, preserved LV function, and no evidence of heart failure may benefit from immediate-release diltiazem. Diltiazem was begun in MDPIT 3 to 15 days after STEMI and in DRS 24 to 72 hours afterward. The results of MDPIT may be confounded by the fact that 53% and 55% of placebo- and diltiazem-treated patients, respectively, received concomitant beta-blocker therapy (643). Also, both the MDPIT and DRS projects were conducted in an era when the use of aspirin was not as prevalent as it is today, which raises further uncertainty about the relevance of their findings for contemporary management of STEMI. Of particular clinical importance is the detrimental mortality effect of diltiazem in patients with LV dysfunction.

Diltiazem was tested in patients with STEMI but without CHF who were undergoing fibrinolytic therapy in the INTERCEPT trial (Incomplete Infarction Trial of European Research Collaborators Evaluating Prognosis post-Thrombolysis) (646). No effect on the cumulative occurrence of fatal and nonfatal end points was demonstrated during a 6-month follow-up, but there was a modest decrease in nonfatal cardiac events, in large part due to reductions in recurrent ischemia.

7. HOSPITAL MANAGEMENT

7.1. Location

7.1.1. Coronary Care Unit

Class I

- **1.** STEMI patients should be admitted to a quiet and comfortable environment that provides for continuous monitoring of the ECG and pulse oximetry and has ready access to facilities for hemodynamic monitoring and defibrillation. (Level of Evidence: C)
- 2. The patient's medication regimen should be reviewed to confirm the administration of aspirin and betablockers in an adequate dose to control heart rate and to assess the need for intravenous nitroglycerin for control of angina, hypertension, or heart failure. (Level of Evidence: A)
- 3. The ongoing need for supplemental oxygen should be assessed by monitoring arterial oxygen saturation. When stable for 6 hours, the patient should be reassessed for oxygen need (i.e., O_2 saturation of less than 90%), and discontinuation of supplemental oxygen should be considered. (Level of Evidence: C)
- 4. Nursing care should be provided by individuals certified in critical care, with staffing based on the specific needs of patients and provider competencies, as well as organizational priorities. (*Level of Evidence: C*)

- 5. Care of STEMI patients in the coronary care unit (CCU) should be structured around protocols derived from practice guidelines. (*Level of Evidence: C*)
- 6. Electrocardiographic monitoring leads should be based on the location and rhythm to optimize detection of ST deviation, axis shift, conduction defects, and dysrhythmias. (Level of Evidence: B)

Class III

It is not an effective use of the CCU environment to admit terminally ill, "do not resuscitate" patients with STEMI, because clinical and comfort needs can be provided outside of a critical care environment. (*Level* of Evidence: C)

Treatment of the patient with STEMI begins in the EMS system/ED/catheterization laboratory and is consolidated in the CCU. On arrival at the CCU, initial patient evaluation includes assessment of vital signs, pulse oximetry, cardiac rhythm and ST segments, and symptoms of acute cardiac ischemia. Research has also shown the importance of assessing anxiety and depression, because outcomes are worse in patients with even moderate elevations in dysphoria. Outstanding (and abnormal) laboratory results should be followed up, and standard orders to CCU should be implemented. All CCUs should have the equipment and personnel necessary to monitor intra-arterial pressure and pulmonary artery catheter pressures (Swan-Ganz catheter). Such monitoring is useful for severely hypotensive patients. An IABP should be available in tertiary care CCUs for treatment of cardiogenic shock. The patient's medication orders should be reviewed to confirm the administration of aspirin and betablockers in an adequate dose to control heart rate and to assess the need for intravenous nitroglycerin for control of angina, hypertension, or acute heart failure.

Ideally, all CCU nursing staff should have certification in critical care nursing (CCRN, Critical Care Registered Nurse) as endorsed by the American Association of Critical-Care Nurses (647). Advanced cardiac life support certification (ACLS) is highly recommended. The need for ACLS certification is usually determined by institutional policy.

There has been documentation of the positive correlation between nurse staffing levels and patient outcome in intensive care units (648-650). The nurse-to-patient ratio has varied depending on nurse availability and the needs of the patient, e.g., 1 nurse per patient for an intubated patient, to 1:2 (651). In 1999, the state of California mandated a ratio of 1:2 in the intensive care unit/CCU, 1:4 in the stepdown unit, and 1:5 in the telemetry unit (652). A study of the effect of the California state law showed a continued wide variation of staffing ratios among hospitals (653). The American Association of Critical Care Nurses has issued a policy statement that staffing decisions in the critical care setting should optimally be based on the specific needs of patients and provider competencies, as well as organizational priorities. Within this framework, staffing should reflect the number and type of staff that meet a group of patients' needs instead of a mandated single staffing ratio or mix (654).

Patients with STEMI may experience heart failure, serious arrhythmias, or recurrent ischemia (655). Accordingly, nursing-related care is commonly prescribed as part of a standing order (i.e., measurement of height and weight on admission and thereafter daily weight, especially if heart failure is present; routine post-STEMI vital signs, including oxygen saturation; ECG monitoring; and activity level such as bathroom privileges and progressive cardiac rehabilitation). Medications such as stool softeners or antianxiety agents should be given based on nursing judgment.

The current medical-legal-economic climate demands that CCUs be operated at peak efficiency, optimizing quality of care and minimizing complications. Practical and ethical dilemmas present the CCU director with issues of futile care, triage, and the administration of appropriate care (656). It is generally accepted that terminal patients (no-code) or patients whose comorbidities make survival unlikely should not be admitted to a CCU. Similarly, patients with improving CHF or stable dysrhythmia should be transferred to a non-CCU, monitored bed.

7.1.1.1. Monitoring and Treatment for Adverse Events

Early general measures focus on monitoring for adverse events, preventing such events through protective measures, and treating these adverse events when they do occur. Electrocardiographic monitoring is an essential role of CCU staff, who must be adept at rhythm interpretation, lead selection based on infarct location and rhythm, and lead placement for detection of RV involvement (657-660). Computer algorithms have proved superior to medical personnel for detection of arrhythmias (661). Accurate and consistent lead placement and careful electrode and skin preparation are important to improve the clinical usefulness of ST monitoring (662).

Nurses should monitor the ST segment for ischemia, particularly during routine morning care, because patients have been reported to have a greater likelihood of ischemic events between 6 AM and noon than at other times (663). Currently, ECG monitors operate with computerized arrhythmia analysis alone or with both arrhythmia and ischemia analysis. STsegment monitoring is generally underutilized in American hospitals.

Because changes in the ST segment can shift among various ECG leads in the same person over time owing to different ischemic mechanisms, a consensus statement on ST monitoring has recommended that 12-lead monitoring be done (662). Patients with acute coronary syndromes, including STEMI, are the highest priority for ST-segment monitoring. It is recommended they be monitored for a minimum of 24 hours and until they remain event-free for 12 to 24 hours. Potential benefits in patients with STEMI include the ability to assess patency of the culprit artery after fibrinolytic therapy (664-668); detect abrupt reocclusion after PCI (669); detect ongoing ischemia (i.e., failed reperfusion therapy), recurrent ischemia, and infarct extension; and detect transient myocardial ischemia. Blood pressure should be measured repeatedly; actual frequency will depend on the severity of the illness. Although invasive arterial monitoring is preferred in the hypotensive patient, noninvasive monitoring is adequate for most patients. Monitoring with an automatic device that inflates and deflates at programmed intervals is useful, but it must be recognized that measurements may be inaccurate because of inappropriate cuff size or muscle contractions; marked peripheral vasoconstriction can result in falsely low readings. Furthermore, many patients report that the device is irritating and disrupts rest.

7.1.2. Stepdown Unit

Class I

- **1.** It is a useful triage strategy to admit low-risk STEMI patients who have undergone successful PCI directly to the stepdown unit for post-PCI care rather than to the CCU. (*Level of Evidence: C*)
- 2. STEMI patients originally admitted to the CCU who demonstrate 12 to 24 hours of clinical stability (absence of recurrent ischemia, heart failure, or hemodynamically compromising dysrhythmias) should be transferred to the stepdown unit. (Level of Evidence: C)

Class IIa

- **1.** It is reasonable for patients recovering from STEMI who have clinically symptomatic heart failure to be managed on the stepdown unit, provided that facilities for continuous monitoring of pulse oximetry and appropriately skilled nurses are available. *(Level of Evidence: C)*
- 2. It is reasonable for patients recovering from STEMI who have arrhythmias that are hemodynamically well tolerated (e.g., AF with a controlled ventricular response; paroxysms of nonsustained VT lasting less than 30 seconds) to be managed on the stepdown unit, provided that facilities for continuous monitoring of the ECG, defibrillators, and appropriately skilled nurses are available. (Level of Evidence: C)
- **Class IIb**

Patients recovering from STEMI who have clinically significant pulmonary disease requiring high-flow supplemental oxygen or noninvasive mask ventilation/bilevel positive airway pressure/continuous positive airway pressure may be considered for care on a stepdown unit provided that facilities for continuous monitoring of pulse oximetry and appropriately skilled nurses with a sufficient nurse:patient ratio are available. (Level of Evidence: C)

Although the CCU was traditionally the hospital location to which patients with STEMI were first admitted, increasing use of catheter-based reperfusion and increasing sophistication of monitoring equipment and staff experience has resulted in a shift toward admitting patients with low-risk STEMI who have undergone successful reperfusion with PCI directly to a stepdown unit. In addition, patients originally admitted to the CCU who demonstrate 12 to 24 hours of clinical stability are typically transferred to the stepdown unit. The same nurse staffing and certification considerations apply to the stepdown unit (coronary observation unit, telemetry unit) as described for the CCU to ensure optimal evaluation and response to any deterioration of the patient with STEMI. Pulse oximetry and ECG monitoring and defibrillation equipment should be available. Optimally, the nursing staff should have a skill set similar to CCU nurses so that they may evaluate and respond to any deterioration of a patient with STEMI. The initial evaluation of patients with STEMI who are admitted directly to the stepdown unit is similar to that described in Section 7.1.1 for the CCU.

7.2. Early, General Measures

7.2.1. Level of Activity

Class IIa

After 12 to 24 hours, it is reasonable to allow patients with hemodynamic instability or continued ischemia to have bedside commode privileges. (Level of Evidence: C)

Class III

Patients with STEMI who are free of recurrent ischemic discomfort, symptoms of heart failure, or serious disturbances of heart rhythm should not be on bed rest for more than 12 to 24 hours. (Level of Evidence: C)

Limiting early physical exertion and minimizing sympathetic stimulation (e.g., acute ischemic-type chest discomfort and anxiety) are methods of minimizing myocardial oxygen demand (670). In an earlier era, the duration of bed rest was extended to several weeks, until it was ascertained that prolonged immobility is harmful because of the physiological deconditioning that occurs after even 6 hours in the supine position (671). Preload decreases because of plasma volume losses that occur early in the bedrest period. Shifts in ventricular filling activate the body's compensatory mechanisms to buffer pressure and volume alterations.

The current literature suggests that the deconditioning effects of bedrest are independent of the patient's clinical condition but rather are associated with the absence of regular exposure to "orthostatic stress" (produced by assumption of an upright posture). The absence of habitual exposure to the upright posture occurs naturally with prolonged periods of bedrest and deprives the cardiovascular system of stimulation needed to maintain adequate blood pressure regulation. As a result, these patients commonly develop orthostatic hypotension or frank syncope during their initial ambulation attempt in the in-patient setting. Studies have found that intermittent but regular exposure to sitting or standing during convalescence can counteract these deconditioning effects (672).

Table 23. Sample Admitting Orders for the Patient With STEMI

1. Condition: Serious

- 2. IV: NS or D_5W to keep vein open. Start a second IV if IV medication is being given. This may be a heparin lock.
- 3. **Vital signs:** every 30 min until stable, then every 4 h as needed. Notify physician if HR is less than 60 bpm or greater than 100 bpm, systolic BP is less than 100 mm Hg systolic or greater than 150 mm Hg, respiratory rate is less than 8 breaths per minute or greater than 22 breaths per minute.
- 4. Monitor: Continuous ECG monitoring for arrhythmia and ST-segment deviation.
- 5. Diet: NPO except for sips of water until stable. Then start 2 grams sodium/d, low saturated fat (less than 7% total calories/d), low cholesterol (less than 200 mg/d) diet, such as Therapeutic Lifestyle Changes (TLC) diet.
- 6. Activity: Bedrest and bedside commode and light activity when stable.
- 7. **Oxygen:** Continuous oximetry monitoring. Nasal cannula at 2 L/min when stable for 6 h, reassess for oxygen need (i.e., O₂ saturation less than 90%), and consider discontinuing oxygen.

8. Medications:

- a. Nitroglycerin (See Section 6.3.1.2 for further discussion.)
 - 1. Use sublingual NTG 0.4 mg every 5 min as needed for chest discomfort.
 - 2. Intravenous NTG for CHF, hypertension, or persistent ischemia.
- b. Aspirin (See Section 6.3.1.4.)
 - 1. If aspirin not given in the ED, chew non-enteric-coated aspirin[†] 162 to 325 mg.
 - 2. If aspirin has been given, start daily maintenance of 75 to 162 mg. May use enteric-coated for gastrointestinal protection.
- c. Beta-Blocker (See Section 6.3.1.5.)
 - 1. If not given in the ED, assess for contraindications, i.e., bradycardia and hypotension. Continue daily assessment to ascertain eligibility for beta-blocker.
 - 2. If given in the ED, continue daily dose and optimize as dictated by HR and BP.
- d. ACE Inhibitor (See Section 6.3.1.6.9.1.)

1. Start ACE inhibitor orally in patients with anterior infarction, pulmonary congestion, or LVEF less than 0.40 if the following are absent: hypotension (SBP less than 100 mm Hg or less than 30 mm Hg below baseline) or known contraindications to this class of medications.

e. Angiotensin Receptor Blocker (See Section 6.3.1.6.9.1.)

1. Start ARB orally in patients who are intolerant of ACE inhibitors and who have either clinical or radiological signs of heart failure or LVEF less than 0.40.

f. Pain Medications (See Section 6.3.1.3.)

2. IV morphine sulfate 2 to 4 mg with increments of 2 to 8 mg IV at 5- to 15-minute intervals as needed to control pain.

- g. Anxiolytics (based on a nursing assessment) (See Section 7.2.4.)
- h. Daily Stool Softener
- 9. Laboratory Tests: Serum biomarkers for cardiac damage,*CBC with platelet count, INR, aPTT, electrolytes, magnesium, BUN, creatinine, glucose, serum lipids (see Table 9).

*Do not wait for results before implementing reperfusion strategy.

†Although some trials have used enteric-coated aspirin for initial dosing, more rapid buccal absorption occurs with non-enteric-coated formulations.

Modified with permission from Ryan et al. J Am Coll Cardiol 1999;34:890-911 (3).

The 1996 ACC/AHA Guidelines for the Management of Patients with Acute MI cited evidence of the continued practice of "coronary precautions" that were advocated in the 1960s (2), and specific directions were given about what were or were no longer considered to be coronary precautions. It is assumed that current practice has advanced sufficiently such that patients with STEMI are no longer kept on bedrest or fed by nurses. Avoidance of the Valsalva maneuver remains an important consideration (673,674), especially in younger patients (e.g., under the age of 45 years) (673-676). Otherwise, current practice routinely entails a short period of bedrest (except for patients who have recurrent ischemic discomfort, symptoms of heart failure, or serious rhythm disturbances). Low-level activities such as toileting, assisted bathing, and light ambulation can prevent physiological deconditioning (673-676). Sample admitting orders reflecting the standard of care are presented in Table 23 (3).

7.2.2. Diet

Class I

1. Patients with STEMI should be prescribed the NCEP Adult Treatment Panel III (ATP III) Therapeutic Lifestyle Changes (TLC) diet, which focuses on reduced intake of fats and cholesterol, less than 7% of total calories as saturated fats, less than 200 mg of cholesterol per day, increased consumption of omega-3 fatty acids, and appropriate caloric intake for energy needs. (Level of Evidence: C)

STEMI = ST-elevation myocardial infarction; IV = intravenous; NS = normal saline; h = hours; bpm = beats per minute; ECG = electrocardiogram; NPO = nothing by mouth; min = minutes; NTG = nitroglycerin; CHF = congestive heart failure; ED = emergency department; HR = heart rate; BP = blood pressure; ACE = angiotensin converting enzyme; LVEF = left ventricular ejection fraction; SBP = systolic blood pressure; ARB = angiotensin receptor blocker; CBC = complete blood count; INR = international normalized ratio; aPTT = activated partial thromboplastin time; BUN = blood urea nitrogen.

- 2. Diabetic patients with STEMI should have an appropriate food group balance and caloric intake. (Level of Evidence: B)
- **3.** Sodium intake should be restricted in STEMI patients with hypertension or heart failure. (*Level of Evidence: B*)

The NCEP ATP III has recommended that a complete blood lipid profile be taken in all patients with established CHD (50). In the STEMI patient, this should be done at the time of admission or within 24 hours of the onset of symptoms, because low-density lipoprotein cholesterol (LDL-C) levels begin to decrease soon after an event and are significantly reduced by 48 hours and remain so for many weeks. Thus, LDL-C measurements several days after STEMI may not be representative of the patient's average LDL-C. In the context of possible early cardiac catheterization, it would be helpful for patients to have been on a clear liquid diet that extended to a period of "nothing by mouth" (NPO) before the procedure. Patients with STEMI should receive a reduced saturated fat and cholesterol diet per the ATP III TLC approach (677). (See Section 7.12.2.) Diabetes experts no longer recommend a single meal plan for all people with diabetes. Instead, they recommend meal plans that are flexible and take into account a person's lifestyle and particular health needs, ideally with consultation from a registered dietitian to design a meal plan. Sodium intake should be restricted in patients with STEMI with hypertension or heart failure to a maximum of 2000 mg/d (Table 23) (3).

Blood pressure increases after caffeine intake (678), but the increase is not clinically significant until 400 mg of caffeine (i.e., 2 to 4 cups of coffee, depending on strength and brewing method) is ingested (679,680). Moderately high doses (450 mg) of caffeine did not increase ventricular arrhythmias in a small group of patients with ischemic heart disease (681). People who drink caffeinated beverages regularly develop a tolerance after 1 to 4 days (682,683), regardless of dose. Withdrawal of caffeine is associated with headache (684,685) and increases in heart rate (686). The available evidence suggests that patients with STEMI who are routine caffeine drinkers be allowed to consume up to 4 to 5 cups of caffeinated coffee a day while in the CCU or progressive care unit, under the surveillance of nursing staff (680,687). As a practical matter in the acute care setting, 1 to 2 cups of coffee, enough to avert caffeine withdrawal, seems appropriate.

7.2.3. Patient Education in the Hospital Setting

Class I

1. Patient counseling to maximize adherence to evidence-based post-STEMI treatments (e.g., compliance with taking medication, exercise prescription, and smoking cessation) should begin during the early phase of hospitalization, occur intensively at discharge, and continue at follow-up visits with providers and through cardiac rehabilitation programs and community support groups, as appropriate. (Level of Evidence: C)

2. Critical pathways and protocols and other qualityimprovement tools (e.g., the ACC's "Guidelines Applied in Practice" and the AHA's "Get with the Guidelines") should be used to improve the application of evidence-based treatments by patients with STEMI, caregivers, and institutions. (Level of Evidence: C)

Patient education should be viewed as a continuous process that should be part of every patient encounter (i.e., on hospital arrival, at inpatient admission, at discharge, and at followup visits). On admission to the coronary care unit, patients should receive an orientation to their surroundings (e.g., location of the call light), an explanation of the equipment used for their care (e.g., electrodes and cardiac monitor, IV line/heparin lock, central line, nasal cannula, and pulse oximetry), nursing care routines and their "round-the-clock" nature, level of activity permitted, an explanation of the pain assessment scale that will be used, and the importance of reporting any symptoms.

In general, effective education involves the use of a combination of good communication skills, patient assessment of prior knowledge and readiness to learn, and effective teaching strategies. One-to-one teaching is the most common patient education method in the inpatient setting and is preferred by patients as well (688). Ideally, family members should be present to hear what patients are being told so they can reinforce the information later (689).

For more complex topics, the most effective educational intervention involves being responsive to the patient's attempts to communicate (e.g., not changing the topic abruptly or engaging in tasks unrelated to the conversation); using good eye contact; not denying the patient's feelings (e.g., "You should not worry about that!"); and being aware of one's own nonverbal cues that encourage or discourage communication (e.g., failure to sit down), as well as those of the patient's.

Inpatient education has been reported to stimulate some lifestyle change after discharge, most frequently in the areas of activity and smoking cessation (690). However, post-MI knowledge alone does not ensure behavioral change (691,692). Providers should be aware that many patients who have had a STEMI may be in the early stages of behavioral change, such as "precontemplation," where they do not yet see a need to change behavior. Others may be in the "contemplation" stage, where, for example, the experience of STEMI may cause them to acknowledge that they need and want to make a change, but that thought is not yet translated into action (691-694).

With 1-to-1 teaching in the hospital, the provider should strike a balance between not using any teaching aids when educating the patient and simply handing out written information or other media without discussion. One-to-one teaching is effectively enhanced and reinforced with the use of appropriate media (688,695).

Table 24. Milestones and Recommended Information for Educating the Patient With ST-Elevation Myocardial Infarction

Following is the range of information providers should address with patients with STEMI and their family members: Before the event-(ongoing by primary care provider, especially with high-risk patients) Assess and manage cardiac risk factors Review recommendations for recognizing and responding to heart attack symptoms (See: http://www.nhlbi.nih.gov/health/prof/heart/ mi/provider.htm) At hospital/ED admission—(day of admission) Explain diagnosis Review plan for inpatient treatment and projected length of stay Inpatient testing/procedures-(day of admission until discharge) Explain the purpose of tests and procedures that are ordered Describe what to expect with tests and procedures: duration; level of discomfort and invasiveness; sensory information Cardiac care unit/stepdown unit-(day of admission) Orient to surroundings/routine of unit Explain nursing care plan Describe the importance of reporting symptoms, needs At discharge—(day of discharge or before) Review risk factor goals and management plan Review prescribed medications and lifestyle changes/recommendations Review information on recognizing and responding to heart attack symptoms Recommend that family member(s) attend a CPR training program and cardiac support group Refer patient to cardiac rehabilitation program Schedule a follow-up appointment with primary care provider Discuss plans for obtaining prescribed medication that day (immediately after discharge) At follow-up visits with primary care provider—(first follow-up appointment and ongoing) Review diagnosis with patient and hospital course/outcome Review medical and lifestyle regimens prescribed Ensure aggressive risk factor modification and follow-up Discuss recognition and response to acute symptoms; review action plan, including taking nitroglycerin in response to acute symptoms if prescribed, and calling 9-1-1

Assess for depression, other psychosocial responses

Suggested milestones for educating patients over the spectrum of cardiac care, both before and after the STEMI event. Risk factor assessment and management and counseling about recognition and response to heart attack symptoms should be done by the primary care provider. Once admitted for STEMI, the provider should review with the patient and family members the diagnosis, treatment plan, and projected length of stay; purpose and what to expect of tests and procedures that are ordered; the coronary care unit (or equivalent) environment and nursing care plan; and at discharge, the prescribed medications and lifestyle changes, follow-up testing, and referral.

STEMI = ST-elevation myocardial infarction; ED = emergency department; CPR = cardiopulmonary resuscitation.

Challenges to patient education in the inpatient setting are shorter lengths of hospital stay (696); older, sicker patients with more psychosocial issues; cultural and literacy barriers (697); and patient anxiety (698). Inpatient education has been shown to reduce anxiety (699,700), improve knowledge (688), and decrease length of hospital stay (700,701).

In standardizing care for these patients, the use of guidelinebased tools has been reported to facilitate improvement in the quality of care for patients with STEMI among a variety of institutions, patients, and caregivers (5). Programs such as the AHA's "Get With the Guidelines" (see the "Get With the Guidelines' Hospital Tool Kit" at: http://www.americanheart.org/downloadable/heart/1107_HospTool.pdf) and the ACC's "Guidelines Applied in Practice" (see sample forms at: http://www.acc.org/gap/mi/ami_permissionprocess.htm) offer tools such as educational plans for patients explaining what they can expect over the course of their hospitalization (e.g., procedures and the treatment plan), care paths, standing orders for providers, discharge protocols that incorporate evidencebased recommendations (for providers and patients), and a data-based patient management tool (702). Support groups provided by some hospitals and cardiac rehabilitation programs may help patients adjust to their diagnosis and newly prescribed lifestyle and medication regimens. For example, Mended Hearts is a national nonprofit organization affiliated with the AHA that offers visiting programs, support group meetings, and educational forums through partnerships with hospitals and rehabilitation clinics. The organization is particularly interested in helping patients deal with the emotional recovery from heart disease through facilitating a positive patient-care experience for heart disease patients, their families, caregivers, and others impacted by heart disease (703). Table 24 shows suggested topics for educating the patient with STEMI.

7.2.4. Analgesia/Anxiolytics

Class IIa

1. It is reasonable to use anxiolytic medications in STEMI patients to alleviate short-term anxiety or altered behavior related to hospitalization for STEMI. (Level of Evidence: C)

2. It is reasonable to routinely assess the patient's anxiety level and manage it with behavioral interventions and referral for counseling. (*Level of Evidence: C*)

It is useful to monitor patients for increased anxiety or altered behavior of the patient in the CCU. Anxiolytics can play an important role in patient management in this setting. Treatment with benzodiazepines should be limited to the minimal dose for a limited period of time (261).

Hospitalized smokers may experience symptoms of nicotine withdrawal, including anxiety, insomnia, depression, difficulty concentrating, irritability, anger, restlessness, and slowed heart rate (704). Patients experiencing nicotine withdrawal can benefit from anxiolytics. Use of bupropion and nicotine replacement therapy in the acute setting should also be considered as options, depending on the severity of the patient's withdrawal syndrome. Agitation and delirium are not uncommon in the CCU, particularly in patients with complicated STEMI and protracted stays in the intensive care setting. In addition, a number of medications used in the CCU, such as lidocaine, mexiletine, procainamide, atropine, cimetidine, and meperidine, can induce delirium. Intravenous haloperidol is a rapidly acting neuroleptic that can be given safely and effectively to cardiac patients with agitation. It rarely produces hypotension or need for assisted ventilation. If patients exhibit altered sensorium and have received fibrinolytics, consideration should be given to ordering a computed axial tomography/magnetic resonance imaging scan to rule out ICH before sedating the patient.

Anxiety and depression are prevalent in patients hospitalized for STEMI because patients are confronted with a diagnosis that is major, both psychologically and physically (705,706). In addition, the experience of a cardiac event is a significant source of stress for family members trying to adjust to the initial diagnosis and confront the uncertainties associated with hospitalization and the initial recovery phase. Anxiety has been demonstrated to predict inhospital recurrent ischemia and arrhythmias (707) and cardiac events during the first year after an MI (708). Physicians' and nurses' subjective judgments of patient anxiety are not accurate when compared with measurements of anxiety on validated scales (709,710).

The provision of information, discussed later in this guideline, and liberal visiting policies can also help patients with STEMI feel more in control (711,712). In addition, psychological support and counseling during hospitalization can decrease anxiety and depression immediately and for up to 6 months after STEMI (713,714). At least 1 randomized controlled trial demonstrated that in-hospital anxiety and depression could be reduced by a structured nursing support intervention (714).

Liberalized visiting rules for patients in critical care can be helpful; several studies have demonstrated no harmful physiological effects attributable to unrestricted visiting policies (711,712). Patients whose anxiety is very severe or persistent in spite of medications should be referred for consultation for formal anxiety assessment and treatment. Consultation could be obtained from a nurse specialist, psychiatric social worker, or a psychiatrist.

7.3. Risk Stratification During Early Hospital Course

Several groups have proposed risk stratification scores for patients with suspected STEMI based on their admission characteristics (241,242,394). Risk stratification is a continuous process and requires the updating of initial assessments with data obtained during the hospital stay. Indicators of failed reperfusion (e.g., recurrence of chest pain, persistence of ECG findings indicating infarction) identify a patient who should undergo coronary angiography. Similarly, findings consistent with mechanical complications (e.g., sudden onset



Figure 27. Summary of data from meta-analysis of trials of beta-blocker therapy from the prefibrinolytic era in patients with myocardial infarction. No. = number; RR = relative risk; CI = confidence interval. Reprinted with permission from Antman and Braunwald. Acute Myocardial Infarction. In: Braunwald, Zipes, Libby, eds. Heart Disease: A Textbook of Cardiovascular Medicine, 6th ed., Philadelphia, PA: W.B. Saunders; 2001:1168 (718).

of heart failure, presence of a new murmur) herald increased risk and suggest the need for rapid intervention. For patients who did not undergo primary reperfusion, changes in clinical status (e.g., development of shock) may herald a worsening clinical status and are an indication for coronary angiography.

Patients with a low risk of complications may be candidates for early discharge. The lowest-risk patients are those who did not have STEMI, despite the initial suspicions. Clinicians should strive to identify such patients within 8 to 12 hours of onset of symptoms. Serial sampling of serum cardiac biomarkers and use of 12-lead ECGs and their interpretation in the context of the number of hours that have elapsed since onset of the patient's symptoms can determine the presence of STEMI better than adherence to a rigid protocol that requires that a specified number of samples be drawn in the hospital. Among those with STEMI treated with reperfusion, it has been suggested that an uncomplicated course after 72 hours of hospitalization identifies a group with a low enough risk for discharge (715). Newby and colleagues calculated that extending the hospital stay of these patients by another day would cost \$105 629 per year of life saved (715). Concerns have been raised that shortening the length of stay to 72 hours may adversely affect patient education regarding STEMI and the identification of the optimum dose of critical medications such as beta-blockers and ACE inhibitors (696). Work on transitional care has shown the effectiveness of an advanced practice nurse intervention in helping patients, many with ischemic heart disease, during the transition from hospital to home and in reducing costs in the process (716).

7.4. Medication Assessment

7.4.1. Beta-Blockers

Class I

- 1. Patients receiving beta-blockers within the first 24 hours of STEMI without adverse effects should continue to receive them during the early convalescent phase of STEMI. (Level of Evidence: A)
- 2. Patients without contraindications to beta-blockers who did not receive them within the first 24 hours after STEMI should have them started in the early convalescent phase. (Level of Evidence: A)
- 3. Patients with early contraindications within the first 24 hours of STEMI should be re-evaluated for candidacy for beta-blocker therapy. (Level of Evidence: C)

There is overwhelming evidence for the benefits of early beta-blockade in patients with STEMI and without contraindications to their use (see Section 6.3.1.5). Benefits have been demonstrated for patients with and without concomitant fibrinolytic therapy, both early and late after STEMI. Metaanalysis of trials from the prefibrinolytic era involving more than 24 000 patients receiving beta-blockers in the convalescent phase showed a 14% RRR in mortality through 7 days and a 23% RRR in long-term mortality (717). These data are summarized in Figure 27 (718).

Beta-blockers should be initiated early in the course of STEMI and continued unless adverse effects have been observed. In appropriately selected patients, these benefits occur at a risk of approximately a 3% incidence of provocation of CHF or complete heart block and a 2% incidence of the development of cardiogenic shock. Beta-blockers are especially beneficial in patients in whom STEMI is complicated by persistent or recurrent ischemia, evidence for infarct extension, or tachyarrhythmias.

Some clinicians do not start beta-blockers in the emergency phase of STEMI management because of error, late presentation, concern about relative or absolute contraindications, or concern about the benefits of beta-blockade in the face of other contemporary treatments. Benefits of beta-blockers initiated in the convalescent phase for secondary prevention of ischemic events are established (717). The Beta-blocker Heart Attack Trial (BHAT) (719) and the more contemporary CAPRICORN trial (273) confirm the substantial benefit of beta-blockers in addition to ACE inhibitor therapy in patients with transient or sustained postinfarction LV dysfunction. A reasonable general rule is to initiate beta-blockade after 24 to 48 hours of freedom from a relative contraindication, such as bradycardia, mild-to-moderate heart failure, or first-degree heart block.

7.4.2. Nitroglycerin

Class I

- 1. Intravenous nitroglycerin is indicated in the first 48 hours after STEMI for treatment of persistent ischemia, CHF, or hypertension. The decision to administer intravenous nitroglycerin and the dose used should take into account that it should not preclude therapy with other proven mortality-reducing interventions such as beta-blockers or ACE inhibitors. (Level of Evidence: B)
- 2. Intravenous, oral, or topical nitrates are useful beyond the first 48 hours after STEMI for treatment of recurrent angina or persistent CHF if their use does not preclude therapy with beta-blockers or ACE inhibitors. (Level of Evidence: B)

Class IIb

The continued use of nitrate therapy beyond the first 24 to 48 hours in the absence of continued or recurrent angina or CHF may be helpful, although the benefit is likely to be small and is not well established in contemporary practice. (*Level of Evidence: B*)

Class III

Nitrates should not be administered to patients with systolic blood pressure less than 90 mm Hg or greater than or equal to 30 mm Hg below baseline, severe bradycardia (less than 50 bpm), tachycardia (more than 100 bpm), or RV infarction. (Level of Evidence: C)

The use of nitrates in the patient with STEMI on presentation and in the early phase of infarction is reviewed in Section 6.3.1.2. Clinical trials have suggested only a modest benefit of nitroglycerin used acutely in STEMI and continued subsequently (152,586). Nitrates are most clearly indicated for persistent or recurrent ischemia and in patients with CHF. Because protocols in large clinical trials included acute followed by sustained therapy, there is little evidence to establish the duration of nitrate therapy after STEMI. Common clinical practice is to continue nitrate therapy for 24 to 48 hours. In view of their marginal treatment benefits, nitrates should not be used if hypotension limits the administration of beta-blockers or ACE inhibitors, which have more powerful benefits for both early use and secondary prevention after STEMI.

Nitrate tolerance develops with prolonged continuous exposure to nitrates, presumably through the mechanism of depletion of sulfhydryl groups in the vessel wall. If sustained therapy with nitrates is planned, intravenous nitrate therapy is usually changed to oral or topical preparations with a nitrate-free interval.

7.4.3. Inhibition of the Renin-Angiotensin-Aldosterone System

Class I

- 1. An ACE inhibitor should be administered orally during convalescence from STEMI in patients who tolerate this class of medication, and it should be continued over the long term. (*Level of Evidence: A*)
- 2. An ARB should be administered to STEMI patients who are intolerant of ACE inhibitors and have either clinical or radiological signs of heart failure or LVEF less than 0.40. Valsartan and candesartan have demonstrated efficacy for this recommendation. (Level of Evidence: B)
- 3. Long-term aldosterone blockade should be prescribed for post-STEMI patients without significant renal dysfunction (creatinine should be less than or equal to 2.5 mg/dL in men and less than or equal to 2.0 mg/dL in women) or hyperkalemia (potassium should be less than or equal to 5.0 mEq/L) who are already receiving therapeutic doses of an ACE inhibitor, have an LVEF of less than or equal to 0.40, and have either symptomatic heart failure or diabetes. (Level of Evidence: A)

Class IIa

In STEMI patients who tolerate ACE inhibitors, an ARB can be useful as an alternative provided there are either clinical or radiological signs of heart failure or LVEF is less than 0.40. Valsartan and candesartan have established efficacy for this recommendation. (Level of Evidence: B)

The use of ACE inhibitors in the initial management of the patient with STEMI is reviewed in Section 6.3.1.6.9.1.

The proportional benefit of ACE inhibitor therapy is largest in higher-risk subgroups, including those with previous infarction, heart failure, depressed LVEF, and tachycardia (585,589,720). Survival benefit for patients more than 75 years old and for a low-risk subgroup without the features noted above is equivocal (589,720).

The duration of ACE inhibitor therapy after STEMI requires detailed analysis of trial results. In general, the trials that administered ACE inhibitors to an unselected population of patients early after MI had a short follow-up of 5 to 7 weeks. In contrast, trials that selected patients with post-MI LV dysfunction or clinical heart failure followed patients up to 5 years. Thus, when there is no evidence of symptomatic or asymptomatic LV dysfunction by 4 to 6 weeks, the indications for long-term ACE inhibitor use should be re-evaluated. The Heart Outcomes Prevention Evaluation (HOPE) trial enrolled patients whose entry characteristics were age 55 years or older, evidence of vascular disease, or diabetes plus 1 other cardiovascular risk factor (721). In the HOPE population, the use of ramipril at doses up to 10 mg daily demonstrated a significant reduction in the primary outcome, a composite of MI, stroke, or death of cardiovascular causes (3.8% absolute risk reduction; RRR 0.78, 95% CI, 0.70 to 0.86; p less than 0.001).

Aldosterone blockade is another means of inhibiting the renin-angiotensin-aldosterone system that has been applied to patients in the post-STEMI setting. Again, additional information can be inferred from heart failure studies that enrolled a large proportion of patients with a history of MI. The RALES study (Randomized Aldactone Evaluation Study) randomized patients with New York Heart Association class III to IV heart failure to either spironolactone (initial dose 25 mg daily with the option to increase to 50 mg daily) or placebo (722). Ischemic heart disease was the cause of heart failure in 55% of patients, and 95% were treated concurrently with an ACE inhibitor. Over 24 months of follow-up, spironolactone treatment was associated with an 11% ARD (24% RRR) in all-cause mortality. The EPHESUS study (Eplerenone Post-Acute Myocardial Infarction Heart Failure Efficacy and Survival Study) randomized 6632 post-MI patients with an ejection fraction of 0.40 or less and heart failure or diabetes to receive the aldosterone blocker eplerenone (target dose 50 mg daily) or placebo in conjunction with routine indicated cardiac medications. There was a significant reduction in overall mortality, cardiovascular mortality, and cardiac hospitalizations (723). The benefit was seen in patients managed with optimal therapy, including reperfusion, aspirin, ACE inhibitors, betablockers, and statins. Thus, the RALES and EPHESUS studies support the long-term use of an aldosterone blocker in patients with STEMI with heart failure and/or an ejection fraction of 0.40 or less, provided the serum creatinine is less than or equal to 2.5 mg/dL in men and less than or equal to 2.0 mg/dL in women and serum potassium concentration is less than or equal to 5.0 mEq/L. The risk of hyperkalemia was greatest in patients with creatinine clearance estimated to be less than 50 mL/min. Close monitoring of potassium levels is indicated for those patients, and the risk-to-benefit ratio should be weighed for those with moderate to severe reductions despite a serum creatinine of less than 2.5 mg/dL.

The use of ARBs after STEMI has not been explored as thoroughly as ACE inhibitors in patients with STEMI. The OPTIMAAL trial (Optimal Trial in Myocardial Infarction with Angiotensin II Antagonist Losartan) found no significant differences between losartan (target dose 50 mg once daily) and captopril (target dose 50 mg 3 times daily) in allcause mortality (724), but there was a trend toward better outcome with captopril. The VALIANT trial (Valsartan in Acute Myocardial Infarction Trial) compared the effects of captopril (target dose 50 mg 3 times daily), valsartan (target dose 160 mg twice daily), and their combination (captopril target dose 50 mg 3 times daily; valsartan target dose 80 mg twice daily) on mortality in post-MI patients with LV dysfunction (725). During a median follow-up of 24.7 months, death occurred in 19.9% of the valsartan group, 19.5% of the captopril group, and 19.3% of the combined-treatment group. The hazard ratio for death in the valsartan group compared with the captopril group was 1.00 (97.5% CI, 0.90 to 1.11; p equals 0.98), and the hazard ratio for death in the valsartan and captopril combined versus the captopril group was 0.98 (97.5% CI, 0.8 to 1.09; p equals 0.73) (725). The combination captopril and valsartan group had the most drugrelated adverse events. In the monotherapy groups, hypotension and renal dysfunction were more common in the valsartan group, and cough, rash, and taste disturbance were more common in the captopril group.

Given the extensive randomized trial and routine clinical experience with ACE inhibitors, they remain the logical first agent for inhibition of the renin-angiotensin-aldosterone system in patients convalescing from STEMI (726). Valsartan monotherapy (target dose 160 mg twice daily) should be administered to patients with STEMI who are intolerant of ACE inhibitors and who have evidence of LV dysfunction. Valsartan monotherapy can be a useful alternative to ACE inhibitors; the decision in individual patients may be influenced by physician and patient preference, cost, and anticipated side-effect profile.

7.4.4. Antiplatelets

Class I

- 1. Aspirin 162 to 325 mg should be given on day 1 of STEMI and in the absence of contraindications should be continued indefinitely on a daily basis thereafter at a dose of 75 to 162 mg. (Level of Evidence: A)
- 2. A thienopyridine (preferably clopidogrel) should be administered to patients who are unable to take aspirin because of hypersensitivity or major gastrointestinal intolerance. (Level of Evidence: C)
- **3.** For patients taking clopidogrel for whom CABG is planned, the drug should be withheld for at least 5 days if possible, and preferably for 7, unless the urgency for revascularization outweighs the risks of bleeding. (Level of Evidence: B)

4. For patients who have undergone diagnostic cardiac catheterization and for whom PCI is planned, clopidogrel should be started and continued for at least 1 month after bare metal stent implantation and for several months after drug-eluting stent implantation (3 months for sirolimus, 6 months for paclitaxel) and for up to 12 months in patients who are not at high risk for bleeding. (Level of Evidence: B)

Aspirin. The use of aspirin in patients with STEMI on initial presentation and in early management is discussed above (Sections 6.3.1.4 and 6.3.1.6.8.2.1). Aspirin should be given to the patient with suspected STEMI as early as possible and continued indefinitely, regardless of the strategy for reperfusion and regardless of whether additional antiplatelet agents are administered. True aspirin allergy is the only exception to this recommendation. Maintenance aspirin doses for secondary prevention of cardiovascular events in large trials have varied from 75 to 325 mg per day, but no trial has directly compared the efficacy of different doses after STEMI. An overview of trials with different doses of aspirin in the longterm treatment of patients with coronary disease suggests similar efficacy for daily doses ranging from 75 to 325 mg (727). An analysis from the CURE trial (Clopidogrel in Unstable angina to prevent Recurrent Events) suggests a dose-dependent increase in bleeding in patients receiving aspirin plus placebo: The major bleeding event rate was 2.0% in patients taking less than 100 mg of aspirin, 2.3% with 100 to 200 mg, and 4.0% with greater than 200 mg per day (728,729). Therefore, lower aspirin doses of 75 to 162 mg per day are preferred for long-term treatment.

The side effects of aspirin are mainly gastrointestinal and dose related (730). Gastric side effects may also be reduced by administration of diluted solutions of aspirin (731), treatment with H2 antagonists (732), antacids (731,733), or use of enteric-coated or buffered aspirin (734,735).

Aspirin should be avoided in those with a known hypersensitivity and used cautiously in those with blood dyscrasias or severe hepatic disease (see additional discussion in Sections 7.12.5 and 7.12.11). If the patient has a history of bleeding peptic ulcers, the use of rectal aspirin suppositories may be safer because it eliminates the local effect of aspirin on the gastric mucosa. However, antiplatelet effects may still pose a risk. Another potentially deleterious effect of aspirin is risk of bleeding from surgical sites. Patients who received aspirin in the Veterans Administration Cooperative Study (736) were noted to have significantly increased postoperative chest drainage and reoperation for bleeding (6.5% for aspirin groups compared with 1.7% for nonaspirin groups; p less than 0.01). Others have noted that preoperative aspirin use has been associated with increased postoperative chest drainage but not an increased rate of reoperation for bleeding (737,738). In another Veterans Administration Cooperative Study (739), starting aspirin 6 hours after surgery conferred the benefits of improved saphenous vein bypass graft patency without the increased postoperative bleeding seen with preoperative administration of aspirin. Aspirin (81 to 365

mg) should be administered as soon as possible (within 24 hours) after CABG unless contraindicated (see Section 7.10.7).

Use of the thienopyridines ticlopidine and clopidogrel in the early management of STEMI is discussed above (Section 6.3.1.6.8.2.2). Clopidogrel 75 mg daily is generally preferred to ticlopidine 250 mg twice daily because of fewer side effects and once-daily dosing (740,741).

In the Clopidogrel vs. Aspirin in Patients at Risk of Ischemic Events (CAPRIE) trial of patients with UA/NSTE-MI, there was a statistically significant risk reduction in vascular death, MI, or stroke in favor of clopidogrel (0.51% ARD; RRR 8.7%) (742). Clopidogrel has also demonstrated efficacy in addition to aspirin versus aspirin alone in patients with acute coronary syndrome (728). Therefore, a thienopyridine (preferably clopidogrel) should be substituted for aspirin in patients with STEMI for whom aspirin is contraindicated because of hypersensitivity or major gastrointestinal intolerance. On the basis of several randomized trials of combination antiplatelet therapy (577,741,743), clopidogrel, in combination with low-dose aspirin (75 to 162 mg, to minimize the risk of bleeding), is recommended for all patients after stent implantation (432).

7.4.5. Antithrombotics

Class I

Intravenous UFH (bolus of 60 U/kg, maximum 4000-U IV bolus; initial infusion of 12 U/kg/h, maximum 1000 U/h) or LMWH should be used in patients after STEMI who are at high risk for systemic emboli (large or anterior MI, AF, previous embolus, known LV thrombus, or cardiogenic shock). (Level of Evidence: C)

Class IIa

It is reasonable that STEMI patients not undergoing reperfusion therapy who do not have a contraindication to anticoagulation be treated with intravenous or subcutaneous UFH or with subcutaneous LMWH for at least 48 hours. In patients whose clinical condition necessitates prolonged bedrest and/or minimized activities, it is reasonable that treatment be continued until the patient is ambulatory. (Level of Evidence: C)

Class IIb

Prophylaxis for DVT with subcutaneous LMWH (dosed appropriately for specific agent) or with subcutaneous UFH, 7500 to 12 500 U twice per day until completely ambulatory, may be useful, but the effectiveness of such a strategy is not well established in the contemporary era of routine aspirin use and early mobilization. (Level of Evidence: C)

In patients treated with fibrinolytic therapy, there is little evidence of the benefit of UFH in the modern era, during which aspirin, beta-blockers, nitrates, and ACE inhibitors are routinely available. Nevertheless, the best available data emanate from a series of randomized clinical trials performed before the reperfusion era. A systematic overview of these studies demonstrated a reduction in mortality (3.5% ARD; 23% RRR) and a reduction in risk of reinfarction (1.5% ARD; 18% RRR) with UFH (537). The control groups in these trials were not treated with other therapies, particularly aspirin, that are now considered routine. Notwithstanding this, it is primarily these randomized data from an earlier era that support the recommendation to use UFH in patients not treated with fibrinolytic therapy.

The occurrence of a large anterior infarction, documentation of thrombus in the LV by echocardiography, history of a previous embolic event, and AF have been associated with a high risk of embolic stroke. Although no randomized trial evidence exists to demonstrate a definite benefit specific to this group, some empirical evidence exists that the risk of systemic emboli in the general population of MI patients can be reduced by early initiation of UFH (744). In the SCATI trial (Studio sulla Calciparina nell'Angina e nella Trombosi Ventricolare nell'Infarto), patients were randomly assigned to a 2000-IU bolus of UFH followed by 12 500 U subcutaneously twice per day or to placebo. In the subgroup also treated with streptokinase, aspirin was withheld. In-hospital mortality was 4.6% in the UFH group and 8.8% in the control group, and a reduction in stroke was observed. Therefore, UFH is recommended for these patients at high risk for systemic arterial emboli, regardless of the fibrinolytic agent given. A LMWH may be used in place of UFH. Initial anticoagulation with UFH or a LMWH should be followed by warfarin in patients at high risk for systemic emboli (see Section 7.12.11 for additional discussion).

The previous ACC/AHA guidelines on acute MI (744) and the American College of Chest Physicians' guidelines (746) recommended 7500 U of subcutaneous UFH twice per day. The empirical basis for this recommendation was the demonstration that DVT was reduced from 12% to 4% in an overview of 3 randomized controlled trials (747). Continued adherence to this practice may be useful, although routine earlier mobilization and use of aspirin may make this treatment unnecessary.

7.4.6. Oxygen

Class I

Supplemental oxygen therapy should be continued beyond the first 6 hours in STEMI patients with arterial oxygen desaturation (SaO₂ less than 90%) or overt pulmonary congestion. (Level of Evidence: C)

The use of oxygen in patients presenting with STEMI is discussed in Section 6.3.1.1. In view of its expense (approximately \$70 per day), there is little justification for continuing its routine use beyond 6 hours in uncomplicated cases.

Pulse oximetry is now routine for continuous monitoring of oxygen saturation and is helpful for providing early warning of hypoxemia. In patients with oxygen saturation less than 90%, supplemental oxygen by nasal prongs is usually administered, especially if the patient is experiencing ongoing or intermittent ischemia. This therapy is based on experimental data that suggest that normal levels of oxygen reduce infarct size. In patients with severe heart failure, pulmonary edema, or mechanical complications of STEMI, significant hypoxemia may require continuous positive pressure breathing or endotracheal intubation and mechanical ventilation.

7.5. Estimation of Infarct Size

Measurement of infarct size is an important element in the overall care of patients with STEMI. The extent of infarction bears a direct relationship to prognosis, assists in establishing the efficacy of reperfusion therapy, guides both shortand long-term therapeutic decision making, and provides a useful surrogate for the investigation of novel experimental therapies.

There are 5 major modalities that can be applied to sizing MI. A discussion of each follows.

7.5.1. ECG Techniques

Class I

All patients with STEMI should have follow-up ECGs at 24 hours and at hospital discharge to assess the success of reperfusion and/or the extent of infarction, defined in part by the presence or absence of new Q waves. (Level of Evidence: B)

The extent of ST-segment deviation on the baseline ECG provides a semiquantitative measure of the amount of jeopardized myocardium and an estimate of the subsequent infarct size likely to ensue in a nonreperfused population. With a QRS scoring system based on the duration and amplitude of individual waveforms within the QRS complex, the size of the infarction can be estimated from a point score derived and weighted from the 12-lead ECG (748). Each point so derived represents approximately 3% of the LV, and the utility of this approach has been validated in postmortem studies of patients with confirmed MI (749). This method, however, is time consuming and is limited in patients with concomitant LV hypertrophy or fascicular or bundle-branch block and when major ST-segment shift distorts the appearance of the QRS complex.

In the fibrinolytic era, simple characterization of the presence or absence of the Q wave has also been used. In the early convalescent period after fibrinolytic therapy, the 20% of patients in the GUSTO angiographic study who did not develop Q waves had better global and regional LV function and improved 2-year survival (6.3% versus 10.1% for patients with developed Q waves; p equals 0.02) (750).

7.5.2. Cardiac Biomarker Methods

The most widely accepted method for quantifying infarction has been the use of serial CK and the CK-MB isoenzyme. With a mathematical formulation based on rates of degradation in specific compartments, the rate of myocardial biomarker release, its volume of distribution, and its clearance rate, it is possible to estimate the quantity of myocardium infarcted (751). Reasonable correlations have been established with anatomic estimates derived from postmortem human studies (752). Whereas other biomarkers of myocardial necrosis exist, such as myoglobin and lactate dehydrogenase, the highly sensitive cardiac troponins (I or T) have greater myocardial tissue specificity and higher sensitivity than conventional biomarkers. Measurement of cardiac troponin T at 72 hours provides an estimate of infarct size in patients with STEMI who do and do not receive reperfusion therapy (753,754). In a consensus document of the Joint European Society of Cardiology and the ACC, the use of cardiac troponins was supported for the assessment of MI. The Joint Committee has emphasized that high-sensitivity cardiac biomarkers, such as troponins, can identify patients with small areas of myocardial necrosis weighing less than 1.0 g (755).

7.5.3. Radionuclide Imaging

The most comprehensive assessment of STEMI with radionuclide imaging was developed with the Technetium sestamibi SPECT approach (756). This technique has been validated extensively and offers the opportunity for both early and late imaging to initially assess the area of ischemic risk as opposed to the ultimate infarct size. This approach is well delineated in the ACC/AHA/ASNC Guidelines on Cardiac Radionuclide Imaging (239). Radionuclide angiography with a variety of radiolabeled isotopes can also provide an estimate of regional and global LV function.

7.5.4. Echocardiography

Global and regional LV function provides an assessment of the functional consequences of STEMI and ischemia. Such measures may be enhanced by an assessment of the extent of regional systolic wall thickening. Readers are referred Section 7.11.1.2 and to the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography (226).

7.5.5. Magnetic Resonance Imaging

Measurement of infarct size with magnetic resonance imaging is a promising new technique that affords enhanced spatial resolution, thereby permitting more accurate assessment of both the transmural and circumferential extent of infarction (757). However, additional experience and comparison with other methods of assessing infarct size are required before any clinical recommendations can be provided.

7.6. Hemodynamic Disturbances

7.6.1. Hemodynamic Assessment

Class I

- **1.** Pulmonary artery catheter monitoring should be performed for the following:
 - a. Progressive hypotension, when unresponsive to fluid administration or when fluid administration

may be contraindicated. (Level of Evidence: C)

- b. Suspected mechanical complications of STEMI, (i.e., VSR, papillary muscle rupture, or free wall rupture with pericardial tamponade) if an echocardiogram has not been performed. (Level of Evidence: C)
- 2. Intra-arterial pressure monitoring should be performed for the following:
 - a. Patients with severe hypotension (systolic arterial pressure less than 80 mm Hg). (*Level of Evidence: C*)
 - b. Patients receiving vasopressor/inotropic agents. (Level of Evidence: C)
 - c. Cardiogenic shock. (Level of Evidence: C)

Class IIa

- **1.** Pulmonary artery catheter monitoring can be useful for the following:
 - a. Hypotension in a patient without pulmonary congestion who has not responded to an initial trial of fluid administration. (Level of Evidence: C)
 - b. Cardiogenic shock. (Level of Evidence: C)
 - c. Severe or progressive CHF or pulmonary edema that does not respond rapidly to therapy. (Level of Evidence: C)
 - d. Persistent signs of hypoperfusion without hypotension or pulmonary congestion. (*Level of Evidence: C*)
 - e. Patients receiving vasopressor/inotropic agents. (Level of Evidence: C)
- 2. Intra-arterial pressure monitoring can be useful for patients receiving intravenous sodium nitroprusside or other potent vasodilators. (*Level of Evidence: C*)

Class IIb

Intra-arterial pressure monitoring might be considered in patients receiving intravenous inotropic agents. (Level of Evidence: C)

Class III

- 1. Pulmonary artery catheter monitoring is not recommended in patients with STEMI without evidence of hemodynamic instability or respiratory compromise. (Level of Evidence: C)
- 2. Intra-arterial pressure monitoring is not recommended for patients with STEMI who have no pulmonary congestion and have adequate tissue perfusion without use of circulatory support measures. (Level of Evidence: C)

Measurements made via pulmonary artery catheter readings may be helpful in the management of STEMI and concomitant hemodynamic instability, including low cardiac output, hypotension, persistent tachycardia, pulmonary edema, and apparent cardiogenic shock. In the patient with hypotension and tachycardia, the pulmonary artery catheter can assist in the differentiation of 1) inadequate intravascular volume, with a resultant low left-sided filling pressure; 2) adequate intravascular volume and a high left-sided filling pressure due to extensive LV dysfunction; and 3) low leftsided filling pressure with elevated right atrial pressure consistent with RV infarction (758). Treatment of the former is prompt expansion of intravascular volume (with normal saline), whereas management of the latter often includes diuresis, inotropic support, afterload reduction, or other supportive measures. In those with extensive LV dysfunction, a pulmonary artery catheter can be used to monitor therapeutic efforts to adjust the left-sided filling pressure so as to maximize cardiac output at the lowest possible filling pressure (758). In some patients, these sophisticated manipulations of intracardiac pressures and cardiac output are facilitated by information provided by a pulmonary artery catheter.

Although the pulmonary artery catheter is quite safe when used by experienced operators, its use has a recognized association with adverse events, including ventricular tachyarrhythmias (during its manipulation) and pulmonary hemorrhage or infarction. Transient right bundle-branch block may develop, which can lead to heart block in those with preexisting LBBB. In addition, it causes some patient discomfort and requires that the patient be relatively immobile. Because the pressure waveform recorded from the catheter tip may be distorted, the clinician should routinely examine the actual waveform rather than rely on the digital display of pressure. Because of the risk of infection, pulmonary artery catheters generally should not remain in the same site for more than 4 to 5 days. The catheter should not be inserted if the patient quickly responds to other interventions or if treatment is expected to be futile. The catheter should be removed expeditiously when it is no longer needed to monitor therapy.

The diagnosis of acute mechanical complications (MR, ventricular septal defect, myocardial rupture) usually can be diagnosed quickly and safely in the ED by transthoracic echocardiography. However, the ordering and performing of an echocardiogram should not delay transfer of a hemodynamically unstable patient to the interventional cardiology laboratory, where hemodynamic stabilization, diagnosis, and treatment can usually best be initiated. Similarly, insertion of a pulmonary artery catheter in the CCU should not delay transfer of the patient to the interventional cardiology laboratory if indicated.

Insertion of a pulmonary artery catheter to measure hemodynamics in patients developing progressive CHF or hypotension may permit the early diagnosis of a preshock state in which appropriate support can prevent the onset of cardiogenic shock (759). Before PCI is performed for cardiogenic shock, the interventional cardiologist should insert a pulmonary artery catheter to maximize the hemodynamic status of the patient and to diagnose unrecognized mechanical complications. After reperfusion therapy, if shock does not rapidly reverse, the pulmonary artery catheter may be used to guide diuretic, inotropic, and vasopressor agents in hemodynamically unstable patients while stunned myocardium is recovering. Unfortunately, there are no data from randomized controlled trials testing whether or not hemodynamic monitoring alters clinical outcome in STEMI.

It is possible that therapeutic interventions in response to erroneous data or inappropriate maneuvers in response to accurate data contribute to the excess mortality rates previously associated with pulmonary artery catheter use (760-762). Clinicians should understand the multiple determinants of pulmonary capillary wedge pressure (PCWP) to avoid equating filling pressure with intravascular volume (compliance, veno-vasodilation, intrathoracic pressure, position, volume). Dynamic changes in ventricular compliance and the use of vasodilating medications result in rapid changes in PCWP that do not reflect intravascular volume. For example, pulmonary artery catheter placement after intravenous nitroglycerin administration for pulmonary edema may reflect PCWP less than 15 mm Hg despite marked elevation 1 hour earlier. An acute ischemic episode may raise PCWP substantially without a concomitant change in volume.

All CCUs should have sufficient equipment and skilled personnel to monitor intra-arterial pressure. Such monitoring is useful in all hypotensive patients, particularly those with cardiogenic shock. Long-term monitoring is best accomplished through the radial artery, although the brachial or femoral arteries may be used as alternatives. Perfusion of the limb or hand distal to the catheter site must be examined carefully and periodically for evidence of ischemia. Intraarterial and central catheters can be left in place with a sterile occlusive dressing. Before insertion, the site should be adequately prepared under sterile conditions. Antibacterial ointments are no longer recommended. Because of the risk of arterial thrombosis and infection, intra-arterial catheters generally should not remain in the same arterial site for longer than 4 to 5 days (763).

7.6.2. Hypotension

Class I

- **1.** Rapid volume loading with an intravenous infusion should be administered to patients without clinical evidence for volume overload. (*Level of Evidence: C*)
- 2. Rhythm disturbances or conduction abnormalities causing hypotension should be corrected. (Level of Evidence: C)
- **3.** Intra-aortic balloon counterpulsation should be performed in patients who do not respond to other interventions, unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (*Level of Evidence: B*)
- 4. Vasopressor support should be administered for hypotension that does not resolve after volume load-ing. (Level of Evidence: C)
- 5. Echocardiography should be used to evaluate mechanical complications unless these are assessed by invasive measures. (*Level of Evidence: C*)

Hypotension (systolic pressure less than 90 mm Hg or 30 points below previous systolic arterial pressure) can result from hypovolemia, arrhythmias, RV or LV failure, mechanical complications of MI, or superimposed complications, such as sepsis or pulmonary embolism. Hypovolemia is a common occurrence and may be due to inadequate intake,

diaphoresis and vomiting, overdiuresis, excessive use of vasodilators, or inappropriate reflex peripheral vasodilation. Hemorrhage is an increasingly common problem associated with the use of invasive procedures, fibrinolytics, antiplatelet agents, and anticoagulant agents. Therefore, rapid volume loading is recommended as an initial therapeutic strategy in all patients without clinical evidence for volume overload. Persistent hypotension should be evaluated with an echocardiogram to define the cardiac anatomy and with a hemoglobin measurement. Correction or control of rhythm disturbances or conduction abnormalities often reverses hypotension. In patients with inotropic failure, vasopressors and inotropic agents are required. Vasopressor agents, including high-dose dopamine and norepinephrine, have alpha-vasoconstricting properties, whereas inotropic agents such as dobutamine have beta-receptor-stimulating properties. Norepinephrine and dopamine have both vasopressor and inotropic properties. When blood pressure is low, dopamine is the agent of first choice. If the patient is markedly hypotensive, intravenous norepinephrine, which is a more potent vasoconstrictor with less potential for tachycardia, should be administered until systolic arterial pressure rises to at least 80 mm Hg, at which time a change to dopamine may be attempted, initiated at 2.5 to 5 mcg/kg/min and titrated as needed to 5 to 15 mcg/kg/min. Once arterial pressure is brought to at least 90 mm Hg, intravenous dobutamine may be given simultaneously in an attempt to reduce the rate of the dopamine infusion. In addition, consideration should be given to initiating intra-aortic balloon counterpulsation. (See Section 7.6.1.)

7.6.3. Low-Output State

Class I

- **1.** Left ventricular function and potential presence of a mechanical complication should be assessed by echocardiography if these have not been evaluated by invasive measures. (*Level of Evidence: C*)
- 2. Recommended treatments for low output states include:
 - a. Inotropic support. (Level of Evidence: B)
 - b. Intra-aortic counterpulsation. (Level of Evidence: B)
 - c. Mechanical reperfusion with PCI or CABG. (Level of Evidence: B)
 - d. Surgical correction of mechanical complications. (Level of Evidence: B)

Class III

Beta-blockers or calcium channel antagonists should not be administered to patients in a low-output state due to pump failure. (*Level of Evidence: B*)

A preshock state of hypoperfusion with normal blood pressure may develop before circulatory collapse and is manifested by cold extremities, cyanosis, oliguria, or decreased mentation (759). Hospital mortality is high, so these patients should be aggressively diagnosed and treated as though they had cardiogenic shock. The initial pharmacological intervention for low cardiac output is often a dobutamine infusion. Intra-aortic counterpulsation therapy may be required to improve coronary artery perfusion pressure if hypotension is present. If the blood pressure permits, afterload-reducing agents should be added to decrease cardiac work and pulmonary congestion. Coronary artery revascularization of ischemic myocardium with either PCI or CABG has been shown to decrease mortality in patients with cardiogenic shock and is strongly recommended in suitable candidates (184,301). Likewise, patients with VSR, papillary muscle rupture, or free wall rupture with pericardial tamponade may benefit from emergency surgical repair.

7.6.4. Pulmonary Congestion

Class I

- 1. Oxygen supplementation to arterial saturation greater than 90% is recommended for patients with pulmonary congestion. (Level of Evidence: C)
- 2. Morphine sulfate should be given to patients with pulmonary congestion. (*Level of Evidence: C*)
- 3. ACE inhibitors, beginning with titration of a shortacting ACE inhibitor with a low initial dose (e.g., 1 to 6.25 mg of captopril) should be given to patients with pulmonary edema unless the systolic blood pressure is less than 100 mm Hg or more than 30 mm Hg below baseline. Patients with pulmonary congestion and marginal or low blood pressure often need circulatory support with inotropic and vasopressor agents and/or intra-aortic balloon counterpulsation to relieve pulmonary congestion and maintain adequate perfusion. (Level of Evidence: A)
- 4. Nitrates should be administered for patients with pulmonary congestion unless the systolic blood pressure is less than 100 mm Hg or more than 30 mm Hg below baseline. Patients with pulmonary congestion and marginal or low blood pressure often need circulatory support with inotropic and vasopressor agents and/or intra-aortic balloon counterpulsation to relieve pulmonary congestion and maintain adequate perfusion. (Level of Evidence: C)
- 5. A diuretic (low- to intermediate-dose furosemide, or torsemide or bumetanide) should be administered to patients with pulmonary congestion if there is associated volume overload. Caution is advised for patients who have not received volume expansion. (Level of Evidence: C)
- 6. Beta-blockade should be initiated before discharge for secondary prevention. For those who remain in heart failure throughout the hospitalization, low doses should be initiated, with gradual titration on an outpatient basis. (Level of Evidence: B)
- 7. Long-term aldosterone blockade should be prescribed for post-STEMI patients without significant renal dysfunction (creatinine should be less than or equal to 2.5 mg/dL in men and less than or equal to 2.0 mg/dL in women) or hyperkalemia (potassium should be less

than or equal to 5.0 mEq/L) who are already receiving therapeutic doses of an ACE inhibitor, have an LVEF of less than or equal to 0.40, and have either symptomatic heart failure or diabetes. (*Level of Evidence: A*)

8. Echocardiography should be performed urgently to estimate LV and RV function and to exclude a mechanical complication. (Level of Evidence: C)

Class IIb

It may be reasonable to insert an IABP for the management of patients with refractory pulmonary congestion. (*Level of Evidence: C*)

Class III

Beta-blockers or calcium channel blockers should not be administered acutely to STEMI patients with frank cardiac failure evidenced by pulmonary congestion or signs of a low-output state. (Level of Evidence: B)

Left ventricular filling pressures, and hence PCWP, may rise rapidly after acute coronary occlusion. This rise is due to acute systolic or diastolic dysfunction that may be associated with superimposed MR. The rise in PCWP leads to rapid redistribution of fluid from the intravascular space into the extravascular space (lung interstitium and alveoli). The presence of any pulmonary congestion on examination or X-ray increases the risk of dying, and pulmonary edema is associated with a 20% to 40% 30-day mortality rate even in the fibrinolytic era (28,240,242,764).

Immediate management goals include adequate oxygenation and preload reduction to relieve pulmonary congestion. Because of sympathetic stimulation, the blood pressure should be elevated in the presence of pulmonary edema. Patients with this appropriate response can typically tolerate the required medications, all of which lower blood pressure. However, iatrogenic cardiogenic shock may result from aggressive simultaneous use of agents that cause hypotension, initiating a cycle of hypoperfusion-ischemia. If acute pulmonary edema is not associated with elevation of the systemic blood pressure, impending cardiogenic shock must be suspected. If pulmonary edema is associated with hypotension, cardiogenic shock is diagnosed. Those patients often need circulatory support with inotropic and vasopressor agents and/or intra-aortic balloon counterpulsation to relieve pulmonary congestion and maintain adequate perfusion (see Section 7.6.5 and Figure 26).

Pulmonary edema may occur as an acute event with the onset of STEMI or reinfarction or as the culmination of slowly progressive CHF over the first several days after infarction. Acute pulmonary edema on presentation with STEMI may occur in a patient with prior myocardial damage and systolic dysfunction, with or without a prior diagnosis of CHF. Alternatively, it may develop in patients with a first STEMI, especially those with preceding diastolic dysfunction due to hypertension or diabetes. Pulmonary edema days after STEMI or on presentation in patients who have prior CHF or LV dysfunction is often associated with hypervolemia. In contrast, patients who present with pulmonary



Figure 28. Emergency management of complicated ST-elevation myocardial infarction (STEMI). The emergency management of patients with cardiogenic shock, acute pulmonary edema or both is outlined. IV = intravenous; SL - sublingual; SBP = systolic BP; BP = blood pressure; ACE = angiotensin converting enzyme; MI = myocardial infarction. *Furosemide less than 0.5 mg/kg for new onset acute pulmonary edema without hypovolemia. 1 mg/kg for acute or chronic volume overload, renal insufficiency. Nesiritide has not been studied adequately in patients with STEMI. Combinations of medications (e.g., dobutamine and dopamine) may be used. Modified with permission from Guidelines 2000 for Cardiopulmonary Resuscitation and Emergency Cardiovascular Care Part 7: The Era of Reperfusion. Section 1: Acute Coronary Syndromes (Acute Myocardial Infarction). Circulation 2000;102: I-172 - I-216 (766).

edema without prior LV dysfunction (and who have not received fluid administration) usually have a normal total body sodium and fluid status. The acute redistribution of fluid into the lungs results in relative intravascular volume depletion in the early phase. Acute diuresis and agents that induce hypotension can precipitate cardiogenic shock.

The cause of pulmonary edema (i.e., systolic, diastolic, or a mechanical complication [MR or VSR]) should be assessed rapidly with a 2-dimensional echocardiograph with color flow Doppler. Indications for right heart catheterization are reviewed in Section 7.6.1. On the basis of data that high-risk patients derive greater benefit from PCI than from fibrinolytic therapy, primary PCI is preferred when available for those who present with pulmonary edema complicating STEMI. There are no randomized trials that assess this comparison. However, analysis of the GUSTO-IIB trial shows similar relative and greater absolute benefit from PCI for Killip class 2 to 3 patients (765). The NRMI registry showed a marked benefit of PCI compared with fibrinolysis for patients with CHF (447). Coronary angiography and revascularization, based on the anatomy, should be performed when late CHF complicates the hospital course. Mechanical ventilation may be required during angiography and PCI, especially for primary PCI during STEMI.

Management includes the use of agents that acutely reduce preload (i.e., nitrates, morphine sulfate, and diuretics) (Figure 28) (766), and avoidance of acute administration of negative inotropic agents (i.e., beta-blockers and calcium channel antagonists). Nitrates are initially administered by sublingual tablets or spray nitroglycerin followed by intravenous nitroglycerin. Intravenous nitroglycerin is a venodilator that acutely reduces ventricular filling pressures. At high doses, it dilates arterioles. It is effective at relieving pulmonary congestion and ischemia and may be used in patients who have normal or elevated systemic arterial pressure. A 10- to 20-mcg bolus should be administered, followed by 10 mcg per minute, increased by 5 to 10 mcg per minute every 5 to 10 minutes until dyspnea is relieved, the mean arterial pressure is lowered by 10% in normotensive patients or 30% in hypertensive patients, or until the heart rate increases by more than 10 bpm. Loop diuretics (furosemide, torsemide, or bumetanide) should be initiated in low to intermediate doses only in patients with associated hypervolemia (see above). Low doses should be used unless there is renal insufficiency, chronic diuretic use, or the presence of chronic CHF and hypervolemia as described above. Typical furosemide doses range from 20 to 80 mg IV (0.5 to 1.0 mg/kg).

Angiotensin converting enzyme inhibitors are indicated for patients with pulmonary congestion. Oral ACE inhibitors, preferably a short-acting agent such as captopril, beginning with 1 to 6.25 mg, should be instituted early in normotensive or hypertensive patients. The dosage may be doubled with each subsequent dose as tolerated up to 25 to 50 mg every 8 hours, then changed to a long-acting agent. Although the risk of hypotension and shock after vasodilator or diuretic administration during the acute phase of MI is substantial for those without a hypertensive response to pulmonary edema, the risk is lower in the late phase after MI. Hence, most patients can tolerate ACE inhibitors before discharge. For patients who presented with CHF complicating MI, ramipril administration between days 3 and 10 significantly reduced 30-day mortality (relative hazard 0.73; 95% CI 0.602 to 0.89; p greater than 0.002) in 2006 patients in the Acute Infarction Ramipril Efficacy Study (767). Given the good tolerability of ACE inhibition within 24 hours of MI in the ISIS-4 and GISSI-3 (lisinopril) studies and the beneficial effects on early infarct expansion, it is recommended that ACE inhibitors be initiated early for those who have pulmonary congestion. However, hypotension should be avoided, particularly during and immediately after reperfusion therapy (767,768). Routine intravenous enalapril is not recommended (769) unless severe hypertension is present. ACE inhibitors are the only adjunctive medication (beyond aspirin and reperfusion therapy) demonstrated to reduce 30-day mortality when CHF complicates STEMI. Therefore, if blood pressure limits use of vasodilators, ACE inhibitors are preferred. Intravenous sodium nitroprusside substantially reduces afterload and preload; however, its use has been associated with coronary steal. Digitalis has no role in the management of pulmonary edema complicating STEMI unless rapid AF is present. Nesiritide (synthetic natriuretic brain peptide) is a new vasodilator agent that promotes diuresis in patients with volume overload and decompensated chronic CHF (class 3 to 4) (770). It has not been investigated in STEMI and is not indicated for treatment of pulmonary edema in these patients. Nesiritide is a potent vasodilator and may result in hypotension, particularly in patients with STEMI, in whom CHF usually is not due to volume overload.

An aldosterone antagonist, eplerenone, was found to be effective for secondary prevention of death and recurrent hospitalization in patients 3 to 14 days after MI with CHF and LVEF less than 0.40. Spironolactone has been demonstrated to improve survival in a population of patients with chronic CHF, which includes those with remote MI (722) (see Section 7.12.6). In contrast to the recommendation to avoid initiation of beta-blockade during pulmonary edema, beta-blockers are strongly recommended before hospital discharge for secondary prevention of cardiac events (273). The initial dose and titration should be based on clinical heart failure status and LVEF. For patients who remain in heart failure during the hospitalization, a low dose should be initiated and gradually titrated as an outpatient, per CHF guidelines (771). This is supported by the beneficial effects of beta-blockade in patients with LV dysfunction after STEMI (771).

See also Sections 7.4.3 (hospital phase) and 7.12.6 (secondary prevention) for recommendations on ARBs.

7.6.5. Cardiogenic Shock

Class I

- 1. Intra-aortic balloon counterpulsation is recommended for STEMI patients when cardiogenic shock is not quickly reversed with pharmacological therapy. The IABP is a stabilizing measure for angiography and prompt revascularization. (Level of Evidence: B)
- 2. Intra-arterial monitoring is recommended for the management of STEMI patients with cardiogenic shock. (Level of Evidence: C)
- 3. Early revascularization, either PCI or CABG, is recommended for patients less than 75 years old with ST elevation or LBBB who develop shock within 36 hours of MI and who are suitable for revascularization that can be performed within 18 hours of shock unless further support is futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (Level of Evidence: A)
- 4. Fibrinolytic therapy should be administered to STEMI patients with cardiogenic shock who are unsuitable for further invasive care and do not have contraindications to fibrinolysis. (Level of Evidence: B)
- 5. Echocardiography should be used to evaluate mechanical complications unless these are assessed by invasive measures. (*Level of Evidence: C*)

Class IIa

- 1. Pulmonary artery catheter monitoring can be useful for the management of STEMI patients with cardiogenic shock. (Level of Evidence: C)
- 2. Early revascularization, either PCI or CABG, is reasonable for selected patients 75 years or older with ST elevation or LBBB who develop shock within 36 hours of MI and who are suitable for revascularization that can be performed within 18 hours of shock. Patients with good prior functional status who agree to invasive care may be selected for such an invasive strategy. (Level of Evidence: B)

Cardiogenic shock in patients with STEMI is most commonly (75% of cases) caused by extensive LV dysfunction, but important other causes include mechanical complications (acute severe MR, VSR, and subacute free-wall rupture with tamponade). Important conditions that may mimic cardiogenic shock include aortic dissection and hemorrhagic shock. Echocardiography with color flow Doppler is extremely use-



Figure 29. Kaplan-Meier survival of cardiogenic shock after early revascularization curve 1-year postrandomization. Survival estimates for early revascularization (n=152) and initial medical stabilization (n=149) groups. Log-rank test P = 0.04. Reprinted with permission from Hochman et al. JAMA 2001;285:190-2. Copyrighted © 2001, American Medical Association. All rights reserved (184).

ful to assess the cause of shock. (See Section 7.6.1 for discussion on hemodynamic assessment.)

Nonrandomized studies have suggested that mechanical reperfusion of occluded coronary arteries by PCI or CABG may improve survival in patients with MI and cardiogenic shock. In large clinical trials, such patients have an in-hospital survival rate that ranges from 20% to 50% when treated with intravenous fibrinolytic therapy (482,483,772,773). In other case series, mechanical reperfusion with PCI has been reported to result in hospital survival rates as high as 70%, but selection bias influenced these findings. However, a multicenter, prospective, randomized study confirmed this general approach (184). The SHOCK trial tested the hypothesis that emergency revascularization for cardiogenic shock due to an ST-elevation/Q-wave or new LBBB MI would result in reduction in all-cause 30-day mortality compared with initial medical stabilization and delayed revascularization as clinically determined. In the SHOCK trial, cardiogenic shock was defined as clinical evidence of systemic hypoperfusion with systolic blood pressure less than 90 mm Hg for at least 30 minutes (or the need for supportive measures to maintain systolic blood pressure greater than 90 mm Hg), cardiac index of no more than 2.2 L/min/m² and PCWP of at least 15 mm Hg.

In the SHOCK trial, 152 patients were randomly assigned to the emergency revascularization strategy, and 150 patients were assigned to a strategy of initial medical stabilization. The 30-day mortality rate for emergency revascularization patients was 46.7% versus 56.0% for initial medical stabilization patients (95% CI minus 20.5 to plus 1.9%, p equals 0.11). However, the mortality rate at 6 and 12 months (secondary end points) was significantly lower in the emergency revascularization group (53.3% versus 66.4%, p less than 0.03, 12 months) (184,494). The prespecified subgroup analysis of patients less than 75 years old showed a 15.4% absolute reduction in the primary end point at 30 days (initial medical stabilization group 56.8% versus emergency revascularization group 41.4%, p less than 0.01), whereas no treatment benefit was apparent for the 56 patients greater than 75 years old. Intra-aortic balloon pump support was used in 86% of both groups; 63% of the initial medical stabilization group received thrombolytic agents, and 25% underwent delayed revascularization (Figure 29) (184). The (S)MASH study [(Swiss) Multicenter Trial of Angioplasty for Shock] randomly assigned 55 refractory shock patients to either PCI or conventional care. The mortality rate in the PCI group was 9 absolute percentage points lower at 30 days (69% versus 78%) than in the conventional therapy group but did not reach statistical significance (495). The group difference was similar to that observed at 30 days in the SHOCK trial (494).

Given the large overall treatment benefit of 13 lives saved per 100 patients treated, emergency revascularization is recommended for those less than 75 years who are suitable for revascularization. Patients with life-shortening illnesses, no vascular access, previously defined coronary anatomy that was unsuitable for revascularization, anoxic brain damage, and prior cardiomyopathy were excluded from the trial. For those enrolled, the treatment benefit was similar for all other subgroups examined: diabetics, women, prior MI or hypertension, and early or late developing shock. The elderly pose a special problem. There were only 56 patients 75 years of age or older in the SHOCK trial, and firm conclusions cannot be drawn. The mortality rate for the elderly patients assigned to initial medical stabilization was similar to younger patients assigned to initial medical stabilization and was therefore unexpectedly low (53.1%). Imbalances in

baseline characteristics of the 56 elderly patients assigned to the emergency revascularization versus initial medical stabilization groups may also have played a role in the apparent lack of treatment effect (301). Those elderly patients (n equals 277) who were clinically selected for early revascularization (17% of the cohort) in the larger, nonrandomized SHOCK registry had a marked survival benefit compared with those with late or no revascularization, even after covariate adjustment and exclusion of early deaths (496). Two other large registries reported a substantial survival benefit for the elderly who were clinically selected on the basis of physician judgment. In these 2 registries, 16% to 33% of the elderly were selected for an invasive strategy (497,498). Although not reported, selection is typically based on prior functional status, comorbidity, suitability for revascularization, and patient and family preferences. An analysis of Medicare patients admitted to hospitals with or without revascularization capability reported no significant reduction in mortality by institution type for patients who presented in shock (0.6%) (774). They did not examine the much larger cohort (approximately 7%) who develop shock. The elderly require individualized judgments, and it is reasonable to consider those with a good functional status and who agree to an aggressive strategy for early revascularization.

Interventions should be performed as soon as possible. However, the time window for early revascularization, as defined in the SHOCK trial, extends to shock that develops up to 36 hours after MI and revascularization within 18 hours of shock. Triple-vessel disease (60%) and left main disease (20%) are often present when shock complicates STEMI. Coronary artery bypass graft surgery is the preferred mode of revascularization for many of these patients on the basis of unsuitability for PCI and to achieve complete revascularization, unload the heart, and administer cardioprotective agents (775,776). The SHOCK trial recommended emergency CABG within 6 hours of randomization for those with severe 3-vessel or left main coronary artery disease. Among the group of patients who underwent emergency early revascularization, 60% received PCI, and 40% had CABG; the 30day mortality rate was 45% and 42%, respectively. Thirtyday outcome was similar despite more severe coronary artery disease and twice the frequency of diabetes in those who underwent CABG. This is in contrast to the 69% in-hospital mortality rate reported for those with 3-vessel disease who underwent PCI for shock (777). Perhaps distal embolization in the non-infarct-related artery is not tolerated by patients in shock. For moderate 3-vessel disease, the SHOCK trial recommended proceeding with PCI of the infarct-related artery, followed by delayed CABG for those who stabilized (Figure 26) (494,502).

It is recommended that patients who arrive at the hospital in cardiogenic shock (15% of cases) or who develop cardiogenic shock after arrival at the hospital (85%) be transferred to a regional tertiary care center with revascularization facilities and experience with these patients. (The SHOCK trial included both transferred [55%] and directly admitted patients and demonstrated the same relative treatment benefit). If skilled personnel are available, IABP placement before transport will help stabilize the patient. If the patient presents in shock within 3 to 6 hours of MI onset and delays in transport and intervention are anticipated, fibrinolytic therapy and IABP may be initiated. Nonrandomized studies suggest that this combination is beneficial (778), and a small randomized trial observed a trend toward benefit for those in classic shock, with acceptable complication rates. Fibrinolytic therapy should be administered to those patients who are not candidates for early revascularization and who do not have a contraindication to fibrinolysis.

When shock has resolved, ACE inhibitors and beta-blockers, initiated in low doses with progressive increases as recommended in the ACC/AHA Guidelines for the Evaluation and Management of Heart Failure, should be administered before discharge (771). (See Section 7.6.7.6 for discussion of mechanical support for the failing heart.)

7.6.6. Right Ventricular Infarction

Class I

- 1. Patients with inferior STEMI and hemodynamic compromise should be assessed with a right precordial V_4R lead to detect ST-segment elevation and an echocardiogram to screen for RV infarction. (See the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography (226)). (Level of Evidence: B)
- 2. The following principles apply to therapy for patients with STEMI and RV infarction and ischemic dysfunction:
 - a. Early reperfusion should be achieved if possible. (Level of Evidence: C)
 - b. Atrioventricular synchrony should be achieved, and bradycardia should be corrected. (Level of Evidence: C)
 - c. Right ventricular preload should be optimized, which usually requires initial volume challenge in patients with hemodynamic instability provided the jugular venous pressure is normal or low. (Level of Evidence: C)
 - d. Right ventricular afterload should be optimized, which usually requires therapy for concomitant LV dysfunction. (Level of Evidence: C)
 - e. Inotropic support should be used for hemodynamic instability not responsive to volume challenge. (Level of Evidence: C)

Class IIa

After infarction that leads to clinically significant RV dysfunction, it is reasonable to delay CABG surgery for 4 weeks to allow recovery of contractile performance. (*Level of Evidence: C*)

Right Ventricular Infarction and Dysfunction. Right ventricular infarction encompasses a spectrum of disease states ranging from asymptomatic mild RV dysfunction through cardiogenic shock. Most patients demonstrate a return of

normal RV function over a period of weeks to months, which suggests that RV stunning, rather than irreversible necrosis, has occurred. In this sense, RV ischemia can be demonstrated in up to half of all inferior STEMIs, although only 10% to 15% of patients show classic hemodynamic abnormalities of clinically significant RV infarction (780,781).

Right ventricular infarction with hemodynamic abnormalities accompanying inferior STEMI is associated with a significantly higher mortality (25% to 30%) and thus identifies a high-risk subgroup of patients with inferior STEMIs (6%) who should be considered high-priority candidates for reperfusion (780). One group of investigators reported a 31% inhospital mortality rate in patients with inferior STEMIs complicated by RV infarction compared with 6% in patients who had an inferior STEMI without RV involvement (780). An analysis of patients with predominant RV infarction and cardiogenic shock from the SHOCK trial registry demonstrated an unexpectedly high mortality rate similar to that for patients with LV shock (53.1% versus 60.8%) (782). The treatment of patients with RV ischemic dysfunction is different and, in several ways, diametrically opposed to management of LV dysfunction.

Anatomic and Pathophysiological Considerations. The right coronary artery usually supplies most of the RV myocardium; thus, occlusion of this artery proximal to the RV branches will lead to RV ischemia (783). Hemodynamically significant RV infarctions occur almost exclusively in the setting of inferior STEMIs (784). Because the RV has a much smaller muscle mass than the LV, owing to the lower vascular resistance of the pulmonary circuit, myocardial oxygen demand is significantly less than that of the LV (785). Coronary perfusion of the RV occurs in both systole and diastole (785). The RV also has a more favorable oxygen supply-demand ratio than the LV because of the more extensive collateral flow from left to right (786,787). These factors likely explain the absence of hemodynamically significant RV ischemia in most patients with proximal right coronary artery occlusions, as well as improvement in RV function observed in the majority of patients after RV ischemia (788).

The severity of the hemodynamic derangements associated with RV ischemia is related to 1) the extent of ischemia and subsequent RV dysfunction, 2) the restraining effect of the surrounding pericardium, and 3) interventricular dependence related to the shared interventricular septum. When the RV becomes ischemic, it dilates acutely, which results in increased intrapericardial pressure caused by the restraining forces of the pericardium. As a consequence, there is a reduction in RV systolic pressure and output, decreased LV preload, a reduction in LV end-diastolic dimension and stroke



Figure 30. Right ventricular infarction. JVP = jugular venous pressure; RA = right atrial; RV = right ventricular; PA = pulmonary artery; PCW = pulmonary capillary wedge. AV = atrioventricular. Modified with permission from Wellens. N Engl J Med 1999;340:381. Copyright © 1999 Massachusetts Medical Society. All rights reserved (803).

volume, and a shifting of the interventricular septum toward the LV (789). Because of this RV systolic and diastolic dysfunction, the pressure gradient between the right and left atria becomes an important driving force for pulmonary perfusion. Factors that reduce preload (volume depletion, diuretics, nitrates) or diminish augmented right atrial contraction (concomitant atrial infarction, loss of AV synchrony) and factors that increase RV afterload (concomitant LV dysfunction) are likely to have profoundly adverse hemodynamic effects (790-794). Goldstein and coworkers (791,794) demonstrated the importance of a paradoxical interventricular septal motion that bulges in piston-like fashion into the RV, generating systolic force, which allows pulmonary perfusion. The loss of this compensatory mechanism, with concomitant septal infarction, may result in further deterioration in patients with RV ischemia.

Clinical Diagnosis. Evidence of RV ischemia/infarction should be sought in all patients with inferior STEMI. The clinical triad of hypotension, clear lung fields, and elevated jugular venous pressure in the setting of a STEMI is characteristic of RV ischemia/infarction. Although specific, this triad has a sensitivity of less than 25% (795). Distended neck veins alone or the presence of Kussmaul's sign (distention of the jugular vein on inspiration) are both sensitive and specific for RV ischemia/infarction in patients with a STEMI (796). These findings may be masked in the setting of volume depletion and may only become evident after adequate volume loading. Right heart catheterization may be helpful in diagnosing RV ischemia/infarction. A right atrial pressure of 10 mm Hg or greater and greater than 80% of pulmonary wedge pressure is a relatively sensitive and specific finding in patients with RV ischemia/infarction (797).

Patients with RV hypertrophy, commonly due to the pulmonary hypertension associated with chronic obstructive pulmonary disease, have increased myocardial demand and may be more likely to suffer RV MI. Demonstration of 1-mm ST-segment elevation in lead V_1 and in the right precordial lead V₄R is the single most predictive ECG finding in patients with RV ischemia (798). The finding may be transient; half of patients show resolution of ST elevation within 10 hours of onset of symptoms (799). It is important for physicians to ensure that hospital personnel (house officer, nurse, technician) recording the ECG in this setting know how to properly record lead $V_4 R$, especially in view of the variety of multilead recording systems available. All patients with inferior STEMI should be screened initially for this finding at the time of admission. Echocardiography can be helpful in patients with suspicious but nondiagnostic findings (226). It can show RV dilation and asynergy, abnormal interventricular and interatrial septal motion, and even right to left shunting through a patent foramen ovale (800-802). Right to left shunting should be suspected when persistent hypoxia is not responsive to supplemental oxygen (802).

Management of RV Ischemia/Infarction. Treatment of RV ischemia/infarction includes early maintenance of RV preload, reduction of RV afterload, inotropic support of the dysfunctional RV, and early reperfusion (Figure 30) (257,803). Because of their influence on preload, drugs routinely used in management of LV infarctions, such as nitrates and diuretics, may reduce cardiac output and produce severe hypotension when the RV is ischemic. Indeed, a common clinical presentation is profound hypotension after administration of sublingual nitroglycerin, with the degree of hypotension often out of proportion to the ECG severity of the infarct. Volume loading with normal saline alone often resolves accompanying hypotension and improves cardiac output (804). The Trendelenburg position may effectively raise preload in patients who develop hypotension after vasodilator administration. Although volume loading is a critical first step in the management of hypotension associated with RV ischemia/infarction, inotropic support (in particular, dobutamine hydrochloride) should be initiated promptly if cardiac output fails to improve after 0.5 to 1 L of fluid have been given. Excessive volume loading may further alleviate the right-sided filling pressure and RV dilatation, resulting in decreased LV output (805-807) through shift of the interventricular septum for RV toward the LV.

Another important factor for sustaining adequate RV preload is maintenance of AV synchrony. High-degree heart block is common, occurring in as many as half of these patients (808). Atrioventricular sequential pacing leads to a significant increase in cardiac output and reversal of shock, even when ventricular pacing alone has not been of benefit (806). Atrial fibrillation may occur in up to one third of patients with RV ischemia/infarction (807) and has profound hemodynamic effects. Prompt cardioversion from AF should be considered at the earliest sign of hemodynamic compromise. When LV dysfunction accompanies RV ischemia/ infarction, the RV is further compromised because of increased RV afterload and reduction in stroke volume (809). In such circumstances, the use of afterload-reducing agents or an intra-aortic counterpulsation device is often necessary to unload the LV and subsequently the RV. Fibrinolytic therapy and primary PCI with subsequent reperfusion have been shown to improve RV ejection fraction (793,810) and reduce the incidence of complete heart block (810-812).

Right ventricular failure secondary to an ischemic RV (either infarction or stunning) presents a particularly hazardous situation (813). The prototypical patient has an occluded right coronary artery proximal to the major RV branches and presents with an inferior MI with or without recognized RV failure (784,795,814-821). Angiography may demonstrate that the coronary anatomy is best treated surgically, but the opportunity for maximal benefit of an emergency operation (initial 4 to 6 hours) has often passed. There is substantial risk in operating after this small window of opportunity but before the recovery of RV function, which usually occurs at 4 weeks after injury (793). During this postinfarct month, the RV is at great risk for severe postoperative dysfunction, which often requires extraordinary levels of perioperative pharmacological and mechanical support and is associated with a very high mortality rate. The nonsurgical postinfarction patient can most often be supported

Table 25. Characteristics of Ventricul Characteristic	lar Septal Rupture (VSR), Rupture of the Ventricular Free We VSR	all, and Papillary Muscle Rupture Rupture of Ventricular Free Wall	Papillary Muscle Rupture
Incidence	1-3% without reperfusion therapy,0.2-0.34% with fibrinolytic therapy,3.9% among patients with cardiogenic shock	0.8-6.2%, Fibrinolytic therapy does not reduce risk; primary PTCA seems to reduce risk	About 1% (posteromedial more frequent than anterolateral papillary muscle)
Time course	Bimodal peak; within 24 hours and 3-5 days; range 1-14 days	Bimodal peak; within 24 hours and 3-5 days; range 1-14 days	Bimodal peak; within 24 hours and 3-5 days; range 1-14 days
Clinical manifestations	Chest pain, shortness of breath, hypotension	Anginal, pleuritic, or pericardial chest pain, syncope, hypotension, arrhythmia, nausea, restlessness, hypotension, sudden death	Abrupt onset of shortness of breath and pulmonary edema; hypotension
Physical findings	Harsh holosystolic murmur, thrill (+), S ₃ , accentuated 2nd heart sound, pulmonary edema, RV and LV failure, cardiogenic shock	Jugulovenous distention (29% of patients), pulsus paradoxus (47%), electromechanical dissociation, cardiogenic shock	A soft murmur in some cases, no thrill, variable signs of RV overload, severe pulmonary edema, cardiogenic shock
Echocardiographic findings	VSR, left-to-right shunt on color flow Doppler echocardiography through the ventricular septum, pattern of RV overload	Greater than 5 mm pericardial effusion not visualized in all cases, layered, high-acoustic echoes within the pericardium (blood clot), direct visualization of tear, signs of tamponade	Hypercontractile LV, torn papillary muscle or chordae tendineae, flail leaflet, severe MR on color flow Doppler echocardiography
Right-heart catheterization	Increase in oxygen saturation from the RA to RV, large V waves	Ventriculography insensitive, classic signs of tamponade not always present (equalization of diastolic pressures among	No increase in oxygen saturation from the RA to RV, large V waves,* very high pulmonary-capillary wedge

ACC - www.acc.org AHA - www.americanheart.org

Ventricular Septal Rupture

Free Wall Rupture

Mitral Regurgitation (Papillary muscle rupture)



Figure 31. Mechanical complications of ST-elevation myocardial infarction (STEMI). See Table 25 for additional information. Images courtesy of William D. Edwards, M.D. (Mayo Clinic).

with pacing, volume loading, and judicious inotropic administration (792). In the surgical setting, the RV takes on different characteristics. There is loss of the pericardial constraint immediately on exposing the heart, which results in acute dilation of the dysfunctional RV. The RV often fails to recover in this setting, even when state-of-the-art myocardial protection schemes and revascularization are used (822). The parallel effects of RV dilatation and dysfunction on LV diastolic and systolic function are magnified and may be associated with the need for high levels of support, inability to close the chest owing to cardiac dilation, need for ventricular assist devices, prolonged convalescence, transplantation, or death (792).

The best defense is an index of suspicion and recognition of the RV dysfunction by physical examination (795,810) ECG (right precordial leads), echocardiography, or radionuclide-gated blood pool study (793,810,823,824). If early PCI of the right coronary artery is indicated on the basis of angiography, this should be performed promptly. It is reasonable to delay coronary bypass surgery for 4 weeks to allow recovery of RV function.

Prognosis. The mere presence of RV ischemia/infarction that is evident by noninvasive criteria is associated with significantly increased short-term morbidity and mortality and may also influence long-term outcome (780,812,825). However, clinical and hemodynamic recovery often occur even in patients with RV dysfunction that persists for weeks or months (796,826-828). This return to normal may be due to improvement of concomitant LV dysfunction, which results in a reduction in RV afterload, or to a gradual stretching of the pericardium with amelioration of its restraining effect (826).

7.6.7. Mechanical Causes of Heart Failure/ Low-Output Syndrome

7.6.7.1. Diagnosis

Mechanical defects, when they occur, usually present within the first week after STEMI. On physical examination, the presence of a new cardiac murmur indicates the possibility of either a VSR or MR. Detailed characteristics of these mechanical defects are listed in Table 25 (829) (Figure 31). A precise diagnosis can usually be established with transthoracic or transesophageal echocardiography. A pulmonary artery monitoring catheter may also be useful in establishing the diagnosis of a mechanical defect and in its subsequent management. In VSR, oxygen saturation will be higher ("step-up") in the pulmonary artery than in the right atrium. With acute MR, a large C-V wave may be evident on the pulmonary artery wedge pressure tracing. However, a prominent V wave does not necessarily indicate the presence of MR and may also be present in patients with severe LV dysfunction associated with decreased left atrial compliance (830). A V wave may also be seen with VSR. In patients with free-wall ventricular rupture and subsequent pericardial tamponade, equalization of diastolic pressures may be seen.

Surgical consultation should be obtained when a mechanical defect is suspected so that an early decision regarding surgical management can be made. In general, prompt surgical repair is indicated in most cases, because medical treatment alone is associated with an extremely high mortality. Insertion of an IABP, particularly in patients with papillary muscle rupture or VSR, can help stabilize the patient. Although there is a need to minimize invasive procedures before early surgical correction of mechanical complications, initial coronary angiography to assess coronary anatomy appears warranted in most cases of VSR and papillary muscle rupture. However, the evidence for the benefit of concomitant CABG associated with surgical repair of acute VSR is inconclusive (831). In the majority of patients, right and left heart catheterization are unnecessary unless other studies (e.g., echocardiography) are not clear in demonstrating a mechanical defect.

7.6.7.2. Mitral Valve Regurgitation

Class I

1. Patients with acute papillary muscle rupture should be considered for urgent cardiac surgical repair, unless further support is considered futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (*Level of Evidence: B*)

2. Coronary artery bypass graft surgery should be undertaken at the same time as mitral valve surgery. (Level of Evidence: B)

Severe MR after STEMI, accompanied by cardiogenic shock, has a poor prognosis. In the SHOCK trial registry, approximately 10% of patients with shock presented with severe MR and had an overall hospital mortality of 55% (832). Mortality with medical treatment only was 71% compared with 40% with surgery (832). In the Survival and Ventricular Enlargement (SAVE) trial, in which patients were treated with an ACE inhibitor after MI, even patients with mild MR experienced a worse prognosis than those without MR (833).

Severe MR may be due to infarction of the posterior papillary muscle, and in such instances, the area of infarction tends to be less extensive than in those patients in whom the MR is due to severe LV dysfunction. Consequently, LV function may also be better preserved in these patients. The presence of pulmonary edema or cardiogenic shock in a patient with inferior STEMI should alert the physician to the possibility of acute MR and papillary muscle rupture (Figure 31). Diagnosis is made by transthoracic or transesophageal echocardiography. All patients with papillary muscle rupture should be considered for urgent surgery. The patient should be stabilized with an IABP, inotropic support, and afterload reduction (to reduce regurgitant volume and pulmonary congestion) while emergency surgery is arranged. Coronary angiography should also be undertaken before surgery. Although emergency mitral valve replacement is associated with a relatively high mortality rate (20%), overall mortality and ventricular function are improved compared with medical therapy alone. Delay in operation appears to increase the risk of further myocardial injury, other organ injury, and subsequent death (834). Most patients will require mitral valve replacement, although mitral valve repair has also been reported in selected circumstances. Five-year survival after surgery has been reported to be 60% to 70% (835-839).

Severe MR, in the absence of papillary muscle rupture, often indicates extensive infarction and severe LV dysfunction. These patients may present a much more difficult management problem, particularly if surgery is required. Initial management should include afterload reduction and possible IABP. In many cases, the MR will improve over the next several days with aggressive medical management. If surgery is required because of critical coronary anatomy or ongoing ischemia, an intraoperative transesophageal echo should be undertaken to assess the mitral valve. Mitral valve surgery, usually annuloplasty, should be undertaken at the same time as CABG for patients with ischemic MR greater than 2+ (840,841). Clearly, operative mortality is increased in such patients, particularly in the elderly, with a marked decrease in LV function.

7.6.7.3. Ventricular Septal Rupture After STEMI

Class I

- 1. Patients with STEMI complicated by the development of a VSR should be considered for urgent cardiac surgical repair, unless further support is considered futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (Level of Evidence: B)
- 2. Coronary artery bypass grafting should be undertaken at the same time as repair of the VSR. (Level of Evidence: B)

The frequency of acute rupture of the intraventricular septum (VSR) (Figure 31) appears to have declined in the reperfusion era. It is estimated to occur in fewer than 1% of patients with STEMI (GUSTO-I) (829,842). Whereas previous pathological and clinical studies indicated the mean time from MI to rupture as 3 to 5 days, data from the GUSTO-I trial and SHOCK registry indicate that the highest risk for development of a postinfarct VSR occurred within the first 24 hours after infarction in patients receiving fibrinolytic therapy (842,843). Although emergency surgical repair was formerly thought to be necessary only in patients with pulmonary edema or cardiogenic shock, it is now recognized as equally important in hemodynamically stable patients (844-846). Because all septal perforations are exposed to shear forces and necrotic tissue removal processes by macrophages, the rupture site can abruptly expand, resulting in sudden hemodynamic collapse even in patients who appear to be clinically stable with normal LV function (846). Insertion of an IABP and prompt surgical referral are recommended for almost every patient with an acute VSR. Invasive monitoring is recommended in all patients, together with judicious use of inotropes and a vasodilator to maintain optimal hemodynamics. Nitroprusside is often used because it provides afterload reduction and can be titrated intravenously. Surgical repair usually involves excision of all necrotic tissue and patch repair of the VSR, together with coronary artery grafting. Surgical mortality remains high and has been reported to be between 20% and 50% (842-845,847,848). Mortality is particularly high in patients with cardiogenic shock (844,849) and was reported to be 87% in the SHOCK registry. However, surgical mortality is significantly less than for medically treated patients. In GUSTO-I (842), the mortality rates for surgical or medically treated patients were 47% and 94%, respectively.

A limited number of patients with postinfarction VSR have been treated by transcatheter closure with a septal occluding device. Most of these cases have been managed several weeks after infarction or have had prior surgical intervention with a residual defect. At this time, surgical closure remains the procedure of choice, although percutaneous closure does offer some hope for the future (850).

7.6.7.4. Left Ventricular Free-Wall Rupture

Class I

- **1.** Patients with free-wall rupture should be considered for urgent cardiac surgical repair, unless further support is considered futile because of the patient's wishes or contraindications/unsuitability for further invasive care. (*Level of Evidence: B*)
- 2. CABG should be undertaken at the same time as repair of free wall rupture. (Level of Evidence: C)

Cardiac rupture may account for recurrent pain and occurs in 1% to 6% of all patients admitted with STEMI (829,851-854) (Figure 31). Left ventricular free-wall rupture is typically heralded by chest pain and ECG ST-T-wave changes, with rapid progression to hemodynamic collapse and electromechanical dissociation. The frequency of cardiac rupture has 2 peaks: an early peak within 24 hours and a late one from 3 to 5 days after STEMI. Early rupture is related to the initial evolution of infarction before significant collagen deposition, and late rupture is related to expansion of the infarctrelated ventricular wall (852,855). Cardiac rupture is observed most frequently in patients with their first MI, those with anterior infarction, the elderly, and women. Other risk factors include hypertension during the acute phase of STEMI, lack of previous angina and MI, lack of collateral blood flow, Q waves on the ECG, use of corticosteroids or nonsteroidal anti-inflammatory drugs (NSAIDs), and use of fibrinolytic therapy more than 14 hours after onset (854,855). Fibrinolytic therapy decreases risk of cardiac rupture (853,856). Although there is an increase in the risk of early rupture after late administration of fibrinolytic therapy (i.e., more than 14 hours), the overall incidence of rupture is reduced. The most important determinants in preventing rupture are successful early reperfusion and the presence of collateral circulation (852,853). Pseudoaneurysm is a serious complication after rupture of the free wall. Clot forms in the pericardial space, and an aneurysmal wall containing clot and pericardium prevents exsanguination. Prompt surgical correction is always indicated for pseudoaneurysm to prevent rupture.

Pericardiocentesis for relief of tamponade and emergency surgical repair may be lifesaving (857,858). Ideally, the pericardium in this case should be opened surgically or tapped in the operating room. Echocardiography is valuable in the diagnosis of free-wall rupture and pseudoaneurysm, but for relief of tamponade in this setting, rapid fluid replacement is essential. Ideally, the patient should be in the operating room as soon as possible and fully prepared for cardiopulmonary bypass to prevent hemodynamic collapse. In these circumstances, delay to perform coronary angiography is inadvisable (859).

Surgery includes repair of the ventricle with a direct suture technique or patch to cover the ventricular perforation (857), in addition to CABG as needed. Alternatively, the use of cyanoacrylate glue has been described to hold the patch in place over necrotic myocardium (860). Most series of patients reaching the operating room for management of this complication are small, with the surgical mortality rate in these patients being up to 60% (859,861).

7.6.7.5. Left Ventricular Aneurysm

Class IIa

It is reasonable that patients with STEMI who develop a ventricular aneurysm associated with intractable ventricular tachyarrhythmias and/or pump failure unresponsive to medical and catheterbased therapy be considered for LV aneurysmectomy and CABG surgery. (Level of Evidence: B)

Ventricular aneurysm after STEMI usually occurs on the anterior aspect of the LV in association with total LAD occlusion and a wide area of infarction. Clinical consequences include angina pectoris, CHF, thromboembolism, and ventricular arrhythmias. Patients with STEMI who receive fibrinolytic therapy and exhibit a patent infarct-related artery have a significantly reduced incidence of LV aneurysm formation compared with those who do not (7.2%)versus 18.8%) (862). The need for surgery for ventricular aneurysm early after STEMI is rare, but it may be necessary for control of heart failure or intractable ventricular arrhythmias unresponsive to conventional therapy (863). Surgical techniques include plication, excision with linear repair, and ventricular reconstruction with endoventricular patches to maintain better physiological function (863-865). The adequacy of the residual LV in terms of size and function is a critical determinant on prognosis. Current mortality rates are reported to be 3.3% to 7.2% (863,864). Patients with severe LV dysfunction have an increase in mortality that has been reported to be as high as 19% for an ejection fraction less than 0.20 (866). Operative survivors have clear improvement in New York Heart Association class and a 60% 5-year survival rate (867).

7.6.7.6. Mechanical Support of the Failing Heart

7.6.7.6.1. INTRA-AORTIC BALLOON COUNTERPULSATION.

Class I

- 1. Intra-aortic balloon counterpulsation should be used in STEMI patients with hypotension (systolic blood pressure less than 90 mm Hg or 30 mm Hg below baseline mean arterial pressure) who do not respond to other interventions, unless further support is futile because of the patient's wishes or contraindications/ unsuitability for further invasive care. See Section 7.6.2. (Level of Evidence: B)
- 2. Intra-aortic balloon counterpulsation is recommended for STEMI patients with low-output state. See Section 7.6.3. (Level of Evidence: B)
- 3. Intra-aortic balloon counterpulsation is recommended for STEMI patients when cardiogenic shock is not quickly reversed with pharmacological therapy. IABP

is a stabilizing measure for angiography and prompt revascularization. See Section 7.6.5. (Level of Evidence: B)

4. Intra-aortic balloon counterpulsation should be used in addition to medical therapy for STEMI patients with recurrent ischemic-type chest discomfort and signs of hemodynamic instability, poor LV function, or a large area of myocardium at risk. Such patients should be referred urgently for cardiac catheterization and should undergo revascularization as needed. See Section 7.8.2. (Level of Evidence: C)

Class IIa

It is reasonable to manage STEMI patients with refractory polymorphic VT with intra-aortic balloon counterpulsation to reduce myocardial ischemia. See Section 7.7.1.2. (Level of Evidence: B)

Class IIb

It may be reasonable to use intra-aortic balloon counterpulsation in the management of STEMI patients with refractory pulmonary congestion. See Section 7.6.4. (Level of Evidence: C)

The IABP improves diastolic coronary blood flow and reduces myocardial work. These physiological effects of the IABP are especially helpful in patients with STEMI with ongoing or recurrent ischemic discomfort, hypotension from ischemia-mediated LV dysfunction, and cardiogenic shock (see section 7.6.5). The IABP is a useful stabilizing measure for patients in whom cardiac catheterization and revascularization are being considered.

Selected patients with cardiogenic shock after STEMI, especially if not candidates for revascularization, may be considered for either a short- or long-term mechanical support device to serve as a bridge to recovery or to subsequent cardiac transplantation. Of the many devices available to support these patients (868), short-term devices include centrifugal pumps and LV assist devices (LVADs) (869). Extracorporeal membrane oxygenation (ECMO), a cardiopulmonary bypass system placed through either the femoral or intrathoracic vessels, serves patients with heart failure and concomitant respiratory failure. These systems are limited by their short-term usefulness of less than 1 week and by problems with bleeding and thrombosis. Some consider ECMO a poor method of support because it does not decompress the LV (869). Patients who are subsequently determined to be transplant candidates may also be converted to a bridge-to-transplant device, thus creating a bridge to a bridge. About 10% of all patients treated with LVADs have them inserted for hemodynamic support after STEMI. This application has not been widely used because of the many comorbidities encountered in such patients, many of whom die before surgery. Experience with LVADs implanted in selected patients within 14 days after infarction has shown a survival rate of 74% to transplantation or explantation (870). This experience suggests that ventricular assist device implantation for cardiogenic shock after STEMI may reduce

the mortality currently associated with medical management. The use of assist devices has been extensively reviewed in a separate ACC consensus conference report (868).

7.6.7.7. Cardiac Transplantation After STEMI

Cardiac transplantation has been reported for patients who have sustained irreversible acute myocardial injury associated with cardiogenic shock (871-873). Most patients so transplanted have been initially treated with an LVAD as a bridge to recovery or transplantation. Many such patients, however, do not meet criteria for transplantation because of advanced age and extensive comorbidity. With appropriate case selection, satisfactory results can be obtained. Of 25 patients with post-STEMI cardiogenic shock and a mean age of 48 years, 3 died while on devices and 18 (72%) survived transplantation (873).

7.7. Arrhythmias After STEMI

Cardiac arrhythmias are common in patients with STEMI and occur most frequently early after development of symptoms. The mechanisms for ventricular tachyarrhythmias include loss of transmembrane resting potential, reentrant mechanisms due to dispersion of refractoriness in the border zones between infarcted and nonischemic tissues (874), and the development of foci of enhanced automaticity. Reperfusion arrhythmias, more commonly seen in the postfibrinolytic era, appear to involve washout of toxic metabolites and of various ions such as lactate and potassium (875). Atrial arrhythmias have as additional causes excessive sympathetic stimulation, increased atrial stretch due to ventricular failure or AV valvular insufficiency, proarrhythmic effects of pericarditis, and atrial infarction. Bradyarrhythmias may be due to overstimulation of vagal afferent receptors and resulting cholinergic stimulation, as well as to ischemic injury of conducting tissues. The treatment of cardiac arrhythmias is based on the presumptive mechanism, the ongoing hemodynamic consequences, and, whenever possible, the results of clinical studies.

7.7.1. Ventricular Arrhythmias

7.7.1.1. Ventricular Fibrillation

Class I

Ventricular fibrillation or pulseless VT should be treated with an unsynchronized electric shock with an initial monophasic shock energy of 200 J; if unsuccessful, a second shock of 200 to 300 J should be given, and then, if necessary, a third shock of 360 J. (*Level of Evidence: B*)

Class IIa

1. It is reasonable that VF or pulseless VT that is refractory to electric shock be treated with amiodarone (300 mg or 5 mg/kg, IV bolus) followed by a repeat unsynchronized electric shock. (*Level of Evidence: B*) 2. It is reasonable to correct electrolyte and acid-base disturbances (potassium greater than 4.0 mEq/L and magnesium greater than 2.0 mg/dL) to prevent recurrent episodes of VF once an initial episode of VF has been treated. (*Level of Evidence: C*)

Class IIb

It may be reasonable to treat VT or shock-refractory VF with boluses of intravenous procainamide. However, this has limited value owing to the length of time required for administration. (Level of Evidence: C)

Class III

Prophylactic administration of antiarrhythmic therapy is not recommended when using fibrinolytic agents. (Level of Evidence: B)

Disturbances of cardiac rhythm are common during STEMI. Early-phase arrhythmias are probably largely a result of microreentry. Although other electrophysiological mechanisms such as enhanced automaticity and triggered activity have been proposed in experimental models of STEMI, convincing evidence for their role in human STEMI is not yet established (876). Important contributory factors include heightened adrenergic nervous system tone, hypokalemia, hypomagnesemia, intracellular hypercalcemia, acidosis, free fatty acid production from lipolysis, and free radical production from reperfusion of ischemic myocardium (876-878). The relative importance of each of these factors in the pathogenesis of arrhythmias during STEMI has not been established, nor has it been clearly shown that aggressive measures specifically targeted at 1 or more of these mechanisms can be relied on clinically to reduce arrhythmia frequency in STEMI.

Primary VF should be distinguished from secondary VF, the latter occurring in the presence of severe CHF or cardiogenic shock (879). Late VF develops more than 48 hours after onset of STEMI. Ventricular fibrillation is more common in the elderly (greater than 75 years of age) (880). The incidence of primary VF is highest (around 3% to 5%) in the first 4 hours after STEMI and declines markedly thereafter (881). Some epidemiological data suggest that the incidence of primary VF in STEMI may be decreasing in the current era, possibly owing to aggressive attempts at infarct-size reduction, correction of electrolyte deficits, and a greater use of beta-adrenoceptor-blocking agents (882). Additional epidemiological data from the Worcester Heart Attack Study more convincingly demonstrate that the case-fatality rate of primary VF has declined over time (883). Contrary to prior belief, primary VF appears to be associated with significantly higher in-hospital mortality, but those persons who survive to hospital discharge, particularly if primary VF occurred within the first 4 hours after STEMI, have the same longterm prognosis as patients who do not experience primary VF (884).

Primary VF remains an important contributor to risk of mortality during the first 24 hours after STEMI. Therefore, a

reliable method for its prediction and prevention remains desirable but has not been established despite extensive clinical investigation. Classification of ventricular arrhythmias in ascending order of risk of primary VF (warning arrhythmias) was proposed, but this approach lacks appropriate specificity and sensitivity (885-887).

Accelerated idioventricular rhythm occurs frequently during the first 12 hours of infarction. Data from the prereperfusion era do not support development of accelerated idioventricular rhythm as a risk factor for development of VF (886,888). In patients receiving fibrinolysis or undergoing primary PCI, accelerated idioventricular rhythm may be a reperfusion arrhythmia and does not indicate an increased risk of VF (889). Thus, it is best managed by observation and should not trigger initiation of antiarrhythmic prophylaxis against VF.

A meta-analysis of randomized trials of prophylaxis with lidocaine has shown a relative reduction in the incidence of primary VF by about 33%, but this was offset by a trend toward increased mortality, probably from fatal episodes of bradycardia and asystole (890). The use of prophylactic lidocaine was assessed in patients with STEMI in the GUSTO-I and GUSTO-IIb trials (891). After adjustment for baseline imbalances, the odds of death were not significantly different with or without lidocaine. Thus, even though it is less clear that lidocaine causes harm, there is no convincing evidence that its prophylactic use reduces mortality, and the prior practice of routine (prophylactic) administration of lidocaine to all patients with known or suspected STEMI has been largely abandoned.

Routine administration of intravenous beta-adrenoceptor blockers to patients without hemodynamic or electrical (AV block) contraindications is associated with a reduction in incidence of early VF (892). In the absence of contraindications (see Section 6.3.1.5), it is reasonable to initiate betablockade intravenously, followed by an oral regimen. Suitable regimens include intravenous metoprolol at 5 mg every 2 minutes for 3 doses, if tolerated, followed by 50 mg orally twice per day for at least 24 hours and then increased to 100 mg twice per day. An alternative regimen is atenolol 5 to 10 mg IV followed by 100 mg orally on a daily basis.

Clinical experience and observational data from CCU populations have identified hypokalemia as an arrhythmogenic risk factor for VF (877,878). Low serum levels of magnesium have not been clearly shown to be associated with an increased risk of VF (878), although tissue depletion of magnesium remains a potential risk factor. Although randomized clinical trial data do not exist to confirm the benefits of repletion of potassium and magnesium deficits in preventing VF, it is sound clinical practice to maintain serum potassium levels at greater than 4.0 mEq/L and magnesium levels at greater than 2.0 mEq/L in patients with acute MI.

Ventricular fibrillation should be treated with an unsynchronized electric shock using an initial monophasic shock energy of 200 J. If this is unsuccessful, a second shock using 200 to 300 J and, if necessary, a third shock using 360 J is indicated (893). The appearance of biphasic waveform defib-
rillators has led to some confusion regarding the comparability of biphasic to monophasic defibrillating energies. Overall, the energy requirement for equivalent biphasic therapeutic effect is about one half of the energy requirement for a monophasic discharge. Given the rapid evolution of resuscitation methodology, clinicians should follow the most current ACLS protocol (629). For example, for patients with VF not easily converted by defibrillation, vasopressin 40 U IV push may be substituted for epinephrine 1 mg. Randomized trials have addressed the use of intravenous amiodarone versus placebo and versus lidocaine for patients with out-ofhospital cardiac arrest. Although this population is somewhat different from the CCU population with primary VF, many patients with out-of-hospital VF have STEMI as the precipitating cause (894). These randomized trials have shown that the use of intravenous amiodarone is superior to placebo (895) and to lidocaine (896) in survival to hospital admission for patients with shock-resistant VF or VT. Neither trial, however, demonstrated improved survival to hospital discharge. In contrast, small studies comparing lidocaine to bretylium failed to show significant differences in the proportion of patients surviving to hospital admission (897,898); the use of procainamide in cardiac arrest is based on a small, 20-patient study (899). The use of lidocaine to treat VF refractory to electric shock or pulseless VT followed by unsynchronized electric shock has not been demonstrated to be beneficial in this setting and has been labeled "class indeterminate" by the Guidelines 2000 for Cardiopulmonary Resuscitation and Emergency Cardiovascular Care (900).

There are no firm data to help define an optimal management strategy for prevention of recurrent VF in patients who have sustained an initial episode of VF in the setting of STEMI. It seems prudent to correct any electrolyte and acidbase disturbances and to administer beta-adrenoceptorblocking agents to inhibit increased sympathetic nervous system tone and prevent ischemia (876). The clinical trials of amiodarone cited above studied bolus administration of amiodarone only during the arrest setting. Thus, in the absence of arrhythmia recurrence, antiarrhythmic drugs should not be maintained beyond a 6- to 24-hour period. They should then be discontinued so that the patient's ongoing need for antiarrhythmic treatment can be reassessed.

7.7.1.2. Ventricular Tachycardia

Class I

- 1. Sustained (more than 30 seconds or causing hemodynamic collapse) polymorphic VT should be treated with an unsynchronized electric shock with an initial monophasic shock energy of 200 J; if unsuccessful, a second shock of 200 to 300 J should be given, and, if necessary, a third shock of 360 J. (Level of Evidence: B)
- 2. Episodes of sustained monomorphic VT associated with angina, pulmonary edema, or hypotension (blood pressure less than 90 mm Hg) should be treated with a synchronized electric shock of 100 J of initial monophasic shock energy. Increasing energies may be

used if not initially successful. Brief anesthesia is desirable if hemodynamically tolerable. (Level of Evidence: B)

- **3.** Sustained monomorphic VT not associated with angina, pulmonary edema, or hypotension (blood pressure less than 90 mm Hg) should be treated with:
 - a. Amiodarone: 150 mg infused over 10 minutes (alternative dose 5 mg/kg); repeat 150 mg every 10 to 15 minutes as needed. Alternative infusion: 360 mg over 6 hours (1 mg/min), then 540 mg over the next 18 hours (0.5 mg/min). The total cumulative dose, including additional doses given during cardiac arrest, must not exceed 2.2 g over 24 hours. (Level of Evidence: B)
 - b. Synchronized electrical cardioversion starting at monophasic energies of 50 J (brief anesthesia is necessary). (Level of Evidence: B)

Class IIa

- It is reasonable to manage refractory polymorphic VT by:
- a. Aggressive attempts to reduce myocardial ischemia, and adrenergic stimulation, including therapies such as beta-adrenoceptor blockade, IABP use, and consideration of emergency PCI/CABG surgery. (Level of Evidence: B)
- **b.** Aggressive normalization of serum potassium to greater than 4.0 mEq/L and of magnesium to greater than 2.0 mg/dL. (*Level of Evidence: C*)
- c. If the patient has bradycardia to a rate less than 60 bpm or long QTc, temporary pacing at a higher rate may be instituted. (*Level of Evidence: C*)

Class IIb

It may be useful to treat sustained monomorphic VT not associated with angina, pulmonary edema, or hypotension (blood pressure less than 90 mm Hg) with a procainamide bolus and infusion (*Level of Evidence:* C)

Class III

- 1. The routine use of prophylactic antiarrhythmic drugs (i.e., lidocaine) is not indicated for suppression of isolated ventricular premature beats, couplets, runs of accelerated idioventricular rhythm, and nonsustained VT. (Level of Evidence: B)
- 2. The routine use of prophylactic antiarrhythmic therapy is not indicated when fibrinolytic agents are administered. (Level of Evidence: B)

Several definitions have been used for VT in the setting of STEMI. Nonsustained VT lasts less than 30 seconds, whereas sustained VT lasts more than 30 seconds and/or causes earlier hemodynamic compromise that requires immediate intervention. On the basis of ECG appearance, VT has also been categorized as monomorphic or polymorphic. Although short bursts (fewer than 5 beats) of nonsustained VT of either monomorphic or polymorphic configuration may be seen frequently, contemporary epidemiological data do not suggest that they are associated with a sufficiently increased risk of sustained VT or VF to warrant prophylactic therapy.

The vast majority of episodes of VT and VF after STEMI occur within the first 48 hours (881). Traditionally, sustained VT or VF that occurs outside of this time frame is thought to deserve especially careful evaluation, including consideration of electrophysiology (EP) studies. In addition, monomorphic VT at rates less than 170 bpm is unusual as an arrhythmia early after STEMI and suggests a more chronic (mature) arrhythmogenic substrate (515,901-903). VT that occurs more than 48 hours after STEMI may denote an arrhythmic substrate deserving of further evaluation by an EP study.

MANAGEMENT STRATEGIES FOR VT. Cardioversion is always indicated for episodes of sustained, hemodynamically compromising VT (876). In the absence of clinical evidence of effective perfusion, urgent electrical conversion of VT is indicated. Rapid, polymorphic-appearing VT should be considered similar to VF and managed with an unsynchronized discharge of 200 J, whereas monomorphic VT with rates greater than 150 bpm can usually be treated with a 100-J synchronized discharge (893). Immediate cardioversion is generally not needed for rates below 150 bpm unless hemodynamic compromise is present.

Episodes of sustained VT that are somewhat better tolerated hemodynamically may initially be treated with drug regimens including amiodarone or procainamide. Unfortunately, the data supporting the use of any specific antiarrhythmic therapy in this setting are scant. Although the Guidelines 2000 for Cardiopulmonary Resuscitation and Emergency Cardiac Care differentiate between patients with normal and impaired LV function, impairment of LV function is likely in the STEMI setting. Thus, recommendations for patients with STEMI parallel the Guidelines 2000 for Cardiopulmonary Resuscitation and Emergency Cardiac Care for patients with impaired LV function and place the use of amiodarone above that of other antiarrhythmics (900). Indeed, amiodarone has produced favorable data in cardiac arrest, and in long-term randomized trials, it is associated with a reduction in arrhythmic death and a small reduction in overall mortality (904). Knowledge of the pharmacokinetics of antiarrhythmic agents in patients with STEMI is important because dosing varies considerably, depending on age, weight, and hepatic and renal function.

Rare episodes of drug-refractory sustained polymorphic VT (electrical storm) have been reported in cases of STEMI. Anecdotal evidence suggests that these episodes may be related to uncontrolled ischemia and increased sympathetic tone and are best treated by intravenous beta-adrenoceptor blockade (905), intravenous amiodarone (906), left stellate ganglion blockade (907), IABP, or emergency revascularization. Intravenous amiodarone and/or intravenous magnesium may also be used.

Nonsustained VT is not a common cause of hemodynamic compromise and consequently does not require acute thera-

py (Figure 10) (223). Nonetheless, it is an important arrhythmia that may denote a high-risk arrhythmic substrate. Indeed, when it occurs more than 4 days after STEMI in patients with a depressed ejection fraction, it may be a harbinger of future risk of sudden death (908). In unusual cases, although the VT is nonsustained, the rate may be so rapid as to reduce cerebral perfusion sufficiently to cause symptoms (i.e., nonsustained VT at a rate of 200 bpm for 10 seconds). In those cases, pharmacotherapy similar to that recommended for sustained VT may be instituted.

7.7.1.3. Ventricular Premature Beats

Class III

Treatment of isolated ventricular premature beats, couplets, and nonsustained VT is not recommended unless they lead to hemodynamic compromise. (Level of Evidence: A)

Before the present era of care of the STEMI patient with antiplatelet therapy, beta-blockade, ACE inhibitors, and, above all, reperfusion strategies, it was thought that ventricular warning arrhythmias preceded VF. Careful monitoring has refuted this concept, and treatment of these rhythm disturbances is not recommended unless they lead to hemodynamic compromise. All post-STEMI patients with ventricular arrhythmias should undergo assessment of electrolyte levels (especially potassium and magnesium) and have other metabolic parameters (i.e., arterial pH) assessed.

The effect on survival of the pharmacological suppression of ventricular premature beats after MI was assessed in the Cardiac Arrhythmia Suppression Trial (909). Patients receiving class I antiarrhythmic drugs for suppression of ventricular premature beats did more poorly than the placebo-treated patients. Therefore, suppression of ventricular premature beats with class I antiarrhythmic drugs is not a goal of post-STEMI therapy.

7.7.1.4. Accelerated Idioventricular Rhythms and Accelerated Junctional Rhythms

Class III

- **1.** Antiarrhythmic therapy is not indicated for accelerated idioventricular rhythm. (*Level of Evidence: C*)
- 2. Antiarrhythmic therapy is not indicated for accelerated junctional rhythm. (*Level of Evidence: C*)

Accelerated idioventricular rhythms are characterized by a wide QRS complex, with a regular rate higher than the atrial rate and lower than 100 bpm. The appearance of an idioventricular rhythm is an inexact indicator of reperfusion. Treatment of idioventricular rhythm is not indicated, and suppression of the rhythm may lead to hemodynamic compromise.

Accelerated junctional rhythms are characterized by a regular narrow QRS not preceded by atrial activity, with rates above 60 bpm. This rhythm may indicate digitalis intoxica-

1			
Study	AVID (912)	CASH (913)	CIDS (914)
Patients	1016	191	659
Prior MI, %	61	51	77
LVEF	0.32 ± 0.13	0.45 ± 0.18	0.34 ± 0.14
Presenting arrhythmia:	100	100	86
VF or VT, %			

Table 26. Clinical Trials of Secondary Prevention of Sudden Death in ICDs: Applicability to the Post-MI Population

ICD = implantable cardioverter defibrillator; MI = myocardial infarction; LVEF = left ventricular ejection fraction; VF = ventricular fibrillation; VT = ventricular tachycardia.

tion and is more often seen in inferior STEMI than in anterior STEMI. In general, treatment of accelerated junctional rhythm is not indicated.

7.7.1.5. Implantable Cardioverter Defibrillator Implantation in Patients After STEMI

Class I

- 1. An implantable cardioverter defibrillator (ICD) is indicated for patients with VF or hemodynamically significant sustained VT more than 2 days after STEMI, provided the arrhythmia is not judged to be due to transient or reversible ischemia or reinfarction. (Level of Evidence: A)
- 2. An ICD is indicated for patients without spontaneous VF or sustained VT more than 48 hours after STEMI whose STEMI occurred at least 1 month previously, who have an LVEF between 0.31 and 0.40, demonstrate additional evidence of electrical instability (e.g., nonsustained VT), and have inducible VF or sustained VT on EP testing. (Level of Evidence: B)

Class IIa

If there is reduced LVEF (0.30 or less) at least 1 month post-STEMI and 3 months after coronary artery revascularization, it is reasonable to implant an ICD in post-STEMI patients without spontaneous VF or sustained VT more than 48 hours after STEMI. (Level of Evidence: B)

Class IIb

- 1. The usefulness of an ICD is not well established in STEMI patients without spontaneous VF or sustained VT more than 48 hours after STEMI who have a reduced LVEF (0.31 to 0.40) at least 1 month after STEMI but who have no additional evidence of electrical instability (e.g., nonsustained VT). (Level of Evidence: B)
- 2. The usefulness of an ICD is not well established in STEMI patients without spontaneous VF or sustained VT more than 48 hours after STEMI who have a reduced LVEF (0.31 to 0.40) at least 1 month after STEMI and additional evidence of electrical instability (e.g., nonsustained VT) but who do not have inducible VF or sustained VT on EP testing. (Level of Evidence: B)

Class III

An ICD is not indicated in STEMI patients who do not experience spontaneous VF or sustained VT more than 48 hours after STEMI and in whom the LVEF is greater than 0.40 at least 1 month after STEMI. (*Level* of Evidence: C)

In general, VF or hemodynamically significant sustained VT taking place more than 2 days after STEMI in the absence of recurrent MI or potentially reversible ischemia indicates electrical instability and portends a poor prognosis (910,911). There are only 3 randomized trials (AVID [Antiarrhythmics Versus Implantable Defibrillators], CASH (Cardiac Arrest Study Hamburg), and CIDS [Canadian Implantable Defibrillator Study]) that compare the ICD with antiarrhythmic therapy in this patient population (912-914). Although the focus of these studies was not specifically the patient with prior or recent STEMI, the populations, as expected, all had a high prevalence of coronary disease and prior MI (Table 26) (912-914).

Only the AVID study showed a benefit of ICD therapy on mortality. However, the meta-analysis by Connolly et al. (914) clearly demonstrated the similarity of the trials and the consistency of overall results for patients with prior MI. The summary hazard ratio was 0.72 (95% CI 0.60 to 0.87; p equals 0.0006) for total mortality and 0.50 (95% CI 0.37 to 0.67; p less than 0.0001) for arrhythmic death. Thus, patients with VF or hemodynamically significant sustained VT that takes place more than 2 to 3 days after STEMI in the absence of recurrent MI or other readily reversible cause should receive an ICD (Table 27) (915-918).

The management of nonsustained VT in patients with prior MI has proven more challenging. In the presence of LV dysfunction, this arrhythmia is associated with a 2-year mortality estimated at 30% (919,920), of which approximately 50% is believed to be arrhythmic in origin. Subsequent studies suggest that this degree of risk occurs principally in patients with inducible sustained VT not treated with an ICD (921). Three large randomized trials have been performed to study primary prevention of sudden death in these patients. However, because these trials were purposely not designed as post-STEMI trials, a conservative interpretation requires knowledge of when patients were randomized in relationship to their last previous known MI.

In the first prospective randomized trial completed in such a patient population, improved survival was documented

Study Name (Reference)	Year	Patients, n	Days After MI	Qualifying Arrhythmia	EF Upper Limit, mean	EPS	Mortality Hazard ICD vs no ICD (95%CI)
MADIT (915)	1996	196	More than 20	3-30 VPBs; rate greater than 120 bpm	35%, 26%	Yes†	0.46 (0.26-0.82)
MUSTT (916)	1999	704	More than 3*	Greater than 2 VPS; rate greater than 100 bpm	40%, 30%	Yes‡	0.42 (0.28-0.62)§ (918)
MADIT-2 (917)	2002	1232	More than 29	None necessary	30%, 23%	No	0.69 (0.51 - 0.93)

lar premature stimulus. [Change to EP study per Dawn]

"Only 16% of the overall MUSTT population was randomized within 1 month of MI.

Randomized MADIT patients had inducible VT not suppressed by a procainamide infusion during EP study.

Randomized MUSTT patients had inducible VT and were randomized to antiarrhythmic drug therapy. On the basis of clinical indications, some patients received ICDs during the course of follow-up.

This hazard ratio compares MUSTT patients in the antiarrhythmic arm who received ICDs with those who received only EP-guided antiarrhythmic therapy

ized until at least 3 weeks after MI. Results of another prospective randomized trial, the Multicenter Unsustained Tachycardia Trial (MUSTT), showed reduced mortality with aggressive therapy for patients with low ejection fraction (0.40 or less), nonsustained VT on Holter monitoring, and inducible sustained ventricular tachyarrhythmias at EP study (916). Most of this benefit appeared to be due to ICD placement (918). In MUSTT, patients could be randomized as early as 4 days after MI; however, only 16% of patients were enrolled within 1 month of MI, which limits conclusions regarding benefit in patients early after STEMI. The MADIT 2 (Multicenter Automatic Defibrillator Implantation Trial 2) study enrolled 1232 post-MI patients with an LVEF of 0.30 or less. Patients were randomized to ICD therapy or not without the requirement for EP screening for inducible ventricular tachyarrhythmia (917). At a mean follow-up of 20 months, mortality was 14.2% in individuals who had ICDs and 19.8% in the conventionally treated group, a 5.6% absolute and 31% relative risk reduction for death. Of potential importance for management of patients recovering from STEMI, ICD therapy was not implemented until at least 1 month after MI and 3 months after coronary artery revascularization (Figure 32).

Thus, evidence from the randomized trials conclusively supports the concept that for patients with coronary disease, LV dysfunction, and high risk of life-threatening ventricular arrhythmias, ICD therapy is more effective than antiarrhythmic therapy. Indeed, an important contribution of the randomized clinical trials has been to refine the estimated contribution of risk factors for arrhythmic death or cardiac arrest in patients with coronary disease and LV dysfunction. For example, in the MUSTT database, each 5% decrease in LVEF from 0.40 to 0.20 conferred a 19% incremental relative risk of arrhythmic death or cardiac arrest. Inducibility at EP study increased risk by 63%. These simple risk factors, therefore, allow the clinician to select those patients most likely to benefit from ICD therapy in the late post-STEMI phase (922). The published studies, however, do not systematically address management considerations within the first month after STEMI. MADIT enrolled patients at least 3 weeks after MI, MUSTT at least 4 days, and MADIT 2 at least 1 month. Nonetheless, there is conceptual uniformity of the results of all 3 trials to support ICD therapy for patients at high risk of sudden cardiac death after STEMI.

Although the ejection fraction criteria used to define a degree of LV dysfunction severe enough to confer high risk are not uniform from study to study, a reduced ejection fraction remains the critical measurement to determine whether a post-STEMI patient is at high risk for late ventricular arrhythmia. Furthermore, ejection fraction is not always stable after STEMI. In a cohort of 252 patients who had sustained an anterior wall STEMI, 53% had at least a 5-point increase in ejection fraction at 90 days, whereas only 16%

after implantation of ICDs in patients with nonsustained VT and EP study-inducible and nonsuppressible ventricular tachyarrhythmias compared with conventional drug therapy, including amiodarone (915). Patients could not be random-



Figure 32. An evidence-based algorithm for primary prevention of sudden death in post-STEMI patients without spontaneous VF or sustained VT at least 1 month post-STEMI to aid in selection of implantable cardioverter/defibrillator (ICD) in patients with STEMI and diminished ejection fraction (EF). The appropriate management path is selected based upon left ventricular ejection fraction (LVEF) measured at least one month after STEMI. These criteria, that are based on the published data, form the basis for the full-text guide-lines in section 7.7.1.5. All patients, whether an ICD is implanted or not, should receive medical therapy as outlined in the full-text guidelines. VF = ventricular fibrillation; VT = ventricular tachycardia; STEMI = ST-elevation myocardial infarction; NSVT = nonsustained VT; LOE = level of evidence; EPS = electrophysiological study; LVEF = left ventricular EF.

had a drop of at least 5 points (923). Thus, ejection fraction should be measured at least 1 month after STEMI before a decision regarding ICD placement is made. Finally, the variability of ejection fraction measured by different techniques (924) should also be taken into account by the clinician caring for the patient.

If there is reduced EF (0.30 or less) at least 1 month after STEMI and 3 months after coronary artery revascularization, it is reasonable to implant an ICD without a preceding diagnostic EP study. If the patient has an EF between 0.31 and 0.40, additional evidence of electrical instability should be sought to help with management decisions. If markers of electrical instability are detected on noninvasive testing (nonsustained on monitoring), an EP study is the next diagnostic step. If inducible VF or sustained VT is found at EP study, an ICD is indicated. The Writing Committee endorses additional research on noninvasive markers of electrical instability and the results of EP testing to help clarify management of decisions in such patients (see Section 7.11). The usefulness of an ICD is less well established for patients in whom no inducible VF or sustained VT is detected at EP study. As shown in Figure 32, such patients do not receive an ICD but should receive medical therapy (post-STEMI) as

discussed in this guideline. An ICD is not indicated in patients with STEMI who do not experience spontaneous VF or sustained VT more than 48 hours after STEMI and in whom the ejection fraction is greater than 0.40 at least 1 month after STEMI.

Unfortunately, published clinical trials have not addressed the patient with a low ejection fraction within the first month after STEMI, which is thought to be a particularly high-risk period. Preliminary findings from DINAMIT (Defibrillator in Acute Myocardial Infarction Trial), a clinical trial of ICD versus conventional therapy in patients 6 to 40 days after MI with ejection fraction less than 0.35 and evidence of impaired autonomic tone, showed a reduction in arrhythmic mortality at the cost of an increase in nonarrhythmic mortality, yielding no net benefit of an ICD implanted in the first month after STEMI (Connelly S; oral presentation, American College of Cardiology 53rd Annual Scientific Session, March 2004, New Orleans, LA). Given this gap in knowledge, the ongoing Home Automatic External Defibrillator Trial is testing the hypothesis that provision of an AED to patients with anterior STEMI for home use will improve survival beyond that achieved from the typical lay response to sudden cardiac arrest (135). Additionally, wearable external defibrillators have been developed that may be applicable for high-risk patients after STEMI (926).

7.7.2. Supraventricular Arrhythmia/AF

Class I

- 1. Sustained AF and atrial flutter in patients with hemodynamic compromise should be treated with one or more of the following:
 - a. Synchronized cardioversion with an initial monophasic shock of 200 J for AF and 50 J for flutter, preceded by brief general anesthesia or conscious sedation whenever possible. (Level of Evidence: C)
 - b. For episodes of AF that do not respond to electrical cardioversion or recur after a brief period of sinus rhythm, the use of antiarrhythmic therapy aimed at slowing the ventricular response is indicated. One or more of these pharmacological agents may be used:
 - i. Intravenous amiodarone (927). (Level of Evidence: C)
 - ii. Intravenous digoxin for rate control, principally for patients with severe LV dysfunction and heart failure. (Level of Evidence: C)
- 2. Sustained AF and atrial flutter in patients with ongoing ischemia but without hemodynamic compromise should be treated with one or more of the following:
 - a. Beta-adrenergic blockade is preferred, unless contraindicated. (Level of Evidence: C)
 - b. Intravenous diltiazem or verapamil. (Level of Evidence: C)
 - c. Synchronized cardioversion with an initial monophasic shock of 200 J for AF and 50 J for flutter, preceded by brief general anesthesia or conscious sedation whenever possible. (Level of Evidence: C)
- 3. For episodes of sustained AF or flutter without hemodynamic compromise or ischemia, rate control is indicated. In addition, patients with sustained AF or flutter should be given therapy with anticoagulants. Consideration should be given to conversion of sinus rhythm in patients without a history of atrial fibrillation or flutter prior to STEMI. (Level of Evidence: C)
- 4. Re-entrant paroxysmal supraventricular tachycardia, because of its rapid rate, should be treated with the following in the sequence shown:
 - a. Carotid sinus massage. (Level of Evidence: C)
 - b. Intravenous adenosine (6 mg × 1 over 1 to 2 seconds; if no response, 12 mg IV after 1 to 2 minutes may be given; repeat 12 mg dose if needed. (Level of Evidence: C)
 - c. Intravenous beta-adrenergic blockade with metoprolol (2.5 to 5.0 mg every 2 to 5 minutes to a total of 15 mg over 10 to 15 minutes) or atenolol (2.5 to 5.0 mg over 2 minutes to a total of 10 mg in 10 to 15 minutes. (Level of Evidence: C)

All AF, including presentation with AF.

GUSTO-I (944) 7.9* Increased age, CK, 6.2% GUSTO-I (944) 7.9* Killip class, heart rate 14.3% TRACE (935) 21† Female; increased age, 9% n TRACE (935) 21† Female; increased age, 9% n TRACE (935) 21† Temale; increased age, 9% n GUSTO-III (945) 6.5* Increased age, Killip class, 6% no ,	e, CK, 6.2% no AF vs. eart rate 14.3% with AF at 30 days used age, 9% no AF vs.	OR 1.4 (95% CI 1.3-1.5) at 30 days OR 15 (95% CI 1 2-1 8)
TRACE (935) 21† Female; increased age, 9% n LV dysfunction; 18% anterior Q-wave MI inh GUSTO-III (945) 6.5* Increased age, Killip class, 6% no <i>i</i>	used age, 9% no AF vs.	OR 15 (95% CT 1 2-1 8)
GUSTO-III (945) 6.5* Increased age, Killip class, 6% no / hymotension. heart block. VF with AF	tion; 18% with AF ave MI inhospital	inhospital
	t block, VF 6% no AF vs. 15% with AF at 30 days	OR 1.49 (95% CI 1.17-1.89) at 30 days
GISSI-3 (946) 7.8* Increased age, heart rate, 5% no A Killip class, hypertension with AF	reart rate, 5% no AF vs. 12.6% bertension with AF inhospital	OR 1.98 (95% CI 1.67-2.64)

- d. Intravenous diltiazem (20 mg [0.25 mg/kg]) over 2 minutes followed by an infusion of 10 mg/h. (*Level of Evidence: C*)
- e. Intravenous digoxin, recognizing that there may be a delay of at least 1 hour before pharmacological effects appear (8 to 15 mcg/kg [0.6 to 1.0 mg in a person weighing 70 kg]). (Level of Evidence: C)

Class III

Treatment of atrial premature beats is not indicated. (Level of Evidence: C)

Atrial fibrillation occurs more frequently than atrial flutter or paroxysmal supraventricular tachycardia in patients with STEMI. The consequences and acute treatment of all 3 arrhythmias may be considered together, recognizing that in atrial flutter and supraventricular tachycardia, atrial pacing may be effective in terminating the tachycardia (928-933). Estimates of the incidence of AF in patients with STEMI vary depending on the population sampled. In the CCP, 22% of Medicare patients aged 65 years or older who were hospitalized for STEMI had AF (934). In the Trandolapril Cardiac Evaluation (TRACE) study of patients with LV dysfunction associated with STEMI, 21% had AF (935). Among the causes of AF in the immediate post-STEMI setting are excessive sympathetic stimulation, atrial stretch due to LV or RV dysfunction, atrial infarction due to circumflex or right coronary lesions, pericarditis, hypokalemia, underlying chronic lung disease, and hypoxia (807,929,936-940). Thus, AF occurs more often in patients with larger infarcts or anterior location of reinfarction and in those whose hospital course is complicated by CHF, complex ventricular arrhythmias, advanced AV block, atrial infarction, or pericarditis. Atrial fibrillation may also occur in patients with inferior STEMI secondary to proximal right coronary artery occlusion with compromise of flow in the sinoatrial nodal artery, the major blood supply to the atria. In some studies, the incidence of AF after STEMI is decreased in patients receiving fibrinolytic therapy (929,941), whereas in other studies, the incidence is similar (942). In the GUSTO trial, patients treated with accelerated alteplase and intravenous UFH had a significantly lower incidence of AF and atrial flutter than patients treated with other fibrinolytic therapies (25). Systemic embolization is more frequent in patients with paroxysmal AF (1.7%) than in those without (0.6%), with half of the embolic events occurring on the first day of hospitalization and more than 90% occurring by the fourth day (943). Because AF can be associated with pericarditis, the development of PR-segment displacement on serial ECGs may predict risk of developing AF during hospitalization (941).

The development of AF is associated with a worse in-hospital and long-term prognosis. In a study of 106 780 elderly (Medicare) patients with AF during MI, about half presented with AF and half developed AF during hospitalization (934). The presence of AF during hospitalization increased shortand long-term relative mortality by 20% and 34%, respectively (Table 28) (935,944-946). Patients who developed AF during hospitalization had a worse prognosis than those with AF on admission (934). Stroke rates are also increased in patients with MI and AF compared with those without AF (944). Outcomes appear to have improved in the fibrinolytic era for patients with AF and STEMI compared with experience between 1981 and 1983 (942), but a stroke rate of 3.1% in the setting of AF and STEMI (944) emphasizes the importance of this association even in the era of fibrinolysis.

When AF occurs, the clinician must consider and correct, if possible, the underlying causes. The initial clinical decision is whether to proceed immediately to electrical cardioversion if the patient is unstable with a rapid heart rate and hypotension, intractable heart failure, or ischemic pain. Cardioversion should be performed when the patient is under adequate general anesthesia or has received medication to produce conscious sedation to avoid pain related to delivery of the electric shock. Short-acting anesthetic drugs or agents that produce conscious sedation are preferred because cardioversion patients should recover rapidly after the procedure (947).

Proper synchronization of the electric shock with the QRS complex calls for triggering by monitoring the R wave with an appropriately selected lead. In addition to R-wave amplitude, it is important that the monitored lead give a good view of P waves, thus facilitating assessment of the outcome of the procedure. The initial energy delivered with a monophasic waveform may be low (50 J) for cardioversion of atrial flutter. Higher monophasic shock energy is required for AF cardioversion, starting with at least 200 J. The monophasic shock energy output is increased successively in increments of 100 J until a maximum of 400 J is reached. Some physicians begin with higher energies to reduce the number of shocks (and thus the total energy) delivered (948). Energy settings should be reduced by about 50% of those noted above for monophasic shocks if a device that delivers a biphasic waveform is used. To avoid myocardial damage, the interval between 2 consecutive shocks should not be less than 1 minute (949). The optimal paddle position remains controversial; however, the electrodes should be in direct contact with the chest wall and not, for example, positioned over breast tissue. There is little experience with novel methods such as transvenous electrical cardioversion in patients with STEMI.

When medical therapy is selected, and in the absence of CHF or severe pulmonary disease, one of the most effective means of slowing the ventricular rate in AF is the use of intravenous beta-adrenoceptor–blocking agents such as metoprolol (2.5 to 5.0 mg every 2 to 5 minutes to a total of 15 mg over 10 to 15 minutes) or atenolol (2.5 to 5.0 mg over 2 minutes to a total of 10 mg in 10 to 15 minutes). Heart rate, blood pressure, and the ECG should be monitored, and treatment should be halted when therapeutic efficacy is achieved or if systolic blood pressure falls below 100 mm Hg or there is excessive bradycardia (e.g., a heart rate below 50 bpm during treatment).

When there are absolute contraindications to beta-adrenergic blockade (bronchospastic lung disease or allergy), rate slowing may also be achieved by intravenous diltiazem (20 mg [0.25 mg/kg]) over 2 minutes followed by an infusion of 10 mg/h) or verapamil (2.5 to 10 mg IV over 2 minutes; may repeat a 5- to 10-mg dose after 15 to 30 minutes). There are concerns regarding the negative inotropic effects of these drugs and reports of increased post-STEMI mortality in patients with LV dysfunction taking long-term, short-acting oral diltiazem (950). Calcium antagonists, therefore, are not recommended for long-term rate control in post-STEMI patients; however, they may be useful for short-term rate control in hospitalized patients when beta-blockers are absolutely contraindicated.

Amiodarone has both sympatholytic and calcium antagonistic properties, depresses AV conduction, and is effective in controlling the ventricular rate in patients with AF. Intravenous amiodarone is effective and well tolerated in critically ill patients who develop rapid atrial tachyarrhythmias refractory to conventional treatment, but its effectiveness has not been evaluated sufficiently in patients with STEMI. In a small observational study of critically ill patients, there was a reduction in ventricular rate of 37 bpm after a 1-hour infusion of 242 mg of amiodarone (951). Amiodarone is considered a first-line agent for heart rate control in critically ill patients (951) and a second-line agent for those patients who are less hemodynamically unstable and can tolerate intravenous diltiazem. Amiodarone is the preferred agent to control repeat AF in patients with CHF or a low-output state.

Although intravenous digoxin may effectively slow the ventricular rate at rest, there is a delay of at least 60 minutes before onset of a therapeutic effect in most patients, and a peak effect does not develop for up to 6 hours. Digoxin is no more effective than placebo in converting AF to sinus rhythm (790,792,795) and may prolong the duration of AF (790,952). The efficacy of digoxin is reduced in states of high sympathetic tone, a common precipitant of paroxysmal AF. In a review of 139 episodes of paroxysmal AF recorded on Holter monitoring, there was no difference in the ventricular rates of patients taking digoxin and those not taking this medication (952). Other investigators, however, have found that digoxin reduces the frequency and severity of AF recurrence (953). Furthermore, the combination of digoxin and atenolol has been shown to be effective for ventricular rate control (954). Given the availability of more effective agents, digoxin is no longer first-line therapy for management of acute AF, but it plays a continuing role in patients with heart failure or LV dysfunction (955). Rapid administration of digoxin to achieve rate slowing may be accomplished by giving intravenous digoxin (8 to 15 mcg/kg [0.6 to 1.0 mg in a person weighing 70 kg]), with half the dose administered initially and the additional increment in 4 hours (956). This method provides a slower response than intravenous betaadrenoceptor blockade or amiodarone; however, some effect on rate slowing may be detectable in 30 minutes to 2 hours.

Given that the databases reporting a marked increase in the risk of stroke in post-STEMI patients with AF do not report the time duration of AF at which stroke risk increases, it is unclear whether all post-STEMI patients with AF, even if transient, should receive anticoagulation or whether this aggressive posture should be reserved only for those with sustained AF of at least a few hours' duration. Indeed, the clinical circumstances, in particular whether there are other risk factors for post-STEMI stroke, should modulate the threshold for anticoagulation in the patient with transient AF.

When it has been determined that anticoagulation is required, either UFH or LMWH may be used. When UFH is selected, an intravenous bolus of 60 U/kg followed by a continuous intravenous infusion at 12 U/kg/h to maintain an aPTT of 50 to 70 seconds (approximately 1.5 to 2 times control) should be given. Alternatively, one of the LMWHs may be used at doses recommended by the manufacturer for full anticoagulation.

Once rate control has been accomplished, clinicians may opt to convert the patient to sinus rhythm to attain optimal hemodynamics and ultimately permit discontinuation of anticoagulants. Guidelines for electrical and chemical cardioversion for the stable patient with AF have been formulated (955). In patients with STEMI, special attention must be given to considerations of proarrhythmia with many commonly used antiarrhythmic agents. The preferred agent for intermediate or long-term use in the STEMI patient, based on the best safety record in post-MI trials, is amiodarone. Indeed, in a meta-analysis of 6553 randomized patients, 78% of whom were in post-MI trials, amiodarone resulted in a relative reduction in risk of death of 13%, which was wholly due to a greater reduction in arrhythmic death (914). The excess risk of pulmonary toxicity was only 1% per year. A post hoc analysis of GUSTO-III patients with AF also addressed the relative safety of other antiarrhythmic agents, including sotalol, in the post-STEMI setting. In a total of 317 patients with AF who received antiarrhythmic agents, no agent or class of agents was associated with increased mortality (957). However, the totality of evidence is much less compelling than for amiodarone after STEMI. Transient AF does not obligate the patient to receive long-term anticoagulation or antiarrhythmic agents, but if such treatment is elected, it is appropriate to limit their use to 6 weeks if sinus rhythm has been restored. In outpatients with paroxysmal AF, a controversy regarding the relative merits of rate control versus maintenance of sinus rhythm has been addressed by a large randomized clinical trial (958). AFFIRM (Atrial Fibrillation Follow-up Investigation of Rhythm Management) demonstrated no clear survival benefit for either strategy. However, patients with STEMI were not enrolled, and only 38.2% of the population had a prior diagnosis of coronary artery disease. Thus, the conclusions of AFFIRM should not be applied to the STEMI population, and the decision to continue chronic adjustable-dose oral anticoagulation indefinitely should be based on an overall assessment of the risk of thromboembolism in the individual patient.

7.7.3. Bradyarrhythmias

See Table 29 for recommendations. Sinus bradycardia occurs frequently, constituting 30% to 40% of AMI-associat-

				First-Deg	ree Block		Mobitz I	Second-	Degree AV Bloc	ck	Mobitz II (Second	Degree AV Blo	[]
	Normal	_	Anterior	MI	Nonanterio	r MI	Anterior N	П	Nonanterio	r MI	Anterior MI		Nonanterio	W
	Action	Class	Action	Class	Action	Class	Action	Class	Action	Class	Action	Class	Action	Class
Normal	Observe	Ι	Observe	Ι	Observe	I	Observe	IIb	Observe	IIa	Observe	III	Observe	E
	A	Ш	A	III	Α	III	A*	III	A	III	A	III	А	Ξ
	TC	Ш	TC	IIb	TC	IIb	TC	Ι	TC	I	TC	Ι	TC	п
	TV	Ш	TV	III	TV	III	TV	III	TV	III	ΛL	IIa	ΤV	Ша
Old or new	Observe	I	Observe	IIb	Observe	IIb	Observe	III	Observe	llb	Observe	Ш	Observe	Ш
fascicular block	A	III	A	Η	A	III	A^*	III	A^*	III	A	Ξ	A	II ,
(LAFB or LPFB)	TC	dII	TC	I	TC	IIa	TC	I	TC	I	JT VT	I IIa	TV	II III
	<u>>.</u> [.	Ш	٨I	III		Ш	TV	Ш	<u>></u> 1	Ξ		114	-	
Old	Observe	I	Observe	Ш	Observe	III	Observe	III	Observe	III	Observe	Ш	Observe	III
bundle-branch	A	III	А	III	A	III	A*	III	A	Ш	A	III -	A	III ·
block	TC	dII	TC	I	TC	Ι	TC	I	TC	I	IC	- +	TC	
	TV	III	ΛL	dIIb	TV	llb	VT	IIb	TV	IIb	1	Ша	1 <	11a
New	Observe	Ш	Observe	III	Observe	III	Observe	Ш	Observe	III	Observe	ШШ	Observe	
bundle-branch	А	III	A	III	A	III	A*	Ш	A	III	UL VL	ΞÉ	UL VL	Ш
block	TC	I	TC	Ι	TC	I	TC	Ι	TC	I		I I	TV	III
	TV	dII	ΛL	IIa	TV	Па	ΤV	IIa	TV	Ша	1 V	I	1 <	I
Fascicular block	Observe	III	Observe	III	Observe	III	Observe	III	Observe	Ш	Observe	Π	Observe	
+	A	III	A	III	A	III	\mathbf{A}^*	Ш	A	Ш	A	II I	A TC	III I
RBBB	TC	I	TC	Ι	TC	I	TC	I	TC	Ι		110 -	1C Tri	all 1
	TV	dII	ΤV	IIIa	TV	Па	TV	IIa	TV	Ша		-	1 <	-
Alternating	Observe	III	Observe	III	Observe	III	Observe	III	Observe	III	Observe	H	Observe	
left and right	A	III	А	III	А	III	A^*	III	А	III	A	∃ E	A T	Π
bundle-branch	TC	IIb	TC	IIb	TC	IIb	TC	llb	TC	, Í		II0	TU	dII
block	TV	I	TV	I	ΤV	I	TV	I	YV	Ι	1 <	-	1 V	I

Table 29. Recommendations for Treatment of Atrioventricular and Intraventricular Conduction Disturbances During ST-Elevation Myocardial Infarction

ACC - www.acc.org AHA - www.americanheart.org

e198

Continued on next page

Table 29. Continued

This table is designed to summarize the atrioventricular (column headings) and intraventricular (row headings) conduction disturbances that may occur during acute anterior or nonanterior STEMI, the possible treatment options, and the indications for each possible therapeutic option.

AV = atrioventricular; MI = myocardial infarction; A = atropine; TC = transcutaneous pacing; TV = temporary transvenous pacing; LAFB = left anterior fascicular block; LPFB = left posterior fascicular block; RBBB = right bundle-branch block.

<u>Action</u>

- There are 4 possible actions, or therapeutic options, listed and classified for each bradyarrhythmia or conduction problem:
- 1. Observe: continued ECG monitoring, no further action planned.
- 2. A, and A*: atropine administered at 0.6 to 1.0 mg IV every 5 minutes to up to 0.04 mg/kg. In general, because the increase in sinus rate with atropine is unpredictable, this is to be avoided unless there is symptomatic bradycardia that will likely respond to a vagolytic agent, such as sinus bradycardia or Mobitz I, as denoted by the asterisk, above.
- 3. TC: application of transcutaneous pads and standby transcutaneous pacing with no further progression to transvenous pacing imminently planned.
- 4. TV: temporary transvenous pacing. It is assumed, but not specified in the table, that at the discretion of the clinician, transcutaneous pads will be applied and standby transcutaneous pacing will be in effect as the patient is transferred to the fluoroscopy unit for temporary transvenous pacing.

<u>Class</u>

Each possible therapeutic option is further classified according to ACC/AHA criteria as I, IIa, IIb, and III. The level of evidence for all cells and all treatments is B or C. There are no randomized trials available that address or compare specific treatment options. Moreover, the data for this table and recommendations are largely derived from observational data of pre–thrombolytic era databases. Thus, the recommendations above must be taken as recommendations and tempered by the clinical circumstances.

Level of Evidence

This table was developed from (1) published observational case reports and case series, (2) published summaries, not meta-analyses, of these data; and (3) expert opinion, largely from the prereperfusion era. There are no published randomized trials comparing different strategies of managing conduction disturbances after STEMI. Thus, the level of evidence for the recommendations in the table is C.

How to use the table

Example: 54-year-old man is admitted with an anterior STEMI and a narrow QRS on admission. On day 1, he develops a right bundle-branch block (RBBB), with a PR interval of 0.28 seconds.

- 1. RBBB is an intraventricular conduction disturbance, so look at row "New bundle-branch block."
- 2. Find the column for "First-Degree AV Block."
- 3. Find the "Action" and "Class" cells at the convergence.
- 4. Note that "Observe" and "Atropine" are class III, not indicated; transcutaneous pacing (TC) is class I. Temporary transvenous pacing (TV) is class IIa.

ed cardiac arrhythmias. It is especially frequent within the first hour of inferior STEMI and with reperfusion of the right coronary artery (Bezold-Jarisch reflex) as a result of increased parasympathetic activity (vagal tone) (959). There are several other potential mechanisms, operating in isolation or in parallel, that account for the high incidence of sinus bradycardia. These include local increases in adenosine, local hyperkalemia, systemic metabolic derangements, and concomitant use of bradycardia-promoting medications (960).

Heart block may develop in approximately 6% to 14% of patients with STEMI. Intraventricular conduction delay has been reported in about 10% to 20% of patients with STEMI

in past reviews (961). The development of AV and intraventricular blocks during STEMI is generally related to the extent of the ischemic/infarcted segment. As such, AV block predicts an increased risk of in-hospital mortality but is less predictive of long-term mortality in those who survive to hospital discharge (962-964). Nonetheless, in fibrinolysis trials, bundle-branch block was present on admission in only 4% but was predictive of a substantially increased in-hospital mortality rate (156). Indeed, the development of sudden AV block in the setting of anterior STEMI, once a feared complication, is quite unusual in the present CCU population in the postreperfusion era. Thus, the use of transvenous pac-

	Location of AV Conduction Disturbance		
Feature	Proximal Distal		
Site of block	Intranodal	Infranodal	
Site of infarction	Inferoposterior	Anteroseptal	
Compromised arterial supply	RCA (90%), LCx (10%)	Septal perforators of LAD	
Pathogenesis	Ischemia, necrosis, hydropic cell swelling, excess parasympathetic activity	Ischemia, necrosis, hydropic cell swelling	
Predominant type of AV nodal block	First degree (pulse rate greater than 200 msec) Mobitz type I second degree	Mobitz type II second degree Third degree	
Common promontory features of third-degree AV block	(a) First-second-degree AV block (b) Mobitz type I pattern		
 Features of escape after third-degree block (a) Location (b) QRS width (c) Rate (d) Stability of escape rhythm 	 (a) Proximal conduction system (His bundle) (b) Less than 0.12 sec* (c) 45-60 per min but may be as low as 30 per min (d) Rate usually stable; asystole uncommon 	 (a) Distal conduction system (bundle branches) (b) Greater than 0.12 sec* (c) Often less than 30 per min (d) Rate often unstable with moderate to high risk of ventricular asystole 	
Duration of high-grade AV block	Usually transient (2-3 days)	Usually transient, but some form of AV conduction disturbances and/or intra- ventricular defect may persist	
Associated mortality rate	Low unless associated with hypotension and/or CHF	High because of extensive infarction associated with power failure or ventricular arrhythmias	
Pacemaker therapy (a) Temporary	(a) Rarely required; may be considered for bradycardia associated with LV power failure, syncope, or angina	(a) Indicated in patients with anteroseptal infarction and acute bifascicular block	
(b) Permanent	(b) Almost never indicated because conduction defect is usually transient	(b) Indicated for patients with high-grade AV block with block in His-Purkinje systems and those with transient advanced AV block and associated bundle-branch block	

Table 30. Features of AV Conduction Disturbances in Acute Myocardial Infarction

AV = atrioventricular; RCA = right coronary artery; LCx = left circumflex artery; LAD = left anterior descending artery; sec = seconds; min = minutes; CHF = congestive heart failure; LV = left ventricular.

*Some studies suggest that a wide QRS escape rhythm (greater than 0.12 sec) after high-grade AV block in inferior infarction is associated with a worse prognosis.

Modified with permission from Antman EM. Cardiovascular Therapeutics: A Companion Guide to Braunwald's Heart Disease. 2nd ed. Philadelphia, PA: WB

ing has diminished, and the reliance on transcutaneous pacing has increased (Table 30) (965).

7.7.3.1. Acute Treatment of Conduction Disturbances and Bradyarrhythmias

Treatment of the conduction disturbances and resulting bradyarrhythmias can have either a prophylactic or therapeutic focus. The purpose of prophylactic pacing is to prevent symptomatic or catastrophic bradycardia by selecting and placing a transcutaneous or transvenous temporary pacemaker. Prophylactic pacing requires the clinician to predict which patients will develop sudden complete heart block with an inadequate ventricular escape mechanism. Fortunately, conduction disturbances generally occur in a stepwise fashion, so that knowledge of the specific ECG pattern can be used to estimate the risk of developing complete heart block and thus to guide the need for prophylactic temporary pacing. These estimates of risk, however, must be interpreted in the context of the risk of performing a procedure, particularly transvenous temporary pacing, on an unstable patient in the acute phase after STEMI, with all the attendant modern antithrombotic therapies increasing the risk of bleeding complications. Additionally, most of the clinically based algorithms to estimate risk of developing complete heart block were developed in the prefibrinolytic era, so they must be interpreted cautiously when applied to a modern post-STEMI population. In some cases, after the development of advanced AV block after STEMI, temporary transcutaneous or transvenous pacing must be used to maintain a stable cardiac rhythm and adequate hemodynamics. When the patient becomes pacemaker-dependent owing to a persistent conduction defect, however, temporary transvenous pacing is preferred compared with long-term transcutaneous pacing.

The pharmacological treatment of bradycardia and AV conduction disturbances during STEMI is a therapeutic, not prophylactic, measure. Pharmacotherapy centers on the use of atropine at doses of 0.6 to 1.0 mg IV repeated every 5 minutes until there is the desired effect or a total dose of 0.04 mg/kg (2 mg for a 50-kg person) has been reached. When there is infranodal block, however, atropine may increase the sinus rate without affecting infranodal conduction, and so the effective ratio of conduction may decrease, and the ventricular rate may decrease.

Other pharmacotherapies to treat bradyarrhythmias, such as isoproterenol and aminophylline, are not recommended because they are arrhythmogenic and increase myocardial oxygen demand. Glucagon has been used to treat bradycardia caused by beta-blockers and calcium antagonists, although principally only when these agents have been used in toxic doses, particularly in combination (966).

The recommendations for prophylactic treatment of AV and intraventricular conduction blocks and the possible combinations are contained in Table 29.

7.7.3.1.1. VENTRICULAR ASYSTOLE.

Class I

Prompt resuscitative measures, including chest compressions, atropine, vasopressin, epinephrine, and temporary pacing, should be administered to treat ventricular asystole. (Level of Evidence: B)

Ventricular asystole may be caused either by failure of the sinus node to generate a cardiac impulse or by the development of complete heart block. In either case, there is concurrent failure of the usual underlying escape mechanisms, whether atrial, junctional, or ventricular. Treatment of the acute event requires prompt institution of transcutaneous pacing, vasopressin, epinephrine, and atropine. It is important to address the underlying cause and discontinue medications that either suppress sinus node function, decrease AV nodal conduction, or suppress a potential escape mechanism. Cardiopulmonary resuscitation according to guidelines must be instituted (900). Transvenous pacing should be instituted unless the asystole is brief and a precipitating cause is found (893,967,968). Vasopressin is an effective vasopressor and is as useful as epinephrine for the treatment of adult shockrefractory VF and pulseless electrical activity (969). In a study of patients with asystole out of the hospital, vasopressin use was associated with significantly higher rates of hospital admission (29.0% versus 20.3% in the epinephrine group; p equals 0.02) and hospital discharge (4.7% versus 1.5%, p equals 0.04). Among patients in whom spontaneous circulation was not restored with 2 injections of either vasopressin or epinephrine, additional treatment with epinephrine resulted in significant improvement in the rates of survival to hospital admission and hospital discharge in the vasopressin group but not in the epinephrine group (hospital admission rate 25.7% versus 16.4%; p equals 0.002; hospital discharge rate 6.2% versus 1.7%; p equals 0.002). Cerebral performance was similar in the 2 groups (969). Thus, in patients with STEMI with ventricular asystole, vasopressin (40 IU) would appear to be the preferable vasoconstrictor to administer first.

7.7.3.2. Use of Permanent Pacemakers

7.7.3.2.1. PERMANENT PACING FOR BRADYCARDIA OR CONDUCTION BLOCKS ASSOCIATED WITH STEMI.

Class I

- 1. Permanent ventricular pacing is indicated for persistent second-degree AV block in the His-Purkinje system with bilateral bundle-branch block or thirddegree AV block within or below the His-Purkinje system after STEMI. (Level of Evidence: B)
- 2. Permanent ventricular pacing is indicated for transient advanced second- or third-degree infranodal AV block and associated bundle-branch block. If the site of block is uncertain, an EP study may be necessary. (Level of Evidence: B)



Figure 33. Twelve-lead ECG in a patient with anterior STEMI complicated by right bundle-branch block (RBBB) and left anterior fascicular block. Note the convex upward ST-elevation in V_2 - V_3 consistent with anterior STEMI RBBB; the anticipated r-wave of the conventional rsR' pattern of RBBB has been replaced by a Q wave in V_1 - V_2 as a result of anterior STEMI. The initial 80-ms forces of the QRS in the frontal plane are deviated to the left as a result of left anterior fascicular block. ECG = electrocardiogram; STEMI = ST-elevation myocardial infarction.

3. Permanent ventricular pacing is indicated for persistent and symptomatic second- or third-degree AV block. (Level of Evidence: C)

Class IIb

Permanent ventricular pacing may be considered for persistent second- or third-degree AV block at the AV node level. (Level of Evidence: B)

Class III

- 1. Permanent ventricular pacing is not recommended for transient AV block in the absence of intraventricular conduction defects. (Level of Evidence: B)
- 2. Permanent ventricular pacing is not recommended for transient AV block in the presence of isolated left anterior fascicular block. (Level of Evidence: B)
- **3.** Permanent ventricular pacing is not recommended for acquired left anterior fascicular block in the absence of AV block. (*Level of Evidence: B*)
- 4. Permanent ventricular pacing is not recommended for persistent first-degree AV block in the presence of bundle-branch block that is old or of indeterminate age. (Level of Evidence: B)

Indications for permanent pacing after STEMI in patients experiencing AV block are related in large measure to the presence of intraventricular conduction defects (Table 29). Unlike some other indications for permanent pacing, the criteria for patients with STEMI and AV block do not necessarily depend on the presence of symptoms. Furthermore, the requirement for temporary pacing in STEMI does not by itself constitute an indication for permanent pacing (3).

The long-term prognosis for survivors of STEMI who have had AV block is related primarily to the extent of myocardial injury and the character of intraventricular conduction disturbances rather than the AV block itself (970-974). Patients with STEMI who have intraventricular conduction defects, with the exception of isolated left anterior fascicular block, have an unfavorable short- and long-term prognosis and an increased risk of sudden death (970,971,973,975). This unfavorable prognosis is not necessarily due to development of high-grade AV block, although the incidence of such block is higher in postinfarction patients with abnormal intraventricular conduction (971,976,977).

When AV or intraventricular conduction block complicates STEMI, the type of conduction disturbance, location of infarction, and relation of electrical disturbance to infarction must be considered if permanent pacing is contemplated. Even with data available, the decision is not always straightforward because the reported incidence and significance of various conduction disturbances vary widely (978). Despite the use of fibrinolytic therapy and primary PCI, which have decreased the incidence of AV block in STEMI, mortality remains high if AV block occurs (962,979-981).

Although more severe disturbances in conduction are generally associated with greater arrhythmic and nonarrhythmic mortality (971-974,976,978), the impact of pre-existing bundle-branch block on mortality after STEMI is controversial (978,982). A particularly ominous prognosis is associated with LBBB combined with advanced second- or third-degree AV block and with right bundle-branch block combined with left anterior or left posterior fascicular block (964,972,974,982) (Figure 33). Irrespective of whether the infarction is anterior or inferior, the development of an intraventricular conduction delay reflects extensive myocardial damage rather than an electrical problem in isolation (974). Although AV block that occurs during inferior STEMI can be associated with a favorable long-term clinical outcome, inhospital survival is impaired, regardless of the use of temporary or permanent pacing in this situation (964,979,980,983). Furthermore, pacemakers should not be implanted if the periinfarctional AV block is expected to resolve or to not have a negative effect on long-term prognosis, as in the case of inferior STEMI (981) with Mobitz I second-degree AV block.

With regard to sinus node dysfunction precipitated or unmasked by STEMI or associated necessary medical therapy, the indications for permanent pacing do not differ from the indications.

7.7.3.2.2. SINUS NODE DYSFUNCTION AFTER STEMI

Class I

Symptomatic sinus bradycardia, sinus pauses greater than 3 seconds, or sinus bradycardia with a heart rate less than 40 bpm and associated hypotension or signs of systemic hemodynamic compromise should be treated with an intravenous bolus of atropine 0.6 to 1.0 mg. If bradycardia is persistent and maximal (2 mg) doses of atropine have been used, transcutaneous or transvenous (preferably atrial) temporary pacing should be instituted. (*Level of Evidence: C*)

Sinus node dysfunction may be unmasked or caused by MI owing to disruption in the blood supply to the sinoatrial node, or owing to the use of medications such as beta-adrenergic blocking agents or calcium antagonists. The overall recommendations, indications, and levels of evidence are not different in non-MI and patients with STEMI, except that transient sinus bradycardia often occurs in the setting of inferior wall infarction, and treatment should avoid permanent pacing whenever possible. Thus, the published ACC/ AHA/NASPE 2002 Guideline Update for Implantation of Cardiac Pacemakers and Antiarrhythmia Devices (984) should be used to guide therapy in patients with STEMI with persistent sinus node dysfunction. These guidelines should be applied with the proviso that new sinus node dysfunction that has first appeared during STEMI may be reversible, and when clinically possible, the decision to implant a permanent pacemaker should be delayed several days.

7.7.3.2.3. PACING MODE SELECTION IN STEMI PATIENTS

Class I

All patients who have an indication for permanent pacing after STEMI should be evaluated for ICD indications. (*Level of Evidence: C*)

Class IIa

- 1. It is reasonable to implant a permanent dual-chamber pacing system in STEMI patients who need permanent pacing and are in sinus rhythm. It is reasonable that patients in permanent AF or atrial flutter receive a single-chamber ventricular device. (Level of Evidence: C)
- 2. It is reasonable to evaluate all patients who have an indication for permanent pacing after STEMI for biventricular pacing (cardiac resynchronization therapy). (Level of Evidence: C)

With regard to permanent pacing mode, there are no randomized trials that specifically address pacing mode selection (dual-chamber versus single-chamber ventricular pacing, with or without rate modulation) in patients with STEMI. In general, patients who need permanent pacing systems and are in sinus rhythm should receive a permanent dual-chamber pacemaker. In contrast, patients who have permanent AF or atrial flutter should receive a ventricular-pacing system. In post-STEMI patients with frequent episodes of paroxysmal AF or a persistent episode for which eventual cardioversion is planned, the clinician must judge the likelihood of the patient being in sinus rhythm chronically before selecting a permanent dual-chamber or ventricular device. The data from randomized trials of pacing mode selection, although admittedly not obtained in a post-STEMI population, suggest specific programming features and goals to improve, or at least not further impair, LV function due to forced ventricular desynchronization by RV pacing. The DAVID (Dual Chamber and VVI Implantable Defibrillator) trial in ICD patients without bradycardia but with LV dysfunction demonstrated that patients programmed to DDD at 70 bpm developed more heart failure than did patients programmed to VVI backup at a low rate (985). A post hoc analysis of the Mode Selection Trial (MOST) reported that in dual-chamber-paced patients with sinus node dysfunction and a median ejection fraction of 0.55, cumulative percent ventricular pacing above 40% was associated with an increase in heart failure hospitalizations. These emerging data suggest that when bradycardia pacing is instituted, the clinician might consider minimizing ventricular dyssynchrony through selection of devices and programming strategies to maintain low cumulative percent ventricular pacing (986). Whether the clinical application of these concepts will improve post-STEMI outcomes is unknown because these studies did not address the specific population on which these guidelines focus. Thus, the Committee has chosen not to incorporate these programming recommendations into the guidelines, pending the design and execution of more applicable prospective trials.

Nonetheless, when a permanent pacemaker is being considered for a post-STEMI patient, the clinician should address 2 additional questions regarding the patient: is there an indication for biventricular pacing, and is there an indication for ICD use? (986) Biventricular pacing has found a place in the treatment of advanced heart failure for patients with a low ejection fraction and a QRS duration greater than 130 ms (984). Patients with severe LV dysfunction may be eligible for implantation of an ICD for primary prevention of life-threatening ventricular arrhythmia, as well as bradycardia support. The algorithm to define whether an ICD is indicated is contained in Figure 32. See Section 7.7.1.5 for further discussion.

7.8. Recurrent Chest Pain After STEMI

The 2 most common cardiac causes of recurrent chest pain after STEMI are pericarditis and ischemia, the latter being the more common and potentially more serious. An ECG taken during the recurrent pain should be compared with ECGs from the index STEMI event (987). Usually, recurrent pain within the first 12 hours after onset of STEMI is considered to be related to the original infarction itself. Pericarditis is probably not responsible for significant chest discomfort in the first 24 hours.

7.8.1. Pericarditis

Class I

- 1. Aspirin is recommended for treatment of pericarditis after STEMI. Doses as high as 650 mg orally (enteric-coated) every 4 to 6 hours may be needed. (*Level of Evidence: B*)
- 2. Anticoagulation should be immediately discontinued if pericardial effusion develops or increases. (Level of Evidence: C)

Class IIa

For episodes of pericarditis after STEMI that are not adequately controlled with aspirin, it is reasonable to administer 1 or more of the following:

- a. Colchicine 0.6 mg orally every 12 hours (Level of Evidence: B)
- b. Acetaminophen 500 mg orally every 6 hours. (Level of Evidence: C)

Class IIb

- 1. Corticosteroids might be considered only as a last resort in patients with pericarditis refractory to aspirin or NSAIDs. Although corticosteroids are effective for pain relief, their use is associated with an increased risk of scar thinning and myocardial rupture. (Level of Evidence: C)
- 2. Nonsteroidal anti-inflammatory drugs may be considered for pain relief; however, they should not be used for extended periods because of their effect on platelet function, an increased risk of myocardial scar thinning, and infarct expansion. (*Level of Evidence: B*)

Class III

Ibuprofen should not be used for pain relief because it blocks the antiplatelet effect of aspirin and it can cause myocardial scar thinning and infarct expansion. (Level of Evidence: B)

Pericarditis in STEMI occurs with extension of necrosis across the full thickness of the myocardial wall to the epicardium. Patients with pericarditis have larger infarcts, a lower ejection fraction, and a higher incidence of CHF (988,989). Pericarditis may appear up to several weeks after STEMI. Anterior chest discomfort mimicking ischemia can occur with pericarditis. However, pericardial pain usually has distinguishing characteristics, such as pleuritic and/or positional discomfort; radiation to the left shoulder, scapula, or trapezius muscle; and a pericardial rub, ECG J-point elevation with concave upward ST-segment elevation and PR depression. Detection of a 3-component rub is diagnostic of pericarditis. Pericardial effusion is evident echocardiographically in more than 40% of cases (990) but is rarely of hemodynamic consequence. A small effusion is not diagnostic of pericarditis because it can be demonstrated in the majority of patients with STEMI (991). On occasion, pericarditis may be a clinical clue to the presence of subacute myocardial rupture (see Section 7.6.7.4).

Focal pericarditis can be diagnosed electrocardiographically by either persistently positive T waves or reversal of initially inverted T waves during the first week after STEMI. However, similar T-wave alterations have also been observed when postinfarction pericardial effusion exists in the absence of clinically recognized pericarditis (992). Pericarditis is not associated with re-elevation of CK-MB. There are data to suggest its incidence has decreased in the reperfusion era (993-995). Interestingly, the Dressler syndrome (post-MI syndrome), an autoimmune-type carditis, has essentially disappeared (996) in the reperfusion era.

Neither the configuration of the ECG nor the absence or presence of the cardiac markers can absolutely establish a diagnosis of, or rule out, pericarditis. Jain described anterior ST-segment elevation secondary to acute pericarditis (997). Bonnefoy et al. studied 69 consecutive patients with idiopathic acute pericarditis (998). Cardiac troponin I was detected in 34 patients (49%), and the level of troponin was beyond the 1.5 ng/ml threshold in 15 (22%). Seven of these 15 patients underwent coronary angiography. All 7 patients had normal coronary angiograms. ST-segment elevation was found in 93% of the patients with troponin I greater than 1.5 ng/ml compared with 57% without troponin elevation (p less than 0.01). Patients with cardiac troponin I higher than 1.5 ng/ml were more likely to have had a recent infarction (66% versus 31%; p equals 0.01) and were younger (age 37 plus or minus 14 years versus 52 plus or minus 16 years; p equals 0.002).

Aspirin (162 to 325 mg/d) is the treatment of choice, but higher doses (650 mg every 4 to 6 hours) may be required (956,999). Nonsteroidal anti-inflammatory drugs may be considered for pain relief; however, they should not be used for extended periods because of their effect on platelet function, an increased risk of myocardial scar thinning, and infarct expansion. Corticosteroids, which are also efficacious for pain relief, are associated with scar thinning in the infarct zone and myocardial rupture (1000,1001). Therefore, corticosteroids should not be used except as a last resort. The risk-



Figure 34. Algorithm for management of recurrent ischemia/infarction after ST-elevation myocardial infarction (STEMI). IABP = intraaortic balloon pump; ECG = electrocardiogram; PCI = percutaneous coronary intervention; CABG = coronary artery bypass graft surgery. *Ideally within 60 minutes from the onset of recurrent discomfort. Modified with permission from Braunwald et al. Heart Disease: A Textbook of Cardiovascular Medicine. 6th ed. Philadelphia, PA: WB Saunders; 2001:1196 (718).

benefit ratio of continuing antithrombotic therapy in the presence of acute pericarditis always presents a clinical challenge. Usually such therapy can be continued safely but requires added vigilance for the detection of enlarging pericardial effusion or signs of hemodynamic instability. Any evidence of impending cardiac tamponade is an indication for prompt termination of antithrombotic therapy.

If NSAIDs are used, misoprostol, 200 mcg every 6 hours, should be used for gastric and renal protection (1002-1004). On the basis of the proven efficacy of colchicine therapy for familial Mediterranean fever, several small studies showed that colchicine could successfully treat or prevent the recurrence of acute pericarditis after conventional therapy, including corticosteroid therapy, had failed (1005). Colchicine may be administered at 0.6 mg every 12 hours, with or without a loading dose (1006-1008).

7.8.2. Recurrent Ischemia/Infarction

Class I

1. Patients with recurrent ischemic-type chest discomfort after initial reperfusion therapy for STEMI should undergo escalation of medical therapy with nitrates and beta-blockers to decrease myocardial oxygen demand and reduce ischemia. Intravenous anticoagulation should be initiated if not already accomplished. (Level of Evidence: B)

- 2. In addition to escalation of medical therapy, patients with recurrent ischemic-type chest discomfort and signs of hemodynamic instability, poor LV function, or a large area of myocardium at risk should be referred urgently for cardiac catheterization and undergo revascularization as needed. Insertion of an IABP should also be considered. (Level of Evidence: C)
- **3.** Patients with recurrent ischemic-type chest discomfort who are considered candidates for revascularization should undergo coronary arteriography and PCI or CABG as dictated by coronary anatomy. (Level of Evidence: B)

Class IIa

It is reasonable to (re)administer fibrinolytic therapy to patients with recurrent ST elevation and ischemictype chest discomfort who are not considered candidates for revascularization or for whom coronary angiography and PCI cannot be rapidly (ideally less than 60 minutes from the onset of recurrent discomfort) implemented. (*Level of Evidence: C*)



Figure 35. Algorithm for postreperfusion ischemic stroke treatment. Daily doses of antithrombotic therapy are shown in the algorithm. ASA = aspirin; INR = international normalized ratio; PTA = percutaneous transluminal angioplasty (carotid); CEA = carotid endarterectomy. *An INR of 2.0-3.0 is acceptable with tight control, but the lower end of the range is preferable. The combination of antiplatelet therapy and warfarin may be considered in patients aged less than 75 years, with low bleeding risk, and who can be monitored reliably.

Class III

Streptokinase should not be readministered to treat recurrent ischemia/infarction in patients who received a non-fibrin-specific fibrinolytic agent more than 5 days previously to treat the acute STEMI event. (Level of Evidence: C)

It is important to differentiate pain due to pericarditis from pain due to ischemia. The latter is more likely when the chest pain is similar to the initial ischemic-type chest discomfort, occurring at rest or with limited activity during hospitalization. This may or may not be associated with re-elevation of the CK-MB, ST-segment depression or elevation, or pseudonormalization of inverted T waves (T-wave inversion on baseline ECG becoming upright during ischemia) (990).

Reinfarction occurs in 4% to 5% of patients who have received fibrinolytic therapy and aspirin (25,43,851,1009,1010). Reinfarction is associated with reelevation of biomarkers after the initial peak of the index infarction. Diagnosis of reinfarction within 18 hours after initiation of fibrinolytic therapy should be based on recurrence of severe ischemic-type chest discomfort that lasts at least 30 minutes, usually but not always accompanied by recurrent ST-segment elevation of at least 0.1 mV in at least 2 contiguous ECG leads and re-elevation of CK-MB to more than the upper limit of normal or increased by at least 50% over the previous value (37). Pathological findings of reinfarction show areas of healing myocardium along with the more recent necrosis, usually in the same vascular risk region of myocardial tissue perfused by the original infarct-related artery. Death, severe CHF, and arrhythmias are early complications of reinfarction, and there is an increased incidence of cardiogenic shock or cardiac arrest (25,43,1011). An algorithm for diagnosing reinfarction both early and late after fibrinolytic therapy is shown in Figure 12. In patients with recurrent MI after fibrinolysis, the mortality rate is increased up to 2 years; however, most of the deaths occur early (in the hospital), with little additional risk of death between the index hospitalization and 2 years later (512).

Patients with recurrent ischemic-type chest discomfort should undergo escalation of medical therapy, including beta-blockers (intravenously and then orally) and nitrates (sublingually and then intravenously); consideration should be given to initiation of intravenous anticoagulation if the patient is not already therapeutically anticoagulated. Secondary causes of recurrent ischemia, such as poorly controlled heart failure, anemia, and arrhythmias, should be corrected (Figure 34) (1012).

With recurrent suspected ischemic-type chest discomfort, coronary arteriography often clarifies the cause of chest discomfort with demonstration of a high-grade coronary obstruction. For patients who are considered candidates for revascularization, prompt reperfusion with PCI or CABG is indicated as dictated by the coronary anatomy (Figure 34) (509,515,1012). For patients who are not considered candidates for revascularization or for whom coronary angiography and PCI cannot be implemented rapidly (ideally in less than 60 minutes), readministration of fibrinolytic therapy is reasonable.

Another cause of chest discomfort to consider in patients recovering from STEMI is infarct expansion. This is characterized by nonspecific repolarization of abnormalities on the ECG, worsening hemodynamics, but no re-elevation of cardiac biomarkers (43). Management of infarct expansion should focus on diuresis and inhibition of the reninangiotensin-aldosterone system (see Section 7.4.3).

7.9. Other Complications

7.9.1. Ischemic Stroke

Class I

- 1. Neurological consultation should be obtained in STEMI patients who have an acute ischemic stroke. (Level of Evidence: C)
- 2. STEMI patients who have an acute ischemic stroke should be evaluated with echocardiography, neuroimaging, and vascular imaging studies to determine the cause of the stroke. (*Level of Evidence: C*)
- 3. STEMI patients with acute ischemic stroke and persistent AF should receive lifelong moderate-intensity (INR 2 to 3) warfarin therapy. (Level of Evidence: A)
- 4. STEMI patients with or without acute ischemic stroke who have a cardiac source of embolism (AF, mural thrombus, or akinetic segment) should receive moderate-intensity (INR 2 to 3) warfarin therapy in addition to aspirin (see Figure 35). The duration of warfarin therapy should be dictated by clinical circumstances (e.g., at least 3 months for patients with an LV mural thrombus or akinetic segment and indefinitely in patients with persistent AF). The patient should receive LMWH or UFH until adequately anticoagulated with warfarin. (Level of Evidence: B)

Class IIa

- **1.** It is reasonable to assess the risk of ischemic stroke in patients with STEMI. (*Level of Evidence: A*)
- 2. It is reasonable that STEMI patients with nonfatal acute ischemic stroke receive supportive care to minimize complications and maximize functional outcome. (Level of Evidence: C)

Class IIb

Carotid angioplasty/stenting, 4 to 6 weeks after ischemic stroke, might be considered in STEMI patients who have an acute ischemic stroke attributable to an internal carotid artery-origin stenosis of at least 50% and who have a high surgical risk of morbidity/mortality early after STEMI. (Level of Evidence: C)

Acute stroke complicates 0.75% to 1.2% of MIs and is one of the most dreaded outcomes of STEMI (326,1013,1014). Although survival from STEMI has been increasing, mortality from post-STEMI stroke remains over 40% (1013). Prior stroke, hypertension, old age, decreased ejection fraction or multiple ulcerated plaques, and AF are the major risk factors for embolic stroke after STEMI (326,944,1015-1017). Anterior STEMI is often cited as a risk factor, but other infarct locations appear to have similar risk (1015,1018). AF is by far the most important of these risk factors (1019). In the SAVE trial (1017), decreased ejection fraction (18%) increased risk per every 5% decrease in ejection fraction) was independently associated with long-term stroke risk in patients with STEMI. Thrombus formation is promoted by extensive wall-motion abnormality, such as anteroapical akinesia or dyskinesia, and Killip class III or IV (1020). Embolic stroke after STEMI originates from LV thrombus or from the left atrium in the setting of AF and occurs even in patients treated with fibrinolysis (1015). It does not appear to be useful to test patients with STEMI, even with mural thrombus formation, for prothrombotic syndromes such as factor V Leiden mutation (1021). Several studies in patients with STEMI suggest that aggressive short- and long-term anticoagulation may reduce but not totally prevent mural thrombus formation (744,1022,1023) and occurrence of stroke (1024-1028). Most ischemic cerebral infarctions after fibrinolytic therapy for STEMI occur more than 48 hours after treatment (218,320,322,586). The highest-risk period is the first 28 days after STEMI (1014), but risk is elevated at least to 1 year. In GUSTO-I (1015), Mahaffey et al. found that the risk of ischemic stroke after coronary fibrinolysis may be predicted (1015). Prospective studies are needed to verify this observation. Compared with ICH, patients with ischemic cerebral infarction present more commonly with focal neurological deficits and less commonly with depressed level of consciousness; headache, vomiting, and coma are uncommon (360).

An algorithm for evaluation and antithrombotic therapy for ischemic stroke is shown in Figure 35. If the STEMI patient has sudden onset of a focal neurological deficit and the initial CT scan is negative for blood or mass effect, then ischemic cerebral dysfunction may be presumed in the absence of a severe metabolic disorder, seizures, autoimmune disease, or cancer. Neurological consultation is recommended to assist with planning the neurovascular evaluation and management issues. The location and nature of the ischemic brain lesion should be defined with repeat CT scan or magnetic resonance imaging scan. Vascular lesions should be evaluated with noninvasive techniques, such as carotid duplex sonography, transcranial Doppler, magnetic resonance angiography, CT angiography, or transesophageal echocardiography. For carotid territory symptoms and signs, evidence for a surgically important stenosis (greater than 50% linear diameter reduction on a catheter-based cerebral angiogram using the North American Symptomatic Carotid Endarterectomy Trial [NASCET] method) (1029) should clearly be sought.

The subacute to chronic pathophysiology of STEMI provides a rationale for clinicians to consider when choosing antithrombotic therapies for stroke prevention and treatment in this setting. A hypercoagulable state may exist for up to 6 months after STEMI (1030). This state may be accentuated by withdrawal of heparin and warfarin therapy (555).

In ISIS-2 (1031), use of aspirin was shown to reduce the occurrence of ischemic stroke. Thus, aspirin administration after STEMI, with or without ischemic cerebral infarction, is appropriate. However, for patients with ischemic cerebral infarction who undergo PCI and have no cardioembolic risk factors, the use of clopidogrel 75 mg/d (for at least 12 months) plus aspirin 75 to 162 mg/d (indefinitely) after STEMI is reasonable (578,728,1032). Patients with STEMI who have ischemic stroke but do not undergo PCI and do not have a cardiac source of embolism or surgically important carotid stenosis may be treated with aspirin/extended-release dipyridamole 25/200 mg plus aspirin 81 mg/d (1033).

A subgroup analysis from the CAPRIE trial (1032) in patients with a prior history of ischemic events suggested a benefit of clopidogrel (mean duration of treatment 1.6 years) over aspirin for the composite end point of ischemic stroke, MI, or vascular death (3.4% ARD, 95% CI 0.2 to 7.0; 14.9% RRR; p equals 0.045). Thus, for the STEMI patient who has an ischemic stroke without a documented cardiac source of embolism while on aspirin therapy, clopidogrel may be added for a period of 18 months.

Patients with cardiogenic sources of embolism, such as AF, LV mural thrombi, or akinetic segment of the LV myocardium, should receive moderate-intensity (INR 2 to 3) warfarin anticoagulation in combination with aspirin. The duration of moderate-intensity warfarin anticoagulation will vary according to the underlying strong source of cardiogenic embolism. Ischemic stroke patients with pre-existing or persistent AF require lifelong warfarin therapy (958), irrespective of 2-dimensional echocardiography findings. In general, patients with STEMI with LV mural thrombus should receive 3 months of warfarin therapy, a time believed to be sufficiently long for clot adherence and re-endothelialization to occur, leading to reduced embolic risk. However, if followup 2-dimensional echocardiography at 3 months in the STEMI patient with acute ischemic stroke shows findings that suggest an ongoing risk of cardiogenic embolism (e.g., new or enlarging mural thrombi or thrombi that are pedunculated or mobile), then long-term moderate-intensity warfarin anticoagulation is recommended.

If a surgically important internal carotid artery stenosis that explains the clinical findings is found, either carotid endarterectomy (1029,1034,1035) or carotid angioplasty with stenting and a distal protection device (1036,1037) (Yadav J; oral presentation, 2002 American Heart Association Annual Scientific Session, November 2002, Chicago, IL) could be performed. If surgical morbidity and mortality are acceptable, carotid endarterectomy may be performed 4 to 6 weeks after cerebral infarction. Preliminary data from 307 randomized patients in the Stenting and Angioplasty with Protection in Patients at High Risk for Endarterectomy (SAPPHIRE) trial (Yadav J; oral presentation, 2002 American Heart Association Annual Scientific Session, November 2002, Chicago, IL) showed a significant reduction in the composite end point of 30-day death, STEMI/NSTEMI, or stroke in the stent arm compared with the endarterectomy arm (5.8% versus 12.6%, p equals 0.047). There was no difference in the occurrence of TIAs or major bleeding between the 2 groups, although the occurrence of cranial nerve injury was significantly higher in the endarterectomy arm (0% versus 5.3%, p less than 0.01). One-year follow-up data are pending.

7.9.2. Deep Venous Thrombosis and Pulmonary Embolism

Class I

- 1. Deep venous thrombosis or pulmonary embolism after STEMI should be treated with full-dose LMWH for a minimum of 5 days and until the patient is adequately anticoagulated with warfarin. Start warfarin concurrently with LMWH and titrate to INR of 2-3. (Level of Evidence: A)
- 2. Patients with CHF after STEMI who are hospitalized for prolonged periods, unable to ambulate, or considered at high risk for DVT and are not otherwise anticoagulated should receive low-dose heparin prophylaxis, preferably with LMWH. (*Level of Evidence: A*)

Prevention. Deep venous thrombosis and pulmonary embolism historically were relatively frequent complications of STEMI, but in the current era in which patients with STEMI almost universally receive anticoagulants, specific prophylaxis is seldom needed (1039). For patients with CHF after STEMI who are hospitalized for prolonged periods or unable to ambulate and who are not otherwise anticoagulated, the best evidence of safety and efficacy supports the use of low-dose LMWH (1040). Dosing depends on the specific LMWH chosen, and product-specific information should be consulted at the time of treatment.

Treatment. A high index of suspicion for DVT and pulmonary embolism is necessary, and patients suspected of having either condition should be evaluated immediately with an appropriate evidence-based diagnostic strategy (1041). Most patients with DVT or pulmonary embolism should be anticoagulated with LMWH. It is at least as effective as UFH in clinical trials, and a meta-analysis suggests it offers a lower total mortality (1042). Outside of carefully

Time Since Operation	n*	Transmural MI Mortality, %	Nontransmural MI Mortality, %
Less than 6 hours	885	12.1	11.5
6-23 hours	556	13.6	6.2
1-7 days	7554	4.3	3.5
8-14 days	6712	2.4	2.7
Greater than or equal to 15 days	28658	2.6	2.7

 Table 31. Comparison of Hospital Mortality for CABG With Respect to Time of Operation:

 Transmural Versus Nontransmural MI

CABG = coronary artery bypass graft surgery; MI = myocardial infarction.

*Number of the 44 365 patients in the NY State Cardiac Surgery Registry who underwent CABG as the sole procedure from 1993-1996 and whose data were used in the analysis.

Modified from Lee et al. Ann Thorac Surg 2001;71:1197-202, Copyright © 2001, with permission from the Society of Thoracic

supervised clinical trials, LMWH is generally superior, because overshooting and undershooting the therapeutic range is commonplace with UFH in routine clinical practice. Low-molecular-weight heparin is less costly overall because it avoids intravenous administration and frequent laboratory testing. Dosing of LMWH depends on the specific product, and product-specific information should be consulted at the time of treatment. Warfarin should be initiated concurrently with LMWH, and LMWH should be continued until the INR reaches the therapeutic range of 2 to 3 (1041,1043). Warfarin should be continued for a duration specific to the individual patient's risk profile (1044). Patients with contraindications to anticoagulation with heparins will require alternative therapies, and some will require placement of an inferior vena cava filter. Detailed discussion of warfarin therapy duration, alternative anticoagulants, and vena cava filter placement criteria are beyond the scope of this guideline, and the reader should consult an evidence-based guideline on venous thromboembolic disease (1041,1043).

7.10. Coronary Artery Bypass Graft Surgery After STEMI

7.10.1. Timing of Surgery

Class IIa

In patients who have had a STEMI, CABG mortality is elevated for the first 3 to 7 days after infarction, and the benefit of revascularization must be balanced against this increased risk. Patients who have been stabilized (no ongoing ischemia, hemodynamic compromise, or life-threatening arrhythmia) after STEMI and who have incurred a significant fall in LV function should have their surgery delayed to allow myocardial recovery to occur. If critical anatomy exists, revascularization should be undertaken during the index hospitalization. (*Level of Evidence: B*)

There are no clear data regarding optimal timing of bypass surgery after STEMI (518,1045-1047). Most published studies are retrospective and observational in nature, with heterogenous patient cohorts and differences in inclusion criteria. Moreover, many patients undergoing emergency surgery tend to have comorbidities and risk factors, including decreased LV function, which are normally associated with an increase in operative mortality.

Coronary artery bypass grafting early after STEMI may carry substantial risk, particularly in unstable patients with Q-wave infarction and decreased LV function. Prior CABG, female gender, and advanced age compound this risk considerably (1045). Lee et al. (1048) reviewed 44 365 patients from New York State undergoing CABG. Mortality for those with or without a history of MI was 3.1% and 1.6%, respectively. Mortality was higher for patients with transmural infarction who underwent surgery within the first 24 hours (12.1% at less than 6 hours, 13.6% at 6 to 23 hours). At 2 weeks, mortality was 2.6%. Mortality for patients with nontransmural MIs was also elevated (11.5% at less than 6 hours, 6.2% at 6 to 23 hours) (Table 31) (1048). In another large series of 2296 patients undergoing CABG after MI, Creswell et al. reported mortality from surgery to be 9.1% at less than 6 hours after MI, 8.3% at 6 to 48 hours, 5.2% at 2 to 14 days, 6.5% at 2 to 6 weeks, and 2.9% at greater than 6 weeks. Urgency of surgery was the most important predictor of operative mortality (1049). When adjustments were made for the independent risk factors of urgency of operation, increased patient age, renal insufficiency, number of previous MIs, and hypertension, the timing between MI and CABG was not a significant predictor for death.

In reviewing 11 retrospective and prospective observational studies regarding CABG and MI, Crossman et al. (1046) concluded that timing of surgery after infarction was not necessarily an independent predictor of outcome. However, these studies did "appear to support an approach of medical stabilization for unstable patients post MI wherever possible to convert high risk emergency operations to lower risk more elective procedures" (1046). The Writing Committee believes that if stable patients with STEMI with preserved LV function require surgical revascularization, then CABG can be undertaken within several days of the infarction without an increased risk. In addition, surgery for patients who have mechanical complications of MI, such as VSR or papillary muscle rupture, or who have ongoing ischemia that has been unresponsive to other medical therapy and have vessels suitable for bypass, cannot usually be delayed. However, patients who have had a significant decrease in LV function as a result of STEMI and who are hemodynamically stable may benefit from a longer period of medical treatment to allow myocardial recovery to occur before surgical revascularization is undertaken. If a patient has critical coronary anatomy, such as greater than 75% left main coronary artery stenosis, then CABG should be undertaken during the same hospitalization. (See Section 3.2.2 of the ACC/AHA Guidelines for Coronary Artery Bypass Graft Surgery (518).)

7.10.2. Arterial Grafting

Class I

An internal mammary artery graft to a significantly stenosed LAD should be used whenever possible in patients undergoing CABG after STEMI. (Level of Evidence: B)

The routine use of the left internal mammary artery (IMA) for LAD grafting, with supplemental saphenous vein grafts to other coronary lesions, is generally accepted as the standard grafting method. A large, long-term follow-up study comparing patients receiving the left IMA-to-LAD and supplemental vein grafts to patients receiving saphenous vein grafts only demonstrated a significantly lower rate of recurrent angina and MI, a lower incidence of reoperation or PCI, and a higher actuarial 10-year survival among patients with IMA grafts (1050). Despite the clear advantage of the IMA graft, not all patients receive arterial grafts. In the PAMI-2 trial (522) of 120 patients undergoing surgery, only 31% received an IMA graft; no patients had cardiogenic shock at the time of surgery. On the other hand, Hirose et al. (1051) used arterial conduits in 96% of 47 patients undergoing emergency CABG within a mean of 27 hours from infarction, and operative mortality was 6.4%. In a further study, Hirotani et al. (1052) performed CABG on 68 patients within 30 days of infarction, with a mortality rate of 7.4%. CABG without arterial grafts was the sole predictor of an adverse survival. Although surgeons may be reluctant to use the IMA in patients after STEMI because of limited flow through the arterial graft and the time necessary to harvest it, the data suggest that IMA grafts can be used safely soon after STEMI with no increase in mortality; their use is associated with better long-term survival.

7.10.3. Coronary Artery Bypass Graft Surgery After Fibrinolytic Therapy

In the 3339 patients enrolled in the TIMI-II trial, CABG was performed as an emergency procedure (1.6%) or electively (10% during initial hospitalization), primarily for left main coronary stenosis or coronary anatomy not amenable to PCI and continuing, recurrent, or exercise-induced ischemia (1053). Of the 41 021 patients enrolled in the GUSTO-I trial, CABG was used in 8.6% at a mean of 8.5 days after fibrinolytic therapy (1054). Unstable patients undergoing CABG shortly after fibrinolytic therapy, primarily for continuing myocardial ischemia, have a higher operative mortality rate (13% to 17%) and increased use of blood products (1053,1055,1056) than hemodynamically stable patients operated on within 8 hours of fibrinolytic therapy, who have a relatively low (2.8%) mortality rate (1057). The only independent predictor of perioperative mortality in TIMI-II was performance of CABG within 24 hours of entry or PCI. The low 1-year mortality rate (2.2%) noted for operative survivors in this group may support the use of emergency operation for selected patients, however (1053). In the SHOCK trial (301), CABG was as successful as PCI in reducing mortality in patients with cardiogenic shock, although they had more complex coronary artery anatomy. The intraoperative use of aprotinin may reduce hemorrhage related to use of thrombolytic agents (1058).

7.10.4. Coronary Artery Bypass Graft Surgery for Recurrent Ischemia After STEMI

Class I

Urgent CABG is indicated if the coronary angiogram reveals anatomy that is unsuitable for PCI. (Level of Evidence: B)

Coronary artery bypass graft surgery should be considered when recurrent ischemia occurs in patients with STEMI whose coronary artery anatomy is not suitable for PCI. Operative mortality in such patients is correlated closely with ejection fraction, and for patients with normal ejection fraction, it is nearly the same as that of elective CABG (1059-1061). The survival benefit for patients with reduced LV function supports the use of CABG in this situation.

7.10.5. Case Selection Concerns in CABG After STEMI

As cardiac surgical programs and individual surgeons come under scrutiny with regard to operative mortality rates, concern has been raised about the possibility that salvageable but high-risk patients may not be offered surgery. Omoigui et al. (1062) suggested that the reduction in mortality noted in New York State was caused by an outmigration of high-risk patients due to the increased scrutiny provoked by public release of mortality data. In a survey of New York State cardiac surgeons, Burack et al. (1063) reported that many surgeons refused to operate on at least 1 high-risk patient over the prior year, primarily because of public reporting. The Writing Committee believes strongly that patients should be offered surgical treatment if the treating team believes that the benefits outweigh the risks and that meaningful survival of the patient could result. Furthermore, appropriately validated risk-adjusted outcome measures should be used when evaluating the performance of an individual surgeon or surgical program. The ACC/AHA Guideline Update on Coronary Artery Bypass Graft Surgery has addressed the issue of institutional and individual surgical caseloads (518). Studies suggest that survival after CABG has improved over the last decade but is negatively affected when the surgery is performed in institutions that do fewer than a threshold of 200 cases per year. However, some institutions and practitioners maintain excellent outcomes despite relatively low volumes. In reviewing the outcome of 13 644 CABG procedures at 56 US hospitals, mortality was significantly lower at high-volume hospitals only for those patients considered as being at high or moderate risk (1064). Therefore, targeted transfer of high- to moderate-risk post-STEMI patients to high-volume institutions may be appropriate.

7.10.6. Elective CABG After STEMI in Patients With Angina

Class I

- 1. Coronary artery bypass graft surgery is recommended for patients with stable angina who have significant left main coronary artery stenosis. (*Level of Evidence: A*)
- 2. Coronary artery bypass graft surgery is recommended for patients with stable angina who have left main equivalent disease: significant (at least 70%) stenosis of the proximal LAD and proximal left circumflex artery. (Level of Evidence: A)
- **3.** Coronary artery bypass graft surgery is recommended for patients with stable angina who have 3-vessel disease. (Survival benefit is greater when LVEF is less than 0.50.) (Level of Evidence: A)
- 4. Coronary artery bypass graft surgery is beneficial for patients with stable angina who have 1- or 2-vessel coronary disease without significant proximal LAD stenosis but with a large area of viable myocardium and high-risk criteria on noninvasive testing. (Level of Evidence: B)
- 5. Coronary artery bypass graft surgery is recommended in patients with stable angina who have 2-vessel disease with significant proximal LAD stenosis and either ejection fraction less than 0.50 or demonstrable ischemia on noninvasive testing. (Level of Evidence: A)

The role of surgical revascularization has been reviewed extensively in the ACC/AHA Guideline Update on Coronary Artery Bypass Graft Surgery (518). Consideration for revascularization after STEMI includes PCI and CABG. Providers should individualize patient management on the basis of clinical circumstances, available revascularization options, and patient preference. Elective CABG should improve survival relative to medical therapy in patients with MI who have 1) left main coronary artery stenosis; 2) left main equivalent (significant [at least 70%] stenosis of the proximal LAD and proximal left circumflex artery); 3) 3-vessel disease, particularly with decreased LV function; 4) 2-vessel disease with significant proximal LAD stenosis not amenable to PCI and either ejection fraction less than 0.50 or demonstrable ischemia on noninvasive testing; and 5) 1- or 2-vessel disease not amenable to PCI without proximal LAD stenosis but with a large area of viable myocardium at risk and high-risk criteria on noninvasive testing. The optimal

timing of surgery has not been established. The risk is greatest within the first 48 hours of infarction and decreases over the next 2 weeks. Risk of operation is greatest for patients with decreased LV function, advanced age, female sex, renal failure, peripheral vascular disease, diabetes, chronic obstructive pulmonary disease, and previous CABG (1048,1061,1065).

7.10.7. Coronary Artery Bypass Surgery After STEMI and Antiplatelet Agents

Class I

- 1. Aspirin should not be withheld before elective or nonelective CABG after STEMI. (Level of Evidence: C)
- 2. Aspirin (75 to 325 mg/d) should be prescribed as soon as possible (within 24 hours) after CABG unless contraindicated. (*Level of Evidence: B*)
- 3. In patients taking clopidogrel in whom elective CABG is planned, the drug should be withheld for 5 to 7 days. (Level of Evidence: B)

Aspirin therapy, particularly within the first 48 hours after CABG, appears to have significant benefits (1066,1067). Mangano et al. (1066) studied 5022 patients in a global registry who survived CABG. Aspirin therapy begun within 48 hours after surgery resulted in a 60% lower death rate at 30 days, as well as decreased rates of MI, stroke, renal failure, and bowel infarction. It is likely that these effects are mediated by both the anti-inflammatory and antithrombotic actions of aspirin (1067). Bleeding complications were also lower in the aspirin-treated group.

Patients with STEMI undergoing revascularization frequently receive 1 or more antiplatelet agents in addition to heparin, all of which may increase risk of serious bleeding during and after cardiac surgery. Because the mechanism and duration of action of the antiplatelet effects of aspirin, the ADP antagonists (ticlopidine and clopidogrel), and GP IIb/IIIa receptor antagonists (abciximab and eptifibatide) differ, the potential exists for an additive effect with a combination of these agents. In the CURE trial (728), in which 12 562 patients with unstable angina or NSTEMI were randomized to placebo plus aspirin or clopidogrel plus aspirin, there was an increased risk of major bleeding in the clopidogrel group (3.7%) compared with the placebo group (2.7%). Also, risk of bleeding was increased in patients undergoing CABG within the first 5 days of stopping clopidogrel. In a prospective study of 224 patients having CABG, Hongo et al. (1068) found reoperation for bleeding to be 10-fold higher in patients who had received clopidogrel within 7 days of surgery than in those who had received no clopidogrel. In the trial (Evaluation Prevention of Ischemic EPIC Complications; placebo or abciximab bolus or abciximab infusion) (1069), transfusion of red blood cells and platelets was significantly higher after treatment with abciximab.

Singh et al. (1070) found transfusion requirements to be higher for patients who underwent urgent CABG after having received abciximab during PCIs. In an analysis of 85 sur-



Figure 36. Evidence-based approach to need for catheterization (cath) and revascularization after ST-elevation myocardial infarction (STEMI). The algorithm shows treatment paths for patients who initially undergo a primary invasive strategy, receive fibrinolytic therapy, or do not undergo reperfusion therapy for STEMI. Patients who have not undergone a primary invasive strategy and have no highrisk features should undergo functional evaluation with one of the noninvasive tests shown. When clinically significant ischemia is detected, patients should undergo catheterization and revascularization as indicated; if no clinically significant ischemia is detected, medical therapy is prescribed after STEMI. *Please see Table 23 of the ACC/AHA 2002 Guideline Update for the Management of Patients With Chronic Stable Angina for further definition (71). †Please see Table 3, Section 6.3.1.6.2 and Section 7.3 in the STEMI guideline for further discussion.

gical patients in the EPILOG (Evaluation of PTCA to Improve Long-Term Outcome by c7E3 GP IIb/IIIa Receptor Blockade) and EPISTENT (Evaluation of Platelet IIb/IIIa Inhibition in Stenting) trials, although platelet transfusions were higher in the abciximab groups, rates of major blood loss and transfusions of whole blood and packed red cells were the same in the abciximab and placebo groups (1071). Management strategies for patients who require surgery and subsequent treatment with antiplatelet agents will differ according to type of agent used and the urgency of surgery.

In many circumstances, it may not be feasible to delay surgery until platelet function has recovered. In patients treated with the small-molecule GP IIb/IIIa receptor antagonists, tirofiban and eptifibatide, platelet function returns toward normal within 4 hours of stopping treatment. Platelet aggregation does not return toward normal for more than 48 hours in patients treated with abciximab. Management strategies other than delaying surgery include platelet transfusions for patients who were recently treated with abciximab, reduced heparin dosing during cardiopulmonary bypass, and possible use of antifibrinolytic agents such as aprotinin or tranexamic acid (1072). Because clopidogrel, when added to aspirin, increases the risk of bleeding during major surgery, clopidogrel should be withheld for at least 5 days (728) and preferably for 7 days before surgery in patients who are scheduled for elective CABG (1073).

7.11. Convalescence, Discharge, and Post-MI Care

7.11.1. Risk Stratification at Hospital Discharge

Informal risk estimates are updated continually as the patient's clinical condition evolves and other information becomes available. For example, tests such as an echocardiogram may be obtained early during hospitalization to define an abnormal physical finding and thus provide data for risk stratification, even though they were not ordered for that purpose. Because patient preference is a critical determinant for any management pathway, it is difficult to produce a rigid algorithm for risk stratification.

There is, nevertheless, value in outlining general strategies for performing risk stratification testing, with an emphasis on avoiding redundancy. The most important immediate management decision is whether to refer a patient for cardiac catheterization. Patients who have not had catheterization as

Antman *et al.* 2004 e213 ACC/AHA Practice Guidelines

Table 32. Secondary Prevention for Patients With STEMI

Goals	Intervention Recommendations
Smoking:	Assess tobacco use. Strongly encourage patient and family to stop smoking and to avoid secondhand smoke
<u>Goal</u> Complete cessation	Provide counseling, pharmacological therapy (including nicotine replacement and bupropion), and formal smoking cessation programs as appropriate. (See Section 7.12.4 in the full-text guidelines for further discussion.)
Blood pressure control: <u>Goal</u> Less than 140/90 mm Hg or less than 130/80 mm Hg if	If blood pressure is 120/80 mm Hg or greater:Initiate lifestyle modification (weight control, physical activity, alcohol moderation, moderate sodium restriction, and emphasis on fruits, vegetables, and low-fat dairy products) in all patients.
chronic kidney disease or diabetes	If blood pressure is 140/90 mmHg or greater or 130/80 mmHg or greater for individuals with chronic kidney disease or diabetes:
	• Add blood pressure-reducing medications, emphasizing the use of beta-blockers and inhibitors of the renin-angiotensin-aldosterone system. (See Sections 7.12.6, 7.12.7, and 7.12.8.)
Lipid management: (TG less than 200 mg/dL)	Start dietary therapy in all patients (less than 7% of total calories as saturated fat and less than 200 mg/d cholesterol). Promote physical activity and weight management. Encourage increased consumption of omega-3 fatty acids.
Primary goal: LDL-C substantially less	Assess fasting lipid profile in all patients, preferably within 24 hours of STEMI. Add drug therapy according to the following guide. (See Section 7.12.2.)
than 100 mg/dL	LDL-C less than 100 mg/dL (baseline or on treatment): LDL-C greater than or equal to 100 mg/dL (baseline or on treatment):
	• Statins should be used to lower LDL-C. • Intensify LDL-C–lowering therapy with drug treatment, giving prefer- ence to statins.
Lipid management: (TG 200 mg/dL or greater)	If TGs are greater than or equal to 150 mg/dL or HDL-C is less than 40 mg/dL: • Emphasize weight management and physical activity. Advise smoking cessation.
<u>Primary goal</u> Non–HDL-C* substantially less than 130 mg/dL	If TG is 200-499 mg/dL: • After LDL-C–lowering therapy,† consider adding fibrate or niacin.‡ If TG is greater than or equal to 500 mg/dL: • Consider fibrate or niacin‡ before LDL-C–lowering therapy.†
	• Consider omega-3 fatty acids as adjunct for high TG. (See Section 7.12.2.)
Physical activity: <u>Minimum goal</u> 30 minutes 3 to 4 days per week Optimal daily	Assess risk, preferably with exercise test, to guide prescription. Encourage minimum of 30 to 60 minutes of activity, preferably daily but at least 3 or 4 times weekly (walking, jogging, cycling, or other aerobic activity) supplemented by an increase in daily lifestyle activities (e.g., walking breaks at work, gardening, household work). Cardiac rehabilitation programs are recommended for patients with STEMI, particularly those with multiple modifiable risk factors and/or those moderate- to high-risk patients in whom supervised exercise training is warranted. (See Sections 7.12.12 and 8.2.)
Continued on next page	

Table 32. Continued

Goals	Intervention Recommendations
Weight management:	Calculate BMI and measure waist circumference as part of evaluation. Monitor response of BMI and
Goal	waist circumference to therapy.
BMI 18.5-24.9 kg/m ²	Start weight management and physical activity as appropriate. Desirable BMI range is 18.5-24.9 kg/m ² .
Waist circumference:	If waist circumference is greater than or equal to 35 inches in women or greater than or equal to 40 inches
Women: less than 35 in	in men, initiate lifestyle changes and treatment strategies for metabolic syndrome.
Men: less than 40 in	(See Section 7.12.3.)
Diabetes management:	
Goal	Appropriate hypoglycemic therapy to achieve near-normal fasting plasma glucose, as indicated by HbA_{1c} .
HbA1C less than 7%	Treatment of other risk factors (e.g., physical activity, weight management, blood pressure, and choles- terol management). (See Section 7.12.9.)
Antiplatelet agents/ anticoagulants	Start and continue indefinitely aspirin 75 to 162 mg/d if not contraindicated. Consider clopidogrel 75 mg/d or warfarin if aspirin is contraindicated. Manage warfarin to INR 2.5 to 3.5 in post-STEMI patients when clinically indicated or for those not able to take aspirin or clopidogrel. (See Sections 7.12.5 and 7.12.11 and Figure 37 for further details of antiplatelet and anticoagulant therapy at hospital discharge.)
Renin-angiotensin-	ACE inhibitors in all patients indefinitely; start early in stable high-risk patients (anterior MI, previous MI, Killin class greater than or equal to U.S. callen rates ratiographic CUE). LVEE loss than 0.40
system blockers	with Kninp class greater than of equal to $\Pi [S_3]$ gamop, rates, radiographic CHFJ, LVEF less than 0.40).
	Angiotensin receptor blockers in patients who are intolerant of ACE inhibitors and with either clinical or
	radiological signs of heart failure or LVEF less than 0.40.
	Aldosterone blockade in patients without significant renal dysfunction§ or hyperkalemial who are already receiving therapeutic doses of an ACE inhibitor, have an LVEF less than or equal to 0.40, and have either
	diabetes or heart failure. (See Section 7.12.6.)
Beta-Blockers	Start in all patients. Continue indefinitely. Observe usual contraindications. (See Section 7.12.7.)

STEMI = ST-elevation myocardial infarction; TG = triglycerides; LDL-C = low-density lipoprotein cholesterol; HDL-C = high-density lipoprotein cholesterol; BMI = body mass index; INR = international normalized ratio; ACE = angiotensin converting enzyme; MI = myocardial infarction; CHF = congestive heart failure; LVEF = left ventricular ejection fraction. *Non-HDL-C = total cholesterol minus HDL-C.

†Treat to a goal of non-HDL-C substantially less than 130 mg/dL.

Dietary-supplement niacin must not be used as a substitute for prescription niacin, and over-the-counter niacin should be used only if approved and monitored by a physician.

Screatinine should be less than or equal to 2.5 mg/dL in men or less than or equal to 2.0 mg/dL in women.

IPotassium should be less than 5.0 mEq/L.Modified with permission from Smith et al. Circulation 2001;104:1577-9 (68).

Modified from Smith et al. Circulation 2001;104:1577-9 (68).

part of their initial treatment strategy should be evaluated for their risk of future cardiac events. Another major decision is referral for EP testing and possible placement of an ICD.

The risk stratification approach for decision making about catheterization is described in Figure 36. For patients who did not have an early catheterization and in whom revascularization therapy would be considered, the approach includes an early quantitative assessment of LVEF. Patients with an LVEF less than 0.40 should be considered for catheterization. For those with a higher LVEF, further stratification with stress testing is recommended. Various approaches are considered acceptable, including a submaximal stress test, a symptom-limited stress imaging test, or a pharmacological stress imaging test. Ambulatory ECG monitoring for ischemia after STEMI has significant limitations (e.g., resting ECG abnormal after infarction) and has not been incorporated into the algorithm in Figure 36 (71,1074).

A symptom-limited stress test was not formerly recommended but is supported by data from DANAMI (515). The decision for referral to cardiac catheterization should then be based on the stress test results, with an indication related to the amount of ischemia quantified by test results. Patients with little or no ischemia may be better treated with medical therapy, a strategy supported by DANAMI.

The suggested algorithm for EP testing and ICD placement is shown in Figure 32. Electrophysiology testing is not indicated for patients with an LVEF less than 0.30 because they are presumed to benefit from an ICD, and an EP test is not necessary. Patients with VF after 48 hours, sustained VT, or hemodynamically significant VT are recommended for ICD placement without an EP test. For patients with an ejection fraction less than 0.40, an EP test is reserved those whose telemetry or Holter monitor show nonsustained VT. Physicians and their patients must make decisions about risk-reduction lifestyle and pharmacological approaches. At this point, however, all patients with STEMI are considered to be at sufficiently high risk to merit use of these secondary prevention interventions, including the use of cardiac rehabilitation, aspirin, appropriate lipid-lowering therapy, and beta-blockers (Table 32) (68).

There is a clear need for tools that can integrate comprehensive risk-stratification information in explicit estimates of absolute risks of cardiovascular outcomes with management decisions. Current risk stratification tests generally provide relative risk information, indicating average increases in risk, but their utility in guiding clinical decisions remains to be defined. The Writing Committee believes that further research in this area is necessary before making recommendations about the best use of these emerging risk stratification tools.

7.11.1.1. Role of Exercise Testing

Class I

- **1.** Exercise testing should be performed either in the hospital or early after discharge in STEMI patients not selected for cardiac catheterization and without high-risk features to assess the presence and extent of inducible ischemia. (*Level of Evidence: B*)
- 2. In patients with baseline abnormalities that compromise ECG interpretation, echocardiography or myocardial perfusion imaging should be added to standard exercise testing. (Level of Evidence: B)

Class IIb

Exercise testing might be considered before discharge of patients recovering from STEMI to guide the postdischarge exercise prescription or to evaluate the functional significance of a coronary lesion previously identified at angiography. (Level of Evidence: C)

Class III

- 1. Exercise testing should not be performed within 2 to 3 days of STEMI in patients who have not undergone successful reperfusion. (Level of Evidence: C)
- 2. Exercise testing should not be performed to evaluate patients with STEMI who have unstable postinfarction angina, decompensated CHF, life-threatening cardiac arrhythmias, noncardiac conditions that severely limit their ability to exercise, or other absolute contraindications to exercise testing (1075). (Level of Evidence: C)
- 3. Exercise testing should not be used for risk stratification in patients with STEMI who have already been selected for cardiac catheterization. (Level of Evidence: C)

Exercise testing after STEMI may be performed to 1) assess functional capacity and the patient's ability to perform tasks at home and at work; 2) establish exercise parameters for cardiac rehabilitation; 3) evaluate the efficacy of the patient's current medical regimen; 4) risk-stratify the post-

patient with STEMI according to the likelihood of a subsequent cardiac event (1076-1080); 5) evaluate chest pain symptoms after STEMI; and 6) provide reassurance to patients regarding their functional capacity after STEMI as a guide to return to work.

Patients who receive reperfusion therapy have a smaller infarct size (1081). Coronary angiography is frequently performed during hospitalization due to recurrent chest pain, which identifies many patients with severe disease who subsequently undergo revascularization (1082). The low cardiac event rate after discharge in patients with STEMI who are successfully reperfused substantially reduces the predictive accuracy of early exercise testing. The ability to perform an exercise test 1 month after STEMI provides a favorable prognosis irrespective of the test results (1075).

Low-level exercise testing appears to be safe if patients have undergone in-hospital cardiac rehabilitation, including low-level exercise, have had no symptoms of angina or heart failure, and have a stable baseline ECG 48 to 72 hours before the exercise test. Two different protocols have been used to determine the end points of these very early exercise tests (1083-1085):

The traditional submaximal exercise test (done at 3 to 5 days in patients without complications) incorporates a series of end points, including a peak heart rate of 120 to 130 bpm or 70% of maximal predicted heart rate for age, a peak work level of 5 metabolic equivalents (METs), or clinical or ECG end points of mild angina or dyspnea, ST-segment depression greater than 2 mm, exertional hypotension, or 3 or more consecutive premature ventricular contractions, whichever end point is reached first. The second protocol is performance of a symptom-limited exercise test (done at 5 days or later) without stopping for target heart rates or MET levels. Although this level will result in a higher frequency of abnormal exercise tests, the prognostic value of ST depression that occurs at higher work levels in deconditioned patients is uncertain. The safety of early symptom-limited exercise testing is based on relatively limited data; therefore, clinical judgment must be used (1075). The results of this symptomlimited test can also be used to establish intensity and target heart rate during cardiac rehabilitation.

The duration of exercise is also known to be an important predictor of outcomes, and the ability to perform at least 5 METs without early exercise ST depression and show a normal rise in systolic blood pressure is important in constituting a negative predictive value (1086,1087). The optimum time for performing the exercise test after STEMI remains unresolved. It is argued that a predischarge exercise test provides psychological benefits to the patient and will permit detection of profound ischemia that could be associated with postdischarge cardiac events that might occur before a scheduled 3- to 6-week postdischarge, symptom-limited stress test. It also provides parameters for cardiac rehabilitation exercise programs. On the other hand, deferring exercise testing until approximately 3 weeks after STEMI in clinically low-risk patients appears safe and reasonable and enables more optimal assessment of functional capacity. It is the consensus of this Writing Committee that patients without complications who have not undergone coronary arteriography before discharge from the hospital and who might be potential candidates for revascularization procedures should undergo exercise electrocardiography before or just after discharge.

7.11.1.2. Role of Echocardiography

Noninvasive imaging in patients recovering from STEMI includes echocardiography and radionuclide imaging. This section discusses the role of echocardiography. (See Sections 7.11.1.3, 7.11.1.4, and 7.11.1.5 for additional discussion on imaging considerations.)

Class I

- 1. Echocardiography should be used in patients with STEMI not undergoing LV angiography to assess baseline LV function, especially if the patient is hemo-dynamically unstable. (Level of Evidence: C)
- 2. Echocardiography should be used to evaluate patients with inferior STEMI, clinical instability, and clinical suspicion of RV infarction. (See the ACC/AHA/ASE 2003 Guideline Update for Clinical Application of Echocardiography (226).) (Level of Evidence: C)
- 3. Echocardiography should be used in patients with STEMI to evaluate suspected complications, including acute MR, cardiogenic shock, infarct expansion, VSR, intracardiac thrombus, and pericardial effusion. (Level of Evidence: C)
- 4. Stress echocardiography (or myocardial perfusion imaging) should be used in patients with STEMI for in-hospital or early postdischarge assessment for inducible ischemia when baseline abnormalities are expected to compromise ECG interpretation. (Level of Evidence: C)

Class IIa

- 1. Echocardiography is reasonable in patients with STEMI to re-evaluate ventricular function during recovery when results are used to guide therapy. (Level of Evidence: C)
- 2. Dobutamine echocardiography (or myocardial perfusion imaging) is reasonable in hemodynamically and electrically stable patients 4 or more days after STEMI to assess myocardial viability when required to define the potential efficacy of revascularization. (Level of Evidence: C)
- 3. In STEMI patients who have not undergone contrast ventriculography, echocardiography is reasonable to assess ventricular function after revascularization. (Level of Evidence: C)

Class III

Echocardiography should not be used for early routine re-evaluation in patients with STEMI in the absence of any change in clinical status or revascularization procedure. Reassessment of LV function 30 to 90 days later may be reasonable. (Level of Evidence: C) The widespread availability, portability, and relatively low cost of echocardiography have resulted in its increased use as a practical and reliable means of assessing both global ventricular function and regional wall-motion abnormalities. Transthoracic imaging and Doppler techniques are generally sufficient for evaluating patients with suspected or documented ischemic heart disease. However, transesophageal echocardiography may be needed in some patients, particularly those with serious hemodynamic compromise but nondiagnostic transthoracic echocardiography studies. The uses of echocardiography in STEMI are discussed in detail in the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography (226).

Assessment of Prognosis and Complication. Echocardiography assesses global and regional ventricular function. The sum of segmental wall-motion abnormalities may overestimate infarct size because it includes regions of prior infarction and areas with ischemic stunning or hibernation of myocardium. However, LVEF and echocardiographically derived infarct size are predictive of early and late complications and mortality (1088-1091).

Echocardiography can be used to evaluate at the bedside, when needed, virtually any complication of STEMI. These complications include acute MR, cardiogenic shock, infarct expansion, VSR, intracardiac thrombus, and pericardial effusion.

Assessment of Therapy. Given the frequent use of reperfusion therapy (fibrinolytic agents or primary angioplasty) in patients with STEMI, assessment of myocardial salvage is an important clinical issue. After successful reperfusion, myocardial stunning may occur and may last for days to months. Wall-motion segments that demonstrate hypokinesia or akinesia at rest but improved function during low-dose dobutamine infusion often recover contractility (1092-1101), which suggests that these segments are stunned. Failure of such segments to show improvement suggests functional recovery is unlikely to occur (1092-1101). These issues are discussed in more detail in the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography (226).

Predischarge Evaluation With Stress Echocardiography. The role of exercise testing in evaluation of patients with STEMI before discharge is discussed in Section 7.11.1.1. As previously noted, the ability to perform an exercise test 1 month after STEMI provides a favorable prognosis regardless of test results (1075). The incremental value of exercise echocardiography over regular exercising testing after STEMI has not been established. Prospective natural history studies are difficult to accomplish because many clinicians now perform angiography and recommend revascularization in patients with significant coronary obstructions. Echocardiography or perfusion imaging (see Section 7.11.1.3) should be added to exercise testing whenever base-

line abnormalities are expected to compromise ECG interpretation.

Although serious complications have been reported (1093), general experience suggests that pharmacological stress echocardiography with a graded protocol and with low doses of dobutamine initially appears to be feasible and safe when performed 4 to 10 days after STEMI (1092,1094-1102). Pharmacological challenge can substitute for exercise in predischarge functional testing for ischemia in patients with limited exercise capacity and can help in assessing myocardial viability early after STEMI (1092,1094,1103-1112). Betablockade may limit the sensitivity of dobutamine echocardiography in assessing ischemia.

7.11.1.3. Exercise Myocardial Perfusion Imaging

Noninvasive imaging in patients recovering from STEMI includes echocardiography and radionuclide imaging. This section discusses the role of exercise myocardial perfusion imaging. (See Sections 7.11.1.2, 7.11.1.4, and 7.11.1.5 for additional discussion on imaging considerations.)

Class I

Dipyridamole or adenosine stress perfusion nuclear scintigraphy or dobutamine echocardiography before or early after discharge should be used in patients with STEMI who are not undergoing cardiac catheterization to look for inducible ischemia in patients judged to be unable to exercise. (Level of Evidence: B)

Class IIa

Myocardial perfusion imaging or dobutamine echocardiography is reasonable in hemodynamically and electrically stable patients 4 to 10 days after STEMI to assess myocardial viability when required to define the potential efficacy of revascularization. (Level of Evidence: C)

Before the use of reperfusion therapy, the prognostic value of exercise myocardial perfusion imaging was found to be superior to that of exercise ECG testing (1113-1116). Pharmacological stress perfusion imaging was shown to have predictive value for postinfarction cardiac events (1117-1119). The key issues are whether these results apply to current patient populations in the reperfusion era and whether myocardial perfusion imaging is worth the additional cost for risk stratification (1120). The same issues outlined previously with respect to exercise echocardiographic testing apply to this methodology. In patients with STEMI who have received fibrinolytic therapy, several studies using myocardial perfusion imaging have found that it is less valuable than previously thought for risk stratification (1121-1123), primarily because of the low rate of late cardiac events.

In patients in the current era who have not received reperfusion therapy, particularly those who have not undergone revascularization, the same considerations regarding subsequent patient outcome that were outlined above for exercise ECG testing apply. There is evidence that myocardial perfusion imaging is useful for risk stratification in such patients, despite their better overall prognosis (1124). It appears likely that the previously demonstrated superiority of stress myocardial perfusion imaging probably continues to apply to this population, although there is limited evidence on this point. Prospective studies are difficult to conduct because clinicians frequently intervene in patients with abnormal predischarge stress perfusion imaging studies.

Myocardial perfusion imaging with either Tl 201 (1125), Tc 99m sestamibi (1126), or Tc 99m tetrofosmin can assess infarct size. The measurement of infarct size by any one of these techniques is significantly associated with subsequent patient mortality after fibrinolytic therapy (1125-1127). Data are also emerging to suggest that vasodilator stress nuclear scintigraphy is safe and can be used for early (48 to 72 hours) risk stratification (1128). The ACC/AHA/ASNC Guidelines for the Clinical Use of Cardiac Radionuclide Imaging, a revision of the 1995 guidelines, stress the ability of pharmacological stress perfusion imaging to risk-stratify patients after STEMI early and safely, thereby facilitating the development of clinical strategies (239).

Recommended strategies for exercise test evaluations after STEMI are presented in Figure 36. These strategies and the data on which they are based are reviewed in more detail in the ACC/AHA 2002 Guideline Update for Exercise Testing (1075).

7.11.1.4. Left Ventricular Function

Noninvasive imaging in patients recovering from STEMI includes echocardiography and radionuclide imaging. This section discusses the importance of measurement of LV function. Either of the above imaging techniques can provide clinically useful information.

Class I

Left ventricular ejection fraction should be measured in all STEMI patients. (*Level of Evidence: B*)

Assessment of LV function after STEMI has been shown to be one of the most accurate predictors of future cardiac events in both the prereperfusion (1129) and the reperfusion (1130,1131) eras. Multiple techniques for assessing LV function of patients after STEMI have important prognostic value. Because of the dynamic nature of LV function recovery after STEMI, clinicians should consider the timing of the imaging study relative to the index event when assessing LV function. (See Table 6 of the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography for further discussion of the impact of timing on assessment of LV function and inducible ischemia) (226). The assessment can include such basic factors as clinical estimates based on patients' symptoms (e.g., exertional dyspnea, functional status), physical findings (e.g., rales, murmurs, elevated jugular venous pressure, cardiomegaly, S₃ gallop), and measurement of ejection fraction by contrast ventriculography, radionuclide ventriculography, and 2-dimensional echocardiography. Zaret and colleagues (1130) found that an

LVEF less than 0.30 as assessed by radionuclide ventriculography was still predictive of mortality in patients surviving infarction treated with fibrinolytic therapy, despite the significantly reduced mortality of these patients compared with those in the prereperfusion era. White and colleagues (1132) performed contrast left ventriculography in 605 patients 1 to 2 months after MI. They found that postinfarction LV dilation, demonstrated by increased end-systolic volume greater than 130 ml, was an even better predictor of mortality after MI than an LVEF less than 0.40 or increased end-diastolic volume. In patients with normal ejection fractions, however, end-systolic volume did not provide any further stratification according to risk.

Assessment of LV function by different techniques may produce different results. The SOLVD (Studies Of Left Ventricular Dysfunction) investigators compared LVEF determined by echocardiography or radionuclide angiography and noted higher mortality for patients with LVEF less than 0.35 by echocardiography than for patients with the same value determined by the radionuclide technique (1133). LVEFs derived by radionuclide angiography and by cardiac catheterization were compared in the SAVE study, with higher values derived by catheterization (924). Radionuclear and catheterization-derived LVEF were poorly correlated, but lower values by either technique predicted a worse prognosis. In some circumstances, assessment of ventricular function by several techniques may be useful.

7.11.1.5. Myocardial Viability

Noninvasive imaging in patients recovering from STEMI includes echocardiography and radionuclide imaging. This section discusses techniques for assessing myocardial viability. Either of the above imaging techniques can provide clinically useful information.

As previously noted, LV function is a well established and powerful predictor of outcome after STEMI. In some patients, LV dysfunction results from necrosis and scar formation. In others, viable but dysfunctional myocardium contributes to LV dysfunction and may be significantly reversible with revascularization. Myocardial hibernation (chronic low-flow state associated with depressed myocardial function) (1134) and stunning (depression of ventricular function after acute ischemia despite adequate restoration of blood flow) (1135) contribute to the potential reversibility of ventricular function. Up to one third of patients with significant ischemic LV dysfunction may improve with revascularization (1136). Therefore, the distinction between ventricular dysfunction caused by fibrosis and that arising from viable but dysfunctional myocardium may have important prognostic and therapeutic implications.

Several noninvasive imaging modalities have been established as predictors of myocardial viability. Radionuclide imaging and dobutamine echocardiography have acceptable accuracy in predicting recovery of regional wall-motion abnormalities (239). More recently, relatively small studies have suggested the ability of viability studies to identify patients most likely to benefit from revascularization in terms of symptoms and natural history (1137-1140). These data are reviewed in more detail in the ACC/AHA/ASNC Guidelines for Cardiac Radionuclide Imaging (239) and the ACC/AHA/ASE 2003 Guideline Update for the Clinical Application of Echocardiography (226). Positron emission tomography, although not currently as widely available as other radionuclide techniques, may provide slightly better overall accuracy in predicting recovery of regional function (1141). More recently, contrast-enhanced magnetic resonance imaging has demonstrated promising results in predicting improvement of regional myocardial function in patients after STEMI (1142-1144).

The data noted above suggest that assessment of myocardial viability after STEMI, particularly in patients with severe LV dysfunction, may identify those with the highest risk, in whom revascularization can be of clinical benefit. However, myocardial viability remains incompletely understood. Viability testing is not a standard until more conclusive diagnostic efficacy studies are performed to demonstrate patient benefit.

7.11.1.6. Invasive Evaluation

Class I

- 1. Coronary arteriography should be performed in patients with spontaneous episodes of myocardial ischemia or episodes of myocardial ischemia provoked by minimal exertion during recovery from STEMI. (Level of Evidence: A)
- 2. Coronary arteriography should be performed for intermediate- or high-risk findings on noninvasive testing after STEMI (see Table 23 of the ACC/AHA 2002 Guideline Update for the Management of Patients With Chronic Stable Angina) (71). (Level of Evidence: B)
- **3.** Coronary arteriography should be performed if the patient is sufficiently stable before definitive therapy of a mechanical complication of STEMI, such as acute MR, VSR, pseudoaneurysm, or LV aneurysm. (Level of Evidence: B)
- 4. Coronary arteriography should be performed in patients with persistent hemodynamic instability. (Level of Evidence: B)
- 5. Coronary arteriography should be performed in survivors of STEMI who had clinical heart failure during the acute episode but subsequently demonstrated well-preserved LV function. (*Level of Evidence: C*)

Class IIa

- 1. It is reasonable to perform coronary arteriography when STEMI is suspected to have occurred by a mechanism other than thrombotic occlusion of an atherosclerotic plaque. This would include coronary embolism, certain metabolic or hematological diseases, or coronary artery spasm. (Level of Evidence: C)
- 2. Coronary arteriography is reasonable in STEMI patients with any of the following: diabetes mellitus, LVEF less than 0.40, CHF, prior revascularization, or

life-threatening ventricular arrhythmias. (Level of Evidence: C)

Class IIb

Catheterization and revascularization may be considered as part of a strategy of routine coronary arteriography for risk assessment after fibrinolytic therapy (see Section 6.3.1.6.4.7). (*Level of Evidence: B*)

Class III

Coronary arteriography should not be performed in survivors of STEMI who are thought not to be candidates for coronary revascularization. (Level of Evidence: A)

In contrast to noninvasive testing, coronary arteriography provides detailed structural information to allow an assessment of prognosis and to provide direction for appropriate management. Indications for coronary arteriography are interwoven with indications for possible therapeutic plans such as PCI or CABG. All survivors of STEMI who are candidates for revascularization therapy with spontaneous ischemia, intermediate- or high-risk findings on noninvasive testing, hemodynamic or electrical instability, mechanical defects, prior revascularization, or high-risk clinical features should be considered for coronary arteriography. Percutaneous coronary intervention or CABG may be considered in these patients if they are found to have significant obstructive coronary artery disease (1145-1147).

Given the adverse clinical consequences of recurrent infarction, it would be highly desirable to minimize the chance of its occurrence after initial treatment for STEMI. Routine referral for angiography of all patients after fibrinolytic therapy is intuitively attractive and supported indirectly by retrospective analyses from trials of fibrinolytic therapy that suggest that patients treated with PCI during the index hospitalization had a lower risk of recurrent MI and a lower 2-year mortality (512). However, previous randomized trials testing a strategy of routine catheterization after fibrinolysis suggested that such an approach was deleterious (480,503,505,516,1148). However, the previous trials were conducted in an era during which aspirin was inconsistently administered, high doses of UFH without monitoring of activated clotting time were used, and the interventional catheters, radiographic imaging equipment, and supportive antiplatelet agents were suboptimal. The Writing Committee encourages contemporary research into the benefit of routine catheterization versus watchful waiting after fibrinolytic therapy in the contemporary era (1149,1149a). See Section 6.3.1.6.4.7.

7.11.1.7. Ambulatory ECG Monitoring for Ischemia

The value of ambulatory ECG monitoring in assessing reversible myocardial ischemia and the risk of a subsequent coronary event early after MI has been evaluated in a number of studies (1150-1157). Up to 25% of patients will show residual ischemia detected on ambulatory ECG monitoring.

Most episodes of transient myocardial ischemia are silent and occur at rest or during times of low-level physical activity or mental stress (1158). During long-term follow-up studies, a number of investigators have reported that the presence of ischemia detected by ambulatory ECG monitoring in the postinfarction period is predictive of a subsequent poor outcome and increases the risk of cardiac events (1150-1157). One study found that the odds ratio for patients with ambulatory ischemia, compared with those without it, was 2.3 for death or nonfatal MI at 1 year (1157).

Despite the promising initial results with ambulatory ECG monitoring, the totality of evidence does not support a general statement about its role in patients with STEMI. Some studies have shown that the results of ambulatory ECG monitoring could be predicted from exercise test data (1152,1155), whereas others have found that additional prognostic information could be obtained by ambulatory ECG monitoring in postinfarction patients (1153). At present, a cost-effective strategy has not been developed to identify patients who are at increased risk for ambulatory ischemia and in whom ambulatory ECG monitoring might be more helpful for stratification into high- and low-risk subgroups for future coronary events.

7.11.1.8. Assessment of Ventricular Arrhythmias

Class IIb

Noninvasive assessment of the risk of ventricular arrhythmias may be considered (including signalaveraged ECG, 24-hour ambulatory monitoring, heart rate variability, micro T-wave alternans, and Twave variability) in patients recovering from STEMI. (Level of Evidence: B)

A number of noninvasive strategies have been used to try to identify patients at high risk for arrhythmic events. Although most of these measure some aspect of the excessive activation of the autonomic nervous system commonly observed in high-risk patients with LV dysfunction, others measure impulse conduction through the infarct zone. Thus, noninvasive determinants of arrhythmia risk encompass measurement of changes in ventricular repolarization, alterations of autonomic tone, and delayed and disordered myocardial conduction.

The most frequently used techniques are signal-averaged or high-resolution ECG, heart rate variability, baroreflex sensitivity, and T-wave alternans. Signal-averaged electrocardiography identifies delayed, fragmented conduction in the infarct zone in the form of late potentials at the terminus of the QRS complex and represents an anatomic substrate that predisposes the patient to re-entrant VT. Kuchar et al. (1159) reported that late potentials predict an increased incidence of sudden death in the post-MI patient population. Gomes et al (1160) found late potentials to be the best single predictor when considering Holter monitoring and ejection fraction and found that they contributed independently to a combined index, although the positive predictive value of each was poor. The filtered QRS duration was the most predictive feature of signal-averaged electrocardiography in a Cardiac Arrhythmia Suppression Trial (CAST) substudy (1161). More recent studies have shown that reperfusion therapy reduces the incidence of late potentials after STEMI (1162). In the setting of frequent use of fibrinolysis, the predictive value of signal-averaged electrocardiography has been variable (1163-1165).

Heart rate variability, an analysis of the beat-to-beat variation in cycle length, largely reflects the sympathovagal interaction that regulates heart rate. Heart rate variability can be quantified in a number of ways, with either time- or frequency-domain parameters (1166). Low heart rate variability, indicative of decreased vagal tone, is a predictor of increased mortality, including sudden death, in patients after MI (1166,1167) and may add significant prognostic information to other parameters (1167). The Autonomic Tone and Reflexes After Myocardial Infarction (ATRAMI) study (1168) was a prospective multicenter study that enrolled 1284 patients with a recent (fewer than 28 days) MI. Decreased heart rate variability was a significant and independent predictor of cardiac mortality (hazard ratio 3.2, p equals 0.005) (1168).

The predictive value of heart rate variability after STEMI, although significant, is modest when used alone. In combination with other techniques, its positive predictive accuracy improves. The most practical, feasible, and cost-efficient combination of noninvasive predictive tests with heart rate variability remains to be determined.

Baroreceptor sensitivity also quantifies the influence of parasympathetic tone on the heart. It is measured as the slope of a regression line relating beat-to-beat heart rate change in response to a change in blood pressure, often accomplished by giving a small bolus of phenylephrine (1169). Reductions in baroreflex sensitivity have been associated with an increased susceptibility to arrhythmic events and sudden death in experimental models and initial clinical reports (1170-1172).

Finally, abnormalities in ventricular repolarization are detectable by microvolt alterations of T-wave amplitude, or T-wave alternans. In experimental preparations, the presence of T-wave alternans was predictive of a lower fibrillation threshold. Clinical studies have demonstrated that T-wave alternans is associated with ventricular arrhythmias during EP testing, as well as with clinically evident arrhythmias. Microvolt T-wave alternans was subsequently shown to be associated with inducible ventricular arrhythmias during programmed ventricular stimulation and spontaneous arrhythmic events (1173,1174).

The clinical applicability of these tests to the post-STEMI patient is in a state of evolution. Although the negative predictive value of most of these tests taken in isolation is high (generally greater than 90%), the positive predictive value is unacceptably low (less than 30%). Whereas the positive predictive value of noninvasive testing for future arrhythmic events can be modestly increased by combining several test results, the therapeutic implications of positive findings are unclear. Insufficient data are available to indicate whether general therapies, such as beta-adrenoceptor blockade, ACE inhibition, and revascularization procedures, or specific interventions, such as treatment with amiodarone or an ICD, targeted toward high-risk patients identified by a combination of noninvasive tests after MI can more favorably impact mortality (1175). The widening indications for ICD implantation in patients with LV dysfunction may encourage the use of these studies to find the lowest-risk patients (i.e., those without any high-risk markers whatsoever) to avoid ICD implantation. Finally, it is difficult to justify the costs of the routine use of these procedures in the absence of demonstrated clinical benefits. Until these issues are resolved, these tests are used only to support routine management and risk assessment.

7.12. Secondary Prevention

Class I

Patients who survive the acute phase of STEMI should have plans initiated for secondary prevention therapies. (Level of Evidence: A)

Secondary prevention therapies, unless contraindicated, are an essential part of the management of all patients with STEMI (Table 32) (68), regardless of sex (69,1176). Inasmuch as atherosclerotic vascular disease is frequently found in multiple vascular beds, the physician should search for symptoms or signs of peripheral vascular disease or cerebrovascular disease in patients presenting with STEMI.

Approximately 70% of CHD deaths and 50% of MIs occur in patients who have previously established coronary artery disease (677). It is estimated that the likelihood of fatal and nonfatal MI is 4 to 7 times higher in patients with apparent coronary disease. The institution of secondary prevention therapies and risk reduction strategies in patients recovering from STEMI represents major opportunities to reduce the toll of cardiovascular disease.

Smoking cessation, aggressive lipid lowering, control of hypertension and diabetes, and prophylactic use of aspirin, beta-blockers, and ACE inhibitors are key components of secondary prevention that have a demonstrated benefit. Dietary cholesterol contributes to elevations in LDL-C and should be included in the list of dietary modifications (reduction in total calories, saturated fat, cholesterol, and salt and increase in vegetables, fruits, and grain fiber) given to the post-STEMI patient. These changes are both directly beneficial and aid in hypertension control and lipid lowering. The specifics of these therapies and recommendations for their use are discussed in the following sections (68).

Secondary prevention services remain discouragingly underutilized (1177) despite the compelling evidence for and large magnitude of their benefits (1178-1180). The ACC and AHA urge all healthcare providers to implement systems that ensure the reliable identification of patients who can benefit from secondary prevention interventions and to ensure that these interventions occur (1181,69). Evidence exists that lifestyle modifications (as discussed below), including regular physical exercise, weight reduction, cessation of cigarette smoking, and stress reduction strategies, have a favorable impact on blood coagulation, fibrinolysis, and platelet reactivity, shifting the patient to a less atherogenic and prothrombotic state (Table 32) (68,1182). Referral to outpatient cardiac rehabilitation can aid in attainment of these goals (1183,1184).

Many institutions have clinical pathways for improving quality of care of patients with STEMI (205). Use of clinical pathways or other management protocols in hospital settings has resulted in improved adherence to therapy by CHD patients and better cholesterol control. The Cardiac Hospitalization Atherosclerosis Management Program (CHAMP) focused on the initiation of therapy with aspirin, beta-blockers, ACE inhibitors, statins, diet, and exercise in persons with established CHD before hospital discharge (1185). There was a significant increase in the use of aspirin, beta-blockers, ACE inhibitors, and statin therapy. The program used postdischarge follow-up visits to titrate the statin dose to achieve an LDL-C level less than 100 mg/dL. One year after discharge, 91% of patients were being treated with cholesterol-lowering therapy, and 58% were at treatment goals. These results suggest that the initiation of treatment during hospitalization for CHD adds needed emphasis to the importance of cholesterol-lowering treatment along with other cardiac medications. There was also a reduction in recurrent MI and 1-year mortality. Evidence exists that adherence to practice guidelines for the management of patients after STEMI is associated with significant reductions in short- and long-term mortality (1186).

7.12.1. Patient Education Before Discharge

Class I

- **1.** Before hospital discharge, all STEMI patients should be educated about and actively involved in planning for adherence to the lifestyle changes and drug therapies that are important for the secondary prevention of cardiovascular disease. (Level of Evidence: B)
- 2. Post-STEMI patients and their family members should receive discharge instructions about recognizing acute cardiac symptoms and appropriate actions to take in response (i.e., calling 9-1-1 if symptoms are unimproved or worsening 5 minutes after onset, or if symptoms are unimproved or worsening 5 minutes after 1 nitroglycerin dose sublingually) to ensure early evaluation and treatment should symptoms recur. (Level of Evidence: C)
- **3.** Family members of STEMI patients should be advised to learn about AEDs and CPR and be referred to a CPR training program. Ideally, such training programs would have a social support component targeting family members of high-risk patients. (Level of Evidence: C)

Once STEMI has occurred, secondary prevention, including aggressive risk factor management through medical and lifestyle therapy, is the mainstay of therapy. It is important for providers in the inpatient setting to communicate with the patient's primary care provider to facilitate coordination of care. The discharge summary should include the patient's medical, psychological, and functional status; prescribed medication and lifestyle regimens; and the patient's living/social support situation (1187).

Effective education is critical to enlist patients' full participation in therapeutic regimens and secondary prevention efforts and to minimize their natural concerns and anxieties. It ideally leads to a patient who is better informed and more satisfied with his or her care and who is also able to achieve a better quality of life and improved survival (71,688,1188-1190). In general, the outcomes of educational interventions depend on the outcome studied (e.g., largest for knowledge change, smallest for behavioral and psychological outcomes) (692) and the strategies applied (688).

Principles of Patient Education. A thorough discussion of the philosophies of and approaches to patient education is beyond the scope of this guideline. There are useful reviews on this topic (688,692,1191), including a few that focus on ischemic heart disease (1192-1194). Well-designed educational programs can improve patients' knowledge and in some instances can improve outcomes (688,1195). These approaches form the basis for commonly used educational programs, such as those conducted before CABG (1196) and after MI (690,1197). A variety of guidelines should be followed to help ensure that educational efforts are successful (71):

- 1. Assess the patient's baseline understanding. This not only establishes a starting point for education but also engages the patient. Healthcare providers are often surprised at the idiosyncratic notions that patients have about their own medical conditions and therapeutic approaches (688,1198,1199).
- 2. Elicit the patient's desire for information. Adults prefer to set their own agendas, and they learn better when they can control the flow of information.
- 3. Use epidemiological and clinical evidence. As clinical decision making becomes increasingly based on scientific evidence, it is reasonable to share that evidence with patients. Epidemiological data can assist in formulating an approach to patient education, such as conveying information on the major cardiovascular disease risk factors. Scientific evidence about the reduction in cardiovascular disease risk to be derived from risk factor modification/lowering can help persuade patients about the effectiveness of various interventions.
- 4. Use allied professional personnel. Even though physicians may feel constrained with the limited amount of time available for a patient encounter, interpersonal contact with and encouragement from physicians is never-theless a vital component of the patient's overall health education. Medical encounters represent powerful "teaching moments" for initiating discussion and shaping perceptions and expectations, and they are perceived as highly credible sources of information (1200,1201).

In many settings, trained health educators, many of whom specialize in diabetes or cardiac disease, are available. Personnel from disciplines such as nursing, physical therapy, nutrition, pharmacy, and others can provide reinforcement, more in-depth information, and ongoing educational support and thus have much to offer patients with ischemic heart disease (1202). Reimbursement for educational activities by these allied providers is poor, although this can also be accomplished as part of a cardiac rehabilitation program for which funding is generally available.

- Use professionally prepared resources when available. 5. Studies have shown that combined teaching strategies, including written and audiovisual materials in addition to verbal instruction, group approaches, and enhanced education methods, are the most effective in achieving desired outcomes (688,692,1192). The decreasing length of hospital stays has raised concern about adequate opportunity for appropriate patient education (369,697), although short educational sequences can produce outcomes comparable to lengthy sessions (690). Innovative presentation styles (e.g., programmed instruction, audiovisual techniques, and health education television programs) can produce benefits comparable to individual educational sessions (1203). Use of a single repository for all educational materials (e.g., a binder that travels with the patient) may provide consistency, document material taught, identify goals that remain, and promote independent study, an effective education strategy (688,1191).
- Although any planned teaching strategy is better than 6. none (688), computer-assisted instruction is the newest approach to patient education. Patients without computer access can be directed to a work station in the physician's office, clinic, hospital education department/ library, or local library, where relevant pages can be printed. A review of the literature on computer-based approaches to patient education supports the use of computer-based education as an effective strategy for transfer of knowledge and skill development for patients (1204). The World Wide Web is convenient for medical personnel and patients with access to personal computers. The AHA maintains a Web site (http://www.americanheart.org) that presents detailed and practical dietary recommendations and information about heart disease, physical activity, heart attacks, and CPR. The NHLBI Web site (http://www.nhlbi.nih.gov) has links to the National High Blood Pressure Education Program, NCEP, and the NHAAP's "Act In Time to Heart Attack Signs," which have information for patients and professionals. Both the ACC's "Guidelines Applied in Practice" program (http://www.acc.org/gap/images/ Discharge.gif) and the AHA's "Get With the Guidelines Hospital Tool Kit" (http://www.americanheart.org/ downloadable/heart/1107_HospTool.pdf) provide discharge information as part of a separate patient-specific discharge form (Guidelines Applied in Practice) or

incorporated into a patient pathway (Get With the Guidelines). The discharge information for patients reviews and documents the important lifestyle changes and drug therapies for the secondary prevention of cardiovascular disease that are part of their discharge instructions. An important issue currently receiving attention is the issue of health literacy. Studies with patients and the public have found average reading abilities do not exceed the eighth-grade level (688,695). The current recommendation is that materials be written at the sixth-grade level to ensure that the maximum number of people can read and understand them.

- 7. Develop an action plan with the patient for their longterm management. If it is necessary to convey a great deal of information to patients about their condition, be cognizant of the patient's level of sophistication, readiness to change, prior educational attainment, language barriers, relevant clinical factors, and social support. A realistic goal might be to motivate an internal change that results in the patient's desire to ask for information on an important topic such as smoking cessation (691). It may be counterproductive to attempt to coax a patient into simultaneously changing several behaviors, such as smoking, diet, and exercise, and taking (and purchasing) multiple new medications. Achieving optimal adherence often requires talking to patients to identify barriers to compliance such as complex regimens and cost of medications (1205) and addressing these barriers through problem solving with the patient.
- 8. Involve family members in educational efforts. It is advisable and often essential to include family members in educational efforts. Many activities require the participation of family members; an example is meal preparation. Efforts to encourage smoking cessation, weight loss, or increased physical activity may be enhanced by enlisting the support of family members who can reinforce the behavior and may themselves benefit from participation.
- 9. Remind, repeat, and reinforce. Almost all learning deteriorates without reinforcement. At regular intervals, the patients' understanding should be reassessed, and key information should be repeated as warranted. Feedback is a powerful motivator, and patients should be congratulated for progress even when their ultimate goals are not fully achieved. Even though the patient who has reduced his or her use of cigarettes from 2 packs to 1 pack per day has not quit smoking, that 50% reduction in exposure is important and may simply represent a milestone on the path to complete cessation (688).

The secondary prevention targets for post-STEMI patients, around which education and follow-up/management should occur, are discussed in the following sections.

Within 6 years after a recognized heart attack, 18% of men and 35% of women will have another heart attack (46); most episodes of cardiac arrest occur within 18 months after hospital discharge for STEMI (1206). Thus, it is critical that patients and family members are educated in advance about recognition of acute ischemic symptoms and the appropriate action steps to take to ensure early evaluation and treatment (see Section 3.3). Furthermore, patients who have had a STEMI have a sudden death rate that is 4 to 6 times that of the general population (46). Thus, family members of patients with STEMI should be advised to learn CPR and should be given community resources to obtain this training. In addition, research has shown that patients whose family members received a social support intervention in conjunction with being taught CPR reported better psychosocial adjustment and less anxiety and hostility than those whose families received CPR training only or CPR training with risk factor education. Family members should be referred to a CPR program that combines CPR training with social support (132,133). (See Section 4.2.)

7.12.2. Lipid Management

Class I

- 1. Dietary therapy that is low in saturated fat and cholesterol (less than 7% of total calories as saturated fat and less than 200 mg/d cholesterol) should be started on discharge after recovery from STEMI. Increased consumption of the following should be encouraged: omega-3 fatty acids, fruits, vegetables, soluble (viscous) fiber, and whole grains. Calorie intake should be balanced with energy output to achieve and maintain a healthy weight. (Level of Evidence: A)
- 2. A lipid profile should be performed, or obtained from recent past records, for all STEMI patients, preferably after they have fasted and within 24 hours of symptom onset. (Level of Evidence: C)
- 3. The target LDL-C level after STEMI should be substantially less than 100 mg/dL. (*Level of Evidence: A*)
 - a. Patients with LDL-C 100 mg/dL or above should be prescribed drug therapy on hospital discharge, with preference given to statins. (*Level of Evidence: A*)
 - b. Patients with LDL-C less than 100 mg/dL or unknown LDL-C levels should be prescribed statin therapy on hospital discharge. (Level of Evidence: B)
- 4. Patients with non-high-density lipoprotein cholesterol (HDL-C) levels less than 130 mg/dL who have an HDL cholesterol level less than 40 mg/dL should receive special emphasis on nonpharmacological therapy (e.g., exercise, weight loss, and smoking cessation) to increase HDL. (Level of Evidence: B)

Class IIa

- 1. It is reasonable to prescribe drug therapy at hospital discharge to patients with non-HDL-C greater than or equal to 130 mg/dL, with a goal of reducing non-HDL-C to substantially less than 130 mg/dL. (Level of Evidence: B)
- 2. It is reasonable to prescribe drugs such as niacin or fibrate therapy to raise HDL-C levels in patients with LDL-C less than 100 mg/dL and non-HDL-C less than 130 mg/dL but HDL-C less than 40 mg/dL

despite dietary and other nonpharmacological therapy. Dietary-supplement niacin must not be used as a substitute for prescription niacin, and over-the-counter niacin should be used only if approved and monitored by a physician. (Level of Evidence: B)

3. It is reasonable to add drug therapy with either niacin or a fibrate to diet regardless of LDL and HDL levels when triglyceride levels are greater than 500 mg/dL. In this setting, non-HDL-C (goal substantially less than 130 mg/dL) should be the cholesterol target rather than LDL-C. Dietary-supplement niacin must not be used as a substitute for prescription niacin, and over-the-counter niacin should be used only if approved and monitored by a physician. (Level of Evidence: B)

Early secondary prevention trials conducted before the use of statin therapy, using the then-available drugs and diet to lower cholesterol, demonstrated significant reductions of 25% in nonfatal MIs and 14% in fatal MIs (59). Subsequently, a growing body of evidence derived mainly from large randomized clinical trials of statin therapy has firmly established the desirability of lowering atherogenic serum lipids in patients who have recovered from a STEMI.

The Scandinavian Simvastatin Survival Study (1207) reported results in 4444 men and women with CHD and moderate hypercholesterolemia observed over 5.4 years. Coronary heart disease mortality was reduced by 42% and total mortality by 30% among those receiving simvastatin compared with placebo. The relative risk reduction seen in this trial was similar among those with the lowest quartile compared with the highest quartile of baseline serum LDL-C. The CARE trial was a similar study in a population of patients who had recovered from an earlier MI and whose total cholesterol (mean 209 mg/dL) and LDL-C (mean 139 mg/dL) were essentially the same as the average for the general US population. In this trial, 4159 patients were randomly assigned to either 40 mg of pravastatin a day or placebo. After a median follow-up of 5 years, there was a significant reduction in the primary end point of fatal CHD and nonfatal confirmed MIs in the pravastatin cohort (3% ARD; 24% RRR; p equals 0.003) (1208).

The results of the large Long-Term Intervention With Pravastatin in Ischemic Disease (LIPID) Study have been reported for more than 9000 patients randomly assigned to either placebo or 40 mg of pravastatin daily (1209). The trial was conducted in patients with a prior history of MI or unstable angina. Patient age at time of entry ranged from 21 to 75 years. The LIPID trial was stopped prematurely because of the efficacy of pravastatin in reducing major cardiovascular events, including significant decreases in CHD deaths (1.9% ARD; 24% RRR), total mortality rate (3.1% ARD; 22% RRR), and stroke (0.8% ARD; 19% RRR). Benefit has also been seen in patients with symptomatic coronary disease who were treated with fluvastatin. In the Lescol in Severe Atherosclerosis (LiSA) Study, patients with symptomatic CHD and hypercholesterolemia who were given fluvastatin had 71% fewer cardiac events than those in the placebo group (1210).

The effect of cholesterol lowering combined with lowintensity oral anticoagulation on late saphenous vein graft status has also been investigated (1211). In an angiographic trial attempting to reduce atherosclerosis in saphenous vein grafts after CABG surgery, aggressive lowering of LDL to less than 100 mg/dL with lovastatin 40 to 80 mg daily (and 8 g of cholestyramine daily, if needed), in addition to a Step I AHA diet, achieved a significant reduction (2.7% ARD; 29% RRR) in obstructive changes in the vein grafts at 4 to 5 years compared with moderate lipid lowering to an LDL-C of 132 to 136 mg/dL (1211). There was no benefit of low-dose warfarin therapy at 5 years. The results of 7.5-year follow-up revealed significant reductions of 30% in revascularization and 24% for the composite end point (death, nonfatal MI, stroke, CABG, or PCI) in the group treated with aggressive lipid-lowering therapy. The investigators concluded that the study provided support for the NCEP recommendation that LDL-C levels should be reduced to less than 100 mg/dL in patients who have coronary artery disease (1212).

The Heart Protection Study (1213) trial randomized more than 20000 men and women aged 40 to 80 years with coronary disease, other vascular disease, diabetes, and/or hypertension to simvastatin 40 mg or placebo for a mean followup of 5 years. The primary end point, total mortality, was significantly reduced with statin treatment (1.8% ARD; 12%) RRR, p equals 0.0003). The advantages of simvastatin therapy were demonstrated in all important prespecified subgroups, including women, patients more than 75 years old, diabetics, and individuals with baseline LDL-C of less than 100 mg/dL. The study supports the benefit of statin therapy for all groups and argues for the use of statins among patients with CHD and LDL-C values 100 to 129 mg/dL who had previously been considered candidates for diet and lifestyle changes to lower LDL-C before implementation of statin therapy (59).

Approximately 25% of patients who have recovered from STEMI demonstrate desirable total cholesterol values but a low HDL-C fraction on a lipid profile. Low HDL-C is an independent risk factor for development of coronary artery disease (1214), and therefore a rationale exists for attempting to raise HDL-C when it is found to be low in the patient with coronary artery disease. The Veterans Affairs High-Density Lipoprotein Cholesterol Intervention Trial (VA-HIT) (1215) revealed that modification of other lipid risk factors can reduce risk for CHD when LDL-C is in the range of 100 to 129 mg/dL. In this trial, male patients with a relatively low LDL-C (mean 112 mg/dL) were treated with gemfibrozil for 5 years. Gemfibrozil therapy, which raised HDL-C and lowered triglycerides, reduced the primary end point of fatal and nonfatal MI (4.4% ARD; 22% RRR) without significantly lowering LDL-C levels. There was no evidence of an increased risk of non-CHD mortality. This trial supports the concept that when LDL-C is in the range of 100 to 129 mg/dL, the use of other lipid-modifying drugs (e.g., fibrates)

is a therapeutic option if the patient has a low HDL-C (less than 40 mg/dL).

The effect of hypertriglyceridemia is somewhat controversial because in many cases, the triglyceride level varies inversely with HDL-C levels. However, if hypertriglyceridemia (triglyceride greater than 200 mg/dL) exists after lifestyle changes such as diet and exercise and statin therapy have been initiated, it is recommended that additional therapy be initiated to lower triglycerides for patients with established CHD (59). In this setting, the target goal of therapy should be non–HDL-C less than 130 mg/dL.

Diet and drug treatments available for the correction of lipid abnormalities are as effective in the elderly as in the young. Clinical trials have shown that such treatment can reduce total mortality up to age 70 years (1207) and the rate of recurrent coronary events up to age 75 years (1208). In addition, the PROSPER trial (1208) studied the value of lipid control for the prevention of initial coronary events in 5804 older persons (2804 men and 3000 women aged 70 to 82 years). Treatment with pravastatin 40 mg daily resulted in a reduction in LDL-C by 34% and significantly reduced the primary end point (CHD death, nonfatal MI, and fatal or nonfatal stroke) over 3.2 years (2.1% ARD; 15% RRR). Unlike previous trials involving elderly patients, the risk of stroke in PROSPER was unaffected, possibly because of the relatively short duration of the trial. Among elderly patients, other risk factors such as high blood pressure, cigarette smoking, and diabetes are frequently present. Thus, it is especially important in this group, as in younger patients, to develop comprehensive treatment programs, such as cardiac rehabilitation, to address all risk factors that might contribute to future CHD events in addition to treating their dyslipidemia.

The early secondary prevention trials of statins specifically excluded patients with STEMI in the acute phase. Thus, although the data supporting the efficacy and benefit of statin therapy for patients after STEMI are robust, statin therapy in the early studies was started 4 to 6 months after STEMI. Subsequent studies have addressed the potential benefits of early initiation of statin therapy during hospitalization for acute coronary syndrome. The results from several of these studies, wherein statins were initiated during the hospital phase of acute coronary syndromes, indicate improved cardiovascular outcomes. The Lipid-Coronary Artery Disease trial (1216) randomized 126 patients with acute coronary syndrome to early treatment with pravastatin with or without cholestyramine or niacin therapy versus usual care. Treatment initiated during hospitalization significantly reduced clinical events at 2 years. The Myocardial Ischemia Reduction with Aggressive Cholesterol Lowering (MIRACL) trial randomized 3086 patients admitted for acute coronary syndrome to treatment with atorvastatin 80 mg per day or placebo within 96 hours after admission (1217). The primary end point of death, nonfatal MI, resuscitated cardiac arrest, or recurrent severe myocardial ischemia was reduced from 17.4% to 14.8% (p equals 0.048) by treatment with atorvastatin. The major clinical benefits were fewer strokes and a lower risk of severe recurrent ischemia among those treated with atorvastatin. In the Swedish Registry of Cardiac Intensive Care of nearly 20000 patients, a 25% reduction in mortality at 1 year was observed among patients when statin therapy was initiated before hospital discharge (1218). Early initiation of therapy has been recommended by the NCEP (59). The Pravastatin or Atorvastatin Evaluation and Infection Therapy-Thrombolysis in Myocardial Infarction 22 study (PROVE-IT-TIMI 22) (1219) compared intensive lipid-lowering therapy to moderate lipid-lowering therapy initiated within 10 days after hospital admission in 4162 patients with acute coronary syndrome. Lipid lowering to a goal of LDL-C less than 100 mg/dL (median 95 mg/dL) with pravastatin 40 mg per day was compared with lipid-lowering to a goal of LDL-C less than 70 mg/dL (median 62 mg/dL) with atorvastatin 80 mg daily. After 2 years of therapy, the primary end point (death due to any cause, MI, unstable angina requiring rehospitalization, coronary revascularization, and stroke) was 26.3% for the group undergoing moderate lipid-lowering therapy and 22.4% in the intensive lipid-lowering group, reflecting a 16% reduction in the hazard ratio favoring intensive lipidlowering therapy (p equals 0.005). Benefits were observed within 30 days of initiation of therapy and occurred at all levels of LDL-C, even among those patients with LDL-C less than 100 mg/dL, although the greatest benefit was seen among patients with LDL-C of 125 mg/dL or greater. The benefits were observed in all important subgroups, including men and women, young and old, and patients with and without diabetes. These data support the early, intensive treatment of patients with acute coronary syndromes to LDL-C goals substantially less than 100 mg/dL with statin therapy. Patients hospitalized with CHD who start lipid-lowering therapy before discharge have been shown to be nearly 3 times as likely to be taking medication at 6 months as those starting therapy after discharge (1220).

7.12.3. Weight Management

Class I

- 1. Measurement of waist circumference and calculation of body mass index are recommended. Desirable body mass index range is 18.5 to 24.9 kg/m². A waist circumference greater than 40 inches in men and 35 inches in women would result in evaluation for metabolic syndrome and implementation of weight-reduction strategies. (Level of Evidence: B)
- 2. Patients should be advised about appropriate strategies for weight management and physical activity (usually accomplished in conjunction with cardiac rehabilitation). (Level of Evidence: B)
- **3.** A plan should be established to monitor the response of body mass index and waist circumference to therapy (usually accomplished in conjunction with cardiac rehabilitation). (Level of Evidence: B)

Obesity is a recognized major risk factor for cardiovascular disease and an important component of the metabolic syndrome (59). The clinical criteria for the metabolic syndrome include any 3 of the following: waist circumference greater than 40 inches in men and greater than 35 inches in women; triglycerides of at least 150 mg/dL; HDL-C less than 40 mg/dL in men or less than 50 mg/dL in women; blood pressure greater than 130 mm Hg systolic and greater than 85 mm Hg diastolic; and fasting blood glucose of at least 110 mg/dL. The metabolic syndrome contributes significantly to CHD risk. Its treatment has weight control and physical activity as primary strategies. Because approximately 65% of the US adult population is overweight or obese (1221), it is suggested that this risk factor be carefully addressed as part of the secondary prevention strategy for all patients after STEMI. The suggested goal should be the achievement or maintenance of a healthy weight, defined in the "Dietary Guidelines for Americans" (1222) as a body mass index from 18.5 to 24.9 kg/m². It is suggested that overweight patients be instructed in a weight loss regimen as part of their cardiac rehabilitation program after STEMI, with emphasis on the importance of regular exercise and a lifelong prudent diet to maintain healthy weight. Weight reduction enhances lowering of other risk factors for cardiovascular disease, including LDL-C, triglycerides, impaired glucose, and blood pressure. An initial weight loss of 10% of body weight achieved over 6 months is a recommended target. The rate of weight loss should be 1 to 2 pounds each week.

7.12.4. Smoking Cessation

Class I

- 1. Patients recovering from STEMI who have a history of cigarette smoking should be strongly encouraged to stop smoking and to avoid secondhand smoke. Counseling should be provided to the patient and family, along with pharmacological therapy (including nicotine replacement and bupropion) and formal smoking cessation programs as appropriate. (Level of Evidence: B)
- 2. All STEMI patients should be assessed for a history of cigarette smoking. (Level of Evidence: A)

Smoking cessation is essential for patients with STEMI. Smoking triggers coronary spasm, reduces the anti-ischemic effects of beta-adrenoceptor blockers, and doubles mortality after STEMI (1223-1225). Smoking cessation reduces rates of reinfarction and death within 1 year of quitting, but one third to one half of patients with STEMI relapse within 6 to 12 months (1226). Because exposure to secondary cigarette smoke has been identified as a risk factor, family members who live in the same household should also be encouraged to quit smoking to help reinforce the patient's effort and to decrease the risk of secondhand smoke (1227).

The most effective strategies for encouraging quitting are those that identify the patient's level or stage of readiness and provide information, support, and, if necessary, pharmacotherapy targeted at the individual's readiness and specific needs (66,1228). Houston-Miller and Taylor (1229) advocate a stepped approach to smoking cessation:
- a. Provide a firm, unequivocal message to quit smoking
- b. Determine whether the patient is willing to quit
- c. Determine the best quitting method
- d. Plan for problems associated with withdrawal
- e. Set a quit date
- f. Help the patient cope with urges to smoke
- g. Provide additional help as needed
- h. Follow-up by telephone call or visit

The stepwise strategies are summarized in Table 30 of the ACC/AHA 2002 Guideline Update for the Management of Patients with Chronic Stable Angina (71).

The most effective pharmacological adjuncts for treating nicotine dependence are nicotine replacement therapy and bupropion in the sustained-release form. When these are combined with behavioral counseling, the best treatment outcomes have been reported (1230). Nicotine replacement therapy (gum and patches) has been shown to increase smoking cessation rates, particularly when combined with counseling (1231). Nicotine replacement therapy can mitigate symptoms of nicotine withdrawal in recovering patients (1232). A population-based case-control study did not find an association between nicotine patches and first-time MI (1233). Several studies have suggested that nicotine replacement therapy does not increase the risk for cardiovascular events, even in people with underlying CHD (1234-1237), although the studies had few or no patients with MI (1235-1237) and were not designed (1235) or sufficiently powered (1234) to examine cardiovascular effects. Thus, routine use of these agents is not recommended during hospitalization for STEMI because of the sympathomimetic effects of the active ingredient, nicotine. However, the dose of nicotine in gums and patches is significantly lower than that found in cigarettes and may be preferable to cigarette smoking if the patient is experiencing acute withdrawal. Therefore, at the time of discharge, if blood pressure and heart rate are stable, these agents may be used in selected patients. Clonidine has been shown to be effective in women but not men (1238); the reason for this finding is unclear. Lobeline has not been shown to have an advantage over placebo (1239-1241) but is again under investigation.

Bupropion has been shown to help smokers quit (1242,1243). Nicotine intake is reinforced by activating the central nervous system to release norepinephrine, dopamine, and other neurotransmitters. Bupropion is a weak inhibitor of the neuronal uptake of neurotransmitters. A study of 615 subjects randomly assigned to take placebo or bupropion achieved good initial quit rates with treatment augmented by brief counseling at baseline, weekly counseling during treatment, and intermittent counseling for up to 1 year (1244). Seven weeks of treatment with bupropion was associated with a smoking cessation rate of 28.8% (100 mg/d), 38.6% (150 mg/d), and 44.2% (300 mg/d); 19.6% of subjects

assigned to placebo quit (p less than 0.001). At 1 year, 12.4% of the placebo group and 19.6% (100 mg/d), 22.9% (150 mg/d), and 23.1% (300 mg/d) of the bupropion group remained abstinent. The drug was well tolerated (37 [8%] of 462 stopped treatment prematurely because of headache, insomnia, or dry mouth), although the study was insufficiently powered to detect an incidence of seizures known to occur with related medications. It reduced the weight gain common in smokers who quit. Bupropion appears to be a valuable option for patients who need to quit smoking after STEMI. In an actual large group-practice setting, the combination of slow-release bupropion and minimal or moderate counseling was associated with 1-year quit rates of 24% and 33%, respectively (1243).

7.12.5. Antiplatelet Therapy

Class I

- 1. A daily dose of aspirin 75 to 162 mg orally should be given indefinitely to patients recovering from STEMI. (Level of Evidence: A)
- 2. If true aspirin allergy is present, preferably clopidogrel (75 mg orally per day) or, alternatively, ticlopidine (250 mg orally twice daily) should be substituted. (Level of Evidence: C)
- **3.** If true aspirin allergy is present, warfarin therapy with a target INR of 2.5 to 3.5 is a useful alternative to clopidogrel in patients less than 75 years of age who are at low risk for bleeding and who can be monitored adequately for dose adjustment to maintain a target INR range. (*Level of Evidence: C*)

Class III

Ibuprofen should not be used because it blocks the antiplatelet effects of aspirin. (Level of Evidence: C)

On the basis of 12 randomized trials in 18788 patients with prior infarction, the Antiplatelet Trialists' Collaboration reported a 25% reduction in the risk of recurrent infarction, stroke, or vascular death in patients receiving prolonged antiplatelet therapy (36 fewer events for every 1000 patients treated) (263). No antiplatelet therapy has proved superior to aspirin in this population, and daily doses of aspirin between 80 and 325 mg appear to be effective (1245). The CAPRIE trial, which compared aspirin with clopidogrel in 19 185 patients at high risk for vascular events, demonstrated a modest but significant reduction in serious vascular events with clopidogrel compared with aspirin (0.51% ARD; 8.6% RRR; p equals 0.043) (742). These data suggest clopidogrel as the best alternative to aspirin in patients with true aspirin allergy.

These compelling data suggest that all patients recovering from STEMI should, in the absence of contraindications, continue taking aspirin for an indefinite period (1246). Clopidogrel or ticlopidine may be substituted in patients with true aspirin allergy.

The use of warfarin therapy for secondary prevention of vascular events in patients after STEMI is discussed in

Section 7.12.11. Large randomized trials have demonstrated that oral anticoagulants, when given in adequate doses, reduce the rates of adverse outcomes, at the cost of a small increase in hemorrhagic events (1247-1249). In the Warfarin, Aspirin, Reinfarction Study (WARIS II), warfarin without aspirin in a dose intended to achieve an INR of 2.8 to 4.2 resulted in a significant reduction in a composite end point (death, nonfatal reinfarction, or thromboembolic stroke) compared with therapy with aspirin alone (16.7% versus 20.0%) (1247). Warfarin therapy resulted in a small but significant increase in major, nonfatal bleeding compared with therapy with aspirin alone (0.62% per year versus 0.17% per year). Chronic therapy with warfarin after STEMI presents an alternative to clopidogrel in patients with aspirin allergy.

A small increase in incidence of stroke in healthy men treated with aspirin was reported in both the American physician and the British doctors primary prevention studies (1250,1251). However, there has been no evidence of an increased incidence of stroke in studies in which aspirin was used for secondary prevention of coronary artery disease. These secondary prevention trials clearly indicate that in patients with clinical manifestations of atherosclerotic disease, aspirin reduces risk of stroke. It is likely that as a consequence of its antihemostatic effect, aspirin produces a small increase in risk of cerebral hemorrhage, which is masked by the beneficial effects of aspirin in patients with an increased risk for thromboembolic stroke but becomes manifest in healthy individuals at very low risk for this event.

Aspirin and NSAIDs are among the most commonly consumed drugs. An interaction between aspirin and ibuprofen on platelet function has been demonstrated in an in vivo model, with concomitant administration of ibuprofen (but not rofecoxib, diclofenac, or acetaminophen) antagonizing the irreversible platelet inhibition induced by aspirin (1252). A subsequent epidemiological study demonstrated increased all-cause and cardiovascular mortality in patients with cardiovascular disease taking aspirin plus ibuprofen compared with those taking aspirin alone, aspirin plus diclofenac, or aspirin plus other NSAIDs (1253). The use of NSAIDs in patients taking aspirin was also assessed in an observational subgroup analysis of the Physician's Health Study (1254). Regular (greater than 60 days per year) use of NSAIDs inhibited the clinical benefits of aspirin, although intermittent use (fewer than 59 days per year) had no effect. These findings suggest that ibuprofen may limit the cardioprotective effects of aspirin. Pending further data, clinicians should discourage patients with cardiovascular disease who are taking aspirin from using ibuprofen on a regular basis.

7.12.6. Inhibition of the Renin-Angiotensin-Aldosterone System

Class I

1. An ACE inhibitor should be prescribed at discharge for all patients without contraindications after STEMI. (Level of Evidence: A)

- 2. Long-term aldosterone blockade should be prescribed for post-STEMI patients without significant renal dysfunction (creatinine should be less than or equal to 2.5 mg/dL in men and less than or equal to 2.0 mg/dL in women) or hyperkalemia (potassium should be less than or equal to 5.0 mEq/L) who are already receiving therapeutic doses of an ACE inhibitor, have an LVEF less than or equal to 0.40, and have either symptomatic heart failure or diabetes. (Level of Evidence: A)
- **3.** An ARB should be prescribed at discharge to those STEMI patients who are intolerant of an ACE inhibitor and have either clinical or radiological signs of heart failure and LVEF less than 0.40. Valsartan and candesartan have established efficacy for this recommendation. (Level of Evidence: B)

Class IIa

In STEMI patients who tolerate ACE inhibitors, an ARB can be useful as an alternative to ACE inhibitors in the long-term management of STEMI patients, provided there are either clinical or radiological signs of heart failure or LVEF less than 0.40. Valsartan and candesartan have established efficacy for this recommendation. (Level of Evidence: B)

Class IIb

The combination of an ACE inhibitor and an ARB may be considered in the long-term management of STEMI patients with persistent symptomatic heart failure and LVEF less than 0.40. (Level of Evidence: B)

The use of ACE inhibitors early in the acute phase of STEMI and in the hospital management phase has been described earlier in Sections 6.3.1.6.9.1 and 7.4.3. Through their ability to interfere with ventricular remodeling, thereby attenuating ventricular dilation over time, ACE inhibitors improve clinical outcomes among patients with LV dysfunction (LVEF less than 0.40) after STEMI. The clinical result is a lessened likelihood for development of CHF, recurrent MI, and death. They are also of value in patients without clinical evidence of CHF but with a history of previous MI, in whom they have been shown to reduce cardiovascular mortality, future MI, and cardiac arrest.

These observations, coupled with experience in both the rat model of STEMI (1255) and large randomized clinical trials (585,1256,1257), have established that use of ACE inhibitors begun after a patient has recovered from STEMI improves long-term survival, provided the infarct was large and anterior in location, and results in significant impairment of LV contractility. Specifically, in the SAVE trial, patients took captopril at a mean 11 days after onset of infarction, which resulted in an approximate 20% reduction in mortality (1256). The Acute Infarction Ramipril Efficacy (AIRE) trial, in which patients who had been in clinical heart failure during the first day of their infarct and then were randomly assigned to either ramipril or placebo an average of 5 days after onset of infarction, resulted in significant risk reduction in all-cause mortality (6% ARD; 27% RRR) (1257). Similarly, the TRACE trial, in which patients with LV dysfunction on echocardiogram were randomly assigned to receive either trandolapril or placebo a median of 4 days after onset of infarction, demonstrated a significant reduction in mortality (7.6% ARD; 22% RRR) (1258).

The SOLVD trial evaluated the ACE inhibitor enalapril in 4228 asymptomatic patients with LVEF less than 0.35, 80% of whom had experienced a prior MI (1259). However, randomization was performed considerably later on the average than in the SAVE and AIRE trials. This prevention arm of the SOLVD trial revealed a trend toward improved mortality but not a statistically significant difference (1260). On the other hand, SOLVD did demonstrate a significant risk reduction of 20% for the combined end points of death or development of CHF requiring hospitalization.

In secondary analyses of the initial ACE inhibitor trials, the benefit of treatment appeared to be primarily in patients with anterior infarctions or LVEF below 0.40. However, based on post hoc analysis of the SAVE trial, in which the likelihood of recurrent MI was reduced by approximately 25% in treated patients, studies were initiated to determine the benefits of ACE inhibitor therapy among patients with known CHD but no clinical evidence for CHF (1261). Compelling evidence now supports the broad chronic use of ACE inhibitors after STEMI. The HOPE trial evaluated the effect of long-term (4 to 6 years) ACE inhibition therapy with ramipril 10 mg per day in 9297 high-risk patients, 2480 of whom were women. Fifty-two percent of the patients had a history of MI, and in 10%, the MI was within 1 year. Overall, there was a highly significant reduction in the combined end point of MI, stroke, and all-cause cardiovascular mortality (3.8% ARD; 22% RRR; p less than 0.001). Significant reductions were seen for each individual component of the primary end point: MI 2.4% ARD, RRR 20%; stroke 1.5% ARD, RRR 32%; and death of any cause 1.8% ARD, RRR 16%. Importantly, the study was performed in patients who were not known to have low ejection fraction or heart failure. The European Trial on Reduction of Cardiac Events With Perindopril in Stable Coronary Artery Disease (EUROPA) evaluated the use of the ACE inhibitor perindopril given 8 mg per day among 13655 patients (age range 26 to 89 years, mean 60 years) with known CHD but no history of clinical heart failure (1262). Nearly two thirds (64%) of the patients had a history of previous MI (more than 3 months before screening). After a mean follow-up of 4.2 years, treatment with perindopril was associated with a significant reduction (2% ARD; 20% RRR; p equals 0.0003) in the combined end point of nonfatal MI, cardiovascular mortality and resuscitated cardiac arrest. The benefit began to appear at 1 year and gradually progressed throughout the trial. Patients in EUROPA all had CHD but a lower risk than those in HOPE, in which the enrollment age was 55 years or older and 39% had diabetes. In EUROPA, nearly one third of the patients were younger than 55 years, and fewer had diabetes and hypertension. Moreover, the benefits of ACE inhibitor therapy with perindopril were observed in spite of relatively high use of other secondary prevention therapies such as platelet inhibitors (91%), lipidlowering medications (69%), and beta-blockers (63%). Given the results of the HOPE trial and the secondary analysis from the initial ACE inhibitor trials, ACE inhibitor therapy is recommended for all patients after STEMI unless otherwise contraindicated.

The results of the VALIANT study evaluating the ARB valsartan are discussed in Section 7.4.3. The series of CHARM studies (Candesartan in Heart Failure Assessment in Reduction of Mortality, although focusing on the evaluation of the ARB candesartan in patients with chronic heart failure, provides information that can be extrapolated to the longterm management of the STEMI patient, because 50% to 60% of the patients studied had ischemic heart disease as the cause of heart failure. In patients with symptomatic heart failure and LVEF 0.40 or less who were intolerant of ACE inhibitors (CHARM-Alternative trial), candesartan (target dose 32 mg once daily) was more effective than placebo in preventing cardiovascular death or hospital admission for heart failure (7% ARD; 23% RRR) (1263). In patients with symptomatic heart failure and LVEF 0.40 or less who were being treated with an ACE inhibitor (CHARM-Added trial), candesartan (target dose 32 mg) was more effective than placebo in preventing cardiovascular death or hospital admission for heart failure (4% ARD; 15% RRR) (1264). In patients with symptomatic heart failure but with a preserved LVEF (greater than 0.40; CHARM-Preserved trial), candesartan had no impact on cardiovascular death compared with placebo but was associated with a trend toward fewer admissions for heart failure (1265).

Given the extensive randomized trial and routine clinical experience with ACE inhibitors, they remain the logical first agent for inhibition of the renin-angiotensin-aldosterone system in the long-term management of patients with STEMI (726). The ARBs valsartan and candesartan should be administered over the long term to patients with STEMI with symptomatic heart failure who are intolerant of ACE inhibitors. As described in Section 7.4.3, the choice between an ACE inhibitor and an ARB in patients who are tolerant of ACE inhibitors over the long term will vary with individual physician and patient preference, as well as cost and anticipated side-effect profile (726).

The results of the most relevant clinical trials testing combinations of ACE inhibitors and ARBs have subtly different but clinically relevant results. Whereas the CHARM-Added (1264) trial demonstrated a reduction in the combined end point of heart failure hospitalization and death over ACE inhibition alone, the VALIANT study (725) reported that the combination of captopril and valsartan was equivalent to either alone, but with a greater number of adverse effects. Thus, when combination ACE inhibition and angiotensin receptor blockade is considered, the preferred ARB is candesartan. Although there is evidence that the combination of an ACE inhibitor and an aldosterone inhibitor is effective at reducing mortality and is well tolerated in patients with a serum creatinine less than or equal to 2.5 mg/dL and serum potassium concentration less than or equal to 5.0 mmol/L (see Section 7.4.3), much less experience exists with the combination of an ARB and an aldosterone inhibitor (24% of 2028 patients in the CHARM-Alternative trial) and the triple combination of an ACE inhibitor, ARB, and an aldosterone antagonist (17% of 2548 patients in the CHARM-Added trial) (1263,1264).

The combination of an ACE inhibitor and an ARB (valsartan 20 mg orally per day initially, titrated up to 160 mg orally twice per day, or candesartan 4 to 8 mg orally per day initially, titrated up to 32 mg orally per day) or an ACE inhibitor and an aldosterone inhibitor may be considered for the longterm management of patients with STEMI with symptomatic heart failure and ejection fraction less than 0.40, provided the serum creatinine is less than or equal to 2.5 mg/dL in men and less than or equal to 2.0 mg/dL in women and serum potassium concentration is less than or equal to 5.0 mEq/L. (See Sections 7.4.3 and 7.6.4.)

7.12.7. Beta-Blockers

Class I

- **1.** All patients after STEMI except those at low risk (normal or near-normal ventricular function, successful reperfusion, absence of significant ventricular arrhythmias) and those with contraindications should receive beta-blocker therapy. Treatment should begin within a few days of the event, if not initiated acutely, and continue indefinitely. (Level of Evidence: A)
- 2. Patients with moderate or severe LV failure should receive beta-blocker therapy with a gradual titration scheme. (*Level of Evidence: B*)

Class IIa

It is reasonable to prescribe beta-blockers to low-risk patients after STEMI who have no contraindications to that class of medications. (*Level of Evidence: A*)

The use of beta-blockers in the early phase of STEMI and in hospital management is reviewed in Sections 6.3.1.6 and 7.4.1. The benefits of beta-blocker therapy in patients without contraindications have been demonstrated with or without reperfusion, initiated early or later in the clinical course, and for all age groups. The greatest mortality benefit is seen in patients with the greatest baseline risk: those with impaired ventricular function or ventricular arrhythmias and those who do not undergo reperfusion (1266,1267). The benefits of beta-blocker therapy for secondary prevention are well established (717,1012). In patients with moderate or severe LV failure, beta-blocker therapy should be administered with a gradual titration scheme (1268). Long-term betablocker therapy should be administered to survivors of STEMI who have subsequently undergone revascularization, because there is evidence of a mortality benefit from their use despite revascularization either with CABG surgery or with PCI (1269).

Given these well-documented benefits, it is disturbing that this therapy continues to be underused, especially in highrisk groups such as the elderly (1270). Beta-blockers should be prescribed for all high-risk patients provided no contraindications are present. In patients with an extremely good prognosis (first STEMI, good ventricular function, no angina, negative stress test, and no complex ventricular ectopy), the effect of beta-blockers on survival will be less (1271). However, beta-blockers are often prescribed for these patients to minimize the likelihood of recurrent ischemic symptoms and to help control surges of heart rate and blood pressure with exertion.

Although relative contraindications may once have been thought to preclude the use of beta-blockers in some patients, evidence now suggests that the benefits of beta-blockers in reducing reinfarctions and mortality may actually outweigh the risks, even in patients with mild asthma not currently active, insulin-dependent diabetes mellitus, chronic obstructive pulmonary disease, severe peripheral vascular disease, PR interval greater than 0.24 seconds, and moderate LV failure. The use of beta-blockers in such patients requires monitoring to be certain that adverse events do not occur (1270,1272,1273).

Some controversy exists as to how long patients should be treated (1274). Data from large trials suggest that therapy should be continued for at least 2 to 3 years (851,1275). Thereafter, if the beta-blocker is well tolerated, such therapy should probably be continued in most patients, although data are lacking.

7.12.8. Blood Pressure Control

Class I

- 1. Blood pressure should be treated with drug therapy to less than 140/90 mm Hg and to less than 130/80 mm Hg for patients with diabetes or chronic kidney disease. (Level of Evidence: B)
- 2. Lifestyle modification (weight control, dietary changes, physical activity, and sodium restriction) should be initiated in all patients with blood pressure greater than or equal to 120/80 mm Hg. (Level of Evidence: B)

Class IIb

A target goal of 120/80 mm Hg for post-STEMI patients may be reasonable. (*Level of Evidence: C*)

Class III

Short-acting dihydropyridine calcium channel blocking agents should not be used for the treatment of hypertension. (Level of Evidence: B)

The Seventh Report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure (JNC-7) (1276) recommends that patients be treated after MI with ACE inhibitors, beta-blockers, and, if necessary, aldosterone antagonists to a target blood pressure of less than 140/90 mm Hg, or less than 130/80 mm Hg for those with chronic kidney disease or diabetes (1276). Most patients will require 2 or more drugs to reach goal, and when the blood pressure is greater than 20/10 mm Hg above goal, 2 drugs should usually be used from the outset. The Antihypertensive and Lipid-Lowering Treatment to Prevent Heart Attack Trial (ALLHAT) (1277) demonstrated that an antihypertensive regimen based on a thiazide diuretic was equal to an ACE inhibitor– or long-acting calcium channel blocker–based regimen in coronary outcomes and superior in other outcomes (CHF, stroke). Most patients received a betablocker in addition to the thiazide.

Beta-blockers reduce the risk for subsequent MI or sudden cardiac death and are indicated in all patients after STEMI absent specific contraindications; they may be especially efficacious in patients with CHF. In general, beta-blockers without intrinsic sympathomimetic activity should be used. Beta-blockers are sufficiently important that they should be withheld only for high-grade heart block, demonstrated intolerance, or other absolute contraindications. Most patients with asthma are able to tolerate cardioselective beta-blockers (1278). Because of the demonstrated effectiveness of thiazide diuretics for reducing adverse outcomes, they should be strongly considered when a second or third agent is needed. ACE inhibitors (or ARBs if ACE inhibitors are not tolerated) should be used long term for the treatment of hypertension when LV dysfunction is present to prevent subsequent heart failure and mortality (1279). ACE inhibitors and ARBs have been shown to favorably affect the progression of diabetic and nondiabetic kidney disease but should be given with careful attention to the development of hyperkalemia (1276). Long-acting calcium channel blockers may be used if other agents are not tolerated or are not sufficient to reach blood pressure goal. Short-acting calcium antagonists should not be used (636).

JNC-7 emphasizes the importance of lifestyle modifications for all patients with blood pressure of 120/80 mm Hg or greater (1276). These modifications include: weight reduction if overweight or obese, consumption of a diet rich in fruits and vegetables and low in total fat and saturated fat, and reduction of sodium to no more than 2.4 g/d (1276).

7.12.9. Diabetes Management

Class I

Hypoglycemic therapy should be initiated to achieve HbA1C less than 7%. (Level of Evidence: B)

Class III

Thiazolidinediones should not be used in patients recovering from STEMI who have New York Heart Association class III or IV heart failure. (Level of Evidence: B)

Tight glucose control in diabetics during and after STEMI has been shown to lower acute and 1-year mortality rates (602). Tight glucose control, defined as an HbA1C of less than 7.0%, reduces microvascular disease and is strongly recommended, although further data are needed regarding its specific benefits for macrovascular disease (1280). Control of hyperglycemia has been shown to reduce diabetes-related events, including MI (2.7% absolute reduction; 16% relative reduction), for patients aged 25 to 65 years with newly

detected type 2 diabetes but without symptomatic macrovascular disease (1280).

Thiazolidinediones are frequently used as monotherapy or in combination with other oral hypoglycemic agents, insulin, and diet for control of diabetes. Although thiazolidinediones may be helpful in combating insulin resistance and thereby may improve vascular, neurohormonal, and myocardial function, their use may also be associated with fluid retention and an increase in LV preload (1281) that is resistant to diuretics. Therefore, thiazolidinediones should not be used in patients recovering from STEMI who have New York Heart Association class III or IV heart failure (1281,1282).

7.12.10. Hormone Therapy

Class III

- 1. Hormone therapy with estrogen plus progestin should not be given de novo to postmenopausal women after STEMI for secondary prevention of coronary events. (Level of Evidence: A)
- 2. Postmenopausal women who are already taking estrogen plus progestin at the time of a STEMI should not continue hormone therapy. However, women who are beyond 1 to 2 years after initiation of hormone therapy who wish to continue such therapy for another compelling indication should weigh the risks and benefits, recognizing a greater risk of cardiovascular events. Hormone therapy should not be continued while patients are on bedrest in the hospital. (*Level of Evidence: B*)

Landmark randomized clinical trials have provided firm evidence that combination estrogen and progestin replacement therapy should not be used for either primary or secondary prevention of cardiovascular disease in women. Observational studies (1283,1284) have been interpreted as indicating that oral unopposed estrogen is effective in primary prevention of cardiovascular disease. Confounding factors such as compliance (1285), selection bias, and baseline health in these studies make it difficult to be certain of the effect of estrogen therapy alone.

Clinical trials have shown that estrogen given alone or in combination with progestin improves the lipid profile and lowers fibrinogen (1286). Favorable effects of estrogen on the lipid profile would theoretically be expected to produce a favorable result in preventing coronary atherosclerosis. However, combining estrogen with a progestin (hormone therapy) (1287) reduces the potential beneficial effect of estrogen given alone on the lipid profile. In addition, hormone therapy has been reported to increase high-sensitivity C-reactive protein levels (1288).

The first large-scale, randomized, double-blind, placebocontrolled trial that addressed the question of estrogen plus progestin for secondary prevention of CHD in postmenopausal women was published by Hulley et al. (1287) for the Heart and Estrogen/progestin Replacement Study (HERS) Research Group. Contrary to conventional wisdom and several observational studies (1289-1292), this trial of 3763 postmenopausal women with established coronary disease and an average age of 66.7 years found no reduction in overall risk for nonfatal MI or coronary death, nor any other cardiovascular outcome, during an average of 4.1 years of follow-up when comparing 0.625 mg of conjugated equine estrogen plus 2.5 mg of medroxyprogesterone acetate in 1 tablet daily (1380 patients) to placebo (1383 patients).

This lack of an overall effect occurred despite a net 11% lower LDL-C level and a 10% higher HDL-C level in the group given hormone therapy compared with the placebo group (p less than 0.001). There was a statistically significant time trend, however, with more primary coronary events in the hormone therapy group than in the placebo group in year 1 and fewer in years 4 and 5. However, the latter was due to nonfatal MI, because CHD deaths were similar in the 2 groups in years 4 and 5. More women in the hormone group than in the placebo group experienced venous thromboembolic events (34 versus 12; RR 2.89; 95% CI 1.50 to 5.58) and gallbladder disease (84 versus 62; RR 1.38; 95% CI 1.00 to 1.92). HERS II extended the unblinded follow-up of the HERS cohort (93% of survivors) for an additional 2.7 years (total 6.8 years of observation) (1293). The lower rates of CHD in the hormone therapy-assigned women in the latter years of HERS did not persist during additional observation. Hormone therapy did not reduce CHD events after 6.8 years. The Estrogen Replacement and Atherosclerosis (ERA) trial showed no effect of estrogen alone or estrogen plus progesterone on the progression of coronary artery disease, despite favorable effects on lipids (1292). A small randomized controlled trial of transdermal estradiol for women after non-STelevation acute coronary syndrome showed no benefit, with a trend toward excess events (1294). The Women's Health Initiative (WHI) Hormone Trial (HT) includes a group of women who have had hysterectomies $(10\,000)$ and receive unopposed estrogen and women with intact uteruses who receive the same estrogen plus progestin used in HERS (1295-1297). Participants are not required to have CHD and are generally younger than those in the HERS cohort. The HT trial completed its enrollment of 27 348 women and planned to report the results of the trial in 2005 after 9 years of treatment. However, the data and safety monitoring board recommended early termination of the combination of estrogen and progestin trial after 5.2 years' average follow-up on the basis of an excess of invasive breast cancer (HR 1.26, 1.00 to 1.59) and CHD (HR 1.29, 1.02 to 1.63), stroke (HR 1.41, 1.07 to 1.85), and pulmonary embolism (HR 2.13, 1.39 to 3.25) in study participants receiving active hormone replacement therapy. The estrogen-only versus placebo trial of the WHI study is continuing. In the Women's Angiographic Vitamin and Estrogen (WAVE) Trial, assignment to conjugated equine estrogen, with or without progestin, resulted in worsening of angiographic and clinical outcomes (1298,1299). In a similar trial, WELL-HART (Women's Estrogen-Progestin Lipid-Lowering Hormone Atherosclerosis Regression Trial), there was no effect of 17beta-estradiol, with or without progestin, on angiographic

progression of disease (1299). Of note, no randomized trial enrolled women within 3 to 6 months of an acute coronary syndrome.

Inherited thrombophilias may increase the risk of thrombosis due to estrogen (1300). However, none have been clearly identified that would permit screening and exclusion of women at excess risk. The role of selective estrogen receptor modulators is yet to be defined (1291,1293,1301). On the basis of WHI, HERS, and HERS II, postmenopausal women should not receive combination estrogen and progestin therapy for primary or secondary prevention of CHD (1287,1293,1297). It is recommended that hormone therapy be discontinued for women who have STEMI (1297). However, women who are beyond 1 to 2 years after initiation of HT and wish to continue it for another compelling indication should weigh the risks and benefits, recognizing a greater risk of cardiovascular events. Hormone therapy should not be continued while patients are on bedrest in the hospital.

7.12.11. Warfarin Therapy

Class I

- **1. Warfarin** should be given to aspirin-allergic post-**STEMI** patients with indications for anticoagulation as follows:
 - a. Without stent implanted (INR 2.5 to 3.5) (Level of Evidence: B).
 - b. With stent implanted and clopidogrel 75 mg/d administered concurrently (INR 2.0 to 3.0) (Level of Evidence: C)
- 2. Warfarin (INR 2.5 to 3.5) is a useful alternative to clopidogrel in aspirin-allergic patients after STEMI who do not have a stent implanted. (Level of Evidence: B)
- **3.** Warfarin (INR 2.0 to 3.0) should be prescribed for post-STEMI patients with either persistent or paroxysmal AF. (*Level of Evidence: A*)
- 4. In post-STEMI patients with LV thrombus noted on an imaging study, warfarin should be prescribed for at least 3 months (*Level of Evidence: B*) and indefinitely in patients without an increased risk of bleeding. (*Level of Evidence: C*)
- 5. Warfarin alone (INR 2.5 to 3.5) or warfarin (INR 2.0 to 3.0) in combination with aspirin (75 to 162 mg) should be prescribed in post-STEMI patients who have no stent implanted and who have indications for anticoagulation. (*Level of Evidence: B*)

Class IIa

1. In post-STEMI patients less than 75 years of age without specific indications for anticoagulation who can have their level of anticoagulation monitored reliably, warfarin alone (INR 2.5 to 3.5) or warfarin (INR 2.0 to 3.0) in combination with aspirin (75 to 162 mg) can be useful for secondary prevention. (Level of Evidence: B)



Figure 37. Long-term antithrombotic therapy at hospital discharge after ST-elevation myocardial infarction (STEMI). ASA = aspirin; INR = international normalized ratio; LOE = level of evidence. *Clopidogrel is preferred over warfarin because of increased risk of bleeding and low patient compliance in warfarin trials. †For 12 months. ‡Discontinue clopidogrel 1 month after implantation of a bare metal stent or several months after implantation of a drug-eluting stent (3 months after sirolimus and 6 months after paclitaxel) because of the potential increased risk of bleeding with warfarin and 2 antiplatelet agents. Continue aspirin and warfarin long term if warfarin is indicated for other reasons such as atrial fibrillation, LV thrombus, cerebral emboli, or extensive regional wall-motion abnormality. §An INR of 2.0 to 3.0 is acceptable with tight control, but the lower end of this range is preferable. The combination of antiplatelet therapy and warfarin may be considered in patients aged less than 75 years with low bleeding risk who can be monitored reliably.

2. It is reasonable to prescribe warfarin to post-STEMI patients with LV dysfunction and extensive regional wall-motion abnormalities. (*Level of Evidence: A*)

Class IIb

Warfarin may be considered in patients with severe LV dysfunction, with or without CHF. (Level of Evidence: C)

The indications for long-term anticoagulation after STEMI remain controversial and are evolving. Although the use of warfarin has been demonstrated to be cost-effective compared with standard therapy without aspirin, the superior safety, efficacy, and cost-effectiveness of aspirin has made it the antithrombotic agent of choice for secondary prevention (Figure 37; Table 33) (1248,1249,1302-1305). Two trials failed to demonstrate a statistically significant reduction in the combined end points of death, reinfarction, or stroke using a regimen of low-dose aspirin in combination with low-dose warfarin (INR less than 2) (1306,1307).

Several trials (1247-1249,1303-1305) have examined the use of moderate- and high-intensity warfarin in secondary prevention. Two of these trials, WARIS II and APRICOT (Antithrombotics in the Prevention of Reocclusion In Coronary Thrombolysis) II, were STEMI specific. In the APRICOT II trial (1249), patients less than 75 years old with

STEMI received UFH, aspirin, and fibrinolytic therapy. Those who achieved TIMI 3 flow were randomized to aspirin alone (80 mg) or warfarin (INR 2 to 3) plus 80 mg of aspirin. The combined group had fewer reocclusions (15% versus 28%; p less than 0.02) and a significant reduction in the combined end points of death, MI, and revascularization (20% ARD; 23% RRR; p less than 0.01) (Figure 37; Table 33) (1248,1249,1303-1305).

The WARIS II study of 3630 subjects compared highintensity warfarin (INR 2.8 to 4.2) alone, medium-intensity warfarin (INR 2 to 2.5) plus aspirin (75 mg), and aspirin alone (160 mg) (1303). Patients were less than 75 years of age. At follow-up in 4 years, the combined group had a lower risk for an event (death, nonfatal reinfarction, thromboembolic cerebral stroke) (3.3% ARD; 29% RRR; p equals 0.03) and the high-intensity warfarin group had a lower risk (5% ARD; 19% RRR; p equals 0.001) than the aspirin group. There was no survival difference, and the benefit resulted from a reduction in nonfatal MI and nonfatal thromboembolic stroke. Bleeding was more common in the warfarin groups, and approximately 35% of patients discontinued warfarin therapy (Figure 37; Table 33) (1248,1249,1303-1305).

In ASPECT (Anticoagulants in the Secondary Prevention of Events in Coronary Thrombosis) II, aspirin (81 mg) Table 33. Aspirin Versus Warfarin Therapy After ST-Elevation Myocardial Infarction (STEMI)

1	SA + rfarin %		15 >=0.001 s. ASA)	0.57	15	6	13	2 86	Ś		ہ 15	17.6 13.3 3.1 1.2†	н	
TESULTS	arfarin A Alone Wa %		16.7 =0.03 vs. (F ASA) vy	0.68	N/A						v x	N/A	N/A	
×	ASA W alone %		20 (P=	0.17	28	20	31	8 66	£		5 1 9	17.3 13.1 3.5 0.72†	0.6	
	Endpoints		Death, nonfatal reinfarction, or thromboembolic stroke	Major, nonfatal bleeding (P less than 0.001)	Reooclusion (TIMI 2 or less) (P less than 0.02)	Total occlusion (TIMI 0-1) (P less than 0.02)	Revascularization (P less than	Reinfarction (P less than 0.05) Event-free survival rate* (P less	than 0.01) Bleeding (TIMI major and minor) (P=NS)		Death, MI, or stroke (P=0.0479) Major Bleeding Minor Bleeding (P less than 0.0001) Almost 20% of the warfarin and combined group discontinued therapy; 40% in therapeutic range	Death (P=0.76) Recurrent MI (P=0.78) Stroke (P=0.52) Májor Bleeding (P less than 0.001)	Ischemic Stroke (P=0.0534)	
	Patients		Age less than 75 yrs.	Hospitalized for acute MI % STEMI=71.8	Age less than or equal to 76 vrs.	Acute STFMI less	than or equal to 6	thrombolytic Tx.	% STEMI=100		Acute MI or UA within preceding 8 wks.	Acute MI within preceding 14 days prior to enrollment	Age 21-85 yrs. (82% less than 70 yrs) MI 3-21 days (mean	9.6 days) prior to enrollment
	3 rd Arm		Dosed to target INR 2.0-	2.5 + ASA 75 mg daily	N/A						Dosed to target INR 2- 2.5 + ASA 80 mg daily	N/A	N/A	
	2 nd Arm		Dosed to target INR 2.8-4.2		Dosed to target INR 2-3	if TIMI 3 post 48 hr 11FH +	160 mg	then 80 mg daily			Dosed to target INR 3-4	Dosed to target INR 1.5-2.5+81 mg ASA daily	1mg + 80 mg ASA (avg INR @ Wk 4=1.05)	
	ASA		160 mg daily		If TIMI 3 post 48 hr UFH	then 160 mg initially and	then 80 mg	uany			80 mg daily	162 mg daily	160 mg daily (avg INR @ Wk 4= 1.02)	
	Drugs Used		ASA monotherapy vs. Warfarin	monoth c rapy vs. Warfarin + ASA	ASA monotherapy vs.	Warfarin + ASA				TEMI	ASA monotherapy vs. Warfarin monotherapy vs. Warfarin + ASA	ASA monotherapy vs. Warfarin + ASA	ASA monotherapy vs. Warfarin 1mg + ASA	
	Study Design	FIC TRIALS	Randomized Open label	N=3630 FU=Mean 4 yrs.	Randomized Open label	N=308	FI I= 3 mo	C-01		SPECIFIC TO ST	Randomized Open label N=999 FU=26 mo.	Randomized Open label N=5059 FU=Median 2.7 yrs.	Randomized Blinded N=8803	FU=33 mo. Median=14
	Study (Ref)	STEMI-SPECI	WARIS II (1303)		APRICOT II (1249)					TRIALS NOT 5	ASPECT II (1248)	CHAMP (1304)	CARS (1305)	

Antman *et al.* 2004 e233 ACC/AHA Practice Guidelines

FU = follow-up: ASA = aspirin; INR = international normalized ratio; N/A = not applicable; Tx = treatment; MI = myocardial infarction. *Events are death, revascularization, or reinfarction. †Reported as number of events per 100 person-years of follow-up.

(1248) was compared with high-intensity anticoagulation (INR 3 to 4) and with medium-intensity warfarin (INR 2 to 2.5) plus aspirin (1248) for secondary prevention in 999 subjects (1308). Significantly fewer patients in the high-dose warfarin and the combined regimen had death, MI, or stroke than in the aspirin group (5%, 5%, and 9%, respectively). Major bleeding was low in all groups. Minor bleeding was higher in the combined group. However, almost 20% of the warfarin and combined group discontinued therapy, and only about 40% remained in the therapeutic range.

Although in these studies, medium-intensity warfarin plus low-dose aspirin clearly reduced the rate of nonfatal reinfarctions and nonfatal strokes, this was achieved at the expense of increased bleeding and significant dropout rates. Patients over 75 years of age have not been studied adequately. Therefore, the Writing Committee wishes to make no definitive recommendation in this age group at this time. The high rate of discontinuation and the relatively large number of patients not on target remain problematic. Unless there are indications for anticoagulation, at this time, the Writing Committee reserves its current level IIa recommendation for those patients under 75 years of age and not at risk of bleeding who are at high risk for reinfarction or thromboembolic events and who can be monitored reliably. For patients in whom anticoagulation is indicated, warfarin (INR 2.5 to 3.5) or medium-intensity warfarin (INR 2.0 to 3.0) plus aspirin (75 to 162 mg) may be used for secondary prevention. When warfarin is used in combination with aspirin, an INR of 2.0 to 3.0 is acceptable with tight control, but the lower end of the range is preferable (Figure 37).

The use of clopidogrel and direct thrombin inhibitors in STEMI remains to be studied more thoroughly. Although definitive data are not available, the consensus of this Writing Committee is that clopidogrel is preferred over warfarin in aspirin-intolerant patients for secondary prevention unless reasons for anticoagulation are present (Figure 37). Patients who have undergone stenting may need to take aspirin, clopidogrel, and warfarin (INR 2.0 to 3.0) if anticoagulation is indicated (i.e., AF, LV thrombus, cerebral emboli, or extensive regional wall-motion abnormality) (Figure 37). In this situation, the Writing Committee believes that clopidogrel may be stopped 1 month after a bare metal stent is implanted and several months after a drug-eluting stent is implanted (3 months after sirolimus and 6 months after paclitaxel) because of the potential risk of bleeding with warfarin and the antiplatelet agents (1309,1310). (See Section 7.12.5.)

The previous ACC/AHA guidelines strongly recommended the use of oral anticoagulants with an INR of 2.0 to 3.0 in patients with a ventricular mural thrombus or a large akinetic region of the LV for at least 3 months. In a meta-analysis and other observational studies (1311-1317), patients with LV thrombus after STEMI had better outcomes and fewer cerebral emboli when they underwent anticoagulation with heparin and warfarin. Despite the absence of controlled studies, late thromboembolism was reduced in postinfarction patients with LV aneurysm treated with warfarin in a number of studies (1314-1316). A cohort analysis from the SOLVD trial (1318) demonstrated that warfarin use was associated with improved survival and reduced morbidity in post-MI patients with LV dysfunction. Other studies and a Cochrane review suggest that we be cautious in recommending warfarin for this indication alone (1017,1319). Warfarin is indicated in patients with persistent AF after STEMI, given the results of multiple trials in other patients with AF (955,958).

7.12.12. Physical Activity

Class I

- 1. On the basis of assessment of risk, ideally with an exercise test to guide the prescription, all patients recovering from STEMI should be encouraged to exercise for a minimum of 30 minutes, preferably daily but at least 3 or 4 times per week (walking, jog-ging, cycling, or other aerobic activity), supplemented by an increase in daily lifestyle activities (e.g., walking breaks at work, gardening, and household work). (Level of Evidence: B)
- 2. Cardiac rehabilitation/secondary prevention programs, when available, are recommended for patients with STEMI, particularly those with multiple modifiable risk factors and/or those moderate- to high-risk patients in whom supervised exercise training is warranted. (Level of Evidence: C)

Federal guidelines recommend that all Americans strive for at least 30 minutes of moderate physical activity most days of the week, preferably daily (1320). The 30 minutes can be spread out over 2 or 3 segments during the day. For postpatients with STEMI, daily walking can be encouraged immediately after discharge. Physical activity is important in efforts to lose weight because it increases energy expenditure and plays an integral role in weight maintenance. Physical activity reduces symptoms in patients with cardiovascular disease and improves other cardiovascular disease risk factors. Beyond the instructions for daily exercise, patients require specific instruction on those strenuous activities (e.g., heavy lifting, climbing stairs, yard work, household activities) that are permissible and those they should avoid. As emphasized by the US Public Health Service, comprehensive cardiac rehabilitation services include long-term programs involving medical evaluation, prescribed exercise, cardiac risk factor modification, education, and counseling (1184). These programs are designed to limit the physiological and psychological effects of cardiac illness, reduce the risk for sudden death or reinfarction, control cardiac symptoms, stabilize or reverse the atherosclerotic process, and enhance the psychosocial and vocational status of selected patients. Enrollment in a cardiac rehabilitation program after discharge may enhance patient education and compliance with the medical regimen and assist with the implementation of a regular exercise program (1183,1322-1324). In addition to aerobic training, mild- to moderate-resistance training is also recommended. This can be started 2 to 4 weeks after aerobic training has begun (1325).

7.12.13. Antioxidants

Class III

Antioxidant vitamins such as vitamin E and/or vitamin C supplements should not be prescribed to patients recovering from STEMI to prevent cardiovascular disease. (Level of Evidence: A)

Earlier observational data from epidemiological studies suggested that an increased intake of lipid-soluble antioxidant vitamins (vitamin E and beta-carotene) is associated with reduced rates of cardiovascular events, including STEMI (1326-1328). In support of these data, one randomized placebo-control study of vitamin E treatment in 2002 patients with documented coronary disease indicated a reduction in nonfatal MI (27% ARD; 77% RRR) but no effect on cardiovascular death or overall mortality (1329). However, a midstudy change in the vitamin E dose and some imbalance in the use of beta-adrenoceptor blockers in subjects receiving vitamin E make interpretation of that study problematic. A prospective cohort study of more than 34000 postmenopausal women suggested that an increase in dietary vitamin E but not supplemental vitamin E was associated with decreased cardiovascular risk (1330). In a randomized trial involving 11 324 patients surviving recent (less than 3 months) MI, patients were assigned to treatment with the following: vitamin E (300 mg daily; n equals 2830); n-3 polyunsaturated fatty acids (1 g daily; n equals 2836); both (n equals 2830); or neither (n equals 2828) for 3.5 years. Treatment with n-3 polyunsaturated fatty acids but not vitamin E significantly lowered the relative risk of the primary end point (death, nonfatal MI, or stroke) by 10% (p equals 0.048) (1331). Thus, although dietary supplementation with n-3 polyunsaturated fatty acids may have clinical benefits for patients after STEMI, this trial failed to demonstrate a treatment benefit for vitamin E. With regard to beta-carotene, several prospective studies have convincingly shown a lack of beneficial effect on cardiovascular disease (1332-1334), and 2 studies have indicated an increase in lung cancer with beta-carotene treatment (1332,1333).

There is even less evidence to support the use of water-soluble enzymatic antioxidants for cardiovascular disease. Although one study suggested reduced cardiovascular risk in subjects taking supplemental vitamin C (1335), the majority of other large epidemiological studies have not indicated a benefit (1326-1328). Thus, routine use of vitamin C after STEMI cannot be recommended.

Despite promising experimental studies, recombinant superoxide dismutase failed to reduce infarct size in a wellcontrolled primary PCI trial (1336). One small study showed a trend for reduced restenosis with vitamin E treatment after coronary angioplasty (restenosis rate 35.5% for treatment group versus 47.5% for placebo group; n equals 100, p equals 0.06) (1337). A larger study evaluating the combination of vitamin E in association with omega-3 fatty acids 2 weeks before elective PCI showed no effect on the restenosis rate (1338). Thus, there is no convincing evidence to support lipid- or water-soluble antioxidant supplementation in patients after STEMI or in patients with or without established coronary disease (1338a). Because these agents are not harmless, the growing practice of recommending antioxidant supplements in these patients should be discouraged until the results of ongoing, well-controlled studies become available and unequivocally indicate a beneficial effect. An extensive review of this subject has been published (1338).

8. LONG-TERM MANAGEMENT

8.1. Psychosocial Impact of STEMI

Class I

The psychosocial status of the patient should be evaluated, including inquiries regarding symptoms of depression, anxiety, or sleep disorders and the social support environment. (Level of Evidence: C)

Class IIa

Treatment with cognitive-behavioral therapy and selective serotonin reuptake inhibitors can be useful for STEMI patients with depression that occurs in the year after hospital discharge. (*Level of Evidence: A*)

Depression is a common consequence of STEMI, with major depression occurring in 15% to 20% of patients with STEMI and some degree of clinically significant depression occurring in up to half (1339). Most studies have found depression to be a significant independent predictor of post-MI mortality (1340-1345), although others have not (1346,1347). An ancillary follow-up study from the ENRICHD trial (Enhancing Recovery in Coronary Heart Disease) of a subsample of 358 depressed patients with an acute MI compared with 408 nondepressed patients found that the depressed patients were at higher risk for all-cause mortality but not until nearly 12 months after the acute event. Depression did not predict nonfatal recurrent infarction (1348). The excess risk associated with depression soon after STEMI remains significant for a longer time than previously thought, and a dose-response relationship exists between depression and mortality. The level of depression symptoms at the time of STEMI admission is more closely linked to long-term (5-year) survival than the level at 1 year, notably so in patients with moderate to severe levels of depression. Even minimal degrees of depression appear to confer risk, and risk increases with degree of depression as measured on the Beck Depression Inventory (1340). Depression markedly decreases quality of life for post-STEMI patients (1339,1346,1347,1349,1350) and is associated with substantially greater costs (1341). Fatigue does not explain the impact of STEMI in producing depression (1343), nor do infarct size, LV function, or other physiological variables predict the degree of depression (1345,1339). Depression rather than physiological variables predicts failure to return to usual activity and failure of social role resumption after MI (1351). Depressed patients are less likely to complete cardiac rehabilitation and less likely to adhere to important lifestyle changes and medications. (1352,1353).

Treatment of depression with combined cognitive-behavioral therapy and selective serotonin reuptake inhibitors improves outcome in terms of depression symptoms and social function (1350,1354,1355). A double-blind study comparing sertraline and placebo found that sertraline was associated with clinically meaningful improvement in multiple quality-of-life domains in patients hospitalized with acute coronary syndrome (74% of which was acute MI) in the previous month who had recurrent depression (1350). Although one randomized controlled trial showed no reduction in mortality or reinfarction (1355), a reanalysis suggested its follow-up was not long enough to demonstrate effect, and indeed, depression was associated with mortality in that study (1348). Therefore, it appears prudent to assess patients with STEMI for depression during hospitalization and during the first month after STEMI and to intervene and reassess yearly in the first 5 years, as appropriate. There is evidence that the STEMI experience, with its sudden and unexpected onset, dramatic changes in lifestyle, and the additive effort of comorbid life events, is a relatively traumatic event and may produce impaired coping during subsequent ischemic events (1356).

Social integration and social support repeatedly have been shown to influence outcomes after STEMI. Social integration refers to the existence of social ties (e.g., spouse, close family members, or friends) and degree of participation in group activities (e.g., family gatherings, religious affiliations). Social support refers to the actual or perceived receipt of information, materials, and/or emotional support.

Mortality from all causes, including ischemic heart disease, is lower in socially integrated individuals (1357). Recurrent cardiac events are also significantly lower among persons reporting high levels of social integration than among socially isolated persons (1358,1359). When social support was clearly defined and measured and the effect of depression was controlled for, a large prospective trial (1360) demonstrated that support did not directly predict post-MI mortality. However, high levels of support mitigated the effect of depression on post-MI mortality. A randomized controlled trial of a social support and depression management intervention similarly did not demonstrate reduced mortality (1355) but did significantly reduce social isolation.

The most effective social support interventions occur naturally. The quality of the support provided is key; support has been shown to facilitate treatment compliance, but only when policing is minimized (1362). Overprotectiveness and withholding of information or worries, either of the patient by family members or vice versa, is associated with worse outcomes (1363,1364). Telephone follow-up, cardiac rehabilitation, or other group events can be effective methods of support for socially isolated individuals (1365).

Anxiety is prevalent among hospitalized patients with STEMI but declines relatively rapidly after discharge to levels typical of the general medical population (1366). Anxiety is predictive of in-hospital recurrent ischemia and arrhyth-

mias after MI (707), and physicians' and nurses' subjective judgments of patient anxiety are not accurate compared with measurement on validated scales (709). At least one randomized controlled trial demonstrates that in-hospital anxiety and depression can be reduced by a structured nursing support intervention (714), and secondary analysis of a longer-term trial suggests that both long-term psychosocial distress and health outcomes may benefit (1367). Anxiety should be assessed at the time of hospital discharge of patients hospitalized for STEMI. A number of studies have examined psychological intervention programs designed to help post-MI patients' psychosocial and emotional adjustment. Two large post-MI programs (1341,1368) failed to achieve positive outcomes on psychological factors or prognosis. Some have observed that the type of approach used with patients recovering from MI varies in terms of its association with anxiety reduction (1367,1369). Nevertheless, one meta-analysis reported that the addition of psychosocial interventions to standard treatment resulted in significantly less depression, anxiety, morbidity, and mortality (1370). Psychosocial interventions in cardiac rehabilitation were found in another review to improve the odds for mortality and recurrence of nonfatal MI, but not necessarily with regard to females and older participants (1369). A secondary analysis of a longerterm trial suggests that both long-term psychosocial prognosis and health outcomes may improve in patients whose psychological status improves (1367).

8.2. Cardiac Rehabilitation

Class I

Cardiac rehabilitation/secondary prevention programs, when available, are recommended for patients with STEMI, particularly those with multiple modifiable risk factors and/or those moderate- to high-risk patients in whom supervised exercise training is warranted. (Level of Evidence: C)

Cardiac rehabilitation programs are designed to limit the physiological and psychological effects of cardiac illness, reduce the risk for sudden death or reinfarction, control cardiac symptoms, stabilize or reverse the atherosclerotic process, and enhance the psychosocial and vocational status of selected patients (1184,1371,1372). Cardiac rehabilitation is a comprehensive long-term program that involves medical evaluation, prescribed exercise, cardiac risk factor modification, education, and counseling (1184,1373). Cardiac rehabilitation can occur in a variety of settings, including supervised groups in a hospital, physician's office, or community facility. In clinically stable lower-risk patients, rehabilitation can be undertaken independently, with regular guidance from a cardiac rehabilitation healthcare professional (1184). The exercise can be supervised or unsupervised and can involve a stationary bicycle, treadmill, calisthenics, walking, or jogging. Home exercise training programs have been shown to be beneficial in certain low-risk patient groups. They offer

The pooled effect estimate for total mortality for the exercise-only intervention demonstrated a reduction in all-cause mortality (random effects model OR 0.73 [0.54, 0.98]) compared with usual care. Comprehensive cardiac rehabilitation reduced all-cause mortality but to a lesser degree (OR 0.87 [0.71, 1.05]). Neither of the interventions had any effect on the occurrence of nonfatal MI. The authors concluded that exercise-based cardiac rehabilitation appeared to be effective in reducing cardiac deaths but that it was still unclear whether an exercise-only or a comprehensive cardiac rehabilitation intervention was more beneficial. The population studied was predominantly male, middle-aged, and low risk. The authors suggested that those who may have benefited from the intervention were excluded owing to age, gender, or comorbidity. The authors cautioned that the results were of limited reliability because the quality of reporting in the studies was generally poor, and there were high losses to follow-up (1373).

Cardiac rehabilitation comprising exercise training and education, counseling, and behavioral interventions yielded improvements in exercise tolerance with no significant cardiovascular complications, improvements in symptoms (decreased anginal pain and improved symptoms of heart failure such as shortness of breath and fatigue), and improvements in blood lipid levels; reduced cigarette smoking in conjunction with a smoking cessation program; decreased stress; and improved psychosocial well-being (1184). In addition to reductions in total cholesterol and LDL-C, increases in HDL levels have been reported (1197).

Existing community studies reveal that fewer than one third of patients with STEMI receive information or counseling about cardiac rehabilitation before being discharged from the hospital (1184,1375). Only 16% of patients in a study of 5 hospitals in 2 Michigan communities were referred to a cardiac rehabilitation program at discharge, and only 26% of the patients later interviewed in the community reported actual participation in such a program. However, 54% of the patients referred at discharge did participate at the time of their follow-up interview (1375). Physician referral (to a cardiologist/cardiac surgeon) was the most powerful predictor of patient participation in a cardiac rehabilitation program.

In a longitudinal study of the use of inpatient cardiac rehabilitation in 5204 Worcester, MA, residents hospitalized with MI in 7 1-year periods between 1986 and 1997, patients not referred to inpatient cardiac rehabilitation were less likely to be prescribed effective cardiac medications and to undergo risk factor modification counseling before discharge (1376).

Patient reasons for nonparticipation and noncompliance include affordability of service, possible insurance coverage, social support from a spouse or other caregiver, gender-specific attitudes, patient-specific internal factors such as anxiety or poor motivation, and logistical and financial constraints, or a combination of these factors (1375,1377). Women and the elderly have been reported to be referred less frequently to cardiac rehabilitation programs, even though they have been reported to derive benefit (1378-1380).

8.3. Follow-Up Visit With Medical Provider

Class I

- **1.** A follow-up visit should delineate the presence or absence of cardiovascular symptoms and functional class. (*Level of Evidence: C*)
- 2. The patients' list of current medications should be reevaluated in a follow-up visit, and appropriate titration of ACE inhibitors, beta-blockers, and statins should be undertaken. (Level of Evidence: C)
- 3. The predischarge risk assessment and planned workup should be reviewed and continued (Figure 36). This should include a check of LV function and possibly Holter monitoring for those patients whose early post-STEMI ejection fraction was 0.31 to 0.40 or lower, in consideration of possible ICD use (Figure 32). (Level of Evidence: C)
- 4. The healthcare provider should review and emphasize the principles of secondary prevention with the patient and family members (Table 32) (68). (Level of Evidence: C)
- 5. The psychosocial status of the patient should be evaluated in follow-up, including inquiries regarding symptoms of depression, anxiety, or sleep disorders and the social support environment. (Level of Evidence: C)
- 6. In a follow-up visit, the healthcare provider should discuss in detail issues of physical activity, return to work, resumption of sexual activity, and travel, including driving and flying. The MET values for various activities are provided as a resource (Table 34). (Level of Evidence: C)
- 7. Patients and their families should be asked if they are interested in CPR training after the patient is discharged from the hospital. (*Level of Evidence: C*)
- 8. Providers should actively review the following issues with patients and their families:
 - a. The patient's heart attack risk. (Level of Evidence: C)
 - b. How to recognize symptoms of STEMI. (Level of Evidence: C)
 - c. The advisability of calling 9-1-1 if symptoms are unimproved or worsening after 5 minutes, despite feelings of uncertainty about the symptoms and fear of potential embarrassment. (Level of Evidence: C)
 - d. A plan for appropriate recognition and response to a potential acute cardiac event, including the phone number to access EMS, generally 9-1-1. (Level of Evidence: C)
- **9.** Cardiac rehabilitation/secondary prevention programs, when available, are recommended for patients with STEMI, particularly those with multiple modifiable risk factors and/or those moderate- to high-risk patients for whom supervised exercise training is warranted. (*Level of Evidence: C*)

	2-C INTERS	5-7 METs	7-9 METS	More Than 9 METs
Self-care Washing Shaving Dressing Desk work Washing dishes Driving auto Light housekeeping	Cleaning windows Raking Power lawn mowing Bed making/stripping Carrying objects (15-30 lb)	Easy digging in garden Level hand lawn mowing Climbing stairs (slowly) Carrying objects (30-60 lb) Digging vigorously	Sawing wood Heavy shoveling Climbing stairs (moderate speed) Carrying objects (60-90 lb)	Carrying loads upstairs (objects more than 90 lb) Climbing stairs (quickly) Shoveling heavy snow
Occupational Sitting (clerical/ assembly) Typing Desk work Standing (store clerk)	Stocking shelves (light objects) Auto repair Light welding/carpentry	Carpentry (exterior) Shoveling dirt Sawing wood Operating pneumatic tools	Digging ditches (pick and shovel)	Lumber jack Heavy labor
Recreational Golf (cart) Knitting Hand sewing	Dancing (social) Golf (walking) Sailing Tennis (doubles) Volleyball (6 persons)	Badminton (competitive) Tennis (singles) Snow skiing (downhill) Light backpacking Basketball Football Stream fishing	Canoeing Mountain climbing Paddle ball	Handball Squash Ski touring Vigorous basketball
Physical conditioning Walking (2 mph) Stationary bike Very light calisthenics	Level walking (3-4 mph) Level biking (6-8 mph) Light calisthenics	Level walking (4.5-5.0 mph) Bicycling (9-10 mph) Swimming, breast stroke	Level jogging (5 mph) Swimming (crawl stroke) Rowing machine Heavy calisthenics Bicycling (12 mph)	Running more than 6 mph Bicycling (more than 13 mph) Rope jumping Walking uphill (5 mph)
Modified from Haskell. Rehabilitation o (1393).	f the coronary patient. In: Wenger and Hellerste	in, eds. Design & implementation of cardia	c conditioning program. New York, NY	: Churchill Livingstone; 1978, Table 9.2, p. 14

ACC - www.acc.org AHA - www.americanheart.org

e238

A caring and supportive doctor-patient relationship is vital to the well-being of the survivor and their families. It is common practice to see these patients 3 to 6 weeks after hospital discharge to assess their progress.

The physician should listen to the concerns of their patients and their patients' families and respond flexibly. Future appointments should be scheduled and should reflect the goals set for each individual patient in accordance with their needs and current guidelines.

8.4. Return to Work and Disability

Return-to-work rates, which currently range from 63% (1381) to 94% (1382), are difficult to influence because they are confounded by factors such as job satisfaction, financial stability, and company policies (1383). In PAMI-II, a study of primary PTCA in low-risk patients with STEMI (i.e., age less than 70 years, ejection fraction greater than 0.45, 1- or 2-vessel disease, and good PTCA result), patients were encouraged to return to work at 2 weeks. The actual timing of return to work was not reported, but no adverse events occurred as a result of this strategy (1384).

Disability. There is some evidence that a cardiac rehabilitation program after STEMI contributes to reduction of mortality and improved physical and emotional well-being (see Section 8.2). Patients whose expectations for return to work were addressed in rehabilitation returned to work at a significantly faster rate than the control group in a prospective study (1385).

A low level of depressive symptoms before STEMI increases the odds of recovery of functional status (1386). Patients with high pre-STEMI functional independence measurement have shorter length of stay and greater likelihood of discharge to home (1387). Pre-STEMI peak aerobic capacity and depression score are the best independent predictors of post-STEMI physical function. Women tend to have lower physical function scores than men of similar age, depression score, and comorbidity. Resting LVEF is not a predictor of physical function score.

Patients' cardiac functional states are not a strong predictor of their probability of returning to work. Diabetes, older age, Q-wave MI, and preinfarction angina are associated with failure to resume full employment (1388). However, psychological variables such as trust, job security, patient feelings about disability, expectations of recovery by both physician and patient, and degree of somatizing are more predictive (1389,1390). Physical requirements of the job play a role as well (1388,1390).

To aid occupational physicians in making return-to-work decisions, Froom et al. (1388) studied the incidence of post-MI events at 1, 2, 4, 6, 9, and 12 months. The events included cardiac death, recurrent infarction, CHF, and unstable angina. They found that the incidence of events reached a low steady state at 10 weeks.

Return to work can be determined by employer regulations rather than by the patient's medical condition. It behooves the physician to provide data to prove that the patient's job does not impose a prohibitive risk for a cardiac event. An example is the case of the Canadian bus drivers reported by Kavanagh et al. (1391). The patients were evaluated with a stress test. The physician and technologist studied the drivers at work and showed that the cardiac stress values during driving were only half of the average values obtained in the stress laboratory. The calculated risk of sudden cardiovascular incidence causing injury or death to passengers, other road users, and the drivers in the first year after recovery from an MI was 1 in 50 000 driving years. The bus drivers were allowed to return to work after they satisfied the Canadian Cardiovascular Society guidelines.

Women have entered the job market and are faced not only with the gender difference but in procedure difference and mortality. Covinsky et al. (1392) performed a mail survey study of patients with MIs. Three months after discharge, women reported worse physical and mental health and were more likely to work less than before the MI. Similarly, women were less likely to return to work than men.

The current aggressive interventional treatment of STEMI will have an impact on mortality, morbidity, and hospital length of stay (696). It remains to be determined whether earlier improvement in cardiac condition after STEMI will have an effect on the rate of return to work because of the multiple noncardiac factors that influence disability and return to work.

8.5. Other Activities

In patients who desire to return to physically demanding activities early, the safety of activity can be determined by comparing performance on a graded exercise test with the MET level required for the desired activity. Table 34 presents energy levels, expressed in METs, required to perform a variety of common activities (1393). This and similar tables can be helpful in translating a patient's performance on a graded exercise test into daily activities that may be undertaken with reasonable safety.

The physician should provide explicit advice about when to return to previous levels of physical activity, sexual activity, and employment. Daily walking can be encouraged immediately (1394). In stable patients without complications (class I), sexual activity with the usual partner can be resumed within 1 week to 10 days. Driving can begin 1 week after discharge if the patient is judged to be in compliance with individual state laws. Each state's Department of Motor Vehicles or its equivalent has mandated certain criteria, which vary from state to state and must be met before operation of a motor vehicle after serious illness (1395) These include such caveats as the need to be accompanied and to avoid stressful circumstances such as rush hour, inclement weather, night driving, heavy traffic, and high speeds. For patients who have experienced a complicated STEMI (one that required CPR or was accompanied by hypotension, serious arrhythmias, high-degree block, or CHF), driving should be delayed 2 to 3 weeks after symptoms have resolved.



Committee Member Name	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant
Dr. Elliott M. Antman	Aventis Bayer Biosite Boehringer Mannheim Bristol-Myers Squibb British Biotech Centor Cor/Millennium Corvas Dade Genentech Lilly Merck Pfizer Sunol	None	None	Aventis
Dr. Daniel T. Anbe	None	None	None	None
Dr. Paul W. Armstrong	Boehringer Ingelheim Eli Lilly Hoffmann-LaRoche Millennium Proctor & Gamble Sanofi	Aventis Hoffman-LaRoche	Medicure, Inc.	Aventis Hoffmann-LaRoche Medicure, Inc.
Dr. Eric R. Bates	Aventis Centocor Cor/Millennium Eli Lilly Genentech The Medicines Co. Merck	None	None	None
Dr. Lee A. Green	None	None	None	None
Ms. Mary Hand	None	None	None	None
Dr. Judith S. Hochman	Merck Aventis Cor/Millennium Guidant Lilly			Daichii Proctor & Gamble
Dr. Harlan M. Krumholz	None	None	None	Astra-Zeneca BMS/Sanofi CVT
Dr. Frederick G. Kushner	Aginamoto Co. Andrx Labs Atherogenics, Inc. Boehringer-Ingelheim Medtronic Novartis Rorer Schering-Plough	Bristol-Myers Squibb Merck Pfizer Reliant	Abbott Labs Baxter Guidant Medtronic Merck Pfizer	Millennium, Inc.

APPENDIX 1. ACC/AHA Committee to Revise the 1999 Guidelines for Management of Acute Myocardial Infarction—Relationships With Industry

Continued on next page

Most commercial aircraft are pressurized to 7500 to 8000 feet and therefore could cause hypoxia due to the reduced alveolar oxygen tension. The maximum level of pressurization is limited to 8000 feet (2440 meters) by Federal Aviation Administration regulation (1396). Therefore, air travel within the first 2 weeks of STEMI should be undertaken only if there is no angina or dyspnea at rest or fear of flying. The individual must have a companion, carry nitroglycerin, and request airport transportation to avoid rushing (personal communication, R.P. Gardner, PhD, November 2002). Air travel for cardiac patients should gradually become safer. Availability of an emergency medical kit and AED has been mandated for April 12, 2004, (1398) in all aircraft that carry at least approximately 30 passengers and have at least 1 flight attendant.

STAFF

American College of Cardiology

Christine W. McEntee, Chief Executive Officer

Katherine D. Doermann, Specialist, Knowledge Development

Kristina N. Petrie, MS, Research Analyst, Knowledge Development

American Heart Association

M. Cass Wheeler, Chief Executive Officer

- Rose Marie Robertson, MD, FACC, FAHA, Chief Science Officer
- Kathryn A. Taubert, PhD, FAHA, Vice President, Science and Medicine

Fernando Costa, MD, FAHA, Staff Scientist

ACC - www.acc.org AHA - www.americanheart.org

APPENDIX 1. Continued

Committee Member Name	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant
Dr. Gervasio A. Lamas	Astra Zeneca Bayer Centocor Bristol-Myers Squibb Guidant Lilly Medtronic Novartis Schering	None	None	None
Dr. Charles J. Mullany	Cardiogenesis	None	None	None
Dr. Joseph P. Ornato	Genentech Meridian Medical Corp. Wyeth	None	None	Bristol-Myers-Squibb Genentech HP/Agilent Medtronic Meridian Medical Corp. Philips PhysioControl/Scios Revivant Corp. Wyeth
Dr. David L. Pearle	None	GlaxoSmithKlein Novartis Pfizer	None	None
Dr. Michael A. Sloan	Abbott Labs Astra Zeneca Bristol-Myers Squibb Genentech Novo-Nordisk	Boehringer-Ingelheim Dupont Genentech Merck Nicolet-EME	None	None
Dr. Sidney C. Smith, Jr.	Merck Pfizer	None	Medtronic	None

This table represents the actual or potential relationships of committee members with industry that were reported orally at the initial writing committee meeting in January 2002 and updated in conjunction with all meetings and conference calls of the writing committee. It does not reflect any actual or potential relationships at the time of publication.

Antman *et al.* 2004 e243 ACC/AHA Practice Guidelines

APPENDIX 2. External Peer Reviewers for the ACC/AHA Guidelines for the Management of Patients With ST-Elevation Myocardial Infarction*

Peer Reviewer Name†	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. John W. Hirshfeld, Jr.	Official Reviewer – ACCF Board of Trustees	None	None	None	None
Dr. W. Barton Campbell	Official Reviewer – ACCF Board of Governors	None	None	None	None
Dr. Joseph S. Alpert	Official Reviewer – ACC/AHA Task Force on Practice Guidelines	None	None	None	None
Dr. Scott Grundy	Official Reviewer – AHA	Abbott Laboratories Astra Zeneca Bristol-Myers Squibb GlaxoSmithKline Kos Pharmaceuticals McNeil Pharmaceuticals Merck Pfizer Veterans Affairs	Abbott Laboratories Astra Zeneca Bristol-Myers Squibb Cooper Aerobics Center Kos Pharmaceuticals McNeil Pharmaceuticals Merck Pfizer Tularik, Inc.	None	None
Dr. L. Kristin Newby	Official Reviewer – AHA; AHA Committee on Acute Cardiac Care	Bristol-Myers Squibb Dade-Behring Millennium Novartis Roche Diagnostics Roche Pharmaceuticals Sanofi Schering-Plough Research Institute	Biosite Bristol-Myers Squibb- Sanofi Ischemia Technologies Institute	None	None
Dr. E. Magnus Ohman	Official Reviewer – AHA; AHA Committee on Acute Cardiac Care	Aventis Berlex Bristol-Myers Squibb Merck Millennium Pfizer Sanofi	None	Medtronic	None
Dr. Deborah Allen	Organizational Reviewer – American Academy of Family Physi <mark>c</mark> ians	None	None	None	None
Dr. Peter Bogaty	Organizational Reviewer – Canadian Cardiovascular Society	Merck Pfizer	Abbott Laboratories	Merck	None
Dr. Jennifer Adgey	Content Reviewer – Individual Review	None	None	None	None
Dr. Jeffrey Anderson	Content Reviewer – ACC/AHA Task Force on Practice Guidelines	None	Johnson & Johnson/Merck Merck Merck/Schering Plough	None	Alteon Merck Johnson & Johnson/Merck
Dr. Jean Barbey	Content Reviewer – Pharmacological Review	None	None	None	None
Dr. Eugene Braunwald	Content Reviewer – ACC/AHA 2002 Guideline Update for Management of Unstable Angina	None	None	None	None

*Participation in the peer review process does not imply endorsement of the document. †Names are listed in alphabetical order within each category of review. ACC - www.acc.org AHA - www.americanheart.org

APPENDIX 2. Continued*

Peer Reviewer Name†	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. Robert Califf	Content Reviewer -	Accumetrics	Aventis	None	GlaxoSmithKline
Di. Robert Califi	ACC/AHA 2002 Guideline	Actelion	Bristol-Myers Squibb	rtone	Pfizer
	Update for Management of	Ajinomoto	Conceptis		
	Unstable Angina	Alsius	GlaxoSmithKline		
		Amgen	Merck		
		Astra Hassle	Millennium		
		Aventis	Novartis		
		Biomarin	Ortho Biotech		
		Boston Scientific	Pennside Partners		
		Bracco	Pfizer		
		Bristol-Myers Squibb	Pharmacia/Upjohn		
		Cambridge Heart	Pharsight		
		Cardiodynamics	Schering Plough		
		Centocor	Wyeth Ayerst		
		Chase Medical			
		Coagulation Diagnostics			
		Corcept			
		Corgentech			
		Critline			
		Dade Behring			
		Daiichi			
		Datascope			
		Flan Pharmaceuticals			
		Enzon			
		Esai			
		Geneceutics			
		Genentech			
		GlaxoSmithKline			
		Guilford Pharmaceuticals			
		Harvard Health Care			
		Hemosol			
		InfraReDx			
		Intracel			
		Lincare			
		Medicure			
		Medivance			
		Medtronic Foundation			
		Merck			
		NABI			
		Novartis			
		Ortho Biotech			
		Otsuka			
		Parke Davis			
		Pfizer			
		Pharmacia/Opjoini Pheromone Science			
		Proctor and Gamble			
		Promethesus			
		Quanam			
		Salix			
		Sanof1 Spectrapetics			
		St. Jude Medical			
		Synaptic			
		The Medicines Company			
		Theravance			
		Vesicor			
		Vicuron Wyeth Averst			
		Yamanouchi			

*Participation in the peer review process does not imply endorsement of the document. †Names are listed in alphabetical order within each category of review.

Antman *et al.* 2004 ACC/AHA Practice Guidelines **e245**

APPENDIX 2. Continued*

Peer Reviewer† Name	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. Kanu Chatterjee	Content Reviewer – ACC/AHA 2002 Guideline Update for Management of Chronic Stable Angina	None	Astra Zeneca CV Therapeutics Bristol-Myers Squibb Merck Novartis Pfizer	None	None
Dr. James Cleeman	Content Reviewer – Individual Review	None	None	None	None
Dr. Stephen Ellis	Content Reviewer – ACCF Clinical Data Standards for Acute Coronary Syndromes	Medtronic	None	None	Cordis Guidant Boston Scientific
Dr. Nathan Every	Content Reviewer – ACCF Clinical Data Standards for Acute Coronary Syndromes	Genentech	None	None	None
Dr. Daniel Fintel	Content Reviewer – Individual Review	None	Merck Aventis Sanofi	None	Aventis
Dr. Valentin Fuster	Content Reviewer – ACC/AHA Task Force on Practice Guidelines	Cordis GlaxoSmithKline Schering AG	None	None	GlaxoSmithKline
Dr. Bernard Gersh	Content Reviewer – Individual Review	None	None	None	None-
Dr. Raymond Gibbons	Content Reviewer – ACCF Cardiovascular Imaging Committee	Alsius Corp Boehringer Ingelheim (pending) Boston Scientific (pending) Innercool Therapies King Pharm Medtronic Radiant Medical Ther Ox Wyeth-Ayerst	None	None	Boehringer Ingelheim Biostream (pending) CV Therapeutics DOV Pharmaceuticals GlaxoSmithKline Hawaii Biotech King Pharm Medicure
Dr. W. Brian Gibler	Content Reviewer – AHA Committee on Cardiac Care	Bristol-Myers Squibb Acute Millennium/Schering Plough	None	None	None
Dr. A. Daniel Glassman	Content Reviewer – Individual Review	None	None	None	None
Dr. Gabriel Gregoratos	Content Reviewer – ACC/AHA Task Force on Practice Guidelines	None	None	None	GlaxoSmithKline
Dr. Cindy Grines	Content Reviewer – Individual Review	Amersham Health Aventis Berlex Eli-Lilly Esperion Therapeutics GlaxoSmithKline Guidant Innercool Therapies Johnson & Johnson Otsuka Pfizer SCIMED	None	None	Aventis Innercool Therapies Guidant The Medicines Company Pfizer
Dr. Timothy Henry	Content Reviewer – ACCF Emergency Cardiac Care Committee	Anges Aventis BMS Berlex Boston Scientific Cordis Genentech Genzyme The Medicines Company Medtronic Millennium	None	None	None
Dr. L. David Hillis	Content Reviewer – ACC/AHA Guidelines Update for Coronary Artery Bypass Graft Surgery (in progre	None (ss)	None	None	None

*Participation in the peer review process does not imply endorsement of the document. †Names are listed in alphabetical order within each category of review. **APPENDIX 2.** Continued*

Peer Reviewer Name†	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. Loren Hiratzka	Content Reviewer – ACC/AHA Task Force on Practice Guidelines	None	None	None	None
Dr. David Holmes	Content Reviewer – ACCF Clinical Competence Statement on Interventional Training Procedures	None	None	None	None
Dr. Alan Jaffe	Content Reviewer – Individual Review	Beckman-Coulter Dade-Behring Roche	None	None	Abbott Laboratories Aventis Beckman Dade-Behring Diadexus Ortho Pfizer Roche Sensera
Dr. Desmond Julian	Content Reviewer – Individual Review	None	None	None	AstraZeneca Aventis Eli Lilly GlaxoSmithKline Merck Novartis Pfizer Servier
Dr. Carlos Kase	Content Reviewer – Neurological Review	None	None	None	None
Dr. Dean Kereiakes	Content Reviewer – ACC/AHA 2002 Guideline Update for Management of Unstable Angina	None	None	None	None
Dr. J. Ward Kennedy	Content Reviewer – ACC/AHA/SCAI Guideline Update for Percutaneous Coronary Intervention (in progress)	None	None	None	None
Dr. Peter Libby	Content Reviewer – ACCF Peripheral Vascular Disease Committee	Astra Zeneca Bayer Bristol-Myers Squibb Merck Novartis Pfizer Sankyo SmithKline Beecham	Astra Zeneca Bayer Bristol-Myers Squibb Merck Novartis Pfizer	None	Astra Zeneca Avant Immunotherapeutics Bayer Bristol-Myers Squibb Interleukin Genetics Merck Millennium Novartis Pfizer Pierre Fabre Sankyo Sanofi Schering-Plough SmithKline Beecham Volcano Therapeutics
Dr. A. Michael Lincoff	Content Reviewer – Individual Review	Eli Lilly Centocor The Medicines Company	None	None	None
Dr. Kathleen McCauley	Content Reviewer – Individual Review	None	None	None	None

*Participation in the peer review process does not imply endorsement of the document. †Names are listed in alphabetical order within each category of review.

Continued on next page

APPENDIX 2. Continued*

Peer Reviewer† Name	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. George McKendall	Content Reviewer – Secondary ACCF Board of Governors Reviewer; ACCF Cardiac Catheterization and Intervention Committee	None	None	None	None
Dr. Richard Pasternak	Content Reviewer – ACC/AHA 2002 Guideline Update for Management of Chronic Stable Angina	None	Bristol-Myers Squibb/Sanofi Merck Merck/Schering-Plough Kos Pfizer	None	Astra Zeneca Bristol-Myers Squibb/Sanofi Iohnson & Johnson-Merck Kos Merck Pfizer Health Solutions
Dr. Marc Pfeffer	Content Reviewer – A ACCF Hypertensive Diseases Brist Committee Mitsub	Astra Zeneca Aventis Aventis ol-Myers Squibb Genzyme oishi Pharma Corp Novartis Pfizer	Astra Zeneca Aventis Bristol-Myers Squibb Genzyme Mitsubishi Pharma Corp Novartis Pfizer		Ortho Biotech
Dr. Charles Pollack	Content Reviewer – Individual Review	Aventis Sanofi	Aventis Bristol-Myers Squibb Key ACS Millennium Scios		Aventis Key ACS
Dr. Barbara Riegel	Content Reviewer – Individual Review	None	None	None	None
Dr. Rose Marie Robertson	Content Reviewer – Individual Review	None	None	None	Merck (spouse)
Dr. Richard Russell	Content Reviewer – ACC/AHA/ASNC 2003 Guideline Update for Radionuclide Imaging	None	None	None	None
Dr. Thomas Ryan	Content Reviewer – ACC/AHA Guidelines for Management of Patients with Acute Myocardial Infarction	None	None	Abbott Labs Amgen ardinal Health Care King Pharm	GenVec
Dr. Cathy A Sila	Content Reviewer – Neurological Review	None	None	None	Centocor Paion Bristol-Myers/Sanofi Squibb
Dr. Burton Sobel	Content Reviewer – Individual Review	None	None	None	None
This table represents the relationship: *Participation in the peer review proc †Names are listed in alphabetical ord.	of peer reviewers with industry that were disclosed ess does not imply endorsement of the document. 2r within each category of review.	l at the time of peer review of this gui	deline. It does not necessarily reflect	relationships with industry	at the time of publication.

Continued on next page

APPENDIX 2. Continued*					
Peer Reviewer† Name	Representation	Research Grant	Speakers Bureau/ Honoraria	Stock Ownership	Consultant/ Advisory Board
Dr. John Stratton	Content Reviewer – Individual Review	None	None	None	None
Dr. Frans Van de Werf	Content Reviewer – ACCF Clinical Data Standards for Acute Coronary Syndromes	Aventis Boehringer Ingelheim Genentech Menarini Novartis Servier	Aventis Bochringer Ingelheim Genentech Menarini Novartis Servier	None	Aventis Bochringer Ingelheim Genentech Menarini Novartis Servier
Dr. Nanette Wenger	Content Reviewer – Individual Review	Astra Zeneca Bristol-Myers Squibb Eli Lilly	Aventis Bristol-Myers Squibb Eli Lilly Merck Pfizer Wyeth-Ayerst	None	Aventis Bristol-Myers Squibb Merck Pfizer Women First Healthcare, Inc.
Dr. Harvey White	Content Reviewer – Individual Review	None	None	None	The Medicines Company
Dr. Clyde Yancy	Content Reviewer – ACC/AHA Guideline Update for Heart Failure (in progress)	None	Scios	None	GlaxoSmithKline
This table represents the relation *Participation in the peer review †Names are listed in alphabetical	ships of peer reviewers with industry that wer process does not imply endorsement of the dt l order within each category of review.	re disclosed at the time of peer review of ocument.	this guideline. It does not necessarily refl	ect relationships with i	ndustry at the time of publication.

ACC - www.acc.org AHA - www.americanheart.org

APPENDIX 3. ABBREVIATIONS

ED

NHAAP

AAFP	= American Academy of Family Physicians	EMS EMT
ACC	= American College of Cardiology	EP
ACE	= angiotensin converting enzyme	EPHE
ACEP	= American College of Emergency	
	Physicians	CIK
ACLS	= advanced cardiac life support	CISSI
AED	= automated external defibrillator	61551
AF	= atrial fibrillation	
AFFIRM	= Atrial Fibrillation Follow-up	CP
	Investigation of Rhythm	CUST
	Management	GUSI
AHA	= American Heart Association	
ALS	= advanced life support	UEDU-C
AMI	= acute myocardial infarction	IIEKU
AMI-SK	= Acute Myocardial Infarction-	UFDS
	Streptokinase	пекз
ARB	= angiotensin receptor blocker	HODE
ARD	= absolute risk difference	HOLF
ASSENT	= Assessment of the Safety and	haCDD
	Efficacy of a New Thrombolytic	
	Regimen	
ATP III	= Adult Treatment Panel III report	
AV	= atrioventricular	
AVID	= Antiarrhythmics Versus Implantable	
-	Defibrillators	
bpm	= beats per minute	ICIC
CABG	= coronary artery bypass graft surgery	1515
CAPRICORN	= Carvedilol Post-infarct Survival	117
CADDIE	Controlled Evaluation	
CAPRIE	= Clopidogrel vs. Aspirin in Patients at	JNC-/
CADTIN	Risk of Ischemic Events	LAD
CAPTIM	= Comparison of Primary Angioplasty	LAD
	A sute Dhase of Muscondial	IRRR
	Acute Phase of Myocardian	I DI .C
CADE	- Chalasteral And Requirement Events	LIPID
	= Choresterol And Recurrent Events	
	- Coperative Caldiovascular Floject	LMW
CCU	= coronary care unit	LUIUI
СНО	= coronary heart disease	LVAD
CHE	= congestive heart failure	LVEF
CIIF	= confidence interval	
CK	- creatine kinase	
C.PORT	- Cardiovascular Patient Outcomes	MAGI
C-I OKI	Research Team	MDPI
CPR	- cardionulmonary resuscitation	
CT	= computed tomography	MET
CURE	= Clonidogrel in Unstable angina to	MI
	nrevent Recurrent Events	MR
DANAMI	= DANish multicenter Trial in Acute	MUST
	Myocardial Infarction	
DRS	= Diltiazem Reinfarction Study	NASCI
DVT	= deep venous thrombosis	
ECG	= electrocardiogram	NCEP

= extracorporeal membrane

oxygenation

ECMO

	= emergency department
	= emergency medical services
	= emergency medical technician
	= electrophysiology
SUS	= Eplerenone Post-Acute Myocardial
	Infarction Heart Failure Efficacy and
	Survival Study
	= glucose-insulin-potassium
	= Gruppo Italiano per lo Studio della
	Sopravvivenza nell'Infarto
	Miocardico
	= glycoprotein
0	= Global Utilization of Strategies To
	Open occluded arteries
C	= high-density lipoprotein cholesterol
) 🖌	= Hirulog and Early Reperfusion or
	Occlusion trial
	= Heart and Estrogen/Progestin
	Replacement Study
	= Heart Outcomes Prevention
	Evaluation
	= high-sensitivity C-reactive protein
	= intra-aortic balloon pump
	= implantable cardioverter defibrillator
	= intracranial hemorrhage
	= internal mammary artery
IE .	= Intravenous nPA for Treatment of
	Infarcting Myocardium Early
	= International Study of Infarct
	Survival
	= intravenous
	= Seventh Joint National Committee on
	High Blood Pressure
	= left anterior descending coronary
	artery
~	= left bundle-branch block
	= low-density lipoprotein cholesterol
	= Long-Term Intervention With
TT.	Pravastatin in Ischemic Disease
H	= low-molecular-weight heparin
	= left ventricular, left ventricle
	= left ventricular assist device
Т	= left ventricular ejection fraction
1	= Multicenter Automatic Denominator
C	- Magnagium in Coronarias trial
С Г	= Magnesium in Coronaries una
1	= Multicenter Difuazem Postimarcuon
	Irial motobalia aquivalant
	- metabolic equivalent
	- myocardiar illiarcuoli - mitral regurgitation
T	- Multicenter Unsuctained Techycordia
1	- municenter Unsustanteu Tachycafula Trial
FT	– North American Symptomatic
	Carotid Endarterectomy Trial
	- National Cholesterol Education
	Program
	1 IUgrain

= National Heart Attack Alert Program

NHLBI	= National Heart, Lung, and Blood	
NRMI	= National Registry of Myocardial	SPECT
NS	= not significant	STEMI
NSTEMI	= non–ST-elevation myocardial	TEE
	infarction	TIA
OR	= odds ratio	TIMI
PCI	= percutaneous coronary intervention	TIMI 0
PCWP	= pulmonary capillary wedge pressure	TIMI 1
PH	= parenchymal hemorrhage	
PRAGUE	= PRimary Angioplasty in patients transferred from General community	TIMI 2
	hospitals to specialized PTCA Units with or without Emergency throm-	TIMI 3
	bolysis	ТІС
Primary		fDA
care provider	= physician, nurse practitioner, physi-	TRACI
DDDD	cian assistant	
RBBB	= right bundle-branch block	
REACT	= Rapid Early Action for Coronary	VALIA
	Treatment	VALIA
RK	= relative risk	VE
RRR	= relative risk reduction	VF
RV	= right ventricular, right ventricle	VSR
SAVE	= Survival and Ventricular Enlargement	VT
SCD	= sudden cardiac death	WARIS
SHOCK	= SHould we emergently revascularize	WHI

	Occluded Coronaries for cardiogenic
	shocK?
Г	= single-photon emission computed
	tomography
I	= ST-elevation myocardial infarction
	= transesophageal echocardiography
	= transient ischemic attack
	= thrombolysis in myocardial infarction
)	= TIMI grade 0 flow, or no perfusion
1	= TIMI grade 1 flow, or penetration
	without perfusion
2	= TIMI grade 2 flow, or partial perfu-
	sion
3	= TIMI grade 3 flow, or complete per-
	fusion
	= Therapeutic Lifestyle Changes
	= tissue plasminogen activator
Е	Trandolapril Cardiac Evaluation
	= unstable angina
	= unfractionated heparin
NT	= Valsartan in Acute Myocardial
	Infarction Trial
	= ventricular fibrillation
	= ventricular septal rupture
	= ventricular tachycardia
S	= Warfarin-Aspirin Reinfarction Study
	= Women's Health Initiative

REFERENCES

- Gunnar RM, Passamani ER, Bourdillon PD, et al. Guidelines for the early management of patients with acute myocardial infarction: a report of the American College of Cardiology/American Heart Association Task Force on Assessment of Diagnostic and Therapeutic Cardiovascular Procedures (Subcommittee to Develop Guidelines for the Early Management of Patients with Acute Myocardial Infarction) J Am Coll Cardiol 1990;16:249-92.
- Ryan TJ, Anderson JL, Antman EM, et al. ACC/AHA guidelines for the management of patients with acute myocardial infarction: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on Management of Acute Myocardial Infarction). *J Am Coll Cardiol* 1996;28:1328-428.
- 3. Ryan TJ, Antman EM, Brooks NH, et al. 1999 update: ACC/AHA guidelines for the management of patients with acute myocardial infarction: a report of the American College of Cardiology/ American Heart Association Task Force on Practice Guidelines (Committee on Management of Acute Myocardial Infarction). J Am Coll Cardiol 1999;34:890-911.
- 4. Braunwald E, Antman E, Beasley J, et al. ACC/AHA 2002 guideline update for the management of patients with unstable angina and non–ST-segment elevation myocardial infarction: summary article: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on the Management of Patients With Unstable Angina). *J Am Coll Cardiol* 2002;40:1366.
- Mehta RH, Montoye CK, Gallogly M, et al, for the GAP Steering Committee of the American College of Cardiology. Improving quality of care for acute myocardial infarction: the Guidelines Applied in Practice (GAP) initiative. *JAMA* 2002;287:1269-76.
- Jencks SF, Huff ED, Cuerdon T. Change in the quality of care delivered to medicare beneficiaries, 1998-1999 to 2000-2001. *JAMA* 2003;289:305-12.
- Schneider EC, Zaslavsky AM, Epstein AM. Racial disparities in the quality of care for enrollees in medicare managed care. *JAMA* 2002;287:1288-94.
- 8. Libby P. Current concepts of the pathogenesis of the acute coronary syndromes. *Circulation* 2001;104:365-72.
- 9. Hamm CW, Bertrand M, Braunwald E. Acute coronary syndrome without ST elevation: implementation of new guidelines. *Lancet* 2001;358:1533-8.
- 10. Davies MJ. The pathophysiology of acute coronary syndromes. *Heart* 2000;83:361-6.
- 11. Falk E. Plaque rupture with severe pre-existing stenosis precipitating coronary thrombosis: characteristics of coronary atherosclerotic plaques underlying fatal occlusive thrombi. *Br Heart J* 1983;50:127-34.
- Little WC, Constantinescu M, Applegate RJ, et al. Can coronary angiography predict the site of a subsequent myocardial infarction in patients with mild-to-moderate coronary artery disease? *Circulation* 1988;78:1157-66.
- Malek AM, Alper SL, Izumo S. Hemodynamic shear stress and its role in atherosclerosis. *JAMA* 1999;282:2035-42.
- Rothwell PM, Villagra R, Gibson R, Donders RC, Warlow CP. Evidence of a chronic systemic cause of instability of atherosclerotic plaques. *Lancet* 2000;355:19-24.
- Libby P, Ridker PM, Maseri A. Inflammation and atherosclerosis. *Circulation* 2002;105:1135-43.
- 16. Dahlbäck B. Blood coagulation. Lancet 2000;355:1627-32.

- 17. Rosenberg RD, Aird WC. Vascular-bed: specific hemostasis and hypercoagulable states. *N Engl J Med* 1999;340:1555-64.
- Reimer KA, Lowe JE, Rasmussen MM, Jennings RB. The wavefront phenomenon of ischemic cell death: 1. Myocardial infarct size vs duration of coronary occlusion in dogs. *Circulation* 1977;56:786-94.
- 19. Hasche ET, Fernandes C, Freedman SB, Jeremy RW. Relation between ischemia time, infarct size, and left ventricular function in humans. *Circulation* 1995;92:710-9.
- DeWood MA, Spores J, Notske R, et al. Prevalence of total coronary occlusion during the early hours of transmural myocardial infarction. N Engl J Med 1980;303:897-902.
- 21. de Feyter PJ, van den Brand M, Serruys PW, Wijns W. Early angiography after myocardial infarction: what have we learned? *Am Heart J* 1985;109:194-9.
- 22. DeWood MA, Stifter WF, Simpson CS, et al. Coronary arteriographic findings soon after non–Q-wave myocardial infarction. *N Engl J Med* 1986;315:417-23.
- 23. Early effects of tissue-type plasminogen activator added to conventional therapy on the culprit coronary lesion in patients presenting with ischemic cardiac pain at rest: results of the Thrombolysis in Myocardial Ischemia (TIMI IIIA) trial. *Circulation* 1993;87:38-52.
- Boersma E, Mercado N, Poldermans D, Gardien M, Vos J, Simoons ML. Acute myocardial infarction. *Lancet* 2003;361:847-58.
- 25. The GUSTO investigators. An international randomized trial comparing four thrombolytic strategies for acute myocardial infarction. *N Engl J Med* 1993;329:673-82.
- 26. The Global Use of Strategies to Open Occluded Coronary Arteries (GUSTO III) Investigators. A comparison of reteplase with alteplase for acute myocardial infarction. N Engl J Med 1997;337:1118-23.
- 27. The Continuous Infusion versus Double-Bolus Administration of Alteplase (COBALT) Investigators. A comparison of continuous infusion of alteplase with double-bolus administration for acute myocardial infarction. *N Engl J Med* 1997;337:1124-30.
- Single-bolus tenecteplase compared with front-loaded alteplase in acute myocardial infarction: the ASSENT-2 double-blind randomised trial. Assessment of the Safety and Efficacy of a New Thrombolytic Investigators. *Lancet* 1999;354:716-22.
- 29. Intravenous NPA for the Treatment of Infarcting Myocardium Early; InTIME-II, a double-blind comparison of single-bolus lanoteplase vs accelerated alteplase for the treatment of patients with acute myocardial infarction. *Eur Heart J* 2000;21:2005-13.
- Topol EJ, for the GUSTO V Investigators. Reperfusion therapy for acute myocardial infarction with fibrinolytic therapy or combination reduced fibrinolytic therapy and platelet glycoprotein IIb/IIIa inhibition: the GUSTO V randomised trial. *Lancet* 2001; 357:1905-14.
- 31. Assessment of the Safety and Efficacy of a New Thrombolytic Regimen (ASSENT)-3 Investigators. Efficacy and safety of tenecteplase in combination with enoxaparin, abciximab, or unfractionated heparin: the ASSENT-3 randomised trial in acute myocardial infarction. *Lancet* 2001;358:605-13.
- 32. Neumann FJ, Blasini R, Schmitt C, et al. Effect of glycoprotein IIb/IIIa receptor blockade on recovery of coronary flow and left ventricular function after the placement of coronary-artery stents in acute myocardial infarction. *Circulation* 1998;98:2695-701.
- 33. White H, for the Hirulog and Early Reperfusion or Occlusion (HERO)-2 Trial Investigators. Thrombin-specific anticoagulation with bivalirudin versus heparin in patients receiving fibrinolytic

therapy for acute myocardial infarction: the HERO-2 randomised trial. *Lancet* 2001;358:1855-63.

- 34. Brener SJ, Barr LA, Burchenal JE, et al, for the ReoPro and Primary PTCA Organization and Randomized Trial (RAPPORT) Investigators. Randomized, placebo-controlled trial of platelet glycoprotein IIb/IIIa blockade with primary angioplasty for acute myocardial infarction. *Circulation* 1998;98:734-41.
- Neumann FJ, Kastrati A, Schmitt C, et al. Effect of glycoprotein IIb/IIIa receptor blockade with abciximab on clinical and angiographic restenosis rate after the placement of coronary stents following acute myocardial infarction. J Am Coll Cardiol 2000;35:915-21.
- 36. Montalescot G, Barragan P, Wittenberg O, et al, for the Abciximab before Direct Angioplasty and Stenting in Myocardial Infarction Regarding Acute and Long-Term Follow-up (ADMIRAL) Investigators. Platelet glycoprotein IIb/IIIa inhibition with coronary stenting for acute myocardial infarction. *N Engl J Med* 2001;344:1895-903.
- Zhu MM, Feit A, Chadow H, Alam M, Kwan T, Clark LT. Primary stent implantation compared with primary balloon angioplasty for acute myocardial infarction: a meta-analysis of randomized clinical trials. *Am J Cardiol* 2001;88:297-301.
- 38. Stone GW, Grines CL, Cox DA, et al, for the Controlled Abciximab and Device Investigation to Lower Late Angioplasty Complications (CADILLAC) Investigators. Comparison of angioplasty with stenting, with or without abciximab, in acute myocardial infarction. N Engl J Med 2002;346:957-66.
- 39. Direct thrombin inhibitors in acute coronary syndromes: principal results of a meta-analysis based on individual patients' data. *Lancet* 2002;359:294-302.
- Keeley EC, Boura JA, Grines CL. Primary angioplasty versus intravenous thrombolytic therapy for acute myocardial infarction: a quantitative review of 23 randomised trials. *Lancet* 2003; 361:13-20.
- 41. Braunwald E, Pfeffer MA. Ventricular enlargement and remodeling following acute myocardial infarction: mechanisms and management. *Am J Cardiol* 1991;68:1D-6D.
- 42. Pfeffer MA. Left ventricular remodeling after acute myocardial infarction. *Annu Rev Med* 1995;46:455-66.
- 43. Weisman HF, Healy B. Myocardial infarct expansion, infarct extension, and reinfarction: pathophysiologic concepts. *Prog Cardiovasc Dis* 1987;30:73-110.
- 44. Weber KT. Aldosterone in congestive heart failure. *N Engl J Med* 2001;345:1689-97.
- 45. Rogers WJ, Canto JG, Lambrew CT, et al. Temporal trends in the treatment of over 1.5 million patients with myocardial infarction in the US from 1990 through 1999: the National Registry of Myocardial Infarction 1, 2 and 3. *J Am Coll Cardiol* 2000;36: 2056-63.
- American Heart Association. Heart Disease and Stroke Statistics — 2004 Update, Dallas, TX: American Heart Association; 2003. Available at http://www.americanheart.org/presenter.jhtml?iden-tifier=3000090. Accessed November 15, 2003.
- 46a. Wiviott SD, Morrow DA, Giugliano RP, et al. Performance of the thrombolysis in myocardial infarction risk index for early acute coronary syndrome in the National Registry of Myocardial Infarction: a simple risk index predicts mortality in both ST and non-ST elevation myocardial infarction. *J Am Coll Cardiol* 2003;41:365A.
- Guidry UC, Evans JC, Larson MG, Wilson PW, Murabito JM, Levy D. Temporal trends in event rates after Q-wave myocardial infarction: the Framingham Heart Study. *Circulation* 1999;100:

2054-9.

- 48. Goldberg RJ, Yarzebski J, Lessard D, Gore JM. A two-decades (1975 to 1995) long experience in the incidence, in-hospital and long-term case-fatality rates of acute myocardial infarction: a community-wide perspective. J Am Coll Cardiol 1999;33:1533-9.
- 49. Rouleau JL, Talajic M, Sussex B, et al. Myocardial infarction patients in the 1990s: their risk factors, stratification and survival in Canada: the Canadian Assessment of Myocardial Infarction (CAMI) Study. *J Am Coll Cardiol* 1996;27:1119-27.
- Executive Summary of The Third Report of The National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). JAMA 2001;285:2486-97.
- National Kidney Foundation practice guidelines for chronic kidney disease: evaluation, classification, and stratification. Ann Intern Med 2003;139:137-47.
- 52. National Heart, Lung, and Blood Institute. Act in Time to Heart Attack. Available at http://www.nhlbi.nih.gov/actintime, 2003. Accessed May 15, 2003.
- 53. Wenger NK. You've come a long way, baby: cardiovascular health and disease in women: problems and prospects. *Circulation* 2004;109:558-60.
- American College of Cardiology. Guidelines Applied in Practice. Available at http://www.acc.org/gap/gap.htm, 2003. Accessed May 15, 2003.
- 55. American Heart Association. Get With the Guidelines. Available at http://www.americanheart.org/presenter.jhtml?identifier=1165, 2003. Accessed May 15, 2003.
- 56. Topol EJ, Kereiakes DJ. Regionalization of care for acute ischemic heart disease: a call for specialized centers. *Circulation* 2003;107:1463-6.
- 57. Califf RM, Faxon DP. Need for centers to care for patients with acute coronary syndromes. *Circulation* 2003;107:1467-70.
- 58. Willerson JT. Editor's commentary: centers of excellence. *Circulation* 2003;107:1471-2.
- 59. National Cholesterol Education Program. Third Report of the Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). NIH Pub. No. 02-5125. Bethesda, MD: National Heart, Lung, and Blood Institute, 2002;284 pages. Guidelines, Related Tools, and Patient Information available at http://www.nhlbi.nih.gov/guidelines/cholesterol/index.htm. Accessed May 15, 2003.
- Wielgosz AT. What do we really know about secondary prevention after myocardial infarction? *Can J Cardiol* 1995;11(Suppl A):31A-32A.
- 61. Ebrahim S, Davey Smith G. Multiple risk factor interventions for primary prevention of coronary heart disease. Cochrane Database Syst Rev 2000CD001561.
- Ross SD, Allen IE, Connelly JE, et al. Clinical outcomes in statin treatment trials: a meta-analysis. *Arch Intern Med* 1999;159: 1793-802.
- 63. Grundy SM, Pasternak R, Greenland P, Smith S, Fuster V. Assessment of cardiovascular risk by use of multiple-risk-factor assessment equations: a statement for healthcare professionals from the American Heart Association and the American College of Cardiology. *Circulation* 1999;100:1481-92.
- 64. Risk Assessment. American Heart Association. Available at: http://www.americanheart.org/presenter.jhtml?identifier=3003499, 2004. Accessed May 15, 2003.
- 65. Pearson TA, Blair SN, Daniels SR, et al. AHA Guidelines for primary prevention of cardiovascular disease and stroke: 2002 update: consensus panel guide to comprehensive risk reduction

for adult patients without coronary or other atherosclerotic vascular diseases. American Heart Association Science Advisory and Coordinating Committee. *Circulation* 2002;106:388-91.

- 66. Wilson K, Gibson N, Willan A, Cook D. Effect of smoking cessation on mortality after myocardial infarction: meta-analysis of cohort studies. *Arch Intern Med* 2000;160:939-44.
- 67. Freemantle N, Cleland J, Young P, Mason J, Harrison J. Beta blockade after myocardial infarction: systematic review and meta regression analysis. *BMJ* 1999;318:1730-7.
- 68. Smith SC, Blair SN, Bonow RO, et al. AHA/ACC scientific statement: AHA/ACC guidelines for preventing heart attack and death in patients with atherosclerotic cardiovascular disease: 2001 update: a statement for healthcare professionals from the American Heart Association and the American College of Cardiology. *Circulation* 2001;104:1577-9.
- Mosca L, Appel LJ, Benjamin EJ, et al. Evidence-based guidelines for cardiovascular disease prevention in women. *Circulation* 2004;109:672-93.
- Treating tobacco use and dependence: a clinical practice guideline. Office of the Surgeon General. Available at http://www.surgeongeneral.gov/tobacco, 2000. Accessed May 15, 2003.
- 71. Gibbons RJ, Abrams J, Chatterjee K, et al. ACC/AHA 2002 guideline update for the management of patients with chronic stable angina: a report of the American College of Cardiology/ American Heart Association Task Force on Practice Guidelines (Committee to Update the 1999 Guidelines for the Management of Patients with Chronic Stable Angina). Available at www.acc.org/clinical/guidelines/stable/stable.pdf, 2002. Accessed December 2, 2003.
- 72. Chobanian AV, Bakris GL, Black HR, et al, for the National Heart, Lung, and Blood Institute Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure, National High Blood Pressure Education Program Coordinating Committee. The seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure: the JNC 7 report. *JAMA* 2003;289:2560-72.
- 73. Pignone M, Mulrow CD. Evidence based management of hypertension: Using cardiovascular risk profiles to individualise hypertensive treatment. *BMJ* 2001;322:1164-6.
- Turnbull F. Blood Pressure Lowering Treatment Trialists' Collaboration. Effects of different blood-pressure-lowering regimens on major cardiovascular events: results of prospectivelydesigned overviews of randomised trials. *Lancet* 2003;362:1527-35.
- 75. SHEP Cooperative Research Group. Prevention of stroke by antihypertensive drug treatment in older persons with isolated systolic hypertension: final results of the Systolic Hypertension in the Elderly Program (SHEP). *JAMA* 1991;265:3255-64.
- Hayden M, Pignone M, Phillips C, Mulrow C. Aspirin for the primary prevention of cardiovascular events: a summary of the evidence for the U.S. Preventive Services Task Force. *Ann Intern Med* 2002;136:161-72.
- U.S. Preventive Services Task Force. Aspirin for the primary prevention of cardiovascular events: recommendation and rationale. *Ann Intern Med* 2002;136:157-60.
- 78. Peters RJ, Mehta SR, Fox KA, et al, for the Clopidogrel in Unstable angina to prevent Recurrent Events (CURE) Trial Investigators. Effects of aspirin dose when used alone or in combination with clopidogrel in patients with acute coronary syndromes: observations from the Clopidogrel in Unstable angina to prevent Recurrent Events (CURE) study. *Circulation* 2003;

108:1682-7.

- 79. Hung J. Medical Issues Committee of the National Heart Foundation of Australia. Aspirin for cardiovascular disease prevention. *Med J Aust* 2003;179:147-52.
- 80. Dracup K, Alonzo AA, Atkins JM, et al. The physician's role in minimizing prehospital delay in patients at high risk for acute myocardial infarction: recommendations from the National Heart Attack Alert Program (Working Group on Educational Strategies to Prevent Prehospital Delay in Patients at High Risk for Acute Myocardial Infarction). Ann Intern Med 1997;126:645-51.
- Hedges JR, Feldman HA, Bittner V, et al, for the REACT Study Group. Impact of community intervention to reduce patient delay time on use of reperfusion therapy for acute myocardial infarction: Rapid Early Action for Coronary Treatment (REACT) trial. *Acad Emerg Med* 2000;7:862-72.
- 82. Canto JG, Zalenski RJ, Ornato JP, et al, for the National Registry of Myocardial Infarction 2 Investigators. Use of emergency medical services in acute myocardial infarction and subsequent quality of care: observations from the National Registry of Myocardial Infarction 2. *Circulation* 2002;106:3018-23.
- Goldberg R, Goff D, Cooper L, et al. Age and sex differences in presentation of symptoms among patients with acute coronary disease: the REACT Trial. Rapid Early Action for Coronary Treatment. *Coron Artery Dis* 2000;11:399-407.
- 84. Hutchings CB, Mann NC, Daya M, et al. Rapid Early Action for Coronary Treatment Study. Patients with chest pain calling 9-1-1 or self-transporting to reach definitive care: which mode is quicker? *Am Heart J* 2004;147:35-41.
- 85. Becker L, Larsen MP, Eisenberg MS. Incidence of cardiac arrest during self-transport for chest pain. *Ann Emerg Med* 1996;28: 612-6.
- 86. Brown AL, Mann NC, Daya M, et al. Demographic, belief, and situational factors influencing the decision to utilize emergency medical services among chest pain patients: Rapid Early Action for Coronary Treatment (REACT) study. *Circulation* 2000;102: 173-8.
- Goff DC, Feldman HA, McGovern PG, et al, for the Rapid Early Action for Coronary Treatment (REACT) Study Group. Prehospital delay in patients hospitalized with heart attack symptoms in the United States: the REACT trial. *Am Heart J* 1999;138: 1046-57.
- Herlitz J, Karlson BW, Liljeqvist JA, Strömbom U, Holmberg S. Early identification of acute myocardial infarction and prognosis in relation to mode of transport to hospital. *Am J Emerg Med* 1992;10:406-12.
- Ho MT, Eisenberg MS, Litwin PE, Schaeffer SM, Damon SK. Delay between onset of chest pain and seeking medical care: the effect of public education. *Ann Emerg Med* 1989;18:727-31.
- Dracup K, Moser DK, Eisenberg M, Meischke H, Alonzo AA, Braslow A. Causes of delay in seeking treatment for heart attack symptoms. *Soc Sci Med* 1995;40:379-92.
- 91. Herlitz J, Blohm M, Hartford M, et al. Follow-up of a 1-year media campaign on delay times and ambulance use in suspected acute myocardial infarction. *Eur Heart J* 1992;13:171-7.
- 92. Luepker RV, Raczynski JM, Osganian S, et al. Effect of a community intervention on patient delay and emergency medical service use in acute coronary heart disease: the Rapid Early Action for Coronary Treatment (REACT) Trial. JAMA 2000;284:60-7.
- 93. Wright RS, Kopecky SL, Timm M, et al, for the Wabasha Heart Attack Team. Impact of community-based education on health care evaluation in patients with acute chest pain syndromes: the

Wabasha Heart Attack Team (WHAT) project. *Fam Pract* 2001; 18:537-9.

- 94. Access to Timely and Optimal Care of Patients with Acute Coronary Syndromes - Community Planning Considerations: a report by the National Heart Attack Alert Program. J Thromb Thrombolysis 1998;6:19-46.
- 95. Leslie WS, Urie A, Hooper J, Morrison CE. Delay in calling for help during myocardial infarction: reasons for the delay and subsequent pattern of accessing care. *Heart* 2000;84:137-41.
- Simon AB, Feinleib M, Thompson HK. Components of delay in the pre-hospital phase of acute myocardial infarction. *Am J Cardiol* 1972;30:476-82.
- 97. Alonzo AA. The impact of the family and lay others on care-seeking during life-threatening episodes of suspected coronary artery disease. *Soc Sci Med* 1986;22:1297-311.
- 98. Guidelines 2000 for cardiopulmonary resuscitation and emergency cardiovascular care: part 12: from science to survival: strengthening the chain of survival in every community. The American Heart Association in collaboration with the International Liaison Committee on Resuscitation. *Circulation* 2000;102:1358-70.
- McDermott MM, Mandapat AL, Moates A, et al. Knowledge and attitudes regarding cardiovascular disease risk and prevention in patients with coronary or peripheral arterial disease. *Arch Intern Med* 2003;163:2157-62.
- 100. Canto JG, Shlipak MG, Rogers WJ, et al. Prevalence, clinical characteristics, and mortality among patients with myocardial infarction presenting without chest pain. *JAMA* 2000;283:3223-9.
- 101. Faxon D, Lenfant C. Timing is everything: motivating patients to call 9-1-1 at onset of acute myocardial infarction. *Circulation* 2001;104:1210-1.
- 102. Guidelines 2000 for cardiopulmonary resuscitation and emergency cardiovascular care: part 7: the era of reperfusion: section 1: acute coronary syndromes (acute myocardial infarction). The American Heart Association in collaboration with the International Liaison Committee on Resuscitation. *Circulation* 2000;102:1172-203.
- 103. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3525. September 2001. Available at: http://www.nhlbi.nih.gov/health/public/heart/mi/ core_bk.pdf. Accessed December 16, 2002.
- 104. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3526. November 2001. Available at: http://www.nhlbi.nih.gov/health/public/heart/mi/ core_sp.pdf. Accessed December 16, 2002.
- 105. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3667. September 2001. Available at: http://www.nhlbi.nih.gov/health/public/heart/mi/ wallet.pdf. Accessed December 16, 2002.
- 106. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3669. September 2001. Available at: http://www.nhlbi.nih.gov/health/public/heart/mi/ act_ plan.pdf. Accessed December 16, 2002.
- 107. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3313. September 2001. Available at: http://www.nhlbi.nih.gov/health/prof/heart/mi/ provider.pdf. Accessed December 16, 2002.

- 108. Act in Time to Heart Attack Signs: Physician Quick Reference Tool for Palm OS. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. 2001. Available at: http://hin.nhlbi.nih.gov/haac_palm/haac_palm.htm. Accessed Decmeber 16, 2002.
- 109. U.S. Department of Health and Human Services. Public Health Service. National Institutes of Health. National Heart, Lung, and Blood Institute. NIH Publication No. 01-3646. September 2001. Available at: http://www.nhlbi.nih.gov/health/public/heart/mi/ poster.pdf. Accessed Decmeber 16, 2002.
- 110. Deleted in press.
- 111. Goff DC, Sellers DE, McGovern PG, et al. Knowledge of heart attack symptoms in a population survey in the United States: the REACT Trial. Rapid Early Action for Coronary Treatment. Arch Intern Med 1998;158:2329-38.
- 112. Welsh RC, Ornato J, Armstrong PW. Prehospital management of acute ST-elevation myocardial infarction: a time for reappraisal in North America. *Am Heart J* 2003;145:1-8.
- 113. Goldberg RJ, Steg PG, Sadiq I, et al. Extent of, and factors associated with, delay to hospital presentation in patients with acute coronary disease (the GRACE registry). *Am J Cardiol* 2002; 89:791-6.
- 114. Finnegan JR, Meischke H, Zapka JG, et al. Patient delay in seeking care for heart attack symptoms: findings from focus groups conducted in five U.S. regions. *Prev Med* 2000;31:205-13.
- 115. Feldman HA, Proschan MA, Murray DM, et al. Statistical design of REACT (Rapid Early Action for Coronary Treatment), a multisite community trial with continual data collection. *Control Clin Trials* 1998;19:391-403.
- 116. McKinley S, Moser DK, Dracup K. Treatment-seeking behavior for acute myocardial infarction symptoms in North America and Australia. *Heart Lung* 2000;29:237-47.
- 117. Rucker D, Brennan T, Burstin H. Delay in seeking emergency care. *Acad Emerg Med* 2001;8:163-9.
- 118. Sheifer SE, Gersh BJ, Yanez ND, Ades PA, Burke GL, Manolio TA. Prevalence, predisposing factors, and prognosis of clinically unrecognized myocardial infarction in the elderly. *J Am Coll Cardiol* 2000;35:119-26.
- 119. Kannel WB. Silent myocardial ischemia and infarction: insights from the Framingham Study. *Cardiol Clin* 1986;4:583-91.
- 120. Rathore SS, Weinfurt KP, Gersh BJ, Oetgen WJ, Schulman KA, Solomon AJ. Treatment of patients with myocardial infarction who present with a paced rhythm. *Ann Intern Med* 2001;134:644-51.
- 121. Cummins RO, Ornato JP, Thies WH, Pepe PE. Improving survival from sudden cardiac arrest: the "chain of survival" concept: a statement for health professionals from the Advanced Cardiac Life Support Subcommittee and the Emergency Cardiac Care Committee, American Heart Association. *Circulation* 1991;83: 1832-47.
- 122. Cummins RO, Eisenberg MS, Hallstrom AP, Litwin PE. Survival of out-of-hospital cardiac arrest with early initiation of cardiopulmonary resuscitation. *Am J Emerg Med* 1985;3:114-9.
- 123. Weisfeldt ML, Becker LB. Resuscitation after cardiac arrest: a 3phase time-sensitive model. *JAMA* 2002;288:3035-8.
- 124. Stapczynski JS, Svenson JE, Stone CK. Population density, automated external defibrillator use, and survival in rural cardiac arrest. Acad Emerg Med 1997;4:552-8.
- 125. Becker LB, Ostrander MP, Barrett J, Kondos GT. Outcome of CPR in a large metropolitan area: where are the survivors? *Ann Emerg Med* 1991;20:355-61.

- 126. Lombardi G, Gallagher J, Gennis P. Outcome of out-of-hospital cardiac arrest in New York City: the Pre-Hospital Arrest Survival Evaluation (PHASE) Study. *JAMA* 1994;271:678-83.
- 127. White RD, Asplin BR, Bugliosi TF, Hankins DG. High discharge survival rate after out-of-hospital ventricular fibrillation with rapid defibrillation by police and paramedics. *Ann Emerg Med* 1996;28:480-5.
- 128. White RD, Hankins DG, Bugliosi TF. Seven years' experience with early defibrillation by police and paramedics in an emergency medical services system. *Resuscitation* 1998;39:145-51.
- 129. Bunch TJ, White RD, Gersh BJ, et al. Long-term outcomes of outof-hospital cardiac arrest after successful early defibrillation. N Engl J Med 2003;348:2626-33.
- 130. Stiell IG, Wells GA, Field BJ, et al. Improved out-of-hospital cardiac arrest survival through the inexpensive optimization of an existing defibrillation program: OPALS study phase II. Ontario Prehospital Advanced Life Support. JAMA 1999;281:1175-81.
- Caffrey SL, Willoughby PJ, Pepe PE, Becker LB. Public use of automated external defibrillators. *N Engl J Med* 2002;347:1242-7.
- 132. Dracup K, Moser DK, Guzy PM, Taylor SE, Marsden C. Is cardiopulmonary resuscitation training deleterious for family members of cardiac patients? *Am J Public Health* 1994;84:116-8.
- 133. Dracup K, Moser DK, Taylor SE, Guzy PM. The psychological consequences of cardiopulmonary resuscitation training for family members of patients at risk for sudden death. *Am J Public Health* 1997;87:1434-9.
- 134. Chen MA, Eisenberg MS, Meischke H. Impact of in-home defibrillators on postmyocardial infarction patients and their significant others: an interview study. *Heart Lung* 2002;31:173-85.
- 135. Home Automatic External Defibrillator Trial. National Heart, Lung, and Blood Institute (NHLBI). Available at: http:// www.clinicaltrials.gov/ct/show/NCT00047411?order=46. 2004. NLM Identifier: NCT00047411. Accessed December 22, 2003.
- 136. Hallstrom AP, Cobb LA, Johnson E, Copass MK. Dispatcher assisted CPR: implementation and potential benefit: a 12-year study. *Resuscitation* 2003;57:123-9.
- 137. Cummins RO. From concept to standard-of-care? Review of the clinical experience with automated external defibrillators. *Ann Emerg Med* 1989;18:1269-75.
- 138. Cummins RO, Schubach JA, Litwin PE, Hearne TR. Training lay persons to use automatic external defibrillators: success of initial training and one-year retention of skills. *Am J Emerg Med* 1989;7:143-9.
- Mosesso VN, Davis EA, Auble TE, Paris PM, Yealy DM. Use of automated external defibrillators by police officers for treatment of out-of-hospital cardiac arrest. Ann Emerg Med 1998;32:200-7.
- 140. Valenzuela TD, Roe DJ, Nichol G, Clark LL, Spaite DW, Hardman RG. Outcomes of rapid defibrillation by security officers after cardiac arrest in casinos. *N Engl J Med* 2000;343:1206-9.
- 141. Page RL, Joglar JA, Kowal RC, et al. Use of automated external defibrillators by a U.S. airline. N Engl J Med 2000;343:1210-6.
- 142. Nichol G, Hallstrom AP, Kerber R, et al. American Heart Association report on the second public access defibrillation conference, April 17-19, 1997. *Circulation* 1998;97:1309-14.
- 143. Nichol G, Hallstrom AP, Ornato JP, et al. Potential cost-effectiveness of public access defibrillation in the United States. *Circulation* 1998;97:1315-20.
- 144. Kerber RE, Becker LB, Bourland JD, et al. Automatic external defibrillators for public access defibrillation: recommendations for specifying and reporting arrhythmia analysis algorithm per-

formance, incorporating new waveforms, and enhancing safety: a statement for health professionals from the American Heart Association Task Force on Automatic External Defibrillation, Subcommittee on AED Safety and Efficacy. *Circulation* 1997;95:1677-82.

- 145. Weisfeldt ML, Kerber RE, McGoldrick RP, et al, for the Automatic Defibrillation Task Force. Public access to defibrillation. *Am J Emerg Med* 1996;14:684-92.
- 146. Weisfeldt ML, Kerber RE, McGoldrick RP, et al, for the Automatic External Defibrillation Task Force. American Heart Association Report on the Public Access Defibrillation Conference December 8-10, 1994. *Circulation* 1995;92:2740-7.
- 147. Weisfeldt ML, Kerber RE, McGoldrick RP, et al. Public access defibrillation: a statement for healthcare professionals from the American Heart Association Task Force on Automatic External Defibrillation. *Circulation* 1995;92:2763.
- 148. Deleted in press.
- 149. Ornato JP, Racht EM, Fitch JJ, Berry JF. The need for ALS in urban and suburban EMS systems. Ann Emerg Med 1990; 19:1469-70.
- 150. Stout J, Pepe PE, Mosesso VN. All-advanced life support vs tiered-response ambulance systems. *Prehosp Emerg Care* 2000;4:1-6.
- 151. Eisenberg MJ, Topol EJ. Prehospital administration of aspirin in patients with unstable angina and acute myocardial infarction. *Arch Intern Med* 1996;156:1506-10.
- **152. ISIS-**4 (Fourth International Study of Infarct Survival) Collaborative Group. ISIS-4: a randomised factorial trial assessing early oral captopril, oral mononitrate, and intravenous magnesium sulphate in 58,050 patients with suspected acute myocardial infarction. *Lancet* 1995;345:669-85.
- 153. Hutter AM, Weaver WD. 31st Bethesda Conference: emergency cardiac care: task force 2: acute coronary syndromes: section 2A–prehospital issues. *J Am Coll Cardiol* 2000;35:846-53.
- 154. Lau J, Ioannidis J, Balk E, et al. Evaluation of Technologies for Identifying Acute Cardiac Ischemia in Emergency Departments. Evidence Report/Technology Assessment Number 26. (Prepared by the New England Medical Center Evidence-based Practice Center under Contract No. 290-97-0019). AHRQ Publication No. 01-E006, Rockville, MD: Agency for Healthcare Research and Quality, May 2001.
- 155. Armstrong PW, Collen D, Antman E. Fibrinolysis for acute myocardial infarction: the future is here and now. *Circulation* 2003;107:2533-7.
- 156. Fibrinolytic Therapy Trialists' (FTT) Collaborative Group. Indications for fibrinolytic therapy in suspected acute myocardial infarction: collaborative overview of early mortality and major morbidity results from all randomised trials of more than 1000 patients. *Lancet* 1994;343:311-22.
- 157. Gruppo Italiano per lo Studio della Streptochinasi nell'Infarto Miocardico (GISSI). Effectiveness of intravenous thrombolytic treatment in acute myocardial infarction. *Lancet* 1986;1:397-402.
- 158. Wallentin L, Goldstein P, Armstrong PW, et al. Efficacy and safety of tenecteplase in combination with the low-molecular-weight heparin enoxaparin or unfractionated heparin in the prehospital setting: the Assessment of the Safety and Efficacy of a New Thrombolytic Regimen (ASSENT)-3 PLUS randomized trial in acute myocardial infarction. *Circulation* 2003;108:135-42.
- 159. Weaver WD, Cerqueira M, Hallstrom AP, et al. Prehospital-initiated vs hospital-initiated thrombolytic therapy: the Myocardial Infarction Triage and Intervention trial. *JAMA* 1993;270:1211-6.
- 160. Hermens WT, Willems GM, Nijssen KM, Simoons ML. Effect of

ACC - www.acc.org

thrombolytic treatment delay on myocardial infarct size. *Lancet* 1992;340:1297.

- 161. Gersh BJ, Anderson JL. Thrombolysis and myocardial salvage: results of clinical trials and the animal paradigm: paradoxic or predictable? *Circulation* 1993;88:296-306.
- 162. Newby LK, Rutsch WR, Califf RM, et al, for the GUSTO-1 Investigators. Time from symptom onset to treatment and outcomes after thrombolytic therapy. J Am Coll Cardiol 1996;27:1646-55.
- 163. Milavetz JJ, Giebel DW, Christian TF, Schwartz RS, Holmes DR, Gibbons RJ. Time to therapy and salvage in myocardial infarction. J Am Coll Cardiol 1998;31:1246-51.
- 164. Chareonthaitawee P, Gibbons RJ, Roberts RS, Christian TF, Burns R, Yusuf S, for the CORE investigators (Collaborative Organisation for RheothRx Evaluation). The impact of time to thrombolytic treatment on outcome in patients with acute myocardial infarction. *Heart* 2000;84:142-8.
- 165. McNeill AJ, Cunningham SR, Flannery DJ, et al. A double-blind placebo controlled study of early and late administration of recombinant tissue plasminogen activator in acute myocardial infarction. *Br Heart J* 1989;61:316-21.
- 166. Castaigne AD, Hervé C, Duval-Moulin AM, et al. Prehospital use of APSAC: results of a placebo-controlled study. *Am J Cardiol* 1989;64:30A-33A; discussion.
- 167. Barbash GI, Roth A, Hod H, et al. Improved survival but not left ventricular function with early and prehospital treatment with tissue plasminogen activator in acute myocardial infarction. *Am J Cardiol* 1990;66:261-6.
- 168. Schofer J, Büttner J, Geng G, et al Prehospital thrombolysis in acute myocardial infarction. *Am J Cardiol* 1990;66:1429-33.
- 169. McAleer B, Ruane B, Burke E, et al. Prehospital thrombolysis in a rural community: short- and long-term survival. *Cardiovasc Drugs Ther* 1992;6:369-72.
- 170. Grampian Region Early Anistreplase Trial (GREAT) Group. Feasibility, safety, and efficacy of domiciliary thrombolysis by general practitioners: *BMJ* 1992;305:548-53.
- 171. The European Myocardial Infarction Project Group. Prehospital thrombolytic therapy in patients with suspected acute myocardial infarction. *N Engl J Med* 1993;329:383-9.
- 172. Morrow DA, Antman EM, Sayah A, et al. Evaluation of the time saved by prehospital initiation of reteplase for ST-elevation myocardial infarction: results of The Early Retavase-Thrombolysis in Myocardial Infarction (ER-TIMI) 19 trial. *J Am Coll Cardiol* 2002;40:71-7.
- 173. Bonnefoy E, Lapostolle F, Leizorovicz A, et al for the Comparison of Angioplasty and Prehospital Thrombolysis in Acute Myocardial Infarction Study Group. Primary angioplasty versus prehospital fibrinolysis in acute myocardial infarction: a randomised study. *Lancet* 2002;360:825-9.
- 174. Spinler SA, Mikhail PA. Prehospital-initiated thrombolysis. *Ann Pharmacother* 1997;31:1339-46.
- 175. Karagounis L, Ipsen SK, Jessop MR, et al. Impact of field-transmitted electrocardiography on time to in-hospital thrombolytic therapy in acute myocardial infarction. *Am J Cardiol* 1990;66:786-91.
- 176. Steg PG, Bonnefoy E, Chabaud S, et al. Impact of Time to Treatment on Mortality After Prehospital Fibrinolysis or Primary Angioplasty: data from the CAPTIM randomized clinical trial. *Circulation* 2003;108:2851-6.
- 177. Widimský P, Budesinský T, Vorác D, et al. Long distance transport for primary angioplasty vs immediate thrombolysis in acute myocardial infarction: final results of the randomized national

multicentre trial-PRAGUE-2. Eur Heart J 2003;24:94-104.

- 178. Deleted in press.
- 179. Emergency department: rapid identification and treatment of patients with acute myocardial infarction. National Heart Attack Alert Program Coordinating Committee, 60 Minutes to Treatment Working Group. *Ann Emerg Med* 1994;23:311-29.
- 180. Cannon CP, Antman EM, Walls R, Braunwald E. Time as an adjunctive agent to thrombolytic therapy. *J Thromb Thrombolysis* 1994;1:27-34.
- 181. Weaver WD, Litwin PE, Martin JS, et al, for the MITI Project Group. Effect of age on use of thrombolytic therapy and mortality in acute myocardial infarction. *J Am Coll Cardiol* 1991;18:657-62.
- 182. Gibler WB, Kereiakes DJ, Dean EN, et al. Prehospital diagnosis and treatment of acute myocardial infarction: a north-south perspective. The Cincinnati Heart Project and the Nashville Prehospital TPA Trial. Am Heart J 1991;121:1-11.
- 183. Pedley DK, Bissett K, Connolly EM, et al. Prospective observational cohort study of time saved by prehospital thrombolysis for ST elevation myocardial infarction delivered by paramedics. *BMJ* 2003;327:22-6.
- 184. Hochman JS, Sleeper LA, White HD, et al, for the Should We Emergently Revascularize Occluded Coronaries for Cardiogenic Shock (SHOCK) Investigators. One-year survival following early revascularization for cardiogenic shock. JAMA 2001;285:190-2.
- 185. Cannon CP. Time to treatment: a crucial factor in thrombolysis and primary angioplasty. *J Thromb Thrombolysis* 1996;3:249-255.
- 186. Steg PG, Goldberg RJ, Gore JM, et al, for the GRACE Investigators. Baseline characteristics, management practices, and in-hospital outcomes of patients hospitalized with acute coronary syndromes in the Global Registry of Acute Coronary Events (GRACE). *Am J Cardiol* 2002;90:358-63.
- 187. Storrow AB, Gibler WB. Chest pain centers: diagnosis of acute coronary syndromes. *Ann Emerg Med* 2000;35:449-61.
- 188. Pope JH, Aufderheide TP, Ruthazer R, et al. Missed diagnoses of acute cardiac ischemia in the emergency department. N Engl J Med 2000;342:1163-70.
- Jesse RL, Kontos MC. Evaluation of chest pain in the emergency department. *Curr Probl Cardiol* 1997;22:149-236.
- 190. Karcz A, Korn R, Burke MC, et al. Malpractice claims against emergency physicians in Massachusetts: 1975-1993. *Am J Emerg Med* 1996;14:341-5.
- 191. Graff L. Missed MI diagnosis. Ann Emerg Med 1994;23:141-2.
- 192. McCarthy BD, Beshansky JR, D'Agostino RB, Selker HP. Missed diagnoses of acute myocardial infarction in the emergency department: results from a multicenter study. *Ann Emerg Med* 1993;22:579-82.
- 193. Tatum JL, Jesse RL, Kontos MC, et al. Comprehensive strategy for the evaluation and triage of the chest pain patient. *Ann Emerg Med* 1997;29:116-25.
- 194. Tatum JL. Cost effective nuclear scanning in a comprehensive and systematic approach to the evaluation of chest pain in the emergency department. *Md Med J* 1997;Suppl:25-9.
- 195. Ornato JP. Chest pain emergency centers: improving acute myocardial infarction care. *Clin Cardiol* 1999;22:IV3-9.
- 196. Newby LK, Storrow AB, Gibler WB, et al. Bedside multimarker testing for risk stratification in chest pain units: the chest pain evaluation by creatine kinase-MB, myoglobin, and troponin I (CHECKMATE) study. *Circulation* 2001;103:1832-7.
- 197. Lateef F, Storrow AB, Gibler BW, Liu T. Heart emergency room: effective for both geriatric and younger patients. *Singapore Med J*

2001;42:259-63.

- 198. Lateef F, Storrow AB, Malone K, Liu T, Gibler BW. Comparison of a 6-hour and 9-hour protocol for evaluation of moderate-to-low risk chest pain patients in an emergency department diagnostic unit. *Singapore Med J* 2001;42:052-6.
- 199. Gibler WB. Chest pain units: do they make sense now? Ann Emerg Med 1997;29:168-71.
- 200. Gibler WB. Evaluation of chest pain in the emergency department. *Ann Intern Med* 1995;123:315; discussion 317.
- 201. Gibler WB. Chest pain evaluation in the ED: beyond triage. *Am J Emerg Med* 1994;12:121-2.
- 202. Gibler WB. Evaluating patients with chest pain in the ED: improving speed, efficiency, and cost-effectiveness, or teaching an old dog new tricks. *Ann Emerg Med* 1994;23:381-2.
- 203. Hoekstra JW, Gibler WB, Levy RC, et al. Emergency-department diagnosis of acute myocardial infarction and ischemia: a cost analysis of two diagnostic protocols. *Acad Emerg Med* 1994;1:103-10.
- 204. Hoekstra JW, Hedges JR, Gibler WB, Rubison RM, Christensen RA, for the National Cooperative CK-MB Project Group. Emergency department CK-MB: a predictor of ischemic complications. Acad Emerg Med 1994;1:17-27.
- 205. Cannon CP, Hand MH, Bahr R, et al, for the. National Heart Attack Alert Program (NHAAP) Coordinating Committee Critical Pathways Writing Group. Critical pathways for management of patients with acute coronary syndromes: an assessment by the National Heart Attack Alert Program. *Am Heart J* 2002;143:777-89.
- 206. Zalenski RJ, Selker HP, Cannon CP, et al. National Heart Attack Alert Program position paper: chest pain centers and programs for the evaluation of acute cardiac ischemia. *Ann Emerg Med* 2000;35:462-71.
- 207. Lambrew CT, Weaver WD, Rogers WJ, Bowlby LJ, Rubison RM, French WJ. Hospital protocols and policies that may delay early identification and thrombolytic therapy of acute myocardial infarction patients. *J Thromb Thrombolysis* 1996;3:301-306.
- 208. Farkouh ME, Smars PA, Reeder GS, et al, for the Chest Pain Evaluation in the Emergency Room (CHEER) Investigators. A clinical trial of a chest-pain observation unit for patients with unstable angina. N Engl J Med 1998;339:1882-8.
- 209. Deleted in press.
- 210. Charney P. Coronary artery disease in young women: the menstrual cycle and other risk factors. *Ann Intern Med* 2001;135: 1002-4.
- 211. Hasdai D, Porter A, Rosengren A, Behar S, Boyko V, Battler A. Effect of gender on outcomes of acute coronary syndromes. *Am J Cardiol* 2003;91:1466-9–A6.
- 212. Kudenchuk PJ, Maynard C, Martin JS, Wirkus M, Weaver WD. Comparison of presentation, treatment, and outcome of acute myocardial infarction in men versus women (the Myocardial Infarction Triage and Intervention Registry) Am J Cardiol 1996;78:9-14.
- 213. Goldberg RJ, McCormick D, Gurwitz JH, Yarzebski J, Lessard D, Gore JM. Age-related trends in short- and long-term survival after acute myocardial infarction: a 20-year population-based perspective (1975-1995). *Am J Cardiol* 1998;82:1311-7.
- 214. Meischke H, Dulberg EM, Schaeffer SS, Henwood DK, Larsen MP, Eisenberg MS. 'Call fast, Call 911': a direct mail campaign to reduce patient delay in acute myocardial infarction. Am J Public Health 1997;87:1705-9.
- 215. McSweeney JC, Cody M, O'Sullivan P, Elberson K, Moser DK, Garvin BJ. Women's early warning symptoms of acute myocar-

dial infarction. Circulation 2003;108:2619-23.

- 216. Bonow RO, Mitch WE, Nesto RW, et al. Prevention Conference VI: Diabetes and Cardiovascular Disease: Writing Group V: management of cardiovascular-renal complications. *Circulation* 2002;105:e159-64.
- 217. Levy DE. How transient are transient ischemic attacks? *Neurology* 1988;38:674-7.
- 218. Sloan MA, Gore JM. Ischemic stroke and intracranial hemorrhage following thrombolytic therapy for acute myocardial infarction: a risk-benefit analysis. *Am J Cardiol* 1992;69:21A-38A.
- 219. Hochman JH, Gersh BJ. Acute myocardial infarction. In: Topol EJ, ed. Textbook of Cardiovascular Medicine. 2nd ed. Philadelphia, PA: Lippincott, Williams, and Wilkins; 2002:438.
- 220. Spodick DH. Diagnostic electrocardiographic sequences in acute pericarditis: significance of PR segment and PR vector changes. *Circulation* 1973;48:575-80.
- 221. Menown IB, Mackenzie G, Adgey AA. Optimizing the initial 12lead electrocardiographic diagnosis of acute myocardial infarction. *Eur Heart J* 2000;21:275-83.
- 222. Mauri F, Gasparini M, Barbonaglia L, et al. Prognostic significance of the extent of myocardial injury in acute myocardial infarction treated by streptokinase (the GISSI trial). *Am J Cardiol* 1989;63:1291-5.
- 223. Antman EM, Rutherford JD. Coronary care medicine: a practical approach. Boston, MA: Martinus Nijhoff Publishing;1986:81.
- 224. Effects of tissue plasminogen activator and a comparison of early invasive and conservative strategies in unstable angina and non–Q-wave myocardial infarction: results of the TIMI IIIB Trial. Thrombolysis in Myocardial Ischemia. *Circulation* 1994;89: 1545-56.
- 225. Matetzky S, Freimark D, Chouraqui P, et al. Significance of ST segment elevations in posterior chest leads (V7 to V9) in patients with acute inferior myocardial infarction: application for thrombolytic therapy. *J Am Coll Cardiol* 1998;31:506-11.
- 226. Cheitlin MD, Armstrong WF, Aurigemma GP, et al. ACC/AHA/ASE 2003 guideline update for the clinical application of echocardiography: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (ACC/AHA/ASE Committee to Update the 1997 Guidelines for the Clinical Application of Echocardiography). American College of Cardiology Web Site. Available at: www.acc.org/clinical/guidelines/echo/index.pdf, 2003. Accessed August 4, 2003.
- 227. Goldman L, Cook EF, Brand DA, et al. A computer protocol to predict myocardial infarction in emergency department patients with chest pain. N Engl J Med 1988;318:797-803.
- 228. Wang K, Asinger RW, Marriott HJ. ST-segment elevation in conditions other than acute myocardial infarction. *N Engl J Med* 2003;349:2128-35.
- 229. Sgarbossa EB, Pinski SL, Barbagelata A, et al, for the GUSTO-1 (Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries) Investigators. Electrocardiographic diagnosis of evolving acute myocardial infarction in the presence of left bundle-branch block. N Engl J Med 1996;334:481-7.
- 230. Sgarbossa EB. Value of the ECG in suspected acute myocardial infarction with left bundle-branch block. *J Electrocardiol* 2000;33(Suppl):87-92.
- 231. Luepker RV, Apple FS, Christenson RH, et al. Case definitions for acute coronary heart disease in epidemiology and clinical research studies: a statement from the AHA Council on Epidemiology and Prevention; AHA Statistics Committee; World Heart Federation

Council on Epidemiology and Prevention; the European Society of Cardiology Working Group on Epidemiology and Prevention; Centers for Disease Control and Prevention; and the National Heart, Lung, and Blood Institute. *Circulation* 2003;108:2543-9.

- 232. Adams JE, Abendschein DR, Jaffe AS. Biochemical markers of myocardial injury: is MB creatine kinase the choice for the 1990s? *Circulation* 1993;88:750-63.
- 233. Alpert JS, Thygesen K, Antman E, Bassand JP. Myocardial infarction redefined: a consensus document of the Joint European Society of Cardiology/American College of Cardiology Committee for the redefinition of myocardial infarction. *J Am Coll Cardiol* 2000;36:959-69.
- 234. Wu AH, Apple FS, Gibler WB, Jesse RL, Warshaw MM, Valdes R. National Academy of Clinical Biochemistry Standards of Laboratory Practice: recommendations for the use of cardiac markers in coronary artery diseases. *Clin Chem* 1999;45:1104-21.
- 235. Ohman EM, Armstrong PW, White HD, et al for the Global Use of Strategies To Open Occluded Coronary Arteries (GUSTO III) Investigators. Risk stratification with a point-of-care cardiac troponin T test in acute myocardial infarction. *Am J Cardiol* 1999;84:1281-6.
- 236. Tanasijevic MJ, Cannon CP, Wybenga DR, et al. Myoglobin, creatine kinase MB, and cardiac troponin-I to assess reperfusion after thrombolysis for acute myocardial infarction: results from TIMI 10A. *Am Heart J* 1997;134:622-30.
- 237. Puleo PR, Meyer D, Wathen C, et al. Use of a rapid assay of subforms of creatine kinase-MB to diagnose or rule out acute myocardial infarction. *N Engl J Med* 1994;331:561-6.
- 238. Antman EM, Sacks DB, Rifai N, McCabe CH, Cannon CP, Braunwald E. Time to positivity of a rapid bedside assay for cardiac-specific troponin T predicts prognosis in acute coronary syndromes: a Thrombolysis in Myocardial Infarction (TIMI) 11A substudy. J Am Coll Cardiol 1998;31:326-30.
- 239. Klocke FJ, Baird MG, Bateman TM, et al. ACC/AHA/ASNC guidelines for the clinical use of cardiac radionuclide imaging: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (ACC/AHA/ASNC Committee to Revise the 1995 Guidelines for the Clinical Use of Radionuclide Imaging). American College of Cardiology Web Site. Available at http://www.acc.org/clinical/guidelines/radio/rni_fulltext.pdf, 2003. Accessed August 18, 2003.
- 240. Lee KL, Woodlief LH, Topol EJ, et al, for the GUSTO-I Investigators. Predictors of 30-day mortality in the era of reperfusion for acute myocardial infarction: results from an international trial of 41,021 patients. *Circulation* 1995;91:1659-68.
- 241. Morrow DA, Antman EM, Giugliano RP, et al. A simple risk index for rapid initial triage of patients with ST-elevation myocardial infarction: an InTIME II substudy. *Lancet* 2001;358:1571-5.
- 242. Morrow DA, Antman EM, Charlesworth A, et al. TIMI risk score for ST-elevation myocardial infarction: a convenient, bedside, clinical score for risk assessment at presentation: an intravenous nPA for treatment of infarcting myocardium early II trial substudy. *Circulation* 2000;102:2031-7.
- 243. Selker HP, Griffith JL, Beshansky JR, et al. Patient-specific predictions of outcomes in myocardial infarction for real-time emergency use: a thrombolytic predictive instrument. *Ann Intern Med* 1997;127:538-56.
- 244. Selker HP, Beshansky JR, Griffith JL, for the TPI Trial Investigators. Use of the electrocardiograph-based thrombolytic predictive instrument to assist thrombolytic and reperfusion ther-

apy for acute myocardial infarction: a multicenter, randomized, controlled, clinical effectiveness trial. *Ann Intern Med* 2002;137: 87-95.

- 245. Vaccarino V, Parsons L, Every NR, Barron HV, Krumholz HM, for the National Registry of Myocardial Infarction 2 Participants. Sex-based differences in early mortality after myocardial infarction. *N Engl J Med* 1999;341:217-25.
- 246. Brass LM, Lichtman JH, Wang Y, Gurwitz JH, Radford MJ, Krumholz HM. Intracranial hemorrhage associated with thrombolytic therapy for elderly patients with acute myocardial infarction: results from the Cooperative Cardiovascular Project. *Stroke* 2000;31:1802-11.
- 247. Krumholz HM, Pasternak RC, Weinstein MC, et al L. Cost effectiveness of thrombolytic therapy with streptokinase in elderly patients with suspected acute myocardial infarction. *N Engl J Med* 1992;327:7-13.
- 248. Maroko PR, Radvany P, Braunwald E, Hale SL. Reduction of infarct size by oxygen inhalation following acute coronary occlusion. *Circulation* 1975;52:360-8.
- 249. Madias JE, Hood WB. Reduction of precordial ST-segment elevation in patients with anterior myocardial infarction by oxygen breathing. *Circulation* 1976;53:I198-200.
- 250. Fillmore SJ, Shapiro M, Killip T. Arterial oxygen tension in acute myocardial infarction: serial analysis of clinical state and blood gas changes. *Am Heart J* 1970;79:620-9.
- 251. Aubier M, Trippenbach T, Roussos C. Respiratory muscle fatigue during cardiogenic shock. *J Appl Physiol* 1981;51:499-508.
- 252. Abrams J. Hemodynamic effects of nitroglycerin and long-acting nitrates. *Am Heart J* 1985;110:216-24.
- 253. Winbury MM. Redistribution of left ventricular blood flow produced by nitroglycerin: an example of integration of the macroand microcirculation. *Circ Res* 1971;28(Suppl 1):140-7.
- 254. Gorman MW, Sparks HV. Nitroglycerin causes vasodilatation within ischaemic myocardium. *Cardiovasc Res* 1980;14:515-21.
- 255. Deleted in press.
- 256. Come PC, Pitt B. Nitroglycerin-induced severe hypotension and bradycardia in patients with acute myocardial infarction. *Circulation* 1976;54:624-8.
- 257. Kinch JW, Ryan TJ. Right ventricular infarction. *N Engl J Med* 1994;330:1211-7.
- 258. Cheitlin MD, Hutter AM, Brindis RG, et al. ACC/AHA expert consensus document on the use of sildenafil (Viagra) in patients with cardiovascular disease. American College of Cardiology/ American Heart Association. *J Am Coll Cardiol* 1999;33:273-82.
- 259. Antman EM, Braunwald E. Acute myocardial infarction. In: Braunwald E, Zipes DP, Libby P, eds. Heart disease: a textbook of cardiovascular medicine, 6th ed. Philadelphia, PA: WB Saunders Co Ltd; 2001:1114-1251.
- 260. Hochman JS, Califf RM. Acute myocardial infarction. In: Smith TW. Cardiovascular therapeutics: a companion to Braunwald's heart disease, 2nd ed. Philadelphia, PA: WB Saunders Co Ltd; 2001:235-291.
- 261. ISIS-2 (Second International Study of Infarct Survival) Collaborative Group. Randomised trial of intravenous streptokinase, oral aspirin, both, or neither among 17,187 cases of suspected acute myocardial infarction: ISIS-2. *Lancet* 1988;2:349-60.
- 262. Roux S, Christeller S, Lüdin E. Effects of aspirin on coronary reocclusion and recurrent ischemia after thrombolysis: a metaanalysis. *J Am Coll Cardiol* 1992;19:671-7.
- 263. Antithrombotic Trialists' Collaboration. Collaborative metaanalysis of randomised trials of antiplatelet therapy for prevention

of death, myocardial infarction, and stroke in high-risk patients. *BMJ* 2002;324:71-86.

- 264. Sagar KA, Smyth MR. A comparative bioavailability study of different aspirin formulations using on-line multidimensional chromatography. *J Pharm Biomed Anal* 1999;21:383-92.
- 265. Yusuf S, Peto R, Lewis J, Collins R, Sleight P. Beta blockade during and after myocardial infarction: an overview of the randomized trials. *Prog Cardiovasc Dis* 1985;27:335-71.
- 266. Randomised trial of intravenous atenolol among 16 027 cases of suspected acute myocardial infarction: ISIS-1. First International Study of Infarct Survival Collaborative Group. *Lancet* 1986;2:57-66.
- 267. The MIAMI Trial Research Group. Metoprolol in acute myocardial infarction: patient population. *Am J Cardiol* 1985;56:10G-14G.
- 268. The TIMI Study Group. Comparison of invasive and conservative strategies after treatment with intravenous tissue plasminogen activator in acute myocardial infarction: results of the thrombolysis in myocardial infarction (TIMI) phase II trial. *N Engl J Med* 1989;320:618-27.
- 269. Roberts R, Rogers WJ, Mueller HS, et al. Immediate versus deferred beta-blockade following thrombolytic therapy in patients with acute myocardial infarction: results of the Thrombolysis in Myocardial Infarction (TIMI) II-B Study. *Circulation* 1991;83: 422-37.
- 270. Van de Werf F, Janssens L, Brzostek T, et al. Short-term effects of early intravenous treatment with a beta-adrenergic blocking agent or a specific bradycardiac agent in patients with acute myocardial infarction receiving thrombolytic therapy. J Am Coll Cardiol 1993;22:407-16.
- 271. Pfisterer M, Cox JL, Granger CB, et al. Atenolol use and clinical outcomes after thrombolysis for acute myocardial infarction: the GUSTO-I experience. Global Utilization of Streptokinase and TPA (alteplase) for Occluded Coronary Arteries. J Am Coll Cardiol 1998;32:634-40.
- 272. Kloner RA, Hale S. Unraveling the complex effects of cocaine on the heart. *Circulation* 1993;87:1046-7.
- 273. Dargie HJ. Effect of carvedilol on outcome after myocardial infarction in patients with left-ventricular dysfunction: the CAPRICORN randomised trial. *Lancet* 2001;357:1385-90.
- 274. De Luca G, Suryapranata H, Zijlstra F, et al, for the ZWOLLE Myocardial Infarction Study Group. Symptom-onset-to-balloon time and mortality in patients with acute myocardial infarction treated by primary angioplasty. J Am Coll Cardiol 2003;42:991-7.
- 275. De Luca G, Suryapranata H, Ottervanger JP, Antman EM. Time delay to treatment and mortality in primary angioplasty for acute myocardial infarction: every minute of delay counts. *Circulation* 2004;109:1223-5.
- 275a. Williams DO. Treatment delayed is treatment denied. *Circulation* 2004;109:1806-8.
- 276. Davies CH, Ormerod OJ. Failed coronary thrombolysis. *Lancet* 1998;351:1191-6.
- 277. Ito H, Okamura A, Iwakura K, Masuyama T, et al. Myocardial perfusion patterns related to thrombolysis in myocardial infarction perfusion grades after coronary angioplasty in patients with acute anterior wall myocardial infarction. *Circulation* 1996;93: 1993-9.
- 278. Wu KC, Zerhouni EA, Judd RM, et al. Prognostic significance of microvascular obstruction by magnetic resonance imaging in patients with acute myocardial infarction. *Circulation* 1998;97: 765-72.
- 279. Zeymer U, Tebbe U, Essen R, Haarmann W, Neuhaus KL, for the

ALKK-Study Group. Influence of time to treatment on early infarct-related artery patency after different thrombolytic regimens. *Am Heart J* 1999;137:34-8.

- 280. Gibson CM. Has my patient achieved adequate myocardial reperfusion? *Circulation* 2003;108:504-7.
- 281. Eagle KA, Goodman SG, Avezum A, Budaj A, Sullivan CM, Lopez-Sendon J, for the GRACE Investigators. Practice variation and missed opportunities for reperfusion in ST-segment-elevation myocardial infarction: findings from the Global Registry of Acute Coronary Events (GRACE). *Lancet* 2002;359:373-7.
- 282. Weaver WD, Simes RJ, Betriu A, et al. Comparison of primary coronary angioplasty and intravenous thrombolytic therapy for acute myocardial infarction: a quantitative review. *JAMA* 1997; 278:2093-8.
- 283. Hasdai D, Behar S, Wallentin L, et al. A prospective survey of the characteristics, treatments and outcomes of patients with acute coronary syndromes in Europe and the Mediterranean basin. The Euro Heart Survey of Acute Coronary Syndromes (Euro Heart Survey ACS). *Eur Heart J* 2002;23:1190-201.
- 284. Boersma E, Maas AC, Deckers JW, Simoons ML. Early thrombolytic treatment in acute myocardial infarction: reappraisal of the golden hour. *Lancet* 1996;348:771-5.
- 285. Morrison LJ, Verbeek PR, McDonald AC, Sawadsky BV, Cook DJ. Mortality and prehospital thrombolysis for acute myocardial infarction: a meta-analysis. *JAMA* 2000;283:2686-92.
- 286. Reimer KA, Jennings RB, Cobb FR, et al. Animal models for protecting ischemic myocardium: results of the NHLBI Cooperative Study: comparison of unconscious and conscious dog models. *Circ Res* 1985;56:651-65.
- 287. Christian TF, Schwartz RS, Gibbons RJ. Determinants of infarct size in reperfusion therapy for acute myocardial infarction. *Circulation* 1992;86:81-90.
- 288. Huber KC, Bresnahan JF, Bresnahan DR, Pellikka PA, Behrenbeck T, Gibbons RJ. Measurement of myocardium at risk by technetium-99m sestamibi: correlation with coronary angiography. J Am Coll Cardiol 1992;19:67-73.
- 289. Klarich KW, Christian TF, Higano ST, Gibbons RJ. Variability of myocardium at risk for acute myocardial infarction. Am J Cardiol 1999;83:1191-5.
- 290. Brodie BR, Stuckey TD, Wall TC, et al. Importance of time to reperfusion for 30-day and late survival and recovery of left ventricular function after primary angioplasty for acute myocardial infarction. *J Am Coll Cardiol* 1998;32:1312-9.
- 291. Brodie BR, Stone GW, Morice MC, et al, for the Stent Primary Angioplasty in Myocardial Infarction Study Group. Importance of time to reperfusion on outcomes with primary coronary angioplasty for acute myocardial infarction (results from the Stent Primary Angioplasty in Myocardial Infarction Trial). Am J Cardiol 2001;88:1085-90.
- 292. Brodie BR, Stuckey TD, Muncy DB, et al. Importance of time-toreperfusion in patients with acute myocardial infarction with and without cardiogenic shock treated with primary percutaneous coronary intervention. *Am Heart J* 2003;145:708-15.
- 293. Antoniucci D, Valenti R, Migliorini A, et al. Relation of time to treatment and mortality in patients with acute myocardial infarction undergoing primary coronary angioplasty. *Am J Cardiol* 2002;89:1248-52.
- 294. Berger PB, Ellis SG, Holmes DR, et al. Relationship between delay in performing direct coronary angioplasty and early clinical outcome in patients with acute myocardial infarction: results from the global use of strategies to open occluded arteries in Acute Coronary Syndromes (GUSTO-IIb) trial. *Circulation* 1999;100:

ACC - www.acc.org

14-20.

- 295. Cannon CP, Gibson CM, Lambrew CT, et al. Relationship of symptom-onset-to-balloon time and door-to-balloon time with mortality in patients undergoing angioplasty for acute myocardial infarction. *JAMA* 2000;283:2941-7.
- 296. Zijlstra F, Patel A, Jones M, et al. Clinical characteristics and outcome of patients with early (less than 2 h), intermediate (2-4 h) and late (greater than 4 h) presentation treated by primary coronary angioplasty or thrombolytic therapy for acute myocardial infarction. *Eur Heart J* 2002;23:550-7.
- 297. Van de Werf F, Ardissino D, Betriu A, et al, for the Task Force on the Management of Acute Myocardial Infarction of the European Society of Cardiology. Management of acute myocardial infarction in patients presenting with ST-segment elevation. The Task Force on the Management of Acute Myocardial Infarction of the European Society of Cardiology. *Eur Heart J* 2003;24:28-66.
- 298. Krumholz HM, Chen J, Wang Y, Radford MJ, Chen YT, Marciniak TA. Comparing AMI mortality among hospitals in patients 65 years of age and older: evaluating methods of risk adjustment. *Circulation* 1999;99:2986-92.
- 299. Granger CB, Goldberg RJ, Dabbous O, et al, for the Global Registry of Acute Coronary Events Investigators. Predictors of hospital mortality in the global registry of acute coronary events. *Arch Intern Med* 2003;163:2345-53.
- 300. Kent DM, Schmid CH, Lau J, Selker HP. Is primary angioplasty for some as good as primary angioplasty for all? *J Gen Intern Med* 2002;17:887-94.
- 301. Hochman JS, Sleeper LA, Webb JG, et al, for the Should We Emergently Revascularize Occluded Coronaries for Cardiogenic Shock (SHOCK) Investigators. Early revascularization in acute myocardial infarction complicated by cardiogenic shock. N Engl J Med 1999;341:625-34.
- 302. Wu AH, Parsons L, Every NR, Bates ER, for the Second National Registry of Myocardial Infarction. Hospital outcomes in patients presenting with congestive heart failure complicating acute myocardial infarction: a report from the Second National Registry of Myocardial Infarction (NRMI-2). J Am Coll Cardiol 2002; 40:1389-94.
- 303. Magid DJ, Calonge BN, Rumsfeld JS, et al, for the National Registry of Myocardial Infarction 2 and 3 Investigators. Relation between hospital primary angioplasty volume and mortality for patients with acute MI treated with primary angioplasty vs thrombolytic therapy. *JAMA* 2000;284:3131-8.
- 304. Canto JG, Every NR, Magid DJ, et al, for the National Registry of Myocardial Infarction 2 Investigators. The volume of primary angioplasty procedures and survival after acute myocardial infarction. N Engl J Med 2000;342:1573-80.
- 305. Nallamothu BK, Bates ER. Percutaneous coronary intervention versus fibrinolytic therapy in acute myocardial infarction: is timing (almost) everything? *Am J Cardiol* 2003;92:824-6.
- 306. Andersen HR, Nielsen TT, Rasmussen K, et al, for the DANAMI-2 Investigators. A comparison of coronary angioplasty with fibrinolytic therapy in acute myocardial infarction. N Engl J Med 2003;349:733-42.
- 307. Aversano T, Aversano LT, Passamani E, et al, for the Atlantic Cardiovascular Patient Outcomes Research Team (C-PORT). Thrombolytic therapy vs primary percutaneous coronary intervention for myocardial infarction in patients presenting to hospitals without on-site cardiac surgery: a randomized controlled trial. *JAMA* 2002;287:1943-51.
- 308. Angeja BG, Gibson CM, Chin R, et al, for the Participants in the National Registry of Myocardial Infarction 2-3. Predictors of

door-to-balloon delay in primary angioplasty. Am J Cardiol 2002;89:1156-61.

- 309. Henriques JP, Haasdijk AP, Zijlstra F, for the Zwolle Myocardial Infarction Study Group. Outcome of primary angioplasty for acute myocardial infarction during routine duty hours versus during off-hours. *J Am Coll Cardiol* 2003;41:2138-42.
- 310. Spencer FA, Becker RC. Circadian variations in acute myocardial infarction: patients or health care delivery? *J Am Coll Cardiol* 2003;41:2143-6.
- 311. Armstrong PW, Antman EM. Coronary angioplasty versus fibrinolytic therapy in acute myocardial infarction. *N Engl J Med* 2003;349:2167-9; author reply.
- 312. Herrick JB. Landmark article (JAMA 1912): clinical features of sudden obstruction of the coronary arteries. *JAMA* 1983;250: 1757-65.
- 313. Chazov EI, Matveeva LS, Mazaev AV, Sargin KE, Sadovskaia GV, Ruda MI. [Intracoronary administration of fibrinolysin in acute myocardial infarct.] *Ter Arkh* 1976;48:8-19.
- 314. Aylward PE, Wilcox RG, Horgan JH, et al, for the GUSTO-I Investigators. Relation of increased arterial blood pressure to mortality and stroke in the context of contemporary thrombolytic therapy for acute myocardial infarction: a randomized trial. *Ann Intern Med* 1996;125:891-900.
- 315. Antman EM. General hospital management. In: Julian DG, Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co Ltd;1994:42-44.
- **316.** Goldberger AL. Hyperacute T waves revisited. *Am Heart J* 1982;104:888-90.
- 317. Boden WE, Kleiger RE, Gibson RS, et al. Electrocardiographic evolution of posterior acute myocardial infarction: importance of early precordial ST-segment depression. *Am J Cardiol* 1987;59: 782-7.
- 318. Zimetbaum PJ, Josephson ME. Use of the electrocardiogram in acute myocardial infarction. *N Engl J Med* 2003;348:933-40.
- 319. Zehender M, Kasper W, Kauder E, et al. Eligibility for and benefit of thrombolytic therapy in inferior myocardial infarction: focus on the prognostic importance of right ventricular infarction. *J Am Coll Cardiol* 1994;24:362-9.
- 320. Gore JM, Sloan M, Price TR, et al. Intracerebral hemorrhage, cerebral infarction, and subdural hematoma after acute myocardial infarction and thrombolytic therapy in the Thrombolysis in Myocardial Infarction Study. Thrombolysis in Myocardial Infarction, Phase II, pilot and clinical trial. *Circulation* 1991; 83:448-59.
- 321. Thrombolytic therapy in thrombosis: a National Institutes of Health consensus development conference. Ann Intern Med 1980; 93:141-4.
- 322. Gore JM, Granger CB, Simoons ML, et al. Stroke after thrombolysis: mortality and functional outcomes in the GUSTO-I trial. Global Use of Strategies to Open Occluded Coronary Arteries. *Circulation* 1995;92:2811-8.
- 323. White HD, Aylward PE, Frey MJ, et al, for the Hirulog Early Reperfusion/Occlusion (HERO) Trial Investigators. Randomized, double-blind comparison of hirulog versus heparin in patients receiving streptokinase and aspirin for acute myocardial infarction (HERO). *Circulation* 1997;96:2155-61.
- 324. Randomised, double-blind comparison of reteplase double-bolus administration with streptokinase in acute myocardial infarction (INJECT): trial to investigate equivalence. International Joint Efficacy Comparison of Thrombolytics. *Lancet* 1995;346:329-36.
- 325. Giugliano RP, McCabe CH, Antman EM, et al. Lower-dose heparin with fibrinolysis is associated with lower rates of intracra-

nial hemorrhage. Am Heart J 2001;141:742-50.

- 326. Wienbergen H, Schiele R, Gitt AK, et al, for the Myocardial Infarction Registry (MIR) and Maximal Individual Therapy in Acute Myocardial Infarction (MITRA) Study Groups. Incidence, risk factors, and clinical outcome of stroke after acute myocardial infarction in clinical practice. *Am J Cardiol* 2001;87:782-5, A8.
- 327. Kandzari DE, Granger CB, Simoons ML, et al, for the Global Utilization of Streptokinase and tPA for Occluded Arteries-I (GUSTO-I) Investigators. Risk factors for intracranial hemorrhage and nonhemorrhagic stroke after fibrinolytic therapy (from the GUSTO-I trial). *Am J Cardiol* 2004;93:458-61.
- 328. Gurwitz JH, Gore JM, Goldberg RJ, et al, for the Participants in the National Registry of Myocardial Infarction 2. Risk for intracranial hemorrhage after tissue plasminogen activator treatment for acute myocardial infarction. *Ann Intern Med* 1998; 129:597-604.
- 329. Simoons ML, Maggioni AP, Knatterud G, et al. Individual risk assessment for intracranial haemorrhage during thrombolytic therapy. *Lancet* 1993;342:1523-8.
- 330. Sloan MA, Guigliano RP, Thompson SL. Prediction of intracranial hemorrhage in the InTIME-II trial. J Am Coll Cardiol 2001; 37:372A.
- 331. Wilcox RG, von der Lippe G, Olsson CG, Jensen G, Skene AM, Hampton JR. Trial of tissue plasminogen activator for mortality reduction in acute myocardial infarction. Anglo-Scandinavian Study of Early Thrombolysis (ASSET). *Lancet* 1988;2:525-30.
- 332. AIMS Trial Study Group. Long-term effects of intravenous anistreplase in acute myocardial infarction: final report of the AIMS study. *Lancet* 1990;335:427-31.
- 333. Lamas GA, Flaker GC, Mitchell G, et al, for the Survival and Ventricular Enlargement Investigators. Effect of infarct artery patency on prognosis after acute myocardial infarction. *Circulation* 1995;92:1101-9.
- 334. Baigent C, Collins R, Appleby P, Parish S, Sleight P, Peto R, for the ISIS-2 (Second International Study of Infarct Survival) Collaborative Group. ISIS-2: 10-year survival among patients with suspected acute myocardial infarction in randomised comparison of intravenous streptokinase, oral aspirin, both, or neither. *BMJ* 1998;316:1337-43.
- 335. Franzosi MG, Santoro E, De Vita C, et al, for the GISSI Investigators. Ten-year follow-up of the first megatrial testing thrombolytic therapy in patients with acute myocardial infarction: results of the Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto-1 study. *Circulation* 1998;98:2659-65.
- 336. Van de Werf F. Thrombolysis for acute myocardial infarction: why is there no extra benefit after hospital discharge? *Circulation* 1995;91:2862-4.
- 337. Deleted in press.
- 338. Gillium BS, Graves EJ, Wood E. National hospital discharge survey: data from the National Health Survey. *Vital and Health Statistics* 1998;13:1-51.
- 339. White HD. Thrombolytic therapy in the elderly. *Lancet* 2000; 356:2028-30.
- 340. Thiemann DR, Coresh J, Schulman SP, Gerstenblith G, Oetgen WJ, Powe NR. Lack of benefit for intravenous thrombolysis in patients with myocardial infarction who are older than 75 years. *Circulation* 2000;101:2239-46.
- 341. Berger AK, Radford MJ, Wang Y, Krumholz HM. Thrombolytic therapy in older patients. *J Am Coll Cardiol* 2000;36:366-74.
- 342. Stenestrand U, Wallentin L. Fibrinolytic therapy in patients 75 years and older with ST-segment-elevation myocardial infarction: one-year follow-up of a large prospective cohort. *Arch Intern Med*

2003;163:965-71.

- 343. Ross AM, Coyne KS, Moreyra E, et al, for the GUSTO-I Angiographic Investigators. Extended mortality benefit of early postinfarction reperfusion. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries Trial. *Circulation* 1998;97:1549-56.
- 344. Ross AM, Lundergan CF, Rohrbeck SC, et al, for the GUSTO-1 Angiographic Investigators. Rescue angioplasty after failed thrombolysis: technical and clinical outcomes in a large thrombolysis trial. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries. *J Am Coll Cardiol* 1998;31:1511-7.
- 345. Clements IP, Christian TF, Higano ST, Gibbons RJ, Gersh BJ. Residual flow to the infarct zone as a determinant of infarct size after direct angioplasty. *Circulation* 1993;88:1527-33.
- 346. Stone GW, Cox D, Garcia E, et al. Normal flow (TIMI-3) before mechanical reperfusion therapy is an independent determinant of survival in acute myocardial infarction: analysis from the primary angioplasty in myocardial infarction trials. *Circulation* 2001; 104:636-41.
- 347. Laster SB, O'Keefe JH, Gibbons RJ. Incidence and importance of thrombolysis in myocardial infarction grade 3 flow after primary percutaneous transluminal coronary angioplasty for acute myocardial infarction. *Am J Cardiol* 1996;78:623-6.
- 348. Stone GW, Peterson MA, Lansky AJ, Dangas G, Mehran R, Leon MB. Impact of normalized myocardial perfusion after successful angioplasty in acute myocardial infarction. *J Am Coll Cardiol* 2002;39:591-7.
- 349. Angeja BG, Gunda M, Murphy SA, et al. TIMI myocardial perfusion grade and ST segment resolution: association with infarct size as assessed by single photon emission computed tomography imaging. *Circulation* 2002;105:282-5.
- 350. Gibbons RJ, Miller DD, Liu P, Guo P, Brooks MM, Schwaiger M. Similarity of ventricular function in patients alive 5 years after randomization to surgery or angioplasty in the BARI trial. *Circulation* 2001;103:1076-82.
- 351. Rovelli F, De Vita C, Feruglio GA, Lotto A, Selvini A, Tognoni G, for the Gruppo Italiano per la Sperimentazione della Streptochinasi nell'Infarto Miocardico. GISSI trial: early results and late follow-up. *J Am Coll Cardiol* 1987;10:33B-39B.
- 352. Maggioni AP, Franzosi MG, Farina ML, et al, for the Gruppo Italiano per lo Studio della Streptochinasi nell'Infarto Miocardico (GISSI). Cerebrovascular events after myocardial infarction: analysis of the GISSI trial. *BMJ* 1991;302:1428-31.
- 353. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto Miocardico. GISSI-2: a factorial randomised trial of alteplase versus streptokinase and heparin versus no heparin among 12,490 patients with acute myocardial infarction. *Lancet* 1990;336:65-71.
- 354. The International Study Group. In-hospital mortality and clinical course of 20,891 patients with suspected acute myocardial infarction randomised between alteplase and streptokinase with or without heparin. *Lancet* 1990;336:71-5.
- 355. Maggioni AP, Franzosi MG, Santoro E, White H, Van de Werf F, Tognoni G, for the Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto Miocardico II (GISSI-2) and the International Study Group. The risk of stroke in patients with acute myocardial infarction after thrombolytic and antithrombotic treatment. *N Engl J Med* 1992;327:1-6.
- 356. De Jaegere PP, Arnold AA, Balk AH, Simoons ML. Intracranial hemorrhage in association with thrombolytic therapy: incidence and clinical predictive factors. J Am Coll Cardiol 1992;19:289-94.
- 357. ISIS-3 (Third International Study of Infarct Survival) Collaborative Group. ISIS-3: a randomised comparison of streptokinase vs tissue plasminogen activator vs anistreplase and of aspirin plus heparin vs aspirin alone among 41,299 cases of suspected acute myocardial infarction. *Lancet* 1992;339:753-70.
- 358. Estudio Multicéntrico Estreptoquinasa Repúblicas de América del Sur (EMERAS) Collaborative Group. Randomised trial of late thrombolysis in patients with suspected acute myocardial infarction. *Lancet* 1993;342:767-72.
- 359. The Global Use of Strategies to Open Occluded Coronary Arteries (GUSTO) IIa Investigators. Randomized trial of intravenous heparin versus recombinant hirudin for acute coronary syndromes. *Circulation* 1994;90:1631-7.
- 360. Sloan MA, Price TR, Terrin ML, et al. Ischemic cerebral infarction after rt-PA and heparin therapy for acute myocardial infarction: the TIMI-II pilot and randomized clinical trial combined experience. *Stroke* 1997;28:1107-14.
- 361. Wilcox RG, von der Lippe G, Olsson CG, Jensen G, Skene AM, Hampton JR. Effects of alteplase in acute myocardial infarction: 6-month results from the ASSET study. Anglo-Scandinavian Study of Early Thrombolysis. *Lancet* 1990;335:1175-8.
- 362. O'Connor CM, Califf RM, Massey EW, et al. Stroke and acute myocardial infarction in the thrombolytic era: clinical correlates and long-term prognosis. *J Am Coll Cardiol* 1990;16:533-40.
- 363. Kase CS, Pessin MS, Zivin JA, et al. Intracranial hemorrhage after coronary thrombolysis with tissue plasminogen activator. *Am J Med* 1992;92:384-90.
- 364. Sloan MA, Price TR, Petito CK, et al. Clinical features and pathogenesis of intracerebral hemorrhage after rt-PA and heparin therapy for acute myocardial infarction: the Thrombolysis in Myocardial Infarction (TIMI) II Pilot and Randomized Clinical Trial combined experience. *Neurology* 1995;45:649-58.
- 365. Longstreth WT, Litwin PE, Weaver WD, for the MITI Project Group. Myocardial infarction, thrombolytic therapy, and stroke: a community-based study. *Stroke* 1993;24:587-90.
- 366. Late Assessment of Thrombolytic Efficacy (LATE) study with alteplase 6-24 hours after onset of acute myocardial infarction. *Lancet* 1993;342:759-66.
- 367. Vermeer F, Bösl I, Meyer J, et al. Saruplase is a safe and effective thrombolytic agent; observations in 1,698 patients: results of the PASS study. Practical Applications of Saruplase Study. *J Thromb Thrombolysis* 1999;8:143-50.
- 368. Neuhaus KL, von Essen R, Tebbe U, et al. Safety observations from the pilot phase of the randomized r-Hirudin for Improvement of Thrombolysis (HIT-III) study: a study of the Arbeitsgemeinschaft Leitender Kardiologischer Krankenhausärzte (ALKK) Circulation 1994;90:1638-42.
- 369. Antman EM, Giugliano RP, Gibson CM, et al, for the TIMI 14 Investigators. Abciximab facilitates the rate and extent of thrombolysis: results of the thrombolysis in myocardial infarction (TIMI) 14 trial. *Circulation* 1999;99:2720-32.
- 370. The National Institute of Neurological Disorders and Stroke (NINDS) rt-PA Stroke Study Group. A systems approach to immediate evaluation and management of hyperacute stroke: experience at eight centers and implications for community practice and patient care. *Stroke* 1997;28:1530-40.
- 371. Broderick JP, Adams HP, Barsan W, et al. Guidelines for the management of spontaneous intracerebral hemorrhage: a statement for healthcare professionals from a special writing group of the Stroke Council, American Heart Association. *Stroke* 1999;30:905-15.
- 372. Gebel JM, Sila CA, Sloan MA, et al. Thrombolysis-related

intracranial hemorrhage: a radiographic analysis of 244 cases from the GUSTO-1 trial with clinical correlation. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries. *Stroke* 1998;29:563-9.

- 373. Gebel JM, Sila CA, Sloan MA, et al. Comparison of the ABC/2 estimation technique to computer-assisted volumetric analysis of intraparenchymal and subdural hematomas complicating the GUSTO-1 trial. *Stroke* 1998;29:1799-801.
- 374. Sane DC, Califf RM, Topol EJ, Stump DC, Mark DB, Greenberg CS. Bleeding during thrombolytic therapy for acute myocardial infarction: mechanisms and management. *Ann Intern Med* 1989; 111:1010-22.
- 375. Mahaffey KW, Granger CB, Sloan MA, et al. Neurosurgical evacuation of intracranial hemorrhage after thrombolytic therapy for acute myocardial infarction: experience from the GUSTO-I trial. Global Utilization of Streptokinase and tissue-plasminogen activator (tPA) for Occluded Coronary Arteries. Am Heart J 1999;138:493-9.
- 376. Sloan MA. Neurologic complications of thrombolytic therapy. In: Biller J, ed. Iatrogenic neurology. Boston, MA: Butterworth Heinemann; 1997:335-78.
- 377. Sloan MA, Sila CA, Mahaffey KW, et al. Prediction of 30-day mortality among patients with thrombolysis-related intracranial hemorrhage. *Circulation* 1998;98:1376-82.
- 378. The thrombolytic agent. In: Sherry S, ed. Fibrinolysis, thrombosis, and hemostasis: concepts, perspectives, and clinical applications. Philadelphia, PA: Lea & Febiger; 1992:119-60.
- 379. Armstrong PW, Collen D. Fibrinolysis for acute myocardial infarction: current status and new horizons for pharmacological reperfusion, part 1. *Circulation* 2001;103:2862-6.
- 380. Cannon CP, Gibson CM, McCabe CH, et al, for the Thrombolysis in Myocardial Infarction (TIMI) 10B Investigators. TNK-tissue plasminogen activator compared with front-loaded alteplase in acute myocardial infarction: results of the TIMI 10B trial. *Circulation* 1998;98:2805-14.
- 381. Mosby's Drug Consult 2004 Internet Subscription, 8th ed. Elsevier, 2003. Available at http://www.mosbysdrugconsult.com. Accessed February 27, 2004.
- 382. Antman EM. Hirudin in acute myocardial infarction: Thrombolysis and Thrombin Inhibition in Myocardial Infarction (TIMI) 9B trial. *Circulation* 1996;94:911-21.
- 383. The Global Use of Strategies to Open Occluded Coronary Arteries (GUSTO) IIb investigators. A comparison of recombinant hirudin with heparin for the treatment of acute coronary syndromes. N Engl J Med 1996;335:775-82.
- 384. Martin GV, Kennedy JW. Choice of thrombolytic agent. In: Julian DG, Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co Ltd; 1994:71-105.
- 385. Fuster V. Coronary thrombolysis: a perspective for the practicing physician. *N Engl J Med* 1993;329:723-5.
- 386. Simoons ML, Arnold AE. Tailored thrombolytic therapy: a perspective. *Circulation* 1993;88:2556-64.
- 387. White HD. Selecting a thrombolytic agent. *Cardiol Clin* 1995; 13:347-54.
- 388. French JK, Williams BF, Hart HH, et al. Prospective evaluation of eligibility for thrombolytic therapy in acute myocardial infarction. *BMJ* 1996;312:1637-41.
- 389. Hillis LD, Forman S, Braunwald E, for the Thrombolysis in Myocardial Infarction (TIMI) Phase II Co-Investigators. Risk stratification before thrombolytic therapy in patients with acute myocardial infarction. *J Am Coll Cardiol* 1990;16:313-5.
- 390. Normand ST, Glickman ME, Sharma RG, McNeil BJ. Using

admission characteristics to predict short-term mortality from myocardial infarction in elderly patients: results from the Cooperative Cardiovascular Project. *JAMA* 1996;275:1322-8.

- 391. Becker RC, Burns M, Gore JM, et al, for the National Registry of Myocardial Infarction (NRMI-2) Participants. Early assessment and in-hospital management of patients with acute myocardial infarction at increased risk for adverse outcomes: a nationwide perspective of current clinical practice. Am Heart J 1998;135:786-96.
- 392. Tu JV, Austin PC, Walld R, Roos L, Agras J, McDonald KM. Development and validation of the Ontario acute myocardial infarction mortality prediction rules. *J Am Coll Cardiol* 2001; 37:992-7.
- 393. Jacobs DR, Kroenke C, Crow R, et al. PREDICT: a simple risk score for clinical severity and long-term prognosis after hospitalization for acute myocardial infarction or unstable angina: the Minnesota heart survey. *Circulation* 1999;100:599-607.
- 394. Morrow DA, Antman EM, Parsons L, et al. Application of the TIMI risk score for ST-elevation MI in the National Registry of Myocardial Infarction 3. *JAMA* 2001;286:1356-9.
- 395. Gibson CM, Cannon CP, Murphy SA, et al. Relationship of TIMI myocardial perfusion grade to mortality after administration of thrombolytic drugs. *Circulation* 2000;101:125-30.
- 395a. Holper EM, Giugliano RP, Antman EM. Glycoprotein IIb/IIIa inhibitors in acute ST-segment myocardial infarction. *Coron Artery Dis* 1999;10:567-73.
- 396. Strategies for Patency Enhancement in the Emergency Department (SPEED) Group. Trial of abciximab with and without low-dose reteplase for acute myocardial infarction. *Circulation* 2000;101:2788-94.
- 397. Brener SJ, Zeymer U, Adgey AA, et al. Eptifibatide and low-dose tissue plasminogen activator in acute myocardial infarction: the integrilin and low-dose thrombolysis in acute myocardial infarction (INTRO AMI) trial. *J Am Coll Cardiol* 2002;39:377-86.
- 398. Savonitto S, Armstrong PW, Lincoff AM, et al. Risk of intracranial haemorrhage with combined fibrinolytic and glycoprotein IIb/IIIa inhibitor therapy in acute myocardial infarction: dichotomous response as a function of age in the GUSTO V trial. *Eur Heart J* 2003;24:1807-14.
- 399. Mortality at 1 year with combination platelet glycoprotein IIb/IIIa inhibition and reduced-dose fibrinolytic therapy vs conventional fibrinolytic therapy for acute myocardial infarction: GUSTO V randomized trial. *JAMA* 2002;288:2130-5.
- 400. Antman EM, Gibson CM, de Lemos JA, et al, for the Thrombolysis in Myocardial Infarction (TIMI) 14 Investigators. Combination reperfusion therapy with abciximab and reduced dose reteplase: results from TIMI 14. *Eur Heart J* 2000;21:1944-53.
- 401. Primary coronary angioplasty compared with intravenous thrombolytic therapy for acute myocardial infarction: six-month follow up and analysis of individual patient data from randomized trials. *Am Heart J* 2003;145:47-57.
- 402. Wilson SH, Bell MR, Rihal CS, Bailey KR, Holmes DR, Berger PB. Infarct artery reocclusion after primary angioplasty, stent placement, and thrombolytic therapy for acute myocardial infarction. *Am Heart J* 2001;141:704-10.
- 403. Grzybowski M, Clements EA, Parsons L, et al. Mortality benefit of immediate revascularization of acute ST-segment elevation myocardial infarction in patients with contraindications to thrombolytic therapy: a propensity analysis. *JAMA* 2003;290:1891-8.
- 404. Zijlstra F, de Boer MJ, Hoorntje JC, Reiffers S, Reiber JH, Suryapranata H. A comparison of immediate coronary angioplas-

ty with intravenous streptokinase in acute myocardial infarction. *N Engl J Med* 1993;328:680-4.

- 405. Ribeiro EE, Silva LA, Carneiro R, et al. Randomized trial of direct coronary angioplasty versus intravenous streptokinase in acute myocardial infarction. *J Am Coll Cardiol* 1993;22:376-80.
- 406. Grinfeld L, Berrocal D, Bellardi J, et al. Fibrinolytics versus primary angioplasty in acute myocardial infarction (FAP): a randomized trial in a community hospital in Argentina. *J Am Coll Cardiol* 1996;27:A222.
- 407. Zijlstra F, Beukema WP, van't Hof AW, et al. Randomized comparison of primary coronary angioplasty with thrombolytic therapy in low risk patients with acute myocardial infarction. *J Am Coll Cardiol* 1997;29:908-12.
- 408. Widimský P, Groch L, Zelízko M, Aschermann M, Bednár F, Suryapranata H. Multicentre randomized trial comparing transport to primary angioplasty vs immediate thrombolysis vs combined strategy for patients with acute myocardial infarction presenting to a community hospital without a catheterization laboratory. The PRAGUE study. *Eur Heart J* 2000;21:823-31.
- 409. de Boer MJ, Ottervanger JP, van't Hof AW, Hoorntje JC, Suryapranata H, Zijlstra F, for the Zwolle Myocardial Infarction Study Group. Reperfusion therapy in elderly patients with acute myocardial infarction: a randomized comparison of primary angioplasty and thrombolytic therapy. *J Am Coll Cardiol* 2002;39:1723-8.
- 410. Akhras F, Ousa AA, Swann G, et al . Primary coronary angioplasty or intravenous thrombolysis for patients with acute myocardial infarction? Acute and late follow-up results in a new cardiac unit [abstr]. *J Am Coll Cardiol* 1997;29:A235-6.
- 411. Deleted in press.
- 412. Grines CL, Browne KF, Marco J, et al, for the Primary Angioplasty in Myocardial Infarction Study Group. A comparison of immediate angioplasty with thrombolytic therapy for acute myocardial infarction. *N Engl J Med* 1993;328:673-9.
- 413. Gibbons RJ, Holmes DR, Reeder GS, Bailey KR, Hopfenspirger MR, Gersh BJ, for the Mayo Coronary Care Unit and Catheterization Laboratory Groups. Immediate angioplasty compared with the administration of a thrombolytic agent followed by conservative treatment for myocardial infarction. *N Engl J Med* 1993;328:685-91.
- 414. Ribichini F, Steffenino G, Dellavalle A, et al. Comparison of thrombolytic therapy and primary coronary angioplasty with liberal stenting for inferior myocardial infarction with precordial ST-segment depression: immediate and long-term results of a randomized study. *J Am Coll Cardiol* 1998;32:1687-94.
- 415. García E, Elízaga J, Pérez-Castellano N, et al. Primary angioplasty versus systemic thrombolysis in anterior myocardial infarction. *J Am Coll Cardiol* 1999;33:605-11.
- 416. The Global Use of Strategies to Open Occluded Coronary Arteries in Acute Coronary Syndromes (GUSTO IIb) Angioplasty Substudy Investigators. A clinical trial comparing primary coronary angioplasty with tissue plasminogen activator for acute myocardial infarction. *N Engl J Med* 1997;336:1621-8.
- 417. Le May MR, Labinaz M, Davies RF, et al. Stenting versus thrombolysis in acute myocardial infarction trial (STAT). *J Am Coll Cardiol* 2001;37:985-91.
- 418. Schömig A, Kastrati A, Dirschinger J, et al, for the Stent versus Thrombolysis for Occluded Coronary Arteries in Patients with Acute Myocardial Infarction Study Investigators. Coronary stenting plus platelet glycoprotein IIb/IIIa blockade compared with tissue plasminogen activator in acute myocardial infarction. *N Engl J Med* 2000;343:385-91.

- 419. Vermeer F, Oude Ophuis AJ, vd Berg EJ, et al. Prospective randomised comparison between thrombolysis, rescue PTCA, and primary PTCA in patients with extensive myocardial infarction admitted to a hospital without PTCA facilities: a safety and feasibility study. *Heart* 1999;82:426-31.
- 420. Myocardial salvage after coronary stenting plus abciximab versus fibrinolysis plus abciximab in patients with acute myocardial infarction: a randomised trial. *Lancet* 2002;359:920-5.
- 421. Grines CL, Westerhausen DR Jr, Grines LL,, et al, for the Air PAMI Study Group. A randomized trial of transfer for primary angioplasty versus on-site thrombolysis in patients with high-risk myocardial infarction: the Air Primary Angioplasty in Myocardial Infarction study. J Am Coll Cardiol 2002;39:1713-9.
- 421a. Melandri G. The obsession with primary angioplasty. *Circulation* 2003;108:e162.
- 422. Rogers WJ, Dean LS, Moore PB, Wool KJ, Burgard SL, Bradley EL, for the Alabama Registry of Myocardial Ischemia Investigators. Comparison of primary angioplasty versus thrombolytic therapy for acute myocardial infarction. *Am J Cardiol* 1994;74:111-8.
- 423. Every NR, Parsons LS, Hlatky M, Martin JS, Weaver WD, for the Myocardial Infarction Triage and Intervention Investigators. A comparison of thrombolytic therapy with primary coronary angioplasty for acute myocardial infarction. N Engl J Med 1996;335: 1253-60.
- 424. Tiefenbrunn AJ, Chandra NC, French WJ, Gore JM, Rogers WJ. Clinical experience with primary percutaneous transluminal coronary angioplasty compared with alteplase (recombinant tissuetype plasminogen activator) in patients with acute myocardial infarction: a report from the Second National Registry of Myocardial Infarction (NRMI-2). J Am Coll Cardiol 1998;31: 1240-5.
- 425. Danchin N, Vaur L, Genès N, et al. Treatment of acute myocardial infarction by primary coronary angioplasty or intravenous thrombolysis in the "real world": one-year results from a nationwide French survey. *Circulation* 1999;99:2639-44.
- 426. Stone GW, Grines CL, Browne KF, et al. Influence of acute myocardial infarction location on in-hospital and late outcome after primary percutaneous transluminal coronary angioplasty versus tissue plasminogen activator therapy. *Am J Cardiol* 1996;78:19-25.
- 427. van't Hof AW, Henriques J, Ottervanger J-P, et al. No mortality benefit of primary angioplasty over thrombolytic therapy in patients with nonanterior myocardial infarction at long-term follow-up: results of the Zwolle trial [abstr]. *J Am Coll Cardiol* 2003;41:369A.
- 428. Brodie BR, Stuckey TD, Hansen C, Muncy D. Benefit of coronary reperfusion before intervention on outcomes after primary angioplasty for acute myocardial infarction. *Am J Cardiol* 2000;85:13-8.
- 429. Juliard JM, Feldman LJ, Golmard JL, et al. Relation of mortality of primary angioplasty during acute myocardial infarction to door-to-Thrombolysis In Myocardial Infarction (TIMI) time. *Am J Cardiol* 2003;91:1401-5.
- 430. Ellis S. Elective coronary intervention: approach, technique, and complications. In: Topol EJ, ed. Textbook of interventional cardiology, 4th ed., Philadelphia, PA: WB Saunders Co Ltd; 2003:163-181.
- 431. Levine GN, Kern MJ, Berger PB, et al, for the American Heart Association Diagnostic and Interventional Catheterization Committee and Council on Clinical Cardiology. Management of patients undergoing percutaneous coronary revascularization. Ann

Intern Med 2003;139:123-36.

- 432. Smith SC Jr, Dove JT, Jacobs AK, et al. ACC/AHA guidelines for percutaneous coronary intervention (revision of the 1993 PTCA guidelines): a report of the American College of Cardiology/ American Heart Association Task Force on Practice Guidelines (Committee to Revise the 1993 Guidelines for Percutaneous Transluminal Coronary Angioplasty). Available at http:// www.acc.org/clinical/guidelines/percutaneous/dirIndex.ht, 2001. Accessed December 16, 2003.
- 433. Cragg DR, Friedman HZ, Bonema JD, et al. Outcome of patients with acute myocardial infarction who are ineligible for thrombolytic therapy. *Ann Intern Med* 1991;115:173-7.
- 434. Brodie BR, Weintraub RA, Stuckey TD, et al. Outcomes of direct coronary angioplasty for acute myocardial infarction in candidates and non-candidates for thrombolytic therapy. *Am J Cardiol* 1991;67:7-12.
- 435. Himbert D, Juliard JM, Steg PG, et al. Primary coronary angioplasty for acute myocardial infarction with contraindication to thrombolysis. *Am J Cardiol* 1993;71:377-81.
- 436. Zahn R, Schuster S, Schiele R, et al, for the Maximal Individual Therapy in Acute Myocardial Infarction (MITRA) Study Group. Comparison of primary angioplasty with conservative therapy in patients with acute myocardial infarction and contraindications for thrombolytic therapy. *Catheter Cardiovasc Interv* 1999;46:127-33.
- 437. Iannone LA, Anderson SM, Phillips SJ. Coronary angioplasty for acute myocardial infarction in a hospital without cardiac surgery. *Tex Heart Inst J* 1993;20:99-104.
- 438. Weaver WD, Litwin PE, Martin JS, for the Myocardial Infarction, Triage, and Intervention Project Investigators. Use of direct angioplasty for treatment of patients with acute myocardial infarction in hospitals with and without on-site cardiac surgery. *Circulation* 1993;88:2067-75.
- 439. Weaver WD, Parsons L, Every N, for the MITI project investigators. Primary coronary angioplasty in hospitals with and without surgery backup. *J Invasive Cardiol* 1995;7:34F-39F.
- 440. Brush JE, Thompson S, Ciuffo AA, et al. Retrospective comparison of a strategy of primary coronary angioplasty versus intravenous thrombolytic therapy for acute myocardial infarction in a community hospital without cardiac surgical backup. *J Invasive Cardiol* 1996;8:91-98.
- 441. Moquet B, Huguet RG, Cami G, et al. [Primary angioplasty in acute myocardial infarction: a one-year experience in a small urban community.] *Arch Mal Coeur Vaiss* 1997;90:11-5.
- 442. Smyth DW, Richards AM, Elliott JM. Direct angioplasty for myocardial infarction: one-year experience in a center with surgical back-up 220 miles away. *J Invasive Cardiol* 1997;9:324-332.
- 443. Ribichini F. Experiences with primary angioplasty without on site-cardiac surgery. *Semin Interv Cardiol* 1999;4:47-53.
- 444. Primary Angioplasty in Acute Myocardial Infarction at Hospitals With No Surgery On-Site (the PAMI-No SOS study) versus transfer to surgical centers for primary angioplasty. *J Am Coll Cardiol* 2004;43:1943-50.
- 445. Wharton TP, McNamara NS, Fedele FA, Jacobs MI, Gladstone AR, Funk EJ. Primary angioplasty for the treatment of acute myocardial infarction: experience at two community hospitals without cardiac surgery. *J Am Coll Cardiol* 1999;33:1257-65.
- 446. Singh M, Garratt KN, Lennon RJ, et al. Emergent angioplasty for acute myocardial infarction at a community hospital without onsite cardiac surgery [abstr]. *J Am Coll Cardiol* 2002;39:40A.
- 447. Sanborn TA, Jacobs AK, Frederick PD, Every NR, French WJ. Nationwide emergent coronary interventions (primary PCI) in

patients with acute myocardial infarction in hospitals with and without on-site cardiac surgery: a report from the National Registry of Myocardial Infarction (NRMI) [abstr]. *Circulation* 2002;106:II-333.

- 448. Aversano T. Primary angioplasty at hospitals without cardiac surgery: C-PORT Registry outcomes [abstr]. *Circulation* 2003;108: IV-613.
- 449. Singh M, Ting HH, Berger PB, Garratt KN, Holmes DR, Gersh BJ. Rationale for on-site cardiac surgery for primary angioplasty: a time for reappraisal. *J Am Coll Cardiol* 2002;39:1881-9.
- 450. Kutcher MA, Klein LW, Wharton TP, et al. Clinical outcomes in coronary angioplasty centers with off-site versus on-site cardiac surgery capabilities: a preliminary report from the American College of Cardiology—National Cardiovascular Data Registry [abstr]. J Am Coll Cardiol 2004;43:96a.
- 451. Deleted in press.
- 452. Lotfi M, Mackie K, Dzavik V, Seidelin PH. Impact of delays to cardiac surgery after failed angioplasty and stenting. *J Am Coll Cardiol* 2004;43:337-42.
- 453. Dehmer GJ, Gantt DS. Coronary intervention at hospitals without on-site cardiac surgery: are we pushing the envelope too far? *J Am Coll Cardiol* 2004;43:343-5.
- 454. Zijlstra F, van't Hof AW, Liem AL, Hoorntje JC, Suryapranata H, de Boer MJ. Transferring patients for primary angioplasty: a retrospective analysis of 104 selected high-risk patients with acute myocardial infarction. *Heart* 1997;78:333-6.
- 455. Zahn R, Schiele R, Seidl K, et al. Primary angioplasty in acute myocardial infarction: differences between referred patients and those treated in hospitals with on-site facilities? *J Invasive Cardiol* 1999;11:213-219.
- 456. Straumann E, Yoon S, Naegeli B, et al. Hospital transfer for primary coronary angioplasty in high-risk patients with acute myocardial infarction. *Heart* 1999;82:415-9.
- 457. Liem AL, van 't Hof AW, Hoorntje JC, de Boer MJ, Suryapranata H, Zijlstra F. Influence of treatment delay on infaret size and clinical outcome in patients with acute myocardial infarction treated with primary angioplasty. J Am Coll Cardiol 1998;32:629-33.
- 458. NRMI-4 Investigators: The National Registry of Myocardial Infarction-4 Quarterly Report. Genentech, South San Francisco, CA, March 2003:2.
- 459. Deleted in press.
- 460. Antoniucci D, Santoro GM, Bolognese L, Valenti R, Trapani M, Fazzini PF. A clinical trial comparing primary stenting of the infarct-related artery with optimal primary angioplasty for acute myocardial infarction: results from the Florence Randomized Elective Stenting in Acute Coronary Occlusions (FRESCO) trial. *J Am Coll Cardiol* 1998;31:1234-9.
- 461. Rodríguez A, Bernardi V, Fernández M, et al. In-hospital and late results of coronary stents versus conventional balloon angioplasty in acute myocardial infarction (GRAMI trial). Gianturco-Roubin in Acute Myocardial Infarction. Am J Cardiol 1998;81:1286-91.
- 462. Suryapranata H, Ottervanger JP, Nibbering E, et al. Long term outcome and cost-effectiveness of stenting versus balloon angioplasty for acute myocardial infarction. *Heart* 2001;85:667-71.
- 463. Saito S, Hosokawa G, Tanaka S, Nakamura S, for the PASTA Trial Investigators. Primary stent implantation is superior to balloon angioplasty in acute myocardial infarction: final results of the primary angioplasty versus stent implantation in acute myocardial infarction (PASTA) trial. *Catheter Cardiovasc Interv* 1999; 48:262-8.
- 464. Grines CL, Cox DA, Stone GW, et al, for the Stent Primary

Angioplasty in Myocardial Infarction Study Group. Coronary angioplasty with or without stent implantation for acute myocardial infarction. *N Engl J Med* 1999;341:1949-56.

- 465. Kawashima A, Ueda K, Nishida Y, et al. Quantitative angiographic analysis of restenosis of primary stenting using Wiktor stent for acute myocardial infarction: results from a multicenter randomised PRISAM study [abstr]. *Circulation* 1999;100:I-856.
- 466. Maillard L, Hamon M, Khalife K, et al, for the STENTIM-2 Investigators. A comparison of systematic stenting and conventional balloon angioplasty during primary percutaneous transluminal coronary angioplasty for acute myocardial infarction. *J Am Coll Cardiol* 2000;35:1729-36.
- 467. Scheller B, Hennen B, Severin-Kneib S, Ozbek C, Schieffer H, Markwirth T. Long-term follow-up of a randomized study of primary stenting versus angioplasty in acute myocardial infarction. *Am J Med* 2001;110:1-6.
- 468. Lemos PA, Saia F, Hofma SH, et al. Short- and long-term clinical benefit of sirolimus-eluting stents compared to conventional bare stents for patients with acute myocardial infarction. *J Am Coll Cardiol* 2004;43:704-8.
- 469. Gibson CM. A union in reperfusion: the concept of facilitated percutaneous coronary intervention. *J Am Coll Cardiol* 2000;36: 1497-9.
- 470. Herrmann HC, Moliterno DJ, Ohman EM, et al. Facilitation of early percutaneous coronary intervention after reteplase with or without abciximab in acute myocardial infarction: results from the SPEED (GUSTO-4 Pilot) Trial. *J Am Coll Cardiol* 2000;36:1489-96.
- 471. Ross AM, Coyne KS, Reiner JS, et al, for the Plasminogen-activator Angioplasty Compatibility Trial (PACT) investigators. A randomized trial comparing primary angioplasty with a strategy of short-acting thrombolysis and immediate planned rescue angioplasty in acute myocardial infarction: the PACT trial. *J Am Coll Cardiol* 1999;34:1954-62.
- 472. Fernandez-Aviles F, Alonso JJ, Castro-Beiras A, et al. Prospective randomized trial comparing a routine invasive strategy within 24 hours to thrombolysis versus an ischemia-guided conservative approach to acute myocardial infarction with ST-segment elevation: the Gracia-1 trial. *Lancet*. In press.
- 473. Kastrati A, Mehilli J, Schlotterbeck K, et al, for the Bavarian Reperfusion Alternatives Evaluation (BRAVE) Study Investigators. Early administration of reteplase plus abciximab vs abciximab alone in patients with acute myocardial infarction referred for percutaneous coronary intervention: a randomized controlled trial. *JAMA*. 2004;291:947-54.
- 473a. Ellis SG, Armstrong P, Betriu A, et al. Facilitated percutaneous coronary intervention versus primary percutaneous coronary intervention: design and rationale of the Facilitated Intervention with Enhanced Reperfusion Speed to Stop Events (FINESSE) trial. *Am Heart J* 2004;147:E16.
- 474. Granger CB, White HD, Bates ER, Ohman EM, Califf RM. A pooled analysis of coronary arterial patency and left ventricular function after intravenous thrombolysis for acute myocardial infarction. *Am J Cardiol* 1994;74:1220-8.
- 475. Lange RA, Cigarroa RG, Wells PJ, Kremers MS, Hills LD. Influence of anterograde flow in the infarct artery on the incidence of late potentials after acute myocardial infarction. *Am J Cardiol* 1990;65:554-8.
- 476. Cigarroa RG, Lange RA, Hillis LD. Prognosis after acute myocardial infarction in patients with and without residual anterograde coronary blood flow. *Am J Cardiol* 1989;64:155-60.
- 477. Ellis SG, da Silva ER, Heyndrickx G, et al. Randomized compar-

ison of rescue angioplasty with conservative management of patients with early failure of thrombolysis for acute anterior myocardial infarction. *Circulation* 1994;90:2280-4.

- 478. Goldman LE, Eisenberg MJ. Identification and management of patients with failed thrombolysis after acute myocardial infarction. *Ann Intern Med* 2000;132:556-65.
- 479. The TIMI Research Group. Immediate vs delayed catheterization and angioplasty following thrombolytic therapy for acute myocardial infarction: TIMI II A results. *JAMA* 1988;260:2849-58.
- 480. Topol EJ, Califf RM, George BS, et al. A randomized trial of immediate versus delayed elective angioplasty after intravenous tissue plasminogen activator in acute myocardial infarction. N Engl J Med 1987;317:581-8.
- 481. Califf RM, Topol EJ, Stack RS, et al, for the TAMI Study Group. Evaluation of combination thrombolytic therapy and timing of cardiac catheterization in acute myocardial infarction: results of thrombolysis and angioplasty in myocardial infarction: phase 5 randomized trial. *Circulation* 1991;83:1543-56.
- 482. Califf RM, Topol EJ, George BS, et al. Characteristics and outcome of patients in whom reperfusion with intravenous tissuetype plasminogen activator fails: results of the Thrombolysis and Angioplasty in Myocardial Infarction (TAMI) I trial. *Circulation* 1988;77:1090-9.
- 483. Lee L, Bates ER, Pitt B, Walton JA, Laufer N, O'Neill WW. Percutaneous transluminal coronary angioplasty improves survival in acute myocardial infarction complicated by cardiogenic shock. *Circulation* 1988;78:1345-51.
- 484. Jeremy RW, Hackworthy RA, Bautovich G, Hutton BF, Harris PJ. Infarct artery perfusion and changes in left ventricular volume in the month after acute myocardial infarction. J Am Coll Cardiol 1987;9:989-95.
- 485. Kersschot IE, Brugada P, Ramentol M, et al. Effects of early reperfusion in acute myocardial infarction on arrhythmias induced by programmed stimulation: a prospective, randomized study. J Am Coll Cardiol 1986;7:1234-42.
- 486. Stadius ML, Davis K, Maynard C, Ritchie JL, Kennedy JW. Risk stratification for 1-year survival based on characteristics identified in the early hours of acute myocardial infarction. The Western Washington Intracoronary Streptokinase Trial. *Circulation* 1986;74:703-11.
- 487. Topol EJ, Califf RM, Vandormael M, et al, for the Thrombolysis and Angioplasty in Myocardial Infarction-6 Study Group. A randomized trial of late reperfusion therapy for acute myocardial infarction. *Circulation* 1992;85:2090-9.
- 488. Dzavik V, Beanlands DS, Davies RF, et al. Effects of late percutaneous transluminal coronary angioplasty of an occluded infarct-related coronary artery on left ventricular function in patients with a recent (less than 6 weeks) Q-wave acute myocardial infarction (Total Occlusion Post-Myocardial Infarction Intervention Study [TOMIIS]: a pilot study). *Am J Cardiol* 1994;73:856-61.
- 489. Zeymer U, Uebis R, Vogt A, et al, for the ALKK-Study Group. Randomized comparison of percutaneous transluminal coronary angioplasty and medical therapy in stable survivors of acute myocardial infarction with single vessel disease: a study of the Arbeitsgemeinschaft Leitende Kardiologische Krankenhausärzte. *Circulation* 2003;108:1324-8.
- 490. Horie H, Takahashi M, Minai K, et al. Long-term beneficial effect of late reperfusion for acute anterior myocardial infarction with percutaneous transluminal coronary angioplasty. *Circulation* 1998;98:2377-82.
- 491. Yousef ZR, Redwood SR, Bucknall CA, Sulke AN, Marber MS. Late intervention after anterior myocardial infarction: effects on

left ventricular size, function, quality of life, and exercise tolerance: results of the Open Artery Trial (TOAT Study). *J Am Coll Cardiol* 2002;40:869-76.

- 492. Deleted in press.
- 493. Sadanandan S, Buller C, Menon V, et al. The late open artery hypothesis: a decade later. *Am Heart J* 2001;142:411-21.
- 494. Hochman JS, Boland J, Sleeper LA, et al, for the SHOCK Registry Investigators. Current spectrum of cardiogenic shock and effect of early revascularization on mortality: results of an international registry. *Circulation* 1995;91:873-81.
- 495. Urban P, Stauffer JC, Bleed D, et al. A randomized evaluation of early revascularization to treat shock complicating acute myocardial infarction. The (Swiss) Multicenter Trial of Angioplasty for Shock-(S)MASH. *Eur Heart J* 1999;20:1030-8.
- 496. Dzavik V, Sleeper LA, Cocke TP, et al, for the SHOCK Investigators. Early revascularization is associated with improved survival in elderly patients with acute myocardial infarction complicated by cardiogenic shock: a report from the SHOCK trial registry. *Eur Heart J* 2003;24:828-37.
- 497. Dauerman HL, Goldberg RJ, Malinski M, Yarzebski J, Lessard D, Gore JM. Outcomes and early revascularization for patients greater than or equal to 65 years of age with cardiogenic shock. *Am J Cardiol* 2001;87:844-8.
- 498. Dauerman HL, Ryan TJ, Piper WD, et al. Outcomes of percutaneous coronary intervention among elderly patients in cardiogenic shock: a multicenter, decade-long experience. *J Invasive Cardiol* 2003;15:380-4.
- 499. Chan AW, Chew DP, Bhatt DL, Moliterno DJ, Topol EJ, Ellis SG. Long-term mortality benefit with the combination of stents and abciximab for cardiogenic shock complicating acute myocardial infarction. *Am J Cardiol* 2002;89:132-6.
- 500. Giri S, Mitchel J, Azar RR, et al. Results of primary percutaneous transluminal coronary angioplasty plus abciximab with or without stenting for acute myocardial infarction complicated by cardiogenic shock. *Am J Cardiol* 2002;89:126-31.
- 501. Antoniucci D, Valenti R, Migliorini A, et al. Abciximab therapy improves survival in patients with acute myocardial infarction complicated by early cardiogenic shock undergoing coronary artery stent implantation. *Am J Cardiol* 2002;90:353-7.
- 502. Hochman JS. Cardiogenic shock complicating acute myocardial infarction: expanding the paradigm. *Circulation* 2003;107:2998-3002.
- 503. Simoons ML, Arnold AE, Betriu A, et al. Thrombolysis with tissue plasminogen activator in acute myocardial infarction: no additional benefit from immediate percutaneous coronary angioplasty. *Lancet* 1988;1:197-203.
- 504. Scheller B, Hennen B, Hammer B, et al, for the SIAM III Study Group. Beneficial effects of immediate stenting after thrombolysis in acute myocardial infarction. *J Am Coll Cardiol* 2003; 42:634-41.
- 505. Should We Intervene Following Thrombolysis? (SWIFT) Trial Study Group. SWIFT trial of delayed elective intervention vs conservative treatment after thrombolysis with anistreplase in acute myocardial infarction. *BMJ* 1991;302:555-60.
- 506. Williams DO, Braunwald E, Knatterud G, et al. One-year results of the Thrombolysis in Myocardial Infarction investigation (TIMI) Phase II Trial. *Circulation* 1992;85:533-42.
- 507. Terrin ML, Williams DO, Kleiman NS, et al. Two- and three-year results of the Thrombolysis in Myocardial Infarction (TIMI) Phase II clinical trial. *J Am Coll Cardiol* 1993;22:1763-72.
- 508. Stenestrand U, Wallentin L. Early revascularisation and 1-year survival in 14-day survivors of acute myocardial infarction: a

prospective cohort study. Lancet 2002;359:1805-11.

- 509. Alter DA, Tu JV, Austin PC, Naylor CD. Waiting times, revascularization modality, and outcomes after acute myocardial infarction at hospitals with and without on-site revascularization facilities in Canada. *J Am Coll Cardiol* 2003;42:410-9.
- 510. Zeymer U, Uebis R, Vogt A, et al, for the ALKK-Study Group. Randomized comparison of percutaneous transluminal coronary angioplasty and medical therapy in stable survivors of acute myocardial infarction with single-vessel disease: a study of the Arbeitsgemeinschaft Leitende Kardiologische Krankenhausärzte. *Circulation* 2003;108:1324-8.
- 511. Gupta M, Chang WC, Van de Werf F, et al, for the ASSENT II Investigators. International differences in in-hospital revascularization and outcomes following acute myocardial infarction: a multilevel analysis of patients in ASSENT-2. *Eur Heart J* 2003; 24:1640-50.
- 512. Gibson CM, Karha J, Murphy SA, et al, for the TIMI Study Group. Early and long-term clinical outcomes associated with reinfarction following fibrinolytic administration in the thrombolysis in myocardial infarction trials. *J Am Coll Cardiol* 2003;42:7-16.
- 513. Deleted in press.
- 514. Deleted in press.
- 515. Madsen JK, Grande P, Saunamäki K, et al. Danish multicenter randomized study of invasive versus conservative treatment in patients with inducible ischemia after thrombolysis in acute myocardial infarction (DANAMI). DANish trial in Acute Myocardial Infarction. *Circulation* 1997;96:748-55.
- 516. Barbash GI, Roth A, Hod H, et al. Randomized controlled trial of late in-hospital angiography and angioplasty versus conservative management after treatment with recombinant tissue-type plasminogen activator in acute myocardial infarction. *Am J Cardiol* 1990;66:538-45.
- 517. Ellis SG, Mooney MR, George BS, et al, for the Treatment of Post-Thrombolytic Stenoses (TOPS) Study Group. Randomized trial of late elective angioplasty versus conservative management for patients with residual stenoses after thrombolytic treatment of myocardial infarction. *Circulation* 1992;86:1400-6.
- 518. Eagle KA, Guyton RA, Davidoff R, et al. ACC/AHA 2004 guideline update for coronary artery bypass graft surgery: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the 1999 Guidelines for Coronary Artery Bypass Graft Surgery). J Am Coll Cardiol In press.
- 519. Deleted in press.
- 520. Caracciolo EA, Davis KB, Sopko G, et al. Comparison of surgical and medical group survival in patients with left main coronary artery disease: long-term CASS experience. *Circulation* 1995;91:2325-34.
- 521. Davis KB, Chaitman B, Ryan T, Bittner V, Kennedy JW. Comparison of 15-year survival for men and women after initial medical or surgical treatment for coronary artery disease: a CASS registry study. Coronary Artery Surgery Study. *J Am Coll Cardiol* 1995;25:1000-9.
- 522. Stone GW, Brodie BR, Griffin JJ, et al. Role of cardiac surgery in the hospital phase management of patients treated with primary angioplasty for acute myocardial infarction. *Am J Cardiol* 2000;85:1292-6.
- 523. Hochman JS, Buller CE, Sleeper LA, et al. Cardiogenic shock complicating acute myocardial infarction: etiologies, management and outcome: a report from the SHOCK Trial Registry. SHould we emergently revascularize Occluded Coronaries for

cardiogenic shocK? J Am Coll Cardiol 2000;36:1063-70.

- 524. Hochman JS, Gersh B. Acute myocardial infarction. In: Topol EJ, ed. Textbook of cardiovascular medicine. Philadelphia, PA: Lippincott-Raven Publishers; 1998:421-62.
- 525. DeBusk RF. Specialized testing after recent acute myocardial infarction. *Ann Intern Med* 1989;110:470-81.
- 526. Holmes DR, Bates ER, Kleiman NS, et al, for the Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries (GUSTO-I) Investigators. Contemporary reperfusion therapy for cardiogenic shock: the GUSTO-I trial experience. J Am Coll Cardiol 1995;26:668-74.
- 527. de Lemos JA, Braunwald E. ST segment resolution as a tool for assessing the efficacy of reperfusion therapy. *J Am Coll Cardiol* 2001;38:1283-94.
- 528. Matetzky S, Novikov M, Gruberg L, et al. The significance of persistent ST elevation versus early resolution of ST segment elevation after primary PTCA. J Am Coll Cardiol 1999;34:1932-8.
- 529. Schröder R, Dissmann R, Brüggemann T, et al. Extent of early ST segment elevation resolution: a simple but strong predictor of outcome in patients with acute myocardial infarction. *J Am Coll Cardiol* 1994;24:384-91.
- 530. Fu Y, Goodman S, Chang WC, Van De Werf F, Granger CB, Armstrong PW. Time to treatment influences the impact of STsegment resolution on one-year prognosis: insights from the assessment of the safety and efficacy of a new thrombolytic (ASSENT-2) trial. *Circulation* 2001;104:2653-9.
- 531. de Lemos JA, Antman EM, Gibson CM, et al. Abciximab improves both epicardial flow and myocardial reperfusion in STelevation myocardial infarction: observations from the TIMI 14 trial. *Circulation* 2000;101:239-43.
- 532. Santoro GM, Antoniucci D, Valenti R, et al. Rapid reduction of ST-segment elevation after successful direct angioplasty in acute myocardial infarction. *Am J Cardiol* 1997;80:685-9.
- 533. Antman EM. The search for replacements for unfractionated heparin. *Circulation* 2001;103:2310-4.
- 534. Eisenberg PR. Role of heparin in coronary thrombolysis. *Chest* 1992;101:131S-139S.
- 535. Rao AK, Pratt C, Berke A, Jaffe A, et al. Thrombolysis in Myocardial Infarction (TIMI) Trial: phase I–hemorrhagic manifestations and changes in plasma fibrinogen and the fibrinolytic system in patients treated with recombinant tissue plasminogen activator and streptokinase. *J Am Coll Cardiol* 1988;11:1-11.
- 536. Popma JJ, Califf RM, Ellis SG, et al. Mechanism of benefit of combination thrombolytic therapy for acute myocardial infarction: a quantitative angiographic and hematologic study. *J Am Coll Cardiol* 1992;20:1305-12.
- 537. Collins R, Peto R, Baigent C, Sleight P. Aspirin, heparin, and fibrinolytic therapy in suspected acute myocardial infarction. *N Engl J Med* 1997;336:847-60.
- 538. The GUSTO Angiographic Investigators. The effects of tissue plasminogen activator, streptokinase, or both on coronary-artery patency, ventricular function, and survival after acute myocardial infarction. *N Engl J Med* 1993;329:1615-22.
- 539. Col J, Decoster O, Hanique G, Deligne B, Boland J, Pirenne B. Infusion of heparin conjunct to streptokinase accelerates reperfusion of acute myocardial infarction: results of a double-blind randomized study (OSIRIS). *Circulation* 1992;86:259.
- 540. Melandri G, Branzi A, Semprini F, Cervi V, Galiè N, Magnani B. Enhanced thrombolytic efficacy and reduction of infarct size by simultaneous infusion of streptokinase and heparin. *Br Heart J* 1990;64:118-20.
- 541. White HD, Yusuf S. Issues regarding the use of heparin following

streptokinase therapy. J Thromb Thrombolysis 1995;2:5-10.

- 542. Simoons M, Krzeminska-Pakula M, Alonso A, et al, for the AMI-SK Investigators. Improved reperfusion and clinical outcome with enoxaparin as an adjunct to streptokinase thrombolysis in acute myocardial infarction: the AMI-SK study. *Eur Heart J* 2002;23:1282-90.
- 543. Bleich SD, Nichols TC, Schumacher RR, Cooke DH, Tate DA, Teichman SL. Effect of heparin on coronary arterial patency after thrombolysis with tissue plasminogen activator in acute myocardial infarction. *Am J Cardiol* 1990;66:1412-7.
- 544. de Bono DP, Simoons ML, Tijssen J, et al. Effect of early intravenous heparin on coronary patency, infarct size, and bleeding complications after alteplase thrombolysis: results of a randomised double-blind European Cooperative Study Group trial. *Br Heart J* 1992;67:122-8.
- 545. Hsia J, Hamilton WP, Kleiman N, Roberts R, Chaitman BR, Ross AM, for the Heparin-Aspirin Reperfusion Trial (HART) Investigators. A comparison between heparin and low-dose aspirin as adjunctive therapy with tissue plasminogen activator for acute myocardial infarction. *N Engl J Med* 1990;323:1433-7.
- 546. Mahaffey KW, Granger CB, Collins R, et al. Overview of randomized trials of intravenous heparin in patients with acute myocardial infarction treated with thrombolytic therapy. *Am J Cardiol* 1996;77:551-6.
- 547. Ogilby JD, Kopelman HA, Klein LW, Agarwal JB. Adequate heparinization during PTCA: assessment using activated clotting times. *Cathet Cardiovasc Diagn* 1989;18:206-9.
- 548. Narins CR, Hillegass WB, Nelson CL, et al. Relation between activated clotting time during angioplasty and abrupt closure. *Circulation* 1996;93:667-71.
- 549. Granger CB, Hirsch J, Califf RM, et al. Activated partial thromboplastin time and outcome after thrombolytic therapy for acute myocardial infarction: results from the GUSTO-I trial. *Circulation* 1996;93:870-8.
- 550. Menon V, Berkowitz SD, Antman EM, Fuchs RM, Hochman JS. New heparin dosing recommendations for patients with acute coronary syndromes. *Am J Med* 2001;110:641-50.
- 551. Hirsh J, Hoak J. Management of deep vein thrombosis and pulmonary embolism: a statement for healthcare professionals. Council on Thrombosis (in consultation with the Council on Cardiovascular Radiology), American Heart Association. *Circulation* 1996;93:2212-45.
- 552. Granger CB, Becker R, Tracy RP, et al, for the Global Utilization of Streptokinase and TPA for Occluded Coronary Arteries (GUSTO-I) Hemostasis Substudy Group. Thrombin generation, inhibition and clinical outcomes in patients with acute myocardial infarction treated with thrombolytic therapy and heparin: results from the GUSTO-I Trial. *J Am Coll Cardiol* 1998;31:497-505.
- 553. Antman EM. Hirudin in acute myocardial infarction: a safety report from the Thrombolysis and Thrombin Inhibition in Myocardial Infarction (TIMI) 9A trial. *Circulation* 1994;90:1624-30.
- 554. Thompson PL, Aylward PE, Federman J, et al, for the National Heart Foundation of Australia Coronary Thrombolysis Group. A randomized comparison of intravenous heparin with oral aspirin and dipyridamole 24 hours after recombinant tissue-type plasminogen activator for acute myocardial infarction. *Circulation* 1991;83:1534-42.
- 555. Granger CB, Miller JM, Bovill EG, et al. Rebound increase in thrombin generation and activity after cessation of intravenous heparin in patients with acute coronary syndromes. *Circulation* 1995;91:1929-35.

- 556. Théroux P, Waters D, Lam J, Juneau M, McCans J. Reactivation of unstable angina after the discontinuation of heparin. *N Engl J Med* 1992;327:141-5.
- 557. Warkentin TE, Levine MN, Hirsh J, et al. Heparin-induced thrombocytopenia in patients treated with low-molecular-weight heparin or unfractionated heparin. *N Engl J Med* 1995;332:1330-5.
- 558. Harrington RA, Sane DC, Califf RM, et al, for the Thrombolysis and Angioplasty in Myocardial Infarction Study Group. Clinical importance of thrombocytopenia occurring in the hospital phase after administration of thrombolytic therapy for acute myocardial infarction. *J Am Coll Cardiol* 1994;23:891-8.
- 559. Ross AM, Molhoek P, Lundergan C, et al, for the HART II Investigators. Randomized comparison of enoxaparin, a lowmolecular-weight heparin, with unfractionated heparin adjunctive to recombinant tissue plasminogen activator thrombolysis and aspirin: second trial of Heparin and Aspirin Reperfusion Therapy (HART II). *Circulation* 2001;104:648-52.
- 560. Baird SH, Menown IB, Mcbride SJ, Trouton TG, Wilson C. Randomized comparison of enoxaparin with unfractionated heparin following fibrinolytic therapy for acute myocardial infarction. *Eur Heart J* 2002;23:627-32.
- 561. Tatu-Chitoiu G, Teodorescu C, Capraru M, et al. Accelerated streptokinase and enoxaparin in ST-segment elevation acute myocardial infarction (the ASENOX study) Polish J Cardiology 2004;60:441-6.
- 562. Antman EM, Louwerenburg HW, Baars HF, et al. Enoxaparin as adjunctive antithrombin therapy for ST-elevation myocardial infarction: results of the ENTIRE-Thrombolysis in Myocardial Infarction (TIMI) 23 Trial. *Circulation* 2002;105:1642-9.
- 563. Wallentin L, Bergstrand L, Dellborg M, et al. Low molecular weight heparin (dalteparin) compared to unfractionated heparin as an adjunct to rt-PA (alteplase) for improvement of coronary artery patency in acute myocardial infarction-the ASSENT Plus study. *Eur Heart J* 2003;24:897-908.
- 564. Kontny F, Dale J, Abildgaard U, Pedersen TR. Randomized trial of low molecular weight heparin (dalteparin) in prevention of left ventricular thrombus formation and arterial embolism after acute anterior myocardial infarction: the Fragmin in Acute Myocardial Infarction (FRAMI) study. *J Am Coll Cardiol* 1997;30:962-9.
- 565. Frostfeldt G, Ahlberg G, Gustafsson G, et al. Low molecular weight heparin (dalteparin) as adjuvant treatment of thrombolysis in acute myocardial infarction: a pilot study: biochemical markers in acute coronary syndromes (BIOMACS II). *J Am Coll Cardiol* 1999;33:627-33.
- 566. Cohen M, Gensini GF, Maritz F, et al, for the TETAMI Investigators. The safety and efficacy of subcutaneous enoxaparin versus intravenous unfractionated heparin and tirofiban versus placebo in the treatment of acute ST-segment elevation myocardial infarction patients ineligible for reperfusion (TETAMI): a randomized trial. *J Am Coll Cardiol* 2003;42:1348-56.
- 567. Kakkar VV, Iyengar SS, De Lorenzo F, Hargreaves JR, Kadziola ZA, for the FAMI Investigator Group. Low molecular weight heparin for treatment of acute myocardial infarction (FAMI): Fragmin (dalteparin sodium) in acute myocardial infarction. *Indian Heart J* 52:533-9.
- 568. Wong GC, Giugliano RP, Antman EM. Use of low-molecularweight heparins in the management of acute coronary artery syndromes and percutaneous coronary intervention. *JAMA* 2003;289:331-42.
- 569. Cannon CP, McCabe CH, Henry TD, et al. A pilot trial of recombinant desulfatohirudin compared with heparin in conjunction

with tissue-type plasminogen activator and aspirin for acute myocardial infarction: results of the Thrombolysis in Myocardial Infarction (TIMI) 5 trial. *J Am Coll Cardiol* 1994;23:993-1003.

- 570. Lidón RM, Théroux P, Lespérance J, et al. A pilot, early angiographic patency study using a direct thrombin inhibitor as adjunctive therapy to streptokinase in acute myocardial infarction. *Circulation* 1994;89:1567-72.
- 571. Théroux P, Pérez-Villa F, Waters D, Lespérance J, Shabani F, Bonan R. Randomized double-blind comparison of two doses of Hirulog with heparin as adjunctive therapy to streptokinase to promote early patency of the infarct-related artery in acute myocardial infarction. *Circulation* 1995;91:2132-9.
- 572. Neuhaus KL, Molhoek GP, Zeymer U, et al. Recombinant hirudin (lepirudin) for the improvement of thrombolysis with streptokinase in patients with acute myocardial infarction: results of the HIT-4 trial. *J Am Coll Cardiol* 1999;34:966-73.
- 573. Fung AY, Lorch G, Cambier PA, et al, for the ESCALAT Investigators. Efegatran sulfate as an adjunct to streptokinase versus heparin as an adjunct to tissue plasminogen activator in patients with acute myocardial infarction. *Am Heart J* 1999; 138:696-704.
- 574. Angiomax (bivalirudin) for injection. Package Insert. The Medicine Company, Parsippany, NJ. Available at http://www. angiomax.com/%7Eproducts_content/PN1002.REV.5.mcR4.pdf, 2003. Accessed December 5, 2003.
- 575. Coussement PK, Bassand JP, Convens C, et al, for the PENTAL-YSE investigators. A synthetic factor-Xa inhibitor (ORG31540/SR9017A) as an adjunct to fibrinolysis in acute myocardial infarction. The PENTALYSE study. *Eur Heart J* 2001;22:1716-24.
- 576. Bertrand ME, Rupprecht HJ, Urban P, Gershlick AH, for the Investigators FT. Double-blind study of the safety of clopidogrel with and without a loading dose in combination with aspirin compared with ticlopidine in combination with aspirin after coronary stenting: the clopidogrel aspirin stent international cooperative study (CLASSICS). *Circulation* 2000;102:624-9.
- 577. Mehta SR, Yusuf S, Peters RJ, et al, for the Clopidogrel in Unstable angina to prevent Recurrent Events trial (CURE) Investigators. Effects of pretreatment with clopidogrel and aspirin followed by long-term therapy in patients undergoing percutaneous coronary intervention: the PCI-CURE study. *Lancet* 2001;358:527-33.
- 578. Steinhubl SR, Berger PB, Mann JT 3rd, et al, for the Clopidogrel for the Reduction of Events During Observation (CREDO) Investigators. Early and sustained dual oral antiplatelet therapy following percutaneous coronary intervention: a randomized controlled trial. *JAMA* 2002;288:2411-20.
- 579. CYPHER[™] sirolimus-eluting coronary stent on RAPTORTM over-the-wire delivery system and CYPHER[™] sirolimus-eluting coronary stent on RAPTORRAIL® rapid exchange delivery system: instructions for Use. Miami, FL: Cordis: A Johnson & Johnson Company, April 2003.
- 580. TAXUS: Paclitaxel-eluting coronary stent system Monorail® and over the wire coronary stent delivery system: directions for use. Natick, MA: Boston Scientific; March 2004.
- 581. Patrono C, Bachmann F, Baigent C, et al. ESC expert consensus document on the use of antiplatelet agents: a report by the Task Force on the Use of Antiplatelet Agents in Patients With Atherosclerotic Cardiovascular Disease of the European Society of Cardiology. *Eur Heart J* 2004;25:166-81.
- 582. Antoniucci D, Rodriguez A, Hempel A, et al. A randomized trial comparing primary infarct artery stenting with or without abcix-

imab in acute myocardial infarction. J Am Coll Cardiol 2003; 42:1879-85.

- 583. Topol EJ, Neumann FJ, Montalescot G. A preferred reperfusion strategy for acute myocardial infarction. *J Am Coll Cardiol* 2003;42:1886-9.
- 584. Lee DP, Herity NA, Hiatt BL, et al. Adjunctive platelet glycoprotein IIb/IIIa receptor inhibition with tirofiban before primary angioplasty improves angiographic outcomes: results of the TIrofiban Given in the Emergency Room before Primary Angioplasty (TIGER-PA) pilot trial. *Circulation* 2003;107:1497-501.
- 585. Latini R, Maggioni AP, Flather M, Sleight P, Tognoni G. ACE inhibitor use in patients with myocardial infarction: summary of evidence from clinical trials. *Circulation* 1995;92:3132-7.
- 586. Gruppo Italiano per lo Studio della Sopravvivenza nell'infarto Miocardico (GISSI). GISSI-3: effects of lisinopril and transdermal glyceryl trinitrate singly and together on 6-week mortality and ventricular function after acute myocardial infarction. *Lancet* 1994;343:1115-22.
- 587. Ambrosioni E, Borghi C, Magnani B, for the Survival of Myocardial Infarction Long-Term Evaluation (SMILE) Study Investigators. The effect of the angiotensin-converting-enzyme inhibitor zofenopril on mortality and morbidity after anterior myocardial infarction. *N Engl J Med* 1995;332:80-5.
- 588. Oral captopril versus placebo among 13,634 patients with suspected acute myocardial infarction: interim report from the Chinese Cardiac Study (CCS-1). *Lancet* 1995;345:686-7.
- 589. ACE Inhibitor Myocardial Infarction Collaborative Group. Indications for ACE inhibitors in the early treatment of acute myocardial infarction: systematic overview of individual data from 100,000 patients in randomized trials. *Circulation* 1998; 97:2202-12.
- 590. Teo K, Yusuf S, Pfeffer M, et al. Effects of long-term treatment with angiotensin-converting-enzyme inhibitors in the presence or absence of aspirin: a systematic review. *Lancet* 2002;360:1037.
- 591. Meune C, Mahe I, Mourad JJ, et al. Interaction between angiotensin-converting enzyme inhibitors and aspirin: a review. *Eur J Clin Pharmacol* 2000;56:609-20.
- 592. Sigurdsson A, Swedberg K. Left ventricular remodelling, neurohormonal activation and early treatment with enalapril (CON-SENSUS II) following myocardial infarction. *Eur Heart J* 1994;15(Suppl B):14-9; discussion 26.
- 593. Sodi-Pallares D, Testelli MR, Fishleder BL, et al. Effects of an intravenous infusion of a potassium-glucose-insulin solution on the electrocardiographic signs of myocardial infarction: a preliminary clinical report. *Am J Cardiol* 1962;9:166-81.
- 594. Fath-Ordoubadi F, Beatt KJ. Glucose-insulin-potassium therapy for treatment of acute myocardial infarction: an overview of randomized placebo-controlled trials. *Circulation* 1997;96:1152-6.
- 595. Díaz R, Paolasso EA, Piegas LS, et al, for the ECLA (Estudios Cardiológicos Latinoamérica) Collaborative Group. Metabolic modulation of acute myocardial infarction. *Circulation* 1998;98:2227-34.
- 596. Ceremuzynski L, Budaj A, Czepiel A, et al. Low-dose glucoseinsulin-potassium is ineffective in acute myocardial infarction: results of a randomized multicenter Pol-GIK trial. *Cardiovasc Drugs Ther* 1999;13:191-200.
- 597. van der Horst IC, Zijlstra F, van't Hof AW, et al. Glucose-insulinpotassium infusion in patients treated with primary angioplasty for acute myocardial infarction: the glucose-insulin-potassium study: a randomized trial. *J Am Coll Cardiol* 2003;42:784-91.
- 598. Coleman GM, Gradinac S, Taegtmeyer H, Sweeney M, Frazier

OH. Efficacy of metabolic support with glucose-insulin-potassium for left ventricular pump failure after aortocoronary bypass surgery. *Circulation* 1989;80:I91-6.

- 599. Melidonis A, Stefanidis A, Tournis S, et al. The role of strict metabolic control by insulin infusion on fibrinolytic profile during an acute coronary event in diabetic patients. *Clin Cardiol* 2000;23:160-4.
- 600. Iwakura K, Ito H, Ikushima M, et al. Association between hyperglycemia and the no-reflow phenomenon in patients with acute myocardial infarction. *J Am Coll Cardiol* 2003;41:1-7.
- 601. Oliver MF, Opie LH. Effects of glucose and fatty acids on myocardial ischaemia and arrhythmias. *Lancet* 1994;343:155-8.
- 602. Malmberg K, Rydén L, Efendic S, et al. Randomized trial of insulin-glucose infusion followed by subcutaneous insulin treatment in diabetic patients with acute myocardial infarction (DIGA-MI study): effects on mortality at 1 year. J Am Coll Cardiol 1995;26:57-65.
- 603. Finney SJ, Zekveld C, Elia A, Evans TW. Glucose control and mortality in critically ill patients. *JAMA* 2003;290:2041-7.
- 603a. Clement S, Braithwaite SS, Magee MF, et al. Management of diabetes and hyperglycemia in hospitals. *Diabetes Care* 2004; 27:553-97
- 604. van den Berghe G, Wouters P, Weekers F, et al. Intensive insulin therapy in the critically ill patients. *N Engl J Med* 2001;345:1359-67.
- 605. Goldberg PA, Siegel MD, Sherwin RS, et al. Implementation of a safe and effective insulin infusion protocol in a medical intensive care unit. *Diabetes Care* 2004;27:461-7.
- 606. Inzucchi SE. Oral antihyperglycemic therapy for type 2 diabetes: scientific review. *JAMA* 2002;287:360-72.
- 607. DeWitt DE, Hirsch IB. Outpatient insulin therapy in type 1 and type 2 diabetes mellitus: scientific review. *JAMA* 2003;289:2254-64.
- 608. Davis SM, Granner DK. Insulin oral hypoglycemic agents and the pharmacology of the endocrine pancreas. In: Hardman J, Limbird L, eds. Goodman and Gilman's pharmacologic basis of therapeutics. 10th ed. USA: McGraw-Hill, 2001:1705.
- 609. Teo KK, Yusuf S, Collins R, Held PH, Peto R, Effects of intravenous magnesium in suspected acute myocardial infarction: overview of randomised trials. *BMJ* 1991;303:1499-503.
- 610. Antman EM, Lau J, Kupelnick B, Mosteller F, Chalmers TC. A comparison of results of meta-analyses of randomized control trials and recommendations of clinical experts: treatments for myocardial infarction. JAMA 1992;268:240-8.
- 611. Woods KL, Fletcher S. Long-term outcome after intravenous magnesium sulphate in suspected acute myocardial infarction: the second Leicester Intravenous Magnesium Intervention Trial (LIMIT-2). *Lancet* 1994;343:816-9.
- 612. Antman EM. Magnesium in acute myocardial infarction: overview of available evidence. *Am Heart J* 1996;132:487-95; discussion 4.
- 613. Baxter GF, Sumeray MS, Walker JM. Infarct size and magnesium: insights into LIMIT-2 and ISIS-4 from experimental studies. *Lancet* 1996;348:1424-6.
- 614. Magnesium in Coronaries (MAGIC) Trial Investigators. Early administration of intravenous magnesium to high-risk patients with acute myocardial infarction in the Magnesium in Coronaries (MAGIC) Trial: a randomised controlled trial. *Lancet* 2002; 360:1189-96.
- 615. Morton BC, Nair RC, Smith FM, McKibbon TG, Poznanski WJ. Magnesium therapy in acute myocardial infarction: a doubleblind study. *Magnesium* 1984;3:346-52.

- 616. Rasmussen HS, McNair P, Norregard P, Backer V, Lindeneg O, Balslev S. Intravenous magnesium in acute myocardial infarction. *Lancet* 1986;1:234-6.
- 617. Smith LF, Heagerty AM, Bing RF, Barnett DB. Intravenous infusion of magnesium sulphate after acute myocardial infarction: effects on arrhythmias and mortality. *Int J Cardiol* 1986;12:175-83.
- 618. Woods KL, Fletcher S, Smith LF. Intravenous magnesium in suspected acute myocardial infarction. *BMJ* 1992;304:119.
- 619. Abraham AS, Rosenmann D, Kramer M, et al. Magnesium in the prevention of lethal arrhythmias in acute myocardial infarction. *Arch Intern Med* 1987;147:753-5.
- 620. Ceremuzynski L, Jurgiel R, Kulakowski P, Gebalska J. Threatening arrhythmias in acute myocardial infarction are prevented by intravenous magnesium sulfate. *Am Heart J* 1989;118:1333-4.
- 621. Shechter M, Hod H, Marks N, et al. Beneficial effect of magnesium sulfate in acute myocardial infarction. *Am J Cardiol* 1990;66:271-4.
- 622. Feldstedt M, Boesgaard S, Bouchelouche P, et al. Magnesium substitution in acute ischaemic heart syndromes. *Eur Heart J* 1991;12:1215-8.
- 623. Woods KL, Fletcher S, Roffe C, Haider Y. Intravenous magnesium sulphate in suspected acute myocardial infarction: results of the second Leicester Intravenous Magnesium Intervention Trial (LIMIT-2). *Lancet* 1992;339:1553-8.
- 624. Thögersen AM, Johnson O, Wester PO. Effects of magnesium infusion on thrombolytic and non-thrombolytic treated patients with acute myocardial infarction. *Int J Cardiol* 1993;39:13-22.
- 625. Shechter M, Hod H, Chouraqui P, Kaplinsky E, Rabinowitz B. Magnesium therapy in acute myocardial infarction when patients are not candidates for thrombolytic therapy. *Am J Cardiol* 1995; 75:321-3.
- 626. Raghu C, Peddeswara Rao P, Seshagiri Rao D. Protective effect of intravenous magnesium in acute myocardial infarction following thrombolytic therapy. *Int J Cardiol* 1999;71:209-15.
- 627. Gyamlani G, Parikh C, Kulkarni AG. Benefits of magnesium in acute myocardial infarction: timing is crucial. *Am Heart J* 2000;139:703.
- 628. Woods KL, Abrams K. The importance of effect mechanism in the design and interpretation of clinical trials: the role of magnesium in acute myocardial infarction. *Prog Cardiovasc Dis* 2002; 44:267-74.
- 629. Guidelines 2000 for Cardiopulmonary Resuscitation and Emergency Cardiovascular Care: part 6: advanced cardiovascular life support: section 5: pharmacology I: agents for arrhythmias. The American Heart Association in collaboration with the International Liaison Committee on Resuscitation. *Circulation* 2000;102:I112-28.
- 630. Muller JE, Morrison J, Stone PH, et al. Nifedipine therapy for patients with threatened and acute myocardial infarction: a randomized, double-blind, placebo-controlled comparison. *Circulation* 1984;69:740-7.
- 631. Sirnes PA, Overskeid K, Pedersen TR, et al. Evolution of infarct size during the early use of nifedipine in patients with acute myocardial infarction: the Norwegian Nifedipine Multicenter Trial. *Circulation* 1984;70:638-44.
- 632. Wilcox RG, Hampton JR, Banks DC, et al. Trial of early nifedipine in acute myocardial infarction: the Trent study. *Br Med J (Clin Res Ed)* 1986;293:1204-8.
- 633. The Israeli Sprint Study Group. Secondary prevention reinfarction Israeli nifedipine trial (SPRINT): a randomized intervention

trial of nifedipine in patients with acute myocardial infarction. *Eur Heart J* 1988;9:354-64.

- 634. Goldbourt U, Behar S, Reicher-Reiss H, Zion M, Mandelzweig L, Kaplinsky E. Early administration of nifedipine in suspected acute myocardial infarction: the Secondary Prevention Reinfarction Israel Nifedipine Trial 2 Study. *Arch Intern Med* 1993;153:345-53.
- 635. Buring JE, Glynn RJ, Hennekens CH. Calcium channel blockers and myocardial infarction: a hypothesis formulated but not yet tested. *JAMA* 1995;274:654-5.
- 636. Furberg CD, Psaty BM, Meyer JV. Nifedipine. Dose-related increase in mortality in patients with coronary heart disease. *Circulation* 1995;92:1326-31.
- 637. Opie LH, Messerli FH. Nifedipine and mortality: grave defects in the dossier. *Circulation* 1995;92:1068-73.
- 638. The Danish Study Group on Verapamil in Myocardial Infarction. Verapamil in acute myocardial infarction. *Eur Heart J* 1984; 5:516-28.
- 639. Gheorghiade M. Calcium channel blockers in the management of myocardial infarction patients. *Henry Ford Hosp Med J* 1991;39:210-6.
- 640. Held PH, Yusuf S. Effects of beta-blockers and calcium channel blockers in acute myocardial infarction. *Eur Heart J* 1993;14:18-25.
- 641. Hilton TC, Miller DD, Kern MJ. Rational therapy to reduce mortality and reinfarction following myocardial infarction. *Am Heart J* 1991;122:1740-50.
- 642. Effect of verapamil on mortality and major events after acute myocardial infarction: the Danish Verapamil Infarction Trial II (DAVIT II). *Am J Cardiol* 1990;66:779-85.
- 643. The Multicenter Diltiazem Postinfarction Trial Research Group. The effect of diltiazem on mortality and reinfarction after myocardial infarction. *N Engl J Med* 1988;319:385-92.
- 644. Gibson RS, Boden WE, Theroux P, et al. Diltiazem and reinfarction in patients with non–Q-wave myocardial infarction: results of a double-blind, randomized, multicenter trial. *N Engl J Med* 1986;315:423-9.
- 645. Boden WE. Non–Q-wave myocardial infarction: a prognostic paradox. *Hosp Pract (Off Ed)* 1992;27:129-33,137-8,140.
- 646. Boden WE, van Gilst WH, Scheldewaert RG, et al. Diltiazem in acute myocardial infarction treated with thrombolytic agents: a randomised placebo-controlled trial. Incomplete Infarction Trial of European Research Collaborators Evaluating Prognosis post-Thrombolysis (INTERCEPT). *Lancet* 2000;355:1751-6.
- 647. American Association of Critical Care Nurses White Paper. Safeguarding the Patient and the Profession: the Value of Critical Care Nurse Certification, Aliso Viejo, CA, December 2002.
- 648. Dimick JB, Swoboda SM, Pronovost PJ, Lipsett PA. Effect of nurse-to-patient ratio in the intensive care unit on pulmonary complications and resource use after hepatectomy. *Am J Crit Care* 2001;10:376-82.
- 649. Needleman J, Buerhaus P, Mattke S, Stewart M, Zelevinsky K. Nurse-staffing levels and the quality of care in hospitals. *N Engl J Med* 2002;346:1715-22.
- 650. Pronovost PJ, Dang D, Dorman T, et al. Intensive care unit nurse staffing and the risk for complications after abdominal aortic surgery. *Eff Clin Pract* 2001;4:199-206.
- 651. Pilcher T, Odell M. Position statement on nurse-patient ratios in critical care. *Nurs Stand* 2000;15:38-41.
- 652. "California Safe Hospital Staffing Law" AB 394.
- 653. Bolton LB, Jones D, Aydin CE, et al. A response to California's mandated nursing ratios. *J Nurs Scholarship* 2001;33:179-84.

- 654. Maintaining patient-focused care in an environment of nursing staff shortages and financial constraints: a statement from the American Association of Critical-Care Nurses (AACN). Available at http://www.aacn.org/pdfLibra.NSF/Files/burson/\$file/burson.pdf, November 2000. Accessed January 7, 2004.
- 655. Newby LK, Califf RM, Guerci A, et al. Early discharge in the thrombolytic era: an analysis of criteria for uncomplicated infarction from the Global Utilization of Streptokinase and t-PA for Occluded Coronary Arteries (GUSTO) trial. *J Am Coll Cardiol* 1996;27:625-32.
- 656. Fein A. Organization and Management of Critical Care Units. In: Irwin RS, Rippe JM, eds. Manual of Intensive Care Medicine. 4th ed. Baltimore, MD: Lippincott-Williams & Wilkins; 1999:2501-9.
- 657. Drew BJ, Wung SF, Adams MG, Pelter MM. Bedside diagnosis of myocardial ischemia with ST-segment monitoring technology: measurement issues for real-time clinical decision making and trial designs. *J Electrocardiol* 1998;30:157-65.
- 658. Bell NN. Clinical significance of ST-segment monitoring. Crit Care Nurs Clin North Am 1992;4:313-23.
- 659. Johanson P, Rössberg J, Dellborg M. Continuous ST monitoring: a bedside instrument? A report from the Assessment of the Safety of a New Thrombolytic (ASSENT 2) ST monitoring substudy. *Am Heart J* 2001;142:58-62.
- 660. Drew BJ, Ide B, Sparacino PS. Accuracy of bedside electrocardiographic monitoring: a report on current practices of critical care nurses. *Heart Lung* 1991;20:597-607.
- 661. Romhilt DW, Bloomfield SS, Chou TC, Fowler NO. Unreliability of conventional electrocardiographic monitoring for arrhythmia detection in coronary care units. *Am J Cardiol* 1973;31:457-61.
- 662. Drew BJ, Krucoff MW, for the ST- Segment Monitoring Practice Guideline International Working Group. Multilead ST-segment monitoring in patients with acute coronary syndromes: a consensus statement for healthcare professionals. *Am J Crit Care* 1999;8:372-86; quiz 387-8.
- 663. Pelter MM, Adams MG, Wung SF, Paul SM, Drew BJ. Peak time of occurrence of myocardial ischemia in the coronary care unit. *Am J Crit Care* 1998;7:411-7.
- 664. Dellborg M, Topol EJ, Swedberg K. Dynamic QRS complex and ST segment vectorcardiographic monitoring can identify vessel patency in patients with acute myocardial infarction treated with reperfusion therapy. *Am Heart J* 1991;122:943-8.
- 665. Krucoff MW, Croll MA, Pope JE, et al. Continuously updated 12lead ST-segment recovery analysis for myocardial infarct artery patency assessment and its correlation with multiple simultaneous early angiographic observations. *Am J Cardiol* 1993;71:145-51.
- 666. Veldkamp RF, Green CL, Wilkins ML, et al, for the Thrombolysis and Angioplasty in Myocardial Infarction (TAMI) 7 Study Group. Comparison of continuous ST-segment recovery analysis with methods using static electrocardiograms for noninvasive patency assessment during acute myocardial infarction. *Am J Cardiol* 1994;73:1069-74.
- 667. Langer A, Krucoff MW, Klootwijk P, et al. Noninvasive assessment of speed and stability of infarct-related artery reperfusion: results of the GUSTO ST segment monitoring study. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries. J Am Coll Cardiol 1995;25:1552-7.
- 668. Santoro GM, Valenti R, Buonamici P, et al. Relation between STsegment changes and myocardial perfusion evaluated by myocardial contrast echocardiography in patients with acute myocardial infarction treated with direct angioplasty. *Am J Cardiol* 1998; 82:932-7.

- 669. Krucoff MW, Parente AR, Bottner RK, et al. Stability of multilead ST-segment "fingerprints" over time after percutaneous transluminal coronary angioplasty and its usefulness in detecting reocclusion. *Am J Cardiol* 1988;61:1232-7.
- 670. Vatner SF, McRitchie RJ, Maroko PR, Patrick TA, Braunwald E. Effects of catecholamines, exercise, and nitroglycerin on the normal and ischemic myocardium in conscious dogs. *J Clin Invest* 1974;54:563-75.
- 671. Chobanian AV, Lille RD, Tercyak A, Blevins P. The metabolic and hemodynamic effects of prolonged bed rest in normal subjects. *Circulation* 1974;49:551-9.
- 672. Convertino VA. Value of orthostatic stress in maintaining functional status soon after myocardial infarction or cardiac artery bypass grafting. *J Cardiovasc Nurs* 2003;18:124-30.
- 673. Metzger BL, Therrien B. Effect of position on cardiovascular response during the Valsalva maneuver. *Nurs Res* 1990;39:198-202.
- 674. Taggart P, Sutton P, John R, Lab M, Swanton H. Monophasic action potential recordings during acute changes in ventricular loading induced by the Valsalva manoeuvre. *Br Heart J* 1992;67:221-9.
- 675. Porth CJ, Bamrah VS, Tristani FE, Smith JJ. The Valsalva maneuver: mechanisms and clinical implications. *Heart Lung* 1984;13:507-18.
- 676. Storm DS, Metzger BL, Therrien B. Effects of age on autonomic cardiovascular responsiveness in healthy men and women. *Nurs Res* 1989;38:326-30.
- 677. Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III) final report. *Circulation* 2002;106:3143-421.
- 678. Goldstein IB, Shapiro D, Hui KK, Yu JL. Blood pressure response to the "second cup of coffee". *Psychosom Med* 1990;52:337-45.
- 679. Aro A, Teirilä J, Gref CG. Dose-dependent effect on serum cholesterol and apoprotein B concentrations by consumption of boiled, non-filtered coffee. *Atherosclerosis* 1990;83:257-61.
- 680. Nawrot P, Jordan S, Eastwood J, Rotstein J, Hugenholtz A, Feeley M. Effects of caffeine on human health. *Food Addit Contam* 2003;20:1-30.
- 681. Myers MG, Harris L. High-dose caffeine and ventricular arrhythmias. *Can J Cardiol* 1990;6:95-8.
- 682. Myers MG, Harris L, Leenen FH, Grant DM. Caffeine as a possible cause of ventricular arrhythmias during the healing phase of acute myocardial infarction. *Am J Cardiol* 1987;59:1024-8.
- 683. Pincomb GA, Lovallo WR, Passey RB, Whitsett TL, Silverstein SM, Wilson MF. Effects of caffeine on vascular resistance, cardiac output and myocardial contractility in young men. Am J Cardiol 1985;56:119-22.
- 684. Hughes JR, Oliveto AH, Bickel WK, Higgins ST, Badger GJ. Caffeine self-administration and withdrawal: incidence, individual differences and interrelationships. *Drug Alcohol Depend* 1993;32:239-46.
- 685. van Dusseldorp M, Katan MB. Headache caused by caffeine withdrawal among moderate coffee drinkers switched from ordinary to decaffeinated coffee: a 12-week double-blind trial. *BMJ* 1990;300:1558-9.
- 686. Höfer I, Bättig K. Cardiovascular, behavioral, and subjective effects of caffeine under field conditions. *Pharmacol Biochem Behav* 1994;48:899-908.
- 687. Lynn LA, Kissinger JF. Coronary precautions: should caffeine be restricted in patients after myocardial infarction? *Heart Lung* 1992;21:365-71.

- 688. Boyd MD. Strategies for effective health teaching. In: Boyd MD, Graham BA, Gleit CJ, Whitman NI, eds. Health teaching in nursing practice: a professional model. 3rd ed. Stanford, CT: Appleton and Lange;1998:201-29.
- 689. Fleury J, Moore SM. Family-centered care after acute myocardial infarction. *J Cardiovasc Nurs* 1999;13:73-82.
- 690. Duryée R. The efficacy of inpatient education after myocardial infarction. *Heart Lung* 1992;21:217-25.
- 691. Saarmann L, Daugherty J, Riegel B. Patient teaching to promote behavioral change. *Nurs Outlook* 2000;48:281-7.
- 692. Cooper H, Booth K, Fear S, Gill G. Chronic disease patient education: lessons from meta-analyses. *Patient Educ Couns* 2001;44:107-17.
- 693. Prochaska JO, DiClemente CC, Norcross JC. In search of how people change: applications to addictive behaviors. *Am Psychol* 1992;47:1102-14.
- 694. Nigg CR, Burbank PM, Padula C, et al. Stages of change across ten health risk behaviors for older adults. *Gerontologist* 1999;39:473-82.
- 695. Glazer HR, Kirk LM, Bosler FE. Patient education pamphlets about prevention, detection, and treatment of breast cancer for low literacy women. *Patient Educ Couns* 1996;27:185-9.
- 696. Antman EM, Kuntz KM. The length of the hospital stay after myocardial infarction. *N Engl J Med* 2000;342:808-10.
- 697. Visser A, Wissow L. From patient education to communication in health care. *Patient Educ Couns* 2003;50:227-8.
- 698. Frazier SK, Moser DK, O'Brien JL, Garvin BJ, An K, Macko M. Management of anxiety after acute myocardial infarction. *Heart Lung* 2002;31:411-20.
- 699. Frazier SK, Moser DK, Daley LK, et al. Critical care nurses' beliefs about and reported management of anxiety. *Am J Crit Care* 2003;12:19-27.
- 700. Devine EC. Effects of psychoeducational care for adult surgical patients: a meta-analysis of 191 studies. *Patient Educ Couns* 1992;19:129-42.
- 701. Bartlett EE. Cost-benefit analysis of patient education. *Patient Educ Couns* 1995;26:87-91.
- 702. LaBresh KA, Gliklich R, Liljestrand J, Peto R, Ellrodt AG. Using "get with the guidelines" to improve cardiovascular secondary prevention. *Jt Comm J Qual Saf* 2003;29:539-50.
- 703. The Mended Hearts, Inc., Dallas, TX. Available at http://www.mendedhearts.org/, November 2003. Accessed December 16, 2002.
- 704. Hughes JR, Higgins ST, Bickel WK. Nicotine withdrawal versus other drug withdrawal syndromes: similarities and dissimilarities. *Addiction* 1994;89:1461-70.
- 705. Malan SS. Psychosocial adjustment following MI: current views and nursing implications. J Cardiovasc Nurs 1992;6:57-70.
- 706. Havik OE, Maeland JG. Patterns of emotional reactions after a myocardial infarction. *J Psychosom Res* 1990;34:271-85.
- 707. Moser DK, Dracup K. Is anxiety early after myocardial infarction associated with subsequent ischemic and arrhythmic events? *Psychosom Med* 1996;58:395-401.
- 708. Frasure-Smith N, Lespérance F, Talajic M. The impact of negative emotions on prognosis following myocardial infarction: is it more than depression? *Health Psychol* 1995;14:388-98.
- 709. O'Brien JL, Moser DK, Riegel B, Frazier SK, Garvin BJ, Kim KA. Comparison of anxiety assessments between clinicians and patients with acute myocardial infarction in cardiac critical care units. *Am J Crit Care* 2001;10:97-103.
- 710. Frazier SK, Moser DK, Riegel B, et al. Critical care nurses' assessment of patients' anxiety: reliance on physiological and

e273 ACC/AHA Practice Guidelines

behavioral parameters. Am J Crit Care 2002;11:57-64.

- 711. Simpson T, Shaver J. Cardiovascular responses to family visits in coronary care unit patients. *Heart Lung* 1990;19:344-51.
- 712. Schulte DA, Burrell LO, Gueldner SH, et al. Pilot study of the relationship between heart rate and ectopy and unrestricted vs restricted visiting hours in the coronary care unit. *Am J Crit Care* 1993;2:134-6.
- 713. Thompson DR, Meddis R. A prospective evaluation of inhospital counselling for first-time myocardial infarction men. *J Psychosom Res* 1990;34:237-48.
- 714. Thompson DR. A randomized controlled trial of in-hospital nursing support for first-time myocardial infarction patients and their partners: effects on anxiety and depression. J Adv Nurs 1989; 14:291-7.
- 715. Newby LK, Eisenstein EL, Califf RM, Thompson TD, Nelson CL, Peterson ED, Armstrong PW, Van de Werf F, White HD, Topol EJ, Mark DB. Cost effectiveness of early discharge after uncomplicated acute myocardial infarction. N Engl J Med 2000;342:749-55.
- 716. Brooten D, Naylor MD, York R, et al. Lessons learned from testing the quality cost model of Advanced Practice Nursing (APN) transitional care. *J Nurs Scholarship* 2002;34:369-75.
- 717. Chae CU, Hennekens CH. Beta blockers. In: Hennekens CH, ed. Clinical trials in cardiovascular disease: a companion to Braunwald's Heart Disease. Philadelphia, PA: WB Saunders Co Ltd; 1999:79-94.
- 718. Antman E, Braunwald E. Acute myocardial infarction. In: Braunwald E, Zipes DP, Libby P, eds. Heart disease: a textbook of cardiovascular medicine. 6th ed. Philadelphia, PA: WB Saunders Co Ltd; 2001:1114-1231.
- 719. Beta-Blocker Heart Attack Study Group. The beta-blocker heart attack trial. *JAMA* 1981;246:2073-4.
- 720. Flather MD, Yusuf S, Køber L, et al, for the ACE-Inhibitor Myocardial Infarction Collaborative Group. Long-term ACEinhibitor therapy in patients with heart failure or left-ventricular dysfunction: a systematic overview of data from individual patients. *Lancet* 2000;355:1575-81.
- 721. Yusuf S, Sleight P, Pogue J, Bosch J, Davies R, Dagenais G, for the Heart Outcomes Prevention Evaluation Study Investigators. Effects of an angiotensin-converting-enzyme inhibitor, ramipril, on cardiovascular events in high-risk patients. N Engl J Med 2000;342:145-53.
- 722. Pitt B, Zannad F, Remme WJ, et al, for the Randomized Aldactone Evaluation Study Investigators. The effect of spironolactone on morbidity and mortality in patients with severe heart failure. N Engl J Med 1999;341:709-17.
- 723. Pitt B, Remme W, Zannad F, et al, for the Eplerenone Post-Acute Myocardial Infarction Heart Failure Efficacy and Survival Study Investigators. Eplerenone, a selective aldosterone blocker, in patients with left ventricular dysfunction after myocardial infarction. N Engl J Med 2003;348:1309-21.
- 724. Dickstein K, Kjekshus J, for the OPTIMAAL Steering Committee of the OPTIMAAL Study Group. Effects of losartan and captopril on mortality and morbidity in high-risk patients after acute myocardial infarction: the OPTIMAAL randomised trial. Optimal Trial in Myocardial Infarction with Angiotensin II Antagonist Losartan. *Lancet* 2002;360:752-60.
- 725. Pfeffer MA, McMurray JJ, Velazquez EJ, et al, for the Valsartan in Acute Myocardial Infarction Trial Investigators. Valsartan, captopril, or both in myocardial infarction complicated by heart failure, left ventricular dysfunction, or both. *N Engl J Med* 2003;349:1893-906.

- 726. Mann DL, Deswal A. Angiotensin-receptor blockade in acute myocardial infarction: a matter of dose. *N Engl J Med* 2003; 349:1963-5.
- 727. Antiplatelet Trialists' Collaboration. Collaborative overview of randomised trials of antiplatelet therap: I–Prevention of death, myocardial infarction, and stroke by prolonged antiplatelet therapy in various categories of patients. *BMJ* 1994;308:81-106.
- 728. Yusuf S, Zhao F, Mehta SR, Chrolavicius S, Tognoni G, Fox KK, for the Clopidogrel in Unstable Angina to Prevent Recurrent Events Trial Investigators. Effects of clopidogrel in addition to aspirin in patients with acute coronary syndromes without ST-segment elevation. N Engl J Med 2001;345:494-502.
- 729. Plavix (clopidogrel) package insert. Sanofi-Synthelabo. New York, NY. March 2002.
- 730. Hirsh J, Dalen JE, Fuster V, Harker LB, Salzman EW. Aspirin and other platelet-active drugs: the relationship between dose, effectiveness, and side effects. *Chest* 1992;102:327S-336S.
- 731. Leonards JR, Levy G. Effect of pharmaceutical formulation on gastrointestinal bleeding from aspirin tablets. *Arch Intern Med* 1972;129:457-60.
- 732. MacKercher PA, Ivey KJ, Baskin WN, Krause WJ. Protective effect of cimetidine on aspirin-induced gastric mucosal damage. *Ann Intern Med* 1977;87:676-9.
- 733. Bowen BK, Krause WJ, Ivey KJ. Effect of sodium bicarbonate on aspirin-induced damage and potential difference changes in human gastric mucosa. *Br Med J* 1977;2:1052-5.
- 734. Graham DY, Smith JL. Aspirin and the stomach. *Ann Intern Med* 1986;104:390-8.
- 735. Mielants H, Verbruggen G, Schelstraete K, Veys EM. Salicylateinduced gastrointestinal bleeding: comparison between soluble buffered, enteric-coated, and intravenous administration. *J Rheumatol* 1979;6:210-8.
- 736. Goldman S, Copeland J, Moritz T, et al. Improvement in early saphenous vein graft patency after coronary artery bypass surgery with antiplatelet therapy: results of a Veterans Administration Cooperative Study. *Circulation* 1988;77:1324-32.
- 737. Chesebro JH, Clements IP, Fuster V, et al. A platelet-inhibitordrug trial in coronary-artery bypass operations: benefit of perioperative dipyridamole and aspirin therapy on early postoperative vein-graft patency. *N Engl J Med* 1982;307:73-8.
- 738. Sanz G, Pajarón A, Alegría E, et al, for the Grupo Español para el Seguimiento del Injerto Coronario (GESIC). Prevention of early aortocoronary bypass occlusion by low-dose aspirin and dipyridamole. *Circulation* 1990;82:765-73.
- 739. Goldman S, Copeland J, Moritz T, et al, for the Department of Veterans Affairs Cooperative Study Group. Starting aspirin therapy after operation: effects on early graft patency. *Circulation* 1991;84:520-6.
- 740. Hass WK, Easton JD, Adams HP, et al, for the Ticlopidine Aspirin Stroke Study Group. A randomized trial comparing ticlopidine hydrochloride with aspirin for the prevention of stroke in highrisk patients. *N Engl J Med* 1989;321:501-7.
- 741. Schömig A, Neumann FJ, Kastrati A, et al. A randomized comparison of antiplatelet and anticoagulant therapy after the placement of coronary-artery stents. *N Engl J Med* 1996;334:1084-9.
- 742. CAPRIE Steering Committee. A randomised, blinded, trial of clopidogrel versus aspirin in patients at risk of ischaemic events (CAPRIE). *Lancet* 1996;348:1329-39.
- 743. Leon MB, Baim DS, Popma JJ, et al, for the Stent Anticoagulation Restenosis Study Investigators. A clinical trial comparing three antithrombotic-drug regimens after coronary-artery stenting. N Engl J Med 1998;339:1665-71.

- 744. The SCATI (Studio sulla Calciparina nell'Angina e nella Trombosi Ventricolare nell'Infarto) Group. Randomised controlled trial of subcutaneous calcium-heparin in acute myocardial infarction. *Lancet* 1989;2:182-6.
- 745. Deleted in press.
- 746. 4th American College of Chest Physicians Consensus Conference on Antithrombotic Therapy. Tucson, Arizona, April 1995. Proceedings. *Chest* 1995;108:225S-522S.
- 747. Chesebro JH, Fuster V. Antithrombotic therapy for acute myocardial infarction: mechanisms and prevention of deep venous, left ventricular, and coronary artery thromboembolism. *Circulation* 1986;74:III1-10.
- 748. Sevilla DC, Wagner NB, Anderson WD, et al. Sensitivity of a set of myocardial infarction screening criteria in patients with anatomically documented single and multiple infarcts. *Am J Cardiol* 1990;66:792-5.
- 749. Ideker RE, Wagner GS, Ruth WK, et al. Evaluation of a QRS scoring system for estimating myocardial infarct size–II: correlation with quantitative anatomic findings for anterior infarcts. *Am J Cardiol* 1982;49:1604-14.
- 750. Goodman SG, Langer A, Ross AM, et al, for the GUSTO-I Angiographic Investigators. Non–Q-wave versus Q-wave myocardial infarction after thrombolytic therapy: angiographic and prognostic insights from the global utilization of streptokinase and tissue plasminogen activator for occluded coronary arteries-I angiographic substudy. *Circulation* 1998;97:444-50.
- 751. Buja LM, Willerson JT. Infarct size: can it be measured or modified in humans? *Prog Cardiovasc Dis* 1987;29:271-89.
- 752. Hackel DB, Reimer KA, Ideker RE, et al. Comparison of enzymatic and anatomic estimates of myocardial infarct size in man. *Circulation* 1984;70:824-35.
- 753. Panteghini M, Cuccia C, Bonetti G, Giubbini R, Pagani F, Bonini E. Single-point cardiac troponin T at coronary care unit discharge after myocardial infarction correlates with infarct size and ejection fraction. *Clin Chem* 2002;48:1432-6.
- 754. Licka M, Zimmermann R, Zehelein J, Dengler TJ, Katus HA, Kübler W. Troponin T concentrations 72 hours after myocardial infarction as a serological estimate of infarct size. *Heart* 2002; 87:520-4.
- 755. Myocardial infarction redefined: a consensus document of the Joint European Society of Cardiology/American College of Cardiology Committee for the redefinition of myocardial infarction. *Eur Heart J* 2000;21:1502-13.
- 756. Gibbons RJ, Miller TD, Christian TF. Infarct size measured by single photon emission computed tomographic imaging with (99m)Tc-sestamibi: a measure of the efficacy of therapy in acute myocardial infarction. *Circulation* 2000;101:101-8.
- 757. Mahrholdt H, Wagner A, Holly TA, et al. Reproducibility of chronic infarct size measurement by contrast-enhanced magnetic resonance imaging. *Circulation* 2002;106:2322-7.
- 758. Mueller HS, Chatterjee K, Davis KB, et al. ACC expert consensus document on the present use of bedside right heart catheterization in patients with cardiac disease. American College of Cardiology. J Am Coll Cardiol 1998;32:840-64.
- 759. Menon V, Slater JN, White HD, Sleeper LA, Cocke T, Hochman JS. Acute myocardial infarction complicated by systemic hypoperfusion without hypotension: report of the SHOCK trial registry. *Am J Med* 2000;108:374-80.
- Hollenberg SM, Hoyt J. Pulmonary artery catheters in cardiovascular disease. *New Horiz* 1997;5:207-13.
- 761. Weil MH. The assault on the Swan-Ganz catheter: a case history of constrained technology, constrained bedside clinicians, and

constrained monetary expenditures. Chest 1998;113:1379-86.

- 762. Dalen JE. The pulmonary artery catheter-friend, foe, or accomplice? *JAMA* 2001;286:348-50.
- 763. O'Grady NP, Alexander M, Dellinger EP, et al. Guidelines for the prevention of intravascular catheter-related infections. Centers for Disease Control and Prevention. *MMWR* Recomm Rep. 2002;51:1-29.
- 764. Killip T, Kimball JT. Treatment of myocardial infarction in a coronary care unit: a two-year experience with 250 patients. *Am J Cardiol* 1967;20:457-64.
- 765. Hochman JS, Jaber W, Bates ER, et al. Angioplasty versus thrombolytics for patients presenting with congestive heart failure: GUSTO IIb substudy findings. J Am Coll Cardiol 1998;31:856-4.
- 766. Guidelines 2000 for cardiopulmonary resuscitation and emergency cardiovascular care: part 7–the era of reperfusion: section 1–acute coronary syndromes (acute myocardial infarction). *Circulation* 2000;102:I172-203.
- 767. Prewitt RM, Gu S, Garber PJ, Ducas J. Marked systemic hypotension depresses coronary thrombolysis induced by intracoronary administration of recombinant tissue-type plasminogen activator. *J Am Coll Cardiol* 1992;20:1626-33.
- 768. Prewitt RM, Gu S, Schick U, Ducas J. Intraaortic balloon counterpulsation enhances coronary thrombolysis induced by intravenous administration of a thrombolytic agent. *J Am Coll Cardiol* 1994;23:794-8.
- 769. Swedberg K, Held P, Kjekshus J, Rasmussen K, Rydén L, Wedel H. Effects of the early administration of enalapril on mortality in patients with acute myocardial infarction: results of the Cooperative New Scandinavian Enalapril Survival Study II (CONSENSUS II) *N Engl J Med* 1992;327:678-84.
- 770. Intravenous nesiritide vs nitroglycerin for treatment of decompensated congestive heart failure: a randomized controlled trial. *JAMA* 2002;287:1531-40.
- 771. Hunt SA, Baker DW, Chin MH, et al. ACC/AHA guidelines for the evaluation and management of chronic heart failure in the adult: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the 1995 Guidelines for the Evaluation and Management of Heart Failure). American College of Cardiology Web site. Available at: http://www.acc.org/clinical/guidelines/failure/hf_index.htm, 2001. Accessed August 5, 2003.
- 772. Goldberg RJ, Gore JM, Alpert JS, et al. Cardiogenic shock after acute myocardial infarction: incidence and mortality from a community-wide perspective, 1975 to 1988. *N Engl J Med* 1991;325:1117-22.
- 773. Bengtson JR, Kaplan AJ, Pieper KS, et al. Prognosis in cardiogenic shock after acute myocardial infarction in the interventional era. *J Am Coll Cardiol* 1992;20:1482-9.
- 774. Berger AK, Radford MJ, Krumholz HM. Cardiogenic shock complicating acute myocardial infarction in elderly patients: does admission to a tertiary center improve survival? *Am Heart J* 2002;143:768-76.
- 775. Allen BS, Rosenkranz E, Buckberg GD, et al. Studies on prolonged acute regional ischemia: VI–myocardial infarction with left ventricular power failure: a medical/surgical emergency requiring urgent revascularization with maximal protection of remote muscle. *J Thorac Cardiovasc Surg* 1989;98:691-702; discussion.
- 776. Allen BS, Buckberg GD, Fontan FM, et al. Superiority of controlled surgical reperfusion versus percutaneous transluminal coronary angioplasty in acute coronary occlusion. J Thorac Cardiovasc Surg 1993;105:864-79; discussion 8.

Antman et al. 2004

e275 ACC/AHA Practice Guidelines

- 777. Dixon SR, Alkafri H, Chami A. Clinical predictors of in-hospital death in patients with cardiogenic shock selected to undergo early revascularization [abstr]. J Am Coll Cardiol 2002;39:808–1.
- 778. Barron HV, Every NR, Parsons LS, et al, for the Investigators in the National Registry of Myocardial Infarction 2. The use of intraaortic balloon counterpulsation in patients with cardiogenic shock complicating acute myocardial infarction: data from the National Registry of Myocardial Infarction 2. *Am Heart J* 2001;141:933-9.
- 779. Deleted in press.
- 780. Zehender M, Kasper W, Kauder E, et al H. Right ventricular infarction as an independent predictor of prognosis after acute inferior myocardial infarction. *N Engl J Med* 1993;328:981-8.
- 781. Berger PB, Ryan TJ. Inferior myocardial infarction: high-risk subgroups. *Circulation* 1990;81:401-11.
- 782. Jacobs AK, Leopold JA, Bates E, et al. Cardiogenic shock caused by right ventricular infarction: a report from the SHOCK registry. *J Am Coll Cardiol* 2003;41:1273-9.
- 783. Weinshel AJ, Isner JM, Salem DN. The coronary anatomy of right ventricular infarction: relationship between the site of right coronary artery occlusion and origin of the right ventricular free wall branches. *Circulation* 1983;68:III-351.
- 784. Andersen HR, Falk E, Nielsen D. Right ventricular infarction: frequency, size and topography in coronary heart disease: a prospective study comprising 107 consecutive autopsies from a coronary care unit. J Am Coll Cardiol 1987;10:1223-32.
- 785. Lee FA. Hemodynamics of the right ventricle in normal and disease states. *Cardiol Clin* 1992;10:59-67.
- 786. Cross CE. Right ventricular pressure and coronary flow. Am J Physiol 1962;202:12-16.
- 787. Haupt HM, Hutchins GM, Moore GW. Right ventricular infarction: role of the moderator band artery in determining infarct size. *Circulation* 1983;67:1268-72.
- 788. Setaro JF, Cabin HS. Right ventricular infarction. *Cardiol Clin* 1992;10:69-90.
- 789. Goldstein JA, Vlahakes GJ, Verrier ED, et al. The role of right ventricular systolic dysfunction and elevated intrapericardial pressure in the genesis of low output in experimental right ventricular infarction. *Circulation* 1982;65:513-22.
- 790. Ferguson JJ, Diver DJ, Boldt M, Pasternak RC. Significance of nitroglycerin-induced hypotension with inferior wall acute myocardial infarction. *Am J Cardiol* 1989;64:311-4.
- 791. Goldstein JA, Barzilai B, Rosamond TL, Eisenberg PR, Jaffe AS. Determinants of hemodynamic compromise with severe right ventricular infarction. *Circulation* 1990;82:359-68.
- 792. Goldstein JA, Tweddell JS, Barzilai B, Yagi Y, Jaffe AS, Cox JL. Importance of left ventricular function and systolic ventricular interaction to right ventricular performance during acute right heart ischemia. J Am Coll Cardiol 1992;19:704-11.
- 793. Bowers TR, O'Neill WW, Grines C, Pica MC, Safian RD, Goldstein JA. Effect of reperfusion on biventricular function and survival after right ventricular infarction. N Engl J Med 1998;338:933-40.
- 794. Bowers TR, O'Neill WW, Pica M, Goldstein JA. Patterns of coronary compromise resulting in acute right ventricular ischemic dysfunction. *Circulation* 2002;106:1104-9.
- 795. Dell'Italia LJ, Starling MR, O'Rourke RA. Physical examination for exclusion of hemodynamically important right ventricular infarction. Ann Intern Med 1983;99:608-11.
- 796. Dell'Italia LJ, Starling MR, Crawford MH, Boros BL, Chaudhuri TK, O'Rourke RA. Right ventricular infarction: identification by hemodynamic measurements before and after volume loading and correlation with noninvasive techniques. J Am Coll Cardiol

1984;4:931-9.

- 797. Cohn JN, Guiha NH, Broder MI, Limas CJ. Right ventricular infarction: clinical and hemodynamic features. *Am J Cardiol* 1974;33:209-14.
- 798. Robalino BD, Whitlow PL, Underwood DA, Salcedo EE. Electrocardiographic manifestations of right ventricular infarction. *Am Heart J* 1989;118:138-44.
- 799. Braat SH, Brugada P, de Zwaan C, Coenegracht JM, Wellens HJ. Value of electrocardiogram in diagnosing right ventricular involvement in patients with an acute inferior wall myocardial infarction. *Br Heart J* 1983;49:368-72.
- 800. Sharkey SW, Shelley W, Carlyle PF, Rysavy J, Cohn JN. M-mode and two-dimensional echocardiographic analysis of the septum in experimental right ventricular infarction: correlation with hemodynamic alterations. *Am Heart J* 1985;110:1210-8.
- 801. López-Sendón J, López de Sá E, Roldán I, Fernández de Soria R, Ramos F, Martín Jadraque L. Inversion of the normal interatrial septum convexity in acute myocardial infarction: incidence, clinical relevance and prognostic significance. J Am Coll Cardiol 1990;15:801-5.
- 802. Manno BV, Bemis CE, Carver J, Mintz GS. Right ventricular infarction complicated by right to left shunt. J Am Coll Cardiol 1983;1:554-7.
- 803. Wellens HJ. The value of the right precordial leads of the electrocardiogram. *N Engl J Med* 1999;340:381-3.
- 804. Goldstein JA, Vlahakes GJ, Verrier ED, et al. Volume loading improves low cardiac output in experimental right ventricular infarction. *J Am Coll Cardiol* 1983;2:270-8.
- 805. Dell'Italia LJ, Starling MR, Blumhardt R, Lasher JC, O'Rourke RA. Comparative effects of volume loading, dobutamine, and nitroprusside in patients with predominant right ventricular infarction. *Circulation* 1985;72:1327-35.
- 806. Love JC, Haffajee CI, Gore JM, Alpert JS. Reversibility of hypotension and shock by atrial or atrioventricular sequential pacing in patients with right ventricular infarction. *Am Heart J* 1984;108:5-13.
- 807. Sugiura T, Iwasaka T, Takahashi N, et al. Atrial fibrillation in inferior wall Q-wave acute myocardial infarction. *Am J Cardiol* 1991;67:1135-6.
- 808. Braat SH, de Zwaan C, Brugada P, Coenegracht JM, Wellens HJ. Right ventricular involvement with acute inferior wall myocardial infarction identifies high risk of developing atrioventricular nodal conduction disturbances. *Am Heart J* 1984;107:1183-7.
- 809. Fantidis P, Castejon R, Fernández Ruiz A, Madero-Jarabo R, Cordovilla G, Sanz Galeote E. Does a critical hemodynamic situation develop from right ventriculotomy and free wall infarct or from small changes in dysfunctional right ventricle afterload? J Cardiovasc Surg (Torino) 1992;33:229-34.
- 810. Braat SH, Ramentol M, Halders S, Wellens HJ. Reperfusion with streptokinase of an occluded right coronary artery: effects on early and late right and left ventricular ejection fraction. *Am Heart* J 1987;113:257-60.
- 811. Schuler G, Hofmann M, Schwarz F, et al. Effect of successful thrombolytic therapy on right ventricular function in acute inferior wall myocardial infarction. *Am J Cardiol* 1984;54:951-7.
- 812. Moreyra AE, Suh C, Porway MN, Kostis JB. Rapid hemodynamic improvement in right ventricular infarction after coronary angioplasty. *Chest* 1988;94:197-9.
- 813. Gott JP, Han DC. Surgical treatment of acute myocardial infarct: clinical considerations. *Semin Thorac Cardiovasc Surg* 1995;7: 198-207.
- 814. Roberts N, Harrison DG, Reimer KA, Crain BS, Wagner GS.

Right ventricular infarction with shock but without significant left ventricular infarction: a new clinical syndrome. *Am Heart J* 1985;110:1047-53.

- 815. Dell'Italia LJ, Starling MR, O'Rourke RA. Physical examination for exclusion of hemodynamically important right ventricular infarction. *Ann Intern Med* 1983;99:608-11.
- 816. Serrano Júnior CV, Ramires JA, César LA, et al. Prognostic significance of right ventricular dysfunction in patients with acute inferior myocardial infarction and right ventricular involvement. *Clin Cardiol* 1995;18:199-205.
- 817. Berger PB, Ruocco NA, Ryan TJ, et al, for the The TIMI Research Group. Frequency and significance of right ventricular dysfunction during inferior wall left ventricular myocardial infarction treated with thrombolytic therapy (results from the thrombolysis in myocardial infarction [TIMI] II trial). Am J Cardiol 1993;71:1148-52.
- 818. Andersen HR, Falk E, Nielsen D. Right ventricular infarction: frequency, size and topography in coronary heart disease: a prospective study comprising 107 consecutive autopsies from a coronary care unit. J Am Coll Cardiol 1987;10:1223-32.
- 819. Roberts N, Harrison DG, Reimer KA, Crain BS, Wagner GS. Right ventricular infarction with shock but without significant left ventricular infarction: a new clinical syndrome. *Am Heart J* 1985;110:1047-53.
- 820. Serrano Júnior CV, Ramires JA, César LA, et al. Prognostic significance of right ventricular dysfunction in patients with acute inferior myocardial infarction and right ventricular involvement. *Clin Cardiol* 1995;18:199-205.
- 821. Berger PB, Ruocco NA, Ryan TJ, et al, for the TIMI Research Group. Frequency and significance of right ventricular dysfunction during inferior wall left ventricular myocardial infarction treated with thrombolytic therapy (results from the thrombolysis in myocardial infarction [TIMI] II trial). Am J Cardiol 1993;71:1148-52.
- 822. Calvin JE. Optimal right ventricular filling pressures and the role of pericardial constraint in right ventricular infarction in dogs. *Circulation* 1991;84:852-61.
- 823. Tobinick E, Schelbert HR, Henning H, et al. Right ventricular ejection fraction in patients with acute anterior and inferior myocardial infarction assessed by radionuclide angiography. *Circulation* 1978;57:1078-84.
- 824. Boldt J, Kling D, Thiel A, Scheld HH, Hempelmann G. Revascularization of the right coronary artery: influence on thermodilution right ventricular ejection fraction. *J Cardiothoracic Anesth* 1998;2:140-146.
- 825. Polak JF, Holman BL, Wynne J, Colucci WS. Right ventricular ejection fraction: an indicator of increased mortality in patients with congestive heart failure associated with coronary artery disease. *J Am Coll Cardiol* 1983;2:217-24.
- 826. Lloyd EA, Gersh BJ, Kennelly BM. Hemodynamic spectrum of "'dominant" right ventricular infarction in 19 patients. *Am J Cardiol* 1981;48:1016-22.
- 827. Klein HO, Tordjman T, Ninio R, et al. The early recognition of right ventricular infarction: diagnostic accuracy of the electrocardiographic V4R lead. *Circulation* 1983;67:558-65.
- 828. Bellamy GR, Rasmussen HH, Nasser FN, Wiseman JC, Cooper RA. Value of two-dimensional echocardiography, electrocardiography, and clinical signs in detecting right ventricular infarction. *Am Heart J* 1986;112:304-9.
- 829. Birnbaum Y, Fishbein MC, Blanche C, Siegel RJ. Ventricular septal rupture after acute myocardial infarction. N Engl J Med 2002;347:1426-32.

- 830. Snyder RW, Glamann DB, Lange RA, et al. Predictive value of prominent pulmonary arterial wedge V waves in assessing the presence and severity of mitral regurgitation. *Am J Cardiol* 1994;73:568-70.
- 831. Dalrymple-Hay MJ, Langley SM, Sami SA, et al. Should coronary artery bypass grafting be performed at the same time as repair of a post-infarct ventricular septal defect? *Eur J Cardiothorac Surg* 1998;13:286-92.
- 832. Thompson CR, Buller CE, Sleeper LA, et al. Cardiogenic shock due to acute severe mitral regurgitation complicating acute myocardial infarction: a report from the SHOCK Trial Registry. SHould we use emergently revascularize Occluded Coronaries in cardiogenic shock? J Am Coll Cardiol 2000;36:1104-9.
- 833. Lamas GA, Mitchell GF, Flaker GC, et al, for the Survival and Ventricular Enlargement Investigators. Clinical significance of mitral regurgitation after acute myocardial infarction. *Circulation* 1997;96:827-33.
- 834. Tepe NA, Edmunds LH. Operation for acute postinfarction mitral insufficiency and cardiogenic shock. *J Thorac Cardiovasc Surg* 1985;89:525-30.
- 835. Tavakoli R, Weber A, Vogt P, Brunner HP, Pretre R, Turina M. Surgical management of acute mitral valve regurgitation due to post-infarction papillary muscle rupture. *J Heart Valve Dis* 2002;11:20-5; discussion 26.
- 836. Chen Q, Darlymple-Hay MJ, Alexiou C, et al. Mitral valve surgery for acute papillary muscle rupture following myocardial infarction. *J Heart Valve Dis* 2002;11:27-31.
- 837. Kishon Y, Oh JK, Schaff HV, Mullany CJ, Tajik AJ, Gersh BJ. Mitral valve operation in postinfarction rupture of a papillary muscle: immediate results and long-term follow-up of 22 patients. *Mayo Clin Proc* 1992;67:1023-30.
- 838. Fasol R, Lakew F, Wetter S. Mitral repair in patients with a ruptured papillary muscle. *Am Heart J* 2000;139:549-54.
- 839. Nishimura RA, Gersh BJ, Schaff HV. The case for an aggressive surgical approach to papillary muscle rupture following myocardial infarction: "from paradise lost to paradise regained". *Heart* 2000;83:611-3.
- 840. Byrne JG, Aranki SF, Cohn LH. Repair versus replacement of mitral valve for treating severe ischemic mitral regurgitation. *Coron Artery Dis* 2000;11:31-3.
- 841. Adams DH, Filsoufi F, Aklog L. Surgical treatment of the ischemic mitral valve. *J Heart Valve Dis* 2002;11:S21-5.
- 842. Crenshaw BS, Granger CB, Birnbaum Y, et al, for the GUSTO-I (Global Utilization of Streptokinase and TPA for Occluded Coronary Arteries) Trial Investigators. Risk factors, angiographic patterns, and outcomes in patients with ventricular septal defect complicating acute myocardial infarction. *Circulation* 2000; 101:27-32.
- 843. Prêtre R, Ye Q, Grünenfelder J, Lachat M, Vogt PR, Turina MI. Operative results of "repair" of ventricular septal rupture after acute myocardial infraction. *Am J Cardiol* 1999;84:785-8.
- 844. Lemery R, Smith HC, Giuliani ER, Gersh BJ. Prognosis in rupture of the ventricular septum after acute myocardial infarction and role of early surgical intervention. *Am J Cardiol* 1992;70:147-51.
- 845. Skillington PD, Davies RH, Luff AJ, et al. Surgical treatment for infarct-related ventricular septal defects: improved early results combined with analysis of late functional status. *J Thorac Cardiovasc Surg* 1990;99:798-808.
- 846. Topaz O, Taylor AL. Interventricular septal rupture complicating acute myocardial infarction: from pathophysiologic features to the role of invasive and noninvasive diagnostic modalities in cur-

e277 ACC/AHA Practice Guidelines

rent management. Am J Med 1992;93:683-8.

- 847. Westaby S, Parry A, Ormerod O, Gooneratne P, Pillai R. Thrombolysis and postinfarction ventricular septal rupture. J Thorac Cardiovasc Surg 1992;104:1506-9.
- 848. Muehrcke DD, Daggett WM, Buckley MJ, Akins CW, Hilgenberg AD, Austen WG. Postinfarct ventricular septal defect repair: effect of coronary artery bypass grafting. *Ann Thorac Surg* 1992;54:876-82; discussion 8.
- 849. Menon V, Webb JG, Hillis LD, et al. Outcome and profile of ventricular septal rupture with cardiogenic shock after myocardial infarction: a report from the SHOCK Trial Registry. SHould we emergently revascularize Occluded Coronaries in cardiogenic shocK? J Am Coll Cardiol 2000;36:1110-6.
- 850. Szkutnik M, Bialkowski J, Kusa J, et al. Postinfarction ventricular septal defect closure with Amplatzer occluders. *Eur J Cardiothorac Surg* 2003;23:323-7.
- 851. Yusuf S, Wittes J, Friedman L. Overview of results of randomized clinical trials in heart disease: I–treatments following myocardial infarction. *JAMA* 1988;260:2088-93.
- 852. Nakamura F, Minamino T, Higashino Y, et al. Cardiac free wall rupture in acute myocardial infarction: ameliorative effect of coronary reperfusion. *Clin Cardiol* 1992;15:244-50.
- 853. Pollak H, Nobis H, Mlczoch J. Frequency of left ventricular free wall rupture complicating acute myocardial infarction since the advent of thrombolysis. *Am J Cardiol* 1994;74:184-6.
- 854. Becker RC, Gore JM, Lambrew C, et al. A composite view of cardiac rupture in the United States National Registry of Myocardial Infarction. J Am Coll Cardiol 1996;27:1321-6.
- 855. Honan MB, Harrell FE, Reimer KA, et al. Cardiac rupture, mortality and the timing of thrombolytic therapy: a meta-analysis. J Am Coll Cardiol 1990;16:359-67.
- 856. Becker RC, Charlesworth A, Wilcox RG, et al. Cardiac rupture associated with thrombolytic therapy: impact of time to treatment in the Late Assessment of Thrombolytic Efficacy (LATE) study. *J Am Coll Cardiol* 1995;25:1063-8.
- 857. Núñez L, de la Llana R, López Sendón J, et al. Diagnosis and treatment of subacute free wall ventricular rupture after infarction. *Ann Thorac Surg* 1983;35:525-9.
- 858. Balakumaran K, Verbaan CJ, Essed CE, et al. Ventricular free wall rupture: sudden, subacute, slow, sealed and stabilized varieties. *Eur Heart J* 1984;5:282-8.
- 859. McMullan MH, Maples MD, Kilgore TL, Hindman SH. Surgical experience with left ventricular free wall rupture. *Ann Thorac Surg* 2001;71:1894-8; discussion 1.
- 860. Padró JM, Mesa JM, Silvestre J, et al. Subacute cardiac rupture: repair with a sutureless technique. *Ann Thorac Surg* 1993;55:20-3; discussion 23.
- 861. Slater J, Brown RJ, Antonelli TA, et al. Cardiogenic shock due to cardiac free-wall rupture or tamponade after acute myocardial infarction: a report from the SHOCK Trial Registry. Should we emergently revascularize occluded coronaries for cardiogenic shock? J Am Coll Cardiol 2000;36:1117-22.
- 862. Tikiz H, Balbay Y, Atak R, Terzi T, Genç Y, Kütük E. The effect of thrombolytic therapy on left ventricular aneurysm formation in acute myocardial infarction: relationship to successful reperfusion and vessel patency. *Clin Cardiol* 2001;24:656-62.
- 863. Premaratne S, Razzuk AM, Koduru SB, Behling A, McNamara JJ. Incidence of postinfarction aneurysm within one month of infarct: experiences with sixteen patients in Hawaii. J Cardiovasc Surg (Torino) 1999;40:473-6.
- 864. Pasini S, Gagliardotto P, Punta G, et al. Early and late results after surgical therapy of postinfarction left ventricular aneurysm. *J*

Cardiovasc Surg (Torino) 1998;39:209-15.

- 865. Ohara K. Current surgical strategy for post-infarction left ventricular aneurysm: from linear aneurysmectomy to Dor's operation. *Ann Thorac Cardiovasc Surg* 2000;6:289-94.
- 866. Di Donato M, Sabatier M, Montiglio F, et al. Outcome of left ventricular aneurysmectomy with patch repair in patients with severely depressed pump function. *Am J Cardiol* 1995;76:557-61.
- 867. Di Donato M, Toso A, Maioli M, et al, for the RESTORE Group. Intermediate survival and predictors of death after surgical ventricular restoration. *Semin Thorac Cardiovasc Surg* 2001;13:468-75.
- 868. Stevenson LW, Kormos RL, Bourge RC, et al. Mechanical cardiac support 2000: current applications and future trial design. June 15-16, 2000 Bethesda, Maryland. J Am Coll Cardiol 2001; 37:340-70.
- 869. Pennington DG, Smedira NG, Samuels LE, Acker MA, Curtis JJ, Pagani FD. Mechanical circulatory support for acute heart failure. *Ann Thorac Surg* 2001;71:S56-9; discussion S8.
- 870. Chen JM, DeRose JJ, Slater JP, et al. Improved survival rates support left ventricular assist device implantation early after myocardial infarction. *J Am Coll Cardiol* 1999;33:1903-8.
- 871. Champagnac D, Claudel JP, Chevalier P, et al. Primary cardiogenic shock during acute myocardial infarction: results of emergency cardiac transplantation. *Eur Heart J* 1993;14:925-9.
- 872. Park SJ, Nguyen DQ, Bank AJ, Ormaza S, Bolman RM. Left ventricular assist device bridge therapy for acute myocardial infarction. *Ann Thorac Surg* 2000;69:1146-51.
- 873. Hendry PJ, Masters RG, Mussivand TV, et al. Circulatory support for cardiogenic shock due to acute myocardial infarction: a Canadian experience. *Can J Cardiol* 1999;15:1090-4.
- 874. Karagueuzian HS, Mandel WJ. Electrophysiologic mechanisms of ischemic ventricular arrhythmias: experimental and clinical correlations. In Mandel WJ, ed. Cardiac arrhythmias: their mechanisms, diagnosis, and management. Philadelphia, PA: JB Lippincott;1995:563-603.
- 875. Carmeliet E. Cardiac ionic currents and acute ischemia: from channels to arrhythmias. *Physiol Rev* 1999;79:917-1017.
- 876. Campbell RWF. Arrhythmias. In: Julian DG. Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co. Ltd;1994:223-240.
- 877. Nordrehaug JE, von der Lippe G. Hypokalaemia and ventricular fibrillation in acute myocardial infarction. *Br Heart J* 1983;50:525-9.
- 878. Higham PD, Adams PC, Murray A, Campbell RW. Plasma potassium, serum magnesium and ventricular fibrillation: a prospective study. *Q J Med* 1993;86:609-17.
- 879. Volpi A, Cavalli A, Santoro E, Tognoni G, for the GISSI Investigators. Incidence and prognosis of secondary ventricular fibrillation in acute myocardial infarction: evidence for a protective effect of thrombolytic therapy. *Circulation* 1990;82:1279-88.
- 880. Ornato JP, Peberdy MA, Tadler SC, Strobos NC. Factors associated with the occurrence of cardiac arrest during hospitalization for acute myocardial infarction in the second national registry of myocardial infarction in the US. *Resuscitation* 2001;48:117-23.
- 881. Campbell RW, Murray A, Julian DG. Ventricular arrhythmias in first 12 hours of acute myocardial infarction: natural history study. *Br Heart J* 1981;46:351-7.
- 882. Antman EM, Berlin JA. Declining incidence of ventricular fibrillation in myocardial infarction: implications for the prophylactic use of lidocaine. *Circulation* 1992;86:764-73.
- 883. Thompson CA, Yarzebski J, Goldberg RJ, Lessard D, Gore JM, Dalen JE. Changes over time in the incidence and case-fatality

rates of primary ventricular fibrillation complicating acute myocardial infarction: perspectives from the Worcester Heart Attack Study. *Am Heart J* 2000;139:1014-21.

- 884. Behar S, Goldbourt U, Reicher-Reiss H, Kaplinsky E, for the Principal Investigators of the SPRINT Study. Prognosis of acute myocardial infarction complicated by primary ventricular fibrillation. *Am J Cardiol* 1990;66:1208-11.
- 885. Lown B, Fakhro AM, Hood WB, Thorn GW. The coronary care unit: new perspectives and directions. *JAMA* 1967;199:188-98.
- 886. Dhurandhar RW, MacMillan RL, Brown KW. Primary ventricular fibrillation complicating acute myocardial infarction. Am J Cardiol 1971;27:347-51.
- 887. El-Sherif N, Myerburg RJ, Scherlag BJ, et al. Electrocardiographic antecedents of primary ventricular fibrillation: value of the R-on-T phenomenon in myocardial infarction. *Br Heart J* 1976;38:415-22.
- Lie KI, Wellens HJ, Durrer D. Characteristics and predictability of primary ventricular fibrillation. *Eur J Cardiol* 1974;1:379-84.
- 889. Solomon SD, Ridker PM, Antman EM. Ventricular arrhythmias in trials of thrombolytic therapy for acute myocardial infarction: a meta-analysis. *Circulation* 1993;88:2575-81.
- 890. MacMahon S, Collins R, Peto R, Koster RW, Yusuf S. Effects of prophylactic lidocaine in suspected acute myocardial infarction: an overview of results from the randomized, controlled trials. *JAMA* 1988;260:1910-6.
- 891. Alexander JH, Granger CB, Sadowski Z, et al, for the GUSTO-I and GUSTO-IIb Investigators. Prophylactic lidocaine use in acute myocardial infarction: incidence and outcomes from two international trials. *Am Heart J* 1999;137:799-805.
- 892. Hjalmarson A, Herlitz J, Holmberg S, et al. The Göteborg metoprolol trial: effects on mortality and morbidity in acute myocardial infarction. *Circulation* 1983;67:I26-32.
- 893. Guidelines for cardiopulmonary resuscitation and emergency cardiac care: part III–adult advanced cardiac life support. Emergency Cardiac Care Committee and Subcommittees, American Heart Association. JAMA 1992;268:2199-241.
- 894. de Vreede-Swagemakers JJ, Gorgels AP, Dubois-Arbouw WI, et al. Circumstances and causes of out-of-hospital cardiac arrest in sudden death survivors. *Heart* 1998;79:356-61.
- 895. Kudenchuk PJ, Cobb LA, Copass MK, et al. Amiodarone for resuscitation after out-of-hospital cardiac arrest due to ventricular fibrillation. N Engl J Med 1999;341:871-8.
- 896. Dorian P, Cass D, Schwartz B, Cooper R, Gelaznikas R, Barr A. Amiodarone as compared with lidocaine for shock-resistant ventricular fibrillation. *N Engl J Med* 2002;346:884-90.
- 897. Haynes RE, Chinn TL, Copass MK, Cobb LA. Comparison of bretylium tosylate and lidocaine in management of out of hospital ventricular fibrillation: a randomized clinical trial. *Am J Cardiol* 1981;48:353-6.
- 898. Olson DW, Thompson BM, Darin JC, Milbrath MH. A randomized comparison study of bretylium tosylate and lidocaine in resuscitation of patients from out-of-hospital ventricular fibrillation in a paramedic system. *Ann Emerg Med* 1984;13:807-10.
- 899. Stiell IG, Wells GA, Hebert PC, Laupacis A, Weitzman BN. Association of drug therapy with survival in cardiac arrest: limited role of advanced cardiac life support drugs. *Acad Emerg Med* 1995;2:264-73.
- 900. Guidelines 2000 for cardiopulmonary resuscitation and emergency cardiovascular care: part 6–advanced cardiovascular life support: section 6–pharmacology II: agents to optimize cardiac output and blood pressure. The American Heart Association in collaboration with the International Liaison Committee on

Resuscitation. Circulation 2000;102:I129-35.

- 901. Eldar M, Sievner Z, Goldbourt U, Reicher-Reiss H, Kaplinsky E, Behar S, for the SPRINT Study Group. Primary ventricular tachycardia in acute myocardial infarction: clinical characteristics and mortality. *Ann Intern Med* 1992;117:31-6.
- 902. Wolfe CL, Nibley C, Bhandari A, Chatterjee K, Scheinman M. Polymorphous ventricular tachycardia associated with acute myocardial infarction. *Circulation* 1991;84:1543-51.
- 903. Berger PB, Ruocco NA, Ryan TJ, Frederick MM, Podrid PJ. Incidence and significance of ventricular tachycardia and fibrillation in the absence of hypotension or heart failure in acute myocardial infarction treated with recombinant tissue-type plasminogen activator: results from the Thrombolysis in Myocardial Infarction (TIMI) Phase II trial. *J Am Coll Cardiol* 1993;22:1773-9.
- 904. Amiodarone Trials Meta-Analysis Investigators. Effect of prophylactic amiodarone on mortality after acute myocardial infarction and in congestive heart failure: meta-analysis of individual data from 6500 patients in randomised trials. *Lancet* 1997;350:1417-24.
- 905. Nademanee K, Taylor R, Bailey WE, et al. Management and longterm outcome of patients with electrical storm [abstr]. J Am Coll Cardiol 1995;125:187A.
- 906. Scheinman MM, Levine JH, Cannom DS, et al, for the Intravenous Amiodarone Multicenter Investigators Group. Doseranging study of intravenous amiodarone in patients with lifethreatening ventricular tachyarrhythmias. *Circulation* 1995; 92:3264-72.
- 907. Nademanee K, Taylor R, Bailey WE, Rieders DE, Kosar EM. Treating electrical storm: sympathetic blockade versus advanced cardiac life support-guided therapy. *Circulation* 2000;102:742-7.
- 908. Correction: a randomized study of the prevention of sudden death in patients with coronary artery disease. *N Engl J Med* 2000;342:1300.
- 909. Shea S, Bigger JT, Campion J, et al. Enrollment in clinical trials: institutional factors affecting enrollment in the cardiac arrhythmia suppression trial (CAST). *Control Clin Trials* 1992;13:466-86.
- 910. Newby KH, Thompson T, Stebbins A, Topol EJ, Califf RM, Natale A, for the GUSTO Investigators. Sustained ventricular arrhythmias in patients receiving thrombolytic therapy: incidence and outcomes. *Circulation* 1998;98:2567-73.
- 911. Al-Khatib SM, Stebbins AL, Califf RM, et al. Sustained ventricular arrhythmias and mortality among patients with acute myocardial infarction: results from the GUSTO-III trial. *Am Heart J* 2003;145:515-21.
- 912. The Antiarrhythmics versus Implantable Defibrillators (AVID) Investigators. A comparison of antiarrhythmic-drug therapy with implantable defibrillators in patients resuscitated from near-fatal ventricular arrhythmias. *N Engl J Med* 1997;337:1576-83.
- 913. Siebels J, Kuck KH. Implantable cardioverter defibrillator compared with antiarrhythmic drug treatment in cardiac arrest survivors (the Cardiac Arrest Study Hamburg). *Am Heart J* 1994;127:1139-44.
- 914. Connolly SJ, Gent M, Roberts RS, et al. Canadian implantable defibrillator study (CIDS): a randomized trial of the implantable cardioverter defibrillator against amiodarone. *Circulation* 2000;101:1297-302.
- 915. Moss AJ, Hall WJ, Cannom DS, et al, for the Multicenter Automatic Defibrillator Implantation Trial Investigators. Improved survival with an implanted defibrillator in patients with coronary disease at high risk for ventricular arrhythmia. *N Engl J Med* 1996;335:1933-40.
- 916. Buxton AE, Lee KL, Fisher JD, et al, for the Multicenter

Unsustained Tachycardia Trial Investigators. A randomized study of the prevention of sudden death in patients with coronary artery disease. *N Engl J Med* 1999;341:1882-90.

- 917. Moss AJ, Zareba W, Hall WJ, et al, for the Multicenter Automatic Defibrillator Implantation Trial II Investigators. Prophylactic implantation of a defibrillator in patients with myocardial infarction and reduced ejection fraction. *N Engl J Med* 2002;346:877-83.
- 918. Lee KL, Hafley G, Fisher JD, et al, for the Multicenter Unsustained Tachycardia Trial Investigators. Effect of implantable defibrillators on arrhythmic events and mortality in the multicenter unsustained tachycardia trial. *Circulation* 2002;106:233-8.
- 919. Bigger JT, Fleiss JL, Kleiger R, Miller JP, Rolnitzky LM. The relationships among ventricular arrhythmias, left ventricular dys-function, and mortality in the 2 years after myocardial infarction. *Circulation* 1984;69:250-8.
- 920. Buxton AE, Marchlinski FE, Waxman HL, Flores BT, Cassidy DM, Josephson ME. Prognostic factors in nonsustained ventricular tachycardia. *Am J Cardiol* 1984;53:1275-9.
- 921. Buxton AE, Lee KL, DiCarlo L, et al, for the Multicenter Unsustained Tachycardia Trial Investigators. Electrophysiologic testing to identify patients with coronary artery disease who are at risk for sudden death. *N Engl J Med* 2000;342:1937-45.
- 922. Buxton AE, Lee KL, Hafley GE, et al, for the MUSTT Investigators. Relation of ejection fraction and inducible ventricular tachycardia to mode of death in patients with coronary artery disease: an analysis of patients enrolled in the multicenter unsustained tachycardia trial. *Circulation* 2002;106:2466-72.
- 923. Solomon SD, Glynn RJ, Greaves S, et al. Recovery of ventricular function after myocardial infarction in the reperfusion era: the healing and early afterload reducing therapy study. *Ann Intern Med* 2001;134:451-8.
- 924. Ureña PE, Lamas GA, Mitchell G, et al, for the Survival and Ventricular Enlargement (SAVE) Investigators. Ejection fraction by radionuclide ventriculography and contrast left ventriculogram: a tale of two techniques. *J Am Coll Cardiol* 1999;33:180-5.
 925. Deleted in press.
- 926. Auricchio A, Klein H, Geller CJ, Reek S, Heilman MS, Szymkiewicz SJ. Clinical efficacy of the wearable cardioverterdefibrillator in acutely terminating episodes of ventricular fibrillation. Am J Cardiol 1998;81:1253-6.
- 927. Kontoyannis DA, Anastasiou-Nana MI, Kontoyannis SA, Zaga AK, Nanas JN. Intravenous amiodarone decreases the duration of atrial fibrillation associated with acute myocardial infarction. *Cardiovasc Drugs Ther* 2001;15:155-60.
- 928. Nielsen FE, Andersen HH, Gram-Hansen P, Sørensen HT, Klausen IC. The relationship between ECG signs of atrial infarction and the development of supraventricular arrhythmias in patients with acute myocardial infarction. *Am Heart J* 1992;123:69-72.
- 929. Kyriakidis M, Barbetseas J, Antonopoulos A, Skouros C, Tentolouris C, Toutouzas P. Early atrial arrhythmias in acute myocardial infarction: role of the sinus node artery. *Chest* 1992;101:944-7.
- 930. Hildebrandt P, Jensen G, Køber L, et al. Myocardial infarction 1979-1988 in Denmark: secular trends in age-related incidence, in-hospital mortality and complications. *Eur Heart J* 1994;15:877-81.
- 931. Waldo AL. An approach to therapy of supraventricular tachyarrhythmias: an algorithm versus individualized therapy. *Clin Cardiol* 1994;17:II21-6.

- 932. Liberthson RR, Salisbury KW, Hutter AM, DeSanctis RW. Atrial tachyarrhythmias in acute myocardial infarction. *Am J Med* 1976;60:956-60.
- 933. Kobayashi Y, Katoh T, Takano T, Hayakawa H. Paroxysmal atrial fibrillation and flutter associated with acute myocardial infarction: hemodynamic evaluation in relation to the development of arrhythmias and prognosis. *Jpn Circ J* 1992;56:1-11.
- 934. Rathore SS, Berger AK, Weinfurt KP, et al. Acute myocardial infarction complicated by atrial fibrillation in the elderly: prevalence and outcomes. *Circulation* 2000;101:969-74.
- 935. Pedersen OD, Bagger H, Køber L, Torp-Pedersen C, for the TRAndolapril Cardiac Evaluation (TRACE) Study Group. The occurrence and prognostic significance of atrial fibrillation/-flutter following acute myocardial infarction. *Eur Heart J* 1999; 20:748-54.
- 936. Goldberg RJ, Seeley D, Becker RC, et al. Impact of atrial fibrillation on the in-hospital and long-term survival of patients with acute myocardial infarction: a community-wide perspective. *Am Heart J* 1990;119:996-1001.
- 937. Hod H, Lew AS, Keltai M, et al. Early atrial fibrillation during evolving myocardial infarction: a consequence of impaired left atrial perfusion. *Circulation* 1987;75:146-50.
- 938. Rechavia E, Strasberg B, Mager A, et al. The incidence of atrial arrhythmias during inferior wall myocardial infarction with and without right ventricular involvement. *Am Heart J* 1992;124:387-91.
- 939. Behar S, Tanne D, Zion M, et al, for the Secondary Prevention Reinfarction Israeli Nifedipine Trial (SPRINT) Study Group. Incidence and prognostic significance of chronic atrial fibrillation among 5,839 consecutive patients with acute myocardial infarction. *Am J Cardiol* 1992;70:816-8.
- 940. James TN. Myocardial infarction and atrial arrhythmias. *Circulation* 1961;24:761-76.
- 941. Nielsen FE, Sørensen HT, Christensen JH, Ravn L, Rasmussen SE. Reduced occurrence of atrial fibrillation in acute myocardial infarction treated with streptokinase. *Eur Heart J* 1991;12:1081-3.
- 942. Eldar M, Canetti M, Rotstein Z, et al, for the SPRINT and Thrombolytic survey groups. Significance of paroxysmal atrial fibrillation complicating acute myocardial infarction in the thrombolytic era. *Circulation* 1998;97:965-70.
- 943. Behar S, Zahavi Z, Goldbourt U, Reicher-Reiss H, for the SPRINT Study Group. Long-term prognosis of patients with paroxysmal atrial fibrillation complicating acute myocardial infarction. *Eur Heart J* 1992;13:45-50.
- 944. Crenshaw BS, Ward SR, Granger CB, Stebbins AL, Topol EJ, Califf RM. Atrial fibrillation in the setting of acute myocardial infarction: the GUSTO-I experience. Global Utilization of Streptokinase and TPA for Occluded Coronary Arteries. *J Am Coll Cardiol* 1997;30:406-13.
- 945. Wong CK, White HD, Wilcox RG, et al. New atrial fibrillation after acute myocardial infarction independently predicts death: the GUSTO-III experience. *Am Heart J* 2000;140:878-85.
- 946. Incidence and prognostic significance of atrial fibrillation in acute myocardial infarction: the GISSI-3 data. *Heart* 2001;86:527-32.
- 947. Lesser MF. Safety and efficacy of in-office cardioversion for treatment of supraventricular arrhythmias. *Am J Cardiol* 1990; 66:1267-8.
- 948. Joglar JA, Hamdan MH, Ramaswamy K, et al. Initial energy for elective external cardioversion of persistent atrial fibrillation. *Am J Cardiol* 2000;86:348-50.
- 949. Dahl CF, Ewy GA, Warner ED, Thomas ED. Myocardial necrosis from direct current countershock. Effect of paddle electrode size

and time interval between discharges. *Circulation* 1974;50:956-61.

- 950. Moss AJ, Oakes D, Benhorin J, Carleen E, for the Multicenter Diltiazem Post-Infarction Research Group. The interaction between diltiazem and left ventricular function after myocardial infarction. *Circulation* 1989;80:IV102-6.
- 951. Clemo HF, Wood MA, Gilligan DM, Ellenbogen KA. Intravenous amiodarone for acute heart rate control in the critically ill patient with atrial tachyarrhythmias. *Am J Cardiol* 1998;81:594-8.
- 952. Rawles JM, Metcalfe MJ, Jennings K. Time of occurrence, duration, and ventricular rate of paroxysmal atrial fibrillation: the effect of digoxin. *Br Heart J* 1990;63:225-7.
- 953. Murgatroyd FD, Gibson SM, Baiyan X, et al. Double-blind placebo-controlled trial of digoxin in symptomatic paroxysmal atrial fibrillation. *Circulation* 1999;99:2765-70.
- 954. Farshi R, Kistner D, Sarma JS, Longmate JA, Singh BN. Ventricular rate control in chronic atrial fibrillation during daily activity and programmed exercise: a crossover open-label study of five drug regimens. *J Am Coll Cardiol* 1999;33:304-10.
- 955. Fuster V, Ryden LE, Asinger RW, et al. ACC/AHA/ESC guidelines for the management of patients with atrial fibrillation: executive summary: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines and the European Society of Cardiology Committee for Practice Guidelines and Policy Conferences (Committee to Develop Guidelines for the Management of Patients With Atrial Fibrillation). J Am Coll Cardiol 2001;38:1231-66.
- 956. Lavie CJ, Gersh BJ. Mechanical and electrical complications of acute myocardial infarction. *Mayo Clin Proc* 1990;65:709-30.
- 957. Wong CK, White HD, Wilcox RG, et al, for the GUSTO-III Investigators. Management and outcome of patients with atrial fibrillation during acute myocardial infarction: the GUSTO-III experience. Global use of strategies to open occluded coronary arteries. *Heart* 2002;88:357-62.
- 958. A comparison of rate control and rhythm control in patients with atrial fibrillation. *N Engl J Med* 2002;347:1825-33.
- 959. Antman EM. General hospital management. In: Julian DG, Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co Ltd; 1994:42-4.
- 960. Brady WJ, Harrigan RA. Diagnosis and management of bradycardia and atrioventricular block associated with acute coronary ischemia. *Emerg Med Clin North Am* 2001;19:371-84, xi-xii.
- 961. Pasternak RC, Braunwald E, Sobel BE. Acute myocardial infarction: In: Braunwald E, ed. Heart disease: a textbook of cardiovascular medicine. Philadelphia, Pa: WB Saunders Co Ltd; 1992:1240-9.
- 962. Berger PB, Ruocco NA, Ryan TJ, Frederick MM, Jacobs AK, Faxon DP. Incidence and prognostic implications of heart block complicating inferior myocardial infarction treated with thrombolytic therapy: results from TIMI II. J Am Coll Cardiol 1992;20:533-40.
- 963. McDonald K, O'Sullivan JJ, Conroy RM, Robinson K, Mulcahy R. Heart block as a predictor of in-hospital death in both acute inferior and acute anterior myocardial infarction. *Q J Med* 1990;74:277-82.
- 964. Nicod P, Gilpin E, Dittrich H, Polikar R, Henning H, Ross J. Long-term outcome in patients with inferior myocardial infarction and complete atrioventricular block. J Am Coll Cardiol 1988;12:589-94.
- 965. Antman EM. Cardiovascular therapeutics: a companion guide to Braunwald's heart disease. 2nd ed. Philadelphia, Pa: WB Saunders Co Ltd; 2001:273.

- 966. Mahr NC, Valdes A, Lamas G. Use of glucagon for acute intravenous diltiazem toxicity. *Am J Cardiol* 1997;79:1570-1.
- 967. Topol EJ, Goldschlager N, Ports TA, et al. Hemodynamic benefit of atrial pacing in right ventricular myocardial infarction. *Ann Intern Med* 1982;96:594-7.
- 968. Mavric Z, Zaputovic L, Matana A, et al. Prognostic significance of complete atrioventricular block in patients with acute inferior myocardial infarction with and without right ventricular involvement. *Am Heart J* 1990;119:823-8.
- 969. Wenzel V, Krismer AC, Arntz HR, Sitter H, Stadlbauer KH, Lindner KH, for the European Resuscitation Council Vasopressor during Cardiopulmonary Resuscitation Study Group. A comparison of vasopressin and epinephrine for out-of-hospital cardiopulmonary resuscitation. *N Engl J Med* 2004;350:105-13.
- 970. Hindman MC, Wagner GS, JaRo M, et al. The clinical significance of bundle branch block complicating acute myocardial infarction: 2.–Indications for temporary and permanent pacemaker insertion. *Circulation* 1978;58:689-99.
- 971. Col JJ, Weinberg SL. The incidence and mortality of intraventricular conduction defects in acute myocardial infarction. Am J Cardiol 1972;29:344-50.
- 972. Ritter WS, Atkins JM, Blomqvist CG, Mullins CB. Permanent pacing in patients with transient trifascicular block during acute myocardial infarction. *Am J Cardiol* 1976;38:205-8.
- 973. Ginks WR, Sutton R, Oh W, Leatham A. Long-term prognosis after acute anterior infarction with atrioventricular block. *Br Heart J* 1977;39:186-9.
- 974. Domenighetti G, Perret C. Intraventricular conduction disturbances in acute myocardial infarction: short- and long-term prognosis. *Eur J Cardiol* 1980;11:51-9.
- 975. Ranganathan N, Dhurandhar R, Phillips JH, Wigle ED. His Bundle electrogram in bundle-branch block. *Circulation* 1972;45:282-94.
- 976. Lamas GA, Muller JE, Turi ZG, et al. A simplified method to predict occurrence of complete heart block during acute myocardial infarction. *Am J Cardiol* 1986;57:1213-9.
- 977. Harthorne JW, Barold SS. Atherosclerosis, the conduction system and cardiac pacing. In: Fuster V, Ross R, Topol EJ, eds. Atherosclerosis and coronary artery disease. Philadelphia, PA: Lippincott-Raven Publishers;1996:1013-30.
- 978. Juma Z, Castellanos A, Myerburg RJ. Prognostic significance of the electrocardiogram in patients with coronary heart disease. In: Wellens HJJ, Kulbertus HE, eds. What's new in electrocardiography. The Hague: Martinus Nijhoff Publishers; 1981:5-22.
- 979. Clemmensen P, Bates ER, Califf RM, et al, for the TAMI Study Group. Complete atrioventricular block complicating inferior wall acute myocardial infarction treated with reperfusion therapy. *Am J Cardiol* 1991;67:225-30.
- 980. Goldberg RJ, Zevallos JC, Yarzebski J, et al. Prognosis of acute myocardial infarction complicated by complete heart block (the Worcester Heart Attack Study). *Am J Cardiol* 1992;69:1135-41.
- 981. Behar S, Zissman E, Zion M, et al, for the SPRINT Study Group. Prognostic significance of second-degree atrioventricular block in inferior wall acute myocardial infarction. *Am J Cardiol* 1993;72:831-4.
- 982. Lie KL, Wellens HJJ, Schuilenburg RM. Bundle branch block and acute myocardial infarction. In: Wellens HJJ, Lie KI, Janse MD, Eds. The conduction system of the heart: structure, function and clinical implications. Philadelphia, Pa: Lea & Febiger;1976:662-72.
- 983. Dubois C, Piérard LA, Smeets JP, Carlier J, Kulbertus HE. Longterm prognostic significance of atrioventricular block in inferior

e281 ACC/AHA Practice Guidelines

acute myocardial infarction. Eur Heart J 1989;10:816-20.

- 984. Gregoratos G, Abrams J, Epstein AE, et al. ACC/AHA/NASPE 2002 guideline update for implantation of cardiac pacemakers and antiarrhythmia devices: summary article: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (ACC/AHA/NASPE Committee to Update the 1998 Pacemaker Guidelines). J Am Coll Cardiol 2002; 40:1703-19.
- 985. Wilkoff BL, Cook JR, Epstein AE, et al, for the Dual Chamber and VVI Implantable Defibrillator Trial Investigators. Dualchamber pacing or ventricular backup pacing in patients with an implantable defibrillator: the Dual Chamber and VVI Implantable Defibrillator (DAVID) Trial. JAMA 2002;288:3115-23.
- 986. Lamas GA, Ellenbogen KA. Evidence base for pacemaker mode selection: from physiology to randomized trials. *Circulation* 2004;109:443-51.
- 987. Rude RE, Poole WK, Muller JE, et al. Electrocardiographic and clinical criteria for recognition of acute myocardial infarction based on analysis of 3,697 patients. *Am J Cardiol* 1983;52:936-42.
- 988. Tofler GH, Muller JE, Stone PH, et al. Pericarditis in acute myocardial infarction: characterization and clinical significance. *Am Heart J* 1989;117:86-92.
- 989. Wall TC, Califf RM, Harrelson-Woodlief L, et al, for the TAMI Study Group. Usefulness of a pericardial friction rub after thrombolytic therapy during acute myocardial infarction in predicting amount of myocardial damage. *Am J Cardiol* 1990;66:1418-21.
- 990. Oliva PB, Hammill SC. The clinical distinction between regional postinfarction pericarditis and other causes of postinfarction chest pain: ancillary observations regarding the effect of lytic therapy upon the frequency of postinfarction pericarditis, postinfarction angina, and reinfarction. *Clin Cardiol* 1994;17:471-8.
- 991. Cheitlin MD, Alpert JS, Armstrong WF, et al. ACC/AHA guidelines for the clinical application of echocardiography: executive summary: a report of the American College of Cardiology/ American Heart Association Task Force on Practice Guidelines (Committee on Clinical Application of Echocardiography). J Am Coll Cardiol 1997;29:862-79.
- 992. Oliva PB, Hammill SC, Talano JV. T wave changes consistent with epicardial involvement in acute myocardial infarction: observations in patients with a postinfarction pericardial effusion without clinically recognized postinfarction pericarditis. J Am Coll Cardiol 1994;24:1073-7.
- 993. Widimský P, Gregor P. Recent atrial fibrillation in acute myocardial infarction: a sign of pericarditis. Cor Vasa 1993;35:230-2.
- 994. Erhardt LR. Clinical and pathological observations in different types of acute myocardial infarction. *Acta Med Scand* 1974; 560:1-78.
- 995. Spodick D. Pericardial complications of myocardial infarction. In: Francis GS, Alpert JS, eds. Modern coronary care. Boston, MA: Little Brown and Co; 1990:331-9.
- 996. Shahar A, Hod H, Barabash GM, Kaplinsky E, Motro M. Disappearance of a syndrome: Dressler's syndrome in the era of thrombolysis. *Cardiology* 1994;85:255-8.
- 997. Jain A. "Tombstone" anterior ST-segment elevations secondary to acute pericarditis: the role of two-dimensional echocardiogram. *Clin Cardiol* 1997;20:404-6.
- 998. Bonnefoy E, Godon P, Kirkorian G, Fatemi M, Chevalier P, Touboul P. Serum cardiac troponin I and ST-segment elevation

in patients with acute pericarditis. Eur Heart J 2000;21:832-6.

- 999. Berman J, Haffajee CI, Alpert JS. Therapy of symptomatic pericarditis after myocardial infarction: retrospective and prospective studies of aspirin, indomethacin, prednisone, and spontaneous resolution. *Am Heart J* 1981;101:750-3.
- 1000. Bulkley BH, Roberts WC. Steroid therapy during acute myocardial infarction: a cause of delayed healing and of ventricular aneurysm. *Am J Med* 1974;56:244-50.
- 1001. Kloner RA, Fishbein MC, Lew H, Maroko PR, Braunwald E. Mummification of the infarcted myocardium by high-dose corticosteroids. *Circulation* 1978;57:56-63.
- 1002. Taylor SD, Chey WY, Scheiman JM. Acid secretion during indomethacin therapy: effect of misoprostol. J Clin Gastroenterol 1995;20:131-5.
- 1003. Weir MR, Klassen DK, Hall PS, et al. Minimization of indomethacin-induced reduction in renal function by misoprostol. J Clin Pharmacol 1991;31:729-35.
- 1004. Spodick DH. Pericardial diseases. Eds. Braunwald E, Zipes DP, Libby P. Heart disease: a textbook of cardiovascular medicine. 6th ed. Philadelphia, PA: WB Saunders Co Ltd; 2001:1823.
- 1005. O'Rourke RA, Fuster V, Alexander RW, Roberts R, King, SB III, Welens HJJ. Hurst's the heart. 10th ed. Manual of cardiology. United States: McGraw-Hill Companies, Inc; 2001:2066.
- 1006. Adler Y, Finkelstein Y, Guindo J, et al. Colchicine treatment for recurrent pericarditis: a decade of experience. *Circulation* 1998;97:2183-5.
- 1007. Brucato A, Cimaz R, Balla E. Prevention of recurrences of corticosteroid-dependent idiopathic pericarditis by colchicine in an adolescent patient. *Pediatr Cardiol* 2000;21:395-6.
- 1008. Troughton RW, Asher CR, Klein AL. Pericarditis. *Lancet* 2004;363:717-27.
- 1009. Antman EM. General hospital management. In: Julian DG, Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co Ltd; 1994:42-44.
- 1010. Hutchins GM, Bulkley BH. Infarct expansion versus extension: two different complications of acute myocardial infarction. Am J Cardiol 1978;41:1127-32.
- 1011. Ohman EM, Califf RM, Topol EJ, et al, for the TAMI Study Group. Consequences of reocclusion after successful reperfusion therapy in acute myocardial infarction. *Circulation* 1990;82:781-91.
- 1012. Antman E, Braunwald E. Acute myocardial infarction. In: Braunwald E, Zipes DP, Libby P, eds. Heart disease: a textbook of cardiovascular medicine, 6th ed. Philadelphia, PA: WB Saunders Co Ltd; 2001:1114-1251.
- 1013. Tanne D, Gottlieb S, Hod H, Reicher-Reiss H, Boyko V, Behar S, for the Secondary Prevention Reinfarction Israeli Nifedipine Trial (SPRINT) and Israeli Thrombolytic Survey Groups. Incidence and mortality from early stroke associated with acute myocardial infarction in the prethrombolytic and thrombolytic eras. J Am Coll Cardiol 1997;30:1484-90.
- 1014. Mooe T, Eriksson P, Stegmayr B. Ischemic stroke after acute myocardial infarction: a population-based study. *Stroke* 1997; 28:762-7.
- 1015. Mahaffey KW, Granger CB, Sloan MA, et al. Risk factors for inhospital nonhemorrhagic stroke in patients with acute myocardial infarction treated with thrombolysis: results from GUSTO-I. *Circulation* 1998;97:757-64.
- 1016. Prieto A, Eisenberg J, Thakur RK. Nonarrhythmic complications of acute myocardial infarction. *Emerg Med Clin North Am* 2001; 19:397-415, xii-xiii.
- 1017. Loh E, Sutton MS, Wun CC, et al. Ventricular dysfunction and

the risk of stroke after myocardial infarction. N Engl J Med 1997;336:251-7.

- 1018. Bodenheimer MM, Sauer D, Shareef B, Brown MW, Fleiss JL, Moss AJ. Relation between myocardial infarct location and stroke. J Am Coll Cardiol 1994;24:61-6.
- 1019. Tanne D, Goldbourt U, Zion M, Reicher-Reiss H, Kaplinsky E, Behar S, for the SPRINT Study Group. Frequency and prognosis of stroke/TIA among 4808 survivors of acute myocardial infarction. *Stroke* 1993;24:1490-5.
- 1020. Spirito P, Bellotti P, Chiarella F, Domenicucci S, Sementa A, Vecchio C. Prognostic significance and natural history of left ventricular thrombi in patients with acute anterior myocardial infarction: a two-dimensional echocardiographic study. *Circulation* 1985;72:774-80.
- 1021. Celik S, Ovali E, Baykan M, et al. Factor V Leiden and its relation to left ventricular thrombus in acute myocardial infarction. *Acta Cardiol* 2001;56:1-6.
- 1022. Stratton JR, Resnick AD. Increased embolic risk in patients with left ventricular thrombi. *Circulation* 1987;75:1004-11.
- 1023. Nordrehaug JE, Johannessen KA, von der Lippe G. Usefulness of high-dose anticoagulants in preventing left ventricular thrombus in acute myocardial infarction. *Am J Cardiol* 1985;55:1491-3.
- 1024. Vaitkus PT, Berlin JA, Schwartz JS, Barnathan ES. Stroke complicating acute myocardial infarction: a meta-analysis of risk modification by anticoagulation and thrombolytic therapy. *Arch Intern Med* 1992;152:2020-4.
- 1025. Arvan S, Boscha K. Prophylactic anticoagulation for left ventricular thrombi after acute myocardial infarction: a prospective randomized trial. *Am Heart J* 1987;113:688-93.
- 1026. Eigler N, Maurer G, Shah PK. Effect of early systemic thrombolytic therapy on left ventricular mural thrombus formation in acute anterior myocardial infarction. Am J Cardiol 1984;54:261-3.
- 1027. Motro M, Barbash GI, Hod H, et al. Incidence of left ventricular thrombi formation after thrombolytic therapy with recombinant tissue plasminogen activator, heparin, and aspirin in patients with acute myocardial infarction. *Am Heart J* 1991;122:23-6.
- 1028. Heik SC, Kupper W, Hamm C, et al. Efficacy of high-dose intravenous heparin for treatment of left ventricular thrombi with high embolic risk. J Am Coll Cardiol 1994;24:1305-9.
- 1029. Barnett HJ, Taylor DW, Eliasziw M, et al, for the North American Symptomatic Carotid Endarterectomy Trial Collaborators. Benefit of carotid endarterectomy in patients with symptomatic moderate or severe stenosis. *N Engl J Med* 1998;339:1415-25.
- 1030. Merlini PA, Bauer KA, Oltrona L, et al. Persistent activation of coagulation mechanism in unstable angina and myocardial infarction. *Circulation* 1994;90:61-8.
- 1031. AIMS Trial Study Group, Effect of intravenous APSAC on mortality after acute myocardial infarction: preliminary report of a placebo-controlled clinical trial. *Lancet* 1988;1:545-9.
- 1032. Ringleb PA, Bhatt DL, Hirsch AT, Topol EJ, Hacke W, for the Clopidogrel Versus Aspirin in Patients at Risk of Ischemic Events Investigators. Benefit of clopidogrel over aspirin is amplified in patients with a history of ischemic events. *Stroke* 2004;35:528-32.
- 1033. Diener HC, Cunha L, Forbes C, Sivenius J, Smets P, Lowenthal A. European Stroke Prevention Study: 2. Dipyridamole and acetylsalicylic acid in the secondary prevention of stroke. J Neurol Sci 1996;143:1-13.
- 1034. Rothwell PM, Gutnikov SA, Warlow CP, for the European Carotid Surgery Trialists Collaboration. Reanalysis of the final

results of the European Carotid Surgery Trial. *Stroke* 2003;34:514-23.

- 1035. Rothwell PM, Eliasziw M, Gutnikov SA, et al, for the Carotid Endarterectomy Trialists' Collaboration. Analysis of pooled data from the randomised controlled trials of endarterectomy for symptomatic carotid stenosis. *Lancet* 2003;361:107-16.
- 1036. Wholey MH, Wholey M, Mathias K, et al. Global experience in cervical carotid artery stent placement. *Catheter Cardiovasc Interv* 2000;50:160-7.
- 1037. Malek AM, Higashida RT, Phatouros CC, et al. Stent angioplasty for cervical carotid artery stenosis in high-risk symptomatic NASCET-ineligible patients. *Stroke* 2000;31:3029-33.
- 1038. Deleted in press.
- 1039. Turpie AG, Chin BS, Lip GY. Venous thromboembolism: pathophysiology, clinical features, and prevention. *BMJ* 2002; 325:887-90.
- 1040. Geerts WH, Heit JA, Clagett GP, et al. Prevention of venous thromboembolism. *Chest* 2001;119:132S-175S.
- 1041. Green L, Fay W, Frey KA, et al. Diagnosis and Treatment of Venous Thromboembolic Disease Guideline. University of Michigan Health System. National Guidelines Clearinghouse 2004. In press.
- 1042. Dolovich LR, Ginsberg JS, Douketis JD, Holbrook AM, Cheah G A meta-analysis comparing low-molecular-weight heparins with unfractionated heparin in the treatment of venous thromboembolism: examining some unanswered questions regarding location of treatment, product type, and dosing frequency. *Arch Intern Med* 2000;160:181-188.
- 1043. Hyers TM, Agnelli G, Hull RD, et al. Antithrombotic therapy for venous thromboembolic disease. *Chest* 2001;119:176S-193S.
- 1044. American Heart Association/American College of Cardiology Foundation guide to warfarin therapy. *Circulation* 2003; 107:1692-711.
- 1045. Boden WE. Is it time to reassess the optimal timing of coronary artery bypass graft surgery following acute myocardial infarction? *Am J Cardiol* 2002;90:35-8.
- 1046. Crossman AW, D'Agostino HJ, Geraci SA. Timing of coronary artery bypass graft surgery following acute myocardial infarction: a critical literature review. *Clin Cardiol* 2002;25:406-10.
- 1047. Braxton JH, Hammond GL, Letsou GV, et al. Optimal timing of coronary artery bypass graft surgery after acute myocardial infarction. *Circulation* 1995;92:II66-8.
- 1048. Lee DC, Oz MC, Weinberg AD, Lin SX, Ting W. Optimal timing of revascularization: transmural versus nontransmural acute myocardial infarction. *Ann Thorac Surg* 2001;71:1197-202; discussion.
- 1049. Creswell LL, Moulton MJ, Cox JL, Rosenbloom M. Revascularization after acute myocardial infarction. *Ann Thorac Surg* 1995;60:19-26.
- 1050. Acinapura AJ, Rose DM, Jacobowitz IJ, et al. Internal mammary artery bypass grafting: influence on recurrent angina and survival in 2,100 patients. *Ann Thorac Surg* 1989;48:186-91.
- 1051. Hirose H, Amano A, Yoshida S, et al. Surgical management of unstable patients in the evolving phase of acute myocardial infarction. *Ann Thorac Surg* 2000;69:425-8.
- 1052. Hirotani T, Kameda T, Shirota S, Nakao Y. Coronary artery bypass grafting within 30 days of an acute myocardial infarction. *Ann Thorac Cardiovasc Surg* 2001;7:28-34.
- 1053. Gersh BJ, Chesebro JH, Braunwald E, et al. Coronary artery bypass graft surgery after thrombolytic therapy in the Thrombolysis in Myocardial Infarction Trial, Phase II (TIMI II). *J Am Coll Cardiol* 1995;25:395-402.

- 1054. Holmes DR, Califf RM, Topol EJ. Lessons we have learned from
 - the GUSTO trial. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Arteries. J Am Coll Cardiol 1995;25:10S-17S.
- 1055. Kereiakes DJ, Topol EJ, George BS, et al, for the TAMI Study Group. Favorable early and long-term prognosis following coronary bypass surgery therapy for myocardial infarction: results of a multicenter trial. Am Heart J 1989;118:199-207.
- 1056. Skinner JR, Phillips SJ, Zeff RH, Kongtahworn C. Immediate coronary bypass following failed streptokinase infusion in evolving myocardial infarction. J Thorac Cardiovasc Surg 1984;87:567-70.
- 1057. Barner HB, Lea JW, Naunheim KS, Stoney WS. Emergency coronary bypass not associated with preoperative cardiogenic shock in failed angioplasty, after thrombolysis, and for acute myocardial infarction. Circulation 1989;79:I152-9.
- 1058. Efstratiadis T, Munsch C, Crossman D, Taylor K. Aprotinin used in emergency coronary operation after streptokinase treatment. Ann Thorac Surg 1991;52:1320-1.
- 1059. Breyer RH, Engelman RM, Rousou JA, Lemeshow S. Postinfarction angina: an expanding subset of patients undergoing coronary artery bypass. J Thorac Cardiovasc Surg 1985; 90:532-40.
- 1060. Hochberg MS, Parsonnet V, Gielchinsky I, Hussain SM, Fisch DA, Norman JC. Timing of coronary revascularization after acute myocardial infarction: early and late results in patients revascularized within seven weeks. J Thorac Cardiovasc Surg 1984;88:914-21.
- 1061. Fremes SE, Goldman BS, Weisel RD, et al. Recent preoperative myocardial infarction increases the risk of surgery for unstable angina. J Card Surg 1991;6:2-12.
- 1062. Omoigui NA, Miller DP, Brown KJ, et al. Outmigration for coronary bypass surgery in an era of public dissemination of clinical outcomes. Circulation 1996;93:27-33.
- 1063. Burack JH, Impellizzeri P, Homel P, Cunningham JN. Public reporting of surgical mortality: a survey of New York State cardiothoracic surgeons. Ann Thorac Surg 1999;68:1195-200; discussion.
- 1064. Nallamothu BK, Saint S, Ramsey SD, Hofer TP, Vijan S, Eagle KA. The role of hospital volume in coronary artery bypass grafting: is more always better? J Am Coll Cardiol 2001;38:1923-30.
- 1065. Applebaum R, House R, Rademaker A, et al. Coronary artery bypass grafting within thirty days of acute myocardial infarction: early and late results in 406 patients. J Thorac Cardiovasc Surg 1991;102:745-52.
- 1066. Mangano DT, for the Multicenter Study of Perioperative Ischemia Research Group. Aspirin and mortality from coronary bypass surgery. N Engl J Med 2002;347:1309-17.
- 1067. Topol EJ. Aspirin with bypass surgery: from taboo to new standard of care. N Engl J Med 2002;347:1359-60.
- 1068. Hongo RH, Ley J, Dick SE, Yee RR. The effect of clopidogrel in combination with aspirin when given before coronary artery bypass grafting. J Am Coll Cardiol 2002;40:231-7.
- 1069. Boehrer JD, Kereiakes DJ, Navetta FI, Califf RM, Topol EJ, for the Evaluation Prevention of Ischemic Complications (EPIC) Investigators. Effects of profound platelet inhibition with c7E3 before coronary angioplasty on complications of coronary bypass surgery. Am J Cardiol 1994;74:1166-70.
- 1070. Singh M, Nuttall GA, Ballman KV, et al. Effect of abciximab on the outcome of emergency coronary artery bypass grafting after failed percutaneous coronary intervention. Mayo Clin Proc 2001;76:784-8.

- 1071. Lincoff AM, LeNarz LA, Despotis GJ, et al. Abciximab and bleeding during coronary surgery: results from the EPILOG and EPISTENT trials. Improve Long-term Outcome with abciximab GP IIb/IIIa blockade. Evaluation of Platelet IIb/IIIa Inhibition in STENTing. Ann Thorac Surg 2000;70:516-26.
- 1072. Dyke C, Bhatia D. Inhibitors of the platelet receptor glycoprotein IIb-IIIa and complications during percutaneous coronary revascularization: management strategies for the cardiac surgeon. J Cardiovasc Surg (Torino) 1999;40:505-16.
- 1073. Clopidigrel. Package insert. Physican's Desk Reference. Medical Economics Co. Montvale, NJ:2756.
- 1074. Crawford MH, Bernstein SJ, Deedwania PC, et al. ACC/AHA guidelines for ambulatory electrocardiography: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the Guidelines for Ambulatory Electrocardiography). J Am Coll Cardiol 1999;34:912-48.
- 1075. Gibbons RJ, Balady GJ, Bricker JT, et al. ACC/AHA 2002 guideline update for exercise testing: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on Exercise Testing). 2002. Available at: http://www.acc.org/clinical/guidelines/exercise/ exercise_clean.pdf. Accessed January 4, 2004.
- 1076. Théroux P, Waters DD, Halphen C, Debaisieux JC, Mizgala HF. Prognostic value of exercise testing soon after myocardial infarction. N Engl J Med 1979;301:341-5.
- 1077. DeBusk RF, Kraemer HC, Nash E, Berger WE, Lew H. Stepwise risk stratification soon after acute myocardial infarction. Am J Cardiol 1983;52:1161-6.
- 1078. Krone RJ, Gillespie JA, Weld FM, Miller JP, Moss AJ. Low-level exercise testing after myocardial infarction: usefulness in enhancing clinical risk stratification. Circulation 1985;71:80-9.
- 1079. Valentine PA, Frew JL, Mashford ML, Sloman JG. Lidocaine in the prevention of sudden death in the pre-hospital phase of acute infarction: a double-blind study. N Engl J Med 1974;291:1327-31.
- 1080. Ross J, Gilpin EA, Madsen EB, et al. A decision scheme for coronary angiography after acute myocardial infarction. Circulation 1989;79:292-303.
- 1081. Ritchie JL, Cerqueira M, Maynard C, Davis K, Kennedy JW. Ventricular function and infarct size: the Western Washington Intravenous Streptokinase in Myocardial Infarction Trial. J Am Coll Cardiol 1988;11:689-97.
- 1082. TIMI Study Group. The Thrombolysis in Myocardial Infarction (TIMI) trial: phase I findings. N Engl J Med 1985;312:932-6.
- 1083. Hamm LF, Crow RS, Stull GA, Hannan P. Safety and characteristics of exercise testing early after acute myocardial infarction. Am J Cardiol 1989;63:1193-7.
- 1084. Juneau M, Colles P, Théroux P, et al. Symptom-limited versus low level exercise testing before hospital discharge after myocardial infarction. J Am Coll Cardiol 1992;20:927-33.
- 1085. Jain A, Myers GH, Sapin PM, O'Rourke RA. Comparison of symptom-limited and low level exercise tolerance tests early after myocardial infarction. J Am Coll Cardiol 1993;22:1816-20.
- 1086. Mark DB, Shaw L, Harrell FE, et al. Prognostic value of a treadmill exercise score in outpatients with suspected coronary artery disease. N Engl J Med 1991;325:849-53.
- 1087. Piccalò G, Pirelli S, Massa D, Cipriani M, Sarullo FM, De Vita C. Value of negative predischarge exercise testing in identifying patients at low risk after acute myocardial infarction treated by systemic thrombolysis. Am J Cardiol 1992;70:31-3.

Antman *et al.* 2004 ACC/AHA Practice Guidelines

e284

- 1088. Heger JJ, Weyman AE, Wann LS, Rogers EW, Dillon JC, Feigenbaum H. Cross-sectional echocardiographic analysis of the extent of left ventricular asynergy in acute myocardial infarction. *Circulation* 1980;61:1113-8.
- 1089. Gibson RS, Bishop HL, Stamm RB, Crampton RS, Beller GA, Martin RP. Value of early two dimensional echocardiography in patients with acute myocardial infarction. *Am J Cardiol* 1982;49:1110-9.
- 1090. Nishimura RA, Tajik AJ, Shub C, Miller FA, Ilstrup DM, Harrison CE. Role of two-dimensional echocardiography in the prediction of in-hospital complications after acute myocardial infarction. *J Am Coll Cardiol* 1984;4:1080-7.
- 1091. Horowitz RS, Morganroth J. Immediate detection of early highrisk patients with acute myocardial infarction using two-dimensional echocardiographic evaluation of left ventricular regional wall motion abnormalities. *Am Heart J* 1982;103:814-22.
- 1092. Minardi G, Di Segni M, Manzara CC, et al. Diagnostic and prognostic value of dipyridamole and dobutamine stress echocardiography in patients with Q-wave acute myocardial infarction. *Am J Cardiol* 1997;80:847-51.
- 1093. Smart S, Wynsen J, Sagar K. Dobutamine-atropine stress echocardiography for reversible dysfunction during the first week after acute myocardial infarction: limitations and determinants of accuracy. *J Am Coll Cardiol* 1997;30:1669-78.
- 1094. Bolognese L, Antoniucci D, Rovai D, et al. Myocardial contrast echocardiography versus dobutamine echocardiography for predicting functional recovery after acute myocardial infarction treated with primary coronary angioplasty. *J Am Coll Cardiol* 1996;28:1677-83.
- 1095. Poli A, Previtali M, Lanzarini L, et al. Comparison of dobutamine stress echocardiography with dipyridamole stress echocardiography for detection of viable myocardium after myocardial infarction treated with thrombolysis. *Heart* 1996;75:240-6.
- 1096. Salustri A, Elhendy A, Garyfallydis P, et al. Prediction of improvement of ventricular function after first acute myocardial infarction using low-dose dobutamine stress echocardiography. *Am J Cardiol* 1994;74:853-6.
- 1097. Watada H, Ito H, Oh H, et al. Dobutamine stress echocardiography predicts reversible dysfunction and quantitates the extent of irreversibly damaged myocardium after reperfusion of anterior myocardial infarction. J Am Coll Cardiol 1994;24:624-30.
- 1098. Previtali M, Poli A, Lanzarini L, Fetiveau R, Mussini A, Ferrario M. Dobutamine stress echocardiography for assessment of myocardial viability and ischemia in acute myocardial infarction treated with thrombolysis. *Am J Cardiol* 1993;72:124G-130G.
- 1099. Smart SC, Sawada S, Ryan T, et al. Low-dose dobutamine echocardiography detects reversible dysfunction after thrombolytic therapy of acute myocardial infarction. *Circulation* 1993;88:405-15.
- 1100. Piérard LA, De Landsheere CM, Berthe C, Rigo P, Kulbertus HE. Identification of viable myocardium by echocardiography during dobutamine infusion in patients with myocardial infarction after thrombolytic therapy: comparison with positron emission tomography. *J Am Coll Cardiol* 1990;15:1021-31.
- 1101. Barilla F, Gheorghiade M, Alam M, Khaja F, Goldstein S. Lowdose dobutamine in patients with acute myocardial infarction identifies viable but not contractile myocardium and predicts the magnitude of improvement in wall motion abnormalities in response to coronary revascularization. *Am Heart J* 1991; 122:1522-31.
- 1102. Lundgren C, Bourdillon PD, Dillon JC, Feigenbaum H.

Comparison of contrast angiography and two-dimensional echocardiography for the evaluation of left ventricular regional wall motion abnormalities after acute myocardial infarction. *Am J Cardiol* 1990;65:1071-7.

- 1103. Applegate RJ, Dell'Italia LJ, Crawford MH. Usefulness of twodimensional echocardiography during low-level exercise testing early after uncomplicated acute myocardial infarction. Am J Cardiol 1987;60:10-4.
- 1104. Picano E, Pingitore A, Sicari R, et al, for the Echo Persantine International Cooperative (EPIC) Study Group. Stress echocardiographic results predict risk of reinfarction early after uncomplicated acute myocardial infarction: large-scale multicenter study. *J Am Coll Cardiol* 1995;26:908-13.
- 1105. Bolognese L, Rossi L, Sarasso G, et al. Silent versus symptomatic dipyridamole-induced ischemia after myocardial infarction: clinical and prognostic significance. *J Am Coll Cardiol* 1992;19:953-9.
- 1106. Quintana M, Lindvall K, Rydén L, Brolund F. Prognostic value of predischarge exercise stress echocardiography after acute myocardial infarction. *Am J Cardiol* 1995;76:1115-21.
- 1107. Orlandini AD, Tuero EI, Diaz R, Vilamajó OA, Paolasso EA. Acute cardiac rupture during dobutamine-atropine echocardiography stress test. J Am Soc Echocardiogr 2000;13:152-3.
- 1108. Carlos ME, Smart SC, Wynsen JC, Sagar KB. Dobutamine stress echocardiography for risk stratification after myocardial infarction. *Circulation* 1997;95:1402-10.
- 1109. Greco CA, Salustri A, Seccareccia F, et al. Prognostic value of dobutamine echocardiography early after uncomplicated acute myocardial infarction: a comparison with exercise electrocar-diography. *J Am Coll Cardiol* 1997;29:261-7.
- 1110. Sclavo MG, Noussan P, Pallisco O, Presbitero P. Usefulness of dipyridamole-echocardiographic test to identify jeopardized myocardium after thrombolysis: limited clinical predictivity of dipyridamole-echocardiographic test in convalescing acute myocardial infarction: correlation with coronary angiography. *Eur Heart J* 1992;13:1348-55.
- 1111. Picano E, Landi P, Bolognese L, et al, for the EPIC Study Group. Prognostic value of dipyridamole echocardiography early after uncomplicated myocardial infarction: a large-scale, multicenter trial. *Am J Med* 1993;95:608-18.
- 1112. Sicari R, Picano E, Landi P, et al. Prognostic value of dobutamine-atropine stress echocardiography early after acute myocardial infarction. Echo Dobutamine International Cooperative (EDIC) Study. J Am Coll Cardiol 1997;29:254-60.
- 1113. Gibson RS, Watson DD, Craddock GB, et al. Prediction of cardiac events after uncomplicated myocardial infarction: a prospective study comparing predischarge exercise thallium-201 scintigraphy and coronary angiography. *Circulation* 1983;68:321-36.
- 1114. Hung J, Goris ML, Nash E, et al. Comparative value of maximal treadmill testing, exercise thallium myocardial perfusion scintigraphy and exercise radionuclide ventriculography for distinguishing high- and low-risk patients soon after acute myocardial infarction. *Am J Cardiol* 1984;53:1221-7.
- 1115. Abraham RD, Freedman SB, Dunn RF, et al. Prediction of multivessel coronary artery disease and prognosis early after acute myocardial infarction by exercise electrocardiography and thallium-201 myocardial perfusion scanning. *Am J Cardiol* 1986;58:423-7.
- 1116. Wilson WW, Gibson RS, Nygaard TW, et al. Acute myocardial infarction associated with single vessel coronary artery disease: an analysis of clinical outcome and the prognostic importance of

vessel patency and residual ischemic myocardium. J Am Coll Cardiol 1988;11:223-34.

- 1117. Leppo JA, O'Brien J, Rothendler JA, Getchell JD, Lee VW. Dipyridamole-thallium-201 scintigraphy in the prediction of future cardiac events after acute myocardial infarction. N Engl J Med 1984;310:1014-8.
- 1118. Pirelli S, Inglese E, Suppa M, Corrada E, Campolo L. Dipyridamole-thallium 201 scintigraphy in the early postinfarction period: safety and accuracy in predicting the extent of coronary disease and future recurrence of angina in patients suffering from their first myocardial infarction. *Eur Heart J* 1988;9:1324-31.
- 1119. Younis LT, Byers S, Shaw L, Barth G, Goodgold H, Chaitman BR. Prognostic value of intravenous dipyridamole thallium scintigraphy after an acute myocardial ischemic event. *Am J Cardiol* 1989;64:161-6.
- 1120. Figueredo V, Cheitlin MD. Risk stratification. In: Julian DG, Braunwald E, eds. Management of acute myocardial infarction. London, England: WB Saunders Co Ltd; 1994:361-91.
- 1121. Tilkemeier PL, Guiney TE, LaRaia PJ, Boucher CA. Prognostic value of predischarge low-level exercise thallium testing after thrombolytic treatment of acute myocardial infarction. *Am J Cardiol* 1990;66:1203-7.
- 1122. Hendel RC, Gore JM, Alpert JS, Leppo JA. Prognosis following interventional therapy for acute myocardial infarction: utility of dipyridamole thallium scintigraphy. *Cardiology* 1991;79:73-80.
- 1123. Miller TD, Gersh BJ, Christian TF, Bailey KR, Gibbons RJ. Limited prognostic value of thallium-201 exercise treadmill testing early after myocardial infarction in patients treated with thrombolysis. *Am Heart J* 1995;130:259-66.
- 1124. Mahmarian JJ, Mahmarian AC, Marks GF, Pratt CM, Verani MS. Role of adenosine thallium-201 tomography for defining longterm risk in patients after acute myocardial infarction. *J Am Coll Cardiol* 1995;25:1333-40.
- 1125. Cerqueira MD, Maynard C, Ritchie JL, Davis KB, Kennedy JW. Long-term survival in 618 patients from the Western Washington Streptokinase in Myocardial Infarction trials. *J Am Coll Cardiol* 1992;20:1452-9.
- 1126. Miller TD, Christian TF, Hopfenspirger MR, Hodge DO, Gersh BJ, Gibbons RJ. Infarct size after acute myocardial infarction measured by quantitative tomographic 99mTc sestamibi imaging predicts subsequent mortality. *Circulation* 1995;92:334-41.
- 1127. The relationships of left ventricular ejection fraction, end-systolic volume index and infarct size to six-month mortality after hospital discharge following myocardial infarction treated by thrombolysis. *J Am Coll Cardiol* 2002;39:30-6.
- 1128. Brown KA, Heller GV, Landin RS, et al. Early dipyridamole (99m)Tc-sestamibi single photon emission computed tomographic imaging 2 to 4 days after acute myocardial infarction predicts in-hospital and postdischarge cardiac events: comparison with submaximal exercise imaging. *Circulation* 1999; 100:2060-6.
- 1129. Risk stratification and survival after myocardial infarction. *N* Engl J Med 1983;309:331-6.
- 1130. Zaret BL, Wackers TH, Terrin M, et al. Does left ventricular ejection fraction following thrombolytic therapy have the same prognostic impact described in the prethrombolytic era? Results of the TIMI II Trial [abstr]. *J Am Coll Cardiol* 1991;17:214A.
- 1131. Roig E, Magriñá J, Garcia A, et al. Prognostic value of exercise radionuclide angiography in low risk acute myocardial infarction survivors. *Eur Heart J* 1993;14:213-8.
- 1132. White HD, Norris RM, Brown MA, Brandt PW, Whitlock RM,

Wild CJ. Left ventricular end-systolic volume as the major determinant of survival after recovery from myocardial infarction. *Circulation* 1987;76:44-51.

- 1133. Rashid H, Exner DV, Mirsky I, Cooper HA, Waclawiw MA, Domanski MJ. Comparison of echocardiography and radionuclide angiography as predictors of mortality in patients with left ventricular dysfunction: studies of left ventricular dysfunction. *Am J Cardiol* 1999;84:299-303.
- 1134. Rahimtoola SH. The hibernating myocardium. Am Heart J 1989;117:211-21.
- 1135. Braunwald E, Kloner RA. The stunned myocardium: prolonged, postischemic ventricular dysfunction. *Circulation* 1982;66: 1146-9.
- 1136. Rahimtoola SH. A perspective on the three large multicenter randomized clinical trials of coronary bypass surgery for chronic stable angina. *Circulation* 1985;72:V123-35.
- 1137. Di Carli MF. Assessment of myocardial viability after myocardial infarction. J Nucl Cardiol 9:229-35.
- 1138. Allman KC, Shaw LJ, Hachamovitch R, Udelson JE. Myocardial viability testing and impact of revascularization on prognosis in patients with coronary artery disease and left ventricular dys-function: a meta-analysis. *J Am Coll Cardiol* 2002;39:1151-8.
- 1139. Afridi I, Grayburn PA, Panza JA, Oh JK, Zoghbi WA, Marwick TH. Myocardial viability during dobutamine echocardiography predicts survival in patients with coronary artery disease and severe left ventricular systolic dysfunction. *J Am Coll Cardiol* 1998;32:921-6.
- 1140. Rohatgi R, Epstein S, Henriquez J, et al. Utility of positron emission tomography in predicting cardiac events and survival in patients with coronary artery disease and severe left ventricular dysfunction. *Am J Cardiol* 2001;87:1096-9, A6.
- 1141. Bax JJ, Wijns W, Cornel JH, Visser FC, Boersma E, Fioretti PM. Accuracy of currently available techniques for prediction of functional recovery after revascularization in patients with left ventricular dysfunction due to chronic coronary artery disease: comparison of pooled data. J Am Coll Cardiol 1997;30:1451-60.
- 1142. Beller GA. Noninvasive assessment of myocardial viability. *N* Engl J Med 2000;343:1488-90.
- 1143. Beek AM, Kühl HP, Bondarenko O, et al. Delayed contrastenhanced magnetic resonance imaging for the prediction of regional functional improvement after acute myocardial infarction. *J Am Coll Cardiol* 2003;42:895-901.
- 1144. Gerber BL, Garot J, Bluemke DA, Wu KC, Lima JA. Accuracy of contrast-enhanced magnetic resonance imaging in predicting improvement of regional myocardial function in patients after acute myocardial infarction. *Circulation* 2002;106:1083-9.
- 1145. Bates DW, Miller E, Bernstein SJ, Hauptman PJ, Leape LL. Coronary angiography and angioplasty after acute myocardial infarction. *Ann Intern Med* 1997;126:539-50.
- 1146. Scott IA, Harden H, Coory M. What are appropriate rates of invasive procedures following acute myocardial infarction: a systematic review. *Med J Aust* 2001;174:130-6.
- 1147. Llevadot J, Giugliano RP, Antman EM, et al, for the InTIME (Intravenous nPA for Treatment of Infarcting Myocardium Early) II Investigators. Availability of on-site catheterization and clinical outcomes in patients receiving fibrinolysis for STelevation myocardial infarction. *Eur Heart J* 2001;22:2104-15.
- 1148. Rogers WJ, Baim DS, Gore JM, et al. Comparison of immediate invasive, delayed invasive, and conservative strategies after tissue-type plasminogen activator: results of the Thrombolysis in Myocardial Infarction (TIMI) Phase II-A trial. *Circulation* 1990;81:1457-76.

- 1149. O'Neill WW. "Watchful waiting" after thrombolysis: it's time for a re-evaluation. *J Am Coll Cardiol* 2003;42:17-9.
- 1149a. Antman EM, Van de Werf F. Pharmacoinvasive therapy: the future of treatment for ST-elevation myocardial infarction. *Circulation* 2004;109:2480-6.
- 1150. Gottlieb SO, Gottlieb SH, Achuff SC, Baumgardner R, Mellits ED, Weisfeldt ML, Gerstenblith G. Silent ischemia on Holter monitoring predicts mortality in high-risk postinfarction patients. *JAMA* 1988;259:1030-5.
- 1151. Tzivoni D, Gavish A, Zin D, et al. Prognostic significance of ischemic episodes in patients with previous myocardial infarction. Am J Cardiol 1988;62:661-4.
- 1152. Bonaduce D, Petretta M, Lanzillo T, et al. Prevalence and prognostic significance of silent myocardial ischaemia detected by exercise test and continuous ECG monitoring after acute myocardial infarction. *Eur Heart J* 1991;12:186-93.
- 1153. Langer A, Minkowitz J, Dorian P, et al, for the Tissue Plasminogen Activator: Toronto (TPAT) Study Group. Pathophysiology and prognostic significance of Holter-detected ST segment depression after myocardial infarction. J Am Coll Cardiol 1992;20:1313-7.
- 1154. Petretta M, Bonaduce D, Bianchi V, et al. Characterization and prognostic significance of silent myocardial ischemia on predischarge electrocardiographic monitoring in unselected patients with myocardial infarction. *Am J Cardiol* 1992;69:579-83.
- 1155. Jereczek M, Andresen D, Schröder J, et al. Prognostic value of ischemia during Holter monitoring and exercise testing after acute myocardial infarction. *Am J Cardiol* 1993;72:8-13.
- 1156. Currie P, Ashby D, Saltissi S. Prognostic significance of transient myocardial ischemia on ambulatory monitoring after acute myocardial infarction. *Am J Cardiol* 1993;71:773-7.
- 1157. Gill JB, Cairns JA, Roberts RS, et al. Prognostic importance of myocardial ischemia detected by ambulatory monitoring early after acute myocardial infarction. N Engl J Med 1996;334:65-70.
- 1158. Deedwania PC. Asymptomatic ischemia during predischarge Holter monitoring predicts poor prognosis in the postinfarction period. *Am J Cardiol* 1993;71:859-61.
- 1159. Kuchar DL, Thorburn CW, Sammel NL. Prediction of serious arrhythmic events after myocardial infarction: signal-averaged electrocardiogram, Holter monitoring and radionuclide ventriculography. J Am Coll Cardiol 1987;9:531-8.
- 1160. Gomes JA, Winters SL, Martinson M, Machac J, Stewart D, Targonski A. The prognostic significance of quantitative signalaveraged variables relative to clinical variables, site of myocardial infarction, ejection fraction and ventricular premature beats: a prospective study. J Am Coll Cardiol 1989;13:377-84.
- 1161. el-Sherif N, Denes P, Katz R, et al, for the Cardiac Arrhythmia Suppression Trial/Signal-Averaged Electrocardiogram (CAST/ SAECG) Substudy Investigators. Definition of the best prediction criteria of the time domain signal-averaged electrocardiogram for serious arrhythmic events in the postinfarction period. *J Am Coll Cardiol* 1995;25:908-14.
- 1162. Vatterott PJ, Hammill SC, Bailey KR, Wiltgen CM, Gersh BJ. Late potentials on signal-averaged electrocardiograms and patency of the infarct-related artery in survivors of acute myocardial infarction. J Am Coll Cardiol 1991;17:330-7.
- 1163. McClements BM, Adgey AA. Value of signal-averaged electrocardiography, radionuclide ventriculography, Holter monitoring and clinical variables for prediction of arrhythmic events in survivors of acute myocardial infarction in the thrombolytic era. J Am Coll Cardiol 1993;21:1419-27.

- 1164. Hohnloser SH, Franck P, Klingenheben T, Zabel M, Just H. Open infarct artery, late potentials, and other prognostic factors in patients after acute myocardial infarction in the thrombolytic era: a prospective trial. *Circulation* 1994;90:1747-56.
- 1165. Farrell TG, Bashir Y, Cripps T, et al. Risk stratification for arrhythmic events in postinfarction patients based on heart rate variability, ambulatory electrocardiographic variables and the signal-averaged electrocardiogram. *J Am Coll Cardiol* 1991; 18:687-97.
- 1166. Bigger JT, Fleiss JL, Rolnitzky LM, Steinman RC. The ability of several short-term measures of RR variability to predict mortality after myocardial infarction. *Circulation* 1993;88:927-34.
- 1167. Kleiger RE, Miller JP, Bigger JT, Moss AJ. Decreased heart rate variability and its association with increased mortality after acute myocardial infarction. *Am J Cardiol* 1987;59:256-62.
- 1168. La Rovere MT, Bigger JT, Marcus FI, Mortara A, Schwartz PJ, for the ATRAMI (Autonomic Tone and Reflexes After Myocardial Infarction) Investigators. Baroreflex sensitivity and heart-rate variability in prediction of total cardiac mortality after myocardial infarction. *Lancet* 1998;351:478-84.
- 1169. Schwartz PJ, La Rovere MT, Vanoli E. Autonomic nervous system and sudden cardiac death: experimental basis and clinical observations for post-myocardial infarction risk stratification. *Circulation* 1992;85:I77-91.
- 1170. Schwartz PJ, Vanoli E, Stramba-Badiale M, et al. Autonomic mechanisms and sudden death: new insights from analysis of baroreceptor reflexes in conscious dogs with and without a myocardial infarction. *Circulation* 1988;78:969-79.
- 1171. La Rovere MT, Specchia G, Mortara A, Schwartz PJ. Baroreflex sensitivity, clinical correlates, and cardiovascular mortality among patients with a first myocardial infarction: a prospective study. *Circulation* 1988;78:816-24.
- 1172. Farrell TG, Paul V, Cripps TR, et al. Baroreflex sensitivity and electrophysiological correlates in patients after acute myocardial infarction. *Circulation* 1991;83:945-52.
- 1173. Gold MR, Bloomfield DM, Anderson KP, et al. A comparison of T-wave alternans, signal averaged electrocardiography and programmed ventricular stimulation for arrhythmia risk stratification. J Am Coll Cardiol 2000;36:2247-53.
- 1174. Ikeda T, Sakata T, Takami M, et al. Combined assessment of Twave alternans and late potentials used to predict arrhythmic events after myocardial infarction: a prospective study. *J Am Coll Cardiol* 2000;35:722-30.
- 1175. Gilman JK, Jalal S, Naccarelli GV. Predicting and preventing sudden death from cardiac causes. *Circulation* 1994;90:1083-92.
- 1176. Dalal H, Evans PH, Campbell JL. Recent developments in secondary prevention and cardiac rehabilitation after acute myocardial infarction. *BMJ* 2004;328:693-7.
- 1177. Qureshi AI, Suri MF, Guterman LR, Hopkins LN. Ineffective secondary prevention in survivors of cardiovascular events in the US population: report from the Third National Health and Nutrition Examination Survey. *Arch Intern Med* 2001; 161:1621-8.
- 1178. Berger AK, Duval S, Krumholz HM. Aspirin, beta-blocker, and angiotensin-converting enzyme inhibitor therapy in patients with end-stage renal disease and an acute myocardial infarction. *J Am Coll Cardiol* 2003;42:201-8.
- 1179. Conti CR. Management of patients with acute myocardial infarction and end-stage renal disease. J Am Coll Cardiol 2003; 42:209-10.
- 1180. Shlipak MG, Heidenreich PA, Noguchi H, Chertow GM,

Browner WS, McClellan MB. Association of renal insufficiency with treatment and outcomes after myocardial infarction in elderly patients. *Ann Intern Med* 2002;137:555-62.

- 1181. LaBresh KA, Ellrodt AG, Gliklich R, Liljestrand J, Peto R. Get with the guidelines for cardiovascular secondary prevention: pilot results. *Arch Intern Med* 2004;164:203-9.
- 1182. Lee KW, Lip GY. Effects of lifestyle on hemostasis, fibrinolysis, and platelet reactivity: a systematic review. *Arch Intern Med* 2003;163:2368-92.
- 1183. Fletcher GF, Balady GJ, Amsterdam EA, et al. Exercise standards for testing and training: a statement for healthcare professionals from the American Heart Association. *Circulation* 2001;104:1694-740.
- 1184. Wenger NK, Froelicher ES, Smith LK, et al. AHCPR Supported Guidelines. Clinical Practice Guideline Number 17: Cardiac Rehabilitation. US Dept of Health and Human Services. Public Health Service. Agency for Health Care Policy and Research. National Heart, Lung and Blood Institute, October 1995. AHCPR publication No. 96-0672.
- 1185. Fonarow GC, Gawlinski A, Moughrabi S, Tillisch JH. Improved treatment of coronary heart disease by implementation of a Cardiac Hospitalization Atherosclerosis Management Program (CHAMP). *Am J Cardiol* 2001;87:819-22.
- 1186. Mehta RH, Montoye CK, Faul J, et al, for the American College of Cardiology Guidelines Applied in Practice Steering Committee. Enhancing quality of care for acute myocardial infarction: shifting the focus of improvement from key indicators to process of care and tool use. *J Am Coll Cardiol* 2004;43:2166-73
- 1187. Bowles KH, Foust JB, Naylor MD. Hospital discharge referral decision making: a multidisciplinary perspective. *Appl Nurs Res* 2003;16:134-43.
- 1188. Plach S, Wierenga ME, Heidrich SM. Effect of a postdischarge education class on coronary artery disease knowledge and self-reported health-promoting behaviors. *Heart Lung* 1996;25:367-72.
- 1189. O'Neill C, Normand C, Cupples M, McKnight A. Cost effectiveness of personal health education in primary care for people with angina in the greater Belfast area of Northern Ireland. J Epidemiol Community Health 1996;50:538-40.
- 1190. Larson CO, Nelson EC, Gustafson D, Batalden PB. The relationship between meeting patients' information needs and their satisfaction with hospital care and general health status outcomes. *Int J Qual Health Care* 1996;8:447-56.
- 1191. Theis SL, Johnson JH. Strategies for teaching patients: a metaanalysis. *Clin Nurse Spec* 1995;9:100-5, 120.
- 1192. Hill J. A practical guide to patient education and information giving. *Baillieres Clin Rheumatol* 1997;11:109-27.
- 1193. Kingsbury K. Taking AIM: how to teach primary and secondary prevention effectively. *Can J Cardiol* 1998;14 Suppl A:22A-26A.
- 1194. Winslow E, Bohannon N, Brunton SA, Mayhew HE. Lifestyle modification: weight control, exercise, and smoking cessation. *Am J Med* 1996;101:4A25S-31S; discussion.
- 1195. The multiple risk factor intervention trial (MRFIT): a national study of primary prevention of coronary heart disease. *JAMA* 1976;235:825-7.
- 1196. Moore SM. Effects of interventions to promote recovery in coronary artery bypass surgical patients. J Cardiovasc Nurs 1997;12:59-70.
- 1197. Oldridge NB, Guyatt GH, Fischer ME, Rimm AA. Cardiac rehabilitation after myocardial infarction: combined experience of

randomized clinical trials. JAMA 1988;260:945-50.

- 1198. Wang WW. The educational needs of myocardial infarction patients. *Prog Cardiovasc Nurs* 1994;9:28-36.
- 1199. Chan V. Content areas for cardiac teaching: patients' perceptions of the importance of teaching content after myocardial infarction. *J Adv Nurs* 1990;15:1139-45.
- 1200. Cogswell B, Eggert MS. People want doctors to give more preventive care: a qualitative study of health care consumers. *Arch Fam Med* 1993;2:611-9.
- 1201. Mittelmark MB, Leupker RV, Grimm R, Kottke TE, Blackburn H. The role of physicians in a community-wide program for prevention of cardiovascular disease: the Minnesota Heart Health Program. *Public Health Rep* 103:360-5.
- 1202. Rhodes KS, Bookstein LC, Aaronson LS, Mercer NM, Orringer CE. Intensive nutrition counseling enhances outcomes of National Cholesterol Education Program dietary therapy. J Am Diet Assoc 1996;96:1003-10; quiz 1011-2.
- 1203. Lindsay C, Jennrich JA, Biemolt M. Programmed instruction booklet for cardiac rehabilitation teaching. *Heart Lung* 1991;20:648-53.
- 1204. Lewis D. Computer-based approaches to patient education: a review of the literature. *J Am Med Inform Assoc* 1999;6:272-82.
- 1205. Ganguli G. Consumers devise drug cost-cutting measures: medical and legal issues to consider. *Health Care Manag* (*Frederick*) 2003;22:275-81.
- 1206. Myerburg RJ, Kessler KM, Castellanos A. Sudden cardiac death: epidemiology, transient risk, and intervention assessment. *Ann Intern Med* 1993;119:1187-97.
- 1207. Randomised trial of cholesterol lowering in 4444 patients with coronary heart disease: the Scandinavian Simvastatin Survival Study (4S) *Lancet* 1994;344:1383-9.
- 1208. Sacks FM, Pfeffer MA, Moye LA, et al, for the Cholesterol and Recurrent Events Trial investigators. The effect of pravastatin on coronary events after myocardial infarction in patients with average cholesterol levels. *N Engl J Med* 1996;335:1001-9.
- 1209. The Long-Term Intervention with Pravastatin in Ischaemic Disease (LIPID) Study Group. Prevention of cardiovascular events and death with pravastatin in patients with coronary heart disease and a broad range of initial cholesterol levels. *N Engl J Med* 1998;339:1349-57.
- 1210. Riegger G, Abletshauser C, Ludwig M, et al. The effect of fluvastatin on cardiac events in patients with symptomatic coronary artery disease during one year of treatment. *Atherosclerosis* 1999;144:263-70.
- 1211. The Post Coronary Artery Bypass Graft Trial Investigators. The effect of aggressive lowering of low-density lipoprotein cholesterol levels and low-dose anticoagulation on obstructive changes in saphenous-vein coronary-artery bypass grafts. *N Engl J Med* 1997;336:153-62.
- 1212. Knatterud GL, Rosenberg Y, Campeau L, et al, for the Post CABG Investigators. Long-term effects on clinical outcomes of aggressive lowering of low-density lipoprotein cholesterol levels and low-dose anticoagulation in the post coronary artery bypass graft trial. *Circulation* 2000;102:157-65.
- 1213. MRC/BHF Heart Protection Study of cholesterol lowering with simvastatin in 20,536 high-risk individuals: a randomised placebo-controlled trial. *Lancet* 2002;360:7-22.
- 1214. Pekkanen J, Linn S, Heiss G, et al. Ten-year mortality from cardiovascular disease in relation to cholesterol level among men with and without preexisting cardiovascular disease. N Engl J Med 1990;322:1700-7.
- 1215. Rubins HB, Robins SJ, Collins D, et al, for the Veterans Affairs

High-Density Lipoprotein Cholesterol Intervention Trial Study Group. Gemfibrozil for the secondary prevention of coronary heart disease in men with low levels of high-density lipoprotein cholesterol. *N Engl J Med* 1999;341:410-8.

- 1216. Arntz HR, Agrawal R, Wunderlich W, et al. Beneficial effects of pravastatin (+/-cholestyramine/niacin) initiated immediately after a coronary event (the randomized Lipid-Coronary Artery Disease [L-CAD] Study). Am J Cardiol 2000;86:1293-8.
- 1217. Schwartz GG, Olsson AG, Ezekowitz MD, et al, for the Myocardial Ischemia Reduction with Aggressive Cholesterol Lowering (MIRACL) Study Investigators. Effects of atorvastatin on early recurrent ischemic events in acute coronary syndromes: the MIRACL study: a randomized controlled trial. *JAMA* 2001;285:1711-8.
- 1218. Stenestrand U, Wallentin L for the Swedish Register of Cardiac Intensive Care (RIKS-HIA). Early statin treatment following acute myocardial infarction and 1-year survival. *JAMA* 2001;285:430-6.
- 1219. Cannon CP, Braunwald E, McCabe CH, et al. Comparison of intensive and moderate lipid lowering with statins after acute coronary syndromes. *N Engl J Med* 2004;350:1562-4.
- 1220. Aronow HD, Novaro GM, Lauer MS, et al. In-hospital initiation of lipid-lowering therapy after coronary intervention as a predictor of long-term utilization: a propensity analysis. *Arch Intern Med* 2003;163:2576-82.
- 1221. Flegal KM, Carroll MD, Ogden CL, Johnson CL. Prevalence and trends in obesity among US adults, 1999-2000. JAMA 2002; 288:1723-7.
- 1222. Nutrition and Your Health: Dietary Guidelines for Americans. Fourth Edition. US Department of Agriculture. US Department of Health and Human Services, Washington, DC, 1995.
- 1223. Winniford MD, Jansen DE, Reynolds GA, Apprill P, Black WH, Hillis LD. Cigarette smoking-induced coronary vasoconstriction in atherosclerotic coronary artery disease and prevention by calcium antagonists and nitroglycerin. *Am J Cardiol* 1987; 59:203-7.
- 1224. Deanfield J, Wright C, Krikler S, Ribeiro P, Fox K. Cigarette smoking and the treatment of angina with propranolol, atenolol, and nifedipine. *N Engl J Med* 1984;310:951-4.
- 1225. Barry J, Mead K, Nabel EG, et al. Effect of smoking on the activity of ischemic heart disease. *JAMA* 1989;261:398-402.
- 1226. Burling TA, Singleton EG, Bigelow GE, Baile WF, Gottlieb SH. Smoking following myocardial infarction: a critical review of the literature. *Health Psychol* 1984;3:83-96.
- 1227. Good Information For Smokers: You Can Quit Smoking. Consumer booklet. U.S. Public Health Service. http://www.surgeongeneral.gov/tobacco/lowlit.htm, September 2002. Accessed December 16, 2002.
- 1228. Treating Tobacco Use and Dependence. Quick Reference Guide for Clinicians. U.S. Public Health Service. http://www.surgeongeneral.gov/tobacco/tobaqrg.htm, October 2000. Accessed December 16, 2002.
- 1229. Houston-Miller N. Smoking cessation. In: Houston-Miller N, Taylor CB, eds. Lifestyle Management for Patients with Coronary Heart Disease. Champaigne, IL: Human Kinetics; 1995:85-104.
- 1230. Glover ED, Glover PN, Payne TJ. Treating nicotine dependence. *Am J Med Sci* 2003;326:183-6.
- 1231. Fiore MC, Smith SS, Jorenby DE, Baker TB. The effectiveness of the nicotine patch for smoking cessation: a meta-analysis. *JAMA* 1994;271:1940-7.
- 1232. Gourlay SG, McNeil JJ. Antismoking products. Med J Aust

1990;153:699-707.

- 1233. Kimmel SE, Berlin JA, Miles C, Jaskowiak J, Carson JL, Strom BL. Risk of acute first myocardial infarction and use of nicotine patches in a general population. *J Am Coll Cardiol* 2001; 37:1297-302.
- 1234. Working Group for the Study of Transdermal Nicotine in Patients with Coronary Artery Disease. Nicotine replacement therapy for patients with coronary artery disease. *Arch Intern Med* 1994;154:989-95.
- 1235. Sønderskov J, Olsen J, Sabroe S, Meillier L, Overvad K. Nicotine patches in smoking cessation: a randomized trial among over-the-counter customers in Denmark. *Am J Epidemiol* 1997;145:309-18.
- 1236. Joseph AM, Norman SM, Ferry LH, et al. The safety of transdermal nicotine as an aid to smoking cessation in patients with cardiac disease. *N Engl J Med* 1996;335:1792-8.
- 1237. Mahmarian JJ, Moyé LA, Nasser GA, et al. Nicotine patch therapy in smoking cessation reduces the extent of exercise-induced myocardial ischemia. J Am Coll Cardiol 1997;30:125-30.
- 1238. Covey LS, Glassman AH. A meta-analysis of double-blind placebo-controlled trials of clonidine for smoking cessation. *Br J Addict* 1991;86:991-8.
- 1239. Bernstein DA. Modification of smoking behavior: an evaluative review. *Psychol Bull* 1969;71:418-40.
- 1240. Davison GC, Rosen RC. Lobeline and reduction of cigarette smoking. *Psychol Rep* 1972;31:443-56.
- 1241. Ford S, Ederer F. Breaking the cigarette habit. JAMA 1965;194:139-42.
- 1242. Bouvy ML, Buurma H, Egberts AC. Determinants for successful smoking cessation with bupropion in daily practice. *Pharm World Sci* 2003;25:207-11.
- 1243. Swan GE, McAfee T, Curry SJ, et al K. Effectiveness of bupropion sustained release for smoking cessation in a health care setting: a randomized trial. *Arch Intern Med* 2003;163:2337-44.
- 1244. Hurt RD, Sachs DP, Glover ED, et al. A comparison of sustainedrelease bupropion and placebo for smoking cessation. *N Engl J Med* 1997;337:1195-202.
- 1245. Gutstein DE, Fuster V. Pathophysiologic bases for adjunctive therapies in the treatment and secondary prevention of acute myocardial infarction. *Clin Cardiol* 1998;21:161-8.
- 1246. Hennekens CH, Dyken ML, Fuster V. Aspirin as a therapeutic agent in cardiovascular disease: a statement for healthcare professionals from the American Heart Association. *Circulation* 1997;96:2751-3.
- 1247. Hurlen M, Smith P, Arnesen H. Effects of warfarin, aspirin and the two combined, on mortality and thromboembolic morbidity after myocardial infarction. The WARIS-II (Warfarin-Aspirin Reinfarction Study) design. *Scand Cardiovasc J* 2000;34:168-71.
- 1248. van Es RF, Jonker JJ, Verheugt FW, Deckers JW, Grobbee DE for the Antithrombotics in the Secondary Prevention of Events in Coronary Thrombosis-2 (ASPECT-2) Research Group. Aspirin and coumadin after acute coronary syndromes (the ASPECT-2 study): a randomised controlled trial. *Lancet* 2002; 360:109-13.
- 1249. Brouwer MA, van den Bergh PJ, Aengevaeren WR, et al. Aspirin plus coumarin versus aspirin alone in the prevention of reocclusion after fibrinolysis for acute myocardial infarction: results of the Antithrombotics in the Prevention of Reocclusion In Coronary Thrombolysis (APRICOT)-2 Trial. *Circulation* 2002;106:659-65.
- 1250. Steering Committee of the Physicians' Health Study Research Group. Final report on the aspirin component of the ongoing

Physicians' Health Study. N Engl J Med 1989;321:129-35.

- 1251. Peto R, Gray R, Collins R, et al. Randomised trial of prophylactic daily aspirin in British male doctors. *Br Med J (Clin Res Ed)* 1988;296:313-6.
- 1252. Catella-Lawson F, Reilly MP, Kapoor SC, et al. Cyclo-oxygenase inhibitors and the antiplatelet effects of aspirin. *N Engl J Med* 2001;345:1809-17.
- 1253. MacDonald TM, Wei L. Effect of ibuprofen on cardioprotective effect of aspirin. *Lancet* 2003;361:573-4.
- 1254. Kurth T, Glynn RJ, Walker AM, et al. Inhibition of clinical benefits of aspirin on first myocardial infarction by nonsteroidal antiinflammatory drugs. *Circulation* 2003;108:1191-5.
- 1255. Pfeffer MA, Pfeffer JM, Steinberg C, Finn P. Survival after an experimental myocardial infarction: beneficial effects of long-term therapy with captopril. *Circulation* 1985;72:406-12.
- 1256. Pfeffer MA, Braunwald E, Moyé LA, Basta L, et al, for the SAVE Investigators. Effect of captopril on mortality and morbidity in patients with left ventricular dysfunction after myocardial infarction: results of the survival and ventricular enlargement trial. *N Engl J Med* 1992;327:669-77.
- 1257. The Acute Infarction Ramipril Efficacy (AIRE) Study Investigators. Effect of ramipril on mortality and morbidity of survivors of acute myocardial infarction with clinical evidence of heart failure. *Lancet* 1993;342:821-8.
- 1258. Køber L, Torp-Pedersen C, Carlsen JE, et al, for the Trandolapril Cardiac Evaluation (TRACE) Study Group. A clinical trial of the angiotensin-converting-enzyme inhibitor trandolapril in patients with left ventricular dysfunction after myocardial infarction. *N Engl J Med* 1995;333:1670-6.
- 1259. Johnstone D, Limacher M, Rousseau M, et al. Clinical characteristics of patients in studies of left ventricular dysfunction (SOLVD). Am J Cardiol 1992;70:894-900.
- 1260. The SOLVD Investigators. Effect of enalapril on mortality and the development of heart failure in asymptomatic patients with reduced left ventricular ejection fractions. *N Engl J Med* 1992;327:685-91.
- 1261. Rutherford JD, Pfeffer MA, Moyé LA, et al, for the SAVE Investigators. Effects of captopril on ischemic events after myocardial infarction: results of the Survival and Ventricular Enlargement trial. *Circulation* 1994;90:1731-8.
- 1262. Fox KM for the EURopean trial On reduction of cardiac events with Perindopril in stable coronary Artery disease Investigators. Efficacy of perindopril in reduction of cardiovascular events among patients with stable coronary artery disease: randomised, double-blind, placebo-controlled, multicentre trial (the EUROPA study). *Lancet* 2003;362:782-8.
- 1263. Granger CB, McMurray JJ, Yusuf S, et al, for the CHARM Investigators and Committees. Effects of candesartan in patients with chronic heart failure and reduced left-ventricular systolic function intolerant to angiotensin-converting-enzyme inhibitors: the CHARM-Alternative trial. *Lancet* 2003;362:772-6.
- 1264. McMurray JJ, Ostergren J, Swedberg K, et al, for the CHARM Investigators and Committees. Effects of candesartan in patients with chronic heart failure and reduced left-ventricular systolic function taking angiotensin-converting-enzyme inhibitors: the CHARM-Added trial. *Lancet* 2003;362:767-71.
- 1265. Yusuf S, Pfeffer MA, Swedberg K, et al, for the CHARM Investigators and Committees. Effects of candesartan in patients with chronic heart failure and preserved left-ventricular ejection fraction: the CHARM-Preserved Trial. *Lancet* 2003;362:777-81.

- 1266. The Beta-Blocker Pooling Project Research Group. The Beta-Blocker Pooling Project (BBPP): subgroup findings from randomized trials in post infarction patients. *Eur Heart J* 1988;9:8-16.
- 1267. Barron HV, Viskin S, Lundstrom RJ, Wong CC, Swain BE, Truman AF, Selby JV. Effect of beta-adrenergic blocking agents on mortality rate in patients not revascularized after myocardial infarction: data from a large HMO. *Am Heart J* 1997;134:608-13.
- 1268. Gheorghiade M, Colucci WS, Swedberg K. Beta-blockers in chronic heart failure. *Circulation* 2003;107:1570-5.
- 1269. Chen J, Radford MJ, Wang Y, Marciniak TA, Krumholz HM. Are beta-blockers effective in elderly patients who undergo coronary revascularization after acute myocardial infarction? *Arch Intern Med* 2000;160:947-52.
- 1270. Krumholz HM, Radford MJ, Wang Y, Chen J, Heiat A, Marciniak TA, National use and effectiveness of beta-blockers for the treatment of elderly patients after acute myocardial infarction: National Cooperative Cardiovascular Project. *JAMA* 1998;280:623-9.
- 1271. Hochman JS, Califf RM. Acute myocardial infarction. In: Antman EA, Colucci WS, Gotto AM, Josephson ME, Loscalzo J, Oparil S, Popma JJ, eds. Cardiovascular therapeutics: a companion to Braunwald's heart disease. 2nd ed. Philadelphia, PA: WB Saunders Co Ltd; 2002:235-91.
- 1272. Marciniak TA, Ellerbeck EF, Radford MJ, et al. Improving the quality of care for Medicare patients with acute myocardial infarction: results from the Cooperative Cardiovascular Project. *JAMA* 1998;279:1351-7.
- 1273. Gottlieb SS, McCarter RJ, Vogel RA. Effect of beta-blockade on mortality among high-risk and low-risk patients after myocardial infarction. *N Engl J Med* 1998;339:489-97.
- 1274. Goldman L, Sia ST, Cook EF, Rutherford JD, Weinstein MC. Costs and effectiveness of routine therapy with long-term betaadrenergic antagonists after acute myocardial infarction. *N Engl J Med* 1988;319:152-7.
- 1275. Brand DA, Newcomer LN, Freiburger A, Tian H. Cardiologists' practices compared with practice guidelines: use of beta-blockade after acute myocardial infarction. *J Am Coll Cardiol* 1995;26:1432-6.
- 1276. Seventh report of the Joint National Committee on the Prevention, Detection, Evaluation, and Treatment of High Blood Pressure (JNC 7): resetting the hypertension sails. *Hypertension* 2003;41:1178-9.
- 1277. ALLHAT Officers and Coordinators for the Antihypertensive and Lipid-Lowering Treatment to Prevent Heart Attack Trial (ALLHAT) Collaborative Research Group. Major outcomes in high-risk hypertensive patients randomized to angiotensin-converting enzyme inhibitor or calcium channel blocker vs diuretic: the Antihypertensive and Lipid-Lowering Treatment to Prevent Heart Attack Trial (ALLHAT). JAMA 2002;288:2981-97.
- 1278. Salpeter S, Ormiston T, Salpeter E. Cardioselective beta-blockers for reversible airway disease. Cochrane Database of Systematic Reviews (4):CD002992, 2002.
- 1279. Hennekens CH, Albert CM, Godfried SL, Gaziano JM, Buring JE. Adjunctive drug therapy of acute myocardial infarction: evidence from clinical trials. *N Engl J Med* 1996;335:1660-7.
- 1280. UK Prospective Diabetes Study (UKPDS) Group. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). *Lancet*

- 1281. Wang CH, Weisel RD, Liu PP, Fedak PW, Verma S. Glitazones and heart failure: critical appraisal for the clinician. *Circulation* 2003;107:1350-4.
- 1282. Nesto RW, Bell D, Bonow RO, et al. Thiazolidinedione use, fluid retention, and congestive heart failure: a consensus statement from the American Heart Association and American Diabetes Association. *Circulation* 2003;108:2941-8.
- 1283. Stevenson JC, Crook D, Godsland IF, Collins P, Whitehead MI. Hormone replacement therapy and the cardiovascular system: nonlipid effects. *Drugs* 1994;47:35-41.
- 1284. Kafonek SD. Postmenopausal hormone replacement therapy and cardiovascular risk reduction: a review. *Drugs* 1994;47:16-24.
- 1285. Petitti DB. Coronary heart disease and estrogen replacement therapy: can compliance bias explain the results of observational studies? *Ann Epidemiol* 1994;4:115-8.
- 1286. Whitehead M. Progestins and androgens. *Fertil Steril* 1994; 62:161S-167S.
- 1287. Hulley S, Grady D, Bush T, et al, for the Heart and Estrogen/progestin Replacement Study (HERS) Research Group. Randomized trial of estrogen plus progestin for secondary prevention of coronary heart disease in postmenopausal women. *JAMA* 1998;280:605-13.
- 1288. Ridker PM, Hennekens CH, Rifai N, Buring JE, Manson JE. Hormone replacement therapy and increased plasma concentration of C-reactive protein. *Circulation* 1999;100:713-6.
- 1289. Stanford JL, Weiss NS, Voigt LF, Daling JR, Habel LA, Rossing MA. Combined estrogen and progestin hormone replacement therapy in relation to risk of breast cancer in middle-aged women. *JAMA* 1995;274:137-42.
- 1290. Colditz GA, Hankinson SE, Hunter DJ, et al. The use of estrogens and progestins and the risk of breast cancer in postmenopausal women. N Engl J Med 1995;332:1589-93.
- 1291. Gorsky RD, Koplan JP, Peterson HB, Thacker SB. Relative risks and benefits of long-term estrogen replacement therapy: a decision analysis. *Obstet Gynecol* 1994;83:161-6.
- 1292. Herrington DM, Reboussin DM, Brosnihan KB, et al. Effects of estrogen replacement on the progression of coronary-artery atherosclerosis. *N Engl J Med* 2000;343:522-9.
- 1293. Grady D, Herrington D, Bittner V, et al, for the HERS Research Group. Cardiovascular disease outcomes during 6.8 years of hormone therapy: Heart and Estrogen/progestin Replacement Study follow-up (HERS II). *JAMA* 2002;288:49-57.
- 1294. Becker LB, Han BH, Meyer PM, et al. Racial differences in the incidence of cardiac arrest and subsequent survival: the CPR Chicago Project. N Engl J Med 1993;329:600-6.
- 1295. van Daele ME, McNeill AJ, Fioretti PM, et al. Prognostic value of dipyridamole sestamibi single-photon emission computed tomography and dipyridamole stress echocardiography for new cardiac events after an uncomplicated myocardial infarction. J Am Soc Echocardiogr 1994;7:370-80.
- 1296. Seligsohn U, Lubetsky A. Genetic susceptibility to venous thrombosis. *N Engl J Med* 2001;344:1222-31.
- 1297. Rossouw JE, Anderson GL, Prentice RL, et al, for the Writing Group for the Women's Health Initiative Investigators. Risks and benefits of estrogen plus progestin in healthy postmenopausal women: principal results from the Women's Health Initiative randomized controlled trial. JAMA 2002;288:321-33.
- 1298. Waters DD, Alderman EL, Hsia J, et al. Effects of hormone replacement therapy and antioxidant vitamin supplements on coronary atherosclerosis in postmenopausal women: a randomized controlled trial. *JAMA* 2002;288:2432-40.

- 1299. Hodis HN, Mack WJ, Azen SP, et al, for the Women's Estrogen-Progestin Lipid-Lowering Hormone Atherosclerosis Regression Trial Research Group. Hormone therapy and the progression of coronary-artery atherosclerosis in postmenopausal women. *N Engl J Med* 2003;349:535-45.
- 1300. Barrett-Connor E, Grady D, Sashegyi A, et al, for the MORE Investigators (Multiple Outcomes of Raloxifene Evaluation). Raloxifene and cardiovascular events in osteoporotic postmenopausal women: four-year results from the MORE (Multiple Outcomes of Raloxifene Evaluation) randomized trial. *JAMA* 2002;287:847-57.
- 1301. Cherry N, Gilmour K, Hannaford P, et al, for the ESPRIT team. Oestrogen therapy for prevention of reinfarction in postmenopausal women: a randomised placebo controlled trial. *Lancet* 2002;360:2001-8.
- 1302. Cairns JA, Markham BA. Economics and efficacy in choosing oral anticoagulants or aspirin after myocardial infarction. JAMA 1995;273:965-7.
- 1303. Hurlen M, Abdelnoor M, Smith P, Erikssen J, Arnesen H. Warfarin, aspirin, or both after myocardial infarction. N Engl J Med 2002;347:969-74.
- 1304. Fiore LD, Ezekowitz MD, Brophy MT, Lu D, Sacco J, Peduzzi P, for the Combination Hemotherapy and Mortality Prevention (CHAMP) Study Group. Department of Veterans Affairs Cooperative Studies Program Clinical Trial comparing combined warfarin and aspirin with aspirin alone in survivors of acute myocardial infarction: primary results of the CHAMP study. *Circulation* 2002;105:557-63.
- 1305. O'Connor CM, Gattis WA, Hellkamp AS, et al. Comparison of two aspirin doses on ischemic stroke in post-myocardial infarction patients in the warfarin (Coumadin) Aspirin Reinfarction Study (CARS). Am J Cardiol 2001;88:541-6.
- 1306. Coumadin Aspirin Reinfarction Study (CARS) Investigators. Randomised double-blind trial of fixed low-dose warfarin with aspirin after myocardial infarction. *Lancet* 1997;350:389-96.
- 1307. Herlitz J, Holm J, Peterson M, et al, for the LoWASA Study Group. Effect of fixed low-dose warfarin added to aspirin in the long term after acute myocardial infarction: the LoWASA Study. *Eur Heart J* 2004;25:232-9.
- 1308. Anticoagulants in the Secondary Prevention of Events in Coronary Thrombosis (ASPECT) Research Group. Effect of long-term oral anticoagulant treatment on mortality and cardiovascular morbidity after myocardial infarction. *Lancet* 1994;343:499-503.
- 1309. Orford JL, Fasseas P, Melby S, et al. Safety and efficacy of aspirin, clopidogrel, and warfarin after coronary stent placement in patients with an indication for anticoagulation. *Am Heart J* 2004;147:463-7.
- 1310. Spencer FA, Becker RC. When guidelines collide... *Am Heart J* 2004;147:395-7.
- 1311. Vaitkus PT, Barnathan ES. Embolic potential, prevention and management of mural thrombus complicating anterior myocardial infarction: a meta-analysis. J Am Coll Cardiol 1993; 22:1004-9.
- 1312. Keating EC, Gross SA, Schlamowitz RA, et al. Mural thrombi in myocardial infarctions: prospective evaluation by two-dimensional echocardiography. *Am J Med* 1983;74:989-95.
- 1313. Cregler LL. Antithrombotic therapy in left ventricular thrombosis and systemic embolism. *Am Heart J* 1992;123:1110-4.
- 1314. Reeder GS, Lengyel M, Tajik AJ, Seward JB, Smith HC, Danielson GK. Mural thrombus in left ventricular aneurysm: incidence, role of angiography, and relation between anticoagu-

e291 ACC/AHA Practice Guidelines

lation and embolization. Mayo Clin Proc 1981;56:77-81.

- 1315. Lip GY. Intracardiac thrombus formation in cardiac impairment: the role of anticoagulant therapy. *Postgrad Med J* 1996;72:731-8.
- 1316. Sherman DG, Dyken ML, Fisher M, Harrison MJ, Hart RG. Antithrombotic therapy for cerebrovascular disorders. *Chest* 1989;95:140S-155S.
- 1317. Weintraub WS, Ba'albaki HA. Decision analysis concerning the application of echocardiography to the diagnosis and treatment of mural thrombi after anterior wall acute myocardial infarction. *Am J Cardiol* 1989;64:708-16.
- 1318. Al-Khadra AS, Salem DN, Rand WM, Udelson JE, Smith JJ, Konstam MA. Warfarin anticoagulation and survival: a cohort analysis from the Studies of Left Ventricular Dysfunction. *J Am Coll Cardiol* 1998;31:749-53.
- 1319. Lip GY, Gibbs CR. Anticoagulation for heart failure in sinus rhythm: a Cochrane systematic review. *QJM* 2002;95:451-9.
- 1320. US Department of Health and Human Services. Physical Activity Fundamental to Preventing Disease. Office of the Assistant Secretary for Planning and Evaluation. http://aspe.hhs.gov/ health/reports/physicalactivity/, June 2002. Accessed December 16, 2002.
- 1321. Deleted in press.
- 1322. Fletcher GF, Balady G, Blair SN, et al. Statement on exercise: benefits and recommendations for physical activity programs for all Americans: a statement for health professionals by the Committee on Exercise and Cardiac Rehabilitation of the Council on Clinical Cardiology, American Heart Association. *Circulation* 1996;94:857-62.
- 1323. Pate RR, Pratt M, Blair SN, et al. Physical activity and public health: a recommendation from the Centers for Disease Control and Prevention and the American College of Sports Medicine. *JAMA* 1995;273:402-7.
- 1324. Thompson PD, Buchner D, Pina IL, et al. Exercise and physical activity in the prevention and treatment of atherosclerotic cardiovascular disease: a statement from the Council on Clinical Cardiology (Subcommittee on Exercise, Rehabilitation, and Prevention) and the Council on Nutrition, Physical Activity, and Metabolism (Subcommittee on Physical Activity). *Circulation* 2003;107:3109-16.
- 1325. Pollock ML, Franklin BA, Balady GJ, et al. AHA science advisory: resistance exercise in individuals with and without cardiovascular disease: benefits, rationale, safety, and prescription: an advisory from the Committee on Exercise, Rehabilitation, and Prevention, Council on Clinical Cardiology, American Heart Association; Position paper endorsed by the American College of Sports Medicine. *Circulation* 2000;101:828-33.
- 1326. Rimm EB, Stampfer MJ, Ascherio A, Giovannucci E, Colditz GA, Willett WC. Vitamin E consumption and the risk of coronary heart disease in men. *N Engl J Med* 1993;328:1450-6.
- 1327. Stampfer MJ, Hennekens CH, Manson JE, Colditz GA, Rosner B, Willett WC. Vitamin E consumption and the risk of coronary disease in women. *N Engl J Med* 1993;328:1444-9.
- 1328. Gey KF, Puska P, Jordan P, Moser UK. Inverse correlation between plasma vitamin E and mortality from ischemic heart disease in cross-cultural epidemiology. *Am J Clin Nutr* 1991;53:326S-334S.
- 1329. Stephens NG, Parsons A, Schofield PM, Kelly F, Cheeseman K, Mitchinson MJ. Randomised controlled trial of vitamin E in patients with coronary disease: Cambridge Heart Antioxidant Study (CHAOS). *Lancet* 1996;347:781-6.
- 1330. Kushi LH, Folsom AR, Prineas RJ, Mink PJ, Wu Y, Bostick RM.

Dietary antioxidant vitamins and death from coronary heart disease in postmenopausal women. *N Engl J Med* 1996;334:1156-62.

- 1331. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto miocardico. Dietary supplementation with n-3 polyunsaturated fatty acids and vitamin E after myocardial infarction: results of the GISSI-Prevenzione trial. *Lancet* 1999;354:447-55.
- 1332. The Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group. The effect of vitamin E and beta carotene on the incidence of lung cancer and other cancers in male smokers. *N Engl J Med* 1994;330:1029-35.
- 1333. Omenn GS, Goodman GE, Thornquist MD, et al. Effects of a combination of beta carotene and vitamin A on lung cancer and cardiovascular disease. *N Engl J Med* 1996;334:1150-5.
- 1334. Hennekens CH, Buring JE, Manson JE, et al. Lack of effect of long-term supplementation with beta carotene on the incidence of malignant neoplasms and cardiovascular disease. N Engl J Med 1996;334:1145-9.
- 1335. Enstrom JE, Kanim LE, Klein MA. Vitamin C intake and mortality among a sample of the United States population. *Epidemiology* 1992;3:194-202.
- 1336. Flaherty JT, Pitt B, Gruber JW, et al. Recombinant human superoxide dismutase (h-SOD) fails to improve recovery of ventricular function in patients undergoing coronary angioplasty for acute myocardial infarction. *Circulation* 1994;89:1982-91.
- 1337. DeMaio SJ, King SB, Lembo NJ, et al. Vitamin E supplementation, plasma lipids and incidence of restenosis after percutaneous transluminal coronary angioplasty (PTCA). *J Am Coll Nutr* 1992;11:68-73.
- 1338. Leaf A, Jorgensen MB, Jacobs AK, et al. Do fish oils prevent restenosis after coronary angioplasty? *Circulation* 1994;90: 2248-57.
- 1338a. Kris-Etherton PM, Lichtenstein AH, Howard BV, Steinberg D, Witzman JL. Antioxidant vitamin supplements and cardiovascular disease. *Circulation*. In press.
- 1339. ENRICHD Investigators. Enhancing Recovery in Coronary Heart Disease (ENRICHD) study intervention: rationale and design. *Psychosom Med* 2001;63:747-55.
- 1340. Bush DE, Ziegelstein RC, Tayback M, et al. Even minimal symptoms of depression increase mortality risk after acute myocardial infarction. *Am J Cardiol* 2001;88:337-41.
- 1341. Frasure-Smith N, Lespérance F, Gravel G, et al. Depression and health-care costs during the first year following myocardial infarction. *J Psychosom Res* 2000;48:471-8.
- 1342. Welin C, Lappas G, Wilhelmsen L. Independent importance of psychosocial factors for prognosis after myocardial infarction. J Intern Med 2000;247:629-39.
- 1343. Irvine J, Basinski A, Baker B, et al. Depression and risk of sudden cardiac death after acute myocardial infarction: testing for the confounding effects of fatigue. *Psychosom Med* 1999;61:729-37.
- 1344. Barefoot JC, Schroll M. Symptoms of depression, acute myocardial infarction, and total mortality in a community sample. *Circulation* 1996;93:1976-80.
- 1345. Ladwig KH, Lehmacher W, Roth R, Breithardt G, Budde T, Borggrefe M. Factors which provoke post-infarction depression: results from the post-infarction late potential study (PILP). *J Psychosom Res* 1992;36:723-9.
- 1346. Lane D, Carroll D, Ring C, Beevers DG, Lip GY. Mortality and quality of life 12 months after myocardial infarction: effects of depression and anxiety. *Psychosom Med* 2001;63:221-30.
- 1347. Mayou RA, Gill D, Thompson DR, Day A, Hicks N, Volmink J,

Neil A. Depression and anxiety as predictors of outcome after myocardial infarction. *Psychosom Med* 2000;62:212-9.

- 1348. Carney RM, Blumenthal JA, Catellier D, et al. Depression as a risk factor for mortality after acute myocardial infarction. *Am J Cardiol* 2003;92:1277-81.
- 1349. Crilley JG, Farrer M. Impact of first myocardial infarction on self-perceived health status. *QJM* 2001;94:13-8.
- 1350. Swenson JR, O'Connor CM, Barton D, et al, for the Sertraline Antidepressant Heart Attack Randomized Trial (SADHART) Group. Influence of depression and effect of treatment with sertraline on quality of life after hospitalization for acute coronary syndrome. *Am J Cardiol* 2003;92:1271-6.
- 1351. Friedman S. Cardiac disease, anxiety, and sexual functioning. *Am J Cardiol* 2000;86:46F-50F.
- 1352. Ziegelstein RC, Fauerbach JA, Stevens SS, Romanelli J, Richter DP, Bush DE. Patients with depression are less likely to follow recommendations to reduce cardiac risk during recovery from a myocardial infarction. *Arch Intern Med* 2000;160:1818-23.
- 1353. Lane D, Carroll D, Ring C, Beevers DG, Lip GY. Predictors of attendance at cardiac rehabilitation after myocardial infarction. *J Psychosom Res* 2001;51:497-501.
- 1354. Sertraline treatment of major depression in patients with acute MI or unstable angina. *JAMA* 2002;288:701-9.
- 1355. Berkman LF, Blumenthal J, Burg M, et al, for Writing Committee for the ENRICHD Investigators. Effects of treating depression and low perceived social support on clinical events after myocardial infarction: the Enhancing Recovery in Coronary Heart Disease Patients (ENRICHD) Randomized Trial. JAMA 2003;289:3106-16.
- 1356. Alonzo AA. Acute myocardial infarction and posttraumatic stress disorder: the consequences of cumulative adversity. J Cardiovasc Nurs 1999;13:33-45.
- 1357. Berkman LF, Syme SL. Social networks, host resistance, and mortality: a nine-year follow-up study of Alameda County residents. *Am J Epidemiol* 1979;109:186-204.
- 1358. Case RB, Moss AJ, Case N, McDermott M, Eberly S. Living alone after myocardial infarction: impact on prognosis. *JAMA* 1992;267:515-9.
- 1359. Ruberman W, Weinblatt E, Goldberg JD, Chaudhary BS. Psychosocial influences on mortality after myocardial infarction. *N Engl J Med* 1984;311:552-9.
- 1360. Frasure-Smith N, Lespérance F, Gravel G, et al. Social support, depression, and mortality during the first year after myocardial infarction. *Circulation* 2000;101:1919-24.
- 1361. Deleted in press.
- 1362. Coppotelli HC, Orleans CT. Partner support and other determinants of smoking cessation maintenance among women. J Consult Clin Psychol 1985;53:455-60.
- 1363. Suls J, Green P, Rose G, Lounsbury P, Gordon E. Hiding worries from one's spouse: associations between coping via protective buffering and distress in male post-myocardial infarction patients and their wives. *J Behav Med* 1997;20:333-49.
- 1364. Clarke DE, Walker JR, Cuddy TE. The role of perceived overprotectiveness in recovery 3 months after myocardial infarction. *J Cardiopulm Rehabil* 1996;16:372-7.
- 1365. Education, communication, and methods of intervention. In: Houston-Miller N, Taylor CB. Lifestyle Management for Patients with Coronary Heart Disease (Current Issues in Cardiac Rehabilitation, Monograph no. 2). Champaigne, IL: Human Kinetics; 1995:21-30.
- 1366. Crowe JM, Runions J, Ebbesen LS, Oldridge NB, Streiner DL. Anxiety and depression after acute myocardial infarction. *Heart*

Lung 1996;25:98-107.

- 1367. Cossette S, Frasure-Smith N, Lespérance F. Clinical implications of a reduction in psychological distress on cardiac prognosis in patients participating in a psychosocial intervention program. *Psychosom Med* 2001;63:257-66.
- 1368. Jones DA, West RR. Psychological rehabilitation after myocardial infarction: multicentre randomised controlled trial. *BMJ* 1996;313:1517-21.
- 1369. Linden W. Psychological treatments in cardiac rehabilitation: review of rationales and outcomes. J Psychosom Res 2000; 48:443-54.
- 1370. Linden W, Stossel C, Maurice J. Psychosocial interventions for patients with coronary artery disease: a meta-analysis. *Arch Intern Med* 1996;156:745-52.
- 1371. Feigenbaum E, Carter E. Health Technology Assessment Report No. 6. US Department of Health and Human Services. Public Health Service. National Center for Health Services, Research, and Health Care Technology Assessment. Publication No. PHS 883427, 1988.
- 1372. Balady GJ, Ades PA, Comoss P, et al. Core components of cardiac rehabilitation/secondary prevention programs: a statement for healthcare professionals from the American Heart Association and the American Association of Cardiovascular and Pulmonary Rehabilitation Writing Group. *Circulation* 2000;102:1069-73.
- 1373. Jolliffe JA, Rees K, Taylor RS, Thompson D, Oldridge N, Ebrahim S. Exercise-based rehabilitation for coronary heart disease. Cochrane Database Syst Rev 2001 CD001800.
- 1374. DeBusk RF, Miller NH, Superko HR, et al. A case-management system for coronary risk factor modification after acute myocardial infarction. *Ann Intern Med* 1994;120:721-9.
- 1375. Barber K, Stommel M, Kroll J, Holmes-Rovner M, McIntosh B. Cardiac rehabilitation for community-based patients with myocardial infarction: factors predicting discharge recommendation and participation. J Clin Epidemiol 2001;54:1025-30.
- 1376. Spencer FA, Salami B, Yarzebski J, Lessard D, Gore JM, Goldberg RJ. Temporal trends and associated factors of inpatient cardiac rehabilitation in patients with acute myocardial infarction: a community-wide perspective. *J Cardiopulm Rehabil* 2001;21:377-84.
- 1377. Frasure-Smith N, Lespérance F, Talajic M. Depression and 18month prognosis after myocardial infarction. *Circulation* 1995;91:999-1005.
- 1378. O'Connor GT, Buring JE, Yusuf S, Goldhaber SZ, Olmstead EM, Paffenbarger RS, Hennekens CH. An overview of randomized trials of rehabilitation with exercise after myocardial infarction. *Circulation* 1989;80:234-44.
- 1379. Lavie CJ, Milani RV. Effects of cardiac rehabilitation and exercise training programs in patients greater than or equal to 75 years of age. Am J Cardiol 1996;78:675-7.
- 1380. Ziegelstein RC. Depression in patients recovering from a myocardial infarction. *JAMA* 2001;286:1621-7.
- 1381. Rost K, Smith GR. Return to work after an initial myocardial infarction and subsequent emotional distress. *Arch Intern Med* 1992;152:381-5.
- 1382. Froelicher ES, Kee LL, Newton KM, Lindskog B, Livingston M. Return to work, sexual activity, and other activities after acute myocardial infarction. *Heart Lung* 1994;23:423-35.
- 1383. Lewin R. Return to work after MI, the roles of depression, health beliefs and rehabilitation. *Int J Cardiol* 1999;72:49-51.
- 1384. Brodie B, Grines CL, Spain M, et al. A prospective, randomized trial evaluating early discharge (day 3) without non-invasive

risk stratification in low risk patients with acute myocardial infarction: PAMI-2. J Am Coll Cardiol 1995;25:5A.

- 1385. Petrie KJ, Cameron LD, Ellis CJ, Buick D, Weinman J. Changing illness perceptions after myocardial infarction: an early intervention randomized controlled trial. *Psychosom Med* 2002;64:580-6.
- 1386. Ostir GV, Goodwin JS, Markides KS, Ottenbacher KJ, Balfour J, Guralnik JM. Differential effects of premorbid physical and emotional health on recovery from acute events. *J Am Geriatr Soc* 2002;50:713-8.
- 1387. Sansone GR, Alba A, Frengley JD. Analysis of FIM instrument scores for patients admitted to an inpatient cardiac rehabilitation program. *Arch Phys Med Rehabil* 2002;83:506-12.
- 1388. Froom P, Cohen C, Rashcupkin J, et al. Referral to occupational medicine clinics and resumption of employment after myocardial infarction. *J Occup Environ Med* 1999;41:943-7.
- 1389. Boudrez H, De Backer G. Recent findings on return to work after an acute myocardial infarction or coronary artery bypass grafting. *Acta Cardiol* 2000;55:341-9.
- 1390. Mittag O, Kolenda KD, Nordman KJ, Bernien J, Maurischat C. Return to work after myocardial infarction/coronary artery bypass grafting: patients' and physicians' initial viewpoints and outcome 12 months later. *Soc Sci Med* 2001;52:1441-50.

- 1391. Kavanagh T, Matosevic V, Thacker L, Belliard R, Shephard RJ. On-site evaluation of bus drivers with coronary heart disease. J Cardiopulm Rehabil 1998;18:209-15.
- 1392. Covinsky KE, Chren MM, Harper DL, Way LE, Rosenthal GE. Differences in patient-reported processes and outcomes between men and women with myocardial infarction. *J Gen Intern Med* 2000;15:169-74.
- 1393. Haskell WL. Rehabilitation of the coronary patient. In: Wenger NK, Hellerstein HK, eds. Design and implementation of cardiac conditioning program. New York, NY: Churchill Livingstone, 1978:147.
- 1394. Usher MC, Dennis CA, Schwartz RG, Ahn DK, DeBusk RF. Physician influences on timing of return to work after myocardial infarction. *Circulation* 1986;74:II-490.
- 1395. US Department of Transportation. Status of Medical Review in Driver Licensing: Policies, Programs and Standards, 1992.
- 1396. Code of Federal Regulation 14CFR 121. Pg 427, Section 25.841.Published by the office of Federal Register, January 1, 2002.
- 1397. Deleted in press.
- 1398. Code of Federal Regulation 14 CFR 121. Appendix A:555-7. Published by the office of the Federal Register, January 1, 2002.